

Documents of the World Administrative Radio Conference for the Planning of the Broadcasting-Satellite Service in Frequency Bands 11.7-12.2 GHz (Regions 2 and 3) and 11.7-12.5 GHz (Region 1) (WARC SAT-77)

(Geneva, 1977)

To reduce download time, the ITU Library and Archives Service has divided the conference documents into sections.

- This PDF includes Document No. 301 388.
- The complete set of conference documents includes Document No. 1 388, DL No. 14 50 (incomplete), DT No. 1 53.

This electronic version (PDF) was scanned by the International Telecommunication Union (ITU) Library & Archives Service from an original paper document in the ITU Library & Archives collections.

La présente version électronique (PDF) a été numérisée par le Service de la bibliothèque et des archives de l'Union internationale des télécommunications (UIT) à partir d'un document papier original des collections de ce service.

Esta versión electrónica (PDF) ha sido escaneada por el Servicio de Biblioteca y Archivos de la Unión Internacional de Telecomunicaciones (UIT) a partir de un documento impreso original de las colecciones del Servicio de Biblioteca y Archivos de la UIT.

(ITU) للاتصالات الدولي الاتحاد في والمحفوظات المكتبة قسم آجر اه الضوئي بالمسح تصوير نتاج (PDF) الإلكترونية النسخة هذه والمحفوظات المكتبة قسم في المتوفرة الوثائق ضمن أصلية ورقية وثقِقة من نقلاً

此电子版(PDF版本)由国际电信联盟(ITU)图书馆和档案室利用存于该处的纸质文件扫描提供。

Настоящий электронный вариант (PDF) был подготовлен в библиотечно-архивной службе Международного союза электросвязи путем сканирования исходного документа в бумажной форме из библиотечно-архивной службы МСЭ.

BROADCASTING SATELLITE CONFERENCE

(Geneva, 1977)

Document No. 301-E 11 February 1977 Original : English

PLENARY MEETING

FINAL PROTOCOL

For the Republic of Kenya

In signing the Final Acts of the World Administrative Broadcasting-Satellite Conference, Geneva 1977, the delegation of the Republic of Kenya, reserves its Government's right to take any measures it may deem necessary to protect its interests if other countries or administrations fail to observe the provisions contained in the Final Acts and its Annexes and the Protocol attached to it, as adopted by the Conference.

Genèn

Corrigendum N° 1 au Document N° 302-F/E/S 11 février 1977

SEANCE PLENIERE PLENARY MEETING SESION PLENARIA

R.3(Corr.1)

...

<u>CORRIGENDUM A LA</u> <u>3ème SERIE DE TEXTES SOUMISE PAR LA COMMISSION DE</u> REDACTION A LA SEANCE PLENIERE

Remplacer les pages 19, 20, 21 et 22 du Document $\operatorname{N}^{\operatorname{O}}$ 302 par les pages ci-annexées.

CORRIGENDUM TO THE

<u>3rd SERIES OF TEXTS SUBMITTED BY THE</u> EDITORIAL COMMITTEE TO THE PLENARY MEETING

Replace the pages 19, 20, 21 and 22 to Document No. 302 by the following pages.

<u>CORRIGÉNDUM A LA</u> <u>3.^a SERIE DE TEXTOS SOMETIDOS POR LA COMISIÓN DE</u> <u>REDACCIÓN AL PLENO DE LA CONFERENCIA</u>

Sustitúyanse las páginas 19, 20, 21 y 22 del Documento N $^{\circ}$ 302 por las adjuntas.



Annexes : 4 pages

PINK PAGES

<u>Corrigendum N^O 1 au</u> <u>Document N^O 302-F/E/S</u> Page 19

TABLEAU 1 - TABLE 1 - CUADRO 1

.

Symboles de pays	Zone hydro- météorologique	Symboles de pays	Zone hydro- météorologiques	Symboles de pays	Zone hydro- météorologiques
Country symbol	Rain-climatic zone	Country symbol	Rain-climatic zone	Country symbol	Rain-climatic zone
Símbolo del país	Zona hidrome- tereológica	Símbolo del país	Zona hidrome- tereológica	Símbolo del país	Zona hidrome- tereológica
ADL	1 _7	BHR	5	/ CPV	5_7
AFG	5	<u>/</u> віо	· 5 _7	CRO	5
AFI	5	BLR	2	CTI	1
		BOL*)		CTR	1
		DOW		CUB	1
AGL	1	BOIL	1 5	CVA	4
ALB	4	BRB	. 1	CYP	4
ALG	*)	BRM	1	D	3
ALS	2	BRU	1	DAH	1
/AMS	> _/	BUL	- 1	(BENIN)	
AND	3	CAF*)		DDR	3
AOE	5	CAN	2	DNK	*)
ARG	5		5	DOM	1
ARS	5	∠ CAR	5_7	E	4
ASC	5	CBG	1	EGY	5 -
AS0	1 5	CHL	*)	EQA	1
ልጥእ		CHN	*)	ETH	1
AUS	± 8)		-34 S	ਸ	3
1.01		/ CHR	5 7	/ FJT	1 7
		CKH	1	/ FLK	5 7
AUT	3	CKN	1	FNT.	2
AZR	3	CLM	1	G	3
В	1	CLN	1	CAB	1
BAH	1	CME	1	GDI.	1
BDI	1	CNR	5	СНА	1
BEL	3	COG	1	CTR	÷ ь
BER	1	СОМ	-		7
BGD	1		-	7 GTP	÷ -′

*) voir renseignements complémentaires dans le Tableau 2

*) see additional information in Table 2.

*) véase información complementaria en el Cuadro 2

R.3(Corr.1)

Corrigendum N^O 1 au Document N^O 302-F/E/S

Page 20

					1
Symboles de pays	Zone hydro- météorologiques	Symboles de pays	Zone hydro- météorologiques	Symboles de pays	Zone hydro- météorologiques
Country symbol	Rain-climatic zone	Country symbol	Rain-climatic zone	Country symbol	Rain-climatic zone
Símbolo del país	Zona hidrome- tereológica	Símbolo del país	Zona hidrome- tereológica	Símbolo. del país	Zona hidrome- tereológica
GMB	- 1	/ ⁻ IWA	1 7	MLT	4
GNE	1	_ _	2	MNG	5
GNP	1	/ JAR	. 5 7	MOZ	1
GRC		JMC	1	/ MRA	1_7
GRL	2	/ JON	5_7	MRC	24
GTM	- 1	JOR	5	_	5
GIB	1	KEN	1	/ MRL	1_7
GUE	1	/ KER	57	/ MRN	1_7
CUT	1	- KOR	2	MRT	1
	1 7	KRE	· 2	MTN	5
7 COM	± _/	KWT	5	MWI	1
HNB	1	LAO	1	MYT	l
HND	1	LBN	5	NCG	1
HNG	4	LBR	1	NCL	1
HOL	3	LBY	5	NGR	1
HTI	1	LUI	3	NHB	3
hvo	1	ISO	1	NIG	1
/HWA	1_7			NTU	1
/ HWL	5_7	TOX	2	NOR	±
I	4	MAC	1	NON	*)
ICO	-	MAU			
IND	1	MCO	4 7	/ NRU	> _/
INP	l	/ MCS	5 _/	NZL	. 3
INS	ı	MDG		OCE	1
IOB	1	MDR	>	OMA	2
IRL	3	∠ MDW	⊥ - ∕	PAK	*)
IRN	5	MEX	*)	PAQ .	5
IRQ	5	MT.A	, I	PHL	
ISL	3	MT.D	,	/ PHX	$\begin{array}{c} 1 \\ -7 \\ -7 \end{array}$
ISR	5	MT.T	1	/ PLM	5 _/
		PILL	5	PNG	1
) voir ren	seignements com	plémentaires	dans le Tablea	u 2	

see additional information in Table 2 *) véase información complementaria en el Cuadro 2 *****)

PINK PAGES Corrigendum Nº 1 au Document Nº 302-F/E/S Page 21

1. I.					
Symboles de pays	Zone hydro- météorologiques	Symboles de pays	Zone hydro- météorologiques	Symboles de pays	Zone hydro- météorologiques
Country symbol	Rain-climatic zone	Country symbol	Rain-climatic zone	Country symbol	Rain-climatic zone
Símbolo del país	Zona hidrome- tereológica	Símbolo del país	Zona hidrome- tereológica	Símbolo del país	Zona hidrome- tereológica
PNR	1	SPI	1		1 7
PNZ		SUD	1	VTN	1
POL.	3	SIIT	÷ 3	/ WAK	1 7
POR	3	SUR	1	WAL	1 -
500	-8-	/SUN	5 7	YEM	5
PRG	-	SW7	· -/ 1	YMS	5
PRU	2	SYR	- 5	YUG	1
/PTC	1 7	TCD	1.5	ZAI	ı
- PTR	1	тсн	-, ,	ZAN	1
QAT	5	тск	1	ZMB	ı
REU	l	TGO	1		·* •
RHS	1	тна	1		
		TKT.	1		
ROU	3	TMP	1	1	
RRW	1	/ TON	1 7		
RYU	2	/ TRC	5 7		
S	2	TRD	1		
	- 20-	TUN	<u>1</u> 4		
SDN	*)	TUR	4		
SEN	1	/ TUV	5 7		
SEV	57	UAE	5		
/SHN	5 7	UGA -	1	5 S O	
/SIM	1 7	UKR	2		
SLV	-'	URG	1		
	1 7	URS	2		
SMO	1			- 40	
SMR	4	USA	1		
SIG	1		3	1	
SOM	5		ц 5		
SPM	2	VEN	1		
N		• • • • • • • • • • • • • • • • • • • •			

*) see additional information in Table 2
*) véase información complementaria en el Cuadro 2

3

÷ '

.

-

<u>Corrigendum N^O l au</u> Document N^O 302-F/E/S

Page 22

....

<u>Tableau 2</u>

Table 2

Cuadro 2

Zones hydrométéorologiques

Rain-climatic zones

Zonas hidrometereológicas

		473111 7 1 0	Tuformonión	acomplementaria
Informations	supplementaires.	Additional informat	ion. Información	comprementar ra.

-	Pays Country País	IFRB No**)	Zone Zona	Pays Country País	IFRB No.	Zone Zona
	ALG	0250 0251 0252	4 4 5	CHN (cont.)	0165 0166 0167 0168	5 2 2 2
	AUS	0004 0005 0007 0008 0009 0010 0011 0012 0013 0014 0015 0315	4 1 5 4 1 4 1 5 4 1 1 4		0169 0170 0171 0172 0173 0174 0175 0176 0177 0178 0179 0180 0181 0182	1 1 1 1 1 2 1 1 1 1 1 2
	:	0316 0317 0318 0319 0320	1 5 4 1 1	00	0183 0184 0185 0186 0187	5 5 1
	BOL	0019 0273	1 5		0188	1
	CAF	0258	1	DNK	0090 0091	2 3
	CHL	0208 0263 0264	5 2 2	MEX	0190 0191	3 1
	CHN	0154	2 5 5	NOR	0120 0121	3 2
		0155 0156 0157 0158 0159 0160	51212	PAK	0127 0210 0281 0282 0283	2 3 5 4 2
		0161 0162 0163 0164	1 2 2	SDN	0230 0231 0232	1 3,4 5
•	¶₹) voir le Pla	an i				$\mathbf{p} \cdot \mathbf{a}/c$

see the Plan

véase el Plan

PINK PAGES

Document No. 302-E 11 February 1977

PLENARY MEETING

R.3

3rd SERIES OF TEXTS SUBMITTED BY THE

EDITORIAL COMMITTEE TO THE PLENARY MEETING

The following texts are submitted to the Plenary Meeting for second reading :

	Title	Document No.	Source
	Annex 3	232	в.б
	Annex 5	231	B.5
7_7	Annex /) 224(Rev.1)	B.3(Rev.1)

Miss M. HUET

Chairman of the Editorial Committee

<u>Annexes</u> : 35 pages

.



ANNEX 3

METHOD FOR DETERMINING THE LIMITING INTERFERING POWER FLUX DENSITY AT THE EDGE OF A BROADCASTING-SATELLITE SERVICE AREA IN THE BAND 11.7 - 12.2 GHz (IN REGIONS 2 AND 3) AND 11.7 - 12.5 GHz (IN REGION 1) AND FOR PREDICTING THE POWER FLUX DENSITY PRODUCED THERE BY A TERRESTRIAL STATION

1. General

1.1 This Annex describes a method of assessing the interference potential from terrestrial transmitters to broadcasting-satellite receivers in the band 11.7 - 12.2 GHz (12.5 GHz in Region 1).

1.2 The method is in two parts :

- the calculation of the maximum permissible interfering power flux density at the edge of the broadcasting-satellite service area concerned;
- ii) the calculation of the likely power flux density produced at any point on the edge of the service area by the terrestrial transmitter of another administration.

1.3 The interference potential of the terrestrial transmitters must be considered case by case; the power flux density produced by each terrestrial transmitter is compared to the limiting power flux density at any point on the edge of the service area of a broadcasting-satellite station of another administration. If, for a given transmitter, the value of the power flux density produced is lower than the value of the limiting power flux density at any point on the edge of the service area, the interference caused to the broadcasting-satellite service by this transmitter is considered to be lower than the permissible value and no coordination is required between administrations before the terrestrial service is brought into use. Where this is not the case, coordination and further, more precise calculations derived from a mutually agreed basis are necessary. 1.4 It is emphasized that, should the calculation described in this Annex indicate that the maximum permissible power flux density is exceeded, it does not necessarily preclude the introduction of the terrestrial service since the calculations are necessarily based on worst case assumptions for :

- a) the nature of the terrain of the interference path,
- b) the off-beam discrimination of the broadcasting-satellite receiving installations,
- c) the necessary protection ratios for the broadcasting-satellite service,
- d) the type of reception in the broadcasting-satellite service, i.e., assuming individual reception, this being more critical than community reception for the angles of elevation concerned.
- e) the value of power flux density to be protected in the broadcastingsatellite service;
- f) the propagation conditions between the terrestrial station and the broadcasting-satellite service area.

2. Limit of power flux density

2.1 General

The limiting power flux density not to be exceeded at the edge of the service area in order to protect the broadcasting-satellite service of an administration is given by the formula :

$$\mathbf{F} = \mathbf{F} - \mathbf{R} \diamond \mathbf{D} + \mathbf{P}$$

where

F = the maximum permissible interfering power flux density (dBW/m²) in the broadcasting-satellite wanted bandwidth

- $\mathbf{F}_{o} =$ the wanted power flux density (dBW/m²) at the edge of the service area
- **R** = the protection ratio (dB) between the wanted and interfering signals
- **D** = angular discrimination (dB) provided by the radiation pattern of the satellite broadcasting receiver antenna
- P = polarization discrimination (dB) between the wanted and interfering signals

The value of F_0 is equal to

- a) -103 dBW/m² for service areas in Regions 1 and 3
- b) -105 dBW/m² for service areas in Region 2

(1)

2.3 <u>Protection ratio</u> (R)

2.3.1 The single entry protection ratio against all types of terrestrial transmissions, with the exception of amplitude-modulation multichannel television systems, is 35 dB for carrier frequency differences between the wanted and interfering signals of up to $\frac{+}{-}$ 10 MHz, decreasing linearly from 35 dB to 0 dB for carrier frequency differences between 10 MHz and 35 MHz, and is 0 dB for frequency differences in excess of 35 MHz (see Figure 1).

2.3.2 The carrier frequency difference should be determined by reference to the frequency assignments in the broadcasting-satellite plan or, in the case of assignments not contained within a plan, by reference to the description of the characteristics of the proposed or operational system. For amplitude-modulation multichannel television systems which produce peaks of high power flux density spread over a wide range of their necessary bandwidth, the protection ratio R is 35 dB and is independent of the carrier frequency difference.

2.3.3 A signal from a terrestrial station should be considered only if its necessary bandwidth overlaps the necessary bandwidth of the broadcasting-satellite assignment.

2.4 Angular discrimination (D)

2.4.1 Broadcasting-satellite service areas in Regions 1 and 3

Where the angle of elevation ϕ selected for the proposed or operational broadcasting-satellite system for the broadcasting-satellite service area concerned is equal to or greater than 19°, the value of D to be assumed in expression (1) is 33 dB. When ϕ is less than 19°, D should be derived from the expression (2.a) below.

<u>Note</u>: If more than one value of ϕ is specified for a particular service area, the appropriate value of ϕ should be used for each section of the edge of the service area under consideration.

 $D = 0 \text{ for } 0 \le \phi \le 0.5^{\circ}$ $D = 3\phi^{2} \text{ for } 0.5^{\circ} \le \phi \le 1.41^{\circ}$ $D = 3 + 20 \log_{10} \phi \text{ for } 1.41^{\circ} \le \phi \le 2.52^{\circ}$ $D = 1 + 25 \log_{10} \phi \text{ for } 2.52^{\circ} \le \phi \le 19^{\circ}$ (2.a)

Note : For the graphical determination of D see Figure 2.

2.4.2 <u>Broadcasting-satellite</u> service areas in Region 2

Where the angle of elevation ϕ , selected for the proposed or operational broadcasting-satellite system, for the broadcasting-satellite service area concerned, is equal to or greater than 27°, the value of D to be assumed in expression (1) is 38 dB. When ϕ is less than 27°, D should be derived from expression (2.b) below.

<u>Note</u> : If more than one value of ϕ is specified for a particular service area, then the appropriate value of ϕ should be used for each section of the edge of the service area under consideration.

 $D = 0 \text{ for } 0 \le \phi \le 0.45^{\circ}$ $D = 3.7\phi^{2} \text{ for } 0.45^{\circ} \le \phi \le 1.27^{\circ}$ $D = 3.9 + 20 \log_{10} \phi \text{ for } 1.27^{\circ} \le \phi \le 2.27^{\circ}$ $D = 2.1 + 25 \log_{10} \phi \text{ for } 2.27^{\circ} \le \phi \le 27^{\circ}$

Note: For the graphical determination of D see Figure 2.

2.5

Polarization discrimination (P)

The value of P is equal to :

a) 3 dB when the interfering terrestrial service uses linear polarization and the broadcasting-satellite service uses circular polarization or vice versa.

b) 0 dB when the interfering terrestrial service and the broadcastingsatellite service both use circular or both use linear polarization.

Power flux density produced by a terrestrial station (F_p)

The power flux density F_p (in dBW/m²) produced at any point on the edge of the service area by the terrestrial station is determined from the following formula :

$$\mathbf{F}_{n} = \mathbf{E} - \mathbf{A} + \mathbf{43}$$

(3)

(2.b)

where E = the equivalent isotropically radiated power (dBW) of the terrestrial station in the direction of the point on the edge of the service area concerned

and A = the total path loss in dB.

R.3

3.1 <u>Evaluation of path loss A for a terrestrial station at a distance</u> greater than 100 km from the edge of the service area of the broadcasting-satellite

For path lengths greater than 100 km, A is given by :

$$A = 137.6 + 0.2324 d_{+} + 0.0814 d_{m}$$

where d_t and d_m are the overland and oversea path lengths respectively, in km.

3.2

Evaluation of path loss A for a terrestrial station at a distance equal to or less than 100 km from the edge of the service area of the broadcasting-satellite

For path lengths equal to or less than 100 km, A is calculated using equations (4) and (5) and the lower value obtained is substituted in formula (3) to calculate the power flux density produced at the point on the edge of the service area :

$$A = 109.5 + 20 \log (d_t + d_m)$$

.

The variation in A for different path lengths and percentage of oversea path is shown in Figure 3.

3.3 Distance beyond which the method need not be applied

The method need not be applied and coordination is unnecessary when the distance between the terrestrial station and the service area of the broadcasting-satellite is greater than :

a) 400 km in the case of all overland paths, or

b) 1200 km in the case of all oversea or mixed paths.

RЗ

PINK PAGES

(4)

(5)



FIGURE 1

Protection ratio R (dB) for a broadcasting-satellite signal against a single entry of interference from a terrestrial service (except for AM multichannel TV systems)

R.3







For service areas in Regions 1 and 3, $\phi_0 = 2^{\circ}$ and Curve A applies For service areas in Region 2, $\phi_0 = 1.8^{\circ}$ and B applies





R.3

.4

4

.

.

ANNEX 5

SHARING CRITERIA BETWEEN SERVICES

1. Protection requirements for sharing between services in the 12 GHz band

1.1 The establishment of sharing criteria for the different services using the 12 GHz band should be based on the protection requirements listed in the table below :

PINK PAGES Document No. 302-E Page 11

Wanted	Wanted	Interfering	Interfering	Protection requirements ²⁾		
service ¹)	signal ¹)	service ¹ /	signal ¹	Total acceptable3)	Single entry	
BSS -	TV/FM	BSS, FSS, FS, BS	TV/FM	$C/I = 30 dB^{4})7)$	$C/I = 35 \ dB^4)$	
FSS	FDM/FM	BSS	TV/FM	$N = 500 \text{ pWOp}^{8}$	N = 300 pWOp	
FSS	TV/FM	BSS, FSS	TV/FM	$C/I = 32 \text{ dB}^{5}$	$C/I = 37 dB^{5})$	
FSS	4ø – PSK	BSS, FSS	TV/FM	C/I = 30 dB	C/I = 35 dB	
FSS	FDM/FM	FSS	FDM/FM	N = 1000 pWOp	N = 400 pWOp	
FS	FDM/FM	BSS	TV/FM	N = 1000 pWOp	-125 dBW/m ² / 4 _{kHz} 6)	
BS	TV/VSB	BSS	TV/FM	C/I = 50 dB	not applicable	

Notes : 1) BSS = broadcasting-satellite service

- FSS = fixed-satellite service
- BS = broadcasting service
- FS = fixed service
- TV = television
- FM = frequency modulation
- FDM = frequency division multiplex
- 4Ø-PSK = four-level phase shift keying VSB = vestigial sideband
- 2) These limits include both up-link and down-link contributions. They are expressed :
 - in dB for carrier-to-interference ratio;
 - in pWOp for noise
 - in dBW/m2/4 kHz for power flux density in a 4 kHz band.
- J) Values in dB are protection ratios for the sum of interfering signals.
 Values in pWOp represent interference noise in the worst telephone channels caused by the sum of interfering signals.
- 4) For BSS satellites located at the interfaces of Regions 1/3 and Region 2, the C/I ratios should be 1 dB higher.
- 5) See CCIR Recommendation 483.
- 6) This value may be suitably modified for tropical regions to take account of rain attenuation. Allowance may also be made for polarization discrimination.
- 7) C/I = ratio of carrier-to-interfering signal
- 8) N = noise power

1.2 The values given as "total acceptable" are those necessary to protect the wanted signal. The "single entry" values are those which should be used as a guide for determining sharing criteria. The total interference from all sources must be calculated, since satisfying the "single entry" criteria for each source may not guarantee that the total interference meets the above protection requirements. A "single entry" is defined as the aggregate of emissions from any one station entering any receiver in the wanted service within the channel to be protected.

1.3 The term C/I refers to the ratio of interfered with to interfering power at the interfered with ground station. The value given shall be exceeded for all but 20 % of the worst month for the fixed-satellite service (FSS), and for all but 1 % of the worst month for the broadcasting service (BS) and the broadcasting-satellite service (BSS).

1.4 The term N refers to the post-demodulation noise power at a point of 0 dBmO relative test tone level in any voice channel of an FDM/FM telephony system. The value given shall not be exceeded for more than 20 % of the worst month.

1.5 The specified values of protection ratio (i.e., the carrier-tointerference power ratio corresponding to a specified picture quality) are applicable, for planning purposes, to television signals of any of the several television standards.

1.6 For BSS systems with FM/TV as the wanted signal, the protection ratios are given for particular reference conditions, the most important of which are :

- a) frequency deviation of the wanted signal (12 MHz peak-to-peak)
- b) quality of the wanted service (grade 4.5)1)
- c) co-channel carriers (no carrier-frequency offset).

1.7 If system design is based on conditions other than a) and b) above, the FM/TV protection ratio is given by :

 $PR = 12.5 - 20 \log (Dv/12) - Q + 1.1 Q^2$ (dB)

where

\$

5

Dv = nominal peak-to-peak frequency deviation (MHz)Q = the impairment grade, concerning the interference only.

1.8 When carriers are offset in frequency, condition c) does not apply and the adjacent channel protection ratios should be adjusted according to the frequency offset as shown in Figure 1. For example, at a frequency offset of 20 MHz, the total acceptable ratio of protection against interference to an FM/TV signal from another FM/TV signal is 13 dB. The corresponding "single entry" value is 18 dB. : ::

¹⁾ Impairment grade on a 5-point scale as defined in CCIR Recommendation 500.



Carrier-frequency offset (MHz)

 $\Delta f = (f_{int.} - f_{wanted})$

Figure 1 - Reference case protection ratios relative to co-channel values

A. TV/VSB-wanted, TV/FM interfering

B. TV/FM-wanted, TV/FM interfering

C. TV/FM-wanted, TV/VSB interfering

2. <u>Reference antenna diameter for a fixed-satellite earth station to be</u> <u>used in calculating interference from space stations in the</u> broadcasting-satellite service

2.1 For antennae larger than 100 λ (2.5 m) in the fixed-satellite service, the gain of the sidelobes is given by the equation $32 - 25 \log \theta$ where θ is the angle from the boresight (CCIR Recommendation 465). The sidelobe gain is independent of antenna diameter.

2.2 However, in the case of transmitting earth stations, the level of interference radiated into the up-link of other satellite systems would be inversely proportional to the square of the antenna diameter. In this case, the interference decreases with increasing antenna diameter. Since the 11.7 - 12.2 GHz band is only assigned in the space-to-earth direction in the fixed-satellite service, this point is not of direct concern to the broad-casting-satellite service.

2.3 Hence it does not appear appropriate, for antenna diameters greater than 100 λ , to specify a minimum antenna diameter for receiving earth stations in the fixed-satellite service sharing the band 11.7 - 12.2 GHz. It may be useful to consider a 4.5 m antenna having an efficiency of 60 % and an on-axis gain of 53 dB as typical for the purpose of planning the sharing of this band; however, it should be noted that administrations in Region 2 are considering the use of antennae 3 m to 10 m in diameter.

Use of energy dispersal in the broadcasting-satellite service

3.1 Artificial energy dispersal is useful in promoting sharing between the broadcasting-satellite service and the other services to which the band is also allocated.

3.2 Such energy dispersal is achieved by the addition at baseband of a triangular waveform to the video signal to form a composite baseband which, in turn; is used to frequency-modulate the up-link carrier. The frequency of the triangular waveform is usually synchronized at a sub-multiple of the television frame frequency. Typical frequencies range from 12.5 Hz to 30 Hz.

3.3 The table below gives the relative reduction in spectral power flux density in a 4 kHz bandwidth as a function of the peak-to-peak deviation due to the energy dispersal signal. This table is based on the following equation :

Relative reduction (in dB) = $10 \log \frac{\Delta F_{pp} + \delta f_{rms}}{4}$

where

3.

 ΔF_{pp} = peak-to-peak deviation due to the energy dispersal signal (kHz)

 δf_{rms} = rms deviation due to "natural" energy dispersal (kHz)

In compiling the table below, a value of 40 kHz has been assumed for δf_{rms} , on the basis of the value of 10 dB for "natural" dispersion given in Table 1 of CCIR draft Report 631(Rev.76).

Peak-to-peak deviation (kHz)	Relative reduction (dB)
0 100 200 300 400 500 600 - 700 800 900 1 000	10 15.44 17.78 19.29 20.41 21.30 22.04 22.67 23.22 23.71 24.15

Reduction of spectral power flux density relative to a 4 kHz bandwidth

3.4 The value of energy dispersal for the broadcasting-satellite service has been determined such that the spectral power flux density measured in a 4 kHz bandwidth is reduced by 22 dB relative to that measured in the entire bandwidth; this reduction corresponds to a peak-to-peak deviation of 600 kHz.

R.3

ANNEX / 7 / 7

TECHNICAL DATA USED IN ESTABLISHING THE PROVISIONS AND ASSOCIATED PLAN AND WHICH SHOULD BE USED FOR THEIR APPLICATION

1. <u>DEFINITIONS</u>

1.1 <u>Service area</u>

The area on the surface of the earth in which the administration responsible for the service has the right to demand that the agreed protection conditions be provided.

<u>Note</u>: In the definition of service area, it is made clear that within the service area the agreed protection conditions can be demanded. This is the area where there should be at least the wanted power flux density and protection against interference based on the agreed protection ratio for the agreed percentage of time should be achieved.

1.2 Coverage area

The area on the surface of the earth delineated by a contour of a constant given value of power flux density which would permit the wanted quality of reception in the absence of interference.

<u>Note 1</u>: In accordance with the provisions of No. 428A of the Radio Regulations, the coverage area must be the smallest area which encompasses the service area.

<u>Note 2</u>: The coverage area, which will normally encompass the entire service area, will result from the intersection of the antenna beam (elliptical or circular) with the surface of the earth, and will be defined by a given value of power flux density. For example, in the case of a Region 1 or 3 country with a service planned for individual reception, it would be the area delineated by the contour corresponding to a level of -103 dBW/m² for 99 % of the worst month. There will usually be an area outside the service area but within the coverage area in which the power flux density will be at least equivalent to the minimum specified value; however, protection against interference will not be provided in this area.

1.3 Beam area

The area delineated by the intersection of the half-power beam of the satellite transmitting antenna with the surface of the earth.

<u>Note</u>: The beam area is simply that area on the earth's surface corresponding to the -3 dB points on the satellite antenna radiation pattern. In many cases the beam area would almost coincide with the coverage area, the discrepancy being accounted for by the permanent difference in path lengths from the satellite throughout the beam area, and also by the permanent variations, if

R.3

PINK PAGES Document No. 302-E Page 17

any, in propagation factors across the area. However, for a service area where the maximum dimension as seen from the satellite position is less than 0.6° (the agreed minimum practicable satellite antenna half-power beamwidth), there could be a significant difference between the beam area and the coverage area.

1.4 Nominal orbital position

The longitude of a position in the geostationary satellite orbit associated with a frequency assignment to a space station in a space radiocommunication service. The position is given in degrees from the Greenwich meridian.

2. RADIO PROPAGATION FACTORS

2.1 The propagation loss on the space to earth path is equal to the free space path loss plus the attenuation exceeded for not more than 1 % of the worst month; the latter being given in Figure 1 for the five rainclimatic zone. The zone (or zones) corresponding to each country is indicated in Tables 1 and 2 below.

2.2 In using the curves of Figure 1, the difference between clear weather attenuation and the attenuation for 99 % of the worst month should be limited to a maximum of 2 dB by appropriate choice of angle of elevation.

2.3 In planning the broadcasting-satellite service, for emissions applying circular polarization, the level of the depolarized component relative to the level of the co-polar component should be taken as :

for rain-climatic zones 1 and 2 : -27 dBfor rain-climatic zones 3, 4 and 5 : -30 dB



Elevation angle (degrees)



Predicted attenuation values exceeded for not more than 1 % of the worst month (0.25 % of the time) at 12 GHz in the rain-climatic zones indicated in Figure 2.

- A : Rain-climatic zone 1
- **B** : Rain-climatic zone 2
- ${f C}$: Rain-climatic zones 3 and 4
- **D** : Rain-climatic zone 5

R.3

1

Document Nº 302-F/E/S Page 19

TABLEAU	1	-	TABLE	1	-	CUADRO	1	

Symboles de pays	Zone hydro- meteorologiques	Symboles de pays	Zone hydro- meteorologiques	Symboles de pays	Zone hydro- meteorologiques
Country symbol	Rain-climatic zone	Country symbol	Rain-climatic zone	Country symbol	Rain-climatic zone
Símbolo del país	Zona hidrome- tereológica	Símbolo del país	Zona hidrome- tereológica	Símbolo del país	Zona hidrome- tereológica
ADL	7	BHR	5	/ CPV	- _7
AFG	5	/_BIO	7	CRO	-
AFI	5	BLR	2	CTI	1
/AFS	1 7	BOL*)		CTR	1
5	5_/			CUB	1
AGL	1	BOT	1 5	CVA	71
ALB	4	BRB	1	CYP	4
ALG	*)	BRM	1	D	3 .
ALS -	2	BRU	1	DAH	1
/AMS	/	BUL	 14	(BENIN)	_
AND	3	CAF*)		DDR	3
AOE	5	CAN	2	DNK	*)
ARG	5		5	DOM	1
ARS	5	∠ CAR	÷7	E	4
ASC	-	CBG	1	EGY	5
ASO	1	CHL	*)	EQA	1
	5	CHN	*)	ETH	1
ATN	1			T.	2
AUS	*)		_ 7	ר - דיד	7
			-'		7
AUT	3		1		-'
AZR	3	CKIN		- FML	2
В	1	CLM		GAD	1
BAH	1		, ⊥ ,	GAB	ר ד ר
BDI	1	CME		GTT GTT	1
BEL	3	CNR	5	GHA	<u>۲</u> ۱.
BER	1	COG		GIB / ⁻ GTF	4
BGD	1	COM	Ţ	7 GTP	/

voir renseignements complémentaires dans le Tableau 2 see additional information in Table 2 véase información complementaria en el Cuadro 2 *)

*)

*)

Document Nº 302-F/E/S

Page 20

Symboles de pays	Zone hydro- meteorologiques	Symboles de pays	Zone hydro- meteorologiques	Symboles de pays	Zone hydro- meteorologiques
Country symbol	Rain-climatic zone	Country symbol	Rain-climatic zone	Country symbol	Rain-climatic zone
Símbolo del país	Zona hidrome- tereológica	Símbolo del país	Zona hidrome- tereológica	Símbolo del país	Zona hidrome- tereológica
GMB	l	/ IWA	7	MLT	4
GNE	1	J	2	MING	5
GNP	1	/ JAR	7	MOZ	1
GRC	4	JMC	1	/ MRA	/
GRL	2	/ JON	/	MRC	4
GTM	l	JOR	5		- 7
GUB	1	KEN	1		7
GUF	1	/ KER	7		
GUI	1	KOR	2	MILLI	5
<u>∕</u> GUM	7	KRE	2	MUT	
HNB	1	KWT	5	MYT	
HND	1	LAO	1	NCG	1
HNG	4	LBN	5	NOT	-
HOL	3	LBR	1	NCD	Ĩ
HTI	1	LBY	5	NGR	
нуо	1	LIE	3		د ر
/ HWA	7	LSO	1	MIG	
/HWL	- 7	LUX	3	NIU	L
I -	ц — ,	MAC	1	NOR	*) *
ICO	-	MAU	1	NPL	1 7
IND	1	MCO	4 -	/ NRU	/
INP	1	/ MCS	/	NZL	. 3
INS	ı	MDG	1	OCE	1
IOB	l	MDR	5 7	OMA	5
IRL	3	∕ MDW	/	PAK	*)
IRN	5	MEX	*)	PAQ .	5
IRQ	5	MT.A	1	PHL	T _
ISL	3	MLD	-	/ PHX	/
ISR	5	MILT	1	/ PLM	/
	l.		5	PNG	1

*) voir renseignements complémentaires dans le Tableau 2

*) see additional information in Table 2
*) véase información complementaria en el Cuadro 2

Document N^O 302-F/E/S Page 21

Symboles de pays	Zone hydro- meteorologiques	Symboles de pays	Zone hydro- meteorologiques	Symboles de pays	Zone hydro- meteorologiques
Country symbol	Rain-climatic zone	Country symbol	Rain-climatic zone	Country symbol	Rain-climatic zone
Símbolo del_país	Zona hidrome- tereológica	Símbolo del país	Zona hidrome- tereológica	Símbolo del país	Zona hidrome- tereológica
PNR	1	SRL	1	/ VIR	7
PNZ	1	STP	1	VTN	1
POL	3	SUI	3	/ WAK	7
POR	3	SUR	1	WAL	1
PRG	1	/ SWN	- 7	YEM	5
PRU	1	SWZ	1	YMS	5
	5	SYR	5	YUG	1
/PTC	- _7	TCD	1,5	ZAI	1
PTR	1	тсн	3	ZAN	1
QAT	5	TGK	1	ZMB	1
REU	l	TGO	l		÷.
RHS	1	THA	1	-3	
<u>/</u> rod	7	TKL	1		
ROU	3	TMP	1		
RRW	1	TON	- 7		
RYU	2	/ TRC	- 7		
S	2	TRD	1		
GDN		TUN	4		
SDN	*)	TUR	4		
SEN	1 5	/ TUV	- 7		
/SEY	- 7	UAE	5		
- /Shn	- 7	UGA	1		(c)
- /SLM	- 7	UKR	2		
- SLV	1	URG	1	- C	
/SMA	- 7	URS	2		
- SMO	1 -		5		
SMR	4	USA	1	*	
SNG	1		3.		
SOM	5	14	_l4 		
SPM	2	VEN	2		
lat o		VEN	+		
) voir ren	seignements com	plémentaires	dans le Tableau	ı 2	ЪЭ

*) see additional information in Table 2
*) véase información complementaria en el Cuadro 2

1

Page 22

•

Tableau 2 Table 2 Cuadro 2

Zones hydrometeorologiques

Rain-climatic zones

Zonas hidrometereológicas

Informations supplementaires. Additional information. Información complem	nentaria.
---	-----------

Pays Country País	IFRB No.**)	Zone Zona	Pays Country Pais	IFRB No.	Zone Zona
ALG	0250 0251 0252	4 4 5	CHN (cont.)	0165 0166 0167 0168	- 5 2 2 2 2
AUS	0004 0005 0006 0007 0008 0009 0010 0011 0012 0013 0014 0015 0315 0316 0317 0318 0319 0320	4 1 5 4 1 4 1 5 4 1 2 4 1 5 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		0169 0170 0171 0172 0173 0174 0175 0176 0177 0178 0179 0180 0181 0182 0183 0181 0182 0183 0184 0185 0186 0187	1 1 1 1 1 1 1 1 2 1 1 1 2 5 5 5 1 1
BOL	0019 0273	1 5		0188	1
CAF	0258	1	DNK	0090 0091	2 3
	0200 0263 0264 0265	5 2 2 3	MEX	0190 0191 0120	3 1 3
CHN	0154 0155 0156	5 5 5	PAK	0121 0127	2 2
	0157 0158 0159 0160 0161 0162 0163	1 2 1 2 1 1 2	SDN	0210 0281 0282 0283 0230 0231	3 5 4 2 1 3,4

**) voir le Plan see the Plan véase el Plan

Document PINK BAGES Page 23

Renseignements encore attendus

Information still required to be furnished

Información que aún debe proporcionarse

۲

-					
		<u> </u>			
AFS	0021 0022 0023 0024				
BOT ·	0603				
ETH	0092			•	
FJI	0193				
NRU	0615				
URS	0059 0060 0061 0064 0065 0066 0067 0068 0069 0070 0071 0072 0073 0074 0075 0076 0076 0077 0078 0079 0080 0081		*		





 $\begin{array}{lll} 0 & & \text{for } 0 \leq \phi \leq 0.25 \ \phi_{0} \\ & -12(\frac{\phi}{\phi_{0}})^{2} & & \text{for } 0.25 \ \phi_{0} \leq \phi \leq 0.707 \ \phi_{0} \\ & -\sqrt{9.0 + 20} \ \log_{10}(\frac{\phi}{\phi_{0}}) \ & \text{for } 0.707 \ \phi_{0} \leq \phi \leq 1.26 \ \phi_{0} \\ & -\sqrt{8.5 + 25} \ \log_{10}(\frac{\phi}{\phi_{0}}) \ & \text{for } 1.26 \ \phi_{0} \leq \phi \leq 9.55 \ \phi_{0} \\ & -33 & & \text{for } 9.55 \ \phi_{0} \leq \phi \end{array}$

Curve A': Co-polar component for community reception without sidelobe suppression

 $\begin{array}{cccc} 0 & & & & \text{for } 0 \leqslant \phi/\phi_0 \leqslant 0.25 \\ & & -12(\phi/\phi_0)^2 & & & \text{for } 0.25 \leqslant \phi/\phi_0 \leqslant 0.86 \\ & & -\sqrt{10.5 + 25 \log_{10}(\phi/\phi_0)} \sqrt{1} \text{ for } 0.86 \leqslant \phi/\phi_0 \text{ up to intersection with } \\ & & & \text{Curve C, (then Curve C)} \\ \end{array}$

 $\begin{array}{l} -25 & \text{for } 0 \leq \phi \leq 0.25 \ \phi_0 \\ -(30 + 40 \ \log_{10} | \frac{\phi}{\phi_0} - 1 |) & \text{for } 0.25 \ \phi_0 < \phi \leq 0.44 \ \phi_0 \\ -20 & \text{for } 0.44 \ \phi_0 < \phi \leq 1.4 \ \phi_0 \\ -(30 + 25 \ \log_{10} | \frac{\phi}{\phi_0} - 1 |) & \text{for } 1.4 \ \phi_0 < \phi \leq 2 \ \phi_0 \\ \end{array}$

- 30 until intersection with co-polar component curve; then as for co-polar component

Curve C : Minus the on-axis gain

Note : for values of ϕ see 3.7.1

b) For Region 2, the relative antenna gain (dB) is given by the curves in Figure 4 for :

- individual reception, for which use should be made of :

Curve A for the co-polar component;

Curve B for the cross-polar component;

- community reception for which Curve B should be used for the cross-.polar component (the co-polar component being given in Figure 3).



AGES

Document No. Page 25

Reference patterns for co-polar and cross-polar components for receiving antennae for individual reception in Region 2

<u>Curve A</u> : Co-polar component without sidelobe suppression

Curve B : Cross-polar component

-25 for
$$0 \le \phi \le 0.25 \phi_0$$

-(30 + 40 $\log_{10} | \frac{\phi}{\phi} | -1 |$) for $0.25 \phi_0 < \phi \le 0.44 \phi_0$
-20 for $0.44 \phi_0 < \phi \le 1.4 \phi_0$
-(30 + 25 $\log_{10} | \frac{\phi}{\phi} | -1 |$) for $1.4 \phi_0 < \phi \le 2 \phi_0$
-30 until intersection with co-polar component curve; then as for co-polar component

<u>Note</u> : for values of ϕ_0 see 3.7.1

•

Document No. 302-E Page 26

3.8 Necessary bandwidth

The necessary bandwidths considered are as follows for :

- 625-line systems : 27 MHz
- 525-line system M of Region 2 : 18 MHz and 23 MHz

3.9 Guard bands

3.9.1 A guard band is defined as the portion of the frequency spectrum between the edge of the allocated band and the edge of the necessary bandwidth of the emission in the nearest channel.

3.9.2 For the planning of the broadcasting-satellite service, the guard bands necessary to protect the services in adjacent frequency bands are shown in the table below.

Regions	Guard band at the lower edge of the band (11.7 GHz)	Guard band at the upper edge of the band (12.2/12.5 GHz)		
l	14 MHz	ll MHz		
2	12 MHz	9 MHz		
3	14 MHz	ll MHz		

These guard bands assume maximum beam centre e.i.r.p. values of 67 dBW for Regions 1 and 3 and 63 dBW for Region 2 (values relating to individual reception), and a filter roll-off of 2 dB/MHz. If smaller e.i.r.p. values are assumed, the guard bands can be reduced in width by 0.5 MHz for each decibel decrease in e.i.r.p.

3.9.3 Since developments in technology or the choice of lower e.i.r.p. values than those given above are likely to permit a reduction in the necessary guard bands, it is recommended that, for purposes other than "a priori" planning at this Conference, the latest CCIR Recommendations concerning spurious emissions from broadcasting satellites should be followed.

3.10 Orbital spacing

The Plan for Regions 1 and 3 has been based on nominal orbital positions spaced uniformly at intervals of 6° .

3.11 Satellite station keeping

Space stations in the broadcasting-satellite service must be maintained in position with an accuracy of better than \pm 0.1° in both the N-S and E-W direction. (These tolerances lead to a maximum excursion of \pm 0.14° from the nominal satellite position).

PINK PAGES Document No. 302-E Page 27

R.3

3.12 Elevation angle of receiving antennae

The Plan has been based on the consideration of a minimum angle of elevation of 20° to minimize the required e.i.r.p. of the satellite and to reduce the effects of shadowing and the possibility of interference from terrestrial services. However, for areas situated in latitudes above about 60° , the angle of elevation is of necessity less than 20° . Attention is also directed to Section 2.2.

For mountainous areas where an angle of 20° may not suffice, an angle of at least 30° has been provided where possible to provide an acceptable service. An angle of elevation of at least 40° has been considered for service areas subject to high precipitation (e.g., rain-climatic zone 1).

Some dry, non-mountainous areas may be given an acceptable service at angles of elevation less than 20°.

In areas with small angles of elevation, the shadowing effect of tall buildings may have to be taken into account.

In choosing a satellite position designed to give the maximum angle of elevation at the ground, the influence of such a position on the eclipse period has been borne in mind.

:

4

Document No. 302-E Page 28

3.13 Transmitting antenna

3.13.1 Cross-section of transmitted beam

Planning has been based on the use of transmitting antennae with beams of elliptical or circular cross-section.

If the cross-section of the transmitted beam is elliptical, the effective beamwidth ϕ_0 is a function of the angle of rotation q between the plane containing the satellite and the major axis of the beam cross-section and the plane in which the beamwidth is required.

The relationship between the maximum gain of an antenna and the half-power beamwidth can be derived from the expression :

$$G_{\rm m} = 27,843/ab$$

or :

$$G_{m}(dB) = 44.44 - 10 \log_{10} a - 10 \log_{10} b$$

where :

a and b are the angles (in degrees) subtended at the satellite by the major and minor axes of the elliptical cross-section of the beam.

An antenna efficiency of 55 % is assumed.

3.13.2 Minimum beamwidth of transmitting antenna

A minimum value of 0.6° for the half-power beamwidth of a transmitting antenna has been agreed for planning.

3.13.3 Transmitting antenna reference patterns

The reference patterns for the co-polar and cross-polar components of satellite transmitting antenna used in preparing the Plan are given in Figure 5.

R.3









<u>Curve A</u> : Co-polar component $-12(\frac{\varphi}{\varphi_0})^2$ for $0 \le \varphi \le 1.58 \phi_0$ -30 for $1.58 \phi_0 \le \varphi \le 3.16 \phi_0$ $-\sqrt{17.5+25} \log_{10}(\frac{\varphi}{\varphi_0})$ for $3.16 \varphi_0 \le \varphi$



after intersection with curve C : as curve C

<u>Curve B</u> : Cross-polar component

 $-(40+40 \log_{10}|\frac{\varphi}{\varphi} - 1|) \text{ for } 0 \le \varphi \le 0.33\varphi_{0}$ -33 for $0.33\varphi_{0} < \varphi \le 1.67\varphi_{0}$ -(4c+40 $\log_{10}|\frac{\varphi}{\varphi_{0}} - 1|$) for $1.67\varphi_{0} < \varphi$

after intersection with curve C : as curve C <u>Curve C</u> : Minus the on-axis gain \cdot

R.3
:

Document No. 302-E Page 30

3.14 Pointing accuracy of satellite antennae

3.14.1 The deviation of the antenna beam from its nominal pointing direction must not exceed a limit of 0.1° in any direction. Moreover, the angular rotation of a transmitting beam about its axis must not exceed a limit of $\pm 2^{\circ}$; this latter limit is not necessary for beams of circular cross-section using circular polarization.

3.14.2 The following factors contribute to the total variation in the area on the surface of the earth illuminated by the satellite beam :

- variations in satellite station-keeping;
- the variations caused by the pointing tolerances, which become more significant for coverage areas with low angles of elevation;
- the effect of the yaw error increases as the beam ellipse lengthens.

3.14.3 The effect of these possible variations should be assessed on a case-by-case basis, since their total effect on the area covered will vary as the geometry of the satellite beam varies, and it would not be reasonable to indicate a single value of shift in the area covered for all situations.

3.14.4 If linear polarization is used for an emission, yaw error makes a significant contribution to increasing the transmitted cross-polarized component; this increases the interference with other carriers which were originally cross-polarized with the emission in question.



3.15 Limitation of output power in the satellite transmitter

The output power of a space station in the broadcasting-satellite service must not rise by more than 0.25 dB relative to its nominal value throughout the life of the satellite.

3.16 Power flux density at edge of coverage area

The value of the power flux density at the edge of the coverage area for 99 % of the worst month is :

- 103 dBW/m² for individual reception in Regions 1 and 3 - 105 dBW/m² for individual reception in Region 2

- lll dBW/m^2 for community reception in all Regions

3.17 <u>Difference between the e.i.r.p. directed towards the edge of the</u> coverage area and that on the axis of the beam

For planning, the absolute value of the difference between the e.i.r.p. directed towards the edge of the coverage area and that on the axis of the beam should preferably be 3 dB.

If the beam area is larger than the coverage area, the value will be less than 3 dB.

3.18 Use of energy dispersal

For planning, an energy dispersal value has been adopted which reduces by 22 dB the spectral power flux density measured in a 4 kHz bandwidth in relation to that measured in the entire bandwidth; this reduction corresponds to a peak-to-peak deviation of 600 kHz.

1

Document No. 302-E Page 32

3. BASIC TECHNICAL CHARACTERISTICS

3.1 Type of modulation

Planning of the broadcasting-satellite service is based on the use of a signal consisting of a video signal with an associated carrier, frequencymodulated by a sound signal, both frequency-modulating a carrier in the 12 GHz band, with a pre-emphasis characteristic in accordance with Figure 2 (from CCIR Recommendation 405).

3.2 Polarization

3.2.1 For the planning of the broadcasting-satellite service, circular polarization shall be used in Regions 1, 2 and 3.1)

3.2.2 If possible, the polarization of different beams intended to serve the same area should be the same.

3.2.3 The terms "direct" and "indirect" used in the Plan to indicate the direction of rotation of circularly polarized waves correspond to the following definitions :

Direct polarization

An elliptically or circularly-polarized wave, in which the electric field-intensity vector, observed in any *fixed plane*, normal to the direction of propagation, whilst looking in (i.e., not against) the direction of propagation, rotates with time in a right-hand or clockwise direction.

Note: For circularly-polarized plane waves, the ends of the electric vectors drawn from any points along a straight line normal to the plane of the wave front, form, at any instant, a left-hand helix.

Indirect polarization

An elliptically or circularly-polarized wave, in which the electric field-intensity vector, observed in any *fixed plane*, normal to the direction of propagation, whilst looking in (i.e., not against) the direction of propagation, rotates with time in a left-hand or anti-clockwise direction.

<u>Note</u>: For circularly-polarized plane waves, the ends of the electric vectors drawn from any points along a straight line normal to the plane of the wave front, form, *at any instant*, a *right-hand* helix.

The administration of Iran expressed a reservation regarding the adoption of circular polarization for planning the broadcasting-satellite service in Region 3 and stated its intention to use linear polarization.

¹⁾ The administration of the United States of America expressed concern regarding the adoption of circular polarization for Region 2 and indicated that the very probable adoption of linear polarization by the fixed-satellite service would preclude the use of cross-polarization to facilitate sharing between the two space services and would affect orbit and spectrum utilization within the Region.



.



Pre-emphasis characteristic for television on 525- and 625-line systems

Curve A: 525-line system B: 625-line system

R. 3

. .

Document No. 302-E Page 33

-+-

.

Document No. 302-E Page 34

3.3 <u>Carrier-to-noise ratio</u>

For the purpose of planning the broadcasting-satellite service, the carrier-to-noise ratio is equal to 14 dB for 99 % of the worst month.

The reduction in quality in the down-link due to thermal noise in the up-link is taken as equivalent to a degradation in the down-link carrierto-noise ratio not exceeding 0.5 dB for 99 % of the worst month.

3.4 Protection ratio between two FM television signals

For planning in Regions 1 and 3 the following protection ratios have been adopted for the purpose of calculating equivalent protection margin (1)

- 31 dB for co-channel signals
- 15 dB for adjacent channel signals.

1) The equivalent protection margin M is given in dB by the formula

 $M = -10 \log \left[10^{-M_1/10} + 10^{-M_2/10} + 10^{-M_3/10} \right]$

where M_1 is the value in dB of the protection margin for the same channel. This is defined in the following expression where the powers are evaluated at the receiver input :

wanted power

(dB) - co-channel protection ratio (dB)

sum of the co-channel interfering powers

 M_2 and M_3 are the values in dB of the upper and lower adjacent channel protection margins.

The definition of the adjacent-channel protection margin is similar to that for the co-channel case except that the adjacent-channel protection ratio and the sum of the interfering powers due to transmissions in the adjacent channel are considered.

R.3

3.5 Channel spacing

ì

3.5.1 Channel spacing in the Plan

The spacing between two adjacent channels is 19.18 MHz. The Plan gives the assigned frequencies for each channel.

3.5.2 Grouping of channels in the same beam

Planning in Region 1 has been carried out by trying to group all the channels radiated within a single antenna beam within a frequency range of 400 MHz, in order to simplify receiver construction.

3.5.3 Spacing between channels of same beam

Owing to technical difficulties in the output circuit of a satellite transmitter, the spacing between two channels feeding a common antenna must be greater than 40 MHz.

3.6 Figure of merit (G/T) of a receiving installation in the broadcastingsatellite_service

In planning the broadcasting-satellite service, the value of the figure of merit (G/T) used is :

- 6 dB/K for individual reception
- 14 dB/K for community reception.

The values are calculated from the following formula which allows for pointing error, polarization effects, and ageing :

$$G/T = \frac{\alpha \beta G_r}{\alpha T_a + (1-\alpha) T_0 + (n-1) T_0}$$

where

- x : the total coupling losses, expressed as a power ratio
- β : the total losses due to the pointing error, polarization effects and ageing, expressed as a power ratio
- G_r : the effective gain of the receiving antenna, expressed as a power ratio and taking account of the method of feeding and the efficiency
- T_a : the effective temperature of the antenna.
- T_0 : the reference temperature = 290 K
- n: the overall noise factor of the receiver, expressed as a power ratio.

See also CCIR Report 473-1 (Annex I).

Document No. 302-E Page 36

3.7 <u>Receiving antenna</u>

3.7.1 Minimum diameter of receiving antenna

For planning the broadcasting-satellite service the minimum diameter of receiving antenna considered is such that the half-power beamwidth, ϕ_{a} , is :

a) for individual reception : 2[°] in Regions 1 and 3
1.8[°] in Region 2

b) for community reception : 1° in all Regions.

3.7.2 Receiving antenna reference patterns

The co-polar and cross-polar reference patterns of receiving antennae are given in Figures 3 and 5.

a) The relative antenna gain (dB) is given by the curves in Figure 3 for :

- individual reception in Regions 1 and 3 :

- Curve A for the co-polar component and

- Curve B for the cross-polar component;

- community reception :

- , in all Regions, Curve A' up to the intersection with Curve C, then Curve C, for the co-polar component,

- in Regions 1 and 3, Curve B for the cross-polar component.

BROADCASTING SATELLITE CONFERENCE

Document No. 303-E 10 February 1977 Original : English

(Geneva, 1977)

PLENARY MEETING

FINAL PROTOCAL

For the People's Republic of Bangladesh

As per Documents Nos. 233 and 265 and the Maps of the Antenna Beam Ellipse, the Beam No. IND0037 allocated to India covers about 90 % of the territory of the People's Republic of Bangladesh. This spillover, which appears to be technically avoidable, is not acceptable to the Bangladesh Administration.

Bangladesh delegation would therefore like to have this reservation recorded by this Conference for necessary corrective measures, with the cooperation of the respected Indian delegation.

Document No. 304-E 11 February 1977 Original : French

(Geneva, 1977)

PLENARY MEETING

FINAL PROTOCOL

For the Republic of Senegal

In signing the Final Acts and associated Plan, the delegation of the Republic of Senegal reserves its Government's right to take any steps or action designed to safeguard its telecommunication services and the rights of the Republic of Senegal.

Furthermore, the delegation of the Republic of Senegal affirms that its Government will continue to cooperate at the international level, on a basis of equality and mutual observance of the rights of countries large and small, in the use of the natural resource constituted by spectrum-orbit space.



Document No. 305-E 11 February 1977 Original : French

(Geneva, 1977)

PLENARY MEETING

FINAL PROTOCOL

For the People's Republic of Benin

The delegation of the People's Republic of Benin reserves its Government's right to : take any action it may consider necessary should any reservations formulated by other administrations jeopardize its interests; take any steps needed to protect its services should any Members of the Union fail to observe the Final Acts of the World Broadcasting-Satellite Administrative Radio Conference, Geneva 1977.

N/È

Document No. 306-E 11 February 1977

PLENARY MEETING

R.4

3

ł

7

4th SERIES OF TEXTS SUBMITTED BY THE EDITORIAL COMMITTEE TO THE PLENARY MEETING

The following texts are submitted to the Plenary Meeting for second reading :

Source	Document No.	Title
B.10	246	Resolutions A - F
B.9	242	Recommendations AA - FF

Miss M. HUET

Chairman of the Editorial Committee

Annexes : 15 pages



6

Ŧ

RESOLUTION No. A

Relating to the annexing to the Radio Regulations of the provisions and associated Plan contained in the Final Acts of the Conference

The World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977,

noting

a) that the World Administrative Radio Conference for Space Telecommunications, Geneva, 1971, adopted Resolution No. Spa2 2 envisaging that stations in the broadcasting-satellite service shall be established and operated in accordance with agreements and associated plans adopted by World or Regional Administrative Radio Conferences;

b) that the present Conference has adopted provisions for all Regions and an associated Plan for Regions 1 and 3;

considering

the wish expressed at the Conference to annex the provisions and associated Plan to the Radio Regulations;

resolves

that the 1979 World Administrative Radio Conference be requested to annex the provisions and associated Plan to the Radio Regulations as an integral part thereof, in the form and to the extent it deems most appropriate without thereby affecting their content or integrity;

requests

the Administrative Council to include the request referred to in the above paragraph in the agenda of the 1979 World Administrative Radio Conference.

R.4-2

RESOLUTION NO. B

Relating to the period between the entry into force of the Final Acts of the Conference and the date on which the provisions and associated Plan are annexed to the Radio Regulations

The World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977,

considering

a) that its Final Acts will come into force on 1 January 1979;

b) that, in its Resolution No. A, it has requested the 1979 World Administrative Radio Conference to annex to the Radio Regulations the provisions and associated Plan established by the Conference;

c) that there will be an interim period between the date of entry into force of these Final Acts and the date on which the provisions and associated Plan are annexed to the Radio Regulations;

further considering

that these Final Acts are regarded as including a World agreement and associated Plan in accordance with Resolution No. Spa2 - 2 of the World Administrative Radio Conference for Space Telecommunications, Geneva, 1971;

resolves

1. that both during this interim period and after the date on which they have been annexed to the Radio Regulations, the provisions and the associated Plan shall retain their integrity as a legal instrument;

2. that during this period the IFRB and the other appropriate organs of the Union shall be guided by the provisions of these Final Acts and the Radio Regulations.

RESOLUTION NO. C

Relating to the updating of the Master International Frequency Register for Regions 1 and 3 on the date of entry into force of the Final Acts

The World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977,

considering

a) that the Final Acts of this Conference will take effect before the entry into force of the revised Radio Regulations adopted by the 1979 World Administrative Radio Conference, and that meanwhile the relevant provisions of the current Radio Regulations and Resolutions Nos. Spa2 _ 2 and Spa2 _ 3 remain valid;

b) that No. 405BA of the Radio Regulations provides that in the band 11.7 _ 12.2 GHz in Region 3 and in the band 11.7 _ 12.5 GHz in Region 1, existing and future fixed, mobile and broadcasting services shall not cause harmful interference to broadcasting-satellite stations operating in accordance with the decisions of the present Conference;

c) that the coordination procedures described in Resolution No. Spa2 _ 3 are to be applied only until the entry into force of plans pursuant to Resolution No. Spa2 _ 2;

resolves

1. that all administrations using or intending to use frequency assignments to terrestrial stations in the bands covered by the Plan shall decide as soon as possible, whether or not these assignments will affect frequency assignments in conformity with the Plan (if necessary, with the assistance of the IFRB);

2. that, if it is found that frequency assignments in accordance with the Plan may be subject to interference, administrations shall inform the IFRB of the measures they intend to take to ensure the protection of the frequency assignments concerned before the date of entry into force of these Final Acts;

R.4-4

3. that administrations may continue to use frequency assignments which are not in conformity with the Plan, provided that agreement is reached with the administrations whose broadcasting-satellite stations are affected;

4. that the administrations seeking agreement shall inform the IFRB of the terms of the agreement reached;

5. that, upon receipt of such information, the IFRB shall insert a symbol in the Remarks column of the Master Register indicating the duration specified in the agreement. The duration specified shall also be published in a special section of its weekly circular;

6. that, on the date of entry into force of the Final Acts, the frequency assignments in the Plan will be entered in the Master Register. The date of signature of these Final Acts will be entered, together with an appropriate symbol, in Column 13c opposite these assignments;

invites the IFRB

to assist administrations in implementing the provisions of this Resolution.

PINK PAGES

2

RESOLUTION NO. D

Relating to the coordination, notification and recording in the Master International Frequency Register of assignments to stations in the fixed-satellite service with respect to stations in the broadcasting-satellite service in Region 2

The World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977,

considering

that the Radio Regulations contain no provisions governing the coordination, notification or recording in the Master International Frequency Register of frequency assignments to stations in the fixed-satellite service in the band 11.7 12.2 GHz with respect to stations in the broadcasting-satellite service in Region 2;

resolves

that the provisions of Article 9A of the Radio Regulations shall be applied in such cases until the matter is considered by a competent Administrative Radio Conference.

R. 4-15

RESOLUTION NO. E

Relating to the coordination, notification and recording in the Master International Frequency Register of frequency assignments to stations in the broadcasting-satellite service in Region 2

The World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977,

considering

a) that a plan will be established for the broadcasting-satellite service in Region 2 in accordance with Recommendation HH;

b) that in Region 2 the broadcasting-satellite service should be operated on the basis of the principles contained in Article 11 and Annexes [...and...] of these Final Acts;

c) that some of the provisions adopted by this Conference concerning the broadcasting-satellite service in Regions 1 and 3 may also be applied in Region 2 prior to the entry into force of the plan for that Region to be established pursuant to Recommendation HH;

<u>d</u>) that, in the interim period, the procedures described in Resolution No. Spa2 - 3 will continue to apply in Region 2;

resolves

1. that an administration intending to bring into use a space station in the broadcasting-satellite service in Region 2 shall, for the purpose of coordination with space systems of other administrations, apply the relevant provisions of Article 9A of the Radio Regulations i.e., Nos. 639AA to 639AI inclusive;

2. that the relevant provisions of Resolution No. Spa2 _ 3 shall apply to the coordination, notification and recording of stations in the broadcasting-satellite service in Region 2, wherever a station in the broadcasting-satellite service or the fixed-satellite service in Region 2 is involved;

2.1 that an administration notifying a frequency assignment to a space station in the broadcasting-satellite service in Region 2 under paragraph 4.1 of Resolution No. Spa2 _ 3 shall also notify a typical receiving earth station;

R. 4-7

3. that the coordination, notification and recording procedures for stations in the fixed-satellite service specified in Article 7 of these Final Acts shall also apply to stations in the broadcasting-satellite service in Region 2 with respect to stations in the broadcasting-satellite service for which a frequency assignment appears in the Plan whenever

- any portion of the necessary bandwidth of the proposed frequency assignment in Region 2 falls within the necessary bandwidth of a frequency assignment in Region 1 or Region 3, and
 - the power flux density which would be produced by the proposed broadcasting-satellite frequency assignment in Region 2 exceeds the value specified in Annex 1.

4. that Annex 2 of these Final Acts shall be used in supplying the information referred to in Section B of Resolution No. Spa2 - 3 and Section II of Article 7;

5. that an individual notice for each frequency assignment shall be drawn up as prescribed in Annex 2 for any frequency assignment notified under paragraph 4.1 of Resolution No. Spa2 _ 3 or paragraph 2.1 of this Resolution or Section III of Article 7 of these Final Acts.

RESOLUTION No. F

Relating to the use, by space stations operating in the frequency bands 11.7 - 12.2 GHz (in Regions 2 and 3) and 11.7 - 12.5 GHz (in Region 1), of the geostationary orbit and no other

The World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977,

considering

a) that a plan designating frequency assignments in the above-mentioned frequency bands and positions in the geostationary orbit has been adopted by the Conference for Regions 1 and 3;

b) +hat a similar plan for Region 2 is expected to result from a Regional Administrative Radio Conference in 1982;

c) that the operation of space radiocommunication services in the frequency bands concerned in orbits other than the geostationary orbit would be incompatible with the plans referred to in a) and b) above;

resolves

that administrations shall ensure that their space stations in these frequency bands are operated in the geostationary orbit and no other.

R.4-9

Document No. 306-E Page 10

RECOMMENDATION No. AA

To the CCIR relating to spurious emissions in the broadcasting-satellite service

The World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977,

considering

a) that space stations in the broadcasting-satellite service operating at high power levels are likely to cause interference to services in adjacent and in harmonically related frequency bands due to spurious emissions;

b) that, in the planning of the broadcasting-satellite service, account must be taken of the need to reduce interference to services operating in adjacent bands to acceptable levels at the lower and upper edges of the bands 11.7 - 12.2 GHz in Regions 2 and 3 and 11.7 - 12.5 GHz in Region 1, and to the radio astronomy service which has an exclusive allocation at 23.6 - 24 GHz in all three Regions;

c) the technical data required to enable the 1979 World Administrative Radio Conference to revise the Radio Regulations;

<u>d)</u> the studies being pursued by the CCIR under the appropriate Study Programme;

invites the CCIR

to continue, as a matter of urgency, the study of the technical and operational aspects of spurious emissions from space stations in the broadcasting-satellite service to enable the Special Joint Meeting of CCIR Study Groups to draw up a report for the 1979 World Administrative Radio Conference.

Document No. 306-E Page 11

RECOMMENDATION No. BB

To the CCIR relating to transmitting antennae for the broadcasting-satellite service

The World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977,

considering

<u>a)</u> the need for ample information on transmitting antennae for the planning of the broadcasting-satellite service;

b) the technical data required to enable the 1979 World Administrative Radio Conference to revise the Radio Regulations;

c) the studies being pursued by the CCIR under the appropriate Questions and Study Programmes;

invites the CCIR

1. to continue the study of reference patterns for the co-polar and cross-polar components of transmitting antennae for the broadcastingsatellite service for both individual and community reception, and in particular the practicable means of achieving various degrees of improved side-lobe suppression and the economic implication thereof;

2. to initiate the study of the technical characteristics designed to achieve a pointing accuracy for transmitting antenna such that :

- the deviation of the antenna beam from its nominal direction of pointing shall not exceed 0.1°;
- the angle of rotation of the transmitting beam about its axis shall not exceed + 2°;

3. to submit as much information as possible on these problems to the 1979 World Administrative Radio Conference.

RECOMMENDATION No. CC

To the CCIR relating to studies of propagation at 12 GHz for the broadcasting-satellite service

The World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977,

considering

a) the need for ample information on the various propagation factors required for the planning of the broadcasting-satellite service;

b) the technical data required to enable the 1979 World Administrative Radio Conference to revise the Radio Regulations;

c) the studies being pursued by the CCIR under the appropriate Study Programmes;

invites the <u>CCIR</u>

1. to continue the study of the effects of precipitation attenuation at low angles of incidence in all rain-climatic zones;

2. to initiate the study of the effects of sand and dust storms;

3. to examine the relationship between the propagation characteristics for 99 % of the worst month and those for the year;

4. to examine, for emissions using circular polarization, the level of the depolarized component relative to the polarized component;

5. to submit as much information as possible on these problems to the 1979 World Administrative Radio Conference.

RECOMMENDATION No. DD

To the CCIR relating to up-links for the broadcasting-satellite service

The World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977,

considering

<u>a)</u> the need for ample information on the characteristics of up-links for planning the broadcasting-satellite service;

b) the technical data required to enable the 1979 World Administrative Radio Conference to revise the Radio Regulations;

<u>c)</u> the studies being pursued by the CCIR under the appropriate Study Programme;

<u>d</u>) that the protection ratios in the up-links to broadcasting satellites need to be approximately one order of magnitude greater than those in the down-links;

.e) that, as regards up-link interference between broadcasting satellites at different orbital positions, adequate up-link protection ratios (approximately 10 dB greater than those in the down-link) would appear to be readily achievable by antenna pattern discrimination in earth station transmitting antennae which would clearly have to be larger in diameter than the receiving antennae used in the down-links;

<u>f</u>) that, where planning is based on isolation parameters such as radiation patterns for space station transmitting antennae, carrier interleaving, and/or polarization discrimination in meeting the down-link carrier-to-interference requirements between service areas served from a single orbital position, the increased carrier-to-interference requirements in the up-links serving the satellite(s) at that same orbital position will have to use the same isolation parameters provided that this produces an improvement of about 10 dB in net isolation. The characteristics of the transmitting earth station will clearly not affect this isolation, apart from the purity of their on-beam polarization;

<u>g</u>) that in the implementation of broadcasting-satellite systems, consideration must be given to all aspects of associated space operation service functions (tracking, telemetry, telecommand and ranging) in connection with the operation of broadcasting satellites;

R.4-13

1

Document No. 306-E Page 14

invites the CCIR

1. to continue the study of those radiation characteristics of receiving antennae of space stations in the broadcasting-satellite service which, singly or in combination with other means of discrimination, would give the necessary protection ratios for the up-links of systems in the broadcasting-satellite service for (a) satellite(s) occupying a given position in the geostationary satellite orbit;

2. to continue the study of those polarization characteristics of receiving antennae of space stations in the broadcasting-satellite service which, singly or in combination with other means of discrimination, would give the necessary protection ratios for the up-links of systems in the broadcasting-satellite service for (a) satellite(s) occupying a given position in the geostationary satellite orbit;

3. to continue the study of the technical up-link characteristics required to implement the plan for this service;

4. to study the technical and design characteristics and requirements which affect the provision of "space operation service functions" of space stations in the broadcasting-satellite service;

5. to study the requirements for adjacent-channel isolation in up-links for (a) satellite(s) in the broadcasting-satellite service occupying a given position in the geostationary satellite orbit;

6. to draw up a report at the Special Joint Meeting of CCIR Study Groups to be held for the preparation of technical data for the 1979 World Administrative Radio Conference.

R.4-05

RECOMMENDATION No. EE

Relating to up-links for the broadcasting-satellite service

The World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977,

considering

<u>a)</u> that, according to the definition given in No. 84AG of the Radio Regulations, the fixed-satellite service includes earth-to-space links for the broadcasting-satellite service;

b) that there is an imbalance between the width of the bands allocated to earth-to-space links and those allocated to space-to-earth links in the fixed-satellite and broadcasting-satellite services between 10 and 15 GHz;

<u>c)</u> that, in consequence, the earth-to-space capacity may be insufficient to meet future demands for space-to-earth links for the broadcastingsatellite and fixed-satellite services;

<u>d</u>) that, due to interference considerations, space stations in both services may be subject to severe up-link constraints;

e) that Recommendation No. DD invites the CCIR to continue the studies on up-links for the broadcasting-satellite service;

invites administrations

to estimate their future technical requirements for such links for the purpose of the studies mentioned in e) above, and to forward them to the appropriate CCIR Study Groups and to the Special Joint Meeting of CCIR Study Groups to be held in preparation for the 1979 World Administrative Radio Conference.

RECOMMENDATION No. FF

To the CCIR relating to the interdependence of receiver design, channel grouping and sharing criteria

The World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977,

considering

a) that receiver design, channel grouping and sharing criteria are interrelated and have a considerable influence on the development of a plan for the broadcasting-satellite service;

b) that, so far, insufficient attention may have been given to these factors and to their influence on the implementation of such a plan;

invites the CCIR

to study the problem of the interdependence of receiver design, channel grouping and sharing criteria, together with the effects of these factors on the operation of the broadcasting-satellite service.

BROADCASTING SATELLITE CONFERENCE

Document No. 307-E 11 February 1977 Original : French

(Geneva, 1977)

PLENARY MEETING

FINAL PROTOCOL

For the Togolese Republic

In signing the Final Acts of the World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977, the delegation of the Togolese Republic reserves its Government's right to take any action it may consider necessary to safeguard its interests should any Member fail to abide by the Final Acts and their Annexes and the Protocol attached thereto or should any reservations formulated by other countries jeopardize the proper functioning of its Broadcasting-Satellite Services.

BROADCASTING SATELLITE CONFERENCE

Document No. 308-E 11 February 1977 Original : French

(Geneva, 1977)

PLENARY MEETING

FINAL PROTOCOL

For the Republic of Guinea

In signing the Final Acts of the World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977, the delegation of the Republic of Guinea reserves its Government's right to take any measures necessary in order to safeguard its interests if reservations made by other delegations on behalf of their administrations or non-compliance with the Final Acts and Annexes should have the effect of jeopardizing the satisfactory operation of the Satellite- and Terrestrial-Broadcasting Services of the Republic of Guinea.

BROADCASTING SATELLITE CONFERENCE

(Geneva, 1977)

Document No. 309-E 11 February 1977 Original : English

PLENARY MEETING

FINAL PROTOCOL

For Australia, New Zealand and Papua New Guinea

The Plan as published shows an Australian system providing three beams each of six channels from orbital position 98°E to the Western States and similarly three beams each with six channels to the Eastern States from orbital position 128°E. The Plan also shows New Zealand systems operating from both 158°E and 128°E and Papua New Guinea systems operating from 110°E and 128°E.

In the case of the systems on 128°E it is proposed, as an interim development, for the three countries to share a satellite which would enable a small number of programmes to be provided into each of the Australian states and in New Zealand and Papua New Guinea.

The planning of this initial system has not yet been finalized and the coordination of the channel allocations will be undertaken outside the Conference using the technical criteria and coordination procedures established at the Conference.

(Geneva, 1977)

Document No. 310-E 11 February 1977 Original : Spanish

PLENARY MEETING

FINAL PROTOCOL

For the Republic of Bolivia

The delegation of the Republic of Bolivia to the World Administrative Radio Conference for the Planning of the Broadcasting-Satellite Service in frequency bands 11.7 - 12.2 GHz (in Regions 2 and 3) and 11.7 - 12.5 GHz (in Region 1), in signing the Final Acts of the Conference, reserves its Government's right to take any action it might consider necessary to safeguard its interests should the lack of a plan for Region 2 in any way jeopardize subsequent plans for the use of the orbit by the broadcasting-satellite or any other telecommunication service.



(Geneva, 1977)

Document No. 311-E 11 February 1977 Original : English

PLENARY MEETING

FINAL PROTOCOL

For the Kingdom of Saudi Arabia, State of Bahrain, Arab Republic of Egypt, United Arab Emirates, State of Kuwait, Kingdom of Morocco, Islamic Republic of Mauritania, Sultanate of Oman, Yemen Arab Republic, People's Democratic Republic of Yemen

The above-mentioned delegations declare that the signature and possible subsequent notification by their respective Governments of the Final Acts of the World Broadcasting-Satellite Administrative Radio Conference, (Geneva, 1977) does not imply the recognition of Israel in any way.



For reasons of economy, this document is printed in a limited number. Participants are therefore kindly asked to bring their copie to the conference since only a few additional copies can be made available.

Document No. 312-E 11 February 1977 Original : English

(Geneva, 1977)

PLENARY MEETING

FINAL PROTOCOL

For Algerian Democratic and Popular Republic, Kingdom of Saudi Arabia, State of Kuwait, Kingdom of Morocco, Democratic Republic of the Sudan, Tunisia, Yemen Arab Republic

The delegations of the above-mentioned countries reserve their rights to take any action required to safeguard their interests to implement and protect their Satellite-Broadcasting and Terrestrial Telecommunication Services in case if any country violates the Final Acts of this Conference.



Document No. 313-E 11 February 1977 Original : English

(Geneva, 1977)

PLENARY MEETING

FINAL PROTOCOL

For Algerian Democratic and Popular Republic, Kingdom of Saudi Arabia, State of Bahrain, People's Republic of Bangladesh, Arab Republic of Egypt, United Arab Emirates, State of Kuwait, Malaysia, Kingdom of Morocco, Sultanate of Oman, Pakistan, Democratic Republic of the Sudan, Tunisia, Yemen Arab Republic, People's Democratic Republic of Yemen

The delegations of the above-mentioned countries who have supported the request submitted by the delegation of Saudi Arabia for the proposed Islamic Programme originating from Saudi Arabia, which was not incorporated in the Plan appropriately, reserve their right to press for the implementation of this proposal in future whenever it is technically feasible.



Document No. 314-E 3 June 1977 Original : English

Corrigendum No. 1 to

(Geneva, 1977)

1

1.

MINUTES

OF THE

SEVENTH PLENARY MEETING

Page 7, item 4 (Documents submitted by Committee 4), <u>replace</u> the third paragraph by the following :

The delegate of Denmark pointed out that in the second formula for curve B the figure 0.33 ϕ should read 0.33 ϕ_0 .



(Geneva, 1977)

Document No. 314-E 15 February 1977 Original : English

PLENARY MEETING

Documents Nos.

241

243 .

217

231, 232, 234, 235

MINUTES

OF THE

SEVENTH PLENARY MEETING

Tuesday, 8 February 1977, at 1415 hrs

Chairman : Mr. Ib LØNBERG (Denmark)

Subjects discussed

 Extension of the Conference and deadlines for the tabling of declarations and counterdeclarations

2. Report of Committee 2

3. Fifth, sixth, seventh and eighth series of texts submitted by the Editorial Committee

4. Document submitted by Committee 4

5. Final Report of the Budget Control Committee

6. Period of validity of the Plan

1. <u>Extension of the Conference and deadlines for the tabling of</u> <u>declarations and counter-declarations</u>

The <u>Chairman</u> announced that the Steering Committee had decided to prolong the Conference by one day, Saturday 12 February. The situation would be reviewed again on Thursday, 10 February and, depending on the progress of work in Committee 5 and the Plenary Meetings, the possibility of a further extension might have to be considered at that time.

Document No. 314-E Page 2

Following the Steering Committee's decision, new deadlines had been set for the tabling of declarations and counter-declarations, namely, Thursday, 10 February at 2000 hrs for declarations and Friday, 11 February at 2000 hrs for counter-declarations. For formal reasons, it would not be possible to make any exceptions in respect of those deadlines.

The <u>Secretary-General</u> drew attention to No. 513 of the Convention, which provided for the tabling of reservations to be included in the Final Protocol to the Final Acts of the Conference. All such reservations must be brought to the notice of the Plenary Meeting in order to enable any delegations wishing to enter counter-reservations to do so. Consequently, a twenty-four hour interval had been allowed between the deadline for declarations and that for counter-declarations.

The <u>delegate of Mexico</u> said that if the Plan or any of the basic documents produced by the Conference were to be modified on Thursday, 10 February in such a way as to affect the interests of his Administration, it would be necessary for his delegation to reconsider the deadlines indicated by the Chairman.

The <u>delegate of Uruguay</u> observed that reservations could only be tabled once the results of the Conference were known. Should any problems still remain unsolved on Thursday, 10 or Friday, 11 February, some provision would presumably have to be made for the tabling of reservations on those matters later than the deadlines to which reference had been made.

The <u>Secretary-General</u> said that a certain amount of flexibility would be introduced should such a situation arise.

The <u>Chairman</u> said that if he heard no objection he would take it that the procedure he had outlined and the explanations given by the Secretary-General were acceptable to delegations.

It was so agreed.

2. Report of <u>Committee</u> 2 (Document No. 241)

The <u>Chairman of Committee 2</u> introduced the Committee's Report, drawing particular attention to the recommendation in paragraph 4 (page 1).

The recommendation, having been supported by the <u>delegate of Mexico</u>, was adopted.

The <u>Chairman of Committee 2</u>, referring to page 3 of the Annex, said that "Roumania (Socialist Republic of)" should be inserted in the appropriate place in paragraph 1.1, and "Central African Empire" and "Congo (People's Republic of)" in paragraph 1.2.1. The names of those three countries should be deleted from pages 4 and 5.

The Report of Committee 2 and its Annex (Document No. 241), as amended, were adopted.
3. Fifth, sixth, seventh and eighth series of texts submitted by the Editorial Committee (Documents Nos. 231, 232, 234, 235)

Fifth series of texts (B.5) (Document No. 231)

The <u>Chairman of Committee 8</u> said that the texts in question, which originated from Document No. 177, were to be incorporated in an Annex to the Final Acts relating to technical sharing criteria.

Paragraph 1.1

On a proposal by the <u>delegate of the United Kingdom</u>, it was <u>agreed</u> to alter the first part of the introductory sentence to read : "The establishment of sharing criteria for the different services using the 12 GHz band....".

On a proposal by the <u>delegate of Spain</u>, it was <u>agreed</u> to change the headings of the first two columns of the table (page 3) to read, respectively, "Wanted service" and "Wanted signal".

The <u>Chairman of Committee 8</u> read out a minor editorial correction to be made in the first line of the first column of the table in the French and Spanish texts.

The <u>representative of the IFRB</u> drew attention to the value 30 dB in the first line of the fifth column of the table, observing that 31 dB was given for the same value in another document. It would perhaps be appropriate to replace "30 dB" by "31 dB" in the table and to delete Note 4.

The <u>Chairman of Committee 5</u> said that Committee 5 had adopted 31 dB as the protection ratio value for the same channel in Regions 1 and 3.

The <u>delegate of the United Kingdom</u> said that the value 31 dB had indeed been used for planning purposes, but that 30 dB was the correct figure to be used as one of the parameters for the establishment of sharing criteria.

The <u>delegate of the United States of America</u> said he would welcome information on the reasons for which Committee 5 had in certain cases adopted values that were different from those which Committee 4, often after lengthy discussions, had recommended for planning purposes.

The <u>Chairman of Committee 5</u> said that Committee 5 had added 1 dB to the protection value after taking into account tube ageing and various other considerations.

The <u>representative of the IFRB</u>, referring to the French text only, said that the word "défavorable" should be replaced by "défavorisée" in Note 3.

It was so <u>agreed</u>.

In reply to a question by the <u>representative of the IFRB</u> concerning Note 5, the <u>Chairman of Working Group 4B</u> said that the Note had been included for the sole purpose of indicating the source of the value in question. The <u>delegate of Egypt</u> observed that Note 6 might be misleading, since it related to the method of calculating the protection ratio and not to the protection requirement itself.

The <u>Chairman of Working Group 4B</u> said that the wording of Note 6 was a compromise solution which had been worked out by a small drafting group and adopted by the Working Group after lengthy discussions on the subject.

The <u>representative of the IFRB</u>, referring to Note 6, said it was his understanding that the IFRB would, if the need arose, have all latitude to take account of the conditions prevailing in tropical regions.

The <u>delegate of Australia</u>, referring to Note 7, proposed that the word "wanted" should be replaced by "carrier".

It was so agreed.

Paragraph 1.2

On a proposal by the <u>delegate of the United Kingdom</u>, it was <u>agreed</u> to replace the word "protected" by "wanted" in the first and last sentences.

On a proposal by the delegate of Canada, it was <u>agreed</u> to replace the phrase "for planning purposes" by the phrase "for determining sharing criteria" in the second sentence.

Paragraph 1.3

The <u>representative of the IFRB</u> suggested that the first sentence should refer to the receiver input rather than to the receiving ground station only.

It was so agreed.

Paragraph 1.4

On a proposal by the <u>delegate of France</u>, it was <u>agreed</u> to replace "post-detection" by "post-demodulation" in the first sentence.

Paragraph 1.6

On a proposal by the <u>delegate of the United Kingdom</u>, it was <u>agreed</u> to replace "protected" by "wanted" in the first line and sub-paragraph a).

Paragraph 1.8

The <u>Chairman of Committee 8</u>, referring to A, B and C underneath the figure, said that "protected" should be replaced by "wanted" in all three cases.

The fifth series of texts (B.5) was <u>approved</u> on first reading, as amended and subject to editorial changes.

Sixth series of texts (B.6) (Document No. 232)

The <u>Chairman of Committee 8</u> introduced Document No. 232, indicating minor editorial corrections, and the <u>Chairman of Committee 6</u> said that, apart from editorial changes, the document was exactly as Committee 6 had received it from Committee 4.

Title

On the proposal of the <u>United Kingdom delegate</u>, it was <u>agreed</u> that on pages 1 and 2 the words "in the band 11.7 - 12.2 GHz (in Regions 2 and 3) and 11.7 - 12.5 GHz (in Region 1)" should be placed after the words "service area".

Paragraph 2.1

It was <u>agreed</u> that the term "wanted bandwidth" should be changed to "necessary bandwidth".

Paragraph 1.4 c)

On the proposal of the <u>United Kingdom delegate</u>, it was <u>agreed</u> that the sub-paragraph should read "the necessary protection ratios for the broadcasting-satellite service."

Paragraph 2.3.3

In reply to the <u>delegate of Australia</u>, the <u>Chairman of Committee 5</u> pointed out that the term "occupied bandwidth" was not used in any other document.

The <u>delegate of the United Kingdom</u> said that the Sub-Working Group under his chairmanship had done the earlier technical work on the point in question. The idea had been that the separation between the carriers of the broadcasting-satellite service and the terrestrial service should be about 30 MHz. In order to express that, the phrase "occupied bandwidth" which was clearly defined in the Radio Regulations, had been taken and there had been considerable discussion on the derivation of the 7 MHz figure, but for typical terrestrial service bandwidths and the broadcasting-satellite bandwidth, that would correspond to carrier separations of about 30 MHz.

The <u>Chairman of Committee 8</u> pointed out that a slight problem arose because the bandwidth occupied by another administration was not generally known, only the necessary bandwidth, and one could not be deduced from the other.

After discussion, it was <u>agreed</u> that paragraph 2.3.3 should be amended to read : "A signal ... only if its necessary bandwidth overlaps the necessary bandwidth of the broadcasting-satellite assignment."

As a consequence, "occupied bandwidth" should be changed to "necessary bandwidth" also in paragraph 2.3.2.

Document No. 314-E Page 6

Paragraphs 2.4.1 and 2.4.2

It was <u>agreed</u> to delete the words " / for protection / ".

At the suggestion of the <u>representative of the IFRB</u>, it was also <u>agreed</u> to delete the words "either as specified in the Plan or".

The sixth series of texts submitted by the Editorial Committee was <u>approved</u> on first reading, as amended, and subject to editorial changes.

Seventh series of texts (B.7) (Document No. 234)

The Chairman of Committee 8 introduced Document No. 234.

Paragraph 1.1

On the proposal of the <u>Chairman of Committee 6</u>, it was <u>agreed</u> to delete the words "...or are separated by less than 7 MHz ...".

Footnote 2) on page 2

On the proposal of the <u>delegate of the United Kingdom</u>, it was <u>agreed</u> that the footnote should be <u>amended</u> to read : ".... except that determination of the need for the coordination referred to in No. 492A of the Radio Regulations shall be based on Appendix / see B.6 /."

Paragraph 3.10 c)

It was agreed to replace the word "station" by "assignment".

The <u>delegate of Italy</u> said that the use of the expression station or frequency assignment "which appears in the Plan" might give the impression that only those which appeared in the Plan at the time of signature were covered. He therefore proposed that those words referring to a station or frequency assignment should be changed to "in conformity with the Plan" wherever they appeared in the document.

It was so agreed.

The seventh series of texts submitted by the Editorial Committee was approved on first reading, as amended, and subject to editorial changes.

Eighth series of texts (B.8) (Document No. 235)

The <u>Chairman</u> said that the term "occupied bandwidth" should be replaced by "necessary bandwidth" throughout the document.

The <u>Chairman of Committee 6</u> said that all references to "the Broadcasting-Satellite Plan" should be shortened to "the Plan".

Section I

The Chairman said that footnote 2) to paragraph 1.1 should be deleted.

Section I, as amended, was approved.

Section II

The <u>Chairman</u> said that the words "new or modified" in the second line of paragraph 2.1 should be deleted.

The <u>delegate of Italy</u> said that the Appendix marked with one asterisk in that paragraph was Appendix D. The same applied to the Appendix mentioned in paragraph 2.3, and footnote 1) on page 4 could now be deleted.

Section II, as amended, was approved.

Sections III to IX

Approved.

Section X

The <u>delegate of Italy</u> said that the Appendix referred to in sub-paragraph a) was Appendix D.

Section X, as amended, was approved.

Document No. 235, as amended, was approved.

4. Document submitted by Committee 4 (Document No. 243)

The <u>Chairman</u> said that, although the Plenary Meeting could not discuss the text of the document until the results of calculations of orbital positions were made known, the Chairman of Committee 5 was anxious for the Plenary to approve the revised reference antenna pattern on page 5, since that pattern had to serve as a basis for planning work.

The <u>representative of the CCIR</u> said that the figure 12.7 in the third formula for curve A should be replaced by 17.5.

The <u>delegate of Denmark</u> said that the figure -33 in the second formula for curve B should be -50.

The reference antenna pattern on page 5 of Document No. 243 was approved.

5. Final Report of the Budget Control Committee (Document No. 217)

The <u>Chairman of Committee 3</u> introduced the report, pointing out that the words "an updated edition of the report ... and the Additional Radio Regulations" at the end of paragraph 3 should be replaced by "the re-arranged Radio Regulations as in 'resolves' paragraph 1 of Resolution No. A." Document No. 314-E Page 8

In reply to a question by the <u>Chairman</u>, the <u>Chairman of Committee 3</u> said that two copies of the Final Acts would be handed to smaller delegations and three or four to larger ones.

In reply to the <u>delegate of Spain</u>, the <u>Deputy Secretary-General</u> said that the additional copies would be sent to administrations within ten days after the end of the Conference, but that surface mail would be used to reduce expenditure.

The final report of Committee 3 was approved.

The <u>Chairman of Committee 3</u> expressed his thanks to all those who had helped him to bring the Committee's work to a successful conclusion.

6. Period of validity of the Plan

The <u>delegate of India</u> said that, in view of the different opinions held on the period of validity of the Plan, it would be advisable to discuss the matter in the Plenary Meeting as soon as possible.

The <u>Chairman of Committee 6</u> said that his Committee had come to no final decision on the question, but that the relevant document would be placed before the Plenary Meeting the following day.

The <u>delegate of the United Kingdom</u> said he saw no reason to single that provision out for especially urgent consideration. In any case, the Plan would remain in force until it was revised by a future competent Administrative Radio Conference.

The <u>Chairman</u> said that the point would be dealt with in connection with the relevant texts from Committee 6.

The meeting rose at 1755 hours.

The Secretary-General :

The Chairman : I. LØNBERG

M. MILI

(Geneva, 1977)

Document No. 315-E 11 February 1977 Original : English

PLENARY MEETING

FINAL PROTOCOL

For the United Republic of Tanzania

The delegation of the United Republic of Tanzania to the World Administrative Broadcasting-Satellite Conference, Geneva, 1977, declares that its Government reserves the right to take any measures it considers necessary to safeguard its interests should failure by other countries to observe the provisions of the Final Acts of the Conference prove prejudicial to the proper functioning of its services.



(Geneva, 1977)

Document No. 316-E 11 February 1977 Original : English

PLENARY MEETING

FINAL PROTOCOL

For the Federative Republic of Brazil

A plan will be established for the Broadcasting-Satellite Service in Region 2 as a result from a Regional Administrative Radio Conference to be held in 1982.

In Region 2 the Broadcasting-Satellite Service should be operated on the basis of principles and provisions contained in Document No. 204.

The Brazilian Administration draws the attention of this Conference to the fact that 5 TV channels for each service area in the territory of Brazil is the minimum number of channels necessary to satisfy its requirements.

(Geneva, 1977)

Document No. 317-E 11 February 1977 Original : English

RCH1

GENÈ

PLENARY MEETING

FINAL PROTOCOL

For the Democratic Republic of the Sudan

1. While signing the Final Acts of the World Administrative Radio Conference for the Planning of the Broadcasting-Satellite Service, Geneva 1977, in the frequency bands 11.7 - 12.2 GHz (in Regions 2 and 3) and 11.7 - 12.5 GHz (in Region 1), the delegation of the Democratic Republic of the Sudan reserves the right of its Government to take such measures as may be necessary to safeguard its interests should any country reserve and/or not accept the provisions of the Final Acts including the Associated Plan.

2. The delegation of the Democratic Republic of the Sudan wishes to point out that the interference caused by the beam ET 0092A may prove to be harmful. The Sudanese delegation reserves the right of its Government to take appropriate measures to ensure that its Satellite Broadcasting Services are free of harmful interference.

Document No. <u>318-E</u> 11 February 1977 Original : English

(Geneva, 1977)

PLENARY MEETING

FINAL PROTOCOL

For the People's Democratic Republic of Yemen

The delegation of the People's Democratic Republic of Yemen to the World Administrative Broadcasting-Satellite Conference, Geneva 1977, reserves its Government's right to take any measures it may deem necessary to protect its interests if other countries fail to respect the Final Acts and its Annexes adopted by this Conference.

(Geneva, 1977)

Document No. 319-E 11 February 1977 Original : French

PLENARY MEETING

FINAL PROTOCOL

For the Central African Empire

The delegation of the Central African Empire to the World Broadcasting-Satellite Administrative Radio Conference, desiring the geostationary orbit to be used in an equitable manner and solely for peaceful purposes, declares that its Government reserves the right to take whatever action may be required to safeguard its national and international interests should the Final Acts and the Plan annexed thereto, as drawn up by the Conference, contravene the provisions of its Constitution or infringe upon its sovereignty.

(Geneva, 1977)

Document No. 320-E 11 February 1977 Original : English

PLENARY MEETING

FINAL PROTOCOL

For the State of Bahrain and the United Arab Emirates

In signing the Final Acts of the World Administrative Radio Conference for the Planning of the Broadcasting-Satellite Service (Geneva, 1977), the delegations of the State of Bahrain and of the United Arab Emirates declare that their Administrations reserve the right to take such measures as they may deem necessary to safeguard their interests, should an Administration fail in any way to observe the provisions of the Final Acts or should the reservations made by other Administrations jeopardize the telecommunication services or liable to infringe upon the sovereign rights of the State of Bahrain and of the United Arab Emirates.



(Geneva, 1977)

Document No. 321-E 11 February 1977 Original : English

GENE

COMMITTEE 6

SUMMARY RECORD

OF THE

EIGHTH MEETING OF COMMITTEE 6

(PROCEDURES)

Saturday, 5 February 1977, at 0915 hrs and 1430 hrs

Chairman : Mr. R.J. BUNDLE (New Zealand)

Subjects discussed :		Documents Nos.
l.	Definition of service area (continued)	159
2.	First Report of Working Group 6B (continued)	185 + Corr.1
3.	Second Report of Working Group 6A	220, 210
4.	Second Report of Working Group 6B	221
5.	Approval of the summary record of the fifth meeting	203
6.	Organization of the work of Committee 6	
7.	First Report of Working Group 6D	227

1. Definition of service area (Document No. 159 (continued))

The <u>Chairman</u> recalled the discussion which had taken place at the Committee's sixth meeting concerning the definition of service area and the related Note (Document No. 159, page 2). Following the consultations he had held on the matter with the Chairman of Committee 4, he was in a position to propose the deletion of the last sentence of the Note.

The <u>delegate of the German Democratic Republic</u>, replying to a question by the <u>Chairman</u>, said that his delegation was prepared to withdraw the amendment it had proposed at the Committee's sixth meeting. Document No. 321-E Page 2

The definition of service area given in Document No. 159 was <u>approved</u> subject to the deletion of the last sentence of the related Note.

2. First Report of Working Group 6B (Document No. 185 + Corr.1 (continued))

The <u>delegate of the United Kingdom</u> said that the consultations regarding the title of the Annex between the delegations concerned and the IFRB had enabled agreement to be reached with regard to his delegation's suggestion. Accordingly, he proposed the insertion of the phrase "affecting broadcastingsatellite stations" after the words "terrestrial stations" in the third line of the title, and the deletion of the phrase "in cases where broadcasting-satellite stations are involved".

In reply to a question by the <u>delegate of the Federal Republic of</u> <u>Germany</u>, the <u>Chairman</u> said it was his understanding that the footnote to the title would not be affected by those changes.

The <u>delegates of Australia</u>, <u>Venezuela</u> and <u>Sweden</u> supported the United Kingdom proposal.

The <u>delegate of China</u> said that to amend the title as proposed by the United Kingdom delegation would be to place Region 3 countries under an obligation to apply the procedures set out in the Annex to frequency assignments outside the band 11.7 - 12.2 GHz. Since it was not within the purview of the Conference to take any action which would affect bands other than the 11.7 - 12.2 GHz band in Region 3, he proposed that the title of the Annex should be retained as it stood.

The delegates of Iran and Indonesia supported that proposal.

The <u>delegate of the United Kingdom</u> said that his delegation understood the term "other services" used in paragraph 2.1 of Administrative Council Resolution No. 762 (Document No. 1) to mean those services in the whole of the band 11.7 - 12.5 GHz which affected Broadcasting-Satellite Services in the bands 11.7 - 12.2 GHz in Regions 2 and 3 and 11.7 - 12.5 GHz in Region 1. In his view, therefore, the proposed amendment was perfectly consistent with the terms of reference of the Conference.

It would no doubt be possible to overcome any differences of opinion concerning the interpretation of the terms of reference of the Conference. However, the following factual situation would arise if the title of the Annex was not altered : the Broadcasting-Satellite Service to the USSR in the area of that country which bordered on Chinese territory would not be protected from Chinese terrestrial services, since it was planned for the USSR services in question to use the band 12.2 - 12.5 GHz. Therefore, if the procedure in the Annex did not apply to Chinese terrestrial services in the band 12.2 - 12.5 GHz, the Conference would have failed to achieve its objectives with regard to sharing criteria and protection provisions.

The delegate of the USSR considered that the title of the Annex, however it was worded, could not affect the terms of reference of the Conference and the substantive issue involved, namely, the establishment of coordination procedures for the bands 11.7 - 12.2 GHz in Regions 2 and 3 and 11.7 - 12.5 GHz in Region 1, and adequate protection of the latter from interference by

Document No. 321-E Page 3

a spinna an

services in Regions 2 and 3. The example used by the United Kingdom delegate to illustrate his point was not the only one that could be quoted; other geographical areas in Region 1 and 3 could also be affected.

The <u>delegate of China</u> said he could not agree with the views expressed by the delegates of the United Kingdom and the USSR. Not only the border area between China and the USSR but Regions 1 and 3 in their totality were involved. If Region 3 were to take on commitments outside its frequency band as laid down in the terms of reference of the Conference, it would be necessary also to discuss the question of protection of services using the 300 MHz band.

CONTRACTORY TO CONTRACTOR STRUCTURE AND A STRUCTURE A STRUCTURE AND A STRUCTURE A STRUCTURE AND A STRUCTURE A

Having called for a show of hands in order to sound out the Committee's reaction to the United Kingdom proposal, the <u>Chairman</u> said that a fairly large number of delegations appeared to be in favour of the changes in question. He therefore proposed to consider the title of the Annex as having been amended in accordance with that proposal.

The <u>delegate of China</u> reserved the right to comment on the question at a later stage.

The <u>Chairman</u> invited the Committee to consider the contents of the Annex paragraph by paragraph.

Paragraph 1

On a proposal by the <u>delegate of Japan</u>, supported by the <u>delegate of</u> the United Kingdom, it was <u>agreed</u> to delete "/ contracting /" from the first line of the paragraph.

On a proposal by the <u>Chairman of Working Group 6B</u>, it was <u>agreed</u> to amend the last phrase of the first sub-paragraph to read : "separated by less than 7 MHz and".

On proposals by the <u>delegates of Japan and the United Kingdom</u>, it was <u>agreed</u> to amend the last sub-paragraph to read : "at one or more points on the edge of the service area which is within the coverage area concerned of the Broadcasting-Satellite Service for that Administration".

> ا میں ایک ان کر انداز ہے۔ ان ہو ہیں ایک اندازہ ہوتا ہے جو ہو کر اندازہ ا

Paragraph 1, as amended, was approved.

Paragraphs 1.1 and 1.2

Approved. Approved a second state of the second sec

Approved subject to the insertion of the words "devant être" before the word "assurés" in the French text.

Paragraphs 1.4 to 1.8

al de la de la

Approved.

Paragraph 1.9

The <u>representative of the IFRB</u> said that the words "its earth station" at the end of the paragraph should be replaced by "its broadcasting-satellite station".

Following a discussion in which the <u>delegates of Algeria</u> and <u>the</u> <u>United Kingdom</u>, the <u>representative of the IFRB</u>, the <u>Vice-Chairman</u> and the <u>Chairman</u> took.part, it was <u>agreed</u> to insert the phrase "or to be rendered" after the words "service rendered" towards the end of the paragraph.

Paragraph 1.9, as amended, was approved.

Paragraph 1.10

Approved.

Paragraph 1.11

The <u>Chairman</u> said that the text in the Corrigendum should be substituted for the text between square brackets in Document No. 185.

On a proposal by the <u>delegate of the United Kingdom</u>, it was <u>agreed</u> to delete the phrase "provided that the assistance of the Board has been requested".

Paragraph 1.11 as set out in the Corrigendum, as amended, was approved.

Paragraph 1.12

On a proposal by the <u>Chairman of Working Group 6B</u>, supported by the delegate of the <u>United Kingdom</u>, it was <u>agreed</u> to delete the paragraph.

Paragraphs 2.1 to 3.14

Approved.

Paragraph 3.15

The <u>Chairman</u> said that the text in the Corrigendum should be substituted for that in Document No. 185.

Paragraph 3.15 as set out in the Corrigendum was <u>approved</u> subject to the substitution of the word "condition" for the word "coordination" in the second sentence, and consequential drafting amendments to the French and Spanish texts.

Paragraphs 3.16 to 3.20

Approved.

Paragraph 3.21

The <u>delegate of Algeria</u> observed that the term "earth stations" in the first sentence was used out of context.

The Chairman said that the appropriate correction would be made.

Paragraph 3.21 was approved on that understanding.

Paragraphs 3.22 and 3.23

Approved.

Paragraph 3.24

Approved, subject to amendment of the term "earth stations" in the first sentence to Broadcasting-Satellite Service.

Paragraphs 3.25 to 3.31

Approved:

Paragraph 3.32

The <u>Chairman</u> said that the text in the Corrigendum should be substituted for that in Document No. 185.

The <u>delegate of the United Kingdom</u> said that the word "coordination" in the second sentence should be replaced by "condition". He proposed the replacement of "IFRB" by "Board" in the last sentence. Furthermore, in order to take into account a point raised by the delegation of the USSR, he proposed that the following sentence should be added at the end of the paragraph : "The notifying administration using the frequency assignment during a specified period shall not use this circumstance subsequently for the continued use of this frequency beyond the period specified if it does not obtain the agreement of the administration or administrations concerned, as the case may be."

Paragraph 3.32 as set out in the Corrigendum, as amended, was approved.

Paragraphs 3.33 to 4

Approved.

Appendix

The <u>Chairman</u> said the Committee might wish to note that, the question having been discussed, Document No. 169 had been considered to be acceptable for the purpose of calculating power flux-density.

It was so agreed.

Document No. 185 and Corrigendum No. 1 as a whole, as amended, was approved.

Document No. 321-E Page 6

The <u>Chairman</u> said it had been proposed that the following text be incorporated in Document No. 185, possibly in the form of a new Article. It read :

"/..../ The procedures for coordination, notification and recording of assignments to terrestrial stations vis-à-vis broadcastingsatellite stations in Region 2 are contained in Article 9 of the Radio Regulations, except that the coordination area referred to in No. 492A of the Radio Regulations is replaced by the power flux-density limits provided for in / Appendix .../."

That text was <u>approved</u> for submission to Committee 8 together with Document No. 185.

3. Second Report of Working Group 6A (Documents Nos. 220, 210)

The <u>Chairman of Working Group 6A</u>, introducing its second Report (Document No. 220), explained that the Ad Hoc 4/6 Group would be reporting direct to the Committee concerning its deliberations on Appendix B which would then have to be attached to the document.

The results of the study by the Working Group on the items and the Miscellaneous Provisions referred back to it (point 1.2 in the Report) were contained in Annex 1. The numbers of the items had been left in square brackets because some renumbering was thought necessary so as to achieve a more logical order : that question could be referred to Committee 8.

Annex 1 was <u>approved</u>.

The <u>Chairman of Working Group 6A</u> drew attention to the additional wording proposed for insertion at the end of Article /4/, paragraph 4.3.2 and at the end of Article 5, paragraph 5.2.5 in Document No. 210. The new texts were given in points 1.3 and 1.4 respectively of the Working Group's Report.

Those texts were approved.

The <u>Chairman of Working Group 6A</u> explained that a set of procedures concerning coordination, notification and registration of broadcastingsatellite stations in Region 2 had been prepared by Sub-Working Group 6A2 and examined since it seemed desirable to have ready at least a preliminary text (Annex 2 of the Report) for possible inclusion in the Final Acts. However, the decision would be contingent upon the outcome of certain discussions in Committee 5, particularly in respect of Document No. 204.

The delegate of the United Kingdom pointed out that Working Group 6D had considered Document No. 204 the previous day and had proceeded on the assumption that there would be no plan. Accordingly, he proposed that Committee 6 defer examination of Annex 2 in Document No. 220 until the Chairman had had an opportunity of ascertaining from the Chairman of Committee 5 whether or not there would be a plan within the meaning of Resolutions Nos. Spa2 - 2 and Spa2 - 3. Otherwise it would be wholly unproductive to discuss Annex 2 in detail.

The views of the United Kingdom delegation concerning various aspects of the regulatory provisions which were being developed and their special implications for broadcasting satellites in Region 2 were set out in Document No. 215, paragraph 3.

The <u>delegates of Mexico and Venezuela</u> agreed with the United Kingdom delegate, particularly in view of the statement in paragraph 15 of the Annex to the Report of Working Group 5B (Document No. 204) which had yet to be considered by Committee 5.

The <u>delegate of the USSR</u> endorsed the United Kingdom delegate's proposal and regretted that the Conference had paid no heed to his delegation's warning at the outset that it would serve no purpose to set up a working group with the terms of reference assigned to Working Group 6A2. Its objection had proved fully justified.

The United Kingdom delegate's proposal to postpone further consideration of Annex 2 in Document No. 220 was <u>adopted</u>.

4. Second Report of Working Group 6B (Document No. 221)

The <u>Chairman of Working Group 6B</u>, presenting its second Report, paid tribute to Mr. Weppler and the Sub-Group over which he had presided for the arduous work done on preparing the texts contained in the Annex.

Document No. 221 was approved, subject to some editorial corrections.

The <u>delegates of Mexico</u>, <u>Cuba</u> and <u>Venezuela</u> reserved their right to comment on Document No. 221 at a later stage in the light of the ultimate decision to be taken in Committee 5 concerning the status of the Fixed-Satellite Service in Region 2.

5. Approval of the summary record of the fifth meeting (Document No. 203)

Document No. 203 was <u>approved</u>, on the understanding that a corrigendum would be issued indicating the final document numbers under item 2.

The meeting was suspended at 1200 hours and was resumed at 1430 hours.

6. Organization of the work of Committee 6

The <u>delegate of the United Kingdom</u> observed that at the present stage of the Conference it was evident that no plan would be produced for Region 2 within the meaning of Resolution No. Spa2 - 2. That being so, it would be advisable to consider what provisions should be included in the Final Acts to cover any situation that might arise as between Region 2 and Regions 1 and 3. He wondered whether Sub-Group 6A2 might not be reconvened for that purpose.

The <u>Chairman of Working Group 6A</u> suggested that, rather than reconvening Sub-Group 6A2, the Committee should set up an ad hoc group for the purpose outlined by the delegate of the United Kingdom.

It was so <u>decided</u>.

Document No. 321-E Page 8

The <u>Chairman</u> requested the Chairman of Working Group 6A to act as the convener of the Ad Hoc Group.

The <u>delegate</u> of the United Kingdom proposed that the terms of reference of the Ad Hoc Group should be as follows :

"To consider what provisions for coordination, notification and recording in the Master International Frequency Register of frequency assignments to stations in the Broadcasting-Satellite Service in Region 2 should be added to the Final Acts to provide for the necessary procedures to supplement, but in no way contradict, Resolution No. Spa2 - 3 and to consider what other provisions are necessary :

a) to protect the Broadcasting-Satellite Service in Regions 1 and 3 and,

b) the broadcasting-satellite stations in Region 2 from the Broadcasting-Satellite Service in Regions 1 and 3."

Those terms of reference were approved.

The <u>delegates of Canada, Mexico, Venezuela, the United Kingdom,</u> Japan and <u>the United States</u> said that they wished to participate in the work of the Ad Hoc Group.

The <u>delegate of the United Kingdom</u> asked the Committee to take note of Document No. 215, prepared by his delegation. Replying to a question by the <u>delegate of the USSR</u>, he explained that the last column of the table in Document No. 215 covered all broadcasting satellites in Region 2.

7. First Report of Working Group 6D (Document No. 227)

The <u>Chairman of Working Group 6D</u>, introducing the document, said that the draft texts had given rise to considerable difference of opinion within the Working Group, but were nevertheless acceptable to the majority of participants. The following errors should be noted in the text of the document, which had been prepared in great haste : in the English text of Article a, the words "an 'agreement and associated plan'" between the words "regarded as" and "in accordance with" should be replaced by "including a world agreement and associated plans for Regions 1 and 3 and for Region 2 respectively"; in the English text of resolves 1 of Resolution No. B, the words "and shall be binding on all Members" should be deleted; in the French text of Resolution No. A, under "notant a)", the words "qui s'y apporte" should be replaced by "associés".

The <u>delegate of the United Kingdom</u> said that since the idea of a Plan for Region 2 had now been abandoned, the words "and for Region 2 respectively" in the corrected text of Article a could be dropped and the "s" of the word "plan" could be deleted.

It was so decided.

The <u>Chairman</u> invited the Committee to consider the draft texts appearing in the Annex to Document No. 227.

Document No. 321-E Page 9

.....

Title and Preamble

The <u>delegate of the United Kingdom</u> said that the letter "s" appearing in brackets after the word "Plan" at the end of the Preamble should be deleted.

The <u>Chairman</u> said that the word "of" between the words "need" and "an orderly development", six lines from the bottom of the page in the English text, should be replaced by the word "for".

The title of the Annex and the draft text of the Preamble were approved with those changes.

 $(2^{n+1},\ldots,n^{n+1})^{n+1} \in \mathbb{R}^{n+1} \cap \mathbb$

Article 1 - GENERAL DEFINITIONS

and the second second

The <u>delegate of the Federal Republic of Germany</u> proposed that the words. "Geneva, 1959" should be added to the definition of the IFRB weekly circular.

The Chairman pointed out that the correct reference was to "edition 1976". He wondered whether the same reference should not be added to the definition of Regions 1, 2 and 3.

After a brief discussion, the <u>delegate of the United States of</u> <u>America</u> proposed that the reference "edition 1976" should be added only to the definition of the Radio Regulations.

It was so decided.

and the second second

The <u>delegate of the USSR</u> said that the definition of the Conference in the English text should be brought into line with the French text by the insertion of the date "1977".

With those changes, the draft text of Article 1 was approved.

Article 2 - FREQUENCY BANDS

and the second secon

The <u>delegate of the United Kingdom proposed</u> that, in view of the decision taken by the Committee in the earlier part of the meeting, the words "the Broadcasting-Satellite Service in" should be inserted between the words "apply to" and "the frequency bands", and that the following words should be added at the end of the Article : "and to the other services to which these bands are allocated".

The <u>delegates</u> of <u>Singapore</u>, the USSR, <u>Denmark</u>, <u>Hungary</u>, <u>Greece</u> and <u>the Federal Republic of Germany</u> supported that proposal.

The delegate of Japan was in favour of leaving the draft text of Article 2 unchanged.

The second of the second s

The <u>delegate of the United States of America</u> also felt that the text should remain as it stood; the point of the proposed amendment was adequately covered in the Preamble. However, if the delegate of the United Kingdom maintained his proposal, the difficulty might be overcome by further adding the , words : "in their relationship to the Bnoadcasting-Satellite Service in these bands".

Document No. 321-E

Page 10

The delegate of Japan seconded that suggestion.

The draft text of Article 2, as amended by the delegates of the United Kingdom and the United States of America, was approved.

Article 3 - EXECUTION OF THE FINAL ACTS

The <u>delegate of Canada</u> proposed the insertion of a new paragraph between paragraphs 1 and 2 reading as follows :

"The Members of the Union in Region 2 shall apply the interim provisions contained in Annex ... to the Final Acts which will govern the Broadcasting-Satellite Service in Region 2 until detailed plans for Region 2, drawn up by a future regional conference, have entered into force."

Furthermore, he proposed the addition of the words "and Annexes" at the end of old paragraph 2.

The <u>delegates of the United Kingdom</u>, the <u>United States of America</u>, Mexico, Netherlands and <u>Japan</u> seconded those proposals.

The <u>delegate of the USSR</u> suggested the deletion of the word "interim" from the text of new paragraph 2 proposed by the delegate of Canada.

The <u>delegate of the United Kingdom</u> said that he could see some value in the word "interim". Those provisions which supplemented Resolution No. Spa2 - 3 would be superseded when a regional conference for Region 2 was held in a few years' time, but others, relating to the Fixed-Satellite and Terrestrial Services, would continue in force.

The draft text of Article 3, as amended by the delegate of Canada, was approved.

Article a - APPROVAL OF THE FINAL ACTS

The <u>Chairman</u> said that the definitive numbering of Articles a, b and c and the question whether the words "final acts" should be capitalized would be entrusted to the Editorial Committee.

The draft text of Article a, as corrected by the Chairman of Working Group 6D and amended by the delegate of the United Kingdom, was <u>approved</u>.

Article b - ENTRY INTO FORCE OF THE FINAL ACTS

The draft text of Article b was approved.

Article c - DURATION OF THE PROVISIONS AND ASSOCIATED PLAN(S)

The <u>delegate of the United Kingdom</u> proposed that the letter "s" at the end of the title and the passage in square brackets at the end of paragraph 1 be deleted.

The <u>delegate of the United States of America</u> proposed that the word "regional" be inserted after "competent" in the second line of paragraph 2.

Document No. 321-E Page 11

Article c was approved, with the above two amendments and subject to the Editorial Committee's consideration of whether "Final Acts" should not read more correctly "final acts" in the text of the Article.

Resolution No. A

The <u>delegate of Canada</u> proposed that the words "for Regions 1 and \mathcal{J} " be inserted after mention of "Plan" throughout the Resolution, but withdrew the proposal in the light of objections by the <u>delegates of the USSR</u> and the United Kingdom.

Resolution No. A was approved.

Resolution No. B

Resolution No. B was <u>approved</u>, subject to deletion of the words "and shall be binding on all Members" in resolves 1.

Document No. 227 was thus approved, as amended.

The <u>Chairman</u> thanked all participants in the Committee's work for their understanding and cooperative attitude.

The meeting rose at 1610 hours.

The Secretary :

The Chairman :

R. PLUSS

R.J. BUNDLE

(Geneva, 1977)

Document No. 322-E 11 February 1977 Original : French

PLENARY MEETING

FINAL PROTOCOL

For the People's Republic of China

In service areas IND 0037 and IND 0038, which the Indian Telecommunication Administration is having included in the Plan, certain regions of Tibet and Sinkiang, both of them Chinese territories, are shown as Indian territory. This is an illegal and inadmissible infringement of Chinese sovereignty.

The Sisha and Nansha Islands, as well as the other islands of the South China Sea, have always been Chinese territory; the People's Republic of China has indisputable sovereignty over these islands and the adjacent waters. The Chinese Government has repeatedly published declarations defining its position on the subject. The use by any foreign country of the broadcastingsatellite service area for the purpose of creating confusion regarding the status of the territories is illegal and invalid.

The Chinese Government reserves the right to take whatever action may be required should the Chinese telecommunication services, during implementation of the Plan, suffer interference resulting from any infringement of the Final Acts.

(Geneva, 1977)

Document No. 323-E 11 February 1977 Original : French

PLENARY MEETING

FINAL PROTOCOL

For the Lao People's Democratic Republic

In signing the Final Acts of the World Broadcasting-Satellite Administrative Radio Conference (Geneva, 1977), the delegation of the Lao People's Democratic Republic reserves its Government's right to take whatever measures it may deem necessary to safeguard the interests of national sovereignty should any neighbouring country use a beam covering Lao territory.

Such countries should obtain the agreement of the Lao People's Democratic Republic before bringing their broadcasting-satellite stations into operation.

(Geneva, 1977)

Document No. 324-E 11 February 1977 Original : English

PLENARY MEETING

FINAL PROTOCOL

For the Kingdom of Saudi Arabia

The equivalent protection margins of Saudi Arabian beams indicated in Document No. 265 are unacceptable to the delegation of Saudi Arabia. In the meeting of Committee 5A on Thursday 10 February 1977 an indication was given by the Chairman of Committee 5A that these margins would be corrected to acceptable values. As the results of these corrective measures are not available as yet, the delegation of Saudi Arabia, therefore, wish to declare that if the results are not satisfactory, the delegation of Saudi Arabia reserve their right, in the interest of their Administration, to take any suitable action technically feasible and deemed fit to achieve satisfactory and good quality reception.



(Geneva, 1977)

Document No. 325-E 11 February 1977 Original : English

PLENARY MEETING

FINAL PROTOCOL

For the Islamic Republic of Pakistan

1. In signing the Final Acts of the World Administrative Radio Conference for the Planning of the Broadcasting-Satellite Service, Geneva, 1977, the delegation of the Islamic Republic of Pakistan hereby declares that its Government reserves the right to take any action which it considers necessary to safe-guard its interests, should any country or countries fail to observe the provisions of the Final Acts and its Annexes or the Protocols attached thereto or should the reservations made by other countries adversely affect the Satellite-Broadcasting and Telecommunications Services of the Islamic Republic of Pakistan.

2. The delegation of the Islamic Republic of Pakistan declares that the decisions of the World Administrative Radio Conference for the Planning of the Broadcasting-Satellite Service, Geneva, 1977, regarding areas falling within the territory of the disputed state of Jammu and Kashmir are without prejudice to the position recognized by the relevant resolutions of the United Nations on the question.

3. The delegation of the Islamic Republic of Pakistan has noted with concern that the Plan has frequency assignments to the Indian Administration for providing Satellite-Broadcasting Services to include coverage of a large area of the territory of Pakistan. This spillover, which is technically avoidable, is not acceptable to the Pakistan Administration. The Government of Pakistan reserves the right to take appropriate measures to ensure that its territory does not come under intentional coverage by such Indian Services.

Document No. 326-E 11 February 1977 Original : Russian

(Geneva, 1977)

PLENARY MEETING

FINAL PROTOCOL

For the Byelorussian Soviet Socialist Republic, the People's Republic of Bulgaria, the Hungarian People's Republic, the People's Republic of Poland, the German Democratic Republic, the Ukrainian Soviet Socialist Republic, the Czechoslovak Socialist Republic and the Union of Soviet Socialist Republics

In signing the Final Acts of the World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977, the above delegations reserve their Governments' right to take whatever action may be required to ensure the normal operation of their telecommunication services, should any countries fail to comply with the provisions adopted by this Conference and the Plan associated with these provisions.



(Geneva, 1977)

Document No. 327-E 11 February 1977 Original : English

PLENARY MEETING

FINAL PROTOCOL

For Thailand

In signing the Final Acts of the World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977, the delegation of Thailand reserves the right of its Government to take any measures it may deem necessary to safeguard its interest should any country or countries fail to observe the provisions of the Final Acts and its Annexes and the Protocol attached thereto or should reservations by other countries jeopardize its Broadcasting-Satellite or other radio Services.

(Geneva, 1977)

Document No. 328-E 11 February 1977 Original : French

PLENARY MEETING

FINAL PROTOCOL

For the Republic of Burundi

The delegation of the Republic of Burundi to the World Broadcasting-Satellite Radio Conference, Geneva, 1977, declares that it reserves its Government's right to take whatever action it may deem necessary to safeguard its interests should they be jeopardized by failure to observe the provisions of the Final Acts, their Annexes and the Protocol attached to the Final Acts, as adopted by this Conference.

(Geneva, 1977)

Document No. 329-E 11 February 1977 Original : French

PLENARY MEETING

FINAL PROTOCOL

For the Republic of the Chad

In signing the Final Acts of this Conference, the Chad delegation reserves its Government's right to take whatever action may be necessary to protect its services should any Member of the Union fail to observe the Final Acts of the World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977.

(Geneva, 1977)

Document No. 330-E 11 February 1977 Original : English

PLENARY MEETING

FINAL PROTOCOL

For Ethiopia

In signing the Final Acts of the World Broadcasting-Satellite Administrative Radio Conference, Geneva 1977, the delegation of Ethiopia declares that its Government reserves the right to take any measures it may deem necessary to safeguard its interests if other countries or administrations fail to observe the provisions contained in the Final Acts and the Protocol attached to it, as adopted by this Conference.



INTERNATIONAL TELECOMMUNICATION UNION

BROADCASTING SATELLITE CONFERENCE

(Geneva, 1977)

Document No. 331-E 11 February 1977 Original : English/ French/ Spanish

PLENARY MEETING

FINAL PROTOCOL

For the Republic of Colombia, the People's Republic of the Congo, the Republic of Equator, the Republic of Gabon, the Republic of Kenya, the Republic of Uganda, the Republic of Zaire

Declare that their Governments are not claiming sovereignty in space, the way it is stated in the letter and spirit of the Treaty on the Peaceful Uses of Outer Space, since there can be no doubt that these countries have always exercised sovereignty over their internationally recognized territories and within their projections.

Consequently, in line with the principles enshrined in the declaration of Bogota, signed on 3 December 1976, they wish to make the following reservations in signing the Final Acts of the Satellite-Broadcasting Conference.

First :

The delegations of the above-mentioned countries officially declare that they do not accept and accordingly are under no circumstances bound, through the signature of the Final Acts, by the Resolutions, Agreements and Decisions of this Conference regarding the location of geostationary satellites on the segments of the orbit over which these States exercise sovereign rights.

Second :

The positioning of such geostationary satellites will require the prior authorization of the equatorial country concerned and their functioning and operation shall be subject to the provisions of the national laws of the Equatorial States concerned.

Third :

The Equatorial countries reserve the right to take whatever steps they may deem fit to preserve and secure the observance of their sovereign rights which include the segments of the geostationary orbit corresponding to their respective national territories, in accordance with the constitutional and legal rules in force in each country.

For reasons of economy, this document is printed in a limited number. Participants are therefore kindly asked to bring their copies to the conference since only a few additional copies can be made available.

(Geneva, 1977)

Document No. 332-E 11 February 1977 Original : English

PLENARY MEETING

FINAL PROTOCOL

For the Federative Republic of Brazil

The Brazilian Administration reserves its position regarding the sentences in paragraph 1 of Document No. 204 which read : "As an exception to the foregoing, it is accepted that, for Greenland, a position in the geostationary satellite orbit between 55°W and 60°W may be used for the Broadcasting-Satellite Service as a primary service. All efforts should be made by the administrations concerned to allow for the sharing of a Broadcasting Satellite for Greenland and the Fixed Satellites of other administrations in Region 2 in this portion of the arc."

In this connection, the Brazilian Administration is of the opinion that a decision about the optimum orbital arc usable for Greenland or any other Administration in Region 2 must only be taken by the Regional Administrative Radio Conference which should be held not later than 1982.

The signature by Brazil of the Final Acts issued from this Conference must be interpreted only as an acceptance of the technical deliberations of this Conference. This acceptance however does not prejudge the Brazilian position concerning any future decision about the juridical regulation of the utilization of the geostationary satellite orbit which could be taken by this or by any other competent forum.



(Geneva, 1977)

Document No. <u>333-E</u> 11 February 1977 <u>Original</u> : French

PLENARY MEETING

FINAL PROTOCOL

For the Algerian Democratic and Popular Republic

The Government of the Algerian Democratic and Popular Republic, which considers invalid any action taken by Spain, the administrative power, or by Morocco or Mauritania with a view to changing in any manner whatsoever the status of Western Sahara with regard to the fundamental provisions of Resolution 1514 (XV) and the other relevant resolutions of the United Nations General Assembly and the provisions of international law, or with a view to infringing any of the lawful and sovereign rights of the Sahraoui people to its own territory, formally contests the notification of the beams : 0209 = MRC and 0223 = MTN.

The resulting deliberate spill-over both over the territory of Western Sahara and consequently over part of Algerian territory is an intentional infringement of the Radio Regulations (No. 428A), strict compliance with which should be ensured.

Western Sahara is still a responsibility of the United Nations, whose duty it is to complete the process of decolonization and to create all the conditions required to guarantee that country's independence and territorial integrity, and no provision of the Broadcasting-Satellite Conference (Geneva, 1977) may exceed the Conference's terms of reference or affect or restrict the Sahraoui people in the exercise of their sovereign rights and in the present case of their rights in matters of broadcasting.



For reasons of economy, this document is printed in a limited number. Participants are therefore kindly asked to bring their copie to the conference since only a few additional copies can be made available.

Document No. <u>334-E</u> 11 February 1977 <u>Original</u> : Spanish

(Geneva, 1977)

PLENARY MEETING

GENE

FINAL PROTOCOL

For the Republic of Venezuela

The delegation of the Republic of Venezuela declares that its Administration reserves the right to adopt whatever measures it may deem appropriate to ensure the development of its telecommunication services in the band 11.7-12.5 GHz, should its interests be adversely affected by the decisions of this Conference.
Corrigendum N° 2 au Document N° 335-F/E/S 12 février 1977



Note du Président de la Commission 5 (Planification)

PROJET DE PLAN POUR LES REGIONS 1 ET 3

Note by the Chairman of Committee 5 (Planning)

DRAFT PLAN FOR REGIONS 1 AND 3

Nota del Presidente de la Comisión 5 (Planificación)

PROYECTO DE PLAN PARA LAS REGIONES 1 y 3



	1	2	3	4	5	6	7	8	9
1	URS 067							66.3	
. 2	ARS 275	•	· . ·		1.2				
0	(2 6 10 14	18)						62 0	
2	(2 6 10 14	18)						03•2	
<u></u>									
3	$\underline{\text{BEN}}_{(DAH)} 233$	1 15 10)	••• •••						
7	URS 067	,ι ι <u>)</u> , ι,γγ						66.4	
12	URS 067				-			66.4	
18	URS 067	<u> </u>	en an eine Station	n an				66.5	
21	MLA 227			ana ang tang tang tang tang tang tang ta	• • • • • •			• • •	2
22	QAT 247			·			,	- ·	2 .
<u>ог</u>				an en	ter industriale e d'Alexie de la constation de la constation			a The card and	
25									2
<u>- 1</u>	$\left \frac{11}{(AFS)} \right (21 2)$	5 29 33	37)	14	· · ·				4.
27	UAE 274						-		2
28								67.7	
28	ÌFB 135								4
	(RHS) (22 26	30 34 3	8)	4 - -		an an ann			Т
29	TUN 150 (22-26	30 31)				135			
			<u> </u>	<u>·</u>				[· · · · · · · · · · · · · · · · · · ·
29	ISL 050	20)		a a a a					3
30	DNK 090	- 271						67.5	
30	IRQ 256								2
-									
31	BEL 018					4	x	64.1	
32	FNL 104							67•5	
33	DNK 091							 	3
34	NOR 121				(· · · · · · ·		•.	66.8	
34	BEL 018							63.5	
35	S 139							67.1	
36	NOR 121							66.9	
37	BEL 108							63.9	
38	S 138							67•4	
41	SYR 339								2
4 7	THN 272]		5.

CONFERENCE DE RADIODIFFUSION PAR SATELLITE

Corrigendum N° 1 au/ Document N° 335-F/E/S 12 février 1977

(Genève, 1977)

COMMISSION 5 COMMITTEE 5 COMISION 5

Note du Président de la Commission 5 (Planification)

PROJET DE PLAN POUR LES REGIONS 1 ET 3

Note by the Chairman of Committee 5 (Planning)

DRAFT PLAN FOR REGIONS 1 AND 3

Nota del Presidente de la Comisión 5 (Planificación)

PROYECTO DE PLAN PARA LAS REGIONES 1 y 3



Pour des raisons d'économie, ce document n'a été tiré qu'en nombre restreint. Les participants sont donc priés de bien vouloir apporter à la conférence leurs documents avec eux, car il n'y aura que fort peu d'exemplaires supplémentaires disponibles.

Canal 14	Channe	14	 Canal	14	_
	- Ondinatio		••••••		

<u> </u>	1	2	3	4	T	5	6	7	8	9 a	9 b	10		11	12		13		14	
\square	+																ļ			
2	9 ALG02510	-25.0	14	4.2	33.2	2.4 1.3	172.	39.4	1	261.	63.6	9.5	30.0	0.	* *TCD0143D	5.	* *LB¥0321D	7.	* *TGJ02260	8
3	0 ARSU2751	17.0	14	48.3	24.6	3.8 1.4	138.	36.9	2	401.	63.0	50.0	, 18 •5	3.	+ *YMS0267D	5.	+ *YEM0266D	7.	*QAT0247D	17
3	1 AUSODO60	98•0	14	135.4	-30.3	2.0 1.4	44.	39.8	1	230.	63.4	140.8	-27.7	1.	* *AUS0008D	3.	* *AUS0005D	8.	* *AUS0004D	14
3	2 AUS00086	128.0	14	145.9	-21.5	2.9 2.0	120.	36.6	2	530.	63.9	149.0	-29.0	3.	≁ ≠AUS0009D	6.	+ *AUS0006D	6.	+ *NZL0287A	17
3	3 BOT02971	-1.0	14	23.3	-22.2	2.1 1.5	36.	39.2	2	296.	63.9	25.3	-17.8	4.	*ZMB0314D	6.	+ *ZAI0323D	10.	+ ≠ROU01360	11
3	4 BRU03308	74.0	14	114.7	4.4	0.6 0.6	٥.	48.7	1	24.	62.6	114.6	4.0	9.	*BGD0220A	14.	*THA0142D	15.	*CHN0180A	18
3	5 CHN0154(62.0	14	83•9	40.5	2.8 2.0	177.	36.7	1	461.	63•4	79.0	34.3	3.	* *PAK0210B	5.	*CHN0155D	8.	+ *IND00450	15
3	6 CHN0172/	92.0	14	120.4	29.1	1.0 0.8	123.	45.2	1	82.	64.3	118.8	27.5	2.	*CHN0174A	6.	*CHN0181A	7.	* *CHN01804	10
3	7 CHN0181	80.0	14	108.5	23•8	1.4 1.1	153.	42•4	2	148.	64.1	104.5	24.5	з.	* *VTN03250	7.	*THA0142D	19.	*CHN0158A	10
3	8 CKH00521	156.0	14	-161.0	-19.8	1.0 0.6	132.	46.1	2	73.	64.8	-163.5	-17.5	15.	*SM00057D	18.	*NZL0055D	19.	≁ *WAL0102D	28
3	9 CLN0219) 50 •0	14	80.6	7.7	1.2 0.6	106.	45.7	1	63.	63.8	80.0	10.0	3.	*IND0043D	6.	*IND0045D	10.		14
4	0 D D 087	-19.0	14	9.6	49.9	1.6 0.7	147.	43.6	2	162.	65.7	10.0	54.8	-0.	*NOR0120A	3.	*ZA10323D	5.	*LUX0114D	11
4	1 GNP0304	-31.0	14	-15.0	12.0	0.9 0.6	172.	46•9	2	43.	63.3	-14.2	12.3	3.	*GMB0302D	7.	*GUI 0192D	7.	¥IRL0211D	14
4	2 GUM0331	122.0	14	144.5	13.1	0.6 0.6	0.	48.7	. 2	30.	63.5	144.7	13.4	7.	*MRA0332D	9.	*CAR0338D	17.	*AUS0008D	20
4	3 IND0037	68.0	14	93.0	25•5	1.5 1.1	40.	42.1	2	159.	64.1	96.0	29.5	2.	*CHN0155D	6.	+ +CHN0154D	7.	*THA0142D	15
4	4 IND0045	56.0	14	76.2	19.5	1.6 1.6	21.	40.3	2	218.	63.7	74.0	15.7	5.	+ *IND0043D	10.	+ *IND0039D	10.	*ARS02750	11
4	5 IRL0211	-31.0	14	-8-2	53•2	0.8 0.6	162.	47•2	1	52.	64.4	-5.5	54.3	5.	*NOR0120A	9.	*D 0087D	10.	+ *GNP03040	15
4	6 KRE0286	A 110.0	14	127.0	39.1	1.3 1.1	31.	42•7	2	134.	64.0	130.0	43.0	0.	*PNG0131D	3.	*J 0111H	7.	+ *J 0111G	7
4	7 MAU0242	29.0	14	59-8	-18.9	1.6 1.2	55.	41.2	1	195.	64.1	60.0	-14.0	5.	*ZAI 0323D	6.	*MDG0236D	14.	*ARS0275D	25
4	8 MLI0327	-37.0	14	-2.0	19.0	2.7 1.3	127.	39.0	1	344.	64.4	-2.5	23.3	· 9•	*ALG0251D	11.	*SMR0311D	18.	*LIE02530	20
4	9 MRL0333	146.0	14	166.7	7.9	1.5 1.5	177.	40.7	1	187.	63 •5	162.1	11.5	13.	+ *PNG01310	19.	+ ★CAR J338D	19.	*NCL01000	22
5	0 NCL0100) i40.0	14	166.0	-21.0	1.1 0.7	146.	45.1	1	75.	63.9	167.5	-23.0	-3.	*WAL0102D	-3.	*NHB3128D	11.	+ *AUS0008D	15
5	1 NOR0120	A 5.0	14	13.1	64•1	1.8 0.9	10.	42.2	2	195.	65 . 0	7.6	58.0	1.	+ ≭D 0087D	2.	+ *IRL0211D	14.	*POL0132D	14
5	2 PAK0210	38.0	14	72.1	30.8	1.2 0.7	90.	45.0	1	71.	63.6	75 •2	32.2	1.	*CHN0154D	2•	* IND30450	15.	+ *AFG0245D	17.
5	3 PNG0131	0 110.0	14	147.7	-6.3	2.5 2.2	169.	36.9	1	585.	64.6	141.0	-9.2	1.	*AUS0008D	6.	*INS0036D	7.	*KRE0286A	9

Canal 14 suite - Channel 14 cont. - Canal 14 cont.

	1	2	3	4		5	6	7	8	9 a	9 b	10		11	12	13	14
Γ	1 1						T			T					l — — —		
5	4 RGU0136D	-1.0	14	25.0	45 •7	1.4 0.7	155.	44•7	1	85.	64.0	20.2	46.1	-1.	*D 0J87D 4. *	*88102970 5. *	*TCH0144⊃ 7 *
5	5 TCD0143D	-13.0	14	18-1	15.5	3.4 1.7	107.	36.6	2	568.	64.1	16.5	8.2	5.	*CME0300D 7.	*MLI0327D 14. *	*GAB32630 15 *
5	6 TG00226D	-25.0	14	0.8	8-6	1.5 0.6	105.	44.6	2	78.	63.5	-0.2	11.1	-2.	*ML103270 -0. *	*ALG0251D 6. *	*DAH02330 8 *
5	7 WAL0102D	140-0	14	-176.8	-14.0	0.7 0.6	2 9 .	47.8	1	48.	64 .6	-178.1	-14.2	-2.	*NCL0100D -2.	*PNG0131D 13. *	*MRL03330 20 *
5	8 YEM0266D	11.0	14	44.3	15.1	1.1 0.7	109.	45.2	1	57.	62.8	44.0	18.0	-1.	*ARS02 75 D 0.	*YMS0267D 7. *	*UGA0051D 11 *
5	9 ZA10323D	-19.0	14	21.3	-6.8	2.8 1.5	149.	38.0	1	483.	64.8	16.3	-1.0	1.	*TCD0143D 5.	*GAB0260D 8.	¥D 0087D 8

Cana	.1 16	Char	nnel	16	. .	Canal 16											
	1	2	3	4		5	6	7	8	9 a	9 b	10)	11	12	13	14
`35	AL GO 252D	-25.0	16	1.6	25•5	3.6 2.2	152.	35•3	1	 583•	63•0	9.5	30 . 0	4.	+LBY03210 7.	+LBY0280E 12.	+MLT01470 15
36	ANDJ341D	-37.0	16	1.6	42.5	J.6 J.6	0.	48.7	2	20.	61.6	1.4	42.4	-1.	# #G 0027D 3.	* *ALG0252D 5.	* *MLI0328D 7
37	AR \$0003D	17.0	16	41.1	23.8	3.5 1.7	134.	36.5	2	421.	62.8	35.5	30.0	-u.	* *:GY00260 1.	* *ALG0252D 12.	* *LBN0279D 12
38	AUS0007D	128.0	16	145.0	-38.1	1.8 1.4	134.	40.2	2	213.	63.5	149.6	-37.5	6.	* *AUS00090 7.	* *NZL02875 14.	* *OCE01010 27
39	AUT0016D	-19.0	16	12.1	47•5	1.1 0.6	166.	45•7	2	73.	64.3	17.1	48 . J	G.	* *ZAI0322D 4.	* *F 0093± 10•	* *URS00600 12
40	BULDO20D	-1.0	16	25.0	43.0	1.0 0.6	165.	46.3	1	57.	63.8	28.1	42.0	-2.	* *MJZ0307D 3.	* *TUR0145E 5.	≄ ¥GRC0105D ε
41	CHN0169A	92.0	15	118.5	36.4	1.2 0.8	11.	44.8	1	98.	64.7	122.8	38.4	-1.	* ≠KRE0286B 1.	* *CHN0167A 8.	* *CHN0186A 8
42	CHN0186A	62.0	16	102.5	30-2	1.9 1.2	147.	40.5	2	311.	65.5	110.0	31.8	7.	* *CHN0169A 9.	* *CHN0158A 17.	* *IND0048D 17
43	CKNJ053D	158.0	16-	-163.0	-11.2	1.8 0.7	30.	43•2	2	133.	64.5	~158.0	-9.0	4.	* *UCE01010 4.	* *TON0215D 29.	* *SM000570 31
44	CPV0301D	-31.0	16	-24.0	16.0	0.9 0.7	144.	46.5	2	39.	62.4	-24.3	14.4	4.	∓G 0027D 6. *	*AZR0134D 14. *	*MLI0328D 16
45	DNK0089B	5.0	16	12.3	57.1	1.2 0.6	177.	45.7	2	74.	64•4	10.0	54 . 5	1.	*G 0027D 5. *	*POL0132E 8.	*AUT0016D 8
46	EGY0026D	-7.0	16	29 .7	26•8	2.3 1.7	136.	38.2	2	319.	63.3	36.0	23.3	-0.	*AR\$00030 0.	*ALG0252D 12.	*MOZO307D 16
47	G 0027D	-31.0	16	-3.5	53.8	1.8 0.7	142.	43.0	1	164.	65.2	1.3	51.1	3.	*ALG02520 6.	*AUT0016D 1).	*DNK00898 13
48	IND0040D	56.0	16	73.0	25.0	1.8 1.5	58.	39.9	2	244.	63.8	78.2	27.0	2.	*NPL0122A 6.	*IND00480 7.	*IND0038A 11
49	IND0048D	68.0	16	86.2	25.0	1.6 0.9	120.	42.8	2	188.	65.5	89.8	26.7	-6.	*CHN0136A -6.	*BRM0298A 9.	*IN00046A 10
50	KRE0286B	110.0	16	127.0	39.1	1.3 1.1	31.	42.7	2	135.	64.0	124.0	39.9	0.	*CHN0169A 2.	*J 0111H 9.	*CHN0186A 10
51	MAU02430	29.0	16	56.8	-13.9	1.6 1.4	65.	43.9	1	197.	63.9	53.0	-15.0	6.	*MDG0236E 9.	÷MOZ0307D 11.	+ ≠COM02070 18
52	MLA0227A	86.0	15	102.1	4.1	1.6 0.8	135.	43.0	ĩ	105.	63 •2	105.4	2.2	7.	*VTN03250 12.	*SNG0151D 13.	*INS0032A 13
53	MLD03068	44.0	16	73.1	6.0	1.0.0.6	90.	46.6	1	51.	63.7	71.0	7.5	5.	* ≄URS00698 6.	*IND0043D 17.	*ARS00030 19
54	MLI0328D	-37.0	16	-7.6	13.2	1.7 1.2	171.	40.9	i	198.	63.9	-12.0	15.0	3.	*GMB0302D 8.	≁ #GUI0192F 8.	*ALG02520 9
55	MLT01470	-13.0	16	14.3	35.9	0.6 0.6	0.	48.7	1	18.	61.2	14.3	35.9	0.	*ALGU252D 2.	*EGY0026D 11.	- *CME03D0E 12 *
56	M020307D	-1.0	16	34.0	-18.0	3.6 1.4	55.	37.3	2	505.	64.4	30.3	-15.0	5.		₩ *BUL0020D 13.	+ ≠ZAI0322D 17
57	OCE0101D	-163.0	16	-145.0	-16.3	4.3 3.5	4.	32.4	2	1352.	63 •7	-154.7	-15.7	17.	*CKN0053D 17.	+ *TON0215D 33. *	*AUS00070 35

Canal 16 suite - Channel 16 cont. - Canal 16 cont.

٠

	1	2	3	4		5	6	7	8	9 a	9 b	10)	11 .	12	<u>.</u>	13	14
													ľ					
5	58 PHL0285A	98 •0	16	121.3	11.1	3.5 1.8	99.	36•4	2	531.	63.7	122.0	21.0	7.	*AUS0005E ≉	11.	*AUSJ004D 13. *	*J 0111H 15 ≭
5	9 RRW0310D	11.0	16	30.0	-2.1	0.7 0.6	42.	48.3	2	47.	65.0	30.3	-1.0	0.	*ZAI0322D *	2.	*UGA0051D 8. *	*YMS0267E 14 *
6	0 STP0241D	-13.0	16	7.0	0.8	0.6 0.6	0.	48.7	2	20.	61.6	5 . J	2.0	-2.	*CME0300E *	2.	*ZAI0322D 3. *	*GAB0260D 6 *
6	1 TUN0215D	170.0	16	5-174.7	-18.0	1.4 0.7	85.	44•4	1	80.	63.5	-173.7	-15.9	4.	*CKN00530 *	6.	*OCE0101D 13. *	*SMA0335E 15 *
6	2 URS00600	23.0	16	41.5	57•4	3.1 1.6	153.	37.4	1	887.	6 6 •9	27.7	60.5	5.	*UR\$0061D *	8.	*DNK00898 11. *	*SOM0312D 19 *
6	3 UR SOO698	44.0	16	5 70.8	38.5	1.4 0.7	161.	44•2	2	98.	64.1	75.1	37.3	4.	*IND0040D *	6.	*MLD0306B 11. *	*IND0038A 14 *
6	4 ZAI0322D	-19.0	16	22.4	0.0	2.2 1.9	48.	38.2	1	468.	64.9	19.5	5.0	4.	*CME0300E	8.	*ALG0252D 9.	*AUT0016D 12

.

UNION INTERNATIONALE DES TELECOMMUNICATIONS CONFERENCE DE RADIODIFFUSION PAR SATELLITE

(Genève, 1977)

Document N° 335-F/E/S 11 février 1977

Page

COMMISSION 5

Note du Président de la Commission 5 (Planification)

PROJET DE PLAN POUR LES REGIONS 1 et 3 (en date du vendredi, 11 février 1977 à 12h.30)

Note en français2Explication des colonnes5Projet de Plan d'assignation de fréquence/position orbitale pour
les Régions 1 et 3 (par ordre de canal)6 - 51

COMMITTEE 5

Note by the Chairman of Committee 5 (Planning)

COMISIÓN 5

ARCHIVES

GENÈN

Nota del Presidente de la Comisión 5 (Planificación)

Pour des raisons d'économie, ce document n'a été tiré qu'en nombre restreint. Les participants sont donc priés de bien vouloir apporter à la conférence leurs documents avec eux, car il n'y aura que fort peu d'exemplaires supplémentaires disponibles. Document N° 335-F Page 2

COMMISSION 5

Note du Président de la Commission 5 (Planification)

ANALYSE DU PROJET DE PLAN PREPARE EN DATE DU VENDREDI, 11 FEVRIER 1977

L'annexe est l'analyse résumée que l'I.F.R.B. a faite du projet de Plan révisé, en date du ll février 1977.

Il a été décidé par la Commission 5 que les colonnes l à 9b, inclusivement, de cette analyse soient considérées comme le projet de Plan pour les Régions l et 3. Pour faciliter la correspondance entre les colonnes de l'analyse et celles du projet de Plan, nous reproduisons ci-après les titres des colonnes du projet de Plan :

- 1. Symbole désignant le pays et numéro de référence de l'I.F.R.B.
- 2. Position nominale sur l'orbite, en degrés
- 3. <u>Numéro du canal</u>
- 4. Coordonnées géographiques du <u>point de visée</u> (degrés et <u>dixièmes</u> de degré)
- 5. <u>Ouverture du faisceau d'antenne</u>. La colonne comporte deux valeurs représentant respectivement le grand axe et le petit axe de la section transversale du faisceau elliptique entre les points à demi-puissance
- 6. Orientation de l'ellipse
- 7. Polarisation (1 = directe, 2 = indirecte) (Rapport 321 du C.C.I.R.)
- 8. P.i.r.e. dans la direction du rayonnement maximal, en dBW
- 9. Observations

Dans l'explication relative aux colonnes du présent document (voir page 5), le numéro se rapportant à la colonne correspondante du projet de Plan est indiqué entre crochets.

> A. PETTI Président Commission 5 (Planification)

Document No. 335-E Page 3

COMMITTEE 5

Note by the Chairman of Committee 5 (Planning)

ANALYSIS OF THE DRAFT PLAN

PREPARED ON FRIDAY, 11 FEBRUARY 1977

Attached is the summary analysis by the I.F.R.B. of the revised draft Plan, as of 11 February 1977.

It has been agreed by Committee 5 that Columns 1 to 9b inclusive of this analysis be considered as the draft Plan for Regions 1 and 3. To assist in correlating the columns of the analysis with those of the draft Plan, the Column headings of the draft Plan are reproduced hereunder:

- 1. Country symbol and I.F.R.B. Serial Number
- 2. Nominal orbital position, in degrees
- 3. Channel number
- 4. Boresight geographical coordinates (in degrees and tenths)
- 5. <u>Antenna aperture</u>. This column contains two figures corresponding to the major axis and the minor axis respectively of the elliptical cross-section half power beam
- 6. Orientation of the ellipse
- 7. Polarization (1 = direct, 2 = indirect) (C.C.I.R. Report 321)
- 8. E.i.r.p. in the maximum direction in dBW
- 9. Remarks

In the explanation of the columns in the present document which follows on page 5, the related column number in the draft Plan is shown in square brackets.

A. PETTI Chairman Committee 5 (Planning

COMISIÓN 5

Nota del Presidente de la Comisión 5 (Planificación)

ANÁLISIS DEL PROYECTO DE PLAN PREPARADO EL VIERNES, 11 DE FEBRERO DE 1977

El análsis resumido que se adjunta, elaborado por la I.F.R.B., es el proyecto de Plan revisado de fecha ll de febrero de 1977.

En el seno de la Comisión 5 se ha decido que las columnas l a 9b inclusive de este análisis se consideren como el proyecto de Plan para las Regiones l y 3. A fin de ayudar en la correlación de las columnas del análisis con las del proyecto de Plan, se reproducen a continuación los encabezamientos de las columnas del proyecto de Plan:

- 1. Símbolo del país y número de orden de la I.F.R.B.
- 2. Posición orbital nominal en grados
- 3. <u>Número del canal</u>
- 4. <u>Coordenadas geográficas del eje de puntería</u> (en grados y <u>décimas</u> de grados)
- 5. <u>Abertura de la antena</u>. Esta columna contiene dos valores que representan los ejes mayor y menor, respectivamente, de la sección transversal elíptica del haz a potencia mitad
- 6. Orientación de la elipse
- 7. Polarización (1 = directa, 2 = indirecta) (Informe 321 del C.C.I.R.)
- 8. P.i.r.e. en la dirección de máxima radiación, en dBW
- 9. Observaciones.

En la explicación de las columnas que se da más adelante (página 5) en el presente documento, el número de columna correspondiente en el proyecto del Plan se indica entre corchetes.

A. PETTI Presidente Comisión 5 (Planificación)

Explication des colonnes dans le présent document

Les numéros des colonnes entre parenthèses carrées correspondent aux numéros des colonnes dans le projet de Plan.

Col.

- 1 Symbole désignant le pays et numéro de référence de l'I.F.R.B. <u>(</u>col. <u>1</u>/
- 2 <u>Position nominale sur l'orbite</u>, en degrés <u>/Col.</u> <u>2</u>/
- 3 Numéro du canal /Col. 37
- 4 Coordonnées géographiques du point de visée /Col. 4/ mais en degrés et dixièmes de degrés
- 5 Ouverture du faisceau d'antenne. Dans le cas d'une antenne elliptique, la colonne comporte deux valeurs représentant respectivement le grand axe et le petit axe de l'allipse /Col. 5_
- 6 Orientation de l'ellipse (conformément à la définition donnée dans la lettrecirculaire de l'I.F.R.B. N° 358) <u>/</u>col. 67
- 7 Gain maximal
- 8 Polarisation [Col. 7]
- 9 9a Puissance en watt_ 9b P.i.r.e. en dBW / Col. 8/
- 10 Coordonnées géographiques du point de contrôle critique pour lequel est indiquée la valeur de la marge de protection
- 11 Marge équivalente de protection
- (12 Indications relatives aux trois princi-
- 13 paux brouilleurs. Chaque colonne contient l'indication du brouilleur (symbole désignant le pays et numéro de série) et la marge de protection qui lui correspond en dB.

Explanat	<u>ion of</u>	the Columns	
in the	present	t document	

Column numbers shown between square brackets are those of the corresponding columns in the draft Plan.

Col.

114

(:L.

- 1 Country symbol and I.F.R.B. Serial Number <u>/</u>Col. <u>1</u>/
- 2 Nominal orbital position, in degrees <u>/Col. 2</u>/
- 3 Channel number / Col. 37
- 4 Boresight geographical coordinates <u>/col.4</u> but in degrees and <u>tenths</u>
- 5 Antenna aperture. For an elliptical antenna this column contains two figures corresponding respectively to the major axis and the minor axis $\sqrt{col} \cdot 5/$
- 6 Orientation of the ellipse, as defined in I.F.R.B. Circular-letter No. 358 /Col. 67
- 7 Maximal gain
- 8 Polarisation [Col. 7]
- 9 9a Power in watts 9b E.i.r.p. in dBW [Col. 3]
- 10 Geographical coordinates of the critical test point for which the protection margin is indicated
- 11 Equivalent protection margin
- (12 Indication of the three main inter-

- 5 -

- {13 ferers; each of these three columns
 - indicates the designation of the interferer (country symbol and serial number) (14 and the corresponding protection margin in dB.

Explicación de las columnas en este documento

Los números de las columnas entre corchetes corresponden a los números de las columnas en el proyecto de Plan.

Col.

- 1 Símbolo de país y número de referencia de la I.F.R.B. /Col. 1/
- 2 <u>Posición orbital nominal</u> en grados $\frac{1}{2}$
- 3 Número del canal /Col. 37
- 4 Coordenadas geográficas del centro de puntería /Col. 4/ pero en grados y décimas
- 5 Abertura del haz de antena. El se trata de una antena elíptica se indicar en la columna dos valores que representan los ejes mayor y menor, respectivamente de la elipse /Col. 5/
- 6 Orientación de la elipse (véase la carta circular N.º 358 de la I.F.R.B.) /Col. 6_7
- 7 Ganancia máxima / Col. 7/
- 8 Polarización
- 9 9a Potencia en vatios 9b P.i.r.e. en dBW 2 Col. 87
- 10 Coordenadas geográficas del punto de prueba crítico para el que se indica el valor del margen de protección
- 11 Margen de protección equivalente
- 12 Indicaciones relativas a las tres
 - fuentes principales de interferencia. En cada columna se indica la fuente de interferencia (símbolo de país y número de referencia) y el margen de protección que le corresponde en dB.

Canal 1 --- Channel 1 --- Canal 1

· ·

	1	2	3	4		5	6	7	8	9 a	9 b	10		11	12		13		14	
1	AFG0246A	50.0	1	64.5 3	 33∙1	1.4 1.4	21.	41.2	1	140.	62•7	57.7	37.3	-2.	 #URS0057A	1.	*CHN0155A	5.	*TURC145A	11.
2	AUS0005A	98.0	1	133.5 -1	18.8	2.7 1.4	76.	38.5	2	379.	64•3	130.8	-12.5	2.	+ * [NS0035A	2.	*AUS0006A	16.	*CHN0162A	18.
3	CARJ338A	122.0	1	149.5	8.0	5.4 0.8	178.	38.1	1	275.	62.5	134.6	7.5	4.	* *1NS0035A	6.	* *J 0111A	8.	* *CHN0162A	20.
4	CHN0155A	62.0	1	88+3 3	31.5	3.4 1.4	162.	37.3	2	356.	62.9	92.4	26.9	1.	* * I ND J937A	3.	* *THA0142A	7.	* *IND0044A	10.
5	5 CHN0162A	92.0	1	115.9 2	21.0	2.7 2.4	23.	36.0	2	612.	63.9	125.0	25.9	0.	* *J 0111A	1.	* *CHN0161A	7.	* *CHN01554	15.
6	5 CHN0163A	80.0	1	116.0	39•2	1.2 0.8	132.	44.4	1	100.	64•4	115.4	36.1	3.	* *CHN0162A	6.	* *J 0111A	8.	* *CHN0155A	11.
	7 CME0300A	-13.0	1	12.7	6.2	2.5 1.7	87.	37.9	1	352.	63.4	14.2	13.2	4.	* *TCD01434	5.	* *LBY0280A	12.	* *F 00934	19.
8	3 F 0093A	-19.0	1	2.6	45.9	2.5 1.0	160.	40.4	1	220.	63.8	9.5	41.2	-1.	* *SMR0311A	1.	* *D 0387A	9.	* *ALG0251A	9.
	E.IT.01934	152-0	1	179.4 -1	17.9	1-0 1-0	67.	44.2	1	89.	63.7	-178-5	-19-9	4.	* *NZ1 0055∆	4.	* *J 0111A	29.	* *INS0035A	30.
	CUT0102A	-37 0	-	-11.0	10.2	1.6 1.0	147.	42.1	2	136.	63.4	-14-5	11.5	. 4.	* *	5.	* *MI 103274	15.	* *SMR0311A	16.
		-51.0		70.7	11 7	1 2 3 4	107	45 5		+50e	42 1	71.0	****~	••	*	12	*	10	*	10
		20.0	1	12.01	11.2		IUI.	42.0		50.	05•1	71.7	12.03		*	12.	*	10.	*	15
12	2 IND0044A	68.0	1	. 79-5 7	22.3	2.2 1.4	146.	37.3	1	253.	63.3	78.2	26.9	4.	*UHNU155A	-	*CHNU154A	12.	*1ND0057A	12.
13	B INSOJ35A	104.0	1	. 124.3 -	-3•2	3.3 1.9	82.	36.1	1	508.	63•2	125.5	4.0	6.	*J 0111A *	<i>(</i> •	*CAR0338A *	15.	*CHNU162A	11.
14	4 J 0111A	113.0	1	134.5	31.5	3.5 3.3	68.	33.6	1	904.	63•2	123.7	24.3	-1.	*CHN3162A *	-0.	*PNG0131A *	11.	*INS0035A *	12.
15	5 LBY0280A	-25.0	1	21.4	26.0	2.5 1.0	119.	40.1	2	218.	63•5	22.0	33.0	5.	*TUR0145A *	8.	*ALG0251A *	11.	*F 0093A *	15.
10	5 MD.GO 236A	2 9- 0	1	46.6 -	18.8	2.7 1.1	65.	39.3	2	250.	63.3	49.0	-12.3	13.	*MAU0242A *	15.	*ZAI 0323A *	22.	*URS0067A *	29.
17	7 NZL0055A	158.0	1	172.3 -	39.7	2.9 1.6	47.	37.7	1	364.	63•3	179.0	-37.6	16.	*FJI0193A *	17.	*CKH0052A	27.	*J 0111A *	29.
13	9 PLM0337A	170.0	1	-161.4	7.0	0.6 0.6	0.	48.7	1	23.	62•4	-162.4	6.1	6.	*CAR 0338A	7.	*SMA 0335A	16.	*NZL0055A	23.
11	9 PUL0132A	-1.0	1	19.3	51.8	1.5 0.6	162.	44.5	2	91.	64.1	22.9	49.0	-0.	+ \$SMR0311A	5.	*TUR0145A	7.	*RGU0136A	7.
2	0 QAT0247A	17.0	1	51.1	25.3	0.6 0.6	0.	48.7	1	20.	61.8	52.3	24.8	2.	*Y MS0267A	5.	*ARS0275A	6.	*L BY0280A	15.
2	1 SMA0335A	173.0	1	-179.1 -	14.2	9.6 9.6	0.	48.7	2	18.	61.1	-171.0	-11.0	1.	* *PLM0337A	2.	* *NZL0055A	12.	*WAL0102A	13.
2	2 SMR0311A	-37.0	1	12.6	43.7	0.6 0.6	0.	48.7	1	24.	62•4	12.0	43.)	э.	* *F 0093A	2.	* *GUIJ192A	9.	* *TUR0145A	12.
2	3 SWZ0313A	-1.0	1	l 31.5 -	26.5	0.6 0.6	66.	48.5	1	27.	62.8	30.8	-26.6	7.	* *B0T0297A	9.	* *POL0132A	11.	* *CME0300A	26.
2	4 THA0142A	74.0	1	100.7	13.2	2.8 1.5	106.	37.9	2	374.	63-6	103.7	18.6	1.	* *CHN0162A	4.	* *LA00284A	5.	* *CHN0155A	11.
2	5 TUR01454	5.0	1	34.4	38.9	2.7 1.0	168-	39.8	1	246.	63.7	42.7	41.5	-2.	+ +UR\$0064A	-1.	* *LBY0280A	11.	* *URS00674	12.
2	6 UD 530644	22.0	1	45.6	40.9	2.2 1) 4	163	43.1	- 2	110.	63.0	41.1	43.4		* *TIR01454	2.	* *!!R\$00674	12-	* *1.8¥02804	12-
12	G UN JUUGHA	2304		L TJ€U	TV 0	202 300	T034		2	** 24	0.50.7			**						

• •

۰.

___ 1 ___

Canal 1 suite — Channel 1 cont. — Canal 1 cont.

	1	2	3	4		5	6	7	8	9 a	9 b	10		11	12	13	14
						1											
27	7 URSD0674	44.0	1	62.4	58.5	3.2 1.5	169.	37.4	1	877.	66.8	50.9	51.9	14.	*TUR0145A 17.	*AFG0246A 21.	*F 0093A 24.
i															*	*	*
28	8 WAK0334A	140.0	1	166.5	19.2	0.6 0.6	0.	48.7	1	31.	63.6	166.5	19.2	17.	*J 01114 23.	*MRL0333A 24.	*NCL0100A 25.
1															*	*	*
2	9 YM30267A	11.0	1	48.8	15.2	1.8 1.5	176.	39.9	2	195.	62.8	49.5	18.4	5.	*ARS0275A 6.	*YEM0266A 13.	*LBY0280A 19.

Canal 2 — Channel 2 — Canal 2

		1	2	3	4		5	6	7	8	9 a	9 b	10		11	12	13	14
							I.					[1		
3	0 /	ALG0251A	-25.0	2	4-2	33•2	2.4 1.3	172.	39.4	1	251.	63.4	9.5	30.0	0.	*TCD0143A 5.	*LBY0321A 7.	*TG00226A 8.
3	1 /	ARS0275A	17.0	2	48.3	2 4.6	3.8 1.4	138.	36.9	2	380.	62.7	42.5	17.0	-11.	*YEM0266A-11.	*YMS0267A -2.	*TCD0143A 12.
3	2	AUS0006A	98.0	2	135.4 -	30.3	2.0 1.4	44.	39.8	1	221.	63.2	140.8	-27.7	1.	* *AUS0008A 3.	* *AUS0005A 8.	* *AUS0004A 14.
3	3 4	AUSOOOBA	128.0	2	145.9 -	21.5	2.9 2.0	120.	36.6	2	510.	63.7	149.0	-29.0	3.	* *AUS0009A 6.	* *AUS0006A 6.	* *AUS0005A 27.
3	4 8	30T0297A	-1.0	2	23.3 -	22.2	2.1 1.5	36.	39.2	2	285.	63.7	25.3	-17.8	4.	* *ZMB0314A 6.	* *ZAI9323A 19.	* *ROU0136A 11.
3	15 (CHN0154Å	62.0	2	83.9	40.5	2.8 2.0	177.	36.7	1	443.	63.2	79.0	34.3	-0.	* *PAK0127A 1.	* *CHN0155A 8.	* *CHN0157A 14.
3	6 (CHN0161A	92.0	2	118-1	31.1	2.5 1.7	117.	38•0	1	436.	64.4	122.6	37.4	1.	* *KOR J112A 3.	* *CHN0160A 7.	* *CHN0162A 13.
3	7 (CKH0052A	158.0	2	-161.0 -	19.8	1.9 0.6	132.	46.1	2	71.	64•6	-163.5	-17.5	15.	* *SM00057A 18.	* *NZL0055A 19.	* *WAL0102A 28.
3	8 (LN0219A	50.0	2	80.6	7.7	1.2 0.6	106.	45•7	1	61.	63.6	80.0	10.0	3.	**************************************	* *IND0045A 10.	* *AFG0246A 14.
3	9 [0087A	-19.0	2	9.6	49.9	1.6 0.7	147.	43.6	2	156.	65•5	10.0	54.8	1.	* *ZAI0323A 5.	*FNL0103A 6.	*LUX0114A 11.
4	0 1	FNL0103A	5•0	2	22.5	64.5	1.4 0.8	171.	44•Û	2	234.	67.7	19•2	60.0	6.	*D 0087A 8.	*POL0132A 13.	* *TUR0145A 20-
4	1 0	GNP0304A	-31.0	2	-15.0	12.0	0.9 0.6	172.	46.9	2	42.	63.1	-14.2	12.3	2.	*GMB0302A 6.	*GUI0192A 7.	*IRL0211A 14.
4	2 (GUM)331A	122.0	2	144.5	13.1	0.6 9.6	0.	48.7	2	29.	63.3	144.7	13.4	7.	* MRA0332A 9.	*CAR 0338A 17.	*AUS0008A 19.
4	3	IND0037A	68.0	2	93.0	25.5	1.5 1.1	40.	42.1	2	153.	63.9	96.0	29•5	-1.	*CHN0157A 4.	*CHN0155A 5.	*CHN0154A 7.
4	4	IND0045A	56.0	2	76.2	19.5	1.6 1.6	21.	40.3	2	213.	63.5	74.4	22•0	3.	*PAK0127A 5.	*CHN0154A 14.	*ARS0275A 15.
4	5 1	INS0028A	80.0	2	101.5	0.0	3.0 1.2	133.	38•7	2	298.	63•3	104.4	0.9	0.	*MLA0228A 4.	*SNG0151A 4.	*THA0142A 11.
4	5	IRL0211A	-31.0	2	-8.2	53.2	0.8 0.6	162.	47.2	1	50.	64.2	-5.5	54.3	7.	*D 0087A 10.	*GNP0304A 15.	*F 0093A 16.
4	7 7	(OR)112A	113.0	2	127.5	36.0	1.2 1.0	168.	43.2	2	108.	63.6	126.2	33.0	-3.	*CHN0161A 1.	*PNG0131A 2.	*J 01113 5.
4	3 1	A002844	74.0	2	103.7	18.1	2.2 2.8	133.	42.)	1	150.	63.8	102.)	22.5	3.	*THA 0142A 7.	+ *IND0037A 9.	*CHN0157A 12.
4	91	14002424	29.0	2	59.8 -	18.9	1.6 1.2	55.	41.2	1	138.	64.0	60.0	-14.0	5.	*ZAI0323A 6.	*MDG0236A 14.	*PAK0127A 21.
5	0 1	ML A0228A	86.0	2	114.1	3.9	2.3 1.1	45.	40.1	1	224.	63.6	109.0	2•4	3.	*INS0028A 5.	*VTN0325A 11.	*CHN0161A 17.
												2 —					suite -	— cont.
	-			•	•))		• .~	

Canal 2 suite — Channel 2 cont. — Canal 2 cont.

	1	2	3	4		5	6	7	8	9 a	9 b	10		11	12		13		14	
51	ML 10327A	-37.0	2	-2.0	19.0	2.7 1.3	127.	39.0	1	330.	64.2	-2.5	23.3	9.	≠ALG0251A	11.	*SMR.0311-	18.	*L IE0253A	20.
52	MRL0333A	146.0	2	166.7	7.9	1.5 1.5	177.	40•7	1	130.	63.3	162.1	11.5	13.	* *PNG0131A	19.	* *CAR0338	19.	* *NCL010J4	22.
53	UCLOIODA	149-0	2	166.0	-21.0	1.1 J.7	146.	45.1	1	72.	63.7	167.5	-23.0	-3.	≠ ¥₩AL 3132A	-3.	* *NHB0128	11.	* *AUS0008A	15.
54	P4K0127A	38.0	2	69.6	29.5	2.3 2.2	14.	37.3	1	461.	63.9	74.5	35.7	2.	* *じHN 0154A	2.	* *4FG0245A	14.	* *AFG0246A	21.
55	PNG0131A	110.0	2	147.7	-6.3	2.5 2.2	169.	36.9	1	562.	64.4	141.0	-9.2	2.	* *AUS0008A	6.	* *INS0036	7.	* *KOR 01124	10.
56	R0U0136A	-1.0	2	25.0	45.7	1.4 0.7	155.	44.7	1	92 .	63.8	20.2	46.1	-1.	* *0 0087A	4.	* *B0T0297A	5.	* *TCH0144A	7.
57	TCD0143A	-13.0	2	18.1	15.5	3.4 1.7	107.	36.6	2	547.	64.0	16.5	8.2	5.	* *CME0300A	7.	* *ML10327~	14.	* *gabo260a	15.
58	TGDD226A	-25.0	2	0. 8	8.6	1.5 0.6	105.	44.6	2	75.	63.4	-0.2	11.1	-2.	* *4LI0327A	-0.	* *ALG0251A	6.	* *DAHJ233A	7.
59	WALJ102A	140.0	2	-176.8	-14.0	J.7 0.6	29.	47.8	1	46.	64.4	-178.1	-14.2	-2.	* *NCL0100A	-2.	* *PNG0131A	13.	* *FJI0193A	18.
63	YEMJ266A	11.0	2	44.3	15.1	1.1 0.7	139.	45.2	1	55.	62.6	44.0	18.0	-1.	* *ARS02 7 5A	0.	* *YMS02674	7.	* *UGA0051A	11.
61	ZA I 0323A	-19.0	2	21.3	-6.8	2.8 1.5	149.	38.0	ł	465.	64•6	16.3	-1.0	1.	* *TCD9143A	5.	* *GAB0260	7.	* *CME0300A	8.
ł	•	_				2												• •		ſ
Cai	nal 3	Channe		·	Canal	ు —			Tat						T					
1	1 1		2		,	L				U 2							1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1			
├		2	3			5	0		╞┻┤	30	30	0			12		13		14	
33	AFGJ245A	50.0	3	70.2	35.5	1.3 1.1	53.	42.5		108.	62.8	68.0	31.6	0.	= 12 ===================================	2.	+PAK01274	11.	+PAK0283A	13.
33 34	AFGJ245A AUSODO4A	50.3 98.0	3	70.2 121.8	35.5 -24.9	1.3 1.1 3.6 1.9	53. 54.	42.5 35.9	1 2	108. 514.	62.8 63.0	68.0 126.6	31.6 -31.0	0. 8.	+IAND1094 * *AUS0006A	2.	*PAK01274 * *AUS00098	11. 21.	+PAK0283A * *INS0036A	13.
33 34 35	AFGJ245A AUSO004A AUS0009A	2 53.3 98.0 128.0	3	3 79.2 3 121.8 3 147.2	35.5 -24.9 -32.0	1.3 1.1 3.6 1.9 2.1 1.4	53. 54. 15.	42.5 35.9 39.6	1 2 1	178. 514. 281.	62.8 63.0 64.1	68.0 126.6 141.0	31.6 -31.0 -34.0	0. 8. 4.	+ i kN01094 * * AUS 0006A * AUS 0007A	2. 8. 6.	*PAK01274 * *AUS00J9A * *AUS00J8A	11. 21. 13.	+PAK02834 * *INS0036A * *PNG0271A	13. 21. 15.
33 34 35 36	AFGJ245A AUSO004A AUS0009A AUS0009A AZR0134A	50.3 98.0 128.0 -31.0	3	70.2 3 121.8 3 147.2 3 -23.4	35.5 -24.9 -32.0 36.1	1.3 1.1 3.6 1.9 2.1 1.4 2.6 0.7	53. 54. 15. 158.	42.5 35.9 39.6 41.7	1 2 1 2	178. 514. 281. 135.	52.8 63.0 64.1 63.J	68.0 126.6 141.0 -16.9	31.6 -31.0 -34.0 32.5	0. 8. 4. -5.	+ i kN01094 * * 4US 3036A * * 4US 0007A * * 20R 0133A	2 • 8 • 6 •	*PAK01274 * *AUS00008A * *AUS0008A * *LBR0244A	11. 21. 13. 9.	+PAK0283A * *INSJ036A * *PNG0271A * *CPV0301A	13. 21. 15. 14.
33 34 39 30 31	AFGJ245A AUSODO4A AUSODO9A AUSODO9A AZRO134A CHN0157A	50.3 98.0 128.0 -31.0 62.3	3	70.2 3 121.8 3 147.2 3 -23.4 3 102.3	35.5 -24.9 -32.0 36.1 27.8	1.3 1.1 3.6 1.9 2.1 1.4 2.6 0.7 2.6 1.6	53. 54. 15. 158. 127.	42.5 35.9 39.6 41.7 38.2	1 2 1 2 2 2	178. 514. 281. 135. 487.	52.8 63.0 64.1 63.0 65.1	68.0 126.6 141.0 -16.9 101.7	31.6 -31.0 -34.0 32.5 21.3	0. 8. 4. -5. 1.	+ i kN01094 * * * *AUS J J J J 6A * * * * 000 01 33A * * * TN0325A	2 • 8 • 6 • • - 4 •	+PAK01274 * *AUS00J9A * +AUS00J8A * +LBR0244A * *	11. 21. 13. 9.	+PAK0283A * *INS0036A * *PNG0271A * *CPV0301A * *LA302844	13. 21. 15. 14.
33 34 35 36 31	AFGJ245A AUSODO4A AUSODO9A AZRO134A CHN0157A B CHN0160A	50.3 98.0 128.0 -31.0 62.3 92.0		70.2 121.8 147.2 -23.4 102.3 122.8	35.5 -24.9 -32.0 36.1 27.8 45.3	1.3 1.1 3.6 1.9 2.1 1.4 2.6 0.7 2.6 1.6 2.5 1.4	53. 54. 15. 158. 127. 150.	42.5 35.9 39.6 41.7 38.2 38.7	1 2 1 2 2 2	178. 514. 281. 135. 487. 438.	52.8 63.0 64.1 63.J 65.1 65.1	68.0 126.6 141.0 -16.9 101.7 122.5	31.6 -31.0 -34.0 32.5 21.3 38.3	0. 8. 4. -5. 1. -0.	+ i kN01094 + AUS J J J 6A + + + AUS J J J 6A + + + + + + + + + +	2 • 8 • 6 • • - 4 • • 2 •	*PAK01274 * *AUS00098A * *LBR0244A * *LAU0284B * *CHN0157A	11. 21. 13. 9. 14. 7.	+PAK0283A + *INS0036A + + PNG0271A + * *CPV0301A + * *LA00284A * *	13. 21. 15. 14. 14. 14. 3 7.
33 34 35 36 31 38	AFGJ245A AUSO004A AUS0009A AZR0134A CHN0157A CHN0157A CHN0160A	50.3 98.0 128.0 -31.0 62.3 92.0 29.0		4 3 70.2 3 121.8 3 147.2 3 -23.4 3 102.3 3 122.8 3 44.1	35.5 -24.9 -32.0 36.1 27.8 45.3 -12.1	1.3 1.1 3.6 1.9 2.1 1.4 2.6 0.7 2.6 1.6 2.5 1.4 0.8 0.6	53. 54. 15. 158. 127. 150. 149.	42.5 35.9 39.6 41.7 38.2 38.7 47.7	1 2 1 2 2 2 2	178. 514. 281. 135. 487. 438. 35.	52.8 63.0 64.1 63.0 65.1 65.1 63.1	68.0 126.6 141.0 -16.9 101.7 122.5 43.1	31.6 -31.0 -34.0 32.5 21.3 38.3 -11.3	0. 8. 4. -5. 1. -0. 5.	+ IKN01094 + AUS DDD6A + + + AUS DDD6A + + + AUS D007A + + + + D01133A + + + - - - - - - - - - - - - -	2. 8. 6. -4. 2. 4. 9.	*PAK01274 *AJS00J9A *AJS00J8A *AUS00J8A * *LBR0244A * *L4D0284B * *CHN0157A *	11. 21. 13. 9. 14. 7. 10.	+PAK0283A *INS0036A * +PNG0271A * +CPV0301A * * +LA00284A * * + + CHNJ161E * *	 13. 21. 15. 14. 14. 4. 14. 7. 4. 16.
33 34 35 36 35 38 38 38 39 40	AFGJ245A AUSO004A AUSO009A AZRO134A CHN0157A CHN0157A CHN0160A COM0207A DAH0233A	50.3 98.0 128.0 -31.0 62.3 92.0 29.0		70.2 3 70.2 3 121.8 3 147.2 3 -23.4 3 102.3 3 122.8 3 44.1 3 2.2	35.5 -24.9 -32.0 36.1 27.8 45.3 -12.1 9.5	1.3 1.1 3.6 1.9 2.1 1.4 2.6 0.7 2.6 1.6 2.5 1.4 0.8 0.6 1.4 0.7	53. 54. 15. 158. 127. 150. 149. 97.	42.5 35.9 39.6 41.7 38.2 38.7 47.7 44.3	1 2 1 2 2 2 2 2	178. 514. 281. 135. 487. 438. 35. 78.	50 62.8 63.0 64.1 63.0 65.1 65.1 63.1 63.3	68.0 126.6 141.0 -16.9 101.7 122.5 43.1 2.2	31.6 -31.0 -34.0 32.5 21.3 38.3 -11.3 6.2	0. 8. 4. -5. 1. -0. 5. 3.	+ i kN01094 * AUS J J J 6A * * AUS J J 36A * * * POR 0133A * * * V TN0325A * * J J111B * * SOM0312A * * * * * * * * * *	2. 8. 6. 4. . 2. . 4. . 9.	+PAK01274 + * * * * * * * * * * * * *	11. 21. 13. 9. 14. 7. 10.	+PAK0283A * *INS0036A * *PNG0271A * *CPV0301A * *CPV0301A * *CHN0161E * * *Aug0284A * * * * * * * * * * * * *	 13. 21. 15. 14. 14. 7. 16. 14.
33 34 35 36 31 38 38 38 40 41	AFGJ245A AUSO004A AUSO009A AZRO134A CHN0157A CHN0157A CHN0160A COM0207A DAH0233A GAB0260A	50.3 98.0 128.0 -31.0 62.0 92.0 29.0 -19.0		4 70.2 3 121.8 3 121.8 3 147.2 3 -23.4 3 102.3 3 102.3 3 122.8 3 44.1 3 2.2 3 11.8	35.5 -24.9 -32.0 36.1 27.8 45.3 -12.1 9.5 -0.6	1.3 1.1 3.6 1.9 2.1 1.4 2.6 0.7 2.6 1.6 2.5 1.4 0.8 0.6 1.4 0.7 1.4 1.1	53. 54. 15. 158. 127. 150. 149. 97. 64.	42.5 35.9 39.6 41.7 38.2 38.7 47.7 44.3 42.2	1 2 2 2 2 2 2 1	178. 514. 281. 135. 487. 438. 35. 78. 130.	52.8 63.0 64.1 63.0 65.1 65.1 63.1 63.3 63.3	68.0 126.6 141.0 -16.9 101.7 122.5 43.1 2.2 11.3	31.6 -31.0 -34.0 32.5 21.3 38.3 -11.3 6.2 -4.0	0. 8. 4. -5. 1. -0. 5. 3. 3.	*IANO1094 * *AUSJJJ6A * *AUSJJJ6A * *AUSJ0007A * *POR0133A * *VTN0325A * *J 0111B * *SOM0312A * * *GG9226A * *	2. 8. 6. 2. 2. 4. 9. 5.	*PAK01274 * *AUS00J8A * *AUS00J8A * *LBR0244A * *LBR0244A * *CHN0157A * *ZMB0314A * *ZAI03224	11. 21. 13. 9. 14. 7. 10. 14. 9.	+PAK0283A * *INS0036A * *PNG0271A * *CPV0301A * *CPV0301A * * *CPV0301A * * * * * * * * * * * * *	 13. 21. 15. 14. 14. 7. 16. 14. 14. 14. 14.
33 34 35 36 31 38 31 34 40 42	AFGJ245A AUSODO4A AUSODO4A AZRO134A CHN0157A CHN0160A CCM0207A DAH0233A GAB0260A CGM0302A	50.3 98.0 128.0 -31.0 62.3 92.0 29.0 -19.0 -13.0 -37.0		70.2 121.8 147.2 147.2 123.4 102.3 122.8 44.1 2.2 11.8 11.8 11.8	35.5 -24.9 -32.0 36.1 27.8 45.3 -12.1 9.5 -0.6 13.4	1.3 1.1 3.6 1.9 2.1 1.4 2.6 0.7 2.6 1.6 2.5 1.4 0.8 0.6 1.4 0.7 1.4 1.1 J.8 0.6	53. 54. 15. 158. 127. 150. 149. 97. 64. 4.	42.5 35.9 39.6 41.7 38.2 38.7 47.7 44.3 42.2 47.5	1 2 2 2 2 2 1 2	178. 514. 281. 135. 487. 438. 35. 78. 139. 38.	52.8 63.0 64.1 63.0 65.1 65.1 63.1 63.3 63.3 63.3	68.0 126.6 141.0 -16.9 101.7 122.5 43.1 2.2 11.3 -14.2	31.6 -31.0 -34.0 32.5 21.3 38.3 -11.3 6.2 -4.0 13.3	0. 8. 4. -5. 1. -0. 5. 3. 3. 3.	+ i kN01094 + i kN01094 + + + AUS J J J 6A + + + + + + + + + + + + +	2 • 8 • 6 • 7 • 4 • 9 • 5 • 7 •	*PAK01274 * *AUS00J8A * *AUS00J8A * *LBR0244A * *LAU0284B * *CHN0157A * *ZMB0314A * *ZAI0323A * *ZAI03224 *	11. 21. 13. 9. 14. 7. 10. 14. 9. 9.	+PAK0283A + *INS0036A + *DNG0271A + *CPV0301A * *LA00284A * *CHN0161E * *AU0243A * * *ZAI0322A * * * * * * * * * * * * *	A 13. A 13. A 15. A 14. A 14. B 7. A 16. A 14. A 14. A 14. A 14.
33 34 35 36 35 36 36 36 46 46 47 47 47	AFGJ245A AUSO004A AUSO004A AUSO009A AZRO134A CHN0157A CHN0157A CHN0160A COM0207A COM0207A DAH0233A GAB0260A CAB0260A CAB0260A CAB0260A	59.3 98.0 128.0 -31.0 62.3 92.0 29.0 -19.0 -13.9 -37.0 5.0		70.2 121.8 121.8 147.2 147.2 123.4 102.3 122.8 44.1 2.2 11.8 2.12.8 11.8 2.2 3.122.8 3.122.8 3.122.8 3.122.8 3.122.8 3.122.8 3.122.8 3.122.8 3.122.8 3.122.8 3.122.8 3.122.8 3.122.8 3.122.8 3.122.8 3.122.8 3.122.8 3.122.8 3.122.8 3.123.8 3.124.7	35.5 -24.9 -32.0 36.1 27.8 45.3 -12.1 9.5 -0.6 13.4 38.2	1.3 1.1 3.6 1.9 2.1 1.4 2.6 0.7 2.6 1.6 2.5 1.4 0.8 0.6 1.4 0.7 1.4 1.1 J.8 J.6 1.6 1.0	53. 54. 15. 158. 127. 150. 149. 97. 64. 4. 156.	42.5 35.9 39.6 41.7 38.2 38.7 47.7 44.3 42.2 47.5 41.8	1 1 2 2 2 2 2 2 1 2 1 2 1	178. 514. 281. 135. 487. 438. 35. 78. 139. 38. 140.	52.8 63.0 64.1 63.0 65.1 65.1 63.1 63.3 63.3 63.3 63.3	68.0 126.6 141.0 -16.9 101.7 122.5 43.1 2.2 11.3 -14.2 26.6	31.6 -31.0 -34.0 32.5 21.3 38.3 -11.3 6.2 -4.0 13.3 41.5	0. 8. 4. -5. 1. -0. 5. 3. 3. 3. 1.	+ i kN01094 + AUS J J J 6A + + + AUS J J J 6A + + + + + + + + + +	2. 8. 6. 2. 4. 9. 5. 5. 7. 4.	*PAK01274 *AUS00J8A *AUS00J8A * *LBR0244A * *LAU0284B * *CHN0157A * *ZMB0314A * *ZAI0323A * *ZAI03224 * *MLI0328A *	11. 21. 13. 9. 14. 7. 10. 14. 9. 9. 6.	+PAK0283A + *INS0036A + *DNG0271A + *CPV0301A * *LA00284A * *LA00284A * * * * * * * * * * * * *	 13. 21. 15. 14. 4. 4.
33 34 35 36 31 38 39 40 41 42 42 42	AFGJ245A AUS0004A AUS0009A AZR0134A CHN0157A CHN0157A CHN0160A COM0207A COM0207A CAB0260A CAB0260A GAB0260A GM50302A GRC0105A IND0043A	59.3 98.0 128.0 -31.0 62.3 92.0 29.0 -19.0 -13.9 -37.0 5.0		4 3 70.2 3 121.8 3 121.8 3 121.8 3 121.8 3 122.3 3 102.3 3 122.8 3 44.1 3 2.2 3 11.8 3 -15.1 3 24.7 3 77.8	35.5 -24.9 -32.0 36.1 27.8 45.3 -12.1 9.5 -0.6 13.4 38.2 11.1	1.3 1.1 3.6 1.9 2.1 1.4 2.6 0.7 2.6 1.6 2.5 1.4 0.8 0.6 1.4 1.1 J.8 0.6 1.8 1.0 1.4 1.3	53. 54. 15. 158. 127. 150. 149. 97. 64. 4. 156. 172.	42.5 35.9 39.6 41.7 38.2 38.7 47.7 44.3 42.2 47.5 41.8 41.8	1 2 2 2 2 2 1 2 1 1 1	178. 514. 281. 135. 487. 438. 35. 78. 139. 38. 140. 141.	50 62.8 63.0 64.1 63.0 65.1 65.1 63.1 63.3 63.3 63.3 63.3 63.3	68.0 126.6 141.0 -16.9 101.7 122.5 43.1 2.2 11.0 -14.2 26.6 79.5	31.6 -31.0 -34.0 32.5 21.3 38.3 -11.3 6.2 -4.0 13.3 41.5 9.1	0. 8. 4. -5. 1. -0. 5. 3. 3. 3. 1. 4.	+ i kN01094 + i kN01094 + AUS J J J 6A + AUS J J 36A + AUS D 007A + J 01133A + VTN0325A + J 01118 + SOM0312A + SOM0312A + AUS J 26A + AUS	2. 8. 6. 2. 4. 9. 5. 5. 7. 4. 5.	*PAK01274 *AUS00008A *AUS00008A *LBR0244A *LBR0244A *CHN0157A *CHN0157A *ZAH00314A *ZAI0322A *MLI0328A *RDU0136A *IND0045A	11. 21. 13. 9. 14. 7. 10. 14. 9. 9. 6.	+PAK0283A * *INS0036A * *PNG0271A * *CPV0301A * *CPV0301A * *CHN0161E * * *CHN0161E * * * * * * * * * * * * *	 13. 21. 15. 14. 14. 7. 14. 7. 14. <l< td=""></l<>

. .

i

.

_ 3 _

Canal 3 suite --- Channel 3 cont. --- Canal 3 cont.

	1	2	3	4		5	6	7	8	9 a	9 b	10		11	12	13	14
46	5 INS0036A	104.0	3	135.2	-3.8	2.5 2.3	147.	37.3	1	439.	63.8	128.)	-7.0	. 4.	*AUS0004A 5.	 ≠J 0111B 15. ≭	*PNG0131A 16.
47	IRND109A	34.0	3	54•2	32•4	3.8 1.8	149.	35.8	2	497.	62.8	61.5	31.0	6.	*AFG0245A 8.	*PAK0127A 15.	*SOM0312A 16.
48	5 J 0111B	110.0	3	134.5	31.5	3.5 3.3	68.	33.6	1	1145.	64.2	129.3	34.7	1.	* *CHN0163A 7.	* *KOR3112B 8. *	* *KOR0112A 8.
49	LBN0279A	11.0	3	35.8	33.9	0.6 0.6	9.	48.7	2	20.	. 61.6	36.8	34.5	-0.	*GRC0105A 6.	*IRN0109A 6.	*ARS0003A 7.
5) LBKJ244A	-31.0	3	-9.3	6.6	1.2 0.7	133.	44.9	1	68.	63•2	-10.2	8.5	4.	*MLI0328A 9.	*AZR0134A 9.	≁ *POR0133A 12. *
51	L BY0321A	-25.0	3	13.1	27.2	2.4 1.1	129.	40.0	2	230.	63.0	11.4	33.5	3.	*ALG7251A 6.	*ALG0252A 9.	*TGD0226A 12.
52	2 LIE0253A	-37.0	3	9.5	47.1	0.6 0.6	0.	48.7	1	24.	62.4	9.5	47.2	1.	*TCH0144A 4.	*LUX0114A 8. *	* *AND0341A 13. *
53	3 LUX0114A	-19.0	3	6.0	49.8	0.6 0.6	0.	48.7	1	26.	62•9	7.0	48•5	-4•	*LIE0253A 1.	*D 0087A 3.	*TCH0144A 4.
54	MRA0332A	122.9	3	145.9	16.9	1.2 0.6	76.	45.7	1	60.	63.5	145.0	20.0	-4.	*J 01118 -4.	*GUM0331A 12.	*AUS0009A 19. *
55	5 NHBJ128A	140.0	3	168.0 -	16.4	1.5 0.7	87.	44.1	2	75.	62•8	169.8	-20.0	7.	*NCL0100A 8.	≠NRU0309A 23.	*INS0036A 23.
56	5 NRU0309A	134.9	3	167.)	-0 . 5	0.6 û.6	0.	48.7	2	24.	62.5	166.0	-2.0	10.	*NHB0128A 14.	* *INS0036A 17. *	* *PNG0131A 19. *
57	PORO133A	-31.0	3	-8.0	39.6	J.9 0.6	112.	46.8	2	45.	63•4	-7.5	37.2	-15.	*AZR0134A-15.	*LBRJ244A 11.	*G 0027A 14.
58	8 SM00057A	158.0	3	-172.3 -	-13.7	0.6 0.6	0.	48.7	1	31.	63.6	-171.0	-14.1	7.	*CKN00534 9. *	+ *CKH0052A 14• *	≁ *WAL0102A`20. *
59	9 SNG0151A	74.0	3	103.8	1.3	0.6 0.6	0.	48.7	2	31.	63.5	106.0	1.1	-1.	*VTN0325A 4.	*INS00288 4.	*INS0028A 4.
60	SOM0312A	23.0	3	45.0	6.4	3.3 1.5	71.	37.2	1	323.	62.3	43.2	11.2	5.	*URS0060A 8.	+ *URS00614 10. *	*YEM0266A 15.
61	TCH0144A	-1.0	3	17.3	49.3	1.5 0.6	170.	44.8	2	80.	63.8	12.1	50.3	-2.	*LIE0253A 3.	*ZMB0314A 5.	*LUX0114A 6.
62	2 UGA3051A	11.0	3	32•3	1.2	1.5 1.1	60.	42.1	1	129.	63.2	29.8	-1.3	1.	* *RRW0310A 4.	*ZMB0314A 8. *	*ZAI03224 16.
63	B URS0061A	23.0	3	24.7	56.6	0.9 0.6	12.	46.7	2	67.	65.0	25.8	54.1	-0.	*TCH0144A 6.	*SOM0312A 6.	*URS0060A 6.
64	URS0073A	44.0	3	54. 3	63.5	1.6 0.7	3.	44.1	1	192.	66.9	49.5	59.3	10.	+IRN0109A 13.	+ *URS0060A 16. ≠	*AFG0245A 21. *
65	5 VTN0325A	86.0	3	105.3	16.1	3.0 1.4	116.	38.0	2	351.	63.4	102.0	22.8	-1.	*CHN0157A -1.	*LAD0284B 14.	*LAD0284A 14.
66	5 ZMB0314A	-1.0	3	27.5 -	-13.1	2.4 1.5	39.	38.8	1	312.	63.7	33.0	-13.8	3.	*40Z0307A 5.	- ≠TCH0144A 11.	*BOT0297A 13.

Canal 4 — Channel 4 — Canal 4

	1	2	3	4	L	5	6	7	8	9 a	9 b	10)	11	12		13	14
3	5 ALGO252A	-25.0	4	1.6	25.5	3.6 2.2	152.	35.3	1	561.	62.8	9.5	30.0	4.	 *LBY0321A *	7.	+LBY)2808 12.	 *MLT0147A 15. *
36	5 AND0341A	-37.0	4	1.6	42.5	0.6 0.6	0.	48.7	2	19.	61.5	1.4	42.4	-1.	≭G 0027A *	3.	*ALG0252A 5.	*MLI0328A 7.
137	AR SOOOBA	17.0	4	41.1	23.8	3.5 1.7	134.	36.5	2	416.	62.7	56.0	21.0	-1.	*¥MS0267B	1.	*2GY00264 9.	*QAT02478 1).
											4 —						suite	cont.

- 4 --

)

٠

<u>،</u> ۲

Canal 4 suite — Channel 4 cont. — Canal 4 cont.

· ·

1

	1	2	3	4	5	6	7	8	9 a	9 b	10		11	12		13		14	
38	AUSDOOTA	ا 128 . 0	4	 145•7 -38•]	1.8 1.4	134.	49.2	2	205•	63.3	149.6	-37.5	1.	 ≠PNG0271A ≠	3.	 *AUS0009A *	7.	*0CE01014	27.
39	AUT0016A	-19.0	4	12.1 47.5	1.1 0.6	166.	45.7	2	.70∙	64.1	17.1	48.0	-0.	*ZAIJ322A	4.	*S 0138A	9.	*F 0093B	10.
40	BUL0020A	-1.0	4	25.0 43.0	1.0 0.6	165.	46.3	1	54.	63.6	28.1	42.0	-2.	≁ *MƊZO307A	3.	* *TUR01458	5.	≠ ≠GRC0105A	6.
41	CHN0156A	62.0	4	97.8 36.3	2.6 1.6	157.	38.2	1	338.	63.5	108.5	35.4	1.	≠ *CHND1ó1B	4.	* *CHN0157A	6.	* *CHN01558	11.
42	CHN0161B	92•0	4	118.1 31.1	2.5 1.7	117.	38.0	1	439.	64.4	122.6	37.4	-0.	* *KOR 0112B	3.	* *CHN016JA	7.	* *CHN0156A	7.
43	CKN0053A	158.0	4	-163.0 -11.2	1.8 0.7	30.	43.2	2	128.	64.3	-158.0	-9.0	4.	≠ *JCE0101A	4.	* *NZL00558	18.	* *TUN0215A	29.
44	CPV0301A	-31.0	4	-24.0 16.0	0.9 0.7	144.	46.5	2	37.	62.2	-24.3	14.4	4.	* *G 0027A	6.	* *AZRJ134A	14.	₩ #MLI0328A	16.
45	EGY0026A	-7.0	4	29.7 26.8	3 2.3 1.7	136.	38.2	2	307.	63.1	36.0	23.3	-0.	* *ARS0003A	0.	* *Algo252A	12.	* *M0Z0307A	16.3
46	G 0027A	-31.0	4	-3.5 53.1	3 1.8 0.7	142.	43.0	1	158.	65.0	1.3	51.1	3.	* *4LG0252A	5.	* *AUT0016A	10.	≠ ≠F 00938	13.
47	IND0040A	56.0	4	73.0 25.0	1.8 1.5	58.	39.9	2	234.	63.6	73.5	30.0	-1-	≠ ≠P 4K 0283A	0.	* *CHN0155B	10.	* *CHN0156A	16.
48	INDOO48A	68.0	4	86.2 25.	0 1.6 0.9	120.	42.8	2	122.	63.7	89.8	26.7	-3.	* *CHN0156A	1.	+ +CHN0155B	3.	*CHN0157A	6.
49	INS0028B	80.0	.4	101.5 0.	3.0 1.2	133.	38 •7	2	290.	63•3	104.4	0.9	Э.	* *MLAJ228B	4.	* *SNG0151A	4.	*THA01428	11.
50	KOR01128	110.0	4	127.5 36.	0 1.2 1.0	168.	43.2	2	109.	63.6	124.6	37.9	-2.	* *CHN0161B	1.	* *CHN0156A	6.	+ ≠J_ 0111C	7
51	LAC0284B	74.0	4	103.7 18.	1 2.2 0.8	133.	42.0	1	151.	63.8	102.0	22.5	3.	* *THA0142B	7.	* *CHN0156A	8.	+ +CHN0157A	12.
52	4AU0243A	29.0	4	56-8 -13-	9 1.6 1.4	65.	40.9	1	19 0.	63.7	53.0	-15.0	6.	≠ *4DG0236B	9.	* *MDZ0307A	11.	+ *COM0207A	18.
53	MLAJ228B	86.0	4	114.1 3.	9 2.3 1.1	45.	43.1	1	226.	63.6	109.0	2.4	3.	* *INS0028B	5.	* *VTN0325A	11.	* *CHN0161B	17.
54	ML10328A	-37.0	4	-7.6 13.	2 1.7 1.2	171.	40.9	1	191.	63.7	-12.0	15.0	3.	*GMBJ302A	8.	* *GUI01928	8.	* *ALG0252A	9.
55	MLT0147A	-13.0	4	14.3 35.	9 0.6 0.6	0.	48.7	1	17.	61.0	14.3	35.9	0.	* *ALG0252A	2.	*EGY0026A	11.	*CME0300B	12.
56	MDZ0307A	-1.0	4	34.0 -18.	0 3.6 1.4	55.	37.3	2	486.	64.2	30.3	-15.0	5.	* *ZMB0314A	6.	* *BUL0020A	13.	* *ZAI0322A	17.
57	90E0101A	-160.0	4	-145.0 -16.	3 4.3 3.5	4.	32.4	2	1301.	63.5	-154.7	-15.7	17.	* *CKN0053A	17.	* *PNG0271A	30.	* *TON0215A	33.
58	PAK0283A	38.0	4	74.7 33.	9 1.3 1.1	160.	42.4	1	152.	64.3	79.1	32.7	2.	* *1ND0040A	4.	* *CHN0156A	10.	* *IND0048A	13.
59	PNG0271A	128.0	4	4 148.3 -6.	7 2.8 2.0	155.	36.7	1	470.	63.4	141.)	-9.2	3.	≠ +AUS 3007A	5.	* *AUS0009A	10.	*INS0036A	16.
60	RRW0310A	11.0	4	+ 30.0 -2.	1 0.7 0.6	42.	48.3	2	45.	64.8	30.3	-1-0	0.	* *ZAI0322A	2.	* *UGA0051A	8.	≠ ≠¥MS02678	14.
61	S 0138A	5.0	4	4 16.2 61.	0 1.0 1.0	14.	44.2	2	195.	67.1	20.1	69.1	2.	* *URS0060A	3.	* *g 0027A	7.	* *POL 01328	3 24.
62	STP0241A	-13.0	4	4 7.0 J.	8 0.6 0.6	0.	48.7	2	19.	61.4	5.0	2.0	-2.	≠ *СМЕ0300В	2.	≠ ≠ZA!0322A	3.	∓ ¥GAB026JA	6.
63	TON0215A	170.0	4	4-174.7 -18.	0 1.4 0.7	85.	44.4	1	77.	63.3	-173.7	-15.9	4.	≠ *CKN0053A	6.	* *OCE01J1A	13.	* *sma03358	15.

• • •

1

-- 5 --

Canal 4	suite —	Channel 4	cont. —	Canal 4	cont.
---------	---------	-----------	---------	---------	-------

Γ		1	2	3	4		5	6	7	8	9 a	9 b	10		11	12		13		14
6	4 URS	0060A	23.0	4	41.5	۱ 57 - 4	3.1 1.6	153.	37.4	1	854.	66.7	28.3	68•9	-0. *	S 0138A	÷ 0. *G	002 7 A	9.	URS0061A 13.
6	5 ZAI)322A	-19.0	4	22.4	0.0	2.2 1.9	48.	38.2	1	451.	64.7	19.5	5.0	4. *	CME0300B	8. *A1	G0252A	9. *	*AUTOJ16A 12.

÷

Canal 5 --- Channel 5 --- Canal 5

	1	2	3	4	5	6	7	8	9 a	9 b	10		11	12		13	14
32	AFG02468	50.0	56	4.5 33.1	1.4 1.4	21.	41.2	1	142.	62.7	67.7	37.3	-2.	*URS0067B	1.	*CHN01558 5.	*TUR01458 11.
33	AUSJOD5B	98.0	5 13	3.5 -18.8	2.7 1.4	76.	38.5	2	383.	64.3	130.8	-12.5	2.	* *INS0035B	2.	* *AUS0006B 16.	* *CHN∂162B 18.
34	CAR03388	122.0	5 14	9•5 8•0	5.4 0.8	178.	38.1	1	278.	62.5	134.6	7.5	3.	* *1NS 3035B	6.	*J 3111C 7.	* *CHN0162B 2J.
35	5 CHN0155B	62.0	58	8.3 31.5	3.4 1.4	162.	37.3	2	360.	62.9	86.0	28.0	-0.	* *IND00448	3.	* *IND0048A 5.	*CHN0156A 12.
36	5 CHN0162B	92.0	5 11	5.9 21.0	2.7 2.4	23.	36.0	2	620.	64• J	125.0	25.9	-1.	≠ ≠J 0111C	0.	* *CHN0161C 7.	* *CHN01618 8.
37	CHN0164A	80.0	5 11	2.2 37.4	1.1 0.8	111.	45.2	1	79.	64.2	113.6	35.6	2.	*CHN01628	6.	*J 0111C 8.	* *CHN0155B 10.
38	6 CME0300B	-13.0	51	2.7 6.2	2.5 1.7	87.	37.9	1	357.	63.5	16.2	1.4	2.	*ZAI0322A	4.	*ZAI0323B 8.	* *TC001438 9.
39	9 F 0093B	-19.0	5	2.6 45.9	2.5 1.0	160.	40.4	1	223.	63.8	9.5	41.2	-2.	* *SMR03118	1.	+ *D))878 9.	* *ALG0251B 9.
40) FJI0193B	152.0	5 17	9.4 -17.9	1.0 1.0	67.	44.2	1	90.	63.7	-178.5	-19.9	4.	* *NZL00558	4.	* *TON0215A 23.	*J 0111C 28.
41	GUI0192B	-37.0	5 -1	1.0 10.2	1.6 1.0	147.	42.1	2	138.	63.5	-14.5	11.5	3.	*GNP0304B	5.	*MLI0328A 8.	*MLI03278 15.
42	2 IND00398	56.0	57	2.7 11.2	1.3 0.6	107.	45.5	1	59.	63.1	71.9	12.3	8.	*IND00458	12.	*IND0040A 16.	*IND00448 18.
43	INDO0448	68.0	57	9.5 22.3	2.2 1.4	146.	39.3	1	256.	63.4	78.2	26.9	э.	* *CHN01558	2.	*CHN0154B 12.	*IND0048A 14.
44	INS00358	104.0	5 12	4.3 -3.2	3.3 1.9	82.	36.1	1	514.	63.2	125.5	4.0	5.	*J 0111C	6.	*CAR0338B 15.	*CHN0162B 17.
45	5 J 0111C	110.0	5 13	4.5 31.5	3.5 3.3	68.	33.6	1	1152.	64.2	123.7	24•3	-0.	*CHN01₀2B	1.	*PNG01318 12.	*INS00358 13.
46	6 LBY02838	-25.0	52	1.4 26.0	2.5 1.0	119.	40.1	2	221.	63.5	22.0	33.0	4.	*TUR 01 458	8.	*ALG02518 11.	*ALG02524 12.
47	MDG0236B	29.0	54	6.6 -18.8	2.7 1.1	65.	39.3	2	253.	63.4	49.)	-12.3	8.	*MAU0243A	10.	*4AJ0242B 15.	*ZAI0323B 22.
48	NZL00558	158.0	5 17	2.3 -39.7	2•9 1`•6	47.	37.7	1	369.	63•4	166.3	-45.5	15.	*AUS00058	21.	*FJ10193B 21.	*CKN0053A 22.
49	PLM03378	170.0	5-16	1.4 7.0	0.6 0.6	0.	48.7	1	24.	ó2•4	-162.4	6.1	5.	*CAR0338B	7.	*TON0215A 14.	*SMA03358 16.
50	POL01328	-1.0	51	9.3 51.8	1.5 0.6	162.	44.5	2	92.	64.2	14.4	53.9	-1.	≠S 0138A *	3.	*F 00938 6.	*FNL0103B 7.
51	QATJ2478	17.0	55	1.1 25.3	0.6 0.6	0.	49.7	1	21.	61.8	52.3	24.8	1.	*YMSJ2673	5.	*ARS02758 6.	*4850003A 9.
										6 —						suite	— cont.
		•)						,					

· · · · ·

Canal 5 suite --- Channel 5 cont. --- Canal 5 cont.

	1	2	3	4		5	6	7	8	9 a	9 b	10		11	12		13		14	
52	SMA0335B	170.0	5-	-170.1 -1	14.2	J.6 J.6	Э.	48.7	2	18.	61.2	-171.0	-11.0	-3.	*TUN0215A	1.	*PLM03378	2.	*CKN0053A	5.
53	SMR0311B	-37.0	5	12.6 4	43.7	J.6 D.6	0.	48.7	1	24.	62.5	12.3	43 .)	-).	* *F 3793B	2.	* *GUI 0192B	9.	* *TURU1458	12.
54	5WZ0313B	-1.0	5	31.5 -2	26.5	0.6 0.6	66.	48.5	1	27.	62•8	31.1	-25.9	4.	* *M0203)7 A	7.	* *8ct02978	9 . :	* *PCL01328	11.
55	THA01428	74.0	5	100.7	13.2	2.8 1.5	106.	37.9	2	379.	63.7	103.7	18.6	-1.	* *CHN0162B	4.	* *lA002840	5.	* *LAG0284B	5.1
56	TUR01458	5.0	5	34.4 3	38.9	2.7 1.0	168.	39.8	1	249•	63•8	42.7	41.5	-2.	* *URS0064B	-1.	* *LBY0280B	11.	* ≭URS0067₿	12.
57	URSJO64B	23.0	5	45.6 4	40•8	2.2 0.6	163.	43.1	2	121.	63.9	40.0	43.,4	-1.	* *TUR0145B	2.	* *URS0060A	6.	* *URS0067B	12.
58	URS0067B	44.0	5	62.4	58•5	3.2 1.5	169.	37.4	1	888.	66.9	50.9	51.9	12.	* *URS0060A	17.	* *TUR 01458	17.	* *AFG0246B	21.
59	WAKJ334 B	140.0	5	166.5	19.2	3.6 9.6	J.	48.7	1	31.	63.6	166.5	19.2	17.	* *J 0111C	22.	* *MRL0333B	24.	* *NCL0100B	25.
60	¥#\$0267B	11.0	5	48.8	15.2	1.8 1.5	176.	39.9	2	197.	62•9	49.5	18.4	3.	* *ARS02 75B	6.	* *ARS0003A	7.	* *YENG2668	13.
Can	al 6	Channe	el 6	— C	anai (6														
	1	2	3	. 4		5	6	7	8	9 a	9 b	10)	11	12		13		14	
30	ALGO2518	-25.0	6	4.2	33.2	2.4 1.3	172.	39.4	1	255.	63.4	9.5	30.0	٥.	*TCD01438	5.	*LBY0321B	7.	*TG00226B	8.
31	AR\$32758	17.0	6	48.3	24.6	3.8 1.4	138.	36.9	2	385.	62.8	42.5	17.0	-11.	* *YEM02668-	-11.	* *YMSJ267B	-1.	* *TCD0143B	12.
32	AUSJOO6B	98.0	6	135.4 -	30.3	2.0 1.4	44.	39.8	1	224.	63.3	140.8	-27.7	1.	* *AUS0008B	3.	* *AUS0005B	8.	* *AUS0004B	14.
33	AUSODOBB	128.0	6	145.9 -	21.5	· 2.9 2.0	120.	36.6	2	516.	63 .7	149.)	-29.0	3.	* *AUS0009B	6.	* *AUS0006B	6.	* *AUS0005B	27.
34	BOT02978	-1.0	6	23.3 -	22.2	2.1 1.5	36.	39•2	2	298.	63.8	25.3	-17.8	4.	* *ZMB0314B	6.	* *ZAI0323B	10-	* *R0U0136B	11.
35	CHNJ1548	62.0	6	83.9	40.5	2.8 2.0	177.	36.7	1	449.	63.3	79.0	34.3	-0.	* *PAK01278	1.	* *CHN0155B	8.	* *CHN01578	14.
36	CHN01610	92.0	6	118.1	31.1	2.5 1.7	117.	38.J	1	442.	64.5	122.6	37.4	1.	* *KOR0112C	3.	* *CHN0160B	7.	* *CHN01628	13.
37	CKH00528	158.0	6	-161.0 -	19.8	1.0 0.6	132.	46.1	2	71.	64.6	-163.5	-17.5	15.	* *SMD0057B	18.	* *NZL0055B	19.	* ≠₩AL01028	28.
38	CLN02198	50.0	6	80.6	7.7	1.2 0.6	106.	45.7	1	ó2.	63.6	80.0	10.0	3.	* ≄∴ND0043B	5.	* *IND0045B	10.	* *AFG0246B	14.
39	D 00878	-19.0	6	9.6	49.9	1.6 0.7	147.	43•6	2	158.	65.6	10.0	54.8	1.	* *ZA10323B	5.	* *FNL0103B	6.	* *LUX01143	6 11.
4)	FNLJ1038	5.0	6	22.5	64.5	1.4 0.8	171.	44.0	2	237.	67.8	19.2	60.0	6.	≭ ≭D 0087B	8.	* *POL01326	13.	* *TUR01458	5 20.
41	GNP33348	-31.0	6	-15.0	12.0	0.9 0.6	172.	46.9	2	42.	63.2	-14.2	12.3	2.	*GMB0302B	6.	*GUI01928	7.	*IRL02119	3 14.
 • 2	GUM03318	122.0	6	144.5	13.1	0.6 0.6	0.	48•7	2	29.	63•4	144.7	13.4	7.	* *MRA0332B	9.	* *CARJ3388	17.	* *AUS0008B	19.
43	INDO0378	68.0	6	93.0	25.5	1.5 1.1	40.	42.1	2	155.	64.0	96.0	29.5	-1.	* ≭CHN0157B	4.	* *CHN01555	6.	* *CHN01548	5 7.
: 144	E 19000458	56.0	6	76.2	19.5	1.6 1.6	21.	40.3	2	213.	63.6	74.4	22.0	3.	* *PAK0127B	5.	* *CHN01548	14.	* ≠ars02759	15.

— 7 ----

1

.

suite — cont.

Canal 6 suite — Channel 6 cont. — Canal 6 cont.

	1	2	3	4		5	6	7	8	9 a	9b	10		11	12		13		14	
45	INSUD28C	80.0	6	101.5	0.0	3.0 1.2	133.	38.7	2	292.	63.3	104.4	0.9	. 0.	≠HLA0228C	4.	*SNG01518	4.	*THA0142B	11.
46	IRL)211B	-31.0	6	-8.2	53•2	0.8 0.6	162.	47.2	1	51.	64.3	-5.5	54.3	7.	*) 00878 :	10.	*GNP0304B	15.	*F 00938	16.
47	KOR01120	113.0	6	127.5	30.0	1.2 1.0	168.	43.2	2	110.	63.6	126.2	33.0	-3.	≁ *CHNJ161C	1.	* *PNG0131B	2.	* *J 0111D	5.
48	LAJJ284C	74.0	6	103.7	18.1	2.2 0.8	133.	42.0	1	152.	63.8	102.0	22.5	3.	≁ *THAJ1428	7.	*IND0037B	9.	* *CHN0157B	12.
49	MAU0242B	29.0	6	59•8	-18.9	1.6 1.2	55.	41.2	1	190.	64.0	60.0	-14.0	5.	+ *ZAI0323B	6.	*MDG0236B	14.	≁ *PAK01278 +	21.
50	MLA0228C	86.0	6	114.1	3.9	2.3 1.1	45.	40.1	1	227.	63.6	109.0	2.4	3.	≁ *≦NS0028C	5.	*VTN0325B	11.	* *CHN0161C	17.
51	MLIJ327B	-37.0	6	-2.0	19.0	2.7 1.3	127.	39.0	1	3 35.	6 4•2	-2.5	23.3	9.	*4LG0251B	11.	*SMR0311B	18.	+ *LIE0253B	20.
52	MRL0333B	146.0	6	166.7	7.9	1.5 1.5	177.	43.7	1	183.	63•3	162.1	11.5	13.	* *PNG01318	19.	*CAR0338B	19.	+ *NCL0100B	22.
53	NCL0100B	140.0	6	166.0	-21.0	1.1 0.7	146.	45.1	1	73.	63.8	167.5	-23.J	-3.	- ≉WAL01028 -	-3.	*NHB0128B	11.	+ *AUS0008B	15.
54	PAK01278	38.0	ć	6 9. 6	29.5	2•3 2•2	14.	37.3	1	468.	64.0	74.5	35.7	2.	≁ ≠CHN0154B	2.	*AFG02458	15.	+ *AFG0246B	21.
55	PNG01318	110.0	6	147.7	-6.3	2.5 2.2	169.	36.9	1	570.	64•4	141.0	-9-2	2.	+ *AUS00088 *	6.	* *INS0036B	7.	* *KOR0112C	10.
56	R0J01368	-1.0	6	25.0	45 •7	1.4 0.7	155.	44.7	1	83.	63.9	20.2	46.1	-1.	≁ *D`00878 ≈	4.	*BCT02978	5.	* *TCH0144B	7.
57	TCD01438	-13.0	6	18.1	15.5	3.4 1.7	107.	36.6	2	554.	64.0	16.5	8.2	5.	*CME0300B	7.	*MLI0327B	14.	*GAB0260B	15.
58	TG30226B	-25.0	6	0.8	8.6	1.5 0.6	105.	44.6	2	76.	63.4	-0.2	11.1	-2.	*MLI03278 ·	-0.	*ALG0251B	6.	*DAH0233B	7.
59	WALD1028	140.0	6	-176.8	-14.0	0.7 0.6	29.	4 7. 8	1	46.	64.4	-178.1	-14.2	-2.	*MCL01008 ·	-2.	*PNG01318	13.	- ≭FJI31938 *	18.
60	YEM02668	11.0	6	44.3	15.1	1.1 0.7	109.	45.2	1	56.	62.7	44.0	18.0	-1.	*4R\$02 75 B	э.	*YMSJ267B	7.		11.
61	ZA10323B	-19.0	6	21.3	-6.8	2.8 1.5	149.	38.0	1	471.	64.7	16.3	-1.0	1.	*TCD01438	5.	*GAB0260B	7.	*D 00 87 B	8.

Canal 7 — Channel 7 — Canal 7

	1	2	3	4	5	6	7	8	9 a	9 b	10	11	12	13	14
33	AFG02458	50.0	1	7 70.2 35.	5 1.3 1.1	53.	42.5	1	109.	62.9	68.0 31.6	0.	≠IRN0109B 2.	*PAK0127B 11.	*PAK02835 13.
34	4US00048	98.0	7	7 121.8 -24.	9 3.6 1.9	54.	35.9	2	520.	63.1	126.6 -31.0	8.	*4US0))6B 8. *	+ +AUSJJJ9B 21. +	*INSCJ36B 22. *
35	Bedoosla 5	128.0	7	7 147.2 -32.	0 2.1 1.4	15.	39.0	1	285.	64.1	141.0 -34.0	4.	*AUS00078 6. *	*AUS0008B 13. *	*PNG0271B 13. *
106	6 AZR01348	-31.0	7	7 -23.4 36.	1 2.6 0.7	158.	41.7	2	136.	63.1	-16.9 32.5	-5.	*POR01338 -4.	*L8802448 9. *	*CPV0301B 14. *
51	7 CHN01578	62 • 3	7	7 192.3 27.	8 2.6 1.6	127.	38.2	2	49 3 .	65.1	1,1.7 21.3	1.	*VTN03259 2. *	*LAU02940 14. *	*LA00284C 14. *
58	CHN0160B	92.0	7	7 122.8 45.	3 2.5 1.4	150.	33.7	2	444.	65.1	122.5 38.3	0.	*J 01119 4. *	*CHNJ157B 7. *	*CHNJ161C 7.
39	9 CCH0207B	29.0	7	7 44.1 -12.	1 0.8 0.6	149.	47•7	2	35.	63.1	43.1 -11.3	5∙	*SOM0312B 9.	*ZMB03148 10.	*MAU0243B 16.
										8				suite	cont.
				· · ·		*								•	, `

Canal 7 suite -- Channel 7 cont. -- Canal 7 cont.

(

	1	2	3	4		5	6	7	8	9 a	9 b	10		11	12		13		14	
4(DAH0233	8 -19.0	7	2.2	9.5	1.4 0.7	97.	44.3	2	73.	53.3	2.2	6.2	3.	*TG002268	5.	*ZA103238 1	4.	*ZAI03228	14.
41	GAB0260	-13.)	7	11.8	-0.6	1.4 1.1	64.	42.2	1	131.	63.4	11.0	-4.0	3.	≠ \$ZAI03233 *	5.	≁ ≭ZAI0322B *	9.	* *TCD01438 *	14.
42	2 GMPJ302	8 -37.0	7	-15.1	13.4	0.8 0.6	4.	47.5	2	39.	63.4	-14.2	13.3	3.	*GNP 03048	7.	*ML103288	9•	*LBK02448	12.
4	GRC0105	5.0	7	24.7	38.2	1.8 1.0	156.	41.8	1	142.	63.4	26.6	41.5	1.	- *BUL 00208	4.	≁ *ROUJ1368 ≭	6.	* *IRN01098 *	13.
4	4 IND0043	8 56. 0	7	77.8	11.1	1.4 1.3	172.	41.8	1	143.	63.4	79.5	9.1	4.	*CLN0219B	5.	*IND00458 1 *	5.	*IND00408	19.
4	5 IND0047	B 68.0	7 🧠	93.3	11.1	1.9 0.6	96.	43.6	1	97.	63.5	93.8	14.8	6.	*CHN0157B	8.	*IND0037B 1	6.	*IND00488	16.
4	6 INSDOG	8 104.0) 7	135.2	-3.8	2.5 2.0	147.	37.3	1	445.	63.8	128.0	-7.0	4.	*4US0004B	5.	+ ≠J 0111D 1	5.	*PNG01318	16.
4	7 IRNO109	B 34.0) ¹	54.2	32.4	3.8 1.8	149.	35.8	2	504.	62.8	61.5	31.0	6.	* *AFG02458 *	8.	* *PAKJ127B 1 *	5.	* *SOM0312B	16.
4	5 J 0111	D 110.0	7	134•5	31.5	3.5 3.3	68.	33 . ó	1	1160.	64•2	129.3	34.7	1.	*CHN01608 *	7.	*KUR0112D *	8.	*KOR0112C *	8-
+	9 LBN0279	B 11.0) 7	35.8	33.9	0.6 0.6	0.	43.7	2	20.	61.7	36.8	34.5	-9•	*GRC01058	6.	*IRN01098 *	6.	*ARS0003B	7.
5) LBFJ244	B -31.0	7	-9+3	6•6	1.2 0.7	133.	44.9	1	69.	63.3	-10.2	8.5	4.	*MLI03288	9.	*AZR0134B	9.	*POR01338	12.
5	1 LBY0321	e -25.0) 7	13.1	27.2	2.4 1.1	129.	40.0	2	202.	63.1	11.4	33.5	3.	*ALG0251B	6.	*ALG0252B	9.	*TG002268	12.
5	2 LIE0253	B -37.0	7 (9.5	47.1	J.6 J.6	э.	48.7	1	24.	62.5	9.5	47•2	1.	*TCH0144B	4.	*LUX0114B *	8.	*AND03418 *	13.
5	3 LUX0114	8 -19.0	7 (6.9	49.8	0.6 0.6	0.	48.7	1	27.	63.0	7.0	48.5	-4.	*LIE02538	1.	*D 00878	з.	*TCH01443	4.
5	4 MRA0332	в 122.0	7 (145.9	16.9	1.2 0.6	76.	45•7	1	61.	63.5	145.0	20.0	-4 •	*J 0111D	-4•	*GUM03318 1	2.	*4US00098	19.
5	5 NHB0128	B 140.(7 (168.0	-1ó.4	1.5 0.7	87.	44.1	2	76.	62•9	169•8	-20.0	7.	*NCL0100B	8.	*NRU03098 2	23.	*INS00368	23.
5	6 NRU0309	B 134.(7 כ	167.0	-0.5	0.6 0.6	0.	48.7	2	25.	62.6	166.0	-2.0	10.	*NHB01283	14.	*INS0036B 1 *	7.	*PNG01318 *	19.
5	7 POR0133	8 -31.0	5 7	-8.0	39.6	3.9 3.6	112.	46.3	2	46.	63.4	-7.5	37.2	-15.	*AZR01348-	-15.	*LBR0244B 1	1.	*G 00278 *	14.
5	8 SM00057	B 158.	ד כ	7-172.3	-13.7	0.6 0.6	0.	48.7	1	32.	63.7	-171.0	-14.1	7.	*CKN00538	9.	≠СКНЭ052В 1	4.	*WAL0102B	20.
5	9 SNG0151	B 74.	0 7	103.8	1.3	0.6 0.6	0.	48.7	2	31.	63.6	106.0	1.1	-1.	*VTN03258	4.	*INS0028D	4.	*INSDO2BC	4.
6	J SOM0312	23.	0 7	7 45.0	5•4	3.3 1.5	71.	37.2	1	327.	62.4	43.2	11.2	5.	≠UR S0060B	8.	*URS0061B 1	э.	*YEM32668	15.
6	1 TCH0144	8 -1.	0 7	17.3	49.3	1.5 0.6	170.	44.9	2	81.	63.9	12.1	50•3	-2.	*LIE02538	3.	≠ZMB0314B *	5.	*LUX01148	6.
6	2 UGAJ05	.8 11.	0 7	7 32.3	1.2	1.5 1.1	60.	42.1	1	130.	63.3	29•8	-1.3	1.	+RRW0310B	4-	*Ζ 430314Β *	8.	*ZA10322B	16.
ċ	3 UR \$006	.B 23.	0 7	7 24•7	56 .6	0.9 0.6	12.	46.7	2	68•	65.1	25.8	54.1	-0.	≠TCHJ144B	6.	*SCM9312B	6.	¥URSJ969B ★	ó.
ó	4 URS007	A 44.	о [.]	7 70.1	61.5	2.4 0.7	173.	42.3	1	300.	67 . 1	63.1	66.2	13.		18.	*IRNJ139B 1	18.	*AFG02458	21.
0	5 VTND32	ib 86.	0 .	7 105.3	16.1	3.0 1.4	116.	38.0	2	356.	63.5	102.0	22.8	-1.	*CHN01578	-1.	*LA30284D]	14.	≭LA ©0284C ≯	14.
່ວ	6 Z +0031	-1.	0	7 27.5	-13.1	2.4 1.5	39.	38.8	1	316.	63.8	33.0	-13.8	3.	*MOZ0307B	5.	*TCH0144B	11.	*80T02978	; 13.

١

- 9 -

Canal 8 --- Channel 8 --- Canal 8

. -

. .

	1	2	3	4	1	5	6	7	8	9 a	9 b	10		11	12		13		14	
44	CPV0301B	-31.0	8	-24.0	16.0	0.9 0.7	144•	46.5	2	38.	62•2	-24.3	14•4	4.	 ≭G 00276	6.	 ≠4ZR01348	14.	 *ML10328B	16.
45	EGY00268	-7.0	8	29•7	26•8	2.3 1.7	136.	38.2	2	311.	63•2	36.0	23.3	-0.	* *ARS0003B	0.	* *ALG02525	12.	* *M0Z03078	16.
40	G 00278	-31.0	8	-3.5	53.8	1.8 0.7	142.	43.0	1	160.	65.1	1.3	51.1	3.	* *ALG02528	5.	* *AUT0016B	13.	* *F 0093C	14.
47	IND0040B	56.0	8	73.0	25.0	1.8 1.5	58.	39.9	2	237.	63.7	73.5	30.0	-1.	* ≉PAK0283B	0.	 ¥chn0155C	10.	* *CHN0156в	16.
48	IND0048B	68.0	8	86.2	25.0	1.6 3.9	120.	42.8	2	124.	63.7	89.8	26.7	-3.	* *CHN01568	1.	* *CHN0155C	3.	* *CHN0157B	6.
49	INS0028D	80.0	8	101.5	0.0	3.0 1.2	133.	38.7	2	294.	63.4	104.4	0•9	0.	≠ ≠MLA0228D	4.	* *SNG01518	4.	* *THA0142C	11.
50	KOR01120	110.0	8	127.5	36.0	1.2 1.0	168.	43.2	2	110.	63.7	126.2	33.0	1.	≠ ≠J 0111E	5.	* *J 01110	6.	₹ ¥CHN0156B	10.
51	LA00284D	7 4.0	8	103.7	18.1	2.2 0.8	133.	42.0	1	153.	63.8	102.0	22•5	3.	* *THA0142C	7.	* *CHN0156B	9.	* *CHN01578	12.
52	MAUJ243B	29.0	8	56.8	-13.9	1.6 1.4	65.	40.9	1	192.	63.8	53.0	-15.0	6.	* *MDG0236C	9.	*40203078	11.	* *COM0207B	18.
53	MLA0228D	86.0	8	114-1	3.9	2.3 1.1	45.	40.1	1	229.	63.7	109.0	2•4	3.	+ +INS0028D	5.	* *VTN03258	11.	* *CHN0173A	19.
54	ML103289	-37.0	8	-7.6	13.2	1.7 1.2	171.	40.9	1	193.	63.8	-12.0	15.0	3.	* *GMB03028	8.	*GUI 0192C	8.	* *ALG02528	۶.
55	MLT01478	-13.0	8	14.3	35.9	0.6 0.6	' 0 .	48.7	1	17.	61.0	14.3	35.9	0.	*ALG32528	2.	*EGY00268	11.	*CME0300C	12.
56	MUZ03078	-1.0	8	34.0	-18.0	3.6 1.4	55.	3 7. 3	2	493.	64•2	30.3	-15.0	5∙	- ≉∠MB0314B	Ó.	+ *3UL0023B	13.	*ZAI0322B	17.
57	GCE01018	-160.0	8	-145.0	-16.3	4.3 3.5	4.	32.4	2	1318.	63.6	-154.7	-15.7	17.	*CKN0053B	17.	*PNGJ2718	30.	+ *TCN02158	33.
58	PAKJ283B	38.9	8	74.7	33.9	1.3 1.1	169.	42.4	1	154.	64.3	79.1	32.7	2.	*IND0040B	4.	+ +CHN01568 *	10.	*IND0048B	13.
59	PNG02718	128.0	8	148.0	-6.7	2.8 2.0	155.	36.7	1	476.	63.4	141.0	-9.2	3.	* \US 30078	5.	*	10.	*INS 2036B	16.
60	RRW03108	11.0	8	30.0	-2.1	0.7 0.6	42.	48.3	2	46.	64.9	30.3	-1.0	0.	*ZAI03228	2.	≭UGA 30518 *	ర.	*YMS02670	14.
61	S 0138B	5.0	8	16.2	61.0	1.0 1.0	14.	44.2	2	198.	67.1	20.1	69.1	2.	*UP.S00608	з.	≭ G 00278 *	7.	*POL0132C	25.
52	STP0241B	-13.0	8	7.0	0.8	0.6 0.6	0.	48.7	2	19.	61.5	5.0	2.0	-2.	*CME0300C *	2.	*ZAI 93228 *	3.	*GAB0260B	6.
63	TON3215B	170.0	8	-174.7	-18.9	1.4 0.7	85.	44.4	1	78.	63.3	-173.7	-15.9	. 4.	*CKN00538 *	6.	*0CE01018 *	13.	*SMA 0335C	15.
64	URS0060B	23.0	8	41.5	57.4	3.1 1.6	153.	37.4	1	865.	66.8	28.3	68.9	-0.	*S 01388 *	1.	*G 00278 *	9.	*URS0061B *	13.
65	ZA103228	-19.0	8	22.4). 0	2.2 1.9	48.	38.2	1	456.	64.3	19.5	5.0	4.	*CME0300C	8.	*ALG02528	9.	*AUT0016B	12.
135	ALG02528	-25.0	8	3 1.6	25.5	3.6 2.2	152.	35.3	1	569•	62.8	9.5	30.0	4.	*L8¥0321B *	7.	*LBY02800 *	12.	*MLT01473	15.
36	AND0341B	-37.0	8	3 1.6	42.5	0.6 0.6	0.	48.7	2	19.	61.5	1.4	42•4	-1.	≭G 0027B ≄	3.	*ALG02528 *	5.	*MLI03288 *	7.
37	ARSJJO3B	17.0	8	8 41.1	23.8	3.5 1.7	134.	36.5	2	421.	62.8	56.0	21.)	-1.	*Y≯\$0267C *	1.	*EGY00268 *	9.	*QAT0247C *	10.
36	AUS00078	128.)	8	3 145.0	-38.1	1.8 1.4	134.	4).2	2	237.	63.4	149.6	-37.5	1.	*"NG02718	3.	*AUS0009B	7.	*3CE01018	27.

- 10 -

.

•-

suite — cont.

۰, ۳

£

Car	nal 8	suite)	Chan	nel 8	cont.	Canal	8 a	ont.				•								
	-		2	3	4		5	6	7	8	9 a	9 b	10		11	12		13		14	
39	AUTO	016B	-19.0	8	12.1	47.5	1.1 0.6	166.	45.7	່2່	71.	64.2	17.1	48.0	-0.	*ZAI 03228	4.	*S 0138B	9.	+F 0093C	10.
40	BULO	0208	-1.0	8	25.0	43.0	1.0 0.6	165.	46.3	1	55.	63•7	28.1	42.0	-2.	* *M0Z0307B	3.	+ ≠TUR0145C	5.	* *GRC0105B	6.
41	CHNO	156 B	62.0	8	97.8	36.3	2.6 1.6	157.	38.2	1	342.	63.5	90.2	38.6	3.	* *PAK0283B	7.	* *CHN0155C	9.	* *CHN01578	12.
42	CHNO	173A	92.0	8	115.7	27.4	1.1 3.9	99.	43.9	1	100.	64.0	113.9	29.0	4.	* *CHN0162C	8.	* *Chn0156B	9.	* *CHN0160B	14.
43	CKNO	0538	158.0	8-	-163.0	-11.2	1.8 9.7	30.	43.2	2	130.	64.3	-158.0	-9.0	4.	* *DCEJ1018	4.	* *NZL9055C	18.	* *TON02158	29.
						,															
				•								•									
1																					
Car	nal 9	—	Channe	BI 9	.—	Canal	9				<u>.</u>								<u>.</u>		
		1	2	3	4	l I	5	6	7	8	9 a	9 b	10		11	12		13		14	
32	2 AFG	9246C	50.0) 9	64.5	33.1	1.4 1.4	21.	41.2	1	144.	ا 62•8	67.7	37•3	-2.	*URS0067C	1.	+CHN0155C	5.	+TUR0145C	11.
3	B AUS	0005C	98.0	9	133.5	-18.8	2.7 1.4	76.	38.5	2	389.	64.4	130.8	-12.5	2.	*INS0035C	2.	*AU50006C	16.	*CHN 0162C	18.
34	4 CAR	0338C	122.0) 9	149.5	8.0	5.4 0.8	178.	38.1	1	282•	62.6	134.6	7.5	3.	* *INS0035C	6.	* *J 0111E	7.	*CHN0162C	20.
3	5 CHN	01550	62.0) 9	88.3	31.5	3.4 1.4	162.	37.3	2	365.	63.0	86.0	28.0	-0.	* *IND0044C	3.	* *IND0048B	5.	* *CHN0156B	12.
3	6 CHN	01620	92.0) 9	115.9	21.0	2.7 2.4	23.	36.0	2	628.	64.0	125.0	25.9	-0.	≠ *J 0111E	9.	*CHN0171A	15.	* *CHN0173A	15.
3	7 CHN	0165A	80.0	9	111.4	41.8	1.6 1.2	15.	41.5	1	163.	63.6	107.6	37.8	5.	* *CHN0155C	8.	* +*CHN0162C	12.		14.
3	в сме	03000	-13.0	9	12.7	6.2	2.5 1.7	87.	37.9	1	361.	63.5	16.2	1.4	2.	* *ZAI0322B	4.	≠ ≠ZA10323C	8.	* *TCD0143C	9.
3	9 F	00930	-19.0) 9	2.6	45.9	2.5 1.0	160.	40.4	1	226.	63.9	9.5	41.2	-2.	* *SMR0311C	1.	.≠D 0087C	9.	* *ALG0251C	9.
4	0 FJI	01930	152.0)	179.4	-17.9	1.0 1.0	67.	44•2	1	91.	63.8	-178.5	-19.9	4.	+ *NZL0055C	4.	* *TCN02158	23.	*J 0111E	28.
4	1 GUI	3192 C	-37.) .9	-11.0	10.2	2 1.6 1.0	147.	42+1	2	140.	63.5	-14.5	11.5	3.	* *GNP0304C	5.	* *MLI03288	8.	* *MLI0327C	15.
4	2 IND	0039C	56.() 9	72.7	11.2	2 1.3 0.6	107.	45.5	1	59.	63.2	71.9	12.3	8.	.*IND0045C	12.	*IND0040B	16.	*I ND0044C	18.
4	3 IND	0044C	68.0) 9	79.5	5 22.3	3 2.2 1.4	146.	39.3	1	259.	63.5	78.2	26.9	0.	* *CHN0155C	2.	* *CHN0154C	13.	* *IND00488	14.
4	4 INS	00350	194.0) 9	124.3	3 -3.2	2 3.3 1.9	82.	36.1	. 1	521.	63.3	125.5	4.0	5.	* *J 0111E	6.	* *CAR03380	15.	* *CHN0162C	17.
4	5 J	0111E	. 110.	0 9	9 134.5	5 31.5	5 3.5 3.3	68.	33.6	. 1	1168.	64.3	123.7	24.3	-0.	* *CHN0162C	1.	* *PNG01310	12.	* *INS00350	13.
4	6 LBY	02800	-25.0) 9	21.4	+ 26.0	2.5 1.0	119.	40.1	. 2	224.	63.6	22.0	33.0	4.	* *TUR0145C	8.	* *ALG02510	11.	* *ALG02528	12.
4	7 MDG	02360	29.0) 9	9 46.6	5 -18.8	3 2.7 1.1	65.	39.3	2	257.	63.4	49.0	-12.3	8.	* *MAU0243B	10.	* *MAU02420	: 15.	* *ZAI03230	22.
4	8 NZL	0055C	158.0	0 9	9 172.3	3 -39.7	7 2.9 1.6	47.	37.7	1	374.	63.4	166.3	-45.5	i 15.	* *AUS0005C	21.	* *FJI01930	21.	* • *CKN00538	3 22.

×.

suite — cont.

'ـ

•

١

Canal 9 suite --- Channel 9 cont. --- Canal 9 cont.

		1	2	3	4		5	6	7	8	9 a	9 b	10		11	12		13		14	
			170 0		-141 4	7 0	0 6 0 6		40 7		24	(2.5	142.4		, Е	*****		0.10.21.50	• •	+0.00005.	1
4	9		170.0	9	-101.4	1.0	0.0 0.0	0.	48• f	I	24.	0200	-102.4	0.1	2•	÷6AKU338L *	*	0.002128	14.	*5M103350 *	16.
5	0	PJL0132C	-1.0	9	19.3	51.8	1.5).6	162.	44•5	2	93.	64.2	14.4	53 •9	-1.	≠S 01388	3. *⊦	= 0093C	6.	*FNL0103C	7.
5	1 (DAT0247C	17.0	9	51.1	25.3	0.6 0.6	0.	48.7	1	21.	61.9	52.3	24.8	1.	*YMS0267C *	5. *A *	NR SO275C	6.	*ARS00033 *	9.
5	2	5MA0335C	179.0	9	-179.1	-14.2	0.6 0.6	э.	48.7	2	18.	61.3	-171.0	-11.0	-3.	¥T JN0215B *	l. *P ≭	PLM0337C	2.	*CKN00538 *	5.
5	3 :	SMR0311C	-37.0	9	12.6	43.7	0.6 0.6	0.	48.7	1	24.	62.5	12.0	43•)	-)•	*F 0093C *	2•*G *	SUI0192C	9.	*TUR0145C *	12.
5	4	5WZ0313C	-1.0	9	31.5	-26.5	0.6 0.6	56.	48.5	1	27.	62.9	31.1	-25.9	4.	*40Z0307B *	ĭ• ≑8 *	30T 3297C	9.	*POL0132C *	11-
5	5	FHA0142C	74.0	9	100.7	13.2	2.8 1.5	106.	37.9	2	384.	63.7	103.7	18.6	-1.	*CHN9162C *	•• *L *	A00234E	5.	*LAG02840 *	5.
5	6	TUR0145C	5.0	9	34.4	38.9	2.7 1.0	168.	39•8	1	252.	63 •8	42.7	41.5	-2.	*URS0064C		BY0283C	11.	*URS0067C *	12.
5	7	JRS3064C	23.0	9	45.6	40.8	2.2 0.6	163.	43.1	2	122.	64.0	40.0	43.4	-1.	*TUR0145C *	2• *i) *	IRS0060B	6.	*URS0067C	12.
5	8	JR 50067C	44.0	9	62.4	58.5	3.2 1.5	169.	37.4	1	900.	66 <u>.</u> 9	50.9	51.9	12.	*URS0060B 1 *	7• ≠T *	UR0145C	17.	*AFG0246C *	21.
5	9	MAK0334C	143.0	9	166.5	19.2	0.6 0.6	0.	48.7	1	32.	63.7	166.5	19.2	17.	*J 0111E 2 *	2. *M *	IRL0333C -	24.	*NCL0100C	25.
6	0	rms0267C	11.0	9	48.8	15,2	1.8 1.5	176.	39.9	2	200.	62.9	49.5	18.4	3.	+ ARS 02 75C	5. *A	AR 50003B	7.	*¥5H0266C	13.
					•			C ·													

Canal 10 — Channel 10 — Canal 10

•

· · ·

Γ	Τ	1	2	3	4	5	6	7	8	9 a	9 b	10	11	12	13	14
		AL CO251C		10	4 2 32	2 4 1 2	172	20.4	Ţ	25.0	(2 5	0 5 70		+10001430 5	1.08033316 7	+******
	U	42602910	-25.0	10	4.2 22.	2 2.4 1.3	172.	37.4	T	228.	0303	9.5 30.	0 0.	#≹€D01436 5• *	*LUY0321L /• *	*1600226C 8• *
3	1	ARS0275C	17.0	10	48.3 24.	5 3.8 1.4	138.	36.9	2	390.	62.9	42.5 17.	0 -11.	*YEM0266C-11. *	≭YMS0267C −1. *	*TCD0143C 12. *
3	2	AUS0006C	98.0	10	135.4 -30.	3 2.0 1.4	44.	39.8	1	227.	63.3	140.8 -27.	7 1.	*AUS0008C 3.	*AUS0005C 8. *	*AUS0004C 14. *
3	3	AUSOOOBC	128.0	10	145.9 -21.	5 2.9 2.3	120.	36.6	2	523.	63.8	149.0 -29.	შ ა	*AUS0079C 6. *	*AUS0006C 6. *	*AUS0005C 27.
3	4	B0T029 7 C	-1.0	10	23.3 -22.	2 2.1 1.5	36.	39.2	2	292.	63.9	25.3 -17.	8 4.	*ZMB0314C 6. *	*ZAI0323C 10.	*R0001360 11. *
3	5	CHN0154C	62.0	10	83.9 40.	5 2.8 2.9	177.	36.7	1	455.	63.3	79.0 34.	3 -0.	*¤AK01270 1. *	*CHN3155C 8. *	*CHN0157C 14. *
3	6	CHNU171A	92.0	10	117.2 32.	1.2 0.7	126.	44.8	1	88.	64.2	116.1 29.	8 3.	*CHN0187A 6. *	*CHN0162C 8. *	*CHN0160C 13. *
3	7	CHN0187A	80.0	10	106.6 26.	1.1 0.9	179.	43.9	2	192.	64.0	104.7 24.	6 1.	*LAO0284E 3. *	*VTN0325C 7.	*THA01420 10.
3	8	CKH9052C	158.0	10	-161.0 -19.	3 1.0 0.6	132.	46 • 1	2	72.	64.7	-163.5 -17.	5 15.	*SM00057C 18.	*NZL3055C 19.	*WALD102C 28-

--- 12 ---

suite --- cont.

	1	2	3	4	1	5	6	7	8	9 a	9 b	10		11	12		13		14	
						1				Ţ.	1						*		±	
39	CLN0219C	50.0	10	80.6	7.7	1.2 0.6	106.	45.7	1	62.	63.7	80.0	10.0	3.	- *IND0043C	6.	*IND0045C	10.	*AFG0246C	14.
4)	5 5 587C	-19.0	15	9.6	49.9	1.6 0.7	147.	43.6	2	150.	65.6	10.0	54.8	1.	*ZA10323C	5.	* *FNL0103C *	6.	*LUX0114C	11.
41	FNL0103C	5.0	10	22.5	64.5	1.4 9.8	171.	44.0	2	241.	67.9	19.2	60.0	Ó.	*D 0087C	8.	*POL0132C	13.	*TUR0145C	20
42	GNP0304C	-31.0	10	-15.0	12.0	0.9 0.6	172.	46.9	2	43.	63.2	-14.2	12.3	3.	*GMB0302C	7.	- *GUI0192C	7.	*IRL0211C	14
43	GUM0331C	122.0	10	144.5	13.1	0.6 0.6	0.	48.7	2	30.	63.4	144.7	13.4	7.	* MRA03320	9.	* *CAR0338C	17.	*AUS0008C	20
44	IND0037C	68.0	10	93•0	25.5	1.5 1.1	49.	42.1	2	157.	64.0	96•0	29.5	-1.	* *CHN0157C	4.	* *CHN0155C	6.	* *CHN0154C	7
45	INDU045C	56.3	10	76.2	19.5	1.6 1.6	21.	40.3	2	216.	63.6	74.4	22.0	3.	≠ *P 4K0127C	5.	* *CHN0154C	14.	* *ARS0275C	15
¥6	IRL02110	-31.0	10	-8.2	53.2	0.8 0.6	162.	47.2	1	52.	64.4	-5.5	54.3	7.	≠ ≠D 0097C	10.	≠ \$GNP0304C	15.	*F 0093C	17
+7	KÜR0112E	110.0	10	127.5	36.0	1.2 1.0	168.	43.2	2	111.	63.7	126.2	33.0	-1.	*PNG0131C	2.	*J 0111F	5.	*J 01113	6
¥8	LAD0284E	74.0	10	103.7	18.1	2.2 0.8	133.	42.0	1	154.	63.9	102.0	22.5	-1.	* *CHN0187A	1.	*THA0142C	7.	*IND0037C	ç
+9	MAU0242C	29.0	10	59•8	-18.9	1.6 1.2	55.	41.2	1	193.	64.1	60.0	-14.0	5.	*ZAI0323C	6.	*MDG0236C	14.	+ + ₽ AK 0127C	21
50	ML 103270	-37.0	10	-2.0	19.0	2.7 1.3	127.	39.0	ì	339.	64.3	-2.5	23.3	9.	*ALG0251C	11.	*SMR 0311C	18.	+ *LIE0253C	20
51	MRL03330	146.0	10	166.7	7.9	1.5 1.5	177.	40.7	1	185.	63.4	162.1	11.5	13.	*PNG0131C	19.	*CAR0338C	19.	*NCL0100C	2
52	NCL01000	140.0	10	166.0	-21.0	1.1 3.7	146.	45.1	1	74.	63.8	167.5	-23.0	-3.	*WAL0102C	-3.	*NHB0128C	11.	*AUS0008C	1
53	PAK01270	38.0	10	69.6	29.5	2.3 2.2	14.	37.3	1	474.	64.0	74.5	35.7	2.	* *CHN0154C	2.	* *AFG0245C	15.	*AFG0246C	2
54	PNG01310	110.0	10	147.7	-6.3	2.5 2.2	169.	36.9	1	577.	64•5	141.0	-9.2	2.		6.	*INS0036C	7.	*KOR01125	1
55	R0U01360	-1.0	10	25.0	45.7	1.4 0.7	155.	44.7	1	84.	63.9	20.2	46.1	-1.	+ +D 0087C	4.	* *BGT0297C	5.	*TCH01440	
56	TCD01430	-13.0	10	18.1	15.5	3.4 1.7	107.	35.6	2	561.	64.1	16.5	8.2	5.	*CME0300C	7.	≁ ≊MLI0327C	14.	*GAB0260C	1
57	TGCJ2260	-25.0	10	0.8	8.6	1.5 0.6	195.	44.6	2	77.	63.5	-0.2	11.1	-2.	*MLI0327C	-0.	*ALG0251C	6.	*DAH02330	
58	WAL01020	: 140.0	10	-176.8	-14.0	0.7 0.6	29.	47.8	1	47.	64.5	-178.1	-14.2	-2.	*NCL0100C	-2.	* *PNG0131C	13.	≁ ¥FJI01930	1
59	YEH02660	: 11.0	10	44.3	15.1	1.1 0.7	109.	45.2	1	56.	62•7	44.0	18.0	-1.	+ +ARS0275C	0.	*YMS0267C	7.	- +UGA00510	1
5)	ZA103230	-19.0	10	21.3	-6.8	2.8 1.5	149.	38.0	1	477.	64.7	16.3	-1.3	1.	- +TC70143C	5.	≁ *GAB026JC	7.	≁⊃ 00870	

١

Canal 10 suite — Channel 10 cont. —[;] Canal 10 cont.

(

Canal 11 — Channel 11 — Canal 11

•

· [1	2	3	4		5	6	7	8	9 a	9 b	10)	11	12		13	14
i i		1	Ì										:	ļ.	•	1		1	
37	2 AFG:	02450	50.0	11	70.2	35.5	1.3 1.1	53.	42.5	1	110.	62.9	68.0	31.6	1.	*!RN0109C *	2•	*РАКО127С 11. *	• ≄ûiNO2190 14. ≑
33	B AUSC	00040	98.0	11	121.8	-24.9	3.6 1.9	54.	35.9	2	527.	63.1	126.6	-31.0	8.	*7000€SP *	8.	*AUS30390 21. *	*INS0036C 22.
34	AUSC	0009C	128.0	11	147.2	-32.0	2.1 1.4	15.	39.6	1	289.	64.2	141.0	-34.0	4•	*AUS 30 37C	6.	*AUSOOO8C 13.	• *PNG02710 13.
3	5 AZRO	01340	-31.0	11	-23.4	36.1	2.6 0.7	158.	41.7	2	138.	63.1	-16.9	32.5	-5.	+ ≠POR0133C	-4.	т *LBк0244С 9. -	*CPV03010 14.
30	5 CHN	0157C	62.0	11	102.3	27.8	2.6 1.6	127.	38.2	2	500.	65.2	101.7	21.3	1.	* *VTN0325C	2.	+ *LA00284₹ 14	. ≠CHN0156C 16.
31	7 CHNC	0160C	92.0	11	122.8	45.3	2.5 1.4	150.	38.7	2	449.	65.2	122.5	38.3	1.	* *J 0111F	4.	* *CHN0157C 7.	* . ≠CHN0170A 12.
31	B COMO	0207C	29.0	11	44.1	-12.1	0.8 0.6	149.	47.7	2	36.	63.2	43.1	-11.3	5.	* *\$0M0312C	9.	* *ZMB0314C 1)	* . ∻MAU02430 16.
آذ	9 DAH0	02330	-19.0	11	2.2	9.5	1.4 0.7	97.	44.3	2	80.	63.4	2.2	6.2	3.	* *TGD02260	5.	* *ZAI0323C 14	* • *ZAI3322C 14•
4	GABC	02600	-13.0	11	11.8	-0.6	1.4 1.1	64.	42-2	1	133.	63.4	11.0	-4.3	3.	* *ZAI0323C	5.	* *ZAIJ322C 94	* • *TCD0143C 14•
4	L GMB:)302C	-37.0	11	-15.1	13.4	0.8 0.6	4.	47.5	2	39.	63•4	-14.2	13.3	3.	* *GNP0304C	7.	* *MLI0328C 9	* *LBR0244C 12.
ĺ																*		*	*
4	2 GRC	3105C	5.0	11	24.7	38-2	1.8 1.3	156.	41.8	1	144.	63.4	26.6	41.5	1.	*BUL 00200 *	4.	*RGU0136C 6 *	- ≠IRN0109C 13. *
÷	3 IND	00430	56.0	11	77.8	11.1	1.4 1.3	172.	41.8	1	145.	63.5	79.5	9.1	4.	*CLN0219C *	5.	*IND00450 15 *	• *IND0040C 19• *
4	4 INC	0047¢	68.0	11	93.3	11.1	1.9 0.6	96.	43.6	1	98.	63.5	93.8	14.9	6.	*CHN0157C	8.	*IND0037C 16	• #IND30480 16•
4	5 INS	0036C	1 04 •0	11	135.2	-3.8	2.5 2.0	147.	37.3	1	451.	63.9	128.0	-7.0	4.	*AUS0004C	5.	*J 0111F 15	• *PNG0131C 16.
4	6 IRN	01090	34.0	11	54•2	32.4	3.8 1.9	149.	35.8	2	510.	62.9	61.5	31.0	6.	+ *AFG0245C	8.	*РАКЭ127C 15	• * SO⊳ 0312C 16•
4	7 J	0111F	110.0	11	134.5	31.5	3.5 3.3	68.	33.6	1	1175.	64.3	129.3	34.7	2.	+ +CHN0160C	7.	*KOR0112F 8	• *KOF 3112E 8.
4	a lbn	02 79 0	11.0	11	35.8	33.9	0.6 0.5	٥.	48.7	2	23.	61.7	36.8	34.5	-0.	* *GRC0105C	6.	* *IRN0109C 6	• #ARS0003C 7.
4	9 LBP	02440	-31.0	11	-9.3	6.6	1.2 0.7	133.	44.9	1	69.	63.3	-10.2	8.5	4.	*AZR 0134C	9.	* ≠MLI0328C 9	. ≠POP01330 12.
5	O LBY	03 21 C	-25.0	11	13.1	27•2	2.4 1.1	129.	40.0	2	205.	63.1	11.4	33.5	3.	* *ALG0251C	6.	* *ALG0252C 9	. *TG00226C 12.
5	1 LIE	0253C	-37.0	11	9.5	47.1	0.6 0.6	0.	48.7	1	24.	62.5	9 •5	47.2	1.	* ≄⊤CH0144C	4.	∓ ≄LUX0114C 8	* • ≄AND03410 13•
. 5	2 L J X	0114 C	-19.0	11	6.0	49.8	0.6 0.6	0.	43.7	1	27.	63.0	7.0	48.5	-4.	≠ *LIE0253C	1.	* ≠D 0087C 3	. ≠ TCH0144C 4.
5	3 MRA	0332C	122.0	11	145.9	16.9	1.2 0.6	76.	45.7	1	62.	63.6	145.0	20.0	-4.	* *J 0111F	-3.	÷ ≭GuM0331C 12	* • *4US0009C 19•
 5	4 NHB	0128C	140.0	11	168.0	-16.4	1.5 0.7	87.	44.1	2	77.	63.0	169.8	-20.0	7.	* *NCL 01 0 0C	8.	* *INSDC36C 23	* *NRUJ3090 23.
د	5 NRU	0309C	134.0	11	167.0	-0.5	0.6 0.6	0.	49.7	2	25.	62.6	165.0	-2.0	10.	* *NHB0128C	14.	* *INS0036C 17	* • *PNG0131C 19•
5	6 POR	0133C	-31.0	11	-8.0	39.6	0.9 0.6	112.	46.8	2	47.	63.5	-7.5	37.2	-15.	* *AZR0134C	-15.	* *LER02440 11	* • *G 00270 14•
1													-	_					

•

,

suite — cont.

e .-

	1	2	3	. 4		E	5	6	7	8	9a	9b	10		11	12		13		14	
							1									*		*		*	
57	SM00057C	158.0	11-	-172.3	-13.7	0.6	0.6	0.	48.7	1	32.	63.8	-171.0	-14-1	7.	*CKN0053C	9.	*CKH0052C	14.	*WAL0102C	2ú
58	SNGJ151C	74.)	11	103.8	1.3	0.6	3.6	0.	48.7	2	31.	63.7	196.0	1.1	3.	*VTN0325C	4.	*BRU0330A	18.	*CHN0157C	20
59	S0M0312C	23.0	11	45.0	6.4	3.3	1.5	71.	37.2	1	332.	62.4	43.2	11.2	5.	+ *URS3360C	8.	*URS0361C	10.	* *YEM0266C	; 15
60	TCH0144C	-1.0	11	17.3	49.3	1.5	0.6	170.	44.8	2	82.	63.9	12.1	50.3	-1.	* *LIE0253C	3.	* *ZMB0314C	5.	* *LUX01140	; 6
61	JGA0051C	11.0	11	32•3	1.2	1.5	1.1	60.	42.1	1	132.	63.3	29.8	-1.3	1.	* *RRW0310C	4.	* *ZMB0314C	8.	* *ZAI0322C	: 16
٥2	URS0061C	23.0	11	24.7	56 .6	0.9	0.6	12.	46.7	2	69.	65.1	25.8	54.1	-0.	* *TCH0144C	6.	* *S0M0312C	6.	* *URS0060C	: 6
63	VTN0325C	86.0	11	105.3	16.1	3.0	1.4	116.	38.0	2	361.	63.5	102.0	22.8	-2.	* *CHN0157C	-1.	* *CHN0187A	9.	* *LAG0284ë	: 14
64	ZMBJ314C	-1.0	11	27.5	-13.1	2.4	1.5	39.	38.8	1	320.	63.8	33.0	-13.8	3.	* ************************************	5.	* *TCH0144C	11.	* *BUT02970	; 13
	• .																				
•	1997 - A.		۰.						1 a						÷.1						
									÷.,						•	• •		а •		14 A. A.	
-					x - 1													· · · ·			
	a ta sea												i.			4				•	
	·		:			۰.					۰.										
	×	* 		. • •								-	÷., •			1. 1. 1. 1.				.•	
			٦								. 4					÷				. · . · ·	-
											5	· · · ,				ана. 1910 — П				· .	a i
	• •			•										ʻ :		tan 1 Ala	•		• • •		:
									•				. •					• 		14月1日) 1月1日年(1月11日)	
	A	· •						· •					· · ·	e.			•				
						·			:			÷.,		6. s.		•.		e Autoria			
	·	×						•										<u>.</u>	•,		
		·																		· · · · ·	
																,					

•

.

ی م ب

Canal 12 — Channel 12 — Canal 12

•

	1	2	3	4	Ī	5	6	7	8	9 a	9b	10		11	12		13	14
34	41.G0252C	-25.0	12	1.6	25 . 5	3.6 2.2	152.	35.3	1	576.	62.9	9.5	30.0	· 4.	+LBY0321C	7.	 ≠LBY0280D 12	• *MLT01470 15.
35	AND0341C	-37.0	12	1.6	42.5	0.6 0.6	0.	48•7	2	19.	61.6	1.4	42.4	-1.	*G 0027C	3.	≠ *ALG0252C 5	* • *MLI0328C 7.
36	AR \$0 003C	17.0	12	41.1	23.8	3.5 1.7	134.	36.5	2	426.	62.8	56.0	21.0	-1.	* *YMS0267D	2.	* *EGY0026C 9	* *QAT0247D 10.
37	AUS0007C	128.0	12	145.0	-38.1	1.8 1.4	134.	40.2	2	210.	63.4	149.6	-37.5	1.	* *PNG0271C	з.	* *AUS0009C 7.	* • *NZL0287A 14.
3B	AUTJ016C	-19.0	12	12.1	47.5	1.1 0.6	166.	45.7	2	72.	64.2	17.1	48.0	ð.	* *ZAI 0322C	4.	* *F 0093D 10.	* *URS0060C 12.
39	BRUD330A	74.0	12	114.7	4.4	0.6 0.6	э.	48.7	1	24.	62.5	114.6	4.0	11.	* *THA0142D	15.	* *CHN0180A 18.	* *VIN0325C 22
40	BUL0J20C	-1.0	12	25.0	43.0	1.0 0.6	165.	46.3	1	56.	63.8	28.1	42.0	-2.	* *MD70307C	3.	* *TUR01450 5.	* * * * GRC 0105C 6-
41	CHN3156C	62.0	12	97.8	36.3	2.6 1.6	157.	38.2	1	347.	63.6	105.2	32.6	3.	* *	5	* * HN0170A 10	* * UN01550 10
42		92.0	12	110 5	33.0	1 2 0 6	155	44 9	1	00	60.0 64 4	110 7	21 2	3.	* +CHNC179A	9• 2	+CUN015(C 0	*
42	CUN01704	92.0	12	111 5	27 4	1 2 0 0	120	44 0	1 2	1.17	66 A	112 6	20.0		*CHN0170A	3. 5	*CHN0156C 9	* * * * * * * * * * * * * * * * * * *
45	LINULIDA	80.0	12	111.5	21.4	1.2 0.9	150.	44.0	2	1010	04.4	115.0	29.0	D •	* +	• C	* *	*
44	CKN0053C	158.0	12-	163.0	-11.2	1.8 0.7	30.	43•2	2	131.	64•4	-158.0	-9.0	4•	*00501010	4•	*NZL0055D 18-	* ************************************
45	CPV0301C	-31.0	. 12	-24.0	16.0	0.9 0.7	144.	46.5	2	38.	62.3	-24.3	14.4	4.	*G 0027C *	6.	*AZR0134C 14	*#LI0328C 16.
46	DNK0089A	5.0	12	12.3	57.1	1.2 0.6	· 177.	45.7	2	74.	64.3	10.0	54.5	1.	*G 0027C *	5.	*POL0132D 8. *	*AUT0016C 8.
47	EGY0026C	-7.0	12	29.7	26.8	2.3 1.7	136.	38.2	2	315.	63.2	36.7	23.3	-0.	*ARS0003C *	0.	*ALG0252C 12	*MGZ0307C 16.
43	G 0027C	-31.0	12	-3.5	53.8	1.8 3.7	142.	43.0	1	162.	65.1	1.3	51.1	3.	*ALG0252C	5.	*AUT0016C 10.	*DNKJ089A 13.
49	IND0040C	56.0	12	73.0	25.0	1.8 1.5	58.	39.9	2	241.	63•8	73.5	30.0	-1.	*PAK0210A	-0.	*CHN0155D 10	*URS0069A 13.
50	IND0048C	68.0	12	86.2	25.0	1.6 0.9	120.	42.8	2	126.	63.8	89•8	26•7	-3.	*CHN 0156C	1.	+ ≠CHN0155D 3	*CHN0157C 6.
51	KORO112F	110.0	12	127.5	36.0	1.2 1.0	168.	43.2	2	112.	63.7	126.2	33.0	-1.	* *CHN0170A	3.	*J 0111G 5.	*J 0111F 6.
52	MAU0243C	29.0	12	56.8	-13.9	1.6 1.4	65.	40.9	1	195.	63.8	53.0	-15.0	6.	* *MDG0236D	9.	* ≄MOZ0307C 11.	* *COM0207C 18.
53	MLD0306A	44.0	12	73.1	6.0	1.0 0.6	90.	46.6	1	50.	63.7	71.0	7.5	2.	* *URS0069A	6.	* *URS0067D 8-	* *INDJ039D 14.
54	4L10328C	-37.0	12	-7.6	13.2	1.7 1.2	171.	40.9	1	196.	63.8	-12.0	15.0	3.	* *GMB0302C	8.	* *GUI3192D 8	* • *Alg0252c 9•
55	MLT0147C	-13.0	12	14.3	35.9	9.6 0.6	0.	48•7	1	17.	61.1	14.3	35.9	0.	* *alg0252C	2.	* *EGY0026C 11	* *CME0300D 12.
56	NOZ0307C	-1.0	12	34.0	-18.0	3.6 1.4	55.	37.3	2	499.	64.3	30.3	-15.0	5.	* *ZMB0314C	6.	* *BUL0020C 13	* • *ZAI0322C 17•
57	0CEJ101C	-160.0	12-	-145.0	-16.3	4.3 3.5	4.	32.4	2	1335.	63.6	-154.7	-15.7	17.	* *CKN0053C	17.	* *PNG0271C 30	* • *TUN0215C 33•
58	PAK0210A	38.0	12	72.1	30.8	1.2 0.7	91-	45.7	1	71-	63.5	72.3	33.7	-7.	* *UR \$00694	0.	* *IND0040C 5	* *AFG0245C 13
60	DNC02710	120 0	12	169 0	7	2920	165	36 7	-	492	42 E	141 0	-0 2	2	# #	5	* ************************************	÷ *ù71 1)2878 12
127	PN602110	158.0	12	140.0	-0.1	2.0 2.0	100.	20+1	T	702.	02+2	141.0	-706	3.	-403-JUJ7C	9.	#402111AC TO	• THELVEOIR IE.

suite — cont.

, *

~

.

` **x** _

- 16 -

Canal 12 suite — Channel 12 cont. — Canal 12 cont.

.

•

Ł

·. ·.

	Ι	1	2	3	4		5	6	7	8	9 a	9 b	10		11	12		13		14	
0	0 7	RRW0310C	11.0	12	30.0	-2.1	0.7 0.6	42•	48.3	2	46.	64.9	30.3	-1.0	0.	#ZAI03220	2.	 ×UGA0051C 8) 3. 4	YMS02670	14.
6	1 9	STP0241C	-13.0	12	7.0	0.8	0.6 0.6	ð.	48.7	2	19.	61.5	5.0	2.0	-2.	* *CME0300D	2.	¥ ¥ZAI0322C 3	4 8. 4	*GAB0260C	6.
6	2 1	ronj215C	170.0	12-	174.7	-18.0	1.4 3.7	85.	44.4	1	79.	63.4	-173.7	-15.9	4.	* *CKN0053C	6• [°]	* ⊭OCE0101C 13	ہ ۲ • 5	*SMA03350	15.
0	3 1	JR 50060C	23.0	12	41.5	57.4	3.1 1.6	153.	37.4	1	876.	66.9	27.7	60.5	5.	* *URS0J61C	8• 3	⊭ *DNK3089A 11		*SC40312C	19.
0	4 i	JR 50069A	44.)	12	70.8	38.5	1.4 0.7	161.	44•2	2	97.	64.1	75.1	37.3	-1.	* ≠PaKU210A	1.	F FINDOO40C &	ہ 1- • •	MLD03064	11.
6	5 7	ZA10322C	-19.0	12	22.4	0.0	2.2 1.9	48.	38.2	1	462.	64.8	19.5	5.0	4.	* *CME0300D	8.	* *Alg0252C 9	•	*AUT0016C	12.
С	ana	ı 13 –	– Chan	nel	13 _	- Car	nal 13														
[1	2	3	4	1	5	6	7	8	9 a	9 b	10		11	12		13		14	
	-				· · · · · · · · · · · · · · · · · · ·											1					
E	3	AFG0246D	50.0	13	64.5	33.1	1.4 1.4	21.	41.2	1	145.	ó2.9	67.7	37.3	-2.	¥URS0067D *	1.	*CHN0155D :	5.	*TUR01450 *	11.
3	34	4 USO 005 D	98.0	13	133.5	-18.8	2.7 1.4	76.	38.5	2	394.	64•4	130.8	-12.5	2.	*INS00350	2.	*AUSJJJ6D 1 *	6.	*CHNU18JA *	18.
3	85	CAR0338D	122.0	13	149.5	8.0	5.4 0.8	178.	38.1	1	286.	62.6	134.6	7.5	3.	*INS0035D *	6.	*J 0111G	7.	*CHN0180A *	19.
1	86	CHN0155D	62.0	13	88.3	31.5	3.4 1.4	162.	37.3	2	370.	63 . J	86•9	28.0	-0.	*IND0044D	3.	*IND0048C	5.	*CHN0156C *	12.
1	57	CHN0183A	92.)	13	113.7	12.9	3.8 2.2	72.	35.1	2	713.	63.6	109.3	17.2	3.	*THA01420	4.	*J 0111G 1	5.	*AUS0005D	19.
3	88	CME0300D	-13.0	13	12.7	6.2	2.5 1.7	87.	37.9	1	366.	6 3.6	16.2	1.4	2.	*ZAI33220	4.	*Z&10323D *	8.	*TCDJ143D *	9.
3	39	F 0093D	-19.0	13	2.6	45.9	2.5 1.0	160.	40.4	1	229.	64.0	9•5	41.2	-2.	≠SMR03110 *	1.	*D 0087D *	9.	*ALG02510 *	9.
4	40	GUI 01920	-37.0	13	-11.0	10.2	1.6 1.0	147.	42.1	2	141.	63.6	-14.5	11.5	3.	*GNP0304D ≭	5.	*ML፤0328C ≑	8.	*4LI0327D *	15.
4	+1	IND0039D	56.0	13	72.7	11.2	1.3 0.6	107.	45.5	1	60.	63.3	73.0	8.3	7.	*MLƏ0306A ≠	13.	*IND0045D 1 *	4.	*IND004JC *	15.
4	42	IND0044D	68.0	13	79.5	22.3	2.2 1.4	146.	39.3	1	263.	63.5	78.2	26.9).	*CHN 0155D *	2.	*CHN0154D 1 *	3.	*IND0048C *	14.
4	43	IN\$00350	104.0	13	124.3	-3.2	3.3 1.9	82.	36.1	1	528.	63.4	125.5	4.0	4.	*J 01116 *	ó.	*CHN9180A *	9.	*CARC338D *	15.
4	44	J 0111G	110.0	13	134.5	31.5	3.5 3.3	68.	33 .6	1	1183.	64.3	123.7	24.3	2.	*CHN0180A *	3.	*PNG0131D 1 *	2.	*INS00359 *	13.
4	45	L8Y0280D	-25.0	13	21.4	26.0	2.5 1.0	119.	40.1	2	226.	63.6	22.0	33•0	4.	*TUR01450 *	8.	*ALG0251D 1 *	1.	*ALG0252C *	12.
-	+6	MDG0236D	29.0	13	46.6	-18.8	2.7 1.1	65.	39.3	2	260.	63.5	49.0	-12.3	. 8.	*MAU0243C	10.	*MAU0242D 1 *	5.	*ZAI0323D ≭	22.
4	47	NZL00550	158.0	13	172.3	-39.7	2.9 1.6	47.	37.7	1	379.	63.5	174.0	-46.)	-1.	*NZL0287A *	-1.	*AUS0005D 2	2.	*CKN0053C *	23.
4	4 8	NZL0287A	128.0	13	173.0	-41.0	3.3 1.3	48.	38.0	1	476.	64.8	160.3	-45.5	1.	*\ZL0055D *	2.	*PNG0271C 1	4.	*AUS0008D *	16.
	49	PLM03370	170.0	13	-161.4	7.0	0.6 0.6	0.	48.7	1	24.	62.6	-162.4	6.1	5.	*CAR0338D *	7.	*TON0215C 1 *	4.	≠SMA0335∪ ≭	16.
	50	PCL01320	-1.0	13	19.3	51.8	1.5 0.6	162.	44.5	2	9 4.	64.3	22.9	49.0	-0.	*5 MR 0311D	5.	*TUR 01 45D	7.	*R0001360	7.

.

suite — cont.

•

--- 17 ---

Canal 13 suite — Channel 13 cont. — Canal 13 cont.

	1	2	3	4		5	6	7	8	9 a	9 b	10)	11	12		13		14	
51	QA TO 247D	17.0	13	51.1	25.3	0.6 0.6	0.	48.7	1	21.	62.0	52.3	24-8	. 1.	" *YMS02670 *	5.	™ *AR\$J2 7 50	6.	+ *4R50003C *	9.
52	SMA0335D	179.0	13-	-170.1	-14.2	0.6 0.6	0.	48.7	2	18.	61.3	-171.0	-11.0	-3.	*TON0215C *	1.	*PLM0337D	2.	*CKN0053C	5.
53	SMR0311D	-37.0	13	12.6	43.7	0.6).6	0.	48.7	1	25.	62.6	12.0	43.)	-J.	≠≓ 00930 ≠	2.	*GUI0192D *	9.	*TUR 31450	12.
54	SWZ0313D	-1.0	13	31.5	-26.5	0.6 0.6	66.	48.5	1	28.	63.0	31.1	-25.9	4.	*MGZ0307C	7.	*801)2970 *	9.	*POL01320	11.
55	THA0142D	74.0	13	100.7	13.2	2.8 1.5	106.	37.9	2	389.	63.8	105.4	14.3	3.	*CHN0180A	3.	*CHN0155D	22.	*J_0111G *	22.
56	TUR0145D	5∙0	13	34.4	38.9	2.7 1.0	168.	39.8	1	255.	63.9	42•7	41.5	-2.	*URS00640 *	-1.	*LBY02800	11.	*URS0067D	12.
57	URS3364D	23 . J	13	45.6	40.8	2.2 0.6	163.	43.1	2	.124.	64.1	40.0	43.4	-1.	*TUP 0145D	2.	*URS0060C *	6.	*URSJ067D *	12.
58	UR\$30675	44.0	13	62.4	58.5	3.2 1.5	169.	37.4	1	912.	67.0	50.9	51.9	12.	*URS0060C	17.	*TUR 0145D *	17.	*AFG0246D	21.
59	WA K0334D	140.0	13	166.5	19.2	3.6 3.6	0.	48.7	1	32.	63.7	166.5	19.2	17.	*J 0111G ≆	22.	*MRL13330	24.	*NCL0100D	25.
6.)	YMS3267D	11.0	13	48.8	15.2	1.8 1.5	176.	39.9	2	202.	63.0	49.5	18.4	3.	* AR 5 02 750	6.	*AR \$0003C	7.	*YEM0266D	13.
1																				1
Cai	al 14 -	– Chan	nel	14 -	– Car	nal 14							•		. . .					
	1	2	3	4		5	6	7	8	9 a	9 b	10)	11	12		13		14	
29	ALG02510	′ −25.0′	14	4-2	33.2	2.4 1.3	' 172 [_]	39.4	' 1'	261.	63-6	9.5	30.0	·).	***********	5.	*I 8V03210	7	+TG002260	
								5761							*		*		*	0•
30	AR \$0275D	17.0	14	48.3	24.6	3.8 1.4	138.	36.9	2	395.	62.9	42.5	17.0	-11.	* *YEM0266D- *	-11.	* * *YMS0267D *	-1.	* *TCD01430 *	12.
30 31	AR S0275D AUS0006D	17•0 98•0	14 14	48.3 135.4	24•6 -30•3	3.8 1.4 2.0 1.4	138. 44.	36.9 39.8	2 1	395. 230.	62 . 9 63 . 4	42.5 140.8	17•0 -27•7	-11. 1.	* *YEM0266D- * *AUS0008D *	-11.	* *YMS0267D * *AUS0005D	-1. 8.	* * *TC001430 * *AUS00040 *	8. 12. 14.
30 31 32	AR SO 2750 AUSO 0060 AUSO 0080	17.0 98.0 128.3	14 14 14	48.3 135.4 145.9	24.6 -30.3 -21.5	3.8 1.4 2.0 1.4 2.9 2.3	138. 44. 120.	36.9 39.8 36.6	2 1 2	395. 230. 530.	62.9 63.4 63.9	42.5 140.8 149.0	17•0 -27•7 -29•0	-11. 1. 3.	* * *YEM0266D- * *AUS0008D * *AUS0009D *	-11. 3. 6.	* * * * * * * * * * * * * * * * * * *	-1. 8. 6.	* * *TCD0143D * *AUS0004C * *NZL0287A *	12. 14. 17.
30 31 32 33	AR S0275D AUS0006D AUS0008D B0T0297D	17.0 98.0 128.3 -1.0	14 14 14 14	48.3 135.4 145.9 23.3	24.6 -30.3 -21.5 -22.2	3.8 1.4 2.0 1.4 2.9 2.0 2.1 1.5	138. 44. 120. 36.	36.9 39.8 36.6 39.2	2 1 2 2	395. 230. 53J. 296.	62.9 63.4 63.9 63.9	42.5 140.8 149.0 25.3	17.0 -27.7 -29.0 -17.8	-11. 1. 3. 4.	* * *YEM0266D- * *AUS0008D * * *AUS0009C * * ZMB0314D	-11. 3. 6.	* * * * * * * * * * * * * * * * * * *	-1. 8. 6. 1).	*TC001430 * *AUS0004C * *NZL0287A * *R0U01360	12. 14. 17. 11.
30 31 32 33 34	AR S0275D AUS0006D AUS0008D B0T0297D BRU0330B	17.0 98.0 128.3 -1.0 74.3	14 14 14 14 14	48.3 135.4 145.9 23.3 114.7	24.6 -30.3 -21.5 -22.2 4.4	3.8 1.4 2.0 1.4 2.9 2.0 2.1 1.5 0.6 0.6	138. 44. 120. 36. 0.	36.9 39.8 36.6 39.2 48.7	2 1 2 2 1	395. 230. 53J. 296. 24.	62.9 63.4 63.9 63.9 63.9	42.5 140.8 149.0 25.3 114.6	17.0 -27.7 -29.0 -17.8 4.0	-11. 1. 3. 4. 9.	* *YEM0266D- * *AUS0008D * *AUS0009E * *ZMB0314D * *BGD0220A *	-11. 3. 6. 14.	* *YMS0267D * *AUS0005D * *AUS0006D * *ZAI0323D * *THA01429 *	-1. 8. 6. 1). 15.	* *TC001430 * *AUS0004C * *NZL0287A * *ROU0136P * *CHN0180A *	 12. 14. 17. 11. 18.
30 31 32 33 34 39	AR S0275D AUS0006D AUS0008D B0T0297D BRU0330B CHN0154D	17.0 98.0 128.3 -1.0 74.3 62.0	14 14 14 14 14	48.3 135.4 145.9 23.3 114.7 83.9	24.6 -30.3 -21.5 -22.2 4.4 40.5	3.8 1.4 2.0 1.4 2.9 2.3 2.1 1.5 0.6 0.6 2.8 2.0	138. 44. 120. 36. 0. 177.	36.9 39.8 36.6 39.2 48.7 36.7	2 1 2 2 1 1	395. 230. 530. 296. 24. 461.	62.9 63.4 63.9 63.9 63.9 62.6 63.4	42.5 140.8 149.0 25.3 114.6 79.0	17.0 -27.7 -29.0 -17.8 4.0 34.3	-11. 1. 3. 4. 9.	* * *YEM0266D- * *AUS0008D * *AUS0009E * *ZMB0314D * *BGD0220A * *PAK02108	-11. 3. 6. 14. 5.	* *YMS0267D * *AUS0005D * *AUS0006D * *ZAI0323D * * THA01429 * *CHN0155D *	-1. 8. 6. 1). 15. 8.	*TCD0143D *AUS0004C * *NZL0287A * *ROU0136D * *CHN0180A * *IND00450 *	 12. 14. 17. 11. 18. 15.
30 31 32 33 34 35 36	AR S02750 AUS00060 AUS00080 BOT02970 BRU03308 CHN01540 CHN0172A	17.0 98.0 128.3 -1.0 74.0 62.0 92.0	14 14 14 14 14 14	48.3 135.4 145.9 23.3 114.7 83.9 120.4	24.6 -30.3 -21.5 -22.2 4.4 40.5 29.1	3.8 1.4 2.0 1.4 2.9 2.3 2.1 1.5 0.6 0.6 2.8 2.0 1.0 J.8	138. 44. 120. 36. 0. 177. 123.	36.9 39.8 36.6 39.2 48.7 36.7 45.2	2 1 2 1 1 1	 395. 230. 530. 296. 24. 461. 82. 	62.9 63.4 63.9 63.9 63.9 62.6 63.4 64.3	42.5 140.8 149.0 25.3 114.6 79.0 118.8	17.0 -27.7 -29.0 -17.8 4.0 34.3 27.5	-11. 1. 3. 4. 9. 3. 2.	* * *YEM02660- * *AUS00080 * *AUS00090 * *ZMB03140 * *BGD0220A * *BGD0220A * *PAK02108 * *CHN0174A	-11. 3. 6. 14. 5. 6.	* * YMS 0267D * * AUS 0005D * * AUS 0005D * * AUS 0006C * * ZAI 0323D * * THA 01429 * * CHN0155D * * CHN0131A *	-1. 8. 6. 1). <u>1</u> 5. 8. 7.	* *TCD0143D * *AUS0004C * *NZL0287A * *ROUJ136D * *CHN0180A * *CHN0180A *	 12. 14. 17. 11. 18. 15. 10.
30 31 32 33 34 35 36 37	AR S0275D AUS0006D AUS0008D B0T0297D BRU0330B CHN0154D CHN0172A CHN0181A	17.0 98.0 128.3 -1.0 74.0 62.0 92.0 80.0	14 14 14 14 14 14 14	48.3 135.4 145.9 23.3 114.7 83.9 120.4 108.5	24.6 -30.3 -21.5 -22.2 4.4 40.5 29.1 23.8	3.8 1.4 2.0 1.4 2.9 2.3 2.1 1.5 0.6 0.6 2.8 2.0 1.3 3.8 1.4 1.1	138. 44. 120. 36. 0. 177. 123. 153.	36.9 39.8 36.6 39.2 48.7 36.7 45.2 42.4	2 1 2 1 1 1 1 2	 395. 230. 530. 296. 24. 461. 82. 148. 	62.9 63.4 63.9 63.9 62.6 63.4 64.3 64.1	42.5 140.8 149.0 25.3 114.6 79.0 118.8 104.5	17.0 -27.7 -29.0 -17.8 4.0 34.3 27.5 24.5	-11. 1. 3. 4. 9. 3. 2. 3.	* *YEM02660- * *AUS00080 * *AUS00090 * *ZMB0314D * *BGD0220A * *BGD0220A * *BGD0220A * * *BGD0220A * * *BGD0220A * * * * * * * * * * * * * * * * * * *	-11. 3. 6. 24. 5. 6. 7.	* *YMS0267D * *AUS0005D * *AUS0006D * *ZAI0323D * *THA01429 * *CHN0155D * *CHN0131A * *THA0142D	-1. 8. 6. 1). <u>1</u> 5. 8. 7.	*TC001430 *AUS0004C * *NZL0287A * *ROU0136D * *CHN0180A * *CHN0180A * *CHN0180A *	 a. 12. 14. 17. 11. 18. 15. 10. 10.
30 31 32 33 34 35 36 37 36	AR S0275D AUS0006D AUS0008D BOT0297D BRU0330B CHN0154D CHN0172A CHN0181A CKH0052D	17.0 98.0 128.3 -1.0 74.3 62.0 92.0 80.0 158.0	14 14 14 14 14 14 14 14	48.3 135.4 145.9 23.3 114.7 83.9 120.4 108.5 -161.0	24.6 -30.3 -21.5 -22.2 4.4 40.5 29.1 23.8 -19.8	3.8 1.4 2.0 1.4 2.9 2.3 2.1 1.5 0.6 0.6 2.8 2.0 1.0 J.8 1.4 1.1 1.0 0.6	138. 44. 120. 36. 0. 177. 123. 153. 132.	36.9 39.8 36.6 39.2 48.7 36.7 45.2 42.4 46.1	2 1 2 1 1 1 2 2 2 2 2	 395. 230. 530. 296. 24. 461. 82. 148. 73. 	62.9 63.4 63.9 63.9 62.6 63.4 64.3 64.1 64.8	42.5 140.8 149.0 25.3 114.6 79.0 118.8 104.5 -163.5	17.0 -27.7 -29.0 -17.8 4.0 34.3 27.5 24.5 -17.5	-11. 1. 3. 4. 9. 3. 2. 3. 15.	* *YEM02660- * *AUS00090 * *AUS00090 * *BGD0220A * *BGD0220A * *BGD0220A * *BGD0220A * * *CHN0174A * * * * * * * * * * * * * * * * * * *	-11. 3. 6. 24. 5. 6. 7. 18.	* *YMS0267D * *AUS0005D * *AUS0006D * *ZAI0323D * *THA0142D * *CHN0155D * *CHN0131A * *THA0142D * *NZL0055D	-1. 8. 6. 1). 15. 8. 7. 10.	*TC001430 *AUS0004C * *NZL0287A * *ROUJ136D * *CHN0180A * *CHN0180A * *CHN0180A * *CHN0188A * * * * * * * * * * * * *	 12. 14. 17. 11. 18. 15. 10. 10. 28.
30 31 32 33 34 35 36 37 36 37 36 39	AR S02750 AUS00060 AUS00080 BOT02970 BRU03308 CHN01540 CHN0172A CHN0181A CKH00520 CLN02190	17.0 98.0 128.3 -1.0 74.3 62.0 92.0 80.0 158.0 50.0	14 14 14 14 14 14 14 14 14	48.3 135.4 145.9 23.3 114.7 83.9 120.4 108.5 -161.0 80.6	24.6 -30.3 -21.5 -22.2 4.4 40.5 29.1 23.8 -19.8 7.7	3.8 1.4 2.0 1.4 2.9 2.0 2.1 1.5 0.6 0.6 2.8 2.0 1.0 J.8 1.4 1.1 1.0 0.6 1.2 0.6	138. 44. 120. 36. 0. 177. 123. 153. 132. 106.	36.9 39.8 36.6 39.2 48.7 36.7 45.2 42.4 46.1 45.7	2 1 2 1 1 1 2 2 1 1 2 2 1	 395. 230. 530. 296. 24. 461. 82. 148. 73. 63. 	62.9 63.4 63.9 63.9 62.6 63.4 64.3 64.3 64.1 64.8 63.8	42.5 140.8 149.0 25.3 114.6 79.0 118.8 104.5 -163.5 80.0	17.0 -27.7 -29.0 -17.8 4.0 34.3 27.5 24.5 -17.5 10.0	-11. 1. 3. 4. 9. 3. 2. 3. 15. 3.	*** *YEM02660- * *AUS00090 * *AUS00090 * *BGD0220A * *BGD0220A * *BGD0220A * *BGD0220A * * * * * * * * * * * * *	-11. 3. 6. 24. 5. 6. 7. 18. 6.	* *YMS 0267D * *AUS 0005D * AUS 0006D * * *ZAI 0323D * THA 0142D * *CHN0155D * *CHN0131A * *THA0142D * *CHN0131A * *THA0142D * * * THA0142D * *	-1. 8. 6. 1). 15. 8. 7. 10. 19. 10.	*TCD01430 *AUS0004C * *AUS0004C * *ROU0136D * *CHN0180A * *CHN0180A * *CHN0180A * *CHN0180A * *CHN0180A * *CHN0180A * *CHN0188A * *HAL0102D * *AFG0246D	 12. 14. 17. 11. 18. 15. 10. 10. 28. 14.
30 31 32 33 34 35 36 37 36 37 36 39 4	AR S02750 AUS00060 AUS00080 BRU03308 CHN01540 CHN0172A CHN0181A CKH00520 CLN02190 D 00870	17.0 98.0 128.3 -1.0 74.3 62.0 92.0 80.0 158.0 50.0 -19.0	14 14 14 14 14 14 14 14 14 14 14	48.3 135.4 145.9 23.3 114.7 83.9 120.4 108.5 -161.0 80.6 9.6	24.6 -30.3 -21.5 -22.2 4.4 40.5 29.1 23.8 -19.8 7.7 49.9	3.8 1.4 2.0 1.4 2.9 2.3 2.1 1.5 0.6 0.6 2.8 2.0 1.3 3.8 1.4 1.1 1.0 0.6 1.2 0.6 1.6 0.7	138. 44. 120. 36. 0. 177. 123. 153. 132. 106. 147.	36.9 39.8 36.6 39.2 48.7 36.7 45.2 42.4 46.1 45.7 43.6	2 1 2 1 1 1 2 2 1 2 1 2	 395. 230. 530. 296. 24. 461. 82. 148. 73. 63. 162. 	62.9 63.4 63.9 63.9 62.6 63.4 64.3 64.3 64.1 64.8 63.8 65.7	42.5 140.8 149.0 25.3 114.6 79.0 118.8 104.5 -163.5 80.0 10.0	17.0 -27.7 -29.0 -17.8 4.0 34.3 27.5 24.5 -17.5 10.0 54.8	-11. 1. 3. 4. 9. 3. 2. 3. 15. 3. -0.	* *YEM02660- * *AUS00090 * *ZMB0314D * *BGD0220A * *BGD0220A * *BGD0220A * * * * * * * * * * * * *	-11. 3. 6. 14. 5. 6. 7. 18. 6. 3.	* YMS 0267D * AUS 0005D * AUS 0006D * AUS 0006D * ZAI 0323D * THA 0142D * CHN0155D * * CHN0155D * * CHN0131A * THA 0142D * * XL0055D * * XL0055D * * XL0025D	-1. 8. 6. 1). 15. 8. 7. 10. 19. 10. 5.	*TC001430 *AUS00046 *NZL0287A *ROU0136D *CHN0180A *CHN0180A *CHN0180A *CHN0180A *CHN0158A *CHN0158A *CHN0158A *CHN0158A *CHN0158A *	 3. 12. 14. 17. 11. 18. 15. 10. 10. 28. 14. 11.
30 31 32 33 34 35 36 37 36 37 36 39 42 41	AR S0275D AUS0006D AUS0008D BRU0330B CHN0154D CHN0154D CHN0172A CHN0181A CKH0052D CLN0219D D 0087D GNPJ304D	17.0 98.0 128.3 -1.0 74.0 62.0 92.0 80.0 158.0 50.0 -19.0 -31.0	14 14 14 14 14 14 14 14 14 14 14 14 14	48.3 135.4 145.9 23.3 114.7 83.9 120.4 108.5 -161.0 80.6 9.6 -15.3	24.6 -30.3 -21.5 -22.2 4.4 40.5 29.1 23.8 -19.8 7.7 49.9 12.0	3.8 1.4 2.0 1.4 2.9 2.3 2.1 1.5 0.6 0.6 2.8 2.0 1.3 J.8 1.4 1.1 1.0 0.6 1.2 0.6 1.6 0.7 J.9 J.6	138. 44. 129. 36. 0. 177. 123. 153. 132. 106. 147. 172.	36.9 39.8 36.6 37.2 48.7 36.7 45.2 42.4 46.1 45.7 43.6 46.9	2 1 2 1 1 1 1 2 2 1 2 2	 395. 230. 530. 296. 24. 461. 82. 148. 73. 63. 162. 43. 	62.9 63.4 63.9 63.9 62.6 63.4 64.3 64.3 64.1 64.8 63.8 65.7 63.3	42.5 140.8 149.0 25.3 114.6 79.0 118.8 104.5 -163.5 80.0 10.0 -14.2	17.0 -27.7 -29.0 -17.8 4.0 34.3 27.5 24.5 -17.5 10.0 54.8 12.3	-11. 1. 3. 4. 9. 3. 2. 3. 15. 3. -0. 3.	* *YEM02660- * *AUS00090 * *AUS00090 * *AUS00090 * *BGD0220A * *BGD0220A * *BGD0220A * *BGD0220A * * * * * * * * * * * * *	-11. 3. 6. 2.4. 5. 6. 7. 18. 6. 3. 7.	* *YMS0267D * *AUS0005D * *AUS0006D * *ZAI0323D * *THA01429 * CHN0155D * *CHN0155D * *CHN0131A * THA0142D * *NZL0055D * *IND0045D * *ZAI0323D * *GUI0192D	-1. 8. 6. 1). 15. 8. 7. 10. 19. 10. 5. 7.	*TC001430 *TC001430 *AUS0004C * *NZL0287A * *CHN0180A * *CHN0180A * *CHN0180A * *CHN0158A * * * * * * * * * * * * *	 12. 14. 17. 11. 18. 15. 10. 10. 28. 14. 14.

- 18 -

suite -- cont.

Canal 14 suite --- Channel 14 cont. --- Canal 14 cont.

1

.

	1	2	3	4		5	6	7	8	9 a	9b	10		11	12	13	14
43	IND3037D	68.0	14	93.0	25.5	1.5 1.1	40•	42.1	2	159.	ó4 . 1	96.0	29.5	2.	×CHN01550 6.	*CHN0154D 7.	*THA01420 15.
44	IND00450	56.0	14	76.2	19.5	1.6 1.6	21.	40.3	2	218.	63.7	74.0	15.7	5.	*1ND0043D 10.	*1ND0039D 10.	*ARS02757 11.
;45 1	IRL02110	-31.0	14	-8.2	53.2	0.8 0.6	162.	47.2	1	52.	64.4	-5.5	54.3	5.	*NOF 3123A 9.	*D 00870 1J.	*GNP03040 15.
46	KREJ286A	110.0	14	127.1	40.1	1.1 0.8	31.	45.2	2	76.	64.0	128.4	38.6	-0.	≠⊃NG0131D 2• *	*J 0111H 6. *	*J 0111G 6. *
47	MAU0242D	2 9. 0	14	59.3	-18.9	1.6 1.2	55.	41.2	1	195.	64.1	63.0 -	-14.3	5.	*2A10323D 6. *	*MDG0236D 14. *	*ARS02750 25.
48	ML10327D	-37.0	14	-2.0	19.0	2.7 1.3	127.	39.0	1	344.	64.4	-2.5	23.3	9.	*\LGJ251D 11. *	*\$MRJ311D 18. *	*L1cJ2530 2). *
49	MRL03330	146.0	14	166.7	7•9	1.5 1.5	177.	40.7	1	187.	63.5	162.1	11.5	13.	*PNG01310 19. *	*CAR0338D 19. *	*NCL01000 22. *
50	NCL0100D	140.0	14	166.0	-21.0	1.1 0.7	146.	45.1	1	75.	63.9	167.5	-23.0	-3.	*₩AL0102D -3. *	*NHB0128D 11. ≠	*4US00080 15. *
51	NORO120A	5.0	14	13.1	64.1	1.8 0.9	10.	42.2	2	195.	65.0	7.6	58.0	1.	*D 0087D 2. *	*IRL02110 14. *	*PCL01320 14. *
52	PAK3210B	38.J	14	72.1	30.8	1.2 0.7	90.	45.0	1	71.	63.6	75.2	32.2	1.	*CHN0154D 2. *	*IND0045D 15. *	*4FG0245D 17. *
53	PNG01310	110.0	14	147.7	-6.3	2.5 2.2	169.	36.9	1	585.	64.6	141.0	-9.2	2.	*4US00080 6. *	*1N30036D 7. *	*KRE0286A 12. *
54	R0U01360	-1.0	14	25.0	45•7	1.4 0.7	155.	44.7	1	85.	64.)	20.2	46.1	-1.	*0 00870 4. ≄	*88732970 5. *	*TCH01449 7. * -
55	TCD0143D	-13.0	14	18.1	15.5	3.4 1.7	107.	36.6	2	568.	64.1	16.5	8.2	5.	*CME03000 7. *	*MLI03270 14. *	*GA802600 15. *
56	16002260	-25.0	14	0.8	8.6	1.5 0.6	105.	44.6	2	78.	63.5	-0.2	11.1	-2.	*ML103270 -0. *	*ALG0251D 6. *	*DAH02330 8. *
57	WALJ102D	140.0	14	-176.8	-14.0	0.7 0.6	29.	47.8	1	48.	64 •6	-178.1	-14.2	-2.	*NCL01000 -2. *	*PNG0131D 13• *	*MRL03330 20. ★
58	YEM02660	11.0	14	44.3	15.1	1.1 0.7	109.	45.2	1	57.	62.9	44.0	18.0	-1.	*4RS0275D 0.	*YMS0267D 7. *	*UGA0951D 11. *
59	ZA 193230	-15.0	14	21.3	-6.8	2.8 1.5	149.	38.7	ì	433.	64.8	16.3	-1.0	1.	**CD01430 5.	*GAB0260D 8.	*D 0087D 8.

Canal 15 — Channel 15 — Canal 15

	1	2	3	4		5	6	7	8	9 a	9 b	10	11	12		13	14	
32	AFG0245D	50.0	15	70•2	35.5	1.3 1.1	53.	42•5	1	112.	63•0	68.0 31.6	1.	+IRN01090 *	2.	 #CLN0219D 14. *	≠PAK02108 15.	i
33	AUS0004D	98.0	15	121.8	-24.9	3.6 1.9	54.	35.9	2	534.	63•2	124.5 -15.5	1.	*PHL0285∆ *	2.	*INS0036D 13. *	≭ AUS0006D 15. *	~!
34	AUS0009D	128.0	15	147•2	-32.0	2.1 1.4	15.	39.6	1	292.	64.2	141.0 -34.0	5.	*AUS0007D *	6.	*AUS0008D 13.	*AUS0D040 13. *	
35	AZK0134D	-31.0	15	-23.4	36.1	2.6 0.7	158.	41.7	2	140.	63-2	-16.9 32.5	-5.	*POR 01330 -	4.	*LBRJ244D 9.	≠CPV03010 14. *	1
36	BGD0220A	74.0	15	90.3	23.6	1.5 0.8	135.	43.4	1	107.	63.7	92.5 21.5	-2.	*IND0047D - *	·2•	*VTN0325D 12.	*IN000370 15. *	ŕ
37	CHNJ158A	83.0	15	111.8	38.0	2.6 1.7	124.	37.7	1	520.	64.9	115.3 31.5	3.	*J 0111H *	7.	*CHN 01744 8.	*CHN0181A 14. ÷	i k
38	CHNO174A	92.0	15	118.1	25.9	1.0 3.8	82.	44.9	2	82.	64.1	117.3 27.8	-2.	*CHN0158A	1.	≠J)111H 4.	*CHNU1724 6.	

- 19 -

suite — cont.

7

Canal 15 suite -- Channel 15 cont. -- Canal 15 cont.

	1	2	2	3	4		5		6	7	8	9 a	9 b	10	· · ·	11	12	<u> </u>	13	14
39	9'CDM020	7D 2	9.0	15'	44.1	-12.1	0.8	0.6'	149•	47.7	2	36.	63.3	43.1	-11.3	5.	#SOM0312D	9.	≠ZMB03140 10.	*M&JJ243) 16.
4(0 CAH023	30 -1	9•0	15	2.2	9.5	1.4	0.7	97.	44.3	2	81.	63•4	2.2	6.2	3.	*TGD0226D	5.	*ZAI0323D 14.	*ZAI03220 14.
4	1 GAB026	DD -1	3.0	15	11.8	-0.6	1.4	1.1	64.	42.2	1	135.	o3∙2	11.0	-4.0	3.	*ZA10323D	5.	*ZAI0322D 9.	*TCD0143) 14.
42	2 GMB030	20 -3	7.0	15 ·	-15.1	13.4	0.8	0.6	4.	47.5	2	40.	63.5	-14.2	13.3	3.	*GNP 0304D	7.	*#L10328D 9.	*LBF.0244D 12.
43	3 GRC010	5D :	5.0	15	24.7	38.2	1.8	1.3	156.	41.8	1	146.	63.5	26.6	41.5	1.	≁ #8UL00200 *	4.	*R090136D 6.	*IRN0109D 13.
44	4 IND004	3D 5	6.0	15	77.8	11.1	1.4	1.3	172.	41.8	1	147.	63.5	79.5	9.1	4.	*CLN0219D	5.	*IND0045D 15.	*IN000400 15.
4	5 IND004	70 6	8.0	15	93.3	11.1	1.9	0.6	96.	43.6	1	99•	63.6	93.8	14.8	2.	*BGD02204	3.	*1ND0048D 14.	*IND00379 16.
48	5 INS003	60 10	4.0	15	135.2	-3.8	2.5	2.9	147.	37.3	1	457.	63.9	128.0	-7.0	4.	+ +AUS0004D *	5.	*J 0111H 15.	*PNG01310 17.
47	7 IRN010	9D 3	4.0	15	54.2	32.4	3.8	1.8	149.	35.8	2	517.	63.0	61.5	31.0	7.	*AFG0245D	8.	*SOM0312D 16.	*ARS02750 26.
48	8 J 911	1H 11	0.0	15	134.5	31.5	3.5	3.3	68.	33.6	1	1190.	64.4	123.7	24.3	5.	*CHN3174A	8.	<pre>*PNG0131D 12.</pre>	*CHN0158A 12.
49	9 LBN027	9D 1	1.0	15	35.8	33.9	0.6	0.6	0.	48.7	2	20.	61.8	36.8	34.5	-0.	+GRC01050	6.	*IRN0109D 6.	*ARS00030 7.
5	0 LBR024	4D -3	1.0	15	-9.3	6.6	1.2	0.7	133.	44.9	1	70.	63.4	-10.2	8.5	4.	*AZR 0134D	9.	*MLI0328D 9.	*POR0133D 12.
5	1 LBY032	1D -2	5.0	15	13.1	27.2	2•4	1.1	129.	40.0	2	208.	63.2	11.4	33.5	3.	*ALG0251D	6.	*ALG0252D 9.	*TG002260 12.
5	2 LIE025	30 -3	7.0 -	15	9.5	47.1	0.6	0.6	0.	48.7	1	24.	62.6	9.5	47.2	1.	*TCH0144D	4.	*LUX0114D 8.	*AND03410 13-
5	3 LUX011	40 -1	9.0	15	6.0	49.8	0.6	ე.6	0.	48.7	1	28.	63.1	7.0	48.5	-4.	*LIE0253D	1.	*D 0087D 3.	*TCH0144∂ 4. *
5	4 MRA033	20 12	2.0	15	145.9	16.9	1.2	9.6	76.	45•7	1	53.	63.6	145.0	20.0	-4.	*J 0111H *	-3.	*GUM0331D 12.	*AUS0009D 19.
5	5 NHB012	8D 14	0.0	15	168.0	-16.4	1.5	0.7	87.	44.1	2	78.	63.0	169.8	-20.0	7.	*NCL 01 000 *	8.	*INS00360 23.	*NRU0309D 23. *
5	6 NRUD30	90 13	4.0	15	167.0	-0.5	0.6	J•6	0.	48.7	2	25.	62.7	166.0	-2.0	10.	*NHB01280 *	14.	*INSD036D 17. *	*PNG0131D 19. *
5	7 POR013	3D -3	1.0	15	-8.0	39 .6	0.9	0.6	112.	46.8	2	47.	63.6	-7.5	37.2	-15.	*AZR01340- *	15.	*LBR0244D 11.	*G 0027D 14.
5	8 SM0005	70 15	8.0	15-	172.3	-13.7	3.6	0.6	0.	48.7	1	33.	63.8	-171.0	-14.1	7.	*CKN0053D *	9.	*CKH0052D 14.	*₩AL0102D 20. *
5	9 SNG015	10 7	4.0	15	103.8	1.3	3.6	3.6	0.	48.7	2	32.	63.7	106.0	1.1	J.	*VTN03250 *	4.	*86002204 4. *	*MLA0227A 12. *
6	0 SOMO31	20 2	3.0	15	. 45 •0	6•4	3.3	1.5	71.	37.2	1	336.	62.5	43•2	11.2	5.	*URS00600 *	8.	*URS0061D 10. *	≭¥E402660 15. *
6	1 TCH014	4D -	1.0	15	17.3	49.3	1.5	0.6	170.	44 •8	2	83.	64• Q	12.1	50.3	-2	*L150253D *	з.	*ZMB0314D 5. *	*LUX9114D 6. *
6	2 UGA005	10 1	1.0	15	32.3	1.2	1.5	1.1	60.	42.1	1	134.	63.4	29.8	-1.3	2.	*RR₩03105 ≠	4.	*ZMB0314D 8. *	*ZAT0322D 16. *
6	3 UR\$006	1D 2	3.0	15	24.7	56.6	0.9	0.6	12.	46.7	2	70.	65.2	25.8	54.1	-0.	*TCH0144D *	6.	¥\$340312D 6. ≭	≉URS00600 6. *
6	4 VTN032	50 8	6.0	15	105.3	16.1	3.9	1.4	116.	38•0	2	365.	63.6	108.0	21.6	0.	*CHN0181A *	3.	*CHN0158A 4. *	*J 0111H 13. *
0	5 ZMB031	4D -	1.0	15	27.5	-13.1	2.4	1.5	39.	38.8	1	3?4•	63.9	33.0	-13.8	3.	*M0Z0307D	5.	*TCHJ144D 11.	*30702970 13.

-- 20 ---

13

• •

Canal 16 — Channel 16 — Canal 16

,

	1	2	3	4		5	6	7	8	9 a	9b	10		11	. 12	13	14
									ΓΤ								
35	ALG0252D	-25.J	16	1.6	25.5	3.6 2.2	152.	35.3	1	583.	63 . 0	9.5	30.0	4.	≭LB¥0321D 7 *	• *LBY02801 12.	*HLF01470 15.
36	AND03410	-37.0	16	1.6	42.5).6 0.6	0.	48.7	2	20.	61.6	1.4	42.4	-1.	*G 0027D 3	• *ALG0252D 5.	*MLI03280 7.
37	AR\$00030	17.0	16	41.1	23.8	3.5 1.7	134.	36.5	2	432.	62.9	56.0	21.0	-1.		• *3GYJJ26D 9.	*QATJ2475 10.
38	3 AUS0007D	128.0	16	145.0	-38.1	1.8 1.4	134.	40.2	2	213.	63•5	149.6	-37.5	6.	*AUS00090 7	• * WZL 02878 14.	*DCE01010 27.
39	AUT0016D	-19+0	16	12•1	47.5	1.1 0.6	166.	45.7	2	73.	64.3	17.1	48•D	٥.	* ¥ZAI0322D 4	• *F 0093E 10.	*URS00600 12.
4:	BULDOZOD	-1.0	16	25.0	43.0	1.0 0.6	165.	46.3	1	57.	63.8	28.1	42.0	-2.	* *M0203070. 3	* • *TUR 0145E 5•	* *GRC01050 6.
41	CHN01694	92.0	16	118.5	36.4	1.2 0.8	11.	44.8	1	98.	64.7	122.8	38.4	-1.	# *KRΞЭ286Β 2	* *CHN0167A 8.	* *CHN0186A 8.
42	CHN0186A	62.0	16	102.5	30.2	1.9 1.2	147.	40.5	2	311.	ئ5 •5	110.0	31.8	7.	* *CHN0169A 9	* • *CHN0158A 17•	* *IN000480 17.
43	CKN00530	158+0	16-	-163.0	-11.2	1.8 3.7	30.	43.2	2	133.	64.5	-158.0	-9.0	4.	* *60501010 4	* • *TON0215D 29•	∓ *SM00057D 31.
44	CPV03010	-31.0	16	-24.0	16.0	0.9 0.7	144.	40.5	2	39.	62•4	-24.3	14.4	4.	* ≢G 00270 €	• *AZR0134D 14•	*MLI0328D 16.
4	5 DNK20898	5.0	16	12.3	· 57•1	1.2 3.6	177.	45.7	2	74.	64.4	13.0	54.5	1.	≠ ≠G 0027D 5	* *POL01325 8.	*AUT00160 8.
4	5 EGY00260	-7.0	16	29.7	26.8	2.3 1.7	136.	38•2	2	319.	63.3	36.0	23.3	-0.	*ARS000000 (• *ALG0252D 12.	* *HOZJ3070 16.
4	7 G 00270	-31.0	16	-3.5	53.8	1.8 0.7	142.	43•J	1	164.	65.2	1.3	51.1	3.	* *ALG0252D 6	• *AUT 0016D 10.	* *DNK00898 13.
48	3 IND00400	56.0	16	73.0	25 . J	1.8 1.5	58.	39.9	2	244.	63.8	78.2	27.9	2.	* *NPL0122A (* • *1ND0048D 7•	*IND0038A 11.
49	9 IND00480	68.0	16	86.2	25.0	1.6 0.9	120.	42•8	2	138.	65 •5	89•8	26.7	-6.	*CHN0186A -0	• #BRM0298A 9.	*IND0046A 1J.
5	0 KRE0286B	115.0	16	127.1	40.1	1.1 0.8	31.	45.2	2	76.	64.0	124.)	39.9	-1.	* *CHN0169A :	• *J 0111H 8.	* *CHN0186A 9.
5	L MAUD243D	29.0	16	56.8	-13.9	1.6 1.4	65.	43.9	1	197.	63.9	53.J	-15.0	6.	* *MOGO236E	. *M∂Z0307D 11.	* *COM0207D 18.
5	2 MLA0227A	86.0	16	102.1	4.1	1.6 0.8	135.	4 3•0	1	105.	63•2	105.4	2.2	7.	* *VTN03250 12	• *SNG0151D 13•	* *INS0032A 13.
5	3 MLD0306B	44.0	16	73.1	6.0	1.0 0.6	90.	46.6	1	51.	63.7	71.0	7.5	5•	* *JRS0069B (* • *IND0043D 17.	* *AR\$00030 19•
54	4 ML 103280	-37.0	16	-7.6	13.2	1.7 1.2	171.	40.9	1	198.	63.9	-12.0	15.0	з.	¥ ≠GMB0302D	≠GUI01925 8.	* *ALG0252D 9.
5	5 MLT0147D	-13.0	16	14.3	35.9	0.6 0.6	0.	48.7	1	18.	61.2	14.3	35.9	0.	* *ALG02520 2	* *EGY0026D 11.	* *CME0300E 12.
50	5 MOZO307D	-1.0	16	34.0	-18.0	3.6 1.4	55.	37.3	2	505.	64.4	37.3	-15.0	5.	* *ZMB03140	* • *BUL0020D 13•	* *ZAI03220 17.
 5	/ UC€01010	-160.0	16	-145.0	-16.3	4.3 3.5	4.	32•4	2	1352.	63.7	-154.7	-15.7	17.	* ≠CKN00530 1	* *TUN 32150 33.	≠ ¥40606070 35.
5	6 PHL0285A	98.0	16	121.3	11.1	3.5 1.8	99.	36.4	2	531.	63.7	122.0	21.0	7.	*AUS00055 1	• *AUS3004D 13•	*J 01114 15.
.5	9 RRW0310D	11.0	16	30.0	-2.1	0.7 0.6	42.	48.3	2	47.	65.0	30.3	-1.0	0.	* *ZAI0322D		* ≑YMS02678 14.
- 6'	D STP3241D	-13.0	16	7.0	0.8	0.6 0.6	0.	48.7	2	20.	61.6	5.0	2.0	-2.	≠ *CME0300€ .	± :• ≠ZAī)322D 3•	* *GAB02600 б.

suite – cont.

•

١,

__ 21 __

Canal 16 suite — Channel 16 cont. — Canal 16 cont.

	1	2	3	4		5	6	7	8	9 a	9 b	10		11	12	13	14
61	13832150	170-0	16	-174.7	-18-0	1.4 0.7	85-	44.4		30.	63-5	-173.7 -1	15.9	. 4.	+CKN00530 6-	±0C+01010 13-	*SM403355 15-
i	1.51 521.50	1, 38 3	10	11401	10.0	101 007	020		•	500	0.2.0.2	11517 1			*	*	*
62	URS0060D	23.0	16	41.5	57.4	3.1 1.6	153.	37.4	1	887.	66•9	27.7 6	50.5	5.	*UPS00610 8. *	*UNK0089B 11. *	*SCM03120 19.
63	UR300698	44.)	16	73.8	38.5	1.4 3.7	161.	44.2	2	98.	64.1	75.1 3	37.3	4.	*IND00400 6. *	*ML003068 11. *	*IND0038A 14. *
54	ZA10322D	-19.0	16	22•4	3.0	2.2 1.9	48.	38.2	1	468.	64•9	19.5	5.0	4.	*CME03008 8.	*ALG32520 9.	*AUT0016D 12.

Canal 17 — Channel 17 — Canal 17

.

	1	2	3	4		5	6	7	8	9 a	9 b	10		11	12	13	14
31	 AUS0005E	98•0	17	133.5	-18.8	2.7 1.4	76.	38.5	2	399.	64.5	130.8	-12.5	-2.	*PHL0285B	. *PHL0285A 1	• *INS0032A 12.
32	BRM0298A	74.0	17	97.1	19.1	3.6 1.5	104.	37.0	2	488.	63•9	92.4	21.4	-0.	*1ND0046A	• *BGD0220B 6	• *IND0048D 10.
33	CAR0338E	122.0	17	149.5	8.0	5.4 9.8	178.	38.1	1	289.	62.7	134.6	7.5	11.	* *INS0032A 12 *	* • *NZL0287B 22 *	* • *AUS0005E 25.
34	- CHN0167A	92.0	17	124.3	43.7	2.0 0.7	156.	42.7	2	159.	64.7	121.0	42.1	7.	*CHN0182A 1	• *CHN0169A 11	• #CHN0159A 21.
35	CHN0182A	83.0	17	108.7	35.1	1.4 0.9	109.	43.3	1	122.	64.2	105.7	32.9	-1.	* *INS0032A 2	* *BRM0298A 4	* +CHN0159A 7.
36	S CME0300E	-13.0	17	12.7	6.2	2.5 1.7	87.	37.9	.1	371.	63•6	16.2	1.4	2.	*ZA10322D 4	• *Z AI0323ë 8	• *TCD0143E 9.
37	F 00935	-19.0	17	2.6	45.9	2.5 1.0	169.	40.4	1	232.	64.)	9.5	41.2	-2•	*SMR0311F 1	• *D 0087E 9	• *4LG02515 9•
38	GUI0192E	-37.0	17	-11.0	10.2	1.6 1.3	147.	42.1	2	143.	63.7	-14.5	11.5	3.	*GNP0304E 5	• *MLIJ328D 8	• *MLI0327E 15.
39	1ND0038A	56.0	17	75.9	33.4	1.5 1.1	33.	42.1	1	166.	64.3	79.0	31.2	-3.	*NPL01224 -3	• *IND0040D 7	• *IND0042A 12.
40	INDOO46A	68.0	17	84•7	20.5	1.6 0.9	30.	42.9	1	118.	63 •6	87.5	21.7	1.	*IND00480 (● *BRM0298A 6	• *BGD0220B 8.
41	INS0032A	80.0	17	112.3	-0.3	2.7 2.3	109.	36.3	2	587.	64.0	109.1	1.8	4.	*BRM0298A (• *CHN01824 9 *	• *CHN0159A 14•
42	LBY02833	-25.0	17	21.4	26.0	2.5 1.9	119.	49.1	2	229.	63.7	22 . Ŭ	33.0	4.	*TUR0145E 8 *	• *ALG0251E 11 *	*ALG02520 12.
43	MDG0236E	29•0	17	46.6	-18.8	2.7 1.1	65.	39.3	2	263.	63.5	49.0	-12.3	8.	*MAUJ2430 10	• *MAU)242E 15 *	*ZAI03235 22.
44	NPL0122A	50.0	17	83.7	28.3	1.7 0.6	163.	44.1	2	112.	64•6	88.0	26.5	3.	*IND0046A 7 *	• *BRM0298A 8 *	• *IND00414 15. *
45	NZL J 287B	128.0	17	173.0	-41.0	3.3 1.3	48.	38.0	1	492.	64•8	166.3	-45.5	12.	*AUS00068 16	• *AUS00070 16 *	*4US00055 22.
46	PLM0337E	170.0	17	-161.4	7.0	0.6 0.6	0.	48.7	1	25.	62•6	-162.4	6.1	6.	*CAR0338E 7	• *T0NJ2150 14 *	• *SMAJ335E 10. *
47	P0L01325	-1•0	17	19.3	51.8	1.5 0.6	162.	44.5	2	95.	64.3	22.9	49.0	-0.	*SMR03116 5	• *TUR01455 7	. *ROU0136≟ 7.
48	JATJ247E	17.0	17	51.1	25.3	0.6 0.6	0.	48.7	1	22.	62.0	52.3	24.8	1.	*YMS02678 5 *	• *ARSJ2752 6 *	* *ARS00030 9.
49	SMA03355	170.0	17	-170.1	-14.2	0.6 0.6	0.	48.7	2	19.	61.4	-171.0	-11.0	-2.	*TBN02150 1 *	• *PLM03372 2	*CKNU053D 5.
150	SMR0311E	-37.0	17	12.6	43.7	0.6 0.6	0.	48.7	1	25.	62.7	12.0	43.0	-0.	* [€] 0093F 2	• *GUI0192E 9	• *TUR0145E 12•

)

٠

)

suite — cont.

Canal 17 suite — Channel 17 cont. — Canal 17 cont.

.

1

	1	2	3	4	5	6	7	8	9 a	9b	10		11	12	13	14
5	1 SWZ0313E	-1.0	1 17	31.5 -26.5	0.6 0.6	66.	48.5	1	1 28.	63.0	31.1 -	-25•9	4.	*MOZ0307D 7 *	*BOT0297E 9.	*POL01325 11.
5	2 TUR0145E	5.0	17	34.4 38.9	2.7 1.0	168.	39.8	1	259.	63.9	42.7	41.5	-2.	*URS0064E −1 *	*LBY0280E 11.	*URS0060D 18. *
5	3 URS3064E	23.0	17	45.6 40.8	2.2 0.6	163.	43.1	2	126.	64.1	40.0	43.4	-0.	*TUR0145E 2 *	*URS00600 6.	*LBY02805 12.
5	4 WAK0334E	140.0	17	166.5 19.2	0.6 0.6	0.	48.7	1	32.	63.8	166.5	19.2	22.	*MRLJ333E 24 *	*NZL9287B 33.	*AUS0005E 31. *
5	5 YMS0267E	11.0	17	48.8 15.2	1.8 1.5	176.	39.9	2	205.	63.0	49.5	18.4	3.	*ARS0275E 6	*ARS0003D 7.	*YEM02668 13.

...

Canal 18 --- Channel 18 --- Canal 18

		1	2	3	4	·	5	6	7	8	9 a	9 b	10		11	12		13		14	
2	26	ALG0251E	-25.0	18	4.2	33.2	2.4 1.3	172.	39.4	1	265.	63.6	9.5	30.0	0.	*TCD0143E	5.	*LBY0321E	7.	*TG00226E	8.
12	27	AR \$0275E	17.0	18	48.3	24.6	3.8 1.4	138.	36.9	2	400.	63.0	42.5	17.0	-11.	+ +YEM0266E- *	11.	* *YMS0267E *	-1.	+ *TCD0143∂ *	12.
12	28	AUS0006E	98.0	18	135.4	-30.3	2.0 1.4	44.	39.8	1	232.	63.4	140.8	-27.7	-3.	*PHL02858	-0.	*AUS0008E	3.	*AUS0005년 *	8.
Ĩ	29	AUS0008E	128.0	18	145.9	-21.5	2.9 2.0	120.	36.6	2	537.	63.9	149.0	-29.0	3.	*AUS0009E	6.	*AUS0006E *	6.	*NZL0287B	17.
	30	BGD0220B	74•0	18	90.3	23.6	1.5 0.8	135.	43.4	1	108.	63.7	88.0	27.0	-2.	¥IND0042A *	1.	*CHN0185A *	7.	*IND0041A *	10.
	31	80T0297E	-1.0	18	23•3	-22.2	2.1 1.5	36.	39.2	2	300.	64.0	25.3	-17.8	4.	*ZMB0314E	6.	*ZAI 0323E	10.	*R0U0136E *	11.
	32	CBG9299A	68.0	18	105.0	12.7	1.0 0.9	110.	44.7	1	91.	64.3	103.5	10.5	2.	*IND0042A	4.	*MLA02278	13.	≭IND0046 B ≠	14.
	33	CHN0159A	80.0	18	139.4	27.3	2.1 1.7	107.	38.6	2	393.	64.5	109.5	33.1	-1.	*INS0030A	5.	*CHN01588 ★	6.	*CHN0182A	6.
	34	CHN0185A	62.0	18	95•7	35•4	2.1 1.1	156.	40.5	1	198.	63.4	101.3	33.1	4.	*CHN0159A *	5.	*BRM02988.	17.	*BRM0298A *	17.
	35	D 0087E	-19.0	18	9.6	49.9	1.6 0.7	147.	43.6	2	164.	65.7	10.0	54.8	-0.	*NOR 01205	3.	*ZAI0323E *	5.	*LUX0114E *	11.
	36	GNPJ304F	-31.0	18	-15.0	12.0	0.9 0.6	172.	46.9	2	44.	63.3	-14.2	12.3	3.	*GMB0302E *	7.	*GUI0192E *	7.	*IRL02115 *	14.
	37	GUM0331E	122.0	18	144.5	13.1	0.6 3.6	0.	48.7	2	31.	63.5	144.7	13.4	8.	*MRA0332E *	9.	*CAR0338E *	17.	≠AUS0008E *	20.
:	38	IND0041A	56.0	18	78.4	16.0	2.1 1.4	35.	39.7	2	258.	63.8	84.9	19.0	2.	‡1ND0042A	3.	*IND0046B	12.	*IND0046A	12.
.	39	IND0042A	68.0	18	79.3	27.7	2.1 1.2	147.	40.3	2	223.	63.8	84.6	25.8	-2.	*BGD02208	2.	*IND 0041A *	4.	*C8G0299A *	15.
1	40	1NS0030A	80.0	18	112.3	-8.1	3.1 1.5	169.	37•6	1	450.	64•2	114.0	-5-8	1.	*CHN0159A ★	4.	≭INS0032B ★	7.	*INS0032A *	7.
	41	IRL9211E	-31.0	18	-8.2	53.2	0.8 0.6	162.	47.2	1	53.	64.5	-5.5	54.3	5.	*NOR 0120B	9.	*D 00875	10.	*GNP0304E	15.
	42	KREJ286C	110.0	18	127.1	40.1	1.1 0.8	31.	45.2	2	77.	64.0	124.0	39.9	4.	*CHN0159A *	7.	*PHL0285B *	13.	*CHN0185A *	13.
	4 3	MAU0242E	29.0	18	59.8	-18.9	1.6 1.2	55.	41.2	1	198.	64.2	60.0	-14.0	5.	*ZAI0323E *	6.	*MDG0236E *	14.	*ARS02755 *	25.
	44	ML A02278	86.0	18	102.1	4.1	1.6 0.8	135.	43.0	1	106.	63.3	103.8	1.2	-1.	*INS0030A	-1.	*INS0032B	15.	*INS00324	15.

• ;

٠,

Canal 18 suite — Channel 18 cont. — Canal 18 cont.

	1	2	3	4	I	5	6	7	8	9 a	9 b	10		11	12	13	14
														•			
4	5 MLI0327E	-37.0	18	-2.0	19.0	2.7 1.3	127.	39.0	1	348.	64.4	-2.5	23.3	9.	*ALG0251E 11.	*SMR0311E 18.	*LIE02538 20.
4	6 MRL0333E	146.0	18	166.7	7.9	1.5 1.5	177.	40.7	1	190.	63 .5	162.1	11.5	16.	*CAR0338E 19.	*AUS0008E 25.	*PHL0285B 25.
4	7 NOR01208	5.0	18	13.1	64.1	1.8 0.9	10.	42.2	2	193.	65.0	7.6	58.0	1.	*D 0087E 2. *	*IRL0211E 14. *	*POL01325 14. *
4	8 PAK0281A	38.0	18	65.2	. 27.9	1.5 1.4	28.	40.9	1	162.	63.0	69.5	32.2	2.	*IND0042A 4.	*URS0070A 10. *	*IRN01095 16. *
4	9 PHL02858	98.0	18	121.3	11.1	3.5 1.8	99.	36.4	2	5 34•	63 .7	122.0	21.0	3.	*AUS0006E 6. *	*CHN0159A 1). *	*AUS0005E 11. *
5	0 ROU0136E	-1.0	18	25.0	45.7	1.4 0.7	155.	44.7	1	86.	64 • 0	20.2	46.1	-1.	*D 9087E 4. *	*BDT0297E 5. *	*TCH0144E 7. *
5	1 TCD0143E	-13.0	18	18.1	15.5	3.4 1.7	107.	36.6	2	576.	64.2	16.5	8.2	5.	*CME0300E 7. *	*MLI0327E 14. *	*GAB0260E 15. ≠
5	2 TG00226E	-25.0	18	0.8	8.6	1.5 0.6	105.	44.6	2	79.	63.6	- 0•2	11.1	-2.	*MLI0327E -0. *	*ALG0251E 6. *	*DAH0233E 8. *
5	3 UR\$0070A	44.0	18	73.9	41.0	1.3 0.8	5.	43.7	2	129.	64.5	73.8	38.4	4.	*PAK0281A 7.	*IND0042A 9. *	*IND00388 17. *
5	4 YEM0266E	11.0	18	44.3	15.1	1.1 0.7	109.	45.2	1	58• ⁻	62.8	44.0	18.0	-1.	*ARSJ275E 0. *	*YMS0267E 7. *	*UGA0051E 11. *
5	5 ZAI0323E	-19.0	18	21.3	-6.8	2.8 1.5	149.	38.0	1	490.	64.9	16.3	-1.0	1.	*TCD0143E 5.	*GAB0260E 8.	*D 0087E 8.

Canal 19 --- Channel 19 --- Canal 19

•

· · ·

	1	2	3	4		5	6	7	8	9 a	9h	10	T	11	12		13		14	
						J			Ħ						·				1	
31	AUS0004E	98.0	19	121.8 -2	4.9	3.6 1.9	54.	35.9	2	541.	63.2	124.5 -	15.5	-2•	*PHL0285C	2.	*PHL0235B *	2.	*INS00328	10.
32	AUSOCO9E	128.0	19	147.2 -3	2.0	2.1 1.4	15.	39.6	1	296.	64.3	141.0 -3	34.0	5.	*AUS0007E *	6.	*AUS0008E *	13.	*AUS0004일 *	13.
33	AZR0134E	-31.0	19	-23.4 3	6.1	2.6 0.7	158.	41.7	2	142.	63.2	-16.9	32.5	-5.	*?OR0133E *	-4.	*CPV0301E *	14.	*GNP0304E *	17.
34	BR M0298B	74.0	19	97.1 1	9.1	3.6 1.5	104.	37.0	2	491.	63.9	92•4	21.4	-1.	*IND0046B *	3.	*BGDJ220C *	6.	*BGD02208 *	6.
35	CHN0158B	80.0	19	111.8 3	8.0	2.6 1.7	124.	37.7	1	527.	64.9	115.3	31.5	-1.	*1NS0032B *	3.	*CHN0159B ≠	6.	*CHN0159A *	7.
36	CHN0179A	92.0	19	112.2 2	1.9	1.8 1.2	37.	40.7	2	204.	63.8	112.0	24.7	1.•	*CHN0158B *	2.	*CHN0159B *	13.	*CHN0159A *	13.
37	DAH0233E	-19.0	19	2.2	9.5	1.4 0.7	97.	44.3	2	82.	63•5	2.2	6.2	3.	≠TGD0226Ε ≭	5.	*ZA10323E *	14.	*ZAI03223 *	14.
38	GAB0260E	-13.0	19	11.8 -	0.6	1.4 1.1	64.	42•2	1	136.	63.6	11.0	-4.0	3.	*ZAI0323E *	6.	*ZAI0322E *	9.	*TCD0143E *	14.
39	GMB0302E	-37.0	19	-15.1 1	3.4	0.8 0.6	4.	47.5	2	40.	63.5	-14.2	13.3	4.	≠GNP 3304E *	7.	*ML10328E *	9.	*MLI0327⊆ *	16.
40	GRC0105E	5.0	19	24.7 3	8.2	1.8 1.0	156.	41.8	1	147.	63.5	26.6	41.5	1.	*BUL0020E	4.	*R0U0136E	6.	*IRN0109E	13.

- 24 --

)

)

— cont.

suite
Can	uai 19	suite —	Cha	annel 1	9 coi	nt. —	Canal 1	9 cor	nt.					• .			
	1	2	3	4		5	6	7	8	9 a	9ь	10)	11	12	13	14
41	I ND0038	B 56.0	19	75.9	33.4	1.5 1.	1 33.	42.1	1	167.	64.3	79.0	31.2	-3.	*NPL01228 -3.	*IRN0109E 9.	*IND00428 12.
42	IND0046	B 68.0	19	84.7	20.5	1.6 0.	9 30.	42.9	1	119.	63.6	87.5	21.7	1.	*BRM0298B 6.	*BGD0220C 8.	*BG002203 8.
43	INS0032	B 80.0	19	112.3	-0.3	2.7 2.	3 109.	36.3	2	591.	64.1	117.5	3.7	0.	*CHN0158B 2.	*CHN0159B 12.	*CHN0159A 12.
44	IN50036	E 104.0	19	135.2	-3.8	2.5 2.	0 147.	37.3	1	463.	64.0	128.0	-7.0	1.	*AUS0004E 5.	*IN\$90328 5.	*INS00308 18.
45	IRN0109	E 34•0	19	54.2	32.4	3.8 1.	8 149.	35.8	2	523.	63.0	60.2	25.2	7.	*SOM0312E 10.	*PAK0281A 14.	*URS0066A 18. *
46	LBN02791	E 11.0	19	35.8	33.9	0.6 0.	.6 0.	48.7	2	21.	61.8	36.8	34.5	-0.	*GRC0105E 6. *	*IRN0109E 6.	*UGA0051E 7. *
47	LBY0321	E -25.0	19	13.1	27.2	2.4 1.	1 129.	40.0	2	210.	63.3	11.4	33.5	3.	*ALG0251E 6.	*ALG0252E 9. *	*TG00226E 12.
48	LIE0253	E -37.0	19	9.5	47.1	0.6 0.	.6 0.	48.7	1	25.	62.6	9.5	47.2	1.	*TCH0144E 4.	*LUX0114E 8. *	*AND0341E 13. *
49	LUX0114	E -19.0	19	6.0	49.8	0.6 0.	.6 0.	48.7	1	28.	63.1	7.0	48.5	-4.	*LIE0253E 1. *	*D 0087E 3. *	*TCH0144E 4. *
50	MRA0332	E 122.0	19	145.9	16.9	1.2 0.	.6 76.	45.7	1	63.	63•7	146.9	15.1	⊧ 7.	*GUM0331E 7. *	*AUS0009E 21. *	*URS0077A 24. *
51	NI 40054	A 158.0	19-	-169.8	-19.0	0.6 0.	6 0.	48.7	2	34.	64.1	-169.9	-19.0	27.	*TKL0058A 31. *	*AUS0009E 31. *	*URS0077A 37. *
52	NPL0122	B 50.0	19	83.7	28.3	1.7 0.	6 163	44.1	2	112.	64.6	88.0	26.5	2.	*IND00468 7. *	*BRM02988 8.	*IND00418 15. *
53	POR0133	E -31.0	19	-8.0	39.6	0.9 0.	6 112	46.8	2	48.	63.6	-7.5	37.2	-15.	*AZR0134E-15. *	*G 0027E 14. *	*LBY03215 18• *
54	SOM0312	E 23.0	19	45.0	6.4	3.3 1.	.5 71.	37.2	1	340.	62.6	42.0	-1.0	7.	*URS0061E 11.	*UGA0051E 13. *	*ZMB0314E 15. *
55	TCH0144	E -1.0	19	17.3	49.3	1.5 0.	6 170	44.8	2	84.	64.0	12.1	50.3	-2.	*LIE02536 3.	*ZMB0314E 5.	*LUX0114E 6.
56	UGA0051	E 11.0	19	32.3	1.2	1.5 1.	1 60	42.1	1	135.	63•4	29•8	-1.3	2.	*RRW0310E 4.	*ZMB0314E 8. *	*ZAI0322E 16.
57	URS0061	E 23.0	19	24.7	56.6	0.9 0.	6 12	46.7	2	70.	65.2	25.8	54.1	1.	*TCH0144E 6.	*SDM0312E 6.	*LIE0253E 7. *
58	UR \$0077	A 110.0	19	112.7	57.3	2.7 1.	8 2	. 37.6	1	450.	64.1	117.8	49.4	1.	*CHN0158B 1. *	*BRM0298B 19.	*INS0036E 19. *
59	ZMB0314	E -1.0	19	27.5	-13.1	2.4 1	5 39	38.8	1	329•	63.9	33.0	-13-8	3.	*M020307E 5.	*TCH0144E 11.	*BOT0297E 13.

• •

Canal 20 --- Channel 20 --- Canal 20

Γ	T	1	2	3	4		5	6	7	8	9 a	9 b	10	11	12	13	14
	30	ALG0252E	-25.0	20	1.6	25.5	3.6 2.2	152.	35.3	1	591.	63.0	-9.0 27.5	5.	 *MRC0209A 6. *	*G 0027E 16.	*ZAI0322E 17.
	31	AND0341E	-37.0	20	1.6	42.5	0.6 0.6	0.	48.7	2	20.	61.7	1.4 42.4	-1.	≭ G 0027E 3• *	*ALG0252E 5. *	*MLI0328E 7. *
	32	AR \$0003E	17.0	20	41.1	23.8	3.5 1.7	134.	36.5	2	437.	62.9	38.0 32.0	-0.	*:GY0026E 1. *	*URS0066A 9. ≠	*LBN0279E 9. *
ļ	33	AUS0007E	128.0	20	145.0	-38.1	1.8 1.4	134.	40•2	2	215.	63.5	149.6 -37.5	6.	*AUS0009E 7. *	*NZL0287C 14. *	*URS0079A 24.

Canal 20 suite — Channel 20 cont. — Canal 20 cont.

.

.

•

٠

	1	2	3	4	T	5	6	7	8	9 a	9 b	10		11	12		13		14]
34	AUT0016F	-19-0	20	12.1	47.5	 1.1 0.6	166.	45.7	2	73.	64-3	9.4	47.2	1.	+ZAI0322E	5.	*AND0341E	8.	*G 0027E	8.
35	86002200	74.0	20	90.3	23.6	1.5 0.8	135.	43.4	-	109.	63.7	88.0	27.0	-2.	* *IND0042B	1.	* *IND0041B	10.	* *URS0066A	12-
24	BUL 0020E	-1-0	20	25-0	43-0	1.0 0.6	165.	46.3	1	57.	63.9	23.0	41.3	-2.	* *M0Z0307E	3.	* *YUG0148A	4.	* *GRC0105F	
	00000200		20	2300	1300	100 000	1020		Ē		0.507	2000			*		*		*	
37	CBG02998	68.0	23	105.0	12.7	1.0 0.9	110.	44.7	1	92.	64.3	103.5	10.5	2.	*IND00428 *	4.	*MLA0227C	13.	*IND0046C *	14.
38	CHN01598	80.0	20	109.4	27.3	2.1 1.7	107.	38.6	2	396.	64.6	109.5	33•1	-2.	*CHNJ184A *	3.	*INS00308	5.	*CHN01588	6.
39	CHN0184A	62.0	20	101.0	37.9	2.8 0.8	144.	40.7	1	201.	63.7	105.8	32.9	0.	*CHN0159B	1.	≉URS0066A *	11.	*CHN01588	16.
40	CPV0301E	-31.0	20	-24.0	16.0	0.9 0.7	144.	46.5	2	39.	62.4	-22.9	16.8	4.	≭G 0027 F	7.	*AZR0134E	13.	*SEN0222A	15.
+1	DNK0089C	5.0	20	12.3	57.1	1.2 0.6	177.	45.7	2	75.	64.4	10.0	54.5	0.	*DDR 0216A	5.	*G 0027E	5.	*AUT00165	8.
42	EGY0026E	-7.0	20	29.7	26.8	2.3 1.7	136.	38.2	2	324.	63.3	36.0	23.3	-0.	* *ARS0003E	0.	*ALG0252E	12.	+ *MGZ0307E	16.
43	G 0027E	-31.0	20	-3.5	53.8	1.8 0.7	142.	43.0	1	166.	65.2	1.3	51.1	2.	*ALG0252E	6.	*AUT 3016E	10.	*BEL0018A	13.
44	IND0041B	56•0	20	78.4	16.0	2.1 1.4	35.	39.7	2	260.	63.8	84•9	19.0	2.	* IND0042B	3.	+ ≠IND0046C	12.	+ *IND0046B	12.
45	IND0042B	68.0	20	, 79.3	27.7	2•Ì 1•2	147.	40.3	2	224.	63.8	84.6	25.8	-2.	*BGD0220C	2.	* *IND0041B	4.	*CBG02998	10.
46	INS0030B	80.0	20	112.3	-8-1	3.1 1.5	169.	37.6	1	453.	64.2	114.0	-5.8	1.	*CHN01598	4.	*INS0032C	7.	*1NS0032B	7.
47	KREJ286D	110.0	20	127.1	40.1	1.1 0.8	31.	45.2	2	77.	64.1	130.0	43.0	-3.	+ *URS0079A -	-2.	- *CHN0159B	11.	+ *URS0077A	14.
48	NLA0227C	86.0	20	102.1	4.1	1.6 3.8	135.	43.0	1	107.	63.3	103.8	1.2	-1.	+ +INS0030B -	-1.	+ *1NS0032C	15.	+ *INS00328	15.
49	ML10328E	-37.0	20	-7.6	13.2	1.7 1.2	171.	40.9	1	201.	63 . 9	-12.0	15.0	2.	*SEN0222A	6.	*GM80392E	8.	*ALG02522	9.
50	M020307E	-1.0	20	34.0	-18.0	3.6 1.4	55.	37.3	2	512.	64•4	32.5	-26.8	4.	*AFS0021A *	6.	*BUL0020E	13.	*ZMB0314E	15.
51	PAK0282A	38.0	20	68.5	25•8	1.3 0.6	133.	45.1	1	66.	63.3	68.3	28.7	-1.	*URS0066A -	-0.	*IND0042B	7.	*EGY0026E	15.
52	PHL0285C	98.0	20	121.3	11.1	3.5 1.8	99.	36.4	2	538.	63.7	122.0	21.0	2.	*CHN0175A	5.	*CHN0159B	10.	*AUS0005F	11.
53	RRW0310E	11.0	20	30.0	-2.1	0.7 0.6	42.	48.3	2	47.	65.0	30.3	-1.9	0.	*ZAI 3322E	2.	*UGA0051E	8.	*KEN0249A	11.
54	STP0241E	-13.0	20	7.0	0.8	0.6 0.6	0.	48.7	2	20.	61.7	5.0	2.3	-0.	*ZAI03225	3.	*GAB0267E	6.	*DAH0233E	9.
55	TKL0058A	158.0	20	-171.8	-8.9	0.7 0.6	35.	48.0	1	38.	63.8	-171.2	-9.5	26.	*NIU0054A :	29.	*URS0079A *	30.	*AUS0007E	35.
56	UR\$0065A	23.0	20	32.4	63.1	1.2 0.6	175.	45.7	1	121.	66.6	29.5	66.6	4.	*URS0066A	9.	*G 0027E	12.	*UR \$0061E	12.
57	UR \$0066A	44.0	20	64.3	44.6	4.6 2.5	169.	33.7	2	1466.	65.4	71.6	37.3	7.	*IND00428 :	10.	*EGY0026E	16.	*IRN0109E	19.
58	UR S0 079A	140.0	20	138.0	53•6	3.2 2.1	62.	36.0	2	1497.	67.7	130.6	42.2	3.	*KRE0286D	3.	*CHN0159B	15.	*CHN0184A	20.
59	ZA10322E	-19.0	23	22.4	0.0	2.2 1.9	48.	38.2	1	474.	64.9	19.5	5.0	6.	*ALG0252E	9.	*AUT0016E	12.	*GAB0260E	16.

- 26 ---

· *

Canal 21 — Channel 21 — Canal 21

	1	2	3	4		5	6	7	8	9 a	9 b	10		11	12		13		14	
31	AFI0099A	23.0	21	42.5	11.6	0.6 0.6	0.	48.7	1	24.	62.5	41.9	10.8	-0.	*KEN0249A	2.	*ETH0092A	6.	*BLRJ062A	10.
32	AFS0021A	5.0	21	24.5	-28.0	3.1 1.7	27.	37.0	2	505.	64.1	30.0	-22.1	1.	* *M0Z0307E	4.	* *RHS0135A	5.	* *C YP 0086A	15.
33	AUS0005F	98.0	21	133.5	-18.8	2.7 1.4	76.	38.5	2	404.	64.5	130.8	-12.5	-2.	* *PHL0285D	1.	* *PHL0285C	1.	* *INS0032C	12.
34	BELOO18A	-19.0	21	4.6	50.6	0.8 0.6	167.	47.3	1	49.	64.2	6.4	50.3	0.	* *DDR 3216A	5.	* *YUG0148A	6.	* *SUI0140A	10.
35	BLR0062A	23.0	21	27.8	52.6	1.1 0.7	1.	45.3	2	89.	64.8	23.4	51.5	3.	* *YUG0148A	7.	* *MCO0116A	11.	* *URS0065A	13.
36	BRM0298C	74.0	21	97.1	19.1	3.6 1.5	104.	37.0	2	494.	63.9	92.4	21.4	-1.	* *IND0046C	3.	* *BGD0220D	6.	* *BGD0220C	6.
37	CHN3175A	92.0	21	121.4	23.8	1.1 0.8	64.	44.5	2	94.	64•3	121.9	21.6	3.	* *PHL0285D	7.	* *PHL0285C	7.	* *CHN0168A	11.
38	CHN0176A	80.0	21	113.7	33.9	1.2 3.8	141.	44.4	1	97.	64.3	110.2	34.5	-1.	* *INS0032C	2.	* *CHN0159C	6.	* *CHN0159B	6.
39	CYP0086A	5.0	21	33.3	35.1	0.6 0.6	0.	48.7	1	31.	63.6	34.5	35 .7	2.	* *AFS00214	4.	* *SYR0229A	7.	* *YUG0148A	17.
40	DDR0216A	-1.0	21	12.6	52.1	0.8 0.6	172.	47.1	2	52.	64•2	11.0	54.0	-3.	+ *FNL0104A	2.	*BEL0018A	2.	+ *DNK0089C	4.
41	HV00107A	-31.0	21	-1.5	12.2	1.4 1.1	29.	42.1	1	155.	64.0	-5.5	12.0	-0.	≁ *ML10328E	5.	*SEN0222A	7.	+ *CTI 0237A	7.
42	IND0038C	56.0	21	75.9	33.4	1.5 1.1	33.	42.1	1	168.	64.4	79.0	31.2	-3.	*NPL0122C	-3.	*IND0042C	12.	*IND00428	12.
43	IND0046C	68.0	21	84.7	20.5	1.6 0.9	30.	42.9	1	120.	63.7	87.5	21.7	1.	+ *BRM0298C	6.	*BGD02200	8.	+ *BGD0220C	8.
44	INS0032C	80.0	21	112.3	-0.3	2.7 2.3	109.	36.3	2	595.	64.1	114.7	-4.2	3.	*1NS0030C	8.	*INS0030B	8.	*CHN0176A	12.
45	ISL0049A	-31.0	21	-19+0	64.9	1.0 0.6	177.	46.5	2	86.	65.8	-13.5	65.1	7.	*HV00107A	10.	*G 0027E	13.	+ +CTI0237A	19.
46	KEN0249A	11.0	21	37.9	1.1	2.3 1.6	94.	38.7	1	317.	63.7	3 3.9	-1.0	7.	*RRW0310E	10.	*BDI0270A	14.	*ZAI0322E	18.
47	MC00116A	-37.0	21	7.4	43.7	0.6 0.6	0.	48.7	1	24.	62.4	5.6	46. 0	-9.	+ *BEL 0018A	-4.	*MRC0209A	-2.	- *YUG0148A	1.
48	MRC0209A	-25.0	21	-9.0	29.2	2.7 1.5	43.	3 8. 2	2	318.	63.3	-1.2	32.2	6.	+ *ALG0252E	7.	*TUN0150A	12.	+ *ISL0049A	22.
49	NPL0122C	50.0	21	83.7	28.3	1.7 9.6	163.	44.1	2	113.	64.6	88•0	26.5	з.	* IND0046C	7.	+ *BRM0298C	8.	*IND0041C	15.
50	NZLO287C	128.0	21	173.0	-41.0	3.3 1.3	48.	38.0	1	488.	64.9	166.3	-45.5	12.	+ *AUS0008F	16.	*AUS0007E	16.	+ *AUS0005F	22.
51	SEN0222A	-37.0	21	-14+4	13.8	1.5 1.0	139.	42.4	2	132.	63.6	-15.0	16.3	2.	*MRC0209A	6.	*MTN0223A	7.	*MLI0328E	10.
52	U4E0274A	17.0	21	53.6	24+2	1.0 0.8	162.	45•3	1	61.	63.2	50.8	24.6	7.	+ *ARS0003E	9.	*KWT0113A	15.	*KEN02494	17.
53	YUG0148A	-7.0	21	18.4	43.7	1.7 0.7	154.	43.8	1	139.	65.2	13.4	46.4	-1.	+ +DDR 0216A	3.	*MCD0116A	3.	*HNG0106A	9.
																				4
											•									

£ -

' '

.

Canal 22 — Channel 22 — Canal 22

YUG0149A 6. AUS0005F 8. NZL0287C 17. SYR0229A 11. IND0046D 6. IND0046D 14. CHN0183A 7. PHL0285D 10. CBG0299C 17.
AUSD005F 8. NZL0287C 17. SYR0229A 11. IND0046D 6. IND0046D 14. CHN0183A 7. PHL0285D 10. CBG0299C 17.
NZL 0287C 17. SYR 0229A 11. IND 0046D 6. IND 0046D 14. CHN 0183A 7. PHL 0285D 10. CBG 0299C 17.
SYR0229A 11. IND0046D 6. IND0046D 14. CHN0183A 7. PHL0285D 10. CBG0299C 17.
IND0046D 6. IND0046D 14. CHN0183A 7. PHL0285D 10. CBG0299C 17.
IND0046D 14. CHN0183A 7. PHL0285D 10. CBG0299C 17.
CHN0183A 7. PHL0285D 10. CBG0299C 17.
PHL0285D 10. CBG0299C 17.
CBG0299C 17.
AGL0295A 1J.
HV00107A 6.
BLR0062A 19.
ISL0050A 16.
SUI0140A 4.
1N90046C 12.
CBG0299C 10.
INS0032C 7.
URS00778 12.
UAE0274A 15.
INS0032C 15.
NIG0119A 12.
SUI 0140A 12.
IND00380 17.
CHN0159C 10.
INS0030C 29.
AGL0295A 20.
ALB0296A 13.
- cont.
•

Canal 22 suite --- Channel 22 cont. --- Canal 22 cont.

1

	1	2	3	4		5	6	7	8	9 a	9 b	10		11	12		13		14	
											T									
5	L SUI0140A	-19.0	22	8•2	46•6	1.0 0.7	171.	45.9	2	66.	64.1	8.5	47.8	2.	*NIG0119A	6.	*FNL0104A	7.	*BEL0018A	12.
5	2 SYRD229A	11.0	22	38.3	34.9	1.0 0.9	7.	44.5	1	74.	63.2	35.9	35.7	2.	*CYP0086A	5.	*JCR0224A	10.	*TGK0225A	12.
5	B TUN01504	-25.0	22	9.5	33.5	1.9 3.7	153.	42.9	1	123.	63.8	12.)	37.5	2.	*GHA0108A	9.	*CVA0085A	9.	*MRC0209A	9.
5	4 UR S00708	44.0	22	73.9	41.0	1.3 0.8	5.	43.7	2	122.	64.6	73.8	38.4	4.	* *PAK02818	7.	+ *IND0042C	9.	+ *IND00380	17.
5	5 URSD081A	140.0	22	168.5	65.5	2.9 0.6	168.	43.5	1	286.	68.1	158.1	67.8	13.	*CHN0168A	13.	+ +URS0077B	26.	*AUS0008F	32.
c	anal 23	Cha	nnel ·	23 -	Ca	nal 23						• .		·						-
																	· · · ·		1	
	1	2	3	4		5	6		8	9 a	9 b	10		11	12		13		14	
	3 AGL02954	-13.0	23	16.5	-12-0	3-1 2-3	84-	35.8	1	691-	64-1	12.5	-6.5	7.	+CEG0235A	9.	+CAF0258A	15.	+SDN0230A	24.
	A NOSO340A	17 0	23	52 3	24 8	2.7 0.8	143.	40.7	1	177.	63.2	48.0	30-0	2.	* * (2002564	5.	* *KWT01134	6.	* *0MA0123A	15.
1	+ AN30340A	1140	23	7205	2400	2.11 0.0	T 430	4001	•	1	0502	4000	5000	2.	*		*	-	*	
3	5 AUS0004F	98.0	23	121.8	-24.9	3.6 1.9	54.	35.9	2	548.	63.3	124.5	-15.5	-2.	*PHL0285E ≠	2.	*PHL0285D *	2•	*INS00320 *	10.
3	6 AUS0009F	128.0	23	147.2	-32.0	2.1 1.4	15.	39.6	1	300.	64.3	141.0	-34.0	5.	*AUS0007F	6.	*AUS0008F	13.	*AUS0004F	: 13.
3	7 BRM0298D	74.0	23	97.1	19.1	3.6 1.5	104.	37.0	2	497.	64.0	92.4	21.4	-1.	*IND 00460	3.	*BGD0220E	6.	*BGD02200) 6.
3	8 CHN01580	80.0	23	111.8	38.0	2.6 1.7	124.	37 •7	1	533.	65.0	119.8	46.8	-2.	*URS00778	1.	*INS0032D	3.	*CHN0168A	12.
3	9 CNR0130A	-31.0	23	-15.7	28•4	1.5 0.6	5.	44.6	2	56.	62.8	-13.5	29.2	-13.	*≘ 0129A- *	-13.	+CT10237A *	13.	*SRL0259A *	13.
4	D CVA00854	-37.0	23	10.8	41.5	2.0 0.6	138.	43.5	1	105.	63.6	18.0	40 •0	-4.	*YUG0149A *	-2.	*E 0129A *	6.	*MTN0288A *	12.
4	1 ā 0129A	-31.0	23	-3.1	39.9	2.1 1.1	154.	40.5	2	223.	63.9	-2.9	35.3	~8.	*CNR0130A	-8.	*CT10237A *	13.	*SRL02594 *	13.
4	2 GHA01084	-25.0	23	-1.2	7.9	1.5 1.1	102.	42•3	1	136.	63.6	0.1	11,0	3.	*NGR0115A	7.	*NIG0119A	8.	*TUN0150#	16.
4	3 GNE03034	-19.0	23	10.3	1.5	0.7 0.6	10.	48.1	2	37.	63.8	11.3	2•4	0.	*AGL0295A	4.	*C0G0235A	8.	*CAF02584	9.
4	4 HJL02134	-19.0	23	5.4	52.0	0.8 0.6	171.	47.7	1	48.	64•4	6.2	50.6	-0.	*E 0129A	5.	*YUG0149A	7.	*CVA0085A	7.
4	5 IND00380	56.0	23	75.9	33.4	1.5 1.1	33.	42.1	1	169.	64•4	7 9.0	31.2	6.	*IND00420	12.	*IND0042C	12.	*IND00410) 14.
4	6 IND00460	68.0	23	84.7	20.5	1.6 3.9	30.	42.9	1	121.	63.7	87.5	21.7	1.	*88.M0298D	6.	+ *BGD0220E	8.	*BGD02200) 8.
4	7 INS00320	80.0	23	112.3	-0.3	2.7 2.3	109.	36.3	2	598.	64.1	117.5	3.7	1.	* *CHN0158C	2.	*CHN0159C	12.	*INS00300) 16.
4	8 ISL00504	5.0	23	-19.5	61.0	2.2 0.8	4.	41.8	1	281.	66•3	-6.2	62.3	7.	*DNK0090A	10.	*FNL0104A	10.	≠ TGK02254 *	1 23.
4	9 JOR02244	11.0	23	35.8	· 31•4	0.8 0.8	114.	46.1	2	50.	63.1	38.3	33.3	-2•	*TGK 0225A	3.	*ARS0340A	5.	*SYR0229/	A 5.
5	0 NIU00548	158.0) 23	8-169.8	-19+0	0.6 0.6	0.	48•7	2	35.	64.1	-169.9	-19.0	27.	*TKL0058B	31.	*AUS0009F	31.	+URS00776	8 35.

• • •

. ...

— 29 —

Canal 23 suite --- Channel 23 cont. --- Canal 23 cont.

	1	2	3	4		5	6	7	8	9 a	9b	10		11	12		13	14
51	SDN0230A	-7.0	23	29.2	7.5	2.3 1.1	148.	40.1	່2່	272.	64.4	23.0	11.0	1.	*CAF0258A	6.	*SDN0231A 7	*YUG0149A 9.
52	SRL0259A	-31.0	23 -	-11.8	8.6	0.8 0.7	114.	47 . J	1	44.	63.4	-10.5	8.5	2.	*∃ 0129A	7.	*CT10237A 8	*GHA0108A 9.
53	TGK0225A	11.0	23	34•6	-6.2	2.4 1.7	129.	38•1	1	366.	63.7	30.5	-1.0	• 3.	*3DI0270A	7.	*AGLJ295A 10	*SDN0233A 1J.
54	URS0061F	23.0	23	24.7	56.6	0.9 0.6	12.	46.7	2	71.	65.3	21.3	55.3	-1.	+ #URS0064F	0.	+ *CVA0085A 10	*YUG0149A 14.
55	URS0064F	23.0	23	45.6	40.8	2.2 0.6	163.	43•1	1	128.	64.2	46.6	38.8	3.	*ARS0340A	4.	+URS0061F 12	*ETH0092A 17.
56	UR \$90778	110.0	23 1	112.7	57.3	2.7 1.8	2.	37.6	1	722.	66.1	117.8	49.4	3.	*CHN0158C	3.	*CHN0168A 17	*BRM02980 21.
57	YUG0149A	-7.0	23	18.4	43.7	1.7 0.7	154.	43.8	1	140.	65.2	13.4	46.4	-1.	*C VA0085A	2•	*SDN0230A 6	*F 0129A 9.
						•					•							
Ca	inal 24 -	– Chai	nnel 2	- 24	- Cai	nal 24												
	1	2	3	4		5	6	7	8	9 a	9b	10		11	12		13	14
26	AUS0007F	128.J	24 1	145.0	-38.1	1.8 1.4	134.	40.2	່2່	218.	63.6	149.6	-37.5	7.	*AUS0009F ★	7.	*URS0079B 24	*PHL0285E 28.
27	BGD0220E	74.0	24	90.3	23.6	1.5 0.8	135.	43.4	1	115.	63•8	92.5	25.0	-2•	*CHN0188A	э.	- *8RM0298D 6	*IND0042D 7.
28	CAF0258A	-13.0	24	21.0	6.3	2.3 1.7	31.	38.5	2	379.	64.3	23.2	11.0	1.	*SDN0232A	5.	*NGR0115A 6	*SDN0230A 6.
29	CBG3299D	68.0	.24	105.0	12.7	1.0 0.9	110.	44.7	1	93.	64.3	103.5	10.5	2.	*IND0042D	4-	*MLA0227E 13	*IND0046D 14.
30	CHN0166A	92 . D	24 :	121.1	41.7	1.5 0.8	154.	43.5	2	126.	64.5	125.6	40.8	٥.	*URS0079B	2.	*PHL02855 7	*CHN0158C 16.
31	CHN0177A	80.0	24	111.8	30.8	1.4 0.8	160.	43.6	2	129.	64.7	109.2	29.1	-1.	*CHN0188A	3.	+INS0030D 4	*CHN0158C 7.
32	2 CHN0188A	62.0	24	101.5	25.1	1.9 1.1	132.	41-2	2	240.	65.0	97.5	24•9	2.	*IND0041D	5.	*BGD0220E 7	*IND0042D 12.
33	DNKOOOOA	5.0	24	17.0	61.5	2.0 1.0	10.	41.2	2	473.	68.0	12.0	54.6	4.	*DDR.02168	8.	*I J082A 8	*!SL0050A 16.
34	A286C I	-19.0	24	12.3	41.3	2.4 1.0	137.	40.6	2	227.	64.1	12.6	35•4	3.	*NGP 0115A	4.	*NMB0025A 15	*CAF0258A 17.
35	5 IND00410	56.0	24	78.4	16.0	2.1 1.4	35.	39.7	2	263.	63.9	84.9	19.0	2.	*IND0042D	3.	*IND00460 12	• *BGD0220E 14•
36	IND0042D	68.0	24	79.3	27.7	2.1 1.2	147.	40.3	2	227.	63.9	84•6	25.8	-1.	*BGD0220E	2.	*IND00410 4	*C8G0299D 10.
37	INS0030D.	80.0	24	112.3	-8.1	3.1 1.5	169.	37•6	1	459.	64.3	114.0	-5-8	4.	*INS0032D	7.	*CHN0177A 10	• *CHN0158C 13.
38	B IRQ0256A	11.0	24	43•6	32.8	1.9 1.0	143.	41.7	1	146.	63.3	44.0	37.3	-0.	*UR\$0066B	1.	*KEN0249B 11	• *TGK0225A 12•
39	9 LS00305A	5.0	24	27.8	-29.8	0.7 0.6	36.	. 48•3	1	39.	64.2	29.5	-29.4	1.	*DNK0090A	3.	*AFS0021B 5	• *NMB0025A 19.
4	MLAD227E	86.0	24	192.1	4.1	1.6 0.8	135.	43.0	1	108.	63.4	103.8	1.2	-1.	*INS0030D	-1.	*INS0032D 15	• *BRM0298D 22•
4	MTN0288A	-37.0	24	-7.8	23.4	1.6 1.1	141.	41.7	1	134.	63.0	-5.6	20.0	3.	*NGR 0115A	5.	*CVA0085A 14	• *SEN0222B 14.
42	2 MW10308A	-1.0	24.	34.1	-13.0	1.5 0.6	87.	44.6	. 2	92.	64.2	35.2	-17.2	10.	*AFS0021B	12.	*TGK02254 21	• *AGL02954 22•
				· .							30						suite	— cont.
					,)	·		ţ, s	د. ۲

	1	2	3	4		5	6	7	8	9 a	9b	10		11	12		13		14	
																_	Ţ			
43	AYT0098A	29.0	24	45.1	-12.8	0.6 0.6	0.	48.7	1	30.	63.4	45.0	-12.8	18.	*TGK0225A	24.	*URS0066B ≭	28.	≠0MA0123A *	29.
44	NGR0115A	-25.0	24	8.3	16.8	2.5 2.1	44.	37.0	2	558.	64.5	13.0	13.0	4.	*CAF0258A	6.	*MRC0209B *	13.	*SDN0232A *	14.
45	5 OMA0123A	17.0	24	55.6	21.9	1.9 1.0	100.	41.4	2	153.	63.3	56.0	26.5	-0.	*URS0066B	6.	*UAE02748	7.	*ARS0340A	7.
46	5 PAK0282B	38.0	24	68.5	25•8	1.3 0.6	133.	45.1	1	67.	63.4	68.3	28.7	-1.	*URS00668	-0.	*IND0042D	7.	*DMA0123A *	10.
47	7 PHL0285E	98.0	24	121.3	11.1	3.5 1.8	99•	36.4	2	545.	63.8	122.0	21.0	9.	*AUS0004F	13.	*CHN0188A	16.	*CHN0166A *	17.
48	B SUNOZ32A	-7.0	24	30•4	19.0	2.4 1.5	176.	38.6	1	295.	63.3	24.0	15.3	-0.	*CAF0258A	3.	*NGR0115A *	5.	*SDN0230A *	11.
49	9 TKL0058B	158.0	24	-171.8	-8.9	0.7 0.6	35.	48.0	1	38.	63.9	-171.2	-9•5	26.	*NIU00548 *	29.	*URS0079B ≭	30.	*AUS0007F *	35.
50	URS0066B	44.0	24	64.3	. 44•6	4.6 2.5	169.	33.7	2	1485.	65.4	53.9	37.3	8.	*IRQ0256A	9 • ַ	*NGR 0115A *	17.	*UR S0064F *	29.
5	L URS)079B	140.0	24	138.0	53.6	3.2 2.1	62.	36.0	2	1516.	67.8	130.6	42•2	6.	*CHN0166A	7.	*PHL0285E	22.	*CHN0158C	24.

1

Canal 24 suite - Channel 24 cont. - Canal 24 cont.

.

1

Canal 25 — Channel 25 — Canal 25

	1	2	3	4		5	6	7	8	9 a	9 b	10		11	12		13		14	
					· · · [
2	7 AFI00998	23.0	25	42.5	11.6	0.6 0.6	0.	48.7	1	25.	62.6	41.9	10.8	-0.	*KEN02498 *	2•	*ETH0092B *	6.	*BLR 0062B	10.
2	8 AFS00218	5.0	25	24.5	-28.0	3.1 1.7	27.	37.0	2	511.	64.1	20•0 ·	-25.0	-1.	*NMB0025A	-0.	*CYP0086B	16.	*L SD 03054	18.
2	9 BEL00188	-19.0	25	4.6	50.6	0.8 0.6	167.	47.3	1	50.	64.3	6.4	50.3	-1.	+ *NMB0025A *	4.	*DDR02168	5.	*YUG01488 *	6.
3	0 BLR00628	23.0	25	27.8	52•6	1.1 0.7	1.	45. 3	2	90.	64.9	23.4	51.5	3.	*YUG0148B *	7.	*MC30116B *	11.	*ETH00928 *	13.
3	1 CYP00868	5.0	25	33.3	35.1	0.6 0.6	0.	48.7	1	31.	63.6	34.5	35.7	-0.	*4FS00218 *	4.	*ISR0110A *	5.	*SYR0229B *	8.
3	2 DDR02168	-1.0	25	12.6	52.1	0.8 0.6	172.	47.1	2	53.	64.3	11.0	54.0	-4.	*FNL01048 *	2.	*B510018B *	2.	*DNK0090A_ *	2•
3	3 HVOQIO78	-31.0	25	-1.5	12 .2	1.4 1.1	29.	42.1	1	156.	64.0	-5.5	12.0	2.	≠SEN02228 *	7.	*CTI0237B *	7.	*MTN0223B *	9.
3	4 ISL00498	-31.0	25	-19.0	64.9	1.0 0.6	177.	46.5	2	87.	65.9	-13.5	65.1	9.	*HV001079 *	10.	*CTI0237B *	19.	*MRC02098 *	20.
13	35 ISR01104	-13.0	25	34.9	31.4	0.9 0.6	117.	46.7	2	51.	63.8	35.2	33.2	2.	*CYP00868 *	3.	*CAF0258A *	13.	≠C0G02358 ≯	14.
3	6 KEN02498	11.0	25	5 37.9	1.1	2.3 1.6	94.	38.7	1	321.	63.8	41.9	3.9	9.	*ETH0092B *	13.	*IRQ0256A *	14.	*S Y R02 29 B *	19.
3	87 MCD31168	-37.0	25	5 7.4	43•7	0.6 0.6	• 0.	48.7	1	24.	62.5	5.6	46.0	-9.	*BEL00188 *	-4.	*MRC0209B *	-2.	*YUG01485 *	1.
	38 MNG02484	74.0	25	5 102.2	46•6	3.6 1.1	169.	38.2	1	393.	64.1	98.8	51.9	-1.	*URS0078A *	J .	*URS0074A *	4.	*BGD 02205 *	13.
	39 MRC02098	-25.0	25	5 -9.0	29.2	2.7 1.5	43.	38.2	2	322.	63.3	-1.2	32.2	3.	*NGR0115A	4.	*TUN01508	12.	*NM800254	29.

suite — cont.

.

· · ·

Canal 25 suite — Channel 25 cont. — Canal 25 cont.

• ••

Г		1	2	3	· 	1	5	a	7		0.	96	10		11	12		13		14	
H	+			-3-1	_	+	2710		′	+ -++++++++++++++++++++++++++++++++++++	- 30		20.0	- 29 6		+ A ES 1021P	-0	* DEL 00190	1.4	*NICOLLOP	21
ľ	+0	NMBUUZSA	-19.0	25	11.5	-21-0	2.1 1.9	40.	31.2	2	202.	04+1	20.0	-20.4	0-	* *	-J.	*DELUUIOD . *	14.	*	21.
-	41	SEN0222B	-37.0	25	-14.4	13.8	1.5 1.0	139.	42•4	2	133.	63.7	-15.0	16.3	3.	*M&C02098	6.	*MTN02238	7.	*MTN0288A	16.
	42	UAE02748	17.0	25	53.6	24.2	1.0 0.8	162.	45.3	1	62.	63.2	54.9	22.4	5.	* *0MA0123A	6.	* *KEN0249B	17.	* *AFI00998	20.
	43	UR SO 078A	110.0	25	108.2	53•4	2.2 0.8	10.	42.0	1	199.	65.0	107.8	50.0	1.	* *MNG0248A *	1.	+ +URS0074A . *	18.	+ *URS0066B *	20.
	44	YUGC148B	-7.0	25	18.4	43.7	1.7 0.7	154.	43.8	1	140.	65.3	13.4	46.4	-1.	*DDR 0216B	3.	≠MC00116B	3.	*HNG01068	9.
			,																		
		· · · · ·											4 - A								
ŀ	Cana	al 26 -	— Char	nel	26 –	Ca	nal 26														
Ċ		1	2	3	4		5	6	7	8	9 a	9b	10)	11	12		13		14	
ſ	19	ALB02968	-7.0	26	19.8	41.3	0.7 0.6	146.	48.1	2	37.	63.8	19.8	42.6	-2.	*HNG01068	2.	*SDN02318	5.	#YUG0149 B	6.
	20	BD10270B	11.0	26	29.9	-3.1	0.7 0.6	80.	48.0	2	35.	63.4	29.1	-2.7	3.	≠ *TGK02258	7.	* *ETH00928	10.	* *SYR02298	11.
	21	COG0235B	-13.0	26	14.6	-0.7	2.0 1.2	59.	40.5	2	217.	63.8	12.6	2.4	-2.	*NIG0119B	0.	+ ≠GN£0303B *	5.	*AGL02958	10.
	22	CTI0237B	-31.0	26	.`−5•6	7.5	1.6 1.2	108.	41.3	2	174.	63.7	-3.5	9.8	-1.	*N1G0119B	4.	*MTN0223B	4.	*HV001078	6.
	23	ETHO092B	23.0	26	39.7	9•1	3.5 2.4	124.	35.0	2	797.	63.5	39.2	17.5	2.	*SDN0231B	3.	*URS0059A	11.	*KWT01138	19.
	24	FNL0104B	5.0	_ 26	17.9	61.5	2.0 1.0	10.	41.2	2	476.	68.0	15.0	55.0	5.	*DDR 0216B	8.	*SUI014JB	11.	*DNK0091A	17.
	25	HNG0106B	-1.0	26	19.5	47.2	0.9 0.6	176.	46.8	1	53.	64.0	16.1	46. 8	-3.	*YUG01498 *	4.	*YUG0148B	4.	*SUI 01408	4.
	26	KWT0113B	17.0	26	47.6	29.2	0.7 0.6	145.	48.1	2	32.	63.1	46.3	28.3	1.	*ETH00928 ≯	2.	*UAE0274B *	15.	*URS0059A *	16.
	27	MTN0223B	-37.0	26	-12.2	18.5	2.6 1.9	150.	37.3	1	355.	62.9	-5.2	15.3	٥.	*CT102378 *	2.	*HV001078 *	7.	*NIG01198 *	12.
	28	NIG0119B	-19.0	26	7.8	9.4	2.2 2.0	45.	37.9	1	407.	63.9	3.4	6•4	4.	*СТ10237∂ *	8.	*GHA01088 *	9.	*SUI0140B *	12.
	29	REU00978	29.0	26	55. 6	-19.2	1.6 0.8	96.	43.4	1	115.	64.9	54.7	-15.7	16.	*ETH0092B	17.	*RHS0135B *	29.	*URS0074A *	30.
	30	RHS01358	-1.0	26	29.6	-18.8	1.5 1.4	37.	41.3	2	197.	64•2	31.2	-22.2	5.	≠4FS00218 *	7.	*HNG0106B *	13.	*NMB00254.	19.
Ì	31	SDN0231B	-7.0	26	28.9	12.7	2.3 2.0	159.	37.8	1	376.	63.5	34.0	8.5	-2.	*ETH00928 *	-0.	*SDN02308 *	6.	*ALB0296B *	13.
	32	SUI0140B	-19.0	26	8.2	46.6	1.0 0.7	171.	45.9	2	67.	64.1	8.5	47 . 8	2.	*NIG01198 *	6.	*FNL0104B *	7.	*BEL00188	12.
	33	S¥R0229B	11.0	26	38.3	34.9	1.0 0.9	7.	44.5	1	75.	63.3	35.9	35.7	2.	*CYP0386B *	5.	*JCR0224B *	13.	*TGK02258 *	12.
	34	TUN0150B	-25.0	26	9.5	33.5	1.9 3.7	153.	42.9	1	125.	63.9	9.0	38.0	3.	*MRC0209B *	9.	*GH40108B *	10.	*CVA0083A *	11.
	35	UR S0 068A	44.0	26	59.0	38.8	2.2 1.3	164.	40.7	2	213.	64.0	52.5	41.8	12.	*SYR 3229B	15.	*URS0059A	18.	*UR 500744	24.
	36	UR 50074A	74.0	26	88.8	57.6	3.1 1.7	162.	37.1	2	1210.	67.9	97.2	49.7	9.	*MNG0248A *	9.	*URS0077C *	21.	*URS0078A *	22.
	37	UR S 0 08 0A	140.0	26	155.3	55.4	2.9 2.4	35.	35.9	1	1584.	67.9	145.6	63.8	22.	*URS 0077C	23.	#URS0078A	34.	*YUG0149 8	89.

e s

ې

					**											**	. ,
Can	al 27	Chanr	nel 2	27	- Cana	al 27						·		•			
	1	2	3	4		5	6	7	8	9 a	9b	10		11	12	13	14
20	AGL0295B	-13.0	27	16.5	-12.0	3.1 2.3	84.	35•8	1	690.	64.2	12.5	-6.5	7.	 ≠CDG0235B 9•	*CAF0258B 15.	*SDN02305 24.
21	BHRJ255A	17.0	27	50•5	26.1	0.6 0.6	0.	48.7	1	16.	60.8	52.3	24•4	-3.	* *3MA0123B 0. *	* ≄URS0059A 6. *	* *IPQ02563 7. *
22	CNR0130B	-31.0	27	-15.7	28.4	1.5 0.6	5.	44•6	2	67.	62.8	-13.5	29.2	-13.	¥≓ 01298-13.	*CTI0237B 13.	*SRL02598 13.
23	CVA0083A	-37.0	27	12.4	41.8	0.6 0.6	0.	48.7	1	45.	65.2	12.5	43.8	-3.	+ *YUG0149B -1. *	≁ ≠E 0129B 2• *	*URS0059A 13. *
24	DNK0091A	5.0	27	-19.5	61.0	2.2 0.8	4.	41.8	1	275.	66.2	-6.2	62•3	6.	*FNL01048 10.	*NOR0121A 10.	*URS0059A 12.
25	E 01298	-31.0	27	-3.1	39.9	2.1 1.1	154.	40.5	2	226.	64.0	-2.9	35.3	-8.	*CNR 01308 -8.	*CT102378 13.	*SRL 02598 13.
26	GHA0108B	-25.0	27	-1.2	7.9	1.5 1.1	192.	42.3	1	137.	63 .7	0.1	11.0	3.	*NGR 0115B 7.	*NIG0119B 8. *	*TUN0150B 16. *
27	GNE0303B	-19.0	27	10.3	1.5	0.7 0.6	10.	48.1	2	37.	63.8	11.3	2.4	0.	*AGL0295B 4.	*COG02358 8.	*NIG0119B 9.
22	HƏL02138	-19.0	27	5.4	52.0	0.8 0.6	171.	47.7	1	48.	64.5	6.2	50.6	0.	*5 0129B 5.	*YUG0149B 7.	*SUID140B 10. *
29	JUR02248	11.0	27	35.8	31.4	0.8 0.8	114.	46.1	2	50.	63.1	35.0	34.0	-4.	*UPS0059A -0. *	*TGK0225B 1.	*SYR0229B 5.
30	SDN0230B	-7.0	27	29•2	7.5	2.3 1.1	148.	40.1	2	275.	64.5	23.0	11.0	1.	*CAF02588 6.	*SDN0231B 7.	*YUG0149B 9. *
31	SRL0259B	-31.0	27	-11.8	8.6	0.8 0.7	114.	47 •0	1	45.	63.5	-10.5	8.5	2.	*E 0129B 7.	*CT10237B 8.	*GHA0108B 5.
32	2 TGK02258	11.0	27	34.6	-6.2	2.4 1.7	129.	38.1	1	371.	63.8	30.5	-1.0	3.	* *3D102708 7. *	*AGL0295B 10. *	*SDN0230B 10. *
33	UR \$0059A	23.0	27	36.0	47.0	3.7 1.4	153.	37.0	2	651.	65.2	22.3	48•4	-1.	*Y UG0149B 0.	*CVA0083A 6.	*ETH00928 12.
34	URS0077C	110.0	27	112.7	57.3	2.7 1.8	2.	37.6	1	920.	67.2	106.2	69.5	12.	*URS0076A 15. *	+ *URS0074A 16. *	+ *URS0066C 22• *
35	5 YUG0149B	-7.0	27	18.4	43.7	1.7 0.7	154.	43•8	1	141.	65.3	20.1	46.3	-3.	*CVA00334 2.	*URS0059A 4.	*HNG0106B 5.

Canal 28 — Channel 28 — Canal 28

	1	2	3	4	5	6	7	8	9 a	9 b	10	11	12	13	14
								ΙŢ			.				
1	7 CAF0258B	-13.0	28	21.0 6.3	2.3 1.7	31.	38.5	2	383.	64.3	23.2 11.0	1.	*SDN02328 5. *	*NGR 01158 6. *	*SDN02308 6. *
1	B I 0082B	-19.0	28	12.3 41.3	2.4 1.0	137.	40.6	2	230.	64.2	12.6 35.4	3.	*NGR0115B 4.	*NMB0025B 15. *	*CAF0258B 18•
1	9 IRQ0256B	11.0	28	43.6 32.8	1.9 1.0	143.	41.7	1	148.	63.4	44.0 37.3	-0.	*URS0066C 1. *	*KENJ249C 11. *	*TGK02258 12+ ≠
2	0 LS003058	5.0	28	27.8 -29.8	0.7 3.6	36.	48.3	1	39.	64.2	29.5 -29.4	1.	*NOR 0121A 3.	*AFS9021C 5.	*NMB0025B 19∘ *
2	1 NTN0288B	-37.0	28	-7.8 23.4	1.6 1.1	141.	41.7	1	136.	63.0	-5.6 20.0	3.	*NGR 01153 5. *	*SEN0222C 14. *	*CVA0083A 14• *
2	2 MWI0308B	-1.0	28	34.1 -13.0	1.5 0.6	87.	44.6	2	94.	64.3	35.2 -17.2	10.	*AFS0021C 12.	*TGK0225B 21. *	*AGL0295B 22.
2	3 MYTO0988	29.0	28	45.1 -12.8	9.6 0.6	0.	48.7	1	30.	63.5	45.0 -12.8	19.	*TGK02258 24.	*URS0066C 28.	*OMA0123B 29.

.

— 33 — [`]

Canal 28 suite - Channel 28 cont. - Canal 28 cont.

		1	2	3	4		5	6	7	8	9 a	9 b	10		11	12		13	14
2	4 NGF	R0115B	-25.0	28	8.3	16.8	2.5 2.1	44.	37.0	2	565.	64.5	13.0	13.0	. 4.	*CAF0258B	6.	*MRC 3209C 13.	*SCN02329 14.
2	5 NOF	R0121A	5.0	28	17.0	61.5	2.0 1.0	10.	41.2	2	453.	67.8	15.0	55.0	4.	*DDR0216C	8.	≑I 0082B 8.	+ ≠URS0066C 15.
2	6 OM/	A0123B	17.0	28	55.6	21.0	1.9 1.0	100.	41.4	2	155.	63.3	56.0	26.5	1.	*UFS0066C	6.	, ≠UAE0274C 7.	*IRQ0256B 7.
2	7 SDI	NJ232B	-7.0	28	30.4	19.0	2.4 1.5	176.	38.6	1	298.	63.3	24.0	15.8	-0-	≑CAF0258B	3.	*NGR0115B 5.	*SDN02308 11.
2	8 UR!	200660	44.0	28	64.3	44.6	4.6 2.5	169.	33.7	2	15)3.	65.5	53.9	37.3	8.	*IRQ0256B	9.	*NGR0115B 17. ≠	*URS0059A 22.
2	9 UR	50076A	74.0	28	98.0	63.2	1.8 0.7	170.	43.2	2	310.	68.1	106.5	69.5	3.	*UR \$0366C	6.	≠URSJ079C 6. *	*URS00770 18.
3	O URS	530790	140.0	28	138.0	53.6	3.2 2.1	62.	36.)	2	1535.	67.8	128.4	73.2	1.	*URS0076A	1.	*URS0077C 18.	*MNG02488 35.
c	anal	29	- Chan	nel 🕯	29 –	- Car	nat 29												
		1	2	3	4		5	6	7	8	9 a	9 b	10)	11	12		13	14
1	15 AF	1009 9 C	23.0	29	42.5	11.6	0.6 0.6	0.	48.7	' 1	25.	62.6	41.9	10.8	-1.	*KEN0249C	2.	*UKR0063A 6	. *ETH00920 6.
1	ί6 AF	S0021C	5.0	29	24.5	-28.0	3.1 1.7	27.	37.0	2	518.	64•2	20.0	-25.0	-1.	*NMB00258	-0.	*CYP0086C 16	*LS003058 18.
1	17 BE	L0018C	-19.0	29	4.6	50.6	3.8 3.6	167.	47.3	1	50.	64.3	6.4	57.3	-1.	*NMB00258	4.	*DDR0216C 5	*YUG0148C 7.
1	L8 CY	P0086C	5.0	29	33.3	35.1	0.6 0.6	0.	48•7	1	31.	63.7	34.5	35 .7	-0.	*AFS0021C	4.	+ISR01108 5	*SYR02290 8.
1	L9 DD	R0216C	-1.0	29	12.6	52.1	0.8 0.6	172.	47.1	2	53.	64.3	11.0	54.0	-4.	*BFL0018C	2.	*S 0139A 2	*NCR0121A 2.
2	20 H V	001070	-31.0	29	-1.5	12.2	1.4 1.1	29.	42•1	1	158.	64.1	-5.5	12.0	2.	*SEN0222C	7.	*CTI0237C 7	. ≠MTN0223C 9.
2	21 15	L0049C	-31.0	29	-19.0	64.9	1.0 0.6	177.	46.5	2	88.	65.9	-13.5	65.1	۶.	*HV00107C	13.	*CTI0237C 19. *	*MRC0209C 2J.
Ż	22 IS	R0110B	-13.0	29	34•9	31.4	0.9 0.6	117.	46•7	2	52.	63.9	35.2	33.2	2.	*C YP 0086C	3.	*C≜F02588 13.	≠COG0235C 14•
2	23 KE	N0249C	11.0	29	37.9	1.1	2.3 1.6	94.	38.7	1	325.	63.8	41.9	3,•9	9.	*ETH0092C	13.	*IRQ02568 14	*SYR0229C 19.
2	24 MC	001160	-37.0	29	7.4	43.7	0.6 0.6	э.	48.7	1	24.	62.5	5.6	46.9	-9.	*BEL0018C	-4.	*MRC0209C -2	*YUG0148C 1.
2	25 MN	G0248B	74•0	29	102.2	46•6	3.6 1.1	169.	38.2	1	398.	64.2	87.9	48.9	2.	*URS00748	3.	*URS0076A 11.	*URS0966C 17.
2	26 MR	C0209C	-25.0	29	-9.0	29.2	2.7 1.5	43.	38.2	2	326.	63.4	-1.2	32.2	3.	*NGR01158	4.	*TUN0150C 12	. #NMB00258 20.
2	27 NM	80025B	-19.0	29	17.5	-21.6	2.7 1.9	48.	37.2	2	572.	64.8	20.0	-28.4	-0.	*AFS0021C	-j.	*BEL0018C 14	*NIG0119C 21.
Z	28 SE	N0222C	-37.0	29	-14.4	13.8	1.5 1.0	139.	42.4	2	135.	63.7	-15.0	16.3	3.	*MRC0209C	6.	*MTN0223C 7.	*MTN0288B 16.
2	29 UA	E0274C	17.0	29	53.6	24.2	1.0 0.8	162.	45.3	1	63.	63.3	54.9	22.4	5.	*GMA01238	6.	*KENJ249C 17	*AF10099C 20.
3	BƏ UK	R0063A	23.0	29	31.2	48•4	2.3 1.0	172.	40.8	2	243.	64.6	22.1	48.4	÷0.	*YUG0148C	2.	*MCD0116C 8	*ETH0092C 13.
13	31 YU	G0148C	-7.0	29	18.4	43•7	1.7 3.7	154.	43.8	1	142.	65.3	13.4	46.4	- ⁻ •	*DDR0216C	3.	- *MCD0116C 3	. *HNG0106C 9.

- 34 --

.

. . .

. . Canal 30 --- Channel 30 --- Canal 30

, **,**

1

	1	2	3	4		5	6	7	8	9 a	9 b	10		11	12		13		14	
18	ALB0296C	-7.0	30.	19.8	41.3	0.7 0.6	146.	48.1	2) 38.	63.9	19.8	42.6	-2.	 *HNG0106C	2.	 *SUN0231C	5 •	 *YUG0149C	6.
19	BD10270C	11.0	30	29.9	-3.1	0.7 0.6	80.	48.0	2	36.	63.5	29.1	-2.7	3.	+ + TGK 0225C	7.	*ETH0092C	10.	+ *SYR02290	11.
20	C0G0235C	-13.0	30	14.6	-0.7	2.0 1.2	59.	40.5	2	219.	63.9	12.6	2.4	-1.	* *NIG0119C	0.	≠ *GNE0303C	5.	≠ ≠AGL0295C	10.
21	CTI0237C	-31.0	30	-5.6	7.5	1.6 1.2	108.	41.3	2	176.	63.8	-3.5	9.8	-1.	* *NIGJ119C - *	4.	* *MTN0223C	4.	* *HVJ0107C	6.
22	ETH0092C	23.0	30	39.7	9.1	3.5 2.4	124.	35.0	2	716.	63.6	39.2	17.5	2.	≁ *SDN0231C *	3.	≁ ≑URS00598	11.	*UKR00634	14.
23	HNG0106C	-1.0	30	19.5	47.2	0.9 0.6	176.	46.8	1	53.	64.1	16.1	46.8	-3.	- ≠¥UG0149C *	4.	+ ≭YUG0148C	4•	+ *\$910140C	4.
24	KWT0113C	17.0	30	47.6	29.2	0.7 0.6	145.	48.1	2	32.	63.2	46.3	28.3	1.	+ *ETH0092C	2.	*UAE0274C	15.	*URS00598	16.
25	MTN02230	-37.0	30	-12.2	18.5	2.6 1.9	150.	37•3	1	360.	62•9	-5.2	15.3	٥.	≠CTI3237C	2.	*HV90107C	7.	*NIG0119C	12.
26	NIGJ119C	-19.0	30	7.8	9•4	2.2 2.0	45.	37.9	1	412.	64.0	3.4	6.4	4.	*CTI0237C	8.	≠GHA0108C *	9.	*SUI0140C	12.
27	REU0097C	29.0	30	55.6	-19.2	1.6 0.8	96.	43•4	1	116.	64.1	54.7	-15.7	16.	*ETH0092C	17.	*RHS0135C	29.	*URS00745	30.
28	PHS0135C	-1.0	30	29+6	-18.8	1.5 1.4	37.	41.3	2	200.	64.3	31.2	-22.2	6.	*AFS0021C	7.	*HNG0106C	13.	*NMB00258	19.
29	S 0139A	5.0	30	17.0	61.5	2.0 1.0	10.	41.2	2	482.	68.1	15.1	55.0	5.	*DDR0216C	8.	*SUI314JC	11.	*ISL0050B	16.
30	SDN0231C	-7.0	30	28.9	12.7	2.3 2.0	159.	37.8	1	381.	6 3.6	34.0	8.5	-2.	*=TH0092C	-0.	*SDN0230C	6.	*ALB0296C	13.
31	SU10140C	-19.0	30	8.2	46.6	1.0 0.7	171.	45.9	2	63.	64.2	8.5	47.8	2.	*NIG0119C	6.	*S 0139A *	7.	*BEL0018C	12.
32	SYR0229C	11.0	30	38.3	34.9	1.0 0.9	7.	44.5	1	76.	63.3	35.9	35.7	2.	+CYP0086C +	5.	*JCR0224C. ≠	13.	*TGK0225C *	12.
33	TUN0150C	-25.0	30	9.5	33.5	1.9 0.7	153.	42.9	1	126.	63.9	9.0	38.0	3.	*MRC0209C	9.	*GHA0108C *	10.	*CVA0083B *	11.
34	UR \$0068B	44.0	30	59.0	38.8	2.2 1.0	164.	40.7	2	216.	64.1	52.5	41.8	12.	*SYR 32290	15.	*UR\$0059B	18.	*URS00748	24.
Car	nal 31 –	– Chan	nel	31	- Can	al 31						•								
	1	2	3	4		5	6	7	8	9 a	9 b	1()	11	12		13		14	
									1.1	T	Т			_					+	
20	D AGL02950	-13.0	31	16.5	-12.0	3.1 2.3	84.	35,8	1	69 8 .	64.2	12.5	-6.5	7.	≭CUG0235C *	9.	≭UAF0258C ¥	15.	#5DN02300 *	. 24•
21	L BHR02558	17.0	31	50.5	26.1	0.6 0.6	0.	48.7	1	17.	60.9	52•3	24•4	-3.	*0MA0123C *	0.	≄URS0059B *	6.	*IRQ02560 *	; 7.
22	2 CNR01300	-31.0	31	-15.7	28•4	1.5 0.6	5.	44•6	2	68.	62.9	-13.5	29 •2	-13.	*E 0129C *	~13.	*CT10237C *	13.	≑SRL02590 *	; 13.
23	3 CVAJ0838	-37.0	31	12.4	41.8	0.6 0.6	0.	48.7	1	45.	65.3	12.5	43.8	-3.	*YUG0149C *	-1.	*E 0129C *	2.	¥URS00598 ≄	3 13.
24	4 E 01290	-31.0) 31	-3.1	39.9	2.1 1.1	154.	40.5	2	228.	64.0	-2.9	35.3	-8.	*CNR)130C	-8.	*CTI0237C	13.	*SRL 02590	: 13.

í ·

25 GHAD108C -25.0 31 -1.2 7.9 1.5 1.1 102. 42.3 1 139. 63.7 0.1 11.0 3. *NGR0115C 7. *NIGD119C 8. *TUN0153C 16. 26 GNED303C -19.0 31 10.3 1.5 0.7 0.6 10. 48.1 2 37. 63.9 11.3 2.4 0. *AGL0295C 4. *CDG0235C 8. *NIG0119C 9.

--- 35 ---

suite — cont.

· · · · ·

	1		2	3	4	I	5		6	7	8	9 a	9 b	10		11	12		13		14	
27	HOLO2	130	-19.0	31	5.4	52.0	0.8	0.6	171.	47.7	1	49.	64.6	6.2	50.6	. 0. 3	E J129C	5.	+ <u>+</u> +YUG0149C	7.	≠SUI01400	10.
28	15130	508	5.0	31	-19.5	61.0	2.2	0. 8	4.	41.8	1	289.	66.4	-6.2	62.3	6.	≭ ¥S 0139A	10.	* *NGR01218	10.	≠ *URS0059B	12.
29	J3802	240	11.0	31	35.8	31.4	0.8	0.8	114.	46.1	2	51.	63.2	35.)	34.0	-4.	* *UR\$00598	-).	* *TGK 0225C	1.	* *SYRJ2290	5.
30	SON02	300	-7.0	31	29.2	7.5	2.3	1.1	148.	40.1	2	279.	64.5	23.0	11.0	1.	* ¢CAF0258C	6.	* *SDN0231C	7.	* *YUG0149C	9.
31	SRL02	590	-31.0	31	-11.8	8.6	0.8	0.7	114.	47.0	1	45.	63.6	-10.5	8.5	2.	¥ ¥E 01290	7.	* *CTI0237C	8.	* *GHA 01 08C	9.
32	TGK32	250	11.0	31	34.6	-6.2	2.4	1.7	129.	38.1	1	375.	63.8	30.5	-1.0	3.	⊭ \$3DI02 70C	7.	* *AGL0295C	10.	* *SDN0230C	10.
33	JRSDO	59B	23.0	31	36.0	47.0	3.7	1.4	153.	37.0	2	660.	65.2	22.0	48.4	-1.	¥ ¥YUG0149C	0.	* *CVA0083B	6.	* *ETH0092C	12.
34	URSOO	77D	110.0	31	112.7	57.3	2.7	1.8	2.	37.6	1	932.	67.2	106.2	69.5	15.	* ≭URS0074B	16.	* *U?S0066D	22.	* *UR\$0079D	22.
35	YUGD1	490	-7.0.	31	18.4	43.7	1.7	0.7	154.	43.8	1	143.	65.4	20.1	46.3	-3.	⊧ ≠ €vaoo8 3B	2.	* *UR <u>S0059</u> 8	4.	≠ ≠HNG0106C	5.
												~										
						* 7	er-			ન	· .			з. Э.						. >	A and the	ŗ
Ca	inal 32	·	- Char	nel	32 -	- Cai	nal 32	2 2		- · ·	- a * -		,			÷ •						-
	1		2	3	. 4		5		6	7	8	9a	9 b	10	,	11	12		13	,	14	
	+										╉═╋						<u>├</u>				-+	
117	7 CAF02	58C	-13.0	³²	21.0	6.3	2.3	1.7	31.	38 •5 -	2	-388.	64.4	23.2	11.0	1.	*SONJ232C	5.	*NGR 0115C	6.	*SDN0230C	6.
11	7 CAF02 B I 00	58C 82C	-13.0 -19.0	32 32	21•0 12•3	6•3 41•3	2•3 2•4	1.7	31 . 137.	38•5 40•6	2 [.] 2	-388. 233.	64•4 64•2	23.2 12.6	11.0 35.4	1. 3.	+ SCN 0232C + + NGR 0115C	5. 4.	*NGR 0115C * *NMB0025C	6. 15.	+ * *CAF0258C	6. 18.
17	7 CAF02 3 I 00 9 IRQ02	58C 82C 56C	-13.0 -19.0 11.0	32 32 32	21.0 12.3 43.6	6.3 41.3 32.8	2•3 2•4 1•9	1.7 1.0 1.0	31. 137. 143.	38.5 40.6 41.7	2 [.] 2 1	388. 233. 149.	64•4 64•2 63•4	23•2 12•6 44•0	11.0 35.4 37.3	1. 3.	×SCNJ232C × ×NGR 0115C × ×URS0066D	5. 4. 1.	+NGR 0115C + *NMB0025C + *KEN0249D	6. 15. 11.	+SDN0230C + *CAF0258C + *TGK0225C	6. 18. 12.
18	7 CAFO2 B I 00 D IRQ02 D LS003	58C 82C 56C 05C	-13.0 -19.0 11.0 5.0	32 32 32 32 32	21.0 12.3 43.6 27.8	6.3 41.3 32.8 -29.8	2•3 2•4 1•9 0•7	1.7 1.0 1.0	31. 137. 143. 36.	38.5 40.6 41.7 48.3	2 2 1 1	-388. 233. 149. 40.	64•4 64•2 63•4 64•3	23.2 12.6 44.0 29.5	11.0 35.4 37.3 -29.4	1. 3. -). 1.	* SCN J232C * * NGR 0115C * *UR S0066D * *NDR 01213	5. 4. 1. 3.	+NGR 0115C + *NMB0025C + *KEN0249D * *AFS0021D	6. 15. 11.	+SDN0230C + *CAF0258C * *TGK0225C *	6. 18. 12. 19.
17 18 19 20 21	7 CAFO2 3 I 00 9 IRQ02 0 LSD03 L MTN02	58C 82C 56C 05C 88C	-13.0 -19.0 11.0 5.0 -37.0	32 32 32 32 32 32	21.0 12.3 43.6 27.8 -7.8	6.3 41.3 32.8 -29.8 23.4	2.3 2.4 1.9 0.7 1.6	1.7 1.0 1.0).6 1.1	31. 137. 143. 36. 141.	38.5 40.6 41.7 48.3 41.7	2 2 1 1	-388. 233. 149. 40. 137.	64.4 64.2 63.4 64.3 63.1	23.2 12.6 44.0 29.5 -5.6	11.0 35.4 37.3 -29.4 20.0	1. 3. -J. 1. 3.	* SCN 3232C * * NGR 0115C * * UR S0066D * * NOR 01213 * * NGR 01215C	5. 4. 1. 3.	*NGR 0115C * *NMB0025C * *KEN0249D * *AFS0021D * * *SEN0222D	6. 15. 11. 5.	+SDN3233C + *CAF0258C * *TGK0225C * *NMB0025C * *CVA0083B	6. 18. 12. 19.
19 18 19 20 21 22	7 CAF02 3 I 00 9 IRQ02 0 LSD03 1 MTN02 2 MWI03	58C 82C 56C 05C 88C 08C	-13.0 -19.0 11.0 5.0 -37.0 -1.0	32 32 32 32 32 32 32	21.0 12.3 43.6 27.8 -7.8 34.1	6.3 41.3 32.8 -29.8 23.4 -13.0	2.3 2.4 1.9 0.7 1.6 1.5	1.7 1.0 1.0).6 1.1 0.6	31. 137. 143. 36. 141. 87.	38.5 40.6 41.7 48.3 41.7 44.6	2 2 1 1 1 2	-388. 233. 149. 40. 137. 95.	64.4 64.2 63.4 64.3 63.1 64.4	23.2 12.6 44.0 29.5 -5.6 35.2	11.0 35.4 37.3 -29.4 20.0 -17.2	1. 3. -J. 1. 3.	× SCN J232C × × NGR 0115C × ×UR S0066D × × NOR 01213 × × NGR 0115C × × ×	5. 4. 1. 3. 5.	*NGR 0115C * *NMB0025C * *KEN0249D * *AFS0021D * *SEN0222D * * *TGK0225C	6. 15. 11. 5. 14. 21.	*SDN323JC * *CAF0258C * *TGK0225C * *NMB0025C * *CVA0083B * *AGL0295C	6. 18. 12. 19. 14. 22.
17 18 19 20 21 22 23	7 CAFO2 3 I 00 9 IRQO2 0 LSD03 1 MTN02 2 MWI03 3 MYTG0	58C 82C 56C 05C 88C 08C	-13.0 -19.0 11.0 5.0 -37.0 -1.0 29.0	32 32 32 32 32 32 32 32	21.0 12.3 43.6 27.8 -7.8 34.1 45.1	6.3 41.3 32.8 -29.8 23.4 -13.0 -12.8	2.3 2.4 1.9 0.7 1.6 1.5 0.6	1.7 1.0 1.0).6 1.1 0.6	31. 137. 143. 36. 141. 87. 0.	38.5 40.6 41.7 48.3 41.7 44.6 48.7	2 2 1 1 1 2 1	388. 233. 149. 40. 137. 95. 30.	64.4 64.2 63.4 64.3 63.1 64.4 63.5	23.2 12.6 44.0 29.5 -5.6 35.2 45.0	11.0 35.4 37.3 -29.4 20.0 -17.2 -12.8	1. 3. -0. 1. 3. 10. 19.	*SCN J232C * NGR 0115C * *UR S0066D * *NDR 01213 * NGR 0115C * *AFS0021D * * * TGK0225C	5. 4. 1. 3. 5. 12.	*NGR 0115C * *NMB0025C * *KEN0249D * *AF30021D * *SEN0222D * *TGK0225C * *UR50066D	6. 15. 11. 5. 14. 21. 28.	+SDN3233C + *CAF0258C + TGK0225C * *NMB0025C * *CVAD083B * *AGL0295C *	 6. 18. 12. 19. 14. 22. 29.
17 18 19 20 21 22 23 23	7 CAFO2 3 I 00 9 IRQO2 0 LSD03 1 MTN02 2 MWI03 3 MYT03 4 NGR01	58C 82C 56C 05C 88C 08C 98C 15C	-13.0 -19.0 11.0 5.0 -37.0 -1.0 29.0 -25.0	32 32 32 32 32 32 32 32 32	21.0 12.3 43.6 27.8 -7.8 34.1 45.1 8.3	6.3 41.3 32.8 -29.8 23.4 -13.0 -12.8 16.8	2.3. 2.4 1.9 0.7 1.6 1.5 0.6 2.5	1.7 1.0 1.0).6 1.1 0.6 0.6 2.1	31. 137. 143. 36. 141. 87. 0. 44.	38.5 40.6 41.7 48.3 41.7 44.6 48.7 37.0	2 2 1 1 2 1 2 1 2	388. 233. 149. 40. 137. 95. 30. 573.	64.4 64.2 63.4 64.3 63.1 64.4 63.5 64.6	23.2 12.6 44.0 29.5 -5.6 35.2 45.0 13.9	11.0 35.4 37.3 -29.4 20.0 -17.2 -12.8 13.0	1. 3. -J. 1. 3. 10. 19. 4.	*SCN J232C * NGR 0115C * *UR S0066D * *NDR 01213 * * NGR 0115C * * AF S0021D * * TGK0225C *	5. 4. 1. 3. 5. 12. 24. 6.	*NGR 0115C * *NMB0025C * *KEN0249D * *AFS0021D * *SEN0222D * *TGK0225C * *URS0066D * *	 6. 15. 11. 5. 14. 21. 28. 13. 	+SDN323JC + *CAF0258C + *TGK0225C * *NMB0025C * *CVA00838 * *AGL0295C * *0MA0123C * *	 6. 18. 12. 19. 14. 22. 29. 14.
17 18 19 20 21 22 23 24 25	7 CAFO2 3 I 00 9 IRQ02 0 LSD03 1 MTN02 2 MWI03 3 MYTG0 4 NGR01 5 NGR01	58C 82C 56C 05C 88C 08C 98C 15C 21B	-13.0 -19.0 11.0 5.0 -37.0 -1.0 29.0 -25.0 5.0	32 32 32 32 32 32 32 32 32 32	21.0 12.3 43.6 27.8 -7.8 34.1 45.1 8.3 17.0	6.3 41.3 32.8 -29.8 23.4 -13.0 -12.8 16.8 61.5	2.3 2.4 1.9 0.7 1.6 1.5 0.6 2.5 2.0	1.7 1.0 1.0).6 1.1 0.6 0.6 2.1 1.0	31. 137. 143. 36. 141. 87. 0. 44. 10.	38.5 40.6 41.7 48.3 41.7 44.6 48.7 37.0 41.2	2 1 1 1 2 1 2 2 2	388. 233. 149. 40. 137. 95. 30. 573. 459.	64.4 64.2 63.4 64.3 63.1 64.4 63.5 64.6 67.9	23.2 12.6 44.0 29.5 -5.6 35.2 45.0 13.0 15.0	11.0 35.4 37.3 -29.4 20.0 -17.2 -12.8 13.0 55.0	1. 3. -J. 1. 3. 10. 19. 4.	*SDN 3232C * NGR 0115C * *UR S0066D * *NOR 01213 * NGR 0115C * * NGR 0115C * * AF S0021D * * TGK0225C * * *CAF 0258C * *	5. 4. 1. 3. 5. 12. 24. 6. 8.	*NGR 0115C * *NMB0025C * *KEN02490 * *AFS0021D * *SFN0222D * *GK0225C * *URS0066D * *MRC0209D * *	 6. 15. 11. 5. 14. 21. 28. 13. 8. 	+SDN323JC + *CAF0258C + *TGK0225C * *NMB0025C * *CVA00838 + *CVA00838 * *CVA00838 * *CVA00838 * *CVA00838 * *CVA00838 * * *CVA00838 * * * * * * * * * * * * * * * * * *	 6. 18. 12. 19. 14. 22. 29. 14. 15.
177 188 199 200 21 22 23 24 25 26	7 CAFO2 8 I 00 9 IRQ02 0 LSQ03 1 MTN02 2 MWI03 8 MYTG0 4 NGR01 5 NGR01 5 DMA01	58C 82C 56C 05C 88C 08C 98C 15C 21B 23C	-13.0 -19.0 11.0 5.0 -37.0 -1.0 29.0 -25.0 5.0 17.0	32 32 32 32 32 32 32 32 32 32 32	21.0 12.3 43.6 27.8 -7.8 34.1 45.1 8.3 17.0 55.6	6.3 41.3 32.8 -29.8 23.4 -13.0 -12.8 16.8 61.5 21.0	2.3 2.4 1.9 0.7 1.6 1.5 0.6 2.5 2.0 1.9	1.7 1.0 1.0).6 1.1 0.6 0.6 2.1 1.0	31. 137. 143. 36. 141. 87. 0. 44. 10. 100.	38.5 40.6 41.7 48.3 41.7 44.6 48.7 37.0 41.2 41.4	2 2 1 1 1 2 1 2 2 2 2	388. 233. 149. 40. 137. 95. 30. 573. 459. 157.	64.4 64.2 63.4 64.3 63.1 64.4 63.5 64.6 67.9 63.4	23.2 12.6 44.0 29.5 -5.6 35.2 45.0 13.0 15.0 56.0	11.0 35.4 37.3 -29.4 20.0 -17.2 -12.8 13.0 55.0 26.5	1. 3. -J. 1. 3. 10. 19. 4. 4. 4.	*SCN 3232C * *NGR 0115C * *UR S0066D * *NOR 01213 * *NGR 0115C * * NGR 0115C * * NGR 0215C * * CAF0258C * * DDR 0216D * *	5. 4. 1. 3. 5. 12. 24. 6. 8. 5.	*NGR 0115C * NMB0025C * *KEN0249D * *AFS0021D * *SEN0222D * *TGK0225C * *URS0066D * *MRC0209D * 1 0082C * *UAE0274G	 6. 15. 11. 5. 14. 21. 28. 13. 8. 7. 	*SDN323JC *CAF0258C *TGK0225C *TGK0225C *CVA0083B *CVA0083B *CVA0083B *CVA0083B *SDN0232C * *URS0066D *	 6. 18. 12. 19. 14. 22. 29. 14. 15. 7.
17 18 19 20 21 22 23 24 25 26 26 27	7 CAFO2 8 I 00 9 IRQ02 0 LS003 1 MTN02 2 MWI03 3 MYTG0 4 NGR01 5 NGR01 5 GMA01 7 SDN02	58C 82C 56C 05C 88C 08C 98C 15C 21B 23C 32C	-13.0 -19.0 11.0 5.0 -37.0 -1.0 29.0 -25.0 5.0 17.0 -7.0	32 32 32 32 32 32 32 32 32 32 32 32 32	21.0 12.3 43.6 27.8 -7.8 34.1 45.1 8.3 17.0 55.6 30.4	6.3 41.3 32.8 -29.8 23.4 -13.0 -12.8 16.8 61.5 21.0 19.0	2.3 2.4 1.9 0.7 1.6 1.5 0.6 2.5 2.0 1.9 2.4	1.7 1.0 1.0).6 1.1 0.6 0.6 2.1 1.0 1.0	31. 137. 143. 36. 141. 87. 0. 44. 10. 100. 176.	38.5 40.6 41.7 48.3 41.7 44.6 48.7 37.0 41.2 41.4 38.6	2 2 1 1 1 2 1 2 2 2 1	388. 233. 149. 40. 137. 95. 30. 573. 459. 157. 302.	64.4 64.2 63.4 64.3 63.1 64.4 63.5 64.6 67.9 63.4 63.4	23.2 12.6 44.0 29.5 -5.6 35.2 45.0 13.0 15.0 56.0 24.0	11.0 35.4 37.3 -29.4 20.0 -17.2 -12.3 13.0 55.0 26.5 15.8	1. 3. -3. 1. 3. 10. 19. 4. 4. 1. -0.	*SCN 3232C * *NGR 0115C * *UR S0066D * *NOR 01213 * NGR 0115C * * NGR 0115C * * NGR 0215C * * CAF0258C * * * * * * * * * * * * *	5. 4. 1. 3. 5. 12. 24. 6. 8. 5. 3.	*NGR 0115C * NMB0025C * KEN0249D * AFS0021D * SEN0222D * SEN0222D * TGK0225C * TGK0225C * TGK0225C * 0082C * + 0082C * + 0082C * + 0082C	 6. 15. 11. 5. 14. 21. 28. 13. 8. 7. 5. 	*SDN323JC * *CAF0258C * TGK0225C * *NMB0025C * *CVAD083B * *AGL0295C * * *0MA0123C * * SDN0232C * *URS0066D * *IRQ0256C *	 6. 18. 12. 19. 14. 22. 29. 14. 15. 7. 11.
17 18 19 20 21 22 23 24 25 26 27 28	7 CAFO2 8 I 00 9 IRQ02 0 LSD03 1 MTN02 2 MWI03 8 MYT03 4 NGR01 5 NGR01 5 NGR01 5 OMA01 7 SDN02 8 URS00	58C 82C 56C 05C 88C 08C 98C 15C 21B 23C 32C 660	-13.0 -19.0 11.0 5.0 -37.0 -1.0 29.0 -25.0 5.0 17.0 -7.0 44.0	32 32 32 32 32 32 32 32 32 32 32 32 32 3	21.0 12.3 43.6 27.8 -7.8 34.1 45.1 8.3 17.0 55.6 30.4 64.3	6.3 41.3 32.8 -29.8 23.4 -13.0 -12.8 16.8 61.5 21.0 19.0 44.6	2.3 2.4 1.9 0.7 1.6 1.5 0.6 2.5 2.0 1.9 2.4 4.6	1.7 1.0 1.0).6 1.1 0.6 0.6 2.1 1.0 1.0 1.5 2.5	 31. 137. 143. 36. 141. 87. 0. 44. 10. 100. 176. 169. 	38.5 40.6 41.7 48.3 41.7 44.6 48.7 37.0 41.2 41.4 38.6 33.7	2 2 1 1 1 2 2 2 2 1 2 2	388. 233. 149. 40. 137. 95. 30. 573. 459. 157. 302. 1522.	64.4 64.2 63.4 64.3 63.1 64.4 63.5 64.6 67.9 63.4 63.4 63.4	23.2 12.6 44.0 29.5 -5.6 35.2 45.0 13.0 15.0 56.0 24.0 87.3	11.0 35.4 37.3 -29.4 20.0 -17.2 -12.8 13.0 55.0 26.5 15.8 49.2	1. 3. -0. 1. 3. 10. 19. 4. 4. 1. -0. 2.	* SCN J232C * NGR 0115C * UR S0066D * UR S0066D * NOR 01213 * NGR 0115C * AF S0021D * TGK0225C * CAF 0258C * JR S0066D * CAF 0258C * JR S0075A	5. 4. 1. 3. 5. 12. 24. 6. 8. 5. 3. 2.	*NGR 0115C *NMB0025C *KEN0249D *AFS0021D *SEN0222D *TGK0225C *TGK0225C *TGK0225C *TGK0225C * *URS0066D * *URS0066D * * *URS0066D * * * * * * * * * * * * *	 6. 15. 11. 5. 14. 21. 28. 13. 8. 7. 5. 19. 	*SDN323JC *CAF0258C *TGK0225C *TGK0225C *CVAD083B *AGL0295C *CVAD083B *AGL0295C *CVAD083B *AGL0295C *CVAD083B *CVAD083B *CVAD083B *CVAD083B *CVAD083B *CVAD083B *CVAD083B *CVAD083C *CVAD085C *CVAD0	 6. 18. 12. 19. 14. 22. 29. 14. 15. 7. 11. 26.
17 18 19 20 21 23 24 25 26 27 28 29	7 CAF02 8 I 00 9 IRQ02 0 LSD03 1 MTN02 2 MWI03 8 MYT03 4 NGR01 5 NGR01 5 NGR01 5 OMA01 7 SDN02 8 URSD0 9 URSD0	58C 82C 56C 05C 88C 08C 98C 15C 21B 23C 32C 66D 75A	-13.0 -19.0 11.0 5.0 -37.0 -1.0 29.0 -25.0 5.0 17.0 -7.0 44.0 74.0	32 32 32 32 32 32 32 32 32 32 32 32 32 3	21.0 12.3 43.6 27.8 -7.8 34.1 45.1 8.3 17.0 55.6 30.4 64.3 94.0	6.3 41.3 32.8 -29.8 23.4 -13.0 -12.8 16.8 61.5 21.0 19.0 44.6 51.7	2.3 2.4 1.9 0.7 1.6 1.5 0.6 2.5 2.0 1.9 2.4 4.6 1.5	1.7 1.0 1.0).6 1.1 0.6 0.6 2.1 1.0 1.5 2.5).6	 31. 137. 143. 36. 141. 87. 0. 44. 10. 100. 176. 169. 172. 	38.5 40.6 41.7 48.3 41.7 44.6 48.7 37.0 41.2 41.4 38.6 33.7 44.6	2 2 1 1 2 1 2 2 1 2 2 1 2 2	388. 233. 149. 40. 137. 95. 30. 573. 459. 157. 302. 1522. 111.	64.4 64.2 63.4 64.3 63.1 64.4 63.5 64.6 67.9 63.4 63.4 65.5 65.1	23.2 12.6 44.0 29.5 -5.6 35.2 45.0 13.0 13.0 15.0 56.0 24.0 87.3 87.9	11.0 35.4 37.3 -29.4 20.0 -17.2 -12.3 13.0 55.0 26.5 15.8 49.2 51.5	1. 3. -J. 1. 3. 10. 19. 4. 4. 1. -0. 2. J.	* SCN 3232C * SCN 3232C * NGR 0115C * UR S0066D * NOR 01213 * NOR 01213 * NGR 0115C * AF S0021D * CAF 0258C * CAF 0258C * JR S0066D * CAF 0258C * JR S0075A * JR S0 366D	5. 4. 1. 3. 5. 12. 24. 6. 8. 5. 3. 2. 1.	*NGR 0115C * NMB0025C * *KEN02490 * *AFS0021D * *SFN0222D * *TGK0225C * *URS0066D * * *URS0066D * * * 0082C * * *NGR0115C * *MNG0248C * *MNG0248C	 6. 15. 11. 5. 14. 21. 28. 13. 8. 7. 5. 19. 8. 	*SDN323JC * *CAF0258C * *TGK0225C * *CVA00838 * *CVA00838 * *CVA00838 * *CVA00838 * *CVA00838 * *CVA00838 * *CVA00838 * *CVA00838 * *CVA00838 * * CVA0025C * * CVA00838 * * CVA00232C * * CVA00232C * * CVA00232C * * CVA00232C * * CVA007320 * * CVA00732C * * CVA007320 * * CVA00732C * * CVA00732C * * CVA00732C * * * CVA00732C * * * CVA00732C * * * CVA00732C * * * CVA00723C * * * * * CVA07770 * * CVA07770 * * CVA07770 * * CVA07770 * * * CVA07770 * * CVA07770 * * CVA07770 * * CVA07770 * * CVA07770 * * CVA07770 * * CVA07770 * * * CVA07770 * * * CVA07770 * * * * * * * * * * * * * * * * *	 6. 18. 12. 19. 14. 22. 29. 14. 15. 7. 11. 26. 23.

مېر فر. م د مرو د

L.-

1

Canal 31 suite — Channel 31 cont. — Canal 31 cont.

--- 36 ---

)

				,	ł												* *			
						(t i							(
Can	al 33 —	Chan	nel 3	3 –	- Can	al 33														
	1	2	3	4		5	6	7	8	9 a	9b	10		11	12		13		14	
15	AF10099D	23.0	33	42.5	11.6	0.6 0.6	0.	48.7	1	25 .	62•7	41.9	10.8	-1.	 *KEN0249D	2.	↓ *UKR0063B	6.	*ETH0092) *	6.
16	AF\$0021D	5.0	33	24.5	-28.0	3.1 1.7	27.	37.0	2	524.	64.2	20.0	-25.0	-1.	*NM80025C	-0.	*CYP 3086C	16.	*LS00305C	18.
17	BEL3018D	-19.0	33	4.6	50.6	0.8 0.6	167.	47.3	1	51.	64.4	6.4	50.3	-1.	*NMB0025C	4.	*DDR0216D	5.	*YUG0148D	7.
18	C¥P0086D	5•0	33	33.3	35.1	0.6 0.6	0.	48.7	1	32.	63.7	34.5	35.7	-0.	*AFS0D210	4.	*ISR D110 C *	5.	*SYR0229D *	8.
19	DDR3216D	-1.0	33	12.6	52 .1	0.8 0.6	172.	47.1	2	54.	64.4	11.0	54.0	-3.	*BEL0018D *	2.	*NOR0121B *	2.	*S 01380 *	3.
20	HV00107D	-31.0	33	-1.5	12.2	1.4 1.1	29.	42.1	1	160.	64.1	-5.5	12.0	2.	*SEN0222D *	7.	*CTIJ237D *	7.	*MTNJ223D *	10.
21	ISLJO49D	-31.0	33	-19.0	64.9	1.0 0.6	177.	46.5	2	89.	66.0	-13.5	65.1	9.	*HV001079 *	10.	*CTI0237D *	19.	*MRC 02090 *	20.
22	1SR0110C	-13.0	33	34.9	31.4	0.9 0.6	117.	46.7	2	53.	63.9	35.2	33•2	2.	*CYP0086D ≠	3.	*CAF0258C *	13.	*COGU235D *	14.
23	KEN02490	11.0	33	37.9	1-1	2.3 1.6	94.	38.7	1	329.	63.9	41.9	3.9	9.	*ETH0092D *	13.	*IRQ0256C *	14.	*SYR0229D *	15.
24	MC00116D	-37.0	33	7•4	43.7	0.6 0.6	0.	48.7	1	24.	62.6	5.6	46.0	-9.	*BEL0018D *	-4•	*MRC0209D *	-2.	*YUG01480 *	1.
25	MNG0248C	74.0	33	102.2	46.6	3.6 1.1	169.	38.2	1	403.	64.2	98.8	51.9	2.	*URS0074C *	4•	*URS0075A *	6.	*UR\$0066D *	18.
26	MRC0209D	-25.0	33	-9.0	29.2	2.7 1.5	43.	38.2	2	330.	63.4	-1.2	32.2	3.	*NGR0115C *	4.	*TUN0150D *	12.	*NMB0025C *	20.
27	NMB0025C	-19.0	33	17.5	-21.6	2.7 1.9	48.	37.2	2	580.	64.8	23.3	-28.4	-).	*AFS00210 *	-).	*BEL3018D *	14.	≭NIG0119 9 *	21.
28	SEN0222D	-37.0	33	-14.4	13.8	1.5 1.0	139.	42.4	2	137.	63.8	-15.0	16.3	з.	*MRC 0209D *	7.	*MTN0223D *	7.	*MTN0288C *	16.
29	UAE0274D	17.0	33	53.6	24.2	1.0 0.8	162.	45.3	1	64.	63.3	54.9	22.4	5.	*UMA0123C *	6.	*KEN0249D *	17.	*AF10099D *	20.
31	UKR0063B	23.0	33	31.2	48.4	2.3 1.0	172.	40.8	2	246.	64.7	22.1	48.4	-0.	*YUG 01 48D *	2.	*MC00116D *	8.	*ETH00920 *	13.
31	YUG0148D	-7.0	33	18.4	43•7	1.7 0.7	154.	43.8	1	144.	65•4	13.4	46.4	-2.	*DDR 02160	з.	*MC00116D	3.	*HNG0106J	9.

Canal 34 — Channel 34 — Canal 34

		1	2	3	4		5	6	7	8	9 a	9 b	10		11	12		13		14	
	 18	ALB0296D	-7.0	34	19.8	41.3	0.7 3.6	146.	48.1	2	38.	63.9	19.8	42.6	-2.	*HNG0106D	2.	*SDN02310	5.	*YUG0149D	6.
	19	B010270D	11.0	34	29.9	-3.1	0.7 0.6	80.	48.0	2	36.	63.5	29.1	-2.7	3.	*TGK 02250 *	7.	*ETH0092D	11.	*SYR0229D *	11.
	20	COG0235D	-13.0	34	14.6	-0.7	2.0 1.2	59.	40.5	2	222.	63.9	12.6	2.4	-1.	*NIG0119D *	0.	*GNE0303D *	5∙્	*4GL02950 *	10.
	21	CT10237D	-31.0	34	-5.6	7.5	1.6 1.2	108.	41.3	2	178.	63.9	-3.5	9.8	-1.	*NIG0119D *	4.	*MTN0223D *	4.	*HV00107D *	5.
	22	ETH3092D	23.0	34	39.7	9.1	3.5 2.4	124.	35.0	2	725.	63.6	39.2	17.5	2.	*SDN0231D *	3.	*URS0059C *	11.	≠UKR00638 *	14.
	23	HNG0106D	-1.0	34	19.5	47.2	0.9 0.6	176.	46.8	1	54.	64.1	16.1	46.8	-3.	*YUG0149D *	4.	*YUG0148D *	4.	*SUI914JD *	4.
ł	24	KWT0113D	17.0	34	47.6	29.2	0.7 0.6	145.	48.1	2	32.	63.2	46.3	28.3	1.	*ETH00920	2.	*UA20274D	15.	≠URS0059C	16.
											_ :	37 —						SL	uite	- cont.	

Canal 34 suite --- Channel 34 cont. --- Canal 34 cont.

		1	2	3	4		5	6	7	8	9 a	9 b	10		11	12		13		14]
Z	25 1	T NO 2 2 3 D	-37.0	34	-12.2	18.5	2.6 1.9	150.	37•3	1	364.	63.0	-5.2	15.3	0.	*CTI0237D	2.	*HV30107D	7.	*NIG01197	12.
2	26 1	NIG0119D	-19.)	34	7.8	9•4	2.2 2.9	45.	37.9	1	417.	64.1	3.4	6.4	4.	*CTI0237D	8.	*GHA0108D	9.	*SUI 01400	12.
4	27 6	REU0097D	29.0	34	55.6	-19.2	1.6 9.8	96.	43.4	1	118.	64.1	54.7	-15.7	16.	*ETH0J92D	17.	*RHS0135D	29.	*URS0074C	3).
2	28 6	RHS0135D	-1.0	34	29.6	-18.8	1.5 1.4	37.	41.3	2	202.	64.3	31.2	~22•2	6.	+ ≉≜FS00210	7.	* *HNG0106D	13.	*NMB0025C	19.
2	29 :	\$ 0 138 C	5.0	34	16.2	61.0	1.0 1.0	14.	44.2	2	215.	67.5	13.3	55.3	5.	+ +DDP0216D	8.	*SUI01400	12.	+ *DNK0091B	16.
1	รับเ	5DNJ2310	-7.0	34	28.9	12.7	2.3 2.0	159.	37.8	1	386.	63.6	34.0	8.5	-2.	+ *ETH0092D	-0.	+ *SDN0230D	6.	+ *AL30296J	13.
1	31	SUI 0140D	-19.0	34	8.2	46.6	1.0 0.7	171.	45•9	2	69.	64.3	10.4	46.9	2.	*NIG0119D	6.	*S 0138C	10.	*HNG0106D	12.
3	62	SYR0229D	11.0	34	38.3	34.9	1.0 0.9	7.	44.5	1	77.	63.4	35.9	35.7	2.		5.	+ *JOR0224D	10.	*7GK02250	12.
-	53 ⁻	TUN0150D	-25.0	34	9.5	33.5	1.9 0.7	153.	42.9	1	128.	64.0	9.0	38.0	3.	*:4RC0209D	9.		10.	≁ ≠CVA0083C	11.
13	34 !	JRS0071A	44.0	34	63.1	42.0	2.6 0.8	170.	40.8	2	230.	64•4	73.1	40.8	9.	*JRS0074C	9.	*UR\$0080C	25.	*ETH00920	29.
3	35	JR 500 74C	74.0	34	88.8	57.6	3.1 1.7	162.	37.1	2	1241.	68.0	97.2	49.7	9.	*MNG0248C	9.	*URS0077E	21.	*URS0071A	22.
	36	URSOOBOC	140.0	34	155.3	55.4	2.9 2.4	35.	35.9	1	1624.	68.0	145.6	63.8	23.	* ≠JRS0077E	23.	* *AF100 99D	89.	* *YUG01490	89.
	Cana	a 35 -	— Cha	nnel	35 -	- Cai	nal 35	4													
	Can	a 35 - 1	— Cha 2	nnel	35 - 4	– Car	nal 35	6	7	8	9 a	9b	10	,	11	12	<u></u>	13		14	
	Cana 2 J	al 35 - <u>1</u> AGL0295D	— Cha <u>2</u> -13.3	nnel 3 35	35 - <u>4</u>	- Car	nal 35 5 3.1 2.3	6 84.	7 35•8	8	9 a 707.	9 b 64•3	10 12•5	-6.5	<u>11</u> 7.	12 *CGG0235D	9.	13 *CAF0258D	15.	14 *SDN0230D	24.
	2 J	al 35 - 1 AGL0295D BHR0255C	- Cha 2 -13.3 17.0	nnel 3 35 35	35 - 4 16.5 50.5	- Car -12.0 26.1	nal 35 5 3.1 2.3 0.6 0.6	6 84. 0.	7 35.8 48.7	8 1 1	9 a 707. 17.	9 b 64.3 61.0	10 12•5 52•3	-6.5 24.4	<u>11</u> 7. -3.	12 *COGO235D * *OMA 01 23D	9. J.	13 *CAF0258D * *URS0059C	15.	14 *SDN0230D * *IRQ3256D	24.
	Cana 2 J 2 J 2 1 2 2	A 35 - 1 AGL0295D BHR0255C CNR0130D	- Cha 2 -13.3 17.0 -31.0	nnel 35 35 35	35 - 4 16.5 50.5 -15.7	- Can -12.0 26.1 28.4	nal 35 5 3.1 2.3 0.6 0.6 1.5 0.6	6 84. 0. 5.	7 35.8 48.7 44.6	8 1 1 2	9 a 707. 17. 69.	9 b 64.3 61.0 63.0	10 12.5 52.3 -13.5	-6.5 24.4 29.2	11 7. -3. -13.	12 *C0G0235D * *0MA 0123D * *∈ 0129D-	9. 0. -13.	13 *CAF0258D * *URS0059C * *CTI0237D	15. 6. 13.	14 *SDN0230D * *IRQ3256D * *SRL0259D	24. 7. 13.
	Cana 2 J 2 J 2 1 2 2 2 3	A 35 AGL02950 BHR0255C CNR01300 CVA0083C	- Cha 2 -13.3 17.0 -31.0 -37.0	nnel 35 35 35 35	35	- Car -12.0 26.1 28.4 41.8	mai 35 5 3.1 2.3 0.6 0.6 1.5 0.6 0.6 0.6	6 84• 0• 5• 0•	7 35.8 48.7 44.6 48.7	8 1 1 2 1	9 a 707. 17. 69. 46.	9b 64.3 61.0 63.0 65.3	10 12.5 52.3 -13.5 12.5	-6.5 24.4 29.2 43.8	11 7. -3. -13. -3.	12 *C0G02350 * *0MA 01230 * *€ 01290- * * YUG01490	9. J. -13. -1.	13 + CAF0258D + UR S0059C + + CTI 0237D + + 5. 0129D +	15. 6. 13.	14 *SDN0230D * *IRQ3256D * *SRL0259D * *URS0059C	24. 7. 13. 15.
	Cana 20 21 22 23 23 24	al 35 - 1 AGL02950 BHR0255C CNR01300 CVA0083C DNK0391B	- Cha 2 -13.0 17.0 -31.0 -37.0 5.0	nnel 35 35 35 35 35 35	35	- Can -12.0 26.1 28.4 41.8 61.0	35 3.1 2.3 0.6 0.6 1.5 0.6 0.6 0.6 2.2 0.8	6 84• 0• 5• 0• 4•	7 35.8 48.7 44.6 48.7 41.8	8 1 2 1	9 a 707. 17. 69. 46. 282.	9 b 64.3 61.0 63.0 65.3 66.3	10 12.5 52.3 -13.5 12.5 -6.2	-6.5 24.4 29.2 43.8 62.3	11 7. -3. -13. -3. 7.	12 *C0G0235D * *0MA0123D * *C0MA0123D * *C0MA0123D * <td< th=""><th>9. J. -13. -1. 10.</th><th>13 *CAF0258D *URS0059C *CTI0237D * *E 0129D * *URS0059C</th><th>) 15. 6. 13. 2. 12.</th><th>14 *SDN0230D * *IRQ3256D * *SRL0259D * *URS0059C * * S 0138C</th><th>24. 7. 13. 13. 14.</th></td<>	9. J. -13. -1. 10.	13 *CAF0258D *URS0059C *CTI0237D * *E 0129D * *URS0059C) 15. 6. 13. 2. 12.	14 *SDN0230D * *IRQ3256D * *SRL0259D * *URS0059C * * S 0138C	24. 7. 13. 13. 14.
	20 21 22 23 24 25	A 35 AGL0295D BHR0255C CNR0130D CVA0083C DNK0091B S 0129D	- Cha -13.3 17.0 -31.0 -37.0 5.0 -31.0	nnel 35 35 35 35 35 35 35 35	35 - 4 16.5 50.5 -15.7 12.4 -19.5 -3.1	- Can -12.0 26.1 28.4 41.8 61.0 39.9	5 3.1 2.3 0.6 0.6 1.5 0.6 0.6 0.6 2.2 3.8 2.1 1.1	6 84• 0• 5• 0• 4• 154•	7 35.8 48.7 44.6 48.7 41.8 40.5	8 1 1 2 1 1 2	9a 707. 17. 69. 46. 282. 231.	9b 64.3 61.0 63.0 65.3 66.3 64.1	10 12.5 52.3 -13.5 12.5 -6.2 -2.9	-6.5 24.4 29.2 43.8 62.3 35.3	11 7. -3. -13. -3. 7. -8.	12 ★COGO235D * *DMA0123D *	9. 0. -13. -1. 10. -8.	13 *CAF0258D * *URS0059C * *CTI0237D * * * 0129D * *URS0059C * *URS0059C) 15. 6. 13. 2. 12.	14 *SDN0230D * *IRQ3256D * *SRL0259D * *URS0059C * *S 0138C * * *SFL 3259J	24. 7. 13. 15. 14.
	20 21 22 23 24 25 26	A 35 - AGL0295D BHR0255C CNR0130D CVA0083C DNK0391B 3 0129D GHA0108D	- Cha 2 -13.3 17.0 -31.0 -37.0 5.0 -31.0 -31.0 -25.0	nnel 35 35 35 35 35 35 35	35 - 4 16.5 50.5 -15.7 12.4 -19.5 -3.1 -1.2	- Can -12.0 26.1 28.4 41.8 61.0 39.9 7.9	5 3.1 2.3 0.6 0.6 1.5 0.6 0.6 0.6 2.2 0.8 2.1 1.1 1.5 1.1	6 84• 0• 5• 0• 4• 154• 102•	7 35.8 48.7 44.6 48.7 41.8 40.5 42.3	8 1 1 2 1 1 2 1	9 a 707. 17. 69. 46. 282. 231. 141.	9b 64.3 61.0 63.0 65.3 66.3 64.1 63.8	10 12.5 52.3 -13.5 12.5 -6.2 -2.9 0.1	-6.5 24.4 29.2 43.8 62.3 35.3 11.0	11 7. -3. -13. -3. 7. -8. 3.	12 *COGO235D * *OMA 01 23D * *COGO129D * *YUG0149D * *DNK0090B * *CNR 0130D * *NGR 0115D	9. J. -13. -1. 10. -8. 7.	13 *CAF0258D * *URS0059C * *CTI0237D * *URS0059C * *URS0059C * *CTI3237D *) 15. 6. 13. 2. 12. 13.	14 *SDN0230D * *IRQ3256D * *SRL0259D * *URS0059C * *SRL3259D * *SRL3259D *	24. 7. 13. 15. 14. 15.
	Cana 23 21 22 23 24 25 26 27	A 35 AGL02950 BHR0255C CNR01300 CVA0083C DNK0091B 3 01290 GHA01080 GNE03030	- Cha 2 -13.0 -31.0 -31.0 -37.0 5.0 -31.0 -25.0 -19.0	nnel 35 35 35 35 35 35 35 35 35	35 4 16.5 50.5 -15.7 12.4 -19.5 -3.1 -1.2 10.3	- Car -12.0 26.1 28.4 41.8 61.0 39.9 7.9 1.5	3.1 2.3 0.6 0.6 1.5 0.6 2.2 0.8 2.1 1.1 1.5 1.1 0.7 0.6	6 84. 0. 5. 0. 4. 154. 102. 10.	7 35.8 48.7 44.6 48.7 41.8 40.5 42.3 48.1	8 1 1 2 1 1 2 1 2	9 a 707. 17. 69. 46. 282. 231. 141. 38.	9b 64.3 61.0 63.0 65.3 66.3 64.1 63.8 63.9	10 12.5 52.3 -13.5 12.5 -6.2 -2.9 0.1 11.3	-6.5 24.4 29.2 43.8 62.3 35.3 11.0 2.4	11 7. -3. -13. -3. 7. -8. 3. 0.	12 *C0G0235D * *0MA 0123D * *C0MA 0123D * *C0MA 0123D *	9. J. -13. -1. 10. -5. 7. 4.	13 +CAF0258D + *URS0059C + CTI0237D + URS0059C + URS0059C + + URS0059C + + CTI3237D + NIG0119D + CCG0235D) 15. 6. 13. 2. 12. 13. 8. 8.	14 *SDN0230D * FIRQ3256D * SRL0259D * URS0059C * S 0138C * SRL 3259D * *SRL 3259D * *NIG0119D	24. 7. 13. 15. 14. 13. 14. 9.
	Cana 20 21 22 23 24 25 26 27 28	A GL02950 BHR0255C CNR0130D CVA0083C DNK0091B E 0129D GHA0108D GNE0303D HDL0213D	- Cha 2 -13.3 17.0 -31.0 -37.0 5.0 -31.0 -31.0 -31.0 -19.0 -19.0	nnel 35 35 35 35 35 35 35 35 35 35	35 4 16.5 50.5 -15.7 12.4 -19.5 -3.1 -1.2 10.3 5.4	- Car -12.0 26.1 28.4 41.8 61.0 39.9 7.9 1.5 52.0	35 5 3.1 0.6 1.5 0.6 0.6 0.6 0.6 0.6 0.6 1.5 0.6 0.6 0.6 0.6 0.6 0.6 0.6 0.6 0.6 0.7 0.8 0.8	6 84. 0. 5. 0. 4. 154. 102. 10. 171.	7 35.8 48.7 44.6 48.7 41.8 40.5 42.3 48.1 47.7	8 1 1 2 1 1 2 1 2 1	9 a 777. 17. 69. 46. 282. 231. 141. 38. 50.	9b 64.3 61.0 63.0 65.3 66.3 64.1 63.8 63.9 64.6	10 12.5 52.3 -13.5 12.5 -6.2 -2.9 0.1 11.3 6.2	-6.5 24.4 29.2 43.8 62.3 35.3 11.0 2.4 50.6	11 7. -3. -13. -3. 7. -8. 3. 0. 9.	12 *C0G0235D * *OMA 0123D * *CMA 0123D * * * YUG0149D * *DNK0090B * *CNR 0130D * *NGR 0115D * *AGL 0295D * *	9. J. -13. -1. 10. -8. 7. 4. 5.	13 *CAF0258D *URS0059C *CTI0237D * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *URS0059C * * *URS0059C * * * * * * * * * * * * *) 15. 6. 13. 2. 12. 13. 8. 8. 7.	14 *SDN0230D * FIRQ3256D * SRL0259D * URS0059C * SO138C * * SKL3259J * NIG0119D * SUI0140D	24. 7. 13. 13. 14. 13. 14. 13. 14. 13. 14. 13.
	Cana 20 21 22 23 24 25 26 27 28 29	A 35 AGL0295D BHR0255C CNR0130D CVA0083C DNK0091B S 0129D GHA0108D GNE0303D H0L0213D J0R0224D	- Cha 2 -13.3 17.0 -31.0 -37.0 5.0 -31.0 -25.0 -19.0 11.0	nnel 35 35 35 35 35 35 35 35 35 35	35 - 4 16.5 50.5 -15.7 12.4 -19.5 -3.1 -1.2 10.3 5.4 35.8	- Can -12.0 26.1 28.4 41.8 61.0 39.9 7.9 1.5 52.0 31.4	Jail 35 3.1 2.3 0.6 0.6 1.5 0.6 0.6 0.6 2.2 0.8 2.1 1.1 1.5 1.1 0.7 0.6 0.8 0.6	6 84- 0- 5- 0- 4- 154- 102- 10- 171- 114-	7 35.8 48.7 44.6 48.7 41.8 40.5 42.3 48.1 47.7 46.1	8 1 1 2 1 1 2 1 2 1 2	9a 707. 17. 69. 46. 282. 231. 141. 38. 50. 52.	9b 64.3 61.0 63.0 65.3 66.3 64.1 63.8 63.9 64.6 63.2	10 12.5 52.3 -13.5 12.5 -6.2 -2.9 0.1 11.3 6.2 35.0	-6.5 24.4 29.2 43.8 62.3 35.3 11.0 2.4 50.6 34.0	11 7. -3. -13. -3. 7. -8. 3. 0. 0. -4.	12 *COGO235D *COMA 01 23D *COMA 01 23D *COMA 01 23D *E *DNK 0090B * *COR 01 30D * *NGR 01 15D * *AGL 02 95D *E 0129D * *URS0059C	9. J. -13. -1. 10. -8. 7. 4. 5. J.	13 *CAF0258D * *CTI0237D * *CTI0237D * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *URS0059C * * *URS0059C * * * * * * * * * * * * * * * * * * *) 15. 6. 13. 2. 12. 13. 8. 8. 7. 1.	14 *SDN0230D * *SRL0259D * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *SUI01259D * * *SFL0259D * * * SFL0259D * * * * STUN01500 * * * * * * * * * * * * * * * * * *	24. 7. 13. 13. 14. 13. 14. 13. 16. 10.
	20 21 22 23 24 25 26 27 28 29 30	A 35 - AGL02950 BHR0255C CNR01300 CVA0083C DNK0391B E 01290 GHA01080 GNE03030 HDL02130 JOR02240 SDN02300	- Cha 2 -13.0 17.0 -31.0 -37.0 5.0 -31.0 -25.0 -19.0 11.0 -7.0	nnel 35 35 35 35 35 35 35 35 35 35	35 - 4 16.5 50.5 -15.7 12.4 -19.5 -3.1 -1.2 10.3 5.4 35.8 29.2	- Can -12.0 26.1 28.4 41.8 61.0 39.9 7.9 1.5 52.0 31.4 7.5	Jail 35 5 3.1 2.3 0.6 0.6 0.6 1.5 0.6 0.6 2.2 0.8 2.1 1.5 1.1 1.5 0.7 0.6 0.8 0.8 0.8 0.8 2.3 1.1 1.1	6 84. 0. 5. 0. 4. 154. 102. 10. 171. 114. 148.	7 35.8 48.7 44.6 48.7 41.8 40.5 42.3 48.1 47.7 46.1 40.1	8 1 1 2 1 1 2 1 2 1 2 2	9a 707. 17. 69. 46. 282. 231. 141. 38. 50. 52. 282.	9b 64.3 61.0 63.0 65.3 66.3 64.1 63.8 63.9 64.6 63.2 64.6	10 12.5 52.3 -13.5 12.5 -6.2 -2.9 0.1 11.3 6.2 35.0 23.0	-6.5 24.4 29.2 43.8 62.3 35.3 11.0 2.4 50.6 34.0 11.0	11 7. -3. -13. -3. 7. -8. 3. 0. 0. 9. -4. 1.	12 *COGO235D * *DMA 01 23D * * 01 29D * * YUG 01 49D * *DNK 00 90B * *CNR 01 30D * * *NGR 01 15D * * * * * * * * * * * * *	9. J. -13. -1. 10. -8. 7. 4. 5. J. 6.	13 *CAF0258D * *URS0059C * *CTI0237D * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *URS0059C * * URS0059C * * URS0059C * * URS0059C * * URS0059C * * URS0059C * * URS0059C * * * URS0059C * * * URS0059C * * * URS0059C * * * URS0059C * * * URS0059C * * * URS0059C * * * * URS0059C * * * * * URS0059C * * * * * * * * * * * * * * * * * * *) 15. 6. 13. 12. 13. 13. 13. 13. 13. 13. 13. 13	14 *SDN0230D * *IRQ3256D * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *URS0059C * *SRL3259D * *SRL3259D * * *SRL3259D * * *SRL3259D * * * * * * * * * * * * * * * * * * *	24. 7. 13. 13. 14. 13. 14. 13. 14. 13. 14. 13. 14. 13. 14. 15. 10. 15. 10.

- 38 --

١,

suite — cont.

Canal 35 suite -- Channel 35 cont. -- Canal 35 cont.

	I	1	2	3	4		5	6	7	8	9 a	9 b	10		11	12	13	14
	32	rgk02250	11.0	35	34.6	-6.2	2•4 1•7	129.	38.1	1	380.	63.9	30.5	-1.0	3.	 *BDI0270D 7.	*AGL0295D 10.	*SDN02300 10.
	33	UR \$0059C	23.0	35	36.0	47.0	3.7 1.4	153.	37.0	2	668.	65.3	22.0	48•4	-1.	*YUG01490 0.	*CVA0083C 6.	* *ETH00920 12. *
	34	JR\$0077E	110.0	35	112.7	57•3	2.7 1.8	2.	37.6	1	944.	67.3	106.2	69.5	15.	*URS0074C 16.	*URS0066E 22. *	*URS0079∃ 22. *
1	35	YUG0149D	-7.0	35	18.4	43.7	1.7 0.7	154.	43.8	1	145.	65.4	20.1	46.3	-3.	*CVA0083C 2.	*URS0059C 4.	*HNG0106D 5.

Ţ

Canal 36 — Channel 36 — Canal 36

Γ		1	2	3	4		5	6	7	8	9 a	9 b	10		11	12		13		14	
	17	CAF3258D	-13.0	36	21.0	6.3	2.3 1.7	31.	38.5	2	393.	64.4	23.2	11.0	1.	*SDN0232D	5.	 ≭NGR0115D *	6.	*SDN02300	6.
	L 8	DNK0099B	5.0	36	17.0	61.5	2.0 1.0	10.	41.2	2	491.	68.2	12.0	54.6	4.	*DDR0216E *	8.	*1 0082D *	8,	*DNK0091B *	16.
	19	I 0082D	-19.0	36	12.3	41.3	2.4 1.0	137.	40.6	2	236.	64.3	12.6	35.4	3.	*NGR 91150 *	4.	*NMB0025D *	15.	*CAF02580 *	18.
	20	IRQ0256D	11.0	36	43.6	32.8	1.9 1.0	143.	41.7	1	151.	63.5	44.0	37.3	-0.	≠URS0066E *	1.	*KEN0249E *	11.	*TGK0225D *	12.
	21	LSG3305D	5.0	36	27.8	-29.8	0.7 0.6	36.	48.3	1	40.	64.3	29.5 -	-29.4	1.	*DNK00908 *	3.	≠AFS0021E *	5.	*NMB0025D *	19.
	22	MTN0288D	-37.0	36	-7.8	23.4	1.6 1.1	141.	41.7	1	139.	63.1	-5.6	20.0	3.	*NGR 0115D *	5.	*SEN0222E *	14.	*CVA0083C *	14.
	23	MWI0308D	-1.0	36	34.1	-13.0	1.5 0.6	87.	44.6	2	96.	64.4	35.2 -	-17.2	10.	*AFS0021E *	12.	*TGK02250 *	21.	*AGL0295D *	22.
	24	MYT0098D	29.0	36	45.1	-12.8	0.6 0.6	0.	48.7	1	31.	63.6	45.0 -	-12.8	19.	*TGK0225D *	24.	*URS0066E *	29.	*0MA0123D *	29.
	25	NGR0115D	-25.0	36	8.3	16.8	2.5 2.1	44.	37.0	2	580.	64.7	13.0	13.0	4.	*CAF0258D *	6.	≠MFC0209E *	13.	*SDN0232D *	14.
	25	OMA01230	17.0	36	55.6	21.0	1.9 1.0	100.	41•4	2	159.	63.4	56.0	26.5	1.	*URS30665 *	6.	*UAE0274E *	7.	*IRQ02560 *	7.
	27	SDN0232D	-7.0	36	30•4	19.0	2.4 1.5	176.	38.6	1	306.	63.4	24.0	15.8	-0.	*CAF0258D *	3.	*NGR0115D *	5.	*SDN02300 *	11,+
	28	UR\$0066E	44.0	36	64.3	44.6	4.6 2.5	169.	33.7	2	1541.	65.6	53.9	37.3	8.	*IRQ0256D *	9.	*NGR0115D *	17.	*URS00590 *	22.
	29	UR \$0079E	140.0	36	138.0	53.6	3.2 2.1	62.	36.0	2	1574.	68.0	128.4	73.2	18.	*URS0077E	18.	*MNG0248D	35.	*YUG0148:	89.

Canal 37 — Channel 37 — Canal 37

	1	2	3	4	5	6	7	8	9 a	9 b	10	11	12	13	14
Γ															
1	4 AFI0099E	23.0	37	42.5 11.	5 0.6 0.6	0.	48.7	1	25.	62.7	41.9 10.8	-1.	*KEN0249E 2. *	*UKR0963C 6. *	¥ETH0092⊾ 6. *
1	5 AFSJ021E	5.0	37	24.5 -28.	3.1 1.7	27.	37.0	2	531.	64.3	20.0 -25.0	-1.	*NMB0025D -0.	*CYP0086E 16.	*LSD0305N 18. *
1	6 BEL0018E	-19.0	37	4.6 50.	5 0.8 0.6	167.	47.3	1	51.	64•4	6.4 50.3	-1.	*NMBJ0255 4. *	*00R02168 5.	*YUG0148⊑ 7. *
1	7 CYP0086E	5.0	37	33.3 35.	1 0.6 0.6	0.	48.7	1	32.	63.8	34.5 35.7	-0.	*AFS0021E 4.	*ISR0110D 5.	*SYR0339A 8.

Canal 37 suite — Channel 37 cont. — Canal 37 cont.

	1	2	3	4		5	6	7	8	9 a	9 b	10		11	12		13		14	
18	DDR0216E	-1.0	37	12.6	52.1	0.8 0.6	172.	47.1	2	55.	64.4	11.J	54.0	-3.	≠BEL00185 *	2.	*DNK0090B	2.	*NORG1200	6.
19	HV001075	-31.0	37	-1.5	12.2	1.4 1.1	29.	42.1	1	162.	64.2	-5.5	12.0	2.	*SEN0222E	7.	*CTI9237E	7.	*MTN0223E	10.
23	ISLJ049E	-31.0	37	-19.0	64.9	1.0 0.6	177.	46.5	2	90.	66.0	-13.5	65.1	9.	*HV00107E	10.	*CTI0237E	19.	*MRC0209E	2û.
21	ISROIIOD	-13.0	37	34.9	31.4	0.9 0.6	117.	46.7	2	53.	64.0	35.2	33.2	2.	*CYP00865 *	3.	*CAF0258D	13.	*COG02355	14.
22	KEN0249E	11.0	37	37.9	1.1	2.3 1.6	94.	38.7	1	333.	63.9	34.0	4•2	9.	*S¥R0339A	14.	*ETH0092E	15.	*1RQ02560	15.
23	MC001163	-37.0	37	7.4	43.7	0.6 0.6	0.	48.7	1	25.	62.6	5.6	46.0	-9.	*BEL0018E	~4•	*MRC0209E	-2.	*YUG0148≓	1.
24	MNG0248D	74.0	37	192.2	46•6	3.6 1.1	169.	38.2	1	409.	64.3	87.9	48•9	3.	*URS0074D	3.	*UP \$0066E	17.	+UKR 0063C	29.
25	MRC02095	-25.0	37	- 9 .0	29•2	2.7 1.5	43.	38.2	2	334.	63.5	-1.2	32.2	2.	*NGR0115D	4.	*TUN0272A	7.	*NMB0025D	2.∔∎
26	NMB0025D	-19.0	37	17.5	-21.6	2.7 1.9	48.	37.2	2	587.	64•9	20.0	-28.4	-0.	*AFS0021E	-0.	+ *BEL0018⋶ *	14.	*NIG0119E	21.
27	SEN0222E	-37.0	37	-14.4	13-8	1.5 1.0	139.	42.4	2	139.	63.9	-15.0	16.3	3.	*MRC02095	7.	*MTN02236	7.	*MTN0288D	16.
28	UAE0274E	17.0	37	53.6	24•2	1.0 0.8	162.	45.3	1	64.	63.4	54.9	22.4	5.	+ ≄⊡MA01230 *	6.	* *KEN0249E	17.	*AF100995	20.
29	UKR0063C	23.0	37	31.2	48.4	2.3 1.0	172.	40.8	2	249.	64.7	22.1	48.4	-0.	*YUG0148E	2.	*MCB0116E	8.	*E7H0092E	13.
30	YUG01485	-7.0	37	18•4	43•7	1.7 0.7	154.	43.8	1	146.	65•4	13.4	46.4	-2.	*DDR 0216E	3.	≠MCO0116E	3.	*HNG01065	9.

Canal 38 — Channel 38 — Canal 38

	1	2	3	4		5	6	7	8	9 a	9 b	10		11	12		13		14
18	ALB0296E	-7.0	38	19.8	41.3	0.7 0.6	146.	48.1	2	39.	64• 0 [′]	19.8	42.6	-2.	*HNG01065	2.	*SDN02316	5.	*YUGJ149: 6.
19	BD102705	11.0	38	29.9	-3.1	0.7 0.6	80.	48.0	2	36.	63.6	29.1	-2•7	з.	*TGK0225č	7.	*SYRJ339A	9.	*ETHJ092E 11.
20	COG0235E	-13.0	38	14.6	-0.7	2.0 1.2	59.	40.5	2	225.	64.0	12.6	2.4	-1.	*NIG0119E	0.	*GNE03038	5.	*AGL02958 10.
21	CT10237E	-31.0	38	-5.6	7.5	1.6 1.2	108.	41.3	2	180.	63.9	-3.5	9.8	-1.	*NIG0119E	4.	*MTN0223E	4.	+ *HVG0107€ 6.
22	ETH0092E	23.0	38	39.7	9.1	3.5 2.4	124.	35.0	2	734.	63.7	39.2	17.5	2.	*SDN0231F	3.	+ +URS0059D	11.	*UKR0063C 14.
23	HN GD 106E	-1.0	38	19.5	47•2	0.9 0.6	176.	46.8	1	55.	64.2	16.1	46.8	-2.	*YUG0149E	4.	*YUG0148E	4.	*SUI01405 4.
24	KWT0113E	17.0	38	47.6	29.2	0.7 0.6	145.	48.1	2	33.	63.3	46 •3	28.3	1.	*ETH0092E	2.	*TUN02724	15.	*UAE0274E 15.
25	MTN0223E	-37.0	38	-12.2	18.5	2.6 1.9	150.	37.3	1	369.	63.0	-5.2	15.3	0.	*CT10237E	2.	≠нV80107E ≠	7.	* *NIG01195 12• *
26	NIG0119E	-19.0	38	7.8	9•4	2.2 2.0	45.	37.9	1	422.	64.1	3.4	6.4	4.	*CTI0237E	8.	*GHA0108E	10.	*SUI01403 12.
27	NOR0120C	5.0	38	13.1	64.1	1.8 0.9	10.	42.2	2	304.	67.0	11.6	59.0	10.	*DDR 0216E	13.	*1SL0050C	15.	*AFS0021.7 23.
28	R EU0097E	29.0	38	55.6	-19.2	1.6 0.8	96.	43.4	1	119.	64.2	54.7 -	-15.7	16.	*:TH0092E	17.	*RHS0135E	29.	*UR500740 3J.

suite — cont.

Canal	38	suite	 Channel	38	cont.	 Canal	38	cont.
Quittui	~~	aurto	onunnun	~~	00110	Contrait	~ ~	oone.

	1	2	3	4		5	6	7	8	9 a	9 b	10)	11	12		13		14	
29	RHSD135E	-1.0	38	29.6 -	-18.8	1.5 1.4	37.	41.3	2	205.	64.4	31.2	-22.2	6.	*AFS0021E	7.	≭HNG0106E	13.	*NMB00250	19.
30	SDN0231E	-7.0	38	28.9	12.7	2.3 2.0	159.	37.8	1	390.	63.7	34.0	8.5	-2.	* *ETH0092≅ *	-0.	* *SDN023JE *	. 6.	* *ALB02968 *	13.
31	SUI0140E	-19.0	38	8.2	46.6	1.0 0.7	171.	45.9	2	69.	64.3	10.4	46.9	2.	*NIG0119E *	6.	*TUN0272A *	13.	*HNG0106E *	12.
32	SYRD339A	11.0	38	37.6	34.2	1.3 3.9	74.	43.6	1	96.	63.4	36.0	36.0	э.	*C¥P0086∈ *	5.	*TUN0272A *	7.	*JOR02241 *	15.
33	TUN0272A	-25.0	38	2.5	32.0	3.6 1.8	175.	36.3	1	367.	61.9	-10.0	30.0	-1.	*MTN02238 *	2.	*MRC0209E *	4.	*GHA0108년 *	10.
34	URSO071B	44+0	38	63.1	42.0	2.6 0.8	170.	40.8	2	233.	64•5	73.1	40.8	9.	*URS 0074D	9.	*UR \$0080D	25.	*ETH0092E	29.
3	5 URS0074D	74.0	38	88.8	57.6	3.1 1.7	162.	37.1	2	1256.	68.1	97.2	49•7	6.	*MNG0248E	9.	*MNG0248D *	9.	*URS0077F *	21.
36	5 URS00800	140.0	38	155.3	55.4	2.9 2.4	35.	35.9	1	1644.	68.1	145.6	63.8	23.	*UR50077F	23.	*AF100995	89.	≭YUG 0149 ≞	89.

Canal 39 — Channel 39 — Canal 39

	1	2	3	4		5	6	7	8	9 a	9 b	10	1	11	12	13	14
	1																
20) AGL0295E	-13.0	39	16.5 -	12.0	3.1 2.3	84.	35.8	1	716.	64.4	12.5	-6.5	7.	*CDG02352 9.	*CAF0258E 15.	*SDN023DE 24.
21	BHR0255D	17.0	39	50.5	26.1	0.6 0.6	0.	48.7	1	17.	61.0	52.3	24•4	-3.	*3MA0123E 0.	*URS0059D 6.	≠IRQ02565 7.
22	2 CNR0130E	-31.0	39	-15.7	28.4	1.5 0.6	5.	44.6	2	69.	63.0	-13.5	29.2	-13.	*E 0129E-13.	*CTI02372 13.	*SRL0259E 13.
23	3 CVA0083D	-37.0	39	12.4	41.8	0.6 0.6	0.	48.7	1	47.	65.4	12.5	43.8	-3.	*YUG0149 € −1 . *	≠E 0129E 2. *	*URS0059D 13. *
24	¥ E 01295	-31.0	39	-3.1	39.9	2.1 1.1	154.	40•5	2	234.	64•2	-2.9	35.3	-8.	*CNR0130E -8.	*CTI0237E 13.	*SRL0259∂ 13. *
2!	5 GHA0108E	-25.0	39	-1.2	7.9	1.5 1.1	102.	42.3	1	143.	63.8	0.1	11.0	3.	*NGR0115E 7.	*NIG0119E 8. *	*TUN0272A 14. *
20	5 GNE0303E	-19.0	3 9	10.3	1.5	0.7 0.6	10.	48.1	2	38.	64.0	11.3	2.4	0.	*AGL0295E 4. *	*CDG0235E 8• *	*NIGO119≞ 9. *
2	7 HOLO213E	-19.0	39	5.4	52.0	0.8 0.6	171.	47.7	1	50.	64.7	6.2	50.6	0.	*E 0129E 5. ≭	*YUG0149E 7. *	*SUI0140€ 10. *
21	B ISL0050C	5.0	39	-19.5	61.0	2.2 0.8	4.	41.8	1	296.	66.5	-6.2	62.3	6.	*NOR0120C 10. *	*S 0139B 1). *	*URS0059D 12. *
2	9 JOR0224E	11.0	39	35.8	31.4	0.8 0.8	114.	46.1	2	52.	63.3	35.)	34.0	-4.	≠URS0059D 0. ≠	*TGK02255 1.	*SYR03394 5. *
3	MNGJ248E	74.0	39	102.2	46.6	3.6 1.1	169.	38.2	1	411.	64•3	98.8	51.9	-2.	*URS0077F -0. *	*URS0074D 4. *	*URS0066F 18. *
3	1 SDN0230E	-7.0	39	29.2	7.5	2.3 1.1	148.	40.1	2	286.	64.6	23.0	11.0	1.	*CAF0258E 6.	*SDN0231E 7.	*YUG0149E 9. *
3	2 SRLJ259E	-31.0	39	-11.8	8.6	J.8 J.7	114.	47.0	1	46.	63.7	-10.5	8.5	2.	≠E 0129E 7 .	*CTI0237E 8.	*GHA01082 9.

Canal 39 suite — Channel 39 cont. — Canal 39 cont.

	1	2	3	4		5	6	7	8	9 a	9 b	10		11	12		13		14	
33	TGK0225E	11.0	39	34.6	-6.2	2.4 1.7	129.	38.1	1	385.	63.9	30.5	-1.0	. 3.	*8010270E	7.	*AGL02955	10.	*SDN0230:	10.
34	UR \$0059D	23.0	39	36.0	47.0	3.7 1.4	153.	37.0	2	676.	65.3	22.0	48•4	-1.	*YUG01495	0.	*C V A0083D *	6.	*ETH0092⊡ *	12.
35	URS0077F	110.0	39	112.7	57.3	2.7 1.8	2.	37.6	1	955 .	67.4	193.6	50.1	2.	*MNG02485	2.	≭U ԶՏ 0074 D	18.	*URS0066F	21.
36	YUG0149E	-7.0	39	18.4	43.7	1.7 0.7	154.	43.8	1	147.	65.5	20.1	46.3	-3.	BEBCCAVD*	2.	≭UR \$0059 D	4.	*HNG01063	5.

Canal 40 — Channel 40 — Canal 40

		1	2	3	4		5	6	7	8	9 a	9 b	10		11	12		13		14	
1	เย่	CAF0258E	-13.0	40	21.0	6.3	2.3 1.7	31.	38.5	2	398.	64.5	23.2	11.0	1.	*SDN02325	5.	*NGR0115E	6.	*SDN02305	6.
1	19	1 00828	-19.0	40	12.3	41.3	2.4 1.0	137.	40.6	2	238.	64.3	12.6	35.4	4.	*NGR0115E	4.	*CAF0253E	18.	*E 0129E	2 2•
1	20	IRQ0256E	11.0	40	43.6	32.8	1.9 1.0	143.	41.7	1	153.	63.5	44.0	37.3	0.	*URS0066F *	1.	*TGK0225E	12.	*URS00590 *	13.
i	21	L\$003055	5.0	40	27.8	-29.8	0.7 0.6	36.	48.3	1	41.	64•4	29.5	-29.4	3.	*S 01398 *	3.	*ISL0050C	22.	*MWI0308E	24.
	22	MTND288E	-37.0	40	-7.8	23•4	1.6 1.1	141.	41.7	1	141.	63.2	-5.6	20.0	4.	*NGR 0115E *	5.	‡CVA008 3D *	14.	*URSJ966F *	25.
á	23	MW10308E	-1.0	40	34.1	-13.0	1.5 0.6	87.	44.6	2	97.	64.5	32.8	-9.3	11.	*TGK02255 1	3.	*AGL0295E *	21.	*CAF0258E *	24.
é	24	MY T0098E	29.0	40	45.1	-12.8	0.6 0.6	0.	48.7	1	31.	63.6	45.0	-12.8	19.	*1GK0225E 2 *	4.	*URS0066F *	29.	*0MA01235 *	29.
	25	NGR0115E	-25.0	40	8.3	16.8	2.5 2.1	44.	37.0	2	587.	64.7	13.0	13.0	5.	*CAF0258E *	6.	*SDiN0232E *	14.	÷1 0082≟ *	18.
1	26	UMA0123E	17.0	40	55.6	21.0	1.9 1.0	100.	41.4	2	161.	63.5	56.0	26.5	2.	*URS0066F *	6.	*IRQ0256E *	7.	*SDN0232≅ *	9.
1	27	S 01398	5.0	40	17.0	61.5	2.0 1.0	10.	41.2	2	497.	68.2	15.1	55.0	6.	≯I 00825 ¥	8.	*URS0066F *	15.	*ISL0050C *	16.
•	28	SDN0232E	-7.0	40	30•4	19.0	2.4 1.5	176.	38.6	1	310.	63.5	24.0	15.8	-0.	*C∆F0258E *	3.	*NGR0115E *	5.	*SDN0230E *	11.
i	29	UR \$0066F	44.0	40	64.3	44.6	4.6 2.5	169.	33.7	2	1560.	65.6	53.9	37.3	8.	*IRQ0256E *	9.	*NGR0115⋶ *	17.	*UR\$0059D *	22.
13	30	URS3079F	140.0	40	138.0	53.6	3.2 2.1	62.	36.0	2	1593.	68.0	128.4	73.2	18.	*URS0077F]	8.	≭MNG0248 E	35.	*UR30066F	89.

.

(Geneva, 1977)

Document No. 336-E 11 February 1977 Original : Spanish

PLENARY MEETING

FINAL PROTOCOL

For the Oriental Republic of Uruguay

The delegation of the Oriental Republic of Uruguay declares that its Government reserves the right to adopt whatever measures it may deem appropriate to ensure the development of its telecommunication services in the band 11.7-12.5 GHz, should its interests be adversely affected by the decisions of this Conference.



(Geneva, 1977)

Document No. 337-E 11 February 1977 Original : Russian

PLENARY MEETING

FINAL PROTOCOL

For the People's Republic of Bulgaria

In signing the Final Acts of the World Broadcasting-Satellite Administrative Radio Conference (Geneva, 1977), the delegation of the People's Republic of Bulgaria reserves its Government's right to take any necessary action of a technical nature, if the negative protection margin obtained for Bulgaria fails to ensure a high-quality service for the territory of the People's Republic of Bulgaria.



(Geneva, 1977)

Document No. <u>338-E</u> 11 February 1977 Original : Spanish

PLENARY MEETING

FINAL PROTOCOL

For the Republic of Panama

In signing the Final Acts of the World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977, the delegation of Panama reserves its Government's right to adopt any measures it may deem necessary to safeguard its interests should one or more Members of the Union formulate reservations liable to affect its sovereign rights and territorial integrity.



(Geneva, 1977)

Document No. 339-E 11 February 1977 Original : Spanish

PLENARY MEETING

FINAL PROTOCOL

For the Argentine Republic

1. The delegation of the Argentine Republic declares that its Government reserves the right to adopt any measures it may deem appropriate to ensure the development of its telecommunication services in the band 11.7-12.5 GHz, should its interests be adversely affected by the decisions of this Conference.

2. In signing these Final Acts, the delegation of the Argentine Republic reiterates its Declaration No. LXXXVII contained in the International Telecommunication Convention, Malaga-Torremolinos, 1973.

Document No. 340-E 11 February 1977 Original : English

(Geneva, 1977)

PLENARY MEETING

FINAL PROTOCOL

For the Republic of Korea

1. The delegation of the Republic of Korea, on behalf of its Government, reserves the right

a) to take any measures as it may deem necessary to safeguard its broadcasting and telecommunication services, should any Contracting Member fail to comply with the provisions, and associated Plan, of the Final Acts of the World Broadcasting-Satellite Administrative Radio Conference, thus causing harmful effects on the services of the Republic of Korea; and

b) to demand, and take due action for, if necessary, the reduction of any Member's coverage area in case it extends beyond a reasonable extent into the territory of the Republic of Korea and the Member fails to take appropriate remedial measures in implementing the Plan under relevant provisions of the Radio Regulations.

2. The delegation further declares that, within its territory, no signals in the Plan other than its own will be protected by its Administration.

Document No. 341-E 11 February 1977 Original : English

(Geneva, 1977)

PLENARY MEETING

FINAL PROTOCOL

For Islamic Republic of Pakistan

The delegation of Pakistan has the honour to refer to paragraph 2 of the Reservations made by the delegation of the Republic of India (Document No. 278) and wishes to make the following comments.

The State of Jammu and Kashmir has been recognized by the United Nations as a disputed territory and its permanent status has yet to be determined by the people of the State, in accordance with the relevant resolutions of the United Nations Security Council. Decisions regarding areas falling within the disputed State are without prejudice to the position recognized by the United Nations in its resolutions on the subject. The areas covered by the Indian Coverage Plan which fall within the State of Jammu and Kashmir are not recognized by Pakistan as being part of Indian territory.

For reasons of economy, this document is printed in a limited number. Participants are therefore kindly asked to bring their copie to the conference since only a few additional copies can be made available.

(Geneva, 1977)

Document No. 342-E 11 February 1977 Original : English

PLENARY MEETING

FINAL PROTOCOL

For India

With reference to the Final Protocol for the People's Republic of Bangladesh as contained in Document No. 303, the delegation of India wishes to state the following :

The Indian beam No. IND 0037, submitted by the Indian Administration, was designed to cover only the Indian territory. In the process of optimization by the IFRB, the beam dimensions have been enlarged. The coverage of Bangladesh territory by this enlarged beam is thus only a technical spillover and does not interfere with any assignment to Bangladesh. However, the Indian Administration is prepared to take any corrective measures as per procedure laid down by the Conference.



For reasons of economy, this document is printed in a limited number. Participants are therefore kindly asked to bring their copie to the conference since only a few additional copies can be made available.

(Geneva, 1977)

Document No. 343-E 11 February 1977 Original : English

PLENARY MEETING

FINAL PROTOCOL

For the Federative Republic of Brazil

In signing the Final Acts of the present Conference, the delegation of the Federative Republic of Brazil wishes to make the following statement.

The Conference has adopted a Plan for the allotment of frequencies and positions in the geostationary satellite orbit for the Broadcasting-Satellite Service for Regions 1 and 3. The delegation of the Federative Republic of Brazil draws the attention of this Conference to the fact that the above-mentioned Plan and its associated sharing criteria do not make adequate provision for the needs of the Fixed-Satellite and Broadcasting-Satellite Services in some countries in Region 2, and that the sharing criteria do not provide the protection required under paragraphs 117 and 428A of the Radio Regulations.

Furthermore, this Plan and its sharing criteria may preclude the successful development of a plan for Region 2 countries in 1982. The Brazilian Administration therefore reserves the right to operate its systems in accordance with the ITU Convention and the Radio Regulations and to claim the protection thus afforded.

(Geneva, 1977)

Document No. 344-E 11 February 1977 Original : English

PLENARY MEETING

FINAL PROTOCOL

For the Democratic People's Republic of Korea

Concerning the Japanese ellipse for Broadcasting-Satellite Service which covers wide areas of the territory of the Democratic People's Republic of Korea, the delegation of the Democratic People's Republic of Korea submitted to the IFRB and Committee 5 the letter requesting that the Japanese ellipse should be reduced to the minimum.

The delegation of the Democratic People's Republic of Korea was informed, however, that the Japanese ellipse could not be reduced under a pretext of technique.

To reduce an ellipse for Broadcasting-Satellite Service or not is not merely a technical matter, but is a matter having political aspect. So, the solution of this problem depends entirely on the attitude of the Japanese authority.

Such unjust attitude of the Japanese authority not only runs counter 428A of the international Radio Regulations, but also is an infringement upon the sovereignty of the Democratic People's Republic of Korea.

Therefore, the delegation of the Democratic People's Republic of Korea cannot tolerate the Japanese ellipse for Broadcasting-Satellite Service which covers the territory of the Democratic People's Republic of Korea and states that all necessary measures will be taken against the coverage by the Japanese ellipse of the territory of the Democratic People's Republic of Korea.

(Geneva, 1977)

Document No. 345-E 11 February 1977 Original : English

PLENARY MEETING

FINAL PROTOCOL

For the Democratic People's Republic of Korea

Concerning the fact that the South Korea "authority" intentionally enlarged its ellipse for Broadcasting Satellite Service to cover even the service area of the Democratic People's Republic of Korea, the delegation of the Democratic People's Republic of Korea participating in the World Administrative Broadcasting-Satellite Conference, Geneva 1977, submitted to the IFRB the proposals requesting that necessary measures should be taken to solve this problem.

However, in disregard of the just demand of the Democratic People's Republic of Korea and the advice of the IFRB, the South Korean "authority" further enlarged the size of the ellipse, thus covering even wider area of the Democratic People's Republic of Korea's service area. This machination of the South Korean "authority" which ignores the Convention of the ITU is aimed at attaining its insidious political goal.

The delegation of the Democratic People's Republic of Korea states that it does not recognize any proposals of the South Korean "authority" and that it cannot tolerate the coverage of the service area of the Democratic People's Republic of Korea by the ellipse for Broadcasting-Satellite Service of the South Korean "authority", and states that all necessary measures will be taken against the coverage of the service areas of the Democratic People's Republic of Korea by the ellipse of the South Korean "authority".



(Geneva, 1977)

Document No. 346-E 11 February 1977 Original : French

PLENARY MEETING

FINAL PROTOCOL

For Tunisia

The Tunisian delegation to the Broadcasting-Satellite Conference (Geneva, 1977), desires to affirm solemnly that strict observance of the provisions of No. 428A of the Radio Regulations is a necessary condition for the application of the Plan and for implementation of the Agreement.

The Tunisian delegation requests the reduction of the coverage area resulting from the assignments to certain neighbouring countries; except for technically unavoidable spillover, coverage should be limited to their national territory in conformity with No. 428A mentioned above.

The Tunisian delegation declares unacceptable any technically avoidable spillover on its territory by the assignments to these countries and reserves its Government's right to take any technical or other measures required to ensure by whatever means the integrity of its national territory in the face of any external interference and to protect its broadcasting services.



For reasons of economy, this document is printed in a limited number. Participants are therefore kindly asked to bring their copies to the conference since only a few additional copies can be made available.

(Geneva, 1977)

Document No. 347-E 11 February 1977 Original : French

PLENARY MEETING

FINAL PROTOCOL

For the Kingdom of Morocco

The beams designated ALG 251 and ALG 252 which have been assigned to the Algerian Democratic and Popular Republic spill over on Moroccan territory to an unacceptable degree. This spill over is by no means technical. A valid technical solution ensuring coverage of the entire Algerian territory, while safeguarding the interests of the Kingdom of Morocco, has been proposed by the Moroccan delegation to the Algerian delegation.

In view of that delegation's refusal to adopt the proposed compromise solution, the Moroccan delegation formulates the most emphatic reservations regarding the assignment of the two above-mentioned beams to the Algerian Democratic and Popular Republic and requests the Conference to note that the provisions of No. 428A of the Radio Regulations have not been respected.

It should also be noted that the beam assigned to the Kingdom of Morocco and designated MRC 209 covers the entire Moroccan territory in the minimal conditions fixed by this Conference. The Moroccan delegation accordingly reserves its Government's right to take any action required in order to ensure the proper operation of its broadcasting-satellite network throughout its territory should any Administration formulate reservations or take measures liable to infringe the sovereign rights of the Kingdom of Morocco.



(Geneva, 1977)

Document No. 348-E 11 February 1977 Original : English

PLENARY MEETING

 $\overline{\mathbb{C}}$

FINAL PROTOCOL

For Turkey

In signing the Final Acts of the World Broadcasting-Satellite Radio Conference, Geneva, 1977, the delegation of Turkey reserves for its government the right to take whatever action may be necessary to safeguard its interests and ensure the proper functioning of its Broadcasting-Satellite and Terrestrial Services should any country fail to comply with the Final Acts or its Annexes or the Protocols attached thereto, or should reservations by other countries jeopardize its above-mentioned services.

For reasons of economy, this document is printed in a limited number. Participants are therefore kindly asked to bring their copie to the conference since only a few additional copies can be made available.

(Geneva, 1977)

Document No. 349-E 11 February 1977 Original : English

PLENARY MEETING

GENÈ

FINAL PROTOCOL

For the United Kingdom of Great Britain and Northern Ireland

With reference to a reservation in relation to the territory of Belize made by Guatemala on signing the International Telecommunication Convention, the United Kingdom does not accept that Guatemala has any rights or any valid claim with respect to that territory.

For reasons of economy, this document is printed in a limited number. Participants are therefore kindly asked to bring their copies to the conference since only a few additional copies can be made available.

Document No. 350-E 11 February 1977 Original : Spanish

(Geneva, 1977)

PLENARY MEETING

FINAL PROTOCOL

For Mexico

The delegation of Mexico declares that its Administration reserves the right to adopt whatever measures it may deem appropriate to ensure the development of its telecommunication services in the band 11.7 - 12.5 GHz should its interests be adversely affected by the decisions of this Conference.



PLENARY MEETING

R.5

5th SERIES OF TEXTS SUBMITTED BY THE EDITORIAL COMMITTEE TO THE PLENARY MEETING

The following texts are submitted to the Plenary Meeting for second reading :

•	Source	Doc	ument No.	Title
	B.ll		261	Recommendation GG
	B.4(Add.2)	225	(Add.2)	Corrigendum to Article 4
	B.13		296	Annexes 2, 4 and 5 Article 9
	B.12		268	Article 11
	B.15		298	Final Protocol
	B.12		268	Annexes 6 and 7 Recommendation HH Resolutions G and H
	B.14		297	Resolution I
	B.12		268 -	Columns headings of the Plan
	в.14		297	Table showing correspondence

Miss M. HUET

Chairman of the Editorial Committee

<u>Annexes</u> : 22 pages



PINK PAGES

RECOMMENDATION NO. GG

relating to the radiation of harmonics of the fundamental frequency by broadcasting-satellite stations

The World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977,

considering

a) that the frequency band 23.6-24 GHz is allocated to the radio astronomy service on a primary basis;

b) that the second harmonic of the fundamental frequency of broadcasting-satellite stations operating within the band 11.8-12.0 GHz may seriously disturb radio astronomy observations in the band 23.6-24.0 GHz if effective steps are not taken to reduce the radiation level produced by this harmonic;

in view of

the provisions of No. 673 of the Radio Regulations;

recommends

that, when defining the characteristics of their space stations operating in the broadcasting-satellite service, particularly within the band 11.8-12.0 GHz, administrations take all necessary steps to reduce the radiation level of the second harmonic below the values indicated in the relevant CCIR Recommendations.

CORRIGENDUM TO ARTICLE 4

1. Page 2, Article 4, Procedures for modifications to the Plan

In the first line of paragraph 4.1, after the word "modification", <u>insert note</u> "1)";

at the bottom of the same page, add the following note 1) :

- 1) The intention not to employ energy dispersal consistent with section 3.18 of Annex 8 shall be treated as a modification and thus subject to the appropriate provisions of this Article.
- 2. <u>The existing footnote 1) shall be renumbered footnote 2), and the following words added at the end</u>:

See Annex 10 for the orbital position limitations.
ANNEX 2

BASIC CHARACTERISTICS TO BE FURNISHED IN NOTICES

RELATING TO SPACE STATIONS IN THE BROADCASTING-SATELLITE SERVICE

- 1. Country and IFRB number
- 2. Nominal orbit position (in degrees from the Greenwich meridian)
- 3. Assigned frequency or channel number
- 4. Date of bringing into use
- 5. Identity of the space station
- 6. Service area (if necessary, the service area may be defined by a number of "test points")
- 7. Geographical coordinates of the intersection of the antenna beam axis with the Earth
- 8. Rain-climatic zone
- 9. Class of station
- 10. Class of emission and necessary bandwidth
- 11. Power supplied to the antenna (Watts)
- 12. Antenna characteristics
 - gain of the antenna referred to an isotropic radiator
 - shape of the beam (elliptical or circular)
 - major axis (degrees) at -3 dB points
 - minor axis (degrees) at -3 dB points
 - orientation of the ellipse
 - ΔG (difference between the maximum gain and the gain in the direction of the point in the service area at which the power flux density is at a minimum)
 - pointing accuracy
 - type of polarization
 - sense of polarization
 - radiation pattern and cross-polar characteristics

PINK PAGES

- 13. Station keeping accuracy
- 14. Modulation characteristics
 - type of modulation
 - pre-emphasis characteristics
 - TV system
 - sound broadcasting characteristics
 - frequency deviation
 - composition of the baseband
 - type of multiplexing of the video and sound signals
 - energy dispersal characteristics
- 15. Minimum angle of elevation in the service area
- 16. Type of reception (individual or community)
- 17. Hours of operation (GMT)
- 18. Coordination
- 19. Agreements
- 20. Other information
- 21. Operating administration or company

ANNEX 4

NEED FOR COORDINATION OF A FIXED-SATELLITE SPACE STATION OR A BROADCASTING-SATELLITE SPACE STATION IN REGION 2 WITH RESPECT TO THE PLAN (ARTICLE 27)

With respect to paragraph 2.1, coordination of a space station in the fixed-satellite service or the broadcasting-satellite service of Region 2 is required when, under assumed free space propagation conditions, the power flux density on the territory of an administration in Region 1 or Region 3 exceeds the value derived from the following expressions :

-147 dBW/m²/27 MHz for $0 \le \theta < 0.44^{\circ}$ -138 + 25 log θ dBW/m²/27 MHz for $0.44^{\circ} \le \theta < 19.1^{\circ}$ -106 dBW/m²/27 MHz for 19.1° $\le \theta$

 θ = the difference in degrees between the longitude of the interfering broadcasting-satellite or fixed-satellite in Region 2 and the longitude of the affected broadcasting-satellite space station in Regions 1 and 3.

R.5

PINK PAGES Document No. 351-E Page 7

ANNEX 5

POWER FLUX DENSITY LIMITS BETWEEN 11.7 AND 12.2 GHz TO PROTECT THE TERRESTRIAL SERVICES IN REGIONS 1 AND 3 FROM INTERFERENCE FROM REGION 2 BROADCASTING-SATELLITE SPACE STATIONS (ARTICLE 9)

The power flux density limits are as follows :

1) for all the territories of administrations in Regions 1 and 3 :

-125 dBW/m²/4 kHz

-128 dBW/m²/4 kHz

for broadcasting-satellite space stations using circular polarization;

for broadcasting-satellite space stations using linear polarization

for all angles of arrival; and

- 2) for territories of administrations in Region 3 and those in the western part of Region 1, West of longitude 30°E :
 - -132 dBW/m²/5 MHz -132 dBW/m²/5 MHz $-132 + 4.2 (\gamma - 10) dBW/m²/5 MHz$ for angles of arrival between 0° and 10° above the horizontal plane; for angles of arrival γ (in degrees) between 10° and 15° above the horizontal plane;

-111 dBW/m²/5 MHz for angles of arrival between 15° and 90° above the horizontal plane.

R.5

ARTICLE 9

Power flux density limits between 11.7 and 12.2 GHz to protect terrestrial services in Regions 1 and 3 from interference from Region 2 broadcasting-satellite space stations

The power flux density at the Earth's surface in Regions 1 and 3, produced by emissions from a space station in the broadcasting-satellite service in Region 2 for all conditions and for all methods of modulation shall not exceed the values given in Annex 5 on the territory of any country unless the administration of that country so agrees.

ARTICLE 11

PROVISIONS GOVERNING THE BROADCASTING-SATELLITE SERVICE IN REGION 2 PENDING THE ESTABLISHMENT OF A DETAILED PLAN

In accordance with the principles set forth in / Annex A /, the following interim provisions shall apply pending the establishment of a detailed plan for the broadcasting-satellite service for Region 2 in the frequency band 11.7 - 12.2 GHz under the terms of paragraphs 8 - 11 below.

1. Space stations in the broadcasting-satellite service shall be located in the following portions of the orbit :

- 75°W to 100°W longitude (however, for service to Canada, the USA and Mexico, the relevant portion shall be only between 75°W and 95°W longitude);

- 140°W to 170°W longitude.

1.1 Space stations in the broadcasting-satellite service may also be located in the remaining portions of the orbit, in which case they shall be operated in accordance with the provisions of No. 139 of the Radio Regulations. As an exception, it is accepted that, for Greenland, a position in the geostationary satellite orbit between 55°W and 60°W may be used for the broadcasting-satellite service as a primary service. The administrations concerned should make every effort to allow for the sharing of this portion of the orbital arc by a broadcasting satellite for Greenland and space stations in the fixed-satellite service of other administrations in Region 2.

2. Space stations in the fixed-satellite service shall be located in portions of the orbit other than those referred to in paragraph 1 above. Such space stations may also be located in the portions of the orbit referred to in paragraph 1 above; they shall then be operated in accordance with the provisions of No. 139 of the Radio Regulations.

2.1 Space stations in the broadcasting-satellite service located in the portions of the orbit referred to in paragraph 1 and space stations in the fixed-satellite service located in the remaining portions of the orbit shall be operated in such a way that no unacceptable interference is caused by stations of one service to stations of other services. The level of unacceptable interference shall be determined by agreement between the administrations concerned, taking the latest CCIR Recommendations and Annexes 8 and 9 of the Final Acts as a guide. Notwithstanding the above, broadcasting-satellite space stations may be located up to the edge of the portion of the orbit referred to in paragraph 1, provided that such stations are operated in accordance with the relevant technical characteristics for Region 2 outlined in Annex 8.

3. Prior to the Regional Administrative Radio Conference, referred to in paragraph 8 below, systems in the broadcasting-satellite service shall be regarded as experimental and shall be operated in accordance with the sharing criteria and technical characteristics contained in Annexes 8 and 9.

٩.

Document No. 351-E Page 10

4. Administrations may implement systems which utilize values for the technical characteristics different from the values in Annex 8 of the Final Acts, provided that such action does not result in interference to operational or planned systems of other administrations in excess of that determined in accordance with Annex 9.

5. Systems in the fixed-satellite service shall be introduced in accordance with the relevant provisions of the Radio Regulations and particularly with those of Article 9A, and where appropriate with the provisions of Article 7 of these Final Acts.

6. Space systems in the frequency band 11.7 - 12.2 GHz shall use, to the maximum extent technically and economically practicable, available techniques in order to make the most efficient use of the geostationary orbit and the frequency spectrum. Examples of such techniques are described in Annex 7.

7. The provisions of Spa2 - 3 shall continue to apply to the broadcastingsatellite service in the frequency band 11.7 - 12.2 GHz in Region 2 until such time as a detailed plan may be adopted for the broadcasting-satellite service.

8. A Regional Administrative Radio Conference is to be held not later than 1982 for the purpose of carrying out detailed planning for the broadcastingsatellite and fixed-satellite services, in accordance with the following terms.

8.1 The said Regional Administrative Radio Conference shall draw up a detailed plan for the orbit/spectrum resource available for the broadcasting-satellite service in the 11.7 - 12.2 GHz band. The plan shall provide for the detailed assignment of the orbital positions and frequency channels available, ensuring that the broadcasting-satellite service requirements submitted by the various administrations are met in an equitable manner satisfactory to all the countries concerned. It should be laid down as a matter of principle that each administration in the Region should be guaranteed a minimum number of channels (4) for the operation of the broadcasting-satellite service. Above this minimum, the special characteristics of the countries (size, time zones, language differences, etc.) shall be taken into account.

8.2 Planning shall be based on individual reception, but each administration may use the reception system which best meets its requirements, namely, individual or community reception, or both. Account shall also be taken of the decisions of the 1977 and 1979 World Administrative Radio Conferences and of the latest CCIR Recommendations in the case of parameters covered by its studies and research.

8.3 When planning the broadcasting-satellite service, it shall be borne in mind that systems should be designed with a view to reducing to a minimum technical differences and incompatibilities with the systems of other Regions.

8.4 The Conference shall also take into account the need to make equitable provision for the requirements of the fixed-satellite service to which this frequency band is also allocated in Region 2.

22

9. All administrations in Region 2 shall submit their broadcastingsatellite service requirements to the IFRB not later than one year before the start of the Regional Administrative Radio Conference responsible for planning this service in Region 2. Each administration may update these requirements as it considers necessary. "Requirements" are understood to include the number and boundaries of service areas and the number of channels requested for each of them. Six months before the deadline for submitting requirements, the IFRB shall remind administrations of the need to submit them by means of a circularletter and/or telegram.

Document Page 11

R.5

10. No systems existing or planned prior to the implementation of any detailed plan such as that referred to above shall cause interference to any systems operating in accordance with such a plan.

11. Existing or previously planned broadcasting-satellite systems will not necessarily be taken into account in the establishment of the detailed plan for the broadcasting-satellite service in the 11.7 - 12.2 GHz band in Region 2. Consequently, the installation or planning of such systems by an administration prior to the establishment of the said plan shall not confer upon that system any rights or recognition.

R.5

FINAL PROTOCOL

At the time of signing the Final Acts containing the provisions, the associated Plan and the decisions concerning the Re-arrangement of the Radio Regulations and the Additional Radio Regulations adopted by the World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977, the undersigned delegates take note of the following statements forming part of the Final Acts of this Conference.

/ Documents No. 269 and No. ... give a list of the Conference documents in which the statements in question are reproduced._/ ANNEX 6

Planning principles in Region 2

The following principles have been applied in drawing up the provisions governing the introduction of space services in the frequency band 11.7 - 12.2 GHz in Region 2 :

1. Equality for allocated services in Region 2

Under Article 5 of the Radio Regulations, the 11.7 - 12.2 GHz band is allocated to broadcasting-satellite, fixed-satellite and terrestrial services on an equal, primary basis. Each administration in Region 2 has the right to decide for itself which of these services are to be implemented within its own territory.

2. Equal rights for services in the various Regions

In accordance with No. 117 of the Radio Regulations, the principle of equal rights for different services in the same category to operate in all the Regions is recognized, provided that no harmful interference is caused to services in the other Regions.

3. Recognition of national requirements

All administrations in Region 2 shall take into consideration the national requirements which have been presented or will be presented in the future.

4. Equitable rights of access to the geostationary orbit spectrum resource

Subject to the provisions of the Convention, the Radio Regulations and the Resolutions in force, it is recognized that all administrations have the right of access to the geostationary orbit spectrum resource in order to fulfil their requirements.

5. <u>Flexible planning approach</u>1)

The plan adopted for Region 2 must be sufficiently flexible to allow for future technical developments, the identification of future requirements, changes in existing or stated requirements, requirements by administrations not represented at the Conference, further information on propagation data and various system design approaches. The plan may be modified only by a competent Administrative Radio Conference.

1) Paragraph 5 does not imply recognition of systems existing prior to the implementation of the plan

R.5

2

Document No. 351-E Page 14

6. Efficient use of the geostationary orbit and the spectrum

The plan for Region 2 shall use, to the maximum extent technically and economically practicable, the techniques available so as to make the most efficient use of the geostationary orbit and the frequency spectrum to fulfil the requirements both of the Region as a whole and of the individual administrations.

7. Consultations among administrations

Administrations planning to bring into operation systems in the 11.7 - 12.2 GHz band, shall consult all the other administrations affected or concerned.

8. Reception

The plan for Region 2 shall have as a basis individual reception, although each administration may choose the reception system that it finds most suited to its requirements, namely, individual or community reception, or both.

PINK PAGES

Document No. 351-E Page 15

ANNEX 7

The second second second second

· . · . . .

Use of the spectrum/orbit resource .

Since the equal sharing of the spectrum/orbit resource between the broadcasting-satellite service and the fixed-satellite service in Region 2 is inherently difficult and may impose some restrictions on both services, it is important that the technical parameters be chosen, and the techniques for efficient use of the spectrum/orbit resource be applied in such a way that both space services will benefit as much as possible.

The following techniques are among those identified as leading to a more efficient use of the spectrum/orbit resource and should therefore be applied to the maximum extent technically and economically practicable consistent with the capability of systems to fulfil the requirements for which they were designed.

Clustering

1.

Extensive analyses have shown that orbit utilization is improved when satellites are grouped according to the sensitivity to interference and the potential for generating interference of the system of which they are a part. In most cases, this means that space stations of similar characteristics should be grouped in the same part of the orbit.

2. Cross-polarization

The proper use of cross-polarization can significantly improve the use of the spectrum/orbit resource by providing additional.isolation between potentially interfering systems.

1 11 12 1

3. Crossed-beam geometry

The principle of crossed-beam geometry is that adjacent satellites should not serve adjacent service areas. In that way, discrimination from both the satellite and the earth station antennae can be used to achieve maximum isolation between systems:

4. Paired service areas

The principle of crossed-beam geometry can be extended : if service areas are far enough apart, then the satellite antenna discrimination alone may be sufficient to permit satellites serving these widely separated service areas to be co-located in the orbit, leading to practical doubling of the orbit capacity.

5. Frequency interleaving

The mutual interference between channels in different systems is usually a maximum when the two carrier frequencies coincide. When channelling design is such that frequencies are interleaved, or, more generally, such that coincidence of carrier frequencies is avoided, mutual interference can in many cases be greatly reduced.

6. Minimum space station spacings

It is obvious that, for maximum orbit utilization, space stations should be placed as close to each other as is consistent with keeping the mutual interference to acceptable levels.

7. Space station antenna discrimination

The discrimination in the sidelobes of the space station antenna determines how much isolation exists between beams serving non-overlapping or non-adjacent service areas. To achieve maximum isolation, every effort should be made to improve the discrimination by technological advances in antenna design.

8. Earth station antenna discrimination

The sidelobe discrimination of the earth station antenna determines how much isolation is obtained from satellite spacing. To achieve maximum isolation, every effort should be made to improve the discrimination by taking advantage of technological advances in antenna design.

9. Minimizing e.i.r.p. differences

The interference caused by relatively high-power space stations (space stations in the broadcasting-satellite service or certain types of space stations in the fixed-satellite service) to the earth station receivers of relatively low-power satellite systems is directly proportional to the difference between their e.i.r.p. Sharing among such systems is greatly facilitated if this difference is kept as small as is consistent with the requirements.

10. Realistic quality and reliability objectives

The quality and reliability objectives have a significant effect on the use of the spectrum/orbit resource. If the objectives are set unnecessarily high, the capacity of the orbit is reduced. Quality and reliability objectives should be set no higher than are absolutely necessary.

RECOMMENDATION No. HH

Relating to the convening of a Regional Administrative Radio Conference for the detailed planning of the space services in the frequency band 11.7 - 12.2 GHz in Region 2

The World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977,

noting

a) that the detailed requirements of all administrations in Region 2 for the broadcasting-satellite service in the frequency band 11.7 - 12.2 GHz are not yet known:

b) that, in view of the large demands expected for the other services with which this band is shared, there is a need to ensure that this frequency band and the geostationary orbit are used as efficiently as possible;

c) that a future Regional Administrative Radio Conference for the detailed planning of space services in the frequency band 11.7 - 12.2 GHz would be able to take advantage of experiments now being carried out, of further technological advances, and of additional studies by the CCIR:

<u>considering</u>

the provisions adopted by this Conference to govern the implementation of space services in the frequency band, 11.7 - 12.2 GHz pending the establishment of a detailed plan for Region 2;

recommends

1. that a Regional Administrative Radio Conference be held not later than 1982 for the purpose of carrying out detailed planning for the broadcasting-satellite and fixed-satellite services in Region 2, in accordance with 2), 3), 4), 5) and 6 below;

2. that the said Regional Administrative Radio Conference draw up a detailed plan for the orbit/spectrum resource available for the broadcastingsatellite services in the 11.7 - 12.2 GHz band. The plan shall provide for the detailed assignment of the orbital positions and frequency channels available, ensuring that the broadcasting-satellite service requirements submitted by the various administrations are met in an equitable manner satisfactory to all the countries concerned. It should be laid down as a matter of principle that each administration in the Region should be guaranteed a minimum number of channels (4) for the operation of the broadcasting-satellite service. Above this minimum, the special characteristics of the countries (size, time zones, language differences, etc.) shall be taken into account;

3. that planning be based on individual reception, but each administration may use the reception system which best meets its requirements, namely, individual or community reception, or both. Account shall also be taken of the decisions of the 1977 and 1979 World Administrative Radio Conferences and of the latest CCIR Recommendations in the case of parameters covered by its studies and research;

4. that, when planning the broadcasting-satellite service, it be borne in mind that systems should be designed with a view to reducing to a minimum technical differences and incompatibilities with the systems of other Regions;

5. that the Conference also take into account the need to make equitable provision for the requirements of the fixed-satellite service to which this frequency band is also allocated in Region 2;

6. that in drafting the above-mentioned detailed plan, account also be taken of the terrestrial radio services sharing the same band;

invites the Administrative Council

to make preparations for convening the said Regional Administrative Radio Conference using the provisions of this Recommendation as a basis for the agenda and the terms of reference of the Conference.

RESOLUTION No. G

Relating to the preparation for an Administrative Radio Conference for the detailed planning of the space services in the frequency band 11.7 - 12.2 GHz in Region 2

The World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977,

considering

a) that a Regional Administrative Radio Conference is to be held not later than 1982 for the detailed planning of the space services in the frequency band 11.7 - 12.2 GHz in Region 2;

b) that the technical criteria and procedures adopted at this Conference, the 1979 World Administrative Radio Conference and the latest CCIR Recommendations will be used in the interim period;

<u>c</u>) that a considerable amount of technical information will be required to ensure the success of this Regional Conference;

invites the CCIR

to carry out such additional studies as are necessary to ensure timely provision of the technical information likely to be required as a basis for the work of the Regional Conference.

RESOLUTION No. H

Relating to the submission of requirements for the broadcasting-satellite service in Region 2

The World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977,

considering

a) the decision taken by the Conference that an Administrative Radio Conference for Region 2 is to be held not later than 1982;

b) that the said Regional Administrative Radio Conference is to draw up a detailed plan for the orbit spectrum resource available for the broadcastingsatellite services in the frequency band 11.7 - 12.2 GHz, taking into account the need to make equitable provision for the requirements of the other services to which this frequency band is also allocated in Region 2;

<u>c</u>) that the plan is to provide for the detailed assignment of the orbital positions and frequency channels available, ensuring that the broadcasting-satellite service requirements of the various administrations are met in an equitable manner satisfactory to all the countries concerned,

invites the IFRB

1. to request all administrations in Region 2 to submit their broadcasting-satellite service requirements to the IFRB not later than one year before the start of the said Regional Administrative Radio Conference. These requirements are understood to include the number and boundaries of service areas and the number of channels requested for each of them. They may be updated as required by each administration;

2. to remind administrations, by means of a circular letter and/or telegram six months before the above deadline for submitting requirements, of the need to submit them;

3. to assemble the information submitted by administrations in a form permitting a comparative study thereof and to communicate it to the Secretary-General for publication and despatch to administrations not later than nine months prior to the said Regional Administrative Radio Conference.

RESOLUTION No. I

Relating to the preparation and publication of information not contained in the broadcasting-satellite . Plan for Regions 1 and 3

The World Broadcasting-Satellite Administrative Radio Conference, Geneva, 1977,

considering

<u>a)</u> that the planning work for Regions 1 and 3 has been based on the calculation of the protection margins at a number of test points;

b) that it would be useful to know the equivalent protection margin at each of these test points for all the assignments in the Plan, in order to assess any degradation which may result from subsequent amendments to the Plan;

c) that it would be helpful, in applying the method set forth in Annex 3, for any administration wishing to bring terrestrial stations into service to know the elevation angle of the receiving antennae of the earth stations in the broadcasting-satellite service;

invites the IFRB

to prepare, with a view to its publication by the Secretary-General in 1977, a document containing the following information :

a) column 1 : country symbol and IFRB serial number for the beam;

b) column 2 : geographical coordinates of the test points as given in Document No. 149 of the Conference;

c) column 3 : elevation angle of the receiving antenna at each of these test points;

d) column 4 : azimuth in degrees clockwise from True North of the major beam axis of the receiving antenna;

e) column 5: the equivalent protection margin¹) in dB at each of these test points for all the assignments in the Plan.

 For the definition of equivalent protection margin, see Annex 8, paragraph 3.4, footnote 1).

R.5

.

-

COLUMNS HEADINGS OF THE PLAN

- 1. <u>Country symbol and IFRB Serial Number</u>(Column 1 contains the symbol designating the country or the geographical area taken from Table No. 1 of the Preface to the International Frequency List).
- 2. Nominal orbital position, in degrees.
- 3. <u>Channel number</u> (see Table showing correspondence between channel numbers and assigned frequencies, page / _/).
- 4. Boresight geographical coordinates, in degrees and tenths of a degree.
- 5. <u>Antenna beamwidth</u>: This column contains two figures corresponding to the major axis and the minor axis respectively of the elliptical cross-section half-power beam, in degrees and tenths of a degree.
- 6. <u>Orientation of the ellipse</u> determined as follows : in a plane normal to the beam axis, the direction of a major axis of the ellipse is specified as the angle measured anti-clockwise from a line parallel to the equatorial plane to the major axis of the ellipse to the nearest degree.
- 7. <u>Polarization</u> (1 = direct, 2 = indirect).¹⁾
- 8. <u>E.i.r.p</u>. in the direction of maximum radiation in dBW.
- 9. Remarks.

1) See Annex 8, paragraph 3.2.3.

PINK PAGES Document Nº 351-F/E/S Page 23

TABLEAU DE CORRESPONDANCE ENTRE LE NUMERO DU CANAL ET LA FREQUENCE ASSIGNEE

TABLE SHOWING CORRESPONDENCE BETWEEN CHANNEL NUMBERS AND ASSIGNED FREQUENCIES

CUADRO DE CORRESPONDENCIA ENTRE EL NÚMERO DEL CANAL Y LA FRECUENCIA ASIGNADA

Canal Channel N°	Fréquence assignée Assigned frequency Frecuencia asignada (MHz)	Canal Channel N°	Fréquence assignée Assigned frequency Frecuencia asignada (MHz)
l	11 727,48	21	12 111,08
2	il 746,66	22	12 130,26
3	11 765,84	23	12 149,44
4	11 785,02	24	12 168,62
5	11 804,20	25	12 187,80
6	11 823,38	26	12 206,98
7	11 842,56	27 [·]	12 226,16
8	11 861,74	28	12 245,34
9	11 880,92	29	12 264,52
10	11 900,10	30	12 283,70
11	11 919,28	31	12 302,88
12	11 938,46	32	12 322,06
13	11 957,64	33	12 341,24
14	11 976.82	34	12 360,42
15	11 996,00	35	12 379,60
16	12 015,18	36	12 398,78
17	12 034,36	37	12 417,96
18	12 053,54	38	12 437,14
19	12 072,72	39	12 456,32
20	12 091,90	40	12 475,50

INTERNATIONAL TELECOMMUNICATION UNION

BROADCASTING SATELLITE CONFERENCE

(Geneva, 1977)

Document No. 352-E 12 February 1977 Original : English

PLENARY MEETING

MINUTES

OF THE

EIGHTH PLENARY MEETING

Wednesday, 9 February 1977, at 1030 hrs

Chairman : Mr. Ib LØNBERG (Denmark)

Subjects discussed

Document No.

242

GENÈVE

1. Ninth series of texts submitted by the Editorial Committee (B.9)

2. Statement by the delegate of Indonesia

1.

<u>Ninth series of texts submitted by the Editorial Committee (B.9)</u> (Document No. 242)

The <u>Chairman of the Editorial Committee</u>, supported by the <u>Chairman</u> of <u>Committee 6</u>, proposed the deletion of the square brackets from the introductory sentence of each of the Recommendations contained in Document No. 242.

It was so <u>decided</u>.

The <u>delegate of India</u> proposed the deletion of the word "more" before the word "ample" in considering 1 of Recommendations Nos. BB, CC and DD.

It was so decided.

Recommendation No. CC

The <u>representative of the CCIR</u> proposed the insertion of the word "rain" between the words "all" and "climatic" in paragraph 1 under "invites the CCIR" and the deletion of the words "99 % of" before the words "the year" in paragraph 3 under "invites the CCIR".

It was so decided.

Recommendation No. DD

The <u>delegate of the United States of America</u> proposed that the word "precision" in the penultimate line of considering f) should be replaced by the word "purity".

It was so decided.

Recommendation No. EE

The <u>delegate of India</u> proposed that the words "to estimate their future requirements for such links" at the beginning of the paragraph under "invites administrations" should be replaced by the words "to send their suggestions and specific proposals".

The <u>delegate of the United States of America</u>, speaking as Chairman of the Working Group in which Recommendation No. EE had been drafted, pointed out that the wording proposed by the delegate of India failed to make clear the kind of suggestions and specific proposals which administrations were being invited to forward to the CCIR.

The <u>Chairman</u>, supported by the <u>delegate of the United Kingdom</u>, proposed that the text should be retained in its original form but with the addition of the word "technical" between the words "future" and "requirements".

It was so agreed.

The ninth series of texts (B.9) was <u>approved</u> on first reading, as amended and subject to editorial changes.

2. Statement by the delegate of Indonesia

The <u>delegate of Indonesia</u> drew attention to the text of Document No. 249, submitted by his delegation. The object of the request was to ensure that the idea underlying the document - namely, that principles previously applied to the use of space should sometimes be discarded in the light of technological progress - might be considered at future conferences of the ITU.

The <u>Chairman</u> thanked the delegate of Indonesia for submitting that very useful document.

The meeting rose at 1120 hours.

The Secretary-General :

The Chairman : Ib LØNBERG

M. MILI

INTERNATIONAL TELECOMMUNICATION UNION

BROADCASTING SATELLITE CONFERENCE

Document No. 353-E 17 March 1977 Original : English

(Geneva, 1977)

PLENARY MEETING

MINUTES

OF THE

NINTH PLENARY MEETING

Wednesday, 9 February 1977, at 1530 hrs

Chairman : Mr. Ib LØNBERG (Denmark)

Subjects discussed

Document Nos.

Texts submitted by the Editorial Committee (Add.1 to B.4, B.10)

225(Add.1), 246

Texts submitted by the Editorial Committee (Add.l to B.4, B.10) (Documents Nos. 225(Add.l), 246)

Addendum 1 to the 4th Series of texts submitted by the Editorial Committee to the Plenary Meeting (Add.1 to B.4) (Document No. 225(Add.1))

Page 2

The <u>Chairman</u> said that all the square brackets in the text could now be removed.

The <u>Chairman</u> of Committee 6 said that the word "occupied" should be replaced by "necessary" in the second and third lines of the text.

The <u>delegate of the United Kingdom</u> proposed that the words "to be" should be inserted before "adopted" in the seventh line and that the words "of Regions 1 and 3 plan" in the footnote should be replaced by "in the Plan".

Page 2, as amended, was approved.

Page 3

The <u>delegate of Italy</u> observed that, since all the assignments in the Plan were to be recorded in the Master Register, it was not clear what entries were to be made in the Register under paragraph 5.2.7.



The <u>Chairman of Committee 6</u>, the <u>delegate of the United Kingdom</u> and the <u>Chairman of the IFRB</u> explained that the provision related to a clearly defined case, that of confirmation of the bringing into use of a frequency assignment notified in advance.

The <u>delegate of Italy</u> proposed that the words "in conformity with paragraph 5.1.3" be inserted after "bringing into use", to avoid any misunderstanding.

Page 3, as amended, was approved.

Page 4

The <u>Chairman of the Editorial Committee</u> pointed out that the English text of the miscellaneous provisions had been changed in her Committee, although some delegations had thought it best to retain the existing text of the Radio Regulations, which had been adopted by Committee 6 in Document No. 220.

The <u>Chairman of Committee 6</u> proposed that the original text should be retained.

The <u>delegate of the United Kingdom</u> supported that proposal. Although the existing texts of the Radio Regulations might leave room for improvement, some of them were the result of compromises reached with difficulty at Administrative Conferences, and it would be inadvisable at that stage of the proceedings to put delegations in a position where they had to examine revised texts to ensure that no substantive changes had been introduced.

It was <u>agreed</u> that the original text of the miscellaneous would be restored in the R. Series.

The Addendum to the Fourth Series of texts (Add.1 to B.4) was <u>approved</u> on first reading, as amended and subject to editorial changes.

10th Series of texts submitted by the Editorial Committee to the Plenary Meeting (B.10) (Document No. 246)

The <u>Chairman of the Editorial Committee</u> said that the French and Spanish versions of the last paragraph on page 2 should be aligned with the English text. The words "(Malaga-Torremolinos, 1973)" should be added after the definition of "Convention" in Article 1 on page 4. Finally, the words "stations of" should be inserted before "existing and future" in "considering" paragraph b) of Resolution No. / C / on page 10.

Page 2

The <u>Chairman of Committee 6</u> said that the words "and the Additional Radio Regulations" at the end of the fourth indented paragraph should be deleted.

The <u>delegate of the United Kingdom</u> said that the words "and in accordance with" should be inserted before "Resolution No. 27" in the first paragraph, to bring the text into line with the one adopted by Committee 6.

The <u>delegate of the Netherlands</u> observed that the words "has been charged" at the end of that paragraph had been erroneously omitted from the English text and proposed that they should be restored.

The <u>delegate of Algeria</u> proposed that the word "<u>même</u>" in the French text of the last paragraph on page 2 be replaced by "y compris".

Page 2, as amended, was approved.

Page 3

The <u>Chairman of Committee 6</u> said that he had discussed the lay-out of the Final Acts with the Chairman of the Editorial Committee and that they had decided to suggest the texts in square brackets in order to take into account the Conference's decisions on the rearrangement of the Radio Regulations. They also considered that it would be more appropriate to place the signatures immediately after the preamble : it had originally been thought that the Final Acts would consist of a few articles on such general matters as definitions and that the procedures would take the form of appendices, but now that it had been decided to include the procedures in the articles themselves, it would be more logical to have the signatures before rather than after the articles. He suggested that the square brackets round the new phrase be removed.

It was so <u>decided</u>.

The <u>delegate of Chile</u> said that it was important to include a specific reference to the provisions whereby the Broadcasting-Satellite Service in Region 2 would be governed until a detailed plan for that Region was drawn up and proposed wording to that effect.

After a discussion during which the <u>Chairman of Committee 6</u> and the <u>delegates of the USSR</u> and <u>the Netherlands</u> pointed out that the broad formulation of the preamble covered the provisions relating to all the Regions, the <u>delegate</u> <u>of the United Kingdom</u> proposed that the phrase "relating to the Broadcasting-Satellite Service in the above-mentioned bands" in the second and third lines be deleted.

The <u>delegate of Chile</u> said that that change would meet his delegation's preoccupations.

Page 3, as amended, was approved.

Article / 1 / - General definitions

The <u>Chairman of Committee 6</u> said that the square brackets could be removed from the last two lines of the definition of "Conference".

Article / 1 / was approved.

Article /2/ - Frequency bands

Article / 2 / was approved.

Article $\overline{/3}$ - Execution of the Final Acts

The <u>Chairman of Committee 6</u> pointed out that the words "and Annexes" should be inserted after "Articles" in the penultimate line.

Article $\overline{3}$, as amended, was <u>approved</u>.

Article $\overline{/8}$ - Approval of the Final Acts

The <u>delegate of the United Kingdom</u>, supported by the <u>delegate of</u> <u>New Zealand</u>, proposed that the first two sentences of the English text should be combined to read "Members shall notify their approval of these Final Acts, as promptly as possible, to the Secretary-General, who shall at once inform the other Members of the Union", as in Article a on page 4 of Document No. 227.

The <u>delegate of Chile</u>, observing that the Final Acts would come into force in 1979 whereas the Conference to establish a plan for Region 2 would not be held until 1982, suggested that a reference to the provisions governing the Broadcasting-Satellite Service in Region 2 be made in the last sentence of Article / 8/.

The <u>Chairman of Committee 6</u> said it had been made clear in the long discussions on the subject in Working Group 6D and Committee 6 that the World Agreement applied to all three Regions and that the associated Plan applied only to Regions 1 and 3.

The <u>delegate of Italy</u> added that the insertion of a specific reference to the provisions governing Region 2 would run counter to Resolution No. 3, providing that Resolution No. Spa2 - 3 would continue to apply to Region 2 until a plan was drawn up for that Region.

The <u>delegate of Chile</u>, supported by the <u>delegate of Bolivia</u>, said he saw no reason why a <u>refer</u>ence to the provisions governing Region 2 should not be made in Article $\underline{/8}$, when they were already specifically mentioned in Article $\underline{/3}$, paragraph 2.

The Deputy Secretary-General suggested that the separate reference to World Agreement was necessary but it was not exclusive. However, if the Plenary Meeting wished to meet the Chilean delegate's wish, the best solution would be to add a separate sentence concerning Region 2 at the end of Article $\frac{78}{0}$.

The <u>delegate of the United Kingdom</u> observed that the purpose of the Article was to draw the attention of Administrations to the way in which the current Conference had implemented Resolution No. Spa2 - 2 of the 1971 Space Conference and to the obligations they would assume in approving the Final Acts.

The proposed reference to interim arrangements for Region 2, whether as an insertion in the existing last sentence or as a separate sentence, would only distort that general purpose; clearly, Resolution No. Spa2 - 2 no longer applied to Regions 1 and 3, because a Plan had been drawn up for these Regions, but continued to apply to Region 2.

Those remarks were endorsed by the <u>Chairman of Committee 6</u>, and the <u>delegate of the Ukrainian SSR</u>.

After a discussion during which the <u>delegates of the United States</u> of <u>America</u> and <u>Mauritania</u> suggested rearrangements of Article 8 to take the Chilean <u>delegation's</u> point into account, the <u>delegate of the USSR</u> proposed that Article / 8 / be divided into two paragraphs, the first consisting of the two opening sentences and the second of the last sentence.

The delegate of Chile accepted that solution.

Article / 8 / , as amended, was approved.

Article $\sqrt{9}$ - Entry into force of the Final Acts

On a proposal by the <u>Chairman of Committee 6</u>, it was <u>agreed</u> to remove the square brackets from the date "1 January 1979".

Article / 9 /, as amended, was approved.

Article / 10 / - Period of validity of the provisions and associated Plan

The <u>Chairman of Committee 6</u>, referring to paragraph 1, said that periods of between 10 and 15 years had been envisaged during the discussions held on the question, but no specific proposal was submitted by the Committee.

The <u>delegate of India</u> said that there were two conflicting requirements to be met when determining the period of validity of the Plan. First, sufficient flexibility must be provided to enable countries, particularly developing ones, to benefit from the rapid development of technology in the broadcastingsatellite field and, second, changes must not be introduced in the Plan too rapidly or too early if certain practical constraints and problems were to be avoided. In order to reconcile those two requirements, his delegation considered that a maximum period, beyond which the Plan would be considered as no longer valid, should also be set. Accordingly, he proposed that 10 and 15 years from the present time (8 and 13 years from the date of entry into force of the Final Acts) should be set, respectively, as the minimum and maximum periods.

The <u>Chairman of Committee 6</u> said that the course suggested by the Indian delegate had been discussed in Committee 6 and rejected in favour of the formula in the text under discussion. If a maximum period were to be set, a Conference would have to be convened at the end of it, even if there was no real need for one.

The <u>delegate of the USSR</u> observed that it would be possible to convene a Conference to revise the Plan at any time, irrespective of what period was specified in the Final Acts, by following the procedure laid down in Article 62 of the Convention. His delegation considered that the period of validity which would best serve the interests of the developing countries was 15 years, but it would be prepared to accept any figure which the Conference might favour.

The <u>delegate</u> of <u>Senegal</u> said that he too preferred 15 years but would accept the wish of the majority.

The <u>delegates of Nigeria</u>, <u>Zaire</u> and <u>Laos</u> said that they favoured 15 years.

The <u>delegate of China</u> said that the period of validity should be at least 15 years.

The <u>delegate of Papua New Guinea</u> expressed a preference for 15 years, and said that he could not agree to the proposal by the Indian delegate.

The <u>delegate of the United Kingdom</u> supported the views of those delegations that favoured a 15-year period. He considered that no changes should be made in the wording of paragraph 1 of the Article.

The <u>delegate of the Federal Republic of Germany</u> said that he too considered 15 years to be an appropriate period. However, there might be some merit in placing that figure between square brackets until delegations had taken cognizance of the Plan itself.

The <u>delegate of Italy</u> observed that, should the need arise, the figure could always be changed when the Article was submitted to the Plenary Meeting for second reading.

Following some further discussion, in which the <u>delegates of India</u>, Zaire, the United Kingdom, France, <u>Mauritania</u> and the USSR took part, it was <u>decided</u> to insert the figure "15" in the blank space in paragraph 1.

On a proposal by the <u>delegate of France</u>, it was <u>agreed</u> to replace the words "Final Acts" by "provisions and associated Plan" in the first line of paragraph 2.

The <u>delegate of Algeria</u> considered that the word "competent" in the second line of paragraph 2 should be replaced by the word "World". If the Final Acts were to be regarded as including a World Agreement, it was difficult to envisage their being revised other than by a World Conference.

Following a discussion during which the <u>delegates of the United</u> <u>Kingdom</u>, <u>Canada</u> and the <u>United States of America</u> opposed the suggested change and the <u>Chairman of Committee 6</u> observed that no Conference, whatever its status, was likely to take decisions that it was not competent to take, the <u>delegate of Algeria</u> said that he would not press his proposal. Article / 10 /, as amended, was approved.

Resolution No. / A /

On a proposal by the <u>delegate of the United Kingdom</u> it was <u>agreed</u> to delete the square brackets from the heading and also the brackets which appeared in the English text of the <u>resolves</u> paragraph.

On a proposal by the <u>delegate of Canada</u>, supported by the <u>delegate</u> <u>of Japan</u>, it was agreed to delete considering b).

Resolution No. / A /, as amended, was approved.

Resolution No. / B_/

<u>Approved</u>, subject to deletion of the brackets in <u>considering</u> a), replacement of the square brackets in <u>considering</u> b) by the letter "A", and substitution of the word "are" for the words "shall be" in the first line of the further considering paragraph.

Article / _/ - Interference

The <u>Chairman of Committee 8</u> said that Committee 8 would examine other texts of a similar nature in order to determine the most appropriate place for the Article in the Final Acts.

The <u>Chairman of Working Group 6D</u> considered that the original wording of the Article as it appeared in Document No. DT/40 was preferable to that proposed by Committee 8. Accordingly, he proposed that the text should be amended to read : "The Members of the Union shall endeavour to agree on the action required to reduce harmful interference caused by the application of these provisions and the associated Plan".

The delegates of Japan and Switzerland supported that proposal.

The <u>delegate of Mauritania</u> said that the proposal would be acceptable to him if the words "which might be" were inserted before the word "caused".

The text proposed by the Chairman of Committee 6, as amended by the delegate of Mauritania, was approved.

In reply to a question by the <u>Chairman of Committee 8</u>, the <u>Chairman</u> said that the English text which had just been approved could be considered as valid and that the other two language versions should be brought into line with it if there were any discrepancies.

Resolution No. C

The <u>Chairman</u> said that the square brackets in the fourth line of the French text of considering a) should be removed.

The <u>Chairman of Committee 6</u> said that "/ subsequent $\overline{/}$ " should be deleted from considering c).

The <u>delegate of Papua New Guinea</u> said that the word "Plans" in <u>considering</u> c) should be replaced by "plans".

Those changes were approved.

The <u>delegate of the United Kingdom</u>, referring to <u>resolves</u> 1, proposed that the phrase "broadcasting-satellite stations provided for in the Plan" should be replaced by the phrase "frequency assignments in accordance with the Plan".

It was so decided.

The <u>Chairman of Committee 6</u> observed that it would be necessary to make consequential amendments to <u>resolves</u> 2 in order to bring the text into line with the changed wording just approved for resolves 1.

It was so agreed.

The <u>Chairman of Committee 6</u> said that the word "which" should be inserted after "frequency assignments" in the first line of <u>resolves</u> 3. In <u>resolves</u> 6, the word "recorded" should be replaced by the word "entered".

It was so agreed.

Resolution No. C, as amended, was approved.

The meeting rose at 1900 hours.

The Secretary-General :

The Chairman : Ib LØNBERG

M. MILI

INTERNATIONAL TELECOMMUNICATION UNION

BROADCASTING SATELLITE CONFERENCE

(Geneva, 1977)

Document No. 354-E 11 February 1977 Original : English

COMMITTEE 6

SUMMARY RECORD

OF THE

NINTH AND LAST MEETING OF COMMITTEE 6

Sunday, 6 February 1977, at 1120 hrs

Chairman : Mr. R.J. BUNDLE (New Zealand)

Subjects discussed :

 Consideration of draft recommendation relating to the use by space radiocommunication services of only the geostationary orbit in the frequency bands 11.7 - 12.2 GHz (in Regions 2 and 3) and 11.7 - 12.5 GHz (in Region 1)

DL/53

DL/55, DL/56, DL/57

227 + Add.1, DL/54

188(Rev.2)

GFNE

Documents Nos.

- Consideration of texts drafted by Committee 6 Ad Hoc Group
- 3. Consideration of Draft texts submitted by Working Group 6D
- 4. Consideration of Recommendations by Working Group 4B
- 5. Completion of the Committee's work

1.

Consideration of draft recommendation relating to the use by space radiocommunication services of only the geostationary orbit in the frequency bands 11.7 - 12.2 GHz (in Regions 2 and 3) and 11.7 - 12.5 GHz (in Region 1) (Document No. DL/53)

The <u>Chairman</u> said that the draft recommendation (Document No. DL/53) might be considered superfluous by some delegations since the point it dealt with could be considered as covered by No. 470VA of the Radio Regulations, but in his view that was not quite adequate for the protection of the Plan under discussion and the draft recommendation constituted an additional safeguard.

The <u>delegate of the United Kingdom</u>, supported by the <u>delegates of</u> <u>Norway</u> and <u>the Federal Republic of Germany</u>, proposed that the draft recommendation be converted into a draft resolution, and that the words "<u>recommends</u>" et seq. be deleted and replaced by :

"resolves

1. that Administrations shall ensure that their space radiocommunication services in these frequency bands are operated only in the geostationary orbit."

That amendment was approved.

The <u>delegate of Colombia</u> said that No. 470VA of the Radio Regulations referred to the Fixed-Satellite Service; in his view the relevant provisions of the Radio Regulations were No. 422 and No. 962. He proposed the addition in <u>considering</u> c) after the words "incompatible with" of the words "the Radio Regulations and.."

The <u>delegate of the United Kingdom</u> opposed the Colombian amendment, and pointed out that although No. 962 dealt with the operation of a broadcasting service by mobile stations on the sea and over the sea, it was clear from the terms and definitions in Article 1 of the Radio Regulations that mobile services were not mobile satellite services.

In the absence of support for the Colombian amendment, the <u>Chairman</u> declared it rejected.

The <u>delegate of Colombia</u> stated that Article 28, No. 962 of the Radio Regulations laid down that "the operation of a broadcasting service by mobile stations at sea and over the sea is prohibited."

It did not specify any restriction on the height at which such stations were located "over the sea"; consequently, such services could only be operated from fixed stations, since the Regulations only provided for those two methods for all the services with which they dealt.

The <u>delegate of Poland</u>, supported by the <u>delegates of the USSR</u> and <u>Colombia</u>, said that <u>considering</u> b) should be placed in square brackets, for reference to the plenary, since the holding of the Regional Administrative Radio Conference to which it referred had not yet been approved by the Conference.

The <u>delegate of the USSR</u> said that the role of the various Regions in the proposed Regional Conference needed to be more clearly specified. Also he was not sure how closely the limited use of the bands by space radiocommunication services corresponded to the Radio Regulations. He therefore proposed that the whole draft resolution should be placed in square brackets to indicate the need for its further consideration by the plenary.

It was so <u>agreed</u>.

The <u>United Kingdom delegate</u> pointed out that the title of the Conference in the first line should be amended to bring it into line with that used in Document No. 227, and the <u>Chairman</u> said that that would be done wherever the title appeared in the documents under consideration.

The draft resolution (Document No. DL/53) as amended by the United Kingdom delegate, and placed in square brackets, was adopted.

2.

Consideration of texts drafted by Committee 6 Ad Hoc Group (Documents Nos. DL/55, DL/56 and DL/57)

The delegate of Australia, speaking as Chairman of the Committee 6 Ad Hoc Group, introduced the Corrigendum to Document No. 210 (Document No. DL/55), the draft resolution relating to the coordination, notification and recording of broadcasting-satellite stations in Region 2 (Document No. DL/56), and the informative draft on power flux-density limits between 11.7 - 12.2 GHz (Document No. DL/57), and announced editorial corrections to Documents Nos. DL/55 and DL/56.

Corrigendum to Document No. 210 (Document No. DL/55)

The delegate of Poland said that, as in the case of the document just discussed, his delegation had objections to the reference to a Region 2 Regional Conference, and he reserved the right to return to the matter in plenary.

The Chairman said that the Polish delegate's reservation had been noted.

The <u>delegate of the United Kingdom</u> proposed the deletion of the words from "to be adopted" to the end of the indented paragraph; the insertion of the word "is" after the words "Region 2 Plan" in the first line of footnote $/ l_{/}$; and the replacement of the word "shall" by the words "and will" in the second line of the footnote.

The delegate of the United States of America opposed that amendment, on the grounds that the words "including modifications thereto" should be retained.

Supported by the delegate of Canada, he proposed deletion of footnote / 1 7 which had been intended merely as an explanatory insertion for Committee 5.

The delegate of Italy said the footnote should remain, since otherwise frequency assignments in Regions 1 and 3 would not be protected from assignments in Region 2, whereas those in Region 2 would be protected from Regions 1 and 3.

The delegate of the USSR endorsed the Italian delegate's comment, and supported the proposed United Kingdom amendment. In view of the somewhat confused nature of the discussion, he reserved the right to return to the subject in plenary.

The United Kingdom delegate withdrew his amendment.

On the proposal of the <u>delegate of India</u>, it was <u>agreed</u> to delete the word "detailed" before the words "Region 2 Plan" in both the indented paragraph and the footnote.

The Chairman said that in view of the lack of unanimity regarding its deletion, footnote 1) would be retained in square brackets.

Document No. DL/55, as amended, and with the editorial corrections given by the Chairman of the Committee 6 Ad Hoc Group, was approved.

Draft resolution relating to the coordination, notification and recording of broadcasting-satellite stations in Region 2 (Document No. DL/56)

The <u>delegate of the United Kingdom</u> proposed that the word "registration" should be changed to "recording" in <u>resolves</u> 2 and 3, to bring the language into conformity with that used in other Conference documents. He further proposed the addition of the words "in Region 2" at the end of considering d).

It was so agreed.

The <u>delegate of Italy</u> proposed that in the third line of <u>resolves</u> 3 the words "for which a frequency assignment appears in the Plan" should be inserted after "broadcasting satellite stations", and agreed that the subsequent reference to Regions 1 and 3 was unnecessary and should be dropped.

The Italian amendment was adopted.

The <u>delegate of the United Kingdom</u>, supported by the <u>delegate of Italy</u>, proposed that in second line of <u>considering</u> c) the words "for that region" should be inserted after the word "Plan".

It was so agreed.

Document No. DL/56, as amended, and with the editorial corrections given by the Chairman of the Committee 6 Ad Hoc Group, was approved.

Power flux-density limits between 11.7 - 12.2 GHz (Document No. DL/57)

The <u>Chairman of Committee 6 Ad Hoc Group</u> said that the document had been prepared by a member of the United Kingdom delegation serving on the ad hoc joint coordination group of Committees 4 and 6.

The <u>delegate of the United Kingdom</u>, introducing the document, said that it contained a draft regulatory text which was designed to ensure that the Broadcasting-Satellite Service in Region 2 did not cause interference to terrestrial services in Regions 1 and 3, and which might be incorporated in the Final Acts as a separate article. The following phrase should be added at the end of the text, after the word "values" : "on the territory of any country unless the administration of that country so agrees". The figures given in paragraphs 1 and 2 were only examples and would have to be replaced by those finally approved by Committee 4. Consequently, he suggested that those two paragraphs might be placed within square brackets pending the outcome of Committee 4's work.

The <u>delegate of Italy</u>, supported by the <u>delegate of the United States</u> of <u>America</u>, considered that the document should be referred to the ad hoc joint coordination group of Committees 4 and 6, which had the task of preparing technical appendices to the articles concerned with coordination and notification procedures. The <u>delegate of the United Kingdom</u> said that the figures would certainly have to be incorporated in the appropriate technical appendix, but that a regulatory text in the form of a separate article was also required. Accordingly, he proposed that the last part of the text should be amended to read "... shall not exceed the values given in Appendix / ... / on the territory of any country ...", and that paragraphs 1 and 2 containing the figures should be deleted.

The <u>delegates of the Federal Republic of Germany</u>, <u>Italy</u> and the <u>USSR</u> supported that proposal.

The <u>Chairman</u> said that if he heard no objection he would take it that the text proposed by the United Kingdom delegate was acceptable to the Committee and could be transmitted to the joint coordination group of Committees 4 and 6 to assist it in its deliberations.

It was so agreed.

Draft Resolution relating to the coordination, notification and recording in the Master International Frequency Register of fixedsatellite stations in relation with broadcasting-satellite stations in Region 2 (Document No. DL/X)

The <u>Chairman of Committee 6 Ad Hoc Group</u> introduced the draft Resolution.

On a proposal by the <u>delegate of Denmark</u>, it was <u>agreed</u> to insert the words "stations in" before the words "the Broadcasting-Satellite Service" at the end of the <u>considering</u> paragraph.

The draft Resolution, as amended, was approved.

3.

Consideration of draft texts submitted by Working Group 6D (Document No. 227 + Add. 1)

The <u>Chairman of Working Group 6D</u> drew attention to Addendum No. 1 to Document No. 227 which contained the text of a draft Resolution relating to the updating of the Master International Frequency Register for Regions 1 and 3. A proposal made towards the end of the Working Group's discussions by the Algerian delegation for the addition of a new paragraph under <u>resolves</u> had been referred to a small group of interested delegations for consideration.

The <u>Chairman</u> said that, as a result of the discussions held between the Algerian and other delegations and the IFRB, agreement had been reached on the following text for a new paragraph <u>resolves</u> 6:

> "that, on the date of entry into force of the Final Acts, the frequency assignments in the Plan will be recorded in the Master Register. The date of signature of the Final Acts, together with an appropriate symbol, will be entered in Column 13c opposite these assignments."

Following a discussion in which the <u>delegates of the United Kingdom</u>, the <u>Federal Republic of Germany</u> and <u>Algeria</u> took part, the text proposed for <u>resolves</u> 6, was <u>approved</u>.

On a proposal by the <u>delegate of Norway</u>, it was <u>agreed</u> to insert the words "in accordance" after the word "operating" in the last line of <u>considering</u> b).

The draft Resolution in Addendum No. 1 to Document No. 227, as amended, was approved.

The <u>delegate of Italy</u>, referring to Article 3 (Document No. 227, page 4), asked what action had been taken in respect of paragraph 3 of the draft which had appeared originally in Document No. DT/40.

The <u>Chairman</u> read out the paragraph in question and said that, if the Committee agreed, it would be included in the Final Acts as a separate Article entitled "Interference". Committee 8 would decide on the most appropriate place for it in the Final Acts.

It was so agreed.

4. <u>Consideration of recommendations by Working Group 4B</u> (Document No. 188(Rev.2))

The <u>delegate of Italy</u> observed that the recommendations in paragraphs 3.1 and 3.2 of Document No. 188(Rev.2) were relevant to the work of Committee 6, since draft Recommendations or Resolutions relating to the matters with which they dealt would have to be prepared.

The <u>Chairman</u> said that the recommendations by Working Group 4B to which the previous speaker had referred could only be included in the Appendices to the procedural provisions, which were being discussed by the joint coordination group of Committees 4 and 6. It would not be possible for those recommendations to be taken into consideration by Committee 6; the only possible course would be to bring them to the attention of the joint group.

Following a discussion in which the <u>delegates of Italy</u> and <u>the</u> <u>United Kingdom</u>, the <u>Chairman of Committee 5</u> and the <u>Chairman</u> took part, it was <u>agreed</u> that the Chairman would discuss the problem with the Chairmen of Committees 4 and 5 in order to determine the most appropriate course to take.

5. Completion of the Committee's work

The <u>Chairman</u> suggested that the joint coordination group of Committees 4 and 6 should be requested to report the results of its work to himself and the Chairman of Committee 4; the draft texts it produced would then be examined by Committee 8 and submitted directly to the Plenary Meeting.

Ł

It was so <u>agreed</u>.
The <u>Chairman</u> said that the Committee had now completed its work, which had not been easy. He wished to thank the Chairmen of the various Working Groups as well as the Secretary and his staff for all their help to the Groups and to the Committee as a whole. His only criticism was that the differing views of the permanent organs of the Union had at times made difficult situations no easier to resolve.

The <u>delegate of the United Kingdom</u> said that the Committee's task had been particularly arduous due to the impossibility of considering regulatory procedures without ambiguity in the absence of full knowledge of all the other factors involved. He paid a warm tribute to the Chairman for the masterly way in which he had conducted the Committee's deliberations.

The meeting rose at 1345 hours.

The Secretary :

R. PLUSS

The Chairman : R.J. BUNDLE

INTERNATIONAL TELECOMMUNICATION UNION

BROADCASTING SATELLITE CONFERENCE

(Geneva, 1977)

Document No. 355-E 12 February 1977 Original : English

PLENARY MEETING

United States of America

DOCUMENT No. 248

With reference to the statement made by the delegation of the Republic of Panama in Document No. 248, the delegation of the United States of America wishes to note that negotiations between the United States and the Republic of Panama on a new treaty governing the Panama Canal are presently under way. It is the hope and aim of the United States of America that a new treaty will be in effect by the time a Broadcasting-Satellite Service is a reality.



Document No. 356-E 12 February 1977 Original : French

(Geneva, 1977)

PLENARY MEETING

FINAL PROTOCOL

For the Kingdom of Morocco and the Islamic Republic of Mauritania

The delegations of the Kingdom of Morocco and the Islamic Republic of Mauritania, having taken note of the declaration made by the Algerian Democratic and Popular Republic in Document No. 333, wish to recall that the dimensions of the beams intended for the coverage of their territories were defined by the planning bodies set up by the Conference. These beams are in full conformity with the provisions of No. 428A of the Radio Regulations.

It should also be noted that the Saharan provinces of the Kingdom of Morocco and the Islamic Republic of Mauritania were recovered legally under the aegis of the international authorities and are an integral part of these countries.

The declaration of the Algerian delegation contained in Document No. 333 is one of the last symbolic acts carried out by the pseudo-revolutionary and expansionist regime of this country after the vain attempts made to impose its will on our Saharan territories.

The Moroccan and Mauritanian delegations consider this declaration to be an act of flagrant interference in their internal affairs and call on the Conference to consider it null and void.



Document No. 357-E 12 February 1977 Original : English

(Geneva, 1977)

PLENARY MEETING

FINAL PROTOCOL

For Japan

The delegation of Japan wishes to state that the ellipse of Japan is established in such a manner as to obtain a most efficient and reasonable broadcasting-satellite system for covering the territory of Japan, while taking every care technically possible so as to reduce to the maximum extent practicable the radiation over the territory of other countries in accordance with the provisions of No. 428A of the Radio Regulations.

In connection with the reservations made by the delegation of any country in this respect, the delegation of Japan reserves for its Administration all the rights necessary to safeguard its interests.

GENÈNE

INTERNATIONAL TELECOMMUNICATION UNION

BROADCASTING SATELLITE CONFERENCE

(Geneva, 1977)

Document No. 358-E 12 February 1977 Original : English

PLENARY MEETING

FINAL PROTOCOL

For Japan

With reference to the reservation made by the delegation of the Democratic People's Republic of Korea which appears in Document No. 344, the delegation of Japan wishes to state as follows :

The ellipse of Japan is established in such a manner as to obtain a most efficient and reasonable broadcasting-satellite system for covering the territory of Japan, while taking every care technically possible so as to reduce to the maximum extent practicable the radiation over the area outside its own territory in accordance with the provisions of No. 428A of the Radio Regulations. As it is established solely from technical point of view and contains no political elements, the delegation of Japan cannot accept the contention of the delegation of the Democratic People's Republic of Korea in this respect.

The delegation of Japan therefore states that its Administration reserves all the rights necessary to safeguard its interest in connection with the reservations made by the delegation of the Democratic People's Republic of Korea.



(Geneva, 1977)

Document No. 359-E 12 February 1977 Original : English

PLENARY MEETING

FINAL PROTOCOL

For Japan

With regard to the declarations made by several equatorial countries in Documents Nos. 340 and 344 in connection with the sovereignty over the geostationary orbit, the delegation of Japan wishes to declare, on behalf of the Japanese Government, that the ITU is not competent to deal with this question and that the delegation of Japan cannot agree with what was expressed in the said documents because of the world-widely accepted principle that outer space is not subject to national appropriation by claim of sovereignty.



INTERNATIONAL TELECOMMUNICATION UNION

BROADCASTING SATELLITE CONFERENCE

Document No. 360-E 12 February 1977 Original : English

(Geneva, 1977)

PLENARY MEETING

FINAL PROTOCOL

For the Federal Republic of Germany, Austria, Belgium, Canada, Denmark, United States of America, Finland, France, Ireland, Italy, Luxembourg, Monaco, Norway, Kingdom of the Netherlands, United Kingdom of Great Britain and Northern Ireland, Sweden

With reference to reservations by the Republic of Colombia, the People's Republic of the Congo, the Republic of Ecuador, the Republic of Gabon, the Republic of Kenya, the Republic of Uganda, the Republic of Zaire the abovementioned delegations consider that such claims cannot be recognized by this Conference and declare that the decisions of this Conference to assign frequencies and orbital positions in the geostationary orbit are fully in accordance with the Convention of the International Telecommunication Union, Malaga-Torremolinos 1973, by which this Conference is bound.

GENÈVE

Document No. 361-E 12 February 1977 Original : English

(Geneva, 1977)

PLENARY MEETING

FINAL PROTOCOL

For the Republic of India

With reference to the Final Protocol for the People's Republic of China as contained in Document No. 322, the delegation of India wishes to state the following.

India categorically refutes the contention of the People's Republic of China that certain regions of Chinese territory are shown as Indian territory in the service areas of beams IND 0037 and IND 0038. All the polygon points for these two beams are within India. The service areas of these beams are an integral part of India.

FNÈ

INTERNATIONAL TELECOMMUNICATION UNION

BROADCASTING SATELLITE CONFERENCE

(Geneva, 1977)

Document No. 362-E 12 February 1977 Original : English

PLENARY MEETING

FINAL PROTOCOL

For the Republic of Korea

With reference to the statement submitted in Document No. 345, the delegation of the Republic of Korea wishes to make the following statement :

1. The delegation of the Republic of Korea categorically rejects the unfounded allegation of the North Korean authorities with regard to the fair and just ellipse of the Republic of Korea for the Broadcasting-Satellite Service.

The delegation of the Republic of Korea, as it stated many times in the various meetings of this Conference, once again makes it clear that the ellipse for the Republic of Korea is the minimum requirement for its broadcasting services and that the Republic of Korea has no intention to affect any intentional spillover to any adjacent countries.

2. The delegation of the Republic of Korea totally rejects the groundless and malicious political polemics of North Korean authorities which have no relevance at all in light of the objectives and technical nature of this Conference.

3. The delegation of the Republic of Korea declares that it does not recognize the ellipse of North Korean authorities which extends into the territory of the Republic of Korea and further declares that it will take all necessary measures to protect the legitimate interests of the Republic of Korea in this regard.



(Geneva, 1977)

Document No. 363-E 12 February 1977 Original : French English Spanish

PLENARY MEETING

Note from the Secretary-General

FINAL PROTOCOL

The following documents (handed in before 1200 hours on 12 February 1977) have been issued in reply to some of the reservations listed in Document No. 269 and contain texts to be published in the Final Protocol of the Final Acts of the Conference :

Document No.	Submitted by
356	Kingdom of Morocco, Islamic Republic of Mau r itania
357	Japan
358	Japan
3 59	Japan
360	Federal Republic of Germany, Austria, Belgium, Canada, Denmark, United States of America, Finland, France, Ireland, Italy, Luxembourg, Monaco, Norway, Kingdom of the Netherlands, United Kingdom of Great Britain and Northern Ireland, Sweden
361	Republic of India
362	Republic of Korea
364	Spain
366	Algerian Democratic and Popular Republic
367	Algerian Democratic and Popular Republic

M. MILI

Secretary-General

For reasons of economy, this document is printed in a limited number. Participants are therefore kindly asked to bring their copies to the conference since only a few additional copies can be made available.

INTERNATIONAL TELECOMMUNICATION UNION

BROADCASTING SATELLITE CONFERENCE

(Geneva, 1977)

Document No. 364-E 11 February 1977 Original : Spanish

PLENARY MEETING

FINAL PROTOCOL

For Spain

The Spanish delegation rejects the allusion made to Spain in a reservation referring to the Western Sahara.

In a communication dated 26 November 1975 to the Secretary-General of the United Nations from the Ambassador and Permanent Representative of Spain to that organization (Document A/31/56/S11997), the Spanish Government announced that it had on that date definitively brought to an end its presence in the territory of the Sahara and noted that it considered itself henceforth released of all international responsibility with regard to the administration of that territory. With regard to the attitude of the Spanish Government in connection with the Western Sahara, the Spanish delegation refers to the communications and declarations submitted by Spain to the competent bodies of the United Nations.

(Geneva, 1977)

Document No. 365-E 12 February 1977 Original : English

PLENARY MEETING

Union of Soviet Socialist Republics

DOCUMENTS Nos. 272 AND 273

In connection with the issue of Documents No. 272 and No. 273 the delegation of the USSR considers it necessary to state its disagreement of principle with their contents, in particular in respect of interpretation of provisions of Quadripartite Agreement of 3 September 1971. Paragraph a) of the Document No. 272 constitutes in fact an attempt of interference in the internal affairs of a sovereign state - the German Democratic Republic. As is well known, the city of Berlin is the capital of the German Democratic Republic and it is only natural that the Quadripartite Agreement of 3 September 1971 does not apply to it.

The delegation of the USSR confirms its declaration made in Committee 2 on 7 February (see Annex to Document No. 271).

Document No. 366-E 12 February 1977 Original : French

(Geneva, 1977)

PLENARY MEETING

FINAL PROTOCOL

For the Algerian Democratic and Popular Republic

In its declaration reproduced in Document No. 346, the delegation of Tunisia alludes to "the coverage area resulting from the assignments to certain neighbouring countries".

In case this allusion should concern the Algerian Democratic and Popular Republic, the Algerian delegation formally draws the attention of the Conference to the fact that the beam 150 TUN allotted to Tunisia intentionally covers part of the territory of Algeria and is therefore subject to the application of No. 428A of the Radio Regulations.



Document No. 367-E 12 February 1977 Original : French

(Geneva, 1977)

PLENARY MEETING

FINAL PROTOCOL

For the Algerian Democratic and Popular Republic

In Document No. 347 the Moroccan delegation claims that it proposed a compromise solution to the Algerian delegation and that the Algerian delegation refused to adopt this solution.

The Algerian delegation wishes to draw attention to the following points :

- 1) The compromise solution was proposed by the Algerian delegation and not by the Moroccan delegation.
- 2) The Algerian delegation consequently submitted the official documents required to the planning groups.
- 3) The Algerian delegation totally rejects the false and misleading statements made in Document No. 347.

GENÈ

INTERNATIONAL TELECOMMUNICATION UNION

BROADCASTING SATELLITE CONFERENCE

(Geneva, 1977)

Document No. 368-E 12 February 1977 Original : English

COMMITTEE 5

Working Group 5A

NOTES CONCERNING NEGATIVE MARGINS AND PROTECTION RATIOS FOR THE BROADCASTING-SATELLITE SERVICE WITHIN THE REGION 1 AND 3 PLAN

1. Introduction

The Conference adopted a protection ratio (the ratio of the wanted signal-to-total interferences) of 30 dB, plus a 1 dB additional planning margin. A value as high as 30 dB was chosen on the basis that a new service, such as the Broadcasting-Satellite Service, would need to provide a good quality of reception, even with pictures which are particularly susceptible to the effects of interference. The 1 dB planning margin had several justifications. It provides a small allowance for transmitter-tube aging and for other factors.

The Working Group 5A have produced a plan for Region 1 and 3 which had as its objective the above standard in respect of protection ratio. It was convenient for planning purposes to refer to any carrier-to-interference ratio which did not achieve 1 dB as a "negative margin". However, it is very important to establish that a negative margin of a few decibels (0-3 dB) does not in any way mean harmful interference. Indeed, the concept of negative and positive margins, whilst convenient for making the plan, may be misleading unless put in the proper perspective.

2. <u>Relationship between protection margin, protection ratio and picture</u> quality

The Annex (produced by the European Broadcasting Union) shows the relationship between protection margin and picture quality. It can be seen from Figure 1 that going from a protection ratio of 31 dB to 28 dB (negative margin of -3) represents a change in the numerical value corresponding to the picture quality of less than 0.5. A value of protection ratio of 20 dB (negative margin of -11) would represent harmful interference since it corresponds to a slightly annoying impairment of the picture.

3. Further considerations

i) The 1 dB planning margin is not essential in all cases. For example, tube aging effects can be overcome by providing a slightly higher power-capability, together with simple power control to keep the emitted power within prescribed limits.

ii) The computer calculation has assumed a situation for printing error where the interfering satellites are all displaced by 0.1° towards the wanted-service area taking, at the same time, an adverse 0.1° error in the wanted-service beam. This will give an over-pessimistic result in some cases.

iii) The protection margin for countries having medium and large geographical areas varies considerably over the service area. Typically, a country having a negative margin of -2 dB at one point may have a positive margin of +6 at another point.

iv) A negative margin of -3 dB corresponds to a protection ratio of 28 dB. This is some 10 to 14 dB below the thermal noise level (assuming a minimum carrier-to-noise ratio of 14 dB).

v) The antenna reference pattern is only an envelope defining levels which are not to be exceeded. In practice, there will probably be nulls in the antenna patterns, providing much higher discrimination against interference from some directions.

4. The quality of the plan

The plan was produced by the Sub-Group 5A 2/3 under the Chairmanship of Mr. Amira of Kenya and comprised experts from 18 different countries from all parts of Regions 1 and 3. It must be put on record that from the viewpoint of achieving a successful compromise between picture quality, technical factors and the requirements presented by administrations to the Conference, the Sub-Group has produced a very good result. The Conference has for its consideration a good Region 1 and 3 plan.

Annex : 1

A N N E X

RELATIONSHIP BETWEEN PROTECTION MARGIN

AND PICTURE-QUALITY

The relationship is derived from equation 1 of CCIR Report 634(Rev.76) : This states that

$$PR_{0} = C - 20 \log \left(\frac{v}{12}\right) - Q + 1 \cdot 1 Q^{2}$$

where

PR : protection ratio, dB

1.30

С

D_v : nominal peak-to-peak frequency deviation

Q : the impairment grade, concerning the effect of interference only, measured on the 5-point scale recommended in draft Recommendation 500(Rev.76)

: a constant depending on the television standard

The appropriate values of PR_0 , C and D_v are given in Document No. 177 of Committee 4, and the resultant relationship is given in Figure 1 of the present document.





Relationship between protection margin/protection ratio and picture quality

Annex to Document No. 368-E Page 4

Document No. 369-E 12 February 1977 Original : English

(Geneva, 1977)

PLENARY MEETING

German Democratic Republic

DOCUMENTS NOS. 272 AND 273

The delegation of the German Democratic Republic confirms its statement made in Committee 2 on 7 February (see Annex to Document No. 271).

In connection with the issue of Documents Nos. 272 and 273, the delegation of the German Democratic Republic deems it necessary to state its disagreement of principle with their contents.

Paragraph a) of Document No. 272 constitutes, in fact, an attempt to interfere with the internal affairs of the German Democratic Republic, a sovereign State. As is well known, Berlin is the capital of the German Democratic Republic and it is only natural that the Quadripartite Agreement of 3 September 1971 does not apply to it.

For reasons of economy, this document is printed in a limited number. Participants are therefore kindly asked to bring their copies to the conference since only a few additional copies can be made available.

PINK PAGES Document No. 370-E 12 February 1977

PLENARY MEETING

6th SERIES OF TEXTS SUBMITTED BY THE EDITORIAL COMMITTEE TO THE PLENARY MEETING

The following texts are submitted to the Plenary Meeting :

Document No.

Title

DT/49, DT/52

Power flux density produced in the territories of Region 2 by satellites in the broadcasting-satellite service in Regions 1 and 3

Miss M. HUET

Chairman of the Editorial Committee

Annexes : 11 pages



POWER FLUX DENSITY PRODUCED IN THE TERRITORIES OF REGION 2 BY SATELLITES IN THE BROADCASTING-SATELLITE SERVICE IN REGIONS 1 AND 3

Method of calculation

1. The power flux density produced, under conditions of free space propagation, at a given point, P, on the surface of the earth, by a satellite in the geostationary orbit, can be calculated from the following data :

1.1 - nominal orbit position,

1.2 - e.i.r.p.

1.3 - characteristics of the antenna beam at half-power points (i.e. the major and minor axes together with the orientation of the associated ellipse),

1.4 - geographical coordinates of the boresight (B),

1.5 - geographical coordinates of the point P.

2. The values for items 1.1 to 1.4 above are obtained from the Plan for the broadcasting-satellite service in Regions 1 and 3. For the calculations which follow, the point P has been taken as the most easterly point of Brazil $(35^{\circ}W, 8^{\circ}S)$.

3.

The power flux density produced at P is then, calculated as follows :

- calculate the distance, d, between the satellite and the point P;

- calculate the spreading attenuation, A, for the distance d

$$(A = \frac{1}{4\pi d^2})$$

- calculate the angle φ , as seen from the satellite , between points $B_{_{\rm I}}$ and P;
- calculate ϕ_0 , the half-power beamwidth in the direction of P (in the case of a circular beam ϕ_0 will be independent of direction);
- using the reference pattern for the co-polar component of the satellite transmitting antenna, determine the relative antenna gain, g, for the calculated values of ϕ and ϕ_{α} ;
- calculate the power flux density, produced at P, from the relation :
 pfd = e.i.r.p. + g A

Results

The power flux densities produced at the most easterly point of Brazil ($35^{\circ}W$, $8^{\circ}S$) from broadcasting satellites of Regions 1 and 3, to which orbital positions from $37^{\circ}W$ to $5^{\circ}E$ and channels 1 to 25 have been assigned in the Plan, are given in the following table.

a light state of the state of the

Document No. 370-E Page 3

ANNEX

PFD Produced in Region 2 at a Point having Longitude = 35° W Latitude 8° S

Nominal orbit position - 37.0			No	minal o	orbit posi [.] • 31.0	tion	Nc	minal o	rbit posi [.] 25.0	tion	No	ominal c	rbit position 19.0			
Country symbol & IFRB No.		Channel Nos.	PFD Produced dBW/m ²	Country symbol & IFRB No.		Channel Nos.	PFD Produced dBW/m ²	Country symbol & IFRB No.		Channel Nos.	PFD Produced dBW/m ²	Country symbol & IFRB No.		Jhannel Nos.	PFD Produced dBW/m ²	
AND	0341	4, 8,12 16, 20	-146.35	AZR	0134	3,7, 11 15, 19	-140.72	ALG	0252	4, 8, 12 16, 20	-130.26	AUT	0016	4, 8, 12 16, 20	-143.67	
GMB	0302	3, 7, 11	-137.17	CPV	0301	4, 8,12, 16, 20	-137.14	TB A	0280	1, 5, 9 13, 17	-138.64	D	0087	2,6,10 14,18	-140.17	
LIE	0253	3, 7, 11 15, 19	-146.00	G	0027	4, 8, 12 16, 20	-140.02	LBY	0321	3, 7, 11 15, 19	-139.00	DAH	0233	3, 7, 11 15, 19	-140.20	
MLI	0327	2, 6, 10 14, 18	-132.79	GNP	0304	2, 6, 10 14, 18	-137.07	TGO	0226	2,6,10,	-141.45	F	0093	1, 5, 9 13, 17	-138.67	
SMR	0311	1, 5, 9, 13, 17	-145.92	IRL	0211	2, 6, 10 14, 18	-144.38			14, 18		LUX	0114	3, 7, 11 15, 19	-145.56	
GUI	0192	1, 5, 9, 13, 17	-132.98	LBŖ	0244	3, 7, 11 15	-137.10					ZAI	0322	4, 8,12 16, 20	-130.94	
MLI	0328	4, 8, 12, 16, 20	-131.06	POR	0133	3, 7, 11 15, 19	-142.35			L L E		ZAI	0323	2, 6,10 14, 18	-130.05	
1					3				0	3						
		1		ł		4	P.								R.6	

PINK PAGES

e,

12

Annex to Document No. 370-E Page 4

-

.

PINK PAGES

*

•3

.

Nominal orbit position - 13.0				Nomi	nálor –	bit positi 7.0	ion	N	ominal	orbit posi — 1.0	tion	No	minal c	orbit posit - 5.0	•ion
Country symbol & IFRB No.		Channel Nos.	PFD Produced dBW/m ²	Country symbol & IFRB No.		Channel Nos.	PFD Produced dBW/m ²	Country symbol & IFRB No.		Channel Nos.	PFD Produced dBW/m ²	Country symbol & IFRB No.		Channel Nos.	PFD Produced dBW/m ²
CVA	0085	23	-141.92	CNR	0130	23,27,31, 35,39	-140.93	GHA	0108	23,27,31, 35,39	-134.45	BEL	0018	21,25,29, 33,37	-144.97
MCO	0116	21,25,29, 33,37	-145.75	CTI	0237	22,26,30, 34,38	-132.20	NGR	0115	24,28,32, 36,40	-127.77	GNE	0303	23,27,31, 35,39	-141.30
MTN	0223	22,26,30, 34,38	-129,20	E	0129	23,27,31, 35,39	-137.48	MRC	0209	21,25,29, 33,37	-128.74	HOL	0213	23,27,31, 35,39	-144.77
MTN	0288	24,28,32, 36,40	-135.68	HVO	0107	21,25,29, 33,37	-131.90	TUN	0150	22,26,30, 34,38	-141.14	I	0082	24,28,32, 36,40	-138.57
SEN	0222	21,25,29, 33,37	-133.19	ISL	0049	21,25,29, 33,37	-142.72					NIG	0119	22,26,30, 34,38	-129.39
				SRL	0259	23,29,31, 35,39	-136.72	×				SUI	0140	22,26,30, 34,38	-143.10
				1 						*	-8-	NMB	0025	25,29,33, 37	-130.13
															R.6

•

٠

.

Annex to Document No.370-E Page 5

in 1

.

.

.......

.

 .

Nominal orbit position - 13.0			Nom	inal or	bit positi 7.0	ion	N	ominal o -	rbit posit	tion	Nomina	al orbit posi + 5.0	tion
Country symbol & IFRB No.	Channel Nos.	PFD Produced dBW/m ²	Cou sym å IFF	ntry ibol : B No.	Channel Nos.	PFD Produced dBW/m ²	Coun symb & IFRB	try ol 8 No.	Channel Nos.	PFD Produced dBW/m ²	Country symbol & IFRB No	Channel Nos.	PFD Produced dBW/m ²
CME 0300 GAB 0260 MLT 0147 STP 0241 TCD 0143	1, 5, 9, 13, 17 3, 7, 11 15, 19 4, 8, 12 16 4, 8, 12 16, 20 2, 6, 10 14, 18	-132.87 -136.65 -148.55 -144.70 -133.89	EGY	0026	4, 8, 12) 16, 2 0	-136.59	BOT BUL MOZ POL ROU	0297 0020 0307 0132 0136	2, 6, 10 14, 18 4, 8, 12 16, 20 4, 8, 12 16, 20 1, 5, 9 13, 17 2, 6, 10 14, 18	-134.49 -144.97 -135.37 -142.67 -143.17	DNK OOS FNL OIO GRC OIO S OI TUR OIA	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	-143.42 -138.17 -140.87 -138.94 -138.47
			-				SWZ TCH ZMB	0313 0144 0314	1, 5, 9 13, 17 3, 7, 11 15, 19 3, 7,11 15, 19	-147.30 -143.27 -134.29	NOR OI	20 14, 18	-139.42
	1	-	I		I I					1		1	R.6

Annex to Document No. 370-E Page 6

PINK PAGES

Nominal orbit position - 13.0				Nomi	inal or	bit positi 7.0	on	N	ominal o	orbit posit - 1.0	tion	Non	ninal o +	rbit posit 5.0	ion			
Con Byr IF	ntry nbol & RB No.	Channel Nos.	PFD Produced dEd/m ²	Cou sym & IFR	ntry bol B No.	Channel Nos.	PFD Produced dEW/m ²	Coun symb & IFRI	ntry pol 3 No.	Channel Nos.	PFD Produced dBW/m ^{2.}	Course Syml & IFR	ntry bol B No.	Channel Nos.	* PFD Produced dBW/m ²			
AGL	0295	23,27,31 35,39	, -129.57	ALB	0296	22,26,30, 3 ⁴ ,38	-146.49	DDR	0216	21,25,29, 33,37	-145.17	AFS	0021	21,25,29, 33,37	-132.06			
CAF	0258	24,28,32 34,40	-130.81	SDN	0231	22,26,30, 34,38	-133.37	HNG	0106	22,26,30, 34, 3 8	-145.07	CYP	0086	21,25,29, 33,37	-147.47			
COG	0235	22,26,30 34,38	, -134.83	SDN	0230	23,27,31, 35,39	-136.84	MWI	0308	24,28,32, 36,40	-142.67	DNK	0090	24	-135.20			
ISR	0110	25,29,33 37	, -145.02	SDN	0232	24,28,32, 36,40	-134.23	RHS	0135	22,26,30, 34,38	-136.51	FNL	0104	22,26	-135.20			
		3		YUG	0148	21,25,29, 33,37	-140.79				-	ISL	0050	23,27,31, 35,39	-137.87			
				YUG	0149	23,27,31, 35,39	-140.79					LSO	0305	24,28,32, 36,40	-145.06			

R.6

10

ANNEX 1

LIMITS FOR DETERMINING WHETHER A SERVICE OF AN ADMINISTRATION IS CONSIDERED TO BE AFFECTED BY A PROPOSED MODIFICATION TO THE PLAN (ARTICLE 4, PARAGRAPH 4.3.1)¹⁾

1.

Limits on the change in the wanted-to-interfering signal ratio with respect to frequency assignments in accordance with the Plan

With respect to paragraph 4.3.1.1, an administration shall be considered as being affected if the effect of the proposed modification to the Plan would result in the wanted-to-interfering signal ratio at any point within the service area associated with any of its frequency assignments in the Plan falling below either 30 dB or the value resulting from the frequency assignments in the Plan at the date of entry into force of the Final Acts, whichever is the lower.

<u>Note</u> : In performing the calculation, the effect at the receiver input of all the co-channel and adjacent channel signals is expressed in terms of one equivalent co-channel interfering signal. This value is usually expressed in decibels.

2. Limits on change in power flux density to protect the broadcastingsatellite service in the band 11.7 - 12.2 GHz in Region 2

With respect to paragraph 4.3.1.2 an administration in Region 2 shall be considered as being affected if the proposed modification to the Plan would result in exceeding the following power flux densities at any point in the service area affected :

-147 dBw/m ² /27 MHz	0° ≤ θ < 0.48°
-139 + 25 log θ dBW/m ² /27 MHz	0.48° ≼ θ < 2 7. 25°
-103 dBW/m ² /27 MHz	θ ≥ 27.25°

where θ is the difference in degrees between the longitudes of the broadcastingsatellite space station in Region 1 or 3 and the broadcasting-satellite space station affected in Region 2.

¹⁾ The limits specified in this Annex relate to the power flux densities which would be obtained assuming free space propagation conditions.

3. Limits on the change in power flux density to protect the terrestrial services of other administrations

With respect to paragraph 4.3.1.3, an administration in Region 1 or 3 shall be considered as being affected if the consequence of the proposed modification to the Plan is to increase the power flux density arriving on any part of the territory of that administration by more than 0.25 dB over that resulting from the frequency assignments in the Plan at the time of entry into force of the Final Acts.

The same administration shall be considered as not being affected if the value of the power flux density anywhere in its territory does not exceed the limits expressed in Annex 5.

An administration in Region 2 shall be considered as being affected if the proposed modification to the Plan would result in exceeding a power flux density, for all angles of arrival, at any point on its territories, of $-125 \text{ dBw/m}^2/4 \text{ kHz}$ when the broadcasting-satellite station uses circular polarization and $-128 \text{ dBw/m}^2/4 \text{ kHz}$ when the broadcasting-satellite station uses linear polarization.

4. Limits on the change in power flux density to protect fixedsatellite service in the band 11.7 - 12.2 GHz in Region 2

With respect to paragraph 4.3.1.4, an administration in Region 2 shall be considered as being affected if the proposed modification to the Plan would result in an increase in the power flux density on its territory of 0.25 dB or more above that resulting from the frequency assignments in the Plan at the time of entry into force of the Final Acts.

Where an assignment in the Plan or its subsequent modification gives a power flux density less than -138 anywhere in the territory of an administration of Region 2, that administration shall be considered as not affected.

ANNEX 10

ORBITAL POSITION LIMITATIONS

In applying the procedure of Article 4 for modifications to the Plan, administrations shall observe the following criteria :

- 1) No broadcasting-satellite serving an area in Region 1 and using a frequency in the range 11.7 12.2 GHz shall occupy a nominal orbital position further West than 37° W or further East than 146° E.
- 2) Any new nominal orbital position in the Plan in the range of orbital arc between 37° W and 10° E associated with a new assignment, or resulting from a modification of an assignment in the Plan, shall be coincident with, or within 1° to the East of, a nominal orbital position in the Plan at the date of entry into force of the Final Acts.

In the event of a modification to an assignment in the Plan, the use of a new nominal orbital position not coincident with any nominal orbital position in the Plan at the date of entry into force of the Final Acts shall be associated with an 8 dB reduction in the e.i.r.p. compared to that appearing in the Plan for the assignment before modification.

R.6

ARTICLE 10

POWER FLUX DENSITY LIMITS BETWEEN 11.7 AND 12.2 GHz TO PROTECT SPACE SERVICES IN REGION 2 FROM INTERFERENCE FROM BROADCASTING-SATELLITE SPACE STATION'S OF REGIONS 1 AND 3

Broadcasting-satellite space stations of Regions 1 and 3 shall employ transmitting antennae whose side lobe characteristics fall within the reference antenna pattern given in Figure 5 of Annex 8. Therefore, the power flux density falling on the territory of any administration of Region 2 in the band 11.7 - 12.2 GHz prior to any modifications to the Plan shall not exceed, under all conditions and methods of modulation, the values produced by broadcastingsatellite stations operating in accordance with the Plan on the date of its entry into force and using the technical characteristics specified in the Plan. The power flux density values shall be calculated using the method described in Annex 11.

In particular, the power flux densities at a reference test point (longitude 35°W, latitude 8°S) prior to any modifications to the Plan shall not exceed the values shown in Annex 12.

ANNEX 11

METHOD OF CALCULATING THE POWER FLUX DENSITY PRODUCED IN THE TERRITORIES OF

REGION 2 BY SPACE STATIONS IN THE BROADCASTING-

SATELLITE SERVICE IN REGIONS 1 AND 3

1. The power flux density produced, under conditions of free space propagation, at a given point, P, on the surface of the earth, by a satellite in the geostationary orbit, can be calculated from the following data :

- 1.1 nominal orbit position,
- 1.2 e.i.r.p.,
- 1.3 characteristics of the antenna beam at half-power points (i.e. the major and minor axes together with the orientation of the corresponding ellipse),
- 1.4 geographical coordinates of the boresight (B),
- 1.5 geographical coordinates of the point P.

2. The values relevant to items 1.1 to 1.4 are indicated in the Plan. The point P can be chosen with reference to the objective of calculation. In the present context P could be, for instance, the most easterly point of Brazil.

3.

- To obtain the power flux density produced at P, calculate :
 - the distance, d, between the satellite and the point P;

- the spreading attenuation, A, for the distance d

$$(A = \frac{1}{4\pi d^2});$$

- the angle ϕ , as seen from the satellite, between points B and P;
- ϕ_0 , the half-power beamwidth, in the direction of P (in the case of a circular beam ϕ_0 will be independent of direction);

R.6

- the relative antenna gain, g, for the calculated values of ϕ and ϕ_0 using the reference pattern for the co-polar component of the satellite transmitting antenna.

•

5

Then apply the expression :

pfd = e.i.r.p. + g - A

to obtain the power flux density produced at P.

Document No. 371-E 12 February 1977 Original : English/French

GENÈ

(Geneva, 1977)

PLENARY MEETING

United States of America, France,

United Kingdom of Great Britain and Northern Ireland

DOCUMENT No. 365

The above-mentioned three delegations hereby state that :

- a) they cannot accept the assertions made in Document No. 365 by the delegation of the USSR; and
- b) they reaffirm the statements made in Document No. 272.

Document No. 372-E 12 February 1977 Original : English

(Geneva, 1977)

PLENARY MEETING

Federal Republic of Germany

DOCUMENTS Nos. 365 AND 369

The delegation of the Federal Republic of Germany fully associates itself with the view stated in Document No. 371 by the delegations of France, the United States of America and the United Kingdom of Great Britain and Northern Ireland.



SEANCE PLENIERE PLENARY MEETING SESION PLENARIA

<u>7ème SERIE DE TEXTES SOUMISE PAR LA COMMISSION DE</u> REDACTION A LA SEANCE PLENIERE

Les textes ci-après sont soumis à la séance plénière <u>en deuxième</u> <u>lecture</u> :

7th SERIES OF TEXTS SUBMITTED BY THE EDITORIAL COMMITTEE TO THE PLENARY MEETING

The following texts are submitted to the Plenary Meeting for second reading :

7.^a SERIE DE TEXTOS SOMETIDOS POR LA COMISIÓN DE REDACCIÓN AL PLENO DE LA CONFERENCIA

Los textos seguidamente relacionados se someten al Pleno de la Conferencia en segunda lectura :

> M. HUET Présidente de la Commission de Rédaction

Annexe : 2 pages

R.7



Document N^O 373-F/E/S Page 2

NOTES - NOTAS

1. Le ΔG de cette assignation est ... dB.

1. The ΔG of this assignment is ... dB.

1. El Δ G de esta asignación es ... dB.

- A affecter au programme islamique prévu dans les documents de la Conférence
 To be allocated to the Islamic programme envisaged in the Conference documents.
 Se destinará al programa islámico previsto en los documentos de la Conferencia
- 3. Cette assignation résulte d'un besoin commun des Administrations du Danemark et de l'Islande. La zone de service comprend les îles Fércé et l'Islande. L'assignation peut, à l'issue de consultations entre les deux administrations, être utilisée par l'une ou l'autre d'entre elles.
- 3. This assignment results from a common requirement of the Administrations of Denmark and Iceland. The service area includes the Faeroe Islands and Iceland. The assignment may after consultations between the two Administrations be used by either of them.
- 3. Esta asignación proviene de una golicitud formulada en común por las Administraciones de Dinamarca e Islandia. La zona de servicio incluye las Islas Feroe e Islandia. Previa consulta entre las dos Administraciones, la asignación podrá ser utilizada por cualquiera de ellas.

4. IFB - IFRB. Cette assignation a été incluse dans le Plan par la Conférence.
4. IFB - IFRB. This assignment has been included in the Plan by the Conference.
4. IFB - IFRB. La Conferencia ha incluido esta asignación en el Plan.

R.7

Document Nº 373-F/E/S Page 3

- 5. Assignation destinée à assurer la couverture de l'Algérie, de la Libye, du Maroc, de la Mauritanie et de la Tunisie, suite à un accord entre ces pays. En cas de besoin, elle peut être utilisée avec les caractéristiques du faisceau TUN150.
- 5. Assignment intended to ensure coverage of Algeria, Libya, Morocco, Mauritania and Tunisia, with the agreement of the countries concerned. If required, this assignment may be used with the characteristics of the beam TUN 150.
- 5. Asignación destinada a asegurar la cobertura de Argelia, Libia, Marruecos, Mauritania y Túnez como consecuencia de un acuerdo entre estos países. De ser necesario, la presente asignación puede utilizarse con las características del haz TUN 150.
- 6. Les assignations inscrites dans le Plan au nom de la Somalie doivent être coordonnées avec chacun des pays intéressés et en particulier avec l'Ethiopie.
- 6. Assignments appearing in the Plan for Somalia should be coordinated with each country concerned and in particular with Ethiopia.
- Esta asignación deberá coordinarse con cada uno de los países interesados y en particular con Etiopía.
BROADCASTING SATELLITE CONFERENCE

(Geneva, 1977)

Document No. 374-E 17 February 1977 Original : English

COMMITTEE 4

SUMMARY RECORD

OF THE

TWELFTH MEETING OF COMMITTEE 4

(TECHNICAL)

Sunday, 6 February 1977 at 0940 hrs

Chairman : F. KRÁLIK (Czechoslovakia)

Subjects discussed

Approval of documents

Document No.

188(Rev.2)

ACHIVE

U.I.T Genè

Approval of documents (Document No. 188(Rev.2) (continued))

The <u>Chairman of the Conference</u>, speaking as Chairman of the <u>ad hoc</u> group convened the previous day to consider paragraph 2.3 of Document No. 188(Rev.2), said that a group of countries in Region 1 had proposed to improve the antenna reference pattern on page 2 of Document No. 108 by 5 dB for all countries in Regions 1 and 3, on condition that paragraph 2.3 would be deleted. That proposal had been supported by other Region 1 and 3 countries in the <u>ad hoc</u> group, but the Region 2 countries, while accepting the first part of the proposal, could not accept the conditional part, and the United States delegation had proposed that the power flux-density figure of - 134 dBW/m² in paragraph 2.3 should be changed to - 132 dBW/m², provided that value was entered in the Plan for only a few countries. That proposal had been supported by other Region 2 countries, but had not been accepted by the other delegations concerned.

In view of that deadlock, he wished to submit a third possibility, namely, that Committee 4 should immediately bring to the notice of Committee 5 for use in its planning work the proposal to improve the antenna reference pattern by 5 dB; that would both facilitate planning and ease the situation with regard to Fixed-Satellite Service in Region 2. At the same time Committee 4 might decide to leave out the figures in square brackets in the first subparagraph and place square brackets round the whole of paragraph 2.3; when the draft Plan was made known, on Tuesday, 8 February, the necessary data - concerning the most westerly positions - would be communicated immediately to Committee 4 Document No. 374-E Page 2

or to Working Group 5C, the IFRB would carry out the necessary calculations and the power flux-density value laid down for Region 2 could be evaluated for all the parties concerned. It was to be hoped that agreement could then be reached on paragraph 2.3 - either to insert figures correlating with Plan calculations, to insert figures to which there might be a few specified exceptions, or to delete the paragraph altogether.

The delegates of Italy and the USSR supported that proposal.

The <u>delegate of Senegal</u> said that, although the proposal had some useful aspects, it should be borne in mind that the proposal to improve the antenna pattern by 5 dB was conditional on the results of the Plan calculations of power flux-density in Region 2. Yet the countries of that Region were in fact awaiting the Plan results before deciding whether they could accept the 5 dB concession; if they decided against acceptance, the Plan, though satisfactory to many countries, would be unsatisfactory to a few, namely the countries of Western Africa directly confronting Region 2, which would be faced with a fait accompli; interference in Region 2 should be regarded from the point of view of its effect on other countries irrespective of whether broadcastingsatellite or fixed-satellite systems were concerned.

The <u>delegate of Mauritania</u> endorsed those remarks. The countries of Region 2 should take into account the concession proposed by the countries of Western Africa and altruistically supported by other Region 1 countries and those of Region 3. The Chairman's proposal would certainly enable the Conference to proceed with its work, but the result of that work must not entail different standards for any countries : ultimately, the values now in square brackets must only reflect the calculations that would emerge from the Plan and could not be based on considerations involving the Fixed-Satellite Service.

The <u>delegate of the United States of America</u> said that his delegation had no objection to the adoption of a new reference antenna pattern by Regions 1 and 3 or to making that pattern applicable to all countries. Indeed, the pattern would be taken as a basis for the Region 2 Plan at the 1982 Planning Conference; it would certainly improve the present Plan, by leading to positive protection margins in most cases. He could support the proposal of the Chairman of the Conference to place the whole of paragraph 2.3 in square brackets until information on expected interference to Region 2 became available. Moreover, he agreed with the Mauritanian delegate that the values in paragraph 2.3 should apply to all countries; that was the spirit in which the Region 2 countries had proposed certain figures - incidentally, similar to those required by Regions 1 and 3 to protect their Terrestrial Services. Where concessions were concerned, it would be recalled that the Region 2 countries had made roughly equivalent ones at a recent meeting of Working Group 4B.

The <u>delegate of Mauritania</u> proposed that the new reference antenna pattern should be assumed to be applicable to all countries and that the IFRB should be asked to calculate on that basis the minimum p.f.d. for the national service in every territory covering a country and the effect that it would have on Region 2. Those data could be listed in columns in paragraph 2.3, which could then be approved, if the Region 2 countries accepted those calculations. The <u>delegates of Senegal and the United Kingdom</u> supported that proposal.

The <u>delegates of Canada</u>, the United States of America and <u>New Zealand</u> said that, although the Mauritanian proposal had some merits, paragraph 2.3 could not be approved until the relevant values were definitively inserted in it. The procedure proposed by the Chairman of the Conference should be followed for the time being.

The <u>delegate of Mauritania</u> said that the Region 2 countries seemed to be waiting for the results of calculations before deciding whether to accept or reject the Plan. That course, however, was also open to the countries of Regions 1 and 3.

The <u>delegate of the United States of America</u> pointed out that it was not for the Region 2 countries to approve the Plan for Regions 1 and 3. All they were concerned with was the establishment of interference levels for their Region.

The <u>delegate of Senegal</u> thought it was not difficult technically to position fixed satellites further over the territory of Region 2 so that the receiving antenna were almost vertical, which would reduce interference without changing the e.i.r.p. He could agree to leaving the whole of paragraph 2.3 in square brackets provided the Plan was established on the basis of the antenna reference diagrams improved by 5 dB.

The <u>delegate of Algeria</u> feared that a Plan would be drawn up and be judged acceptable or not by Region 2 depending on a certain protection value; he proposed that that value be calculated by a method adopted by the Conference as a whole as soon as possible and not as the result of a unilateral decision.

That view was supported by the <u>delegates of the Ivory Coast</u> and <u>of</u> <u>the United Kingdom</u>, the latter agreeing that the contents of the square brackets should be worded along the lines proposed by Mauritania and Senegal.

The <u>delegate of New Zealand</u> thought it preferable to await the information which would become available on Tuesday 8 February before changing the paragraph.

In view of the support expressed for the Mauritanian proposal, the <u>Chairman</u> asked the delegate of the United Kingdom to chair a drafting group to produce a new version and it was <u>agreed</u> that the delegations of Mauritania, Senegal, the United States of America, Brazil and India would also take part.

In answer to the <u>Chairman of Committee 5</u>, the <u>Chairman</u> confirmed that the antenna diagrams should be improved. The <u>delegate of Switzerland</u> felt that an improvement of 5 dB would not be possible for the left-hand part of the curves. The <u>Chairman</u> said that a diagram had already been prepared and would be presented to the meeting at a later stage.

Following a short break to allow the drafting group to meet, the <u>delegate of the United Kingdom</u> read out the following new paragraph, to be retained in square brackets for the time being :

"2.3 To protect space services in Region 2

The Broadcasting Satellite Service of Regions 1 and 3 shall employ satellite antennae whose side lobe characteristics fall within the <u>/</u> new <u>/</u> reference antenna pattern given in Figure <u>/</u> _/ and the power flux-density falling on the territory of any administration of Region 2 in the band 11.7-12.2 GHz shall not exceed, under all conditions and methods of modulation, the following :

Country	Assignment in the Plan	Max. P.F.D. over Region 2
	<pre>/ Note l - assignment refers to orbit position, channel, beam major axis, azimuth, sub-satellite point etc. /</pre>	/ Note 2 - the values of max. P.F.D. over Region 2 would be those values resulting from the Plan_/"

The delegate of Algeria said he could accept that text.

The <u>delegate of the USSR</u> said that the heading should be altered to read "To protect Fixed-Satellite Services in Region 2". Nothing had been said about the Broadcasting Service because the Fixed-Satellite Service, with which the paragraph was concerned, was much more sensitive to interference.

The <u>Chairman of Working Group 4B</u> said he understood the paragraph to cover protection of both space services in Region 2. The title had not given rise to any discussions in the working group and he saw no justification for changing it.

The <u>delegates of the United Kingdom</u>, <u>Spain</u> and <u>Brazil</u> agreed with that view and the <u>delegate of the USSR</u> withdrew his suggestion.

At the proposal of the <u>delegate of Denmark</u>, it was agreed to insert the words "(except Greenland)" after "Region 2" in the fourth line.

Paragraph 2.3, as amended, was <u>approved</u>, subject to retention of the square brackets.

At the <u>Chairman</u>'s invitation, the <u>Chairman of Working Group 4B</u> introduced the remainder of Document No. 188(Rev.2).

He proposed further equations to make the curve in Figure 1 more explicit and indicated changes to the text of paragraph 2.2, which had not been approved in the Working Group due to lack of agreement on the content of paragraph 2.3 and difficulties in indicating protection levels for Terrestrial Services in Region 2. The square brackets around the figure of - 111 dBW should be deleted and the text of the right-hand column read :

> "For territories of administrations in the western part of Region 1 (west of longitude 30^oE) and for the Broadcasting and the Mobile (except aeronautical) Services and systems in the Fixed Service (excluding FDM/FM systems in Region 3).

For territories of administrations in Region 1 and for the FDM/FM systems in the Fixed Service in Region 3."

The <u>Chairman of the Working Group</u> said that the values in paragraphs 2.1 and 2.2 had been provided by the Administrations of Regions 1 and 3.

In reply to the <u>delegate of the United States of America</u>, the <u>delegates of the Federal Republic of Germany</u> and the <u>Netherlands</u> confirmed that they were prepared to modify the figure of - 134 dBW/m²/5 MHz to - 132 dBW.

The <u>Chairman of the Working Group</u> said that paragraph 3 was intended for the guidance of Committee 6, the figures having been provided by the Chairman of Committee 5 and the delegation of the USSR. The intention of the group in drafting the paragraph had been to reduce restrictions on modifications to the Plan to a minimum.

The <u>delegate of the USSR</u> thought it had been decided that paragraph 3 would be considered by Working Group 5C and might be subject to alteration as a result of planning activities. The <u>Chairman of Committee 5</u> said that the paragraph could not be included in the final acts in its present form and in his view it was a matter for Committee 6.

The <u>Chairman of Working Group 4B</u> did not recall any decision on the disposition of the paragraph but could not agree that its contents would be affected by planning.

It was <u>agreed</u> to continue discussion on the document at the following meeting of the Committee.

The meeting rose at 1300 hours.

The Co-Secretaries :

The Chairman :

F. KRALIK

J. RUTKOWSKI/M. AHMAD

BROADCASTING SATELLITE CONFERENCE

(Geneva, 1977)

Document No. 375-E 17 February 1977 Original : English

COMMITTEE 4

SUMMARY RECORD

OF THE

THIRTEENTH AND LAST MEETING OF COMMITTEE 4

(TECHNICAL)

Monday, 7 February 1977, at 0925 hrs

Chairman : Mr. F. KRÁLÍK (Czechoslovakia)

Subjects discussed

1. Approval of documents

2. Completion of the Committee's work

1. Approval of documents (Document No. 188(Rev.2)) (continued)

The <u>Chairman of Working Group 4B</u> read out the following text of a new paragraph 2.4 to be inserted in Document No. 188(Rev.2) :

"2.4 Protection of Terrestrial Services in Region 2 from the Broadcasting-Satellite Service in Regions 1 and 3

2.4.1 A power flux-density limit should be adopted to protect Terrestrial Services in Region 2 from the Broadcasting-Satellite Service in Regions 1 and 3. This limit should be :

2.4.2 To protect FDM/FM systems in the Fixed Service :

- 125 dBW/m² in any kHz band, for circular polarization. for all angles of arrival;

2.4.3 This limit may be exceeded with the agreement of the Administrations concerned, that is, in those territories where the p.f.d. would be exceeded".

For reasons of economy, this document is printed in a limited number. Participants are therefore kindly asked to bring their copie to the conference since only a few additional copies can be made available.

 \mathcal{D}

Document No.

188(Rev.2)

Document No. 375-E Page 2

On second thoughts, he would suggest that sub-paragraph 2.4.3 might be omitted, since the relatively high figure of - 125 dBW/m² could be regarded as the upper limit and since the same figure approved for the protection of Terrestrial Services in Regions 1 and 3 in paragraph 2.2 represented an absolute limit rather than a trigger level for coordination.

The <u>delegate of Iceland</u>, supported by the <u>delegates of the United</u> <u>Kingdom</u>, <u>Japan</u> and <u>Finland</u>, said that, in view of his country's proximity to Region 2, his delegation would prefer that provision to remain and to be worded identically with the second sentence of paragraph 2.2.

The <u>delegate of India</u> observed that spacecraft in Regions 1 and 3 would be penalized in relation to those of Region 2 by the proposed 2 dB difference in the p.f.d. limits for protection of space services in Region 2 and in Regions 1 and 3.

The <u>Chairman of Working Group 4B</u> said that the proposed level was high enough to afford protection without any significant penalty for any Region 1 or 3 country with respect to the sidelobe performance of a spacecraft antenna.

The <u>delegate of India</u> said that the goal should be to achieve uniform sidelobe suppression.

The <u>delegate of the USSR</u>, supported by the <u>delegate of India</u>, said that, since the Conference documents showed a strong probability of the use of linear polarization in Region 2, the words "- 128 dBW/m²/4 kHz for linear polarization" should be added after "- 125 dBW/m²/4 kHz for circular polarization" in paragraph 2.2, to cover all eventualities.

The <u>delegate of the United States of America</u> agreed with that proposal and suggested that the same addition be made in paragraph 2.4.

Paragraph 2.4, as amended, was approved.

After a discussion on the action to be taken with regard to paragraph 3, during which the Chairman of Working Group 4B and the delegates of Canada, the Federal Republic of Germany, Brazil and Mexico expressed the view that the provisions of the paragraph should take the form of a recommendation by Committee 4 to the Plenary Meeting of specific technical criteria for interregional sharing, to be observed by Committee 5 and Joint Working Group 4/6, while the <u>delegates of India</u> and <u>the USSR</u> considered that such a recommendation would unnecessarily restrict the freedom of action of Committee 5 and that the best course would be to refer paragraph 3 to that Committee and to Joint Working Group 4/6, the Chairman of Committee 5 proposed that the contents of paragraph 3 should be incorporated in the cover page of Committee 4's report to the Plenary Meeting, that after approval by the Plenary the text should be forwarded to Committee 5 for planning purposes and to Joint Working Group 4/6 for incorporation in the Appendix on modification procedures and that the remainder of Document No. 188(Rev.2) should be annexed to the report and forwarded to the Editorial Committee for publication in a B Series document.

Document No. 375-E Page 3

That proposal, supported by the <u>delegates of Canada and Brazil</u>, was <u>approved</u>.

Document No. 188(Rev.2), as amended, was approved.

2. <u>Completion of the Committee's work</u>

The <u>Chairman</u> thanked all those concerned for their valuable contributions and announced that Committee 4 had completed its work.

The meeting rose at 1040 hours.

The Co-Secretaries :

The Chairman

J. RUTKOWSKI/M. AHMAD

F. KRÁLÍK

BROADCASTING SATELLITE CONFERENCE

(Geneva, 1977)

Document No. 376-E 17 February 1977 Original : English

COMMITTEE 5

SUMMARY RECORD

OF THE

SIXTH MEETING OF COMMITTEE 5

(Planning)

Monday, 7 February 1977, at 1410 and at 1945 hrs

Chairman : Mr. A. PETTI (Italy)

Subjects discussed

1. Characteristics given in the Plan

2. Report by Working Group 5B

3. Technical data used for drawing up the Plan/s/ and to be used in implementing the Plan/s/ (paragraphs 3.7 and 3.8.1)

1. Characteristics given in the Plan (Document No. DT/47)

The <u>Chairman</u> invited the Committee to consider the column headings he had prepared in conjunction with the Secretariat which were needed for preparing copies of the Plan in time for the signature of the Final Acts. They were set out in Document No. DT/47 and might be taken seriatim.

It was so <u>agreed</u>.

Column 1 - Country symbol and IFRB serial number

The <u>Technical Secretary</u> said that it might be useful for administrations not represented at the Conference if the Secretariat were authorized to insert a cross-reference to the source of the symbols used in the preface.

Subject to such an insertion, the heading for column 1 was approved.

For reasons of economy, this document is printed in a limited number. Participants are therefore kindly asked to bring their copies to the conference since only a few additional copies can be made available.

Documents Nos.

DT/47

204(Rev.1) and Corr.1, 226

DT/48

ACHIVE

U.I.T.

Document No. 376-E Page 2

Column 2 - Nominal longitude in degrees

At the suggestion of the <u>United Kingdom delegate</u>, it was <u>agreed</u> to modify the heading of column 2 to read : "Nominal orbital position in degrees" so that it should tally with the definition.

Column 3 - Channel number

The <u>Chairman</u> observed that under column 3 data would also have to be given concerning the frequency.

The heading of column 3 was approved.

Column 4 - Boresight geographical coordinates (in degrees and tenths)

The <u>delegate of Chile</u> considered that in the interests of uniformity both columns 2 and 4 should be expressed in degrees and tenths.

The <u>delegate of the United States of America</u> agreed that degrees and tenths would be more in line with computer usage.

The heading of column 4 was approved as it stood.

<u>Column 5</u> - Antenna aperture. For an elliptical antenna this column contains two figures corresponding to the major axis and the minor axis respectively

The <u>delegate of Denmark</u> said that the words "these figures contain pointing errors" should be added in brackets at the end of the heading. That addition was necessary because as indicated in Document No. 103(Rev.1) in many cases, if not all, the figures included such errors.

The <u>delegate of France</u> feared that such an amendment would complicate matters because the column would then no longer indicate the antenna aperture.

The <u>representative of the IFRB</u> suggested that the point be met by the insertion of the words "half-power" before the words "antenna aperture" and the insertion of the words "in degrees and tenths" at the end of the first sentence in the heading of Column 5.

Those amendments were approved.

The heading of column 5, as amended, was approved.

<u>Column 6</u> - Orientation of the ellipse determined as follows : in a plane normal to the beam axis, the direction of a major axis of the ellipse is specified as the angle measured anti-clockwise from a line parallel to the equatorial plane to the major axis of the ellipse, to the nearest degree

Approved.

<u>Column 7</u> - ΔG (Difference between the e.i.r.p. directed towards the edge of the coverage area and the e.i.r.p. in the beam axis)

The <u>delegate of the USSR</u> considered that column 7 was superfluous because the exceptions to values of less than 3 dBW would be so few.

The <u>Chairman</u> explained that a value of 3 dBW had been fixed except for small countries where it would be smaller. Perhaps column 7 might be deleted and an explanatory footnote on that point added under column 11 "Remarks".

It was so agreed.

<u>Column 8</u> - Polarization (1 = direct, 2 = indirect)

The <u>delegate of the Federal Republic of Germany</u>, referring to point 3.2 concerning polarization (Document No. 224, page 6) said that a definition was lacking; the proposed heading for column 8 did not make clear whether direct or circular polarization was to be used.

The <u>Chairman</u> suggested the insertion of a footnote to column 8 containing a definition of direct or indirect polarization on the lines of that contained in Document No. 321 of the CCIR.

It was so agreed.

Subject to that addition, the heading of column 8 was approved.

Column 9 - E.i.r.p. in dBW

The delegate of France said that the heading was too imprecise.

The <u>representative</u> of the IFRB suggested that the heading be amended to bring it into line with the wording already used in column headings of the IFRB Frequency Register. It might read "E.i.r.p. in the direction of maximum radiation in dBW".

It was so agreed.

The heading of Column 9, as amended, was approved.

Column 10 - Equivalent protection margin

The <u>delegate of the Federal Republic of Germany</u> observed that columns 1 to 9 dealt with the basic characteristics of certain beams whereas column 10 was more concerned with qualitative data. For example a country might wish to change its type of beam which would have repercussions on others. Perhaps column 10 should be dropped and some indication given that data on equivalent protection margins would be published from time to time either by IFRB or in some other form. If column 10 were retained some difficulties of definition would undoubtedly arise. The <u>delegate of Japan</u> believed that data on equivalent protection margins was essential for administrations and should be published in a separate document and not included in the Final Acts.

The <u>delegate of the USSR</u> agreed that such data was extremely useful for the assessment of channel qualities.

The <u>delegate of Italy</u>, while agreeing that a definition of values would have to be attached to column 10, considered it to be indispensable for the application of procedures for modifying the Plan.

The <u>Technical Secretary</u> pointed out that it would also be necessary to mention the test point otherwise the values given under column 10 would be meaningless.

The <u>delegate of France</u> considered that it would be difficult to include a protection margin because if a modification were made in the Plan and subsequently accepted some margins would then be altered and would become obsolete.

The <u>delegate of Italy</u> pointed out that the IFRB would have to keep the Plan up to date with any modifications and consequential adjustments in protection margins.

The delegate of France said that would meet his objection.

The delegate of the Federal Republic of Germany pointed out that in Document No. DT/34 the three main interferers were not mentioned in column 11 or identified.

The <u>representative of the IFRB</u> pointed out that the term "protection margin" had not yet been defined. Test points might differ from those in the Plan.

The <u>Chairman</u> said that six test points were to be examined in the Annex. Perhaps the problem of protection margins in service areas could be dealt with in a draft resolution.

The <u>delegate of France</u> expressed doubts about such a solution : six test points were too few because a deterioration in a small service area could affect a large part of a territory in a large service area but it would be too complicated to choose numerous test points.

The <u>Chairman</u> suggested that a draft resolution might be prepared for examination at a later stage requesting the IFRB to publish a separate document on test points and protection margins which would not form part of the Final Acts. Column 10 might then be eliminated.

It was so agreed.

Column 11 - Remarks

The <u>delegate of the United Kingdom</u> pointed out that it was important to include a column indicating the minimum angle of elevation for protection

Document No. 376-E Page 5

of the Broadcasting-Satellite Service. Delegations could indicate before the conclusion of the Conference what value they regarded as useful and there should be no difficulty in making the calculations for small service areas not far from the equator or for large service areas in more northerly or more southerly latitudes.

The <u>delegate of France</u> said that the angle of elevation was difficult to define because it depended on the point from which it was calculated and he questioned whether such data needed to be included either in the Plan or its annexes.

The delegate of Chile agreed with the United Kingdom delegate.

The <u>delegate of the USSR</u> observed that the values in question could vary widely according to the point from which measurements were taken. Surely they could be calculated by reference to the formula set out in point 2.4.1 of Document No. 232. He was unable to understand the reason for the United Kingdom proposal since it was hoped that by the end of the week there would be a plan indicating the nominal orbital position for each satellite within a country or area within its territory. When the problem of the protection of Broadcasting-Satellite Services from Terrestrial Services arose then the angle of elevation would have to be determined as from different points and that would become a matter of mere trigonometrical calculation and geographical coordinates in accordance with the actual data set out in the Plan and the formulae established.

The <u>delegate of Belgium</u> considered that the best course would be for the IFRB to prepare a formula for the angle of elevation on the basis of which each administration could make its own calculation.

The <u>delegate of Mauritania</u> believed that an average value should be chosen which would meet the requirements of both small and large countries covered by large beams.

The <u>delegate of France</u> endorsed the judicious proposal advocated by the Belgian delegate and shared the doubts expressed by the USSR delegate concerning the United Kingdom proposal.

The <u>delegate of the United Kingdom</u> explained that he was only concerned with the need for an administration to specify a coordination angle in order for another administration to protect its Terrestrial Services. If any administration found it difficult to specify a single value it could choose more.

It was up to administrations to make their own calculations but his own would like to see included in the Plan an elevation angle for its Broadcasting-Satellite Service so that its neighbours had a value which the United Kingdom Administration knew was a good one in order that they could determine whether or not to coordinate their Terrestrial Services with it. It was not for Committee 5 to go into the problem of coordination which was being dealt with in another committee. Administrations ought to be in a position to provide the data by 9 February. Document No. 376-E Page 6

The <u>representative of the IFRB</u> pointed out that the relevant data could not be assembled in the time available.

After some discussion, the <u>delegate of the United Kingdom</u> proposed that the IFRB be requested to prepare a list of all the elevation angles for different countries after the conclusion of the Conference.

It was so decided.

The <u>delegate of the USSR</u>, supported by the <u>delegate of India</u>, proposed that the Committee should request the IFRB to supply all delegations with maps on which ellipses were plotted, not only for their own countries' services, but also for the neighbouring areas.

After a discussion in which the <u>delegates of France, Italy</u> and <u>Turkey</u>, the <u>representative of the IFRB</u> and the <u>Deputy Secretary-General</u> took part, that proposal was <u>adopted</u>.

2. Report by Working Group 5B (Document No. 204(Rev.1))

The <u>Chairman of Working Group 5B</u>, introducing Document No. 204(Rev.1) said that it reflected a consensus of all Region 2 countries represented at the Conference. In addition to providing the basis for a future Conference of Region 2 Administrations to be held not later than 1982, the document also included provisions to govern the introduction of Space Services in the 11.7-12.2 GHz band in Region 2 in the interim period. Annex A set out the principles used in drafting the agreement, while Annex B dealt with the efficient use of the geostationary orbit and frequency spectrum, without, however, drawing any conclusions with regard to the problem of inter-regional sharing, which was left to the Committees and Working Groups directly concerned with it. Drawing special attention to paragraph 4 of the document, he indicated that the words "at or" should be inserted before the word "within" in the penultimate line, so that the paragraph as a whole now read as follows :

"The intent of the second paragraph of Section 2 of the Annex is to ensure that, if broadcasting satellites are operated in accordance with the technical characteristics for the Broadcasting-Satellite Service for Region 2 as given in the relevant Appendix of the Final Acts, they may operate without further restrictions at or within the limits of the orbit, referred to in Section 1 of the Annex, for the Broadcasting-Satellite Service."

The document, if approved, should in his opinion appear as a separate section of Article III of the Final Acts, all references to "Appendix / of the Final Acts /" being replaced by references to the contents of Documents Nos. 224 and 231 wherever and in whatever form they appeared in the Final Acts. That point would undoubtedly be taken into account by the Editorial Committee. Lastly, a number of corrections to the Spanish text of Document No. 204(Rev.1) would be handed in to the Secretariat.

The <u>Chairman</u> invited the Committee to consider the Annex to the document.

The title and introductory paragraph of the Annex were approved.

Paragraph 1

After a brief discussion in which the <u>delegates of Chile</u> and <u>Guatemala</u>, as well as the <u>Chairman</u>, took part, the <u>Chairman of Working Group 5B</u> proposed the insertion of the word "however", before the words "for service to" in the third line.

It was so decided.

Paragraph 1, as amended, was approved.

Paragraph 2

The <u>delegate of Australia</u> wondered why the provisions for the location of space stations in the Fixed-Satellite Service were not as specific as those for the Broadcasting-Satellite Service.

The <u>delegate of the United States of America</u> explained that the decision not to provide specific limits for the location of fixed-satellite stations had been taken deliberately because of inter-regional sharing considerations.

Paragraph 2 was approved without change.

Paragraphs 2.1 and 3

Approved.

Paragraph 4

The <u>delegate of Mauritania</u> proposed the deletion of the word "relevant" in the second line.

With that amendment, paragraph 4 was approved.

Paragraph 5

The representative of the IFRB suggested that the paragraph should be placed in square brackets pending the publication of the report of Committee 6.

The delegate of Venezuela supported that suggestion.

It was so <u>decided</u>.

Paragraph 6

Approved.

The <u>delegate of Chile</u> proposed that paragraphs 7.1, 7.2, 7.3, 7.4 and 7.5 should be renumbered 7, 7.1, 7.2, 7.3 and 7.4 respectively. Document No. 376-E Page 8

It was so <u>decided</u>.

Paragraph 7

The <u>delegate of Finland</u>, drawing attention to Document No. 215 of the Convention, pointed out that the Conference could only recommend the holding of a regional administrative radio conference. Accordingly, the word "shall" in the first line should be replaced by "should".

It was so decided.

The <u>Chairman of Working Group 5B</u> said that the asterisk in the first line of the English text should be deleted.

Paragraph 7, as amended, was approved.

Paragraphs 7.1, 7.2, 7.3, 7.4, 8, 9 and 10

Approved.

Paragraph 11

The <u>Chairman of Working Group 5B</u> said that the passage in square brackets should be deleted, as the matter was being dealt with by Committee 6.

Paragraph 11, thus amended, was approved.

Annex A - PLANNING PRINCIPLES

Title and introductory paragraph

Approved.

Paragraph 1

Approved.

Paragraph 2

The <u>delegate of the United Kingdom</u> recalled that at the Committee's third meeting his delegation had already made the point that the text of paragraph 2 was based on a misapprehension. The geostationary orbital arc was not shared between Regions but between Services. In his opinion, the paragraph was irrelevant and should be deleted.

The delegate of Portugal agreed.

The <u>delegate of the United States of America</u> opposed the deletion of the paragraph. Annex A was, in part, historical in nature; Paragraph 2 represented a statement of fact, and could not be suppressed. Furthermore, it was in accordance with the Convention, which called for equitable access to the geostationary orbit. After further discussion, the <u>Chairman</u> suggested that an attempt to reconcile the conflicting views with regard to paragraph 2 should be made during the suspension of the meeting.

The meeting was suspended at 1725 hours and resumed at 1945 hours.

The Chairman of Working Group 5B said that following informal discussions a compromise solution appeared to be acceptable whereby the title and the text of paragraph 2 would be reworded to read :

"Equality of the right to operate by services in different regions

In accordance with No. 117 of the Radio Regulations, the principle is recognized of equality of the right to operate by different services of the same category in the various regions without causing harmful interference to services in the other regions".

Adoption of that text would require a consequential amendment in the title of paragraph 1 whereby the words "in Region 2" should be added at the end of the existing title. That would explain the scope of paragraph 1.

The <u>delegates</u> of the United Kingdom and the United States of America signified their acceptance of such a compromise.

The amended text of paragraph 2 and the consequential amendment to the title of paragraph 1 were <u>approved</u>.

Paragraph 3

Approved.

Paragraph 4

The <u>delegate of the United Kingdom</u>, supported by the <u>delegates of</u> <u>Norway</u>, Portugal, France, Switzerland, Italy, Ireland, Japan, Sweden, India and Luxembourg, proposed the deletion of the sentence in brackets at the end of paragraph 4 on the grounds that its contents were outside the competence of Working Group 5B. The matter should be dealt with elsewhere in the Final Acts.

The <u>delegate of Colombia</u> opposed the deletion of the sentence in brackets as a matter of principle since it referred to any agreement that might be reached by the countries of Region 2.

The delegates of Guatemala and Chile supported the previous speaker.

The <u>Chairman</u> said there appeared to be a clear majority in favour of the United Kingdom proposal.

The <u>delegate of Colombia</u> said that the content of Annex A to Document No. 204 was the result of a gentleman's agreement among all the Region 2 delegations, and consequently all administrations in that Region should support it. Document No. 376-E Page 10

Despite the fact that the agreement related exclusively to Region 2, it was administrations in Regions 1 and 3 who were supporting the United Kingdom proposal for the deletion of the sentence in brackets at the end of paragraph 4.

In his view, the principle stated therein, whose deletion was being requested by some developed countries in Western Europe, had been discussed, agreed, accepted and stated by the countries in Region 2.

Paragraph 4 was <u>approved</u> with the deletion of the last sentence within brackets.

Paragraph 5 and the footnote

The <u>delegation of Finland</u> supported by the <u>delegate of the United</u> <u>States of America</u> said that in order to avoid ambiguity references to the "plan" for Regions 1 and 3 throughout the text should be given a capital letter so as to distinguish it from the plan for Region 2. Perhaps the point could be referred to Committee 8.

It was so agreed.

At the Chairman's suggestion it was <u>agreed</u> to insert the words "for Region 2" after the word "plan" in the first and penultimate lines of paragraph 5.

Paragraph 5, as amended, and the footnote was approved.

Paragraph 6

The <u>delegate of France</u> proposed that the penultimate line be corrected to read : "the requirements of Region 2 ...".

It was so agreed.

Paragraph 6, as amended, was approved.

Paragraphs 7 and 8

Approved.

The <u>delegate of Chile</u> proposed, in the interests of clarity, the addition of the words "in Region 2" in the title of Annex A.

It was so agreed.

<u>Annex</u> B

First two paragraphs

Approved.

Paragraphs 1 - 9

Approved.

Paragraph 10

In order to meet an objection raised by the <u>delegate of France</u>, the <u>delegate of the United States of America</u> proposed the <u>deletion</u> of the words "for a TV channel (or any channel in the Fixed-Satellite Service)".

That amendment was approved.

Paragraph 10, as amended, was approved.

Document No. 226

The <u>Chairman of Working Group 5B</u>, presenting its second report (Document No. 226) said that some corrections to the Spanish version would be handed direct to the Secretariat. The report contained four draft resolutions in Annexes A to D.

Annex A

The <u>delegate of Finland</u>, referring to the heading of the first draft resolution considered that it should be designated as a "recommendation" in conformity with No. 215 of the 1973 Malaga-Torremolinos Convention. As a consequential amendment the word "recommends" should be substituted for the word "resolves" in the fourth paragraph.

It was so decided.

The <u>delegate of Sweden</u> proposed the deletion of <u>considering</u> a) which was superfluous.

It was so decided.

In order to meet a point made by the <u>delegate of Bolivia</u>, the <u>Deputy Secretary-General</u> suggested the substitution of the words "present Conference" for the words "WARC-BS (Geneva, 1977)" and the addition of the words "for Region 2" at the end of considering b).

That amendment was approved.

The <u>delegate of Algeria</u> pointed out that as the text was now in the form of a recommendation and not a draft resolution consequential amendments would have to be made in <u>recommends</u> a) and b) so that the verbs were changed to the correct tense. Clearly the provisions of a recommendation could not be mandatory. That comment applied to both recommends a) and b).

It was so agreed.

"recommends" paragraph c)

The <u>representative of the IFRB</u> suggested that the Recommendation might be greatly simplified by deleting the last two sentences of paragraph b) and paragraphs c) and d) in their entirety and replacing those texts by a paragraph c) reading "that the agenda of this Conference shall take into account the principles contained in Annexes A and B to the present Final Acts".

The <u>Chairman of Working Group 5B</u> said that the Region 2 countries had unanimously decided that the relevant provisions of Document No. 204 should be repeated in the operative part of the Recommendation, since there was still some controversy concerning the form that the plan for Region 2 would take in the Final Acts.

The <u>delegates of Cuba</u>, <u>Guatemala</u>, <u>Mexico</u>, <u>Chile</u>, <u>Bolivia</u>, <u>Uruguay</u>, <u>Paraguay</u>, <u>Venezuela</u> and <u>Brazil</u> endorsed that statement.

The <u>delegate of the United States of America</u> also endorsed the statement of the Chairman of Working Group 5B, adding that the provision in paragraph d) did not appear in Document No. 204.

The <u>delegate of Algeria</u>, supported by the <u>delegate of Benin</u>, proposed that the second sentence should begin with the words "Account should also be taken of the provisions of No. 47 of the International Telecommunication Convention (Malaga-Torremolinos, 1973), of the decisions ...".

The <u>delegate of the United States of America</u> said that a reference to the Convention was out of place in paragraph c), since the 1977 and 1979 WARCs and the relevant CCIR Recommendations had been mentioned there for very specific reasons - their direct effect on planning parameters in 1982. Moreover, it was self-evident that the Convention governed the proceedings of all ITU conferences.

The <u>Chairman</u> suggested that the phrase proposed by the Algerian delegate might be added at the end of paragraph a).

The <u>Deputy Secretary-General</u> suggested that the opening words of paragraph a) might be amended to read "that a Regional Administrative Radio Conference should be convened, in accordance with the provisions of Articles 7 and 54 of the Convention, ...".

The <u>delegate of Algeria</u> said he would prefer the specific reference to No. 47 of the Convention to be inserted after the words "in accordance with" in paragraph a).

The delegate of Togo supported the Algerian proposal.

The <u>delegate of the United States of America</u> supported the text suggested by the Deputy Secretary-General.

The <u>delegates of Canada, Chile, Uruguay, Portugal, Mexico</u> and <u>Paraguay</u> said that the proposed insertion was superfluous, since the Convention in any case governed all ITU activities.

Document No. 376-E Page 13

The <u>Chairman</u> suggested that, in view of the considerable opposition to the proposal, the phrase should not be inserted in either paragraph a) or c), but that he should refer to the Algerian amendment in the Committee's report to the Plenary Meeting.

It was so <u>decided</u>.

Paragraph c) was approved.

"recommends" paragraph d)

Approved.

"recommends" paragraph e)

The <u>delegate</u> of France proposed that the words "Fixed-Satellite Service" be replaced by "other services", since that would correspond more closely to the relevant provisions of Document No. 204(Rev.1).

The <u>delegates of Cuba</u>, Togo and <u>Algeria</u> supported that proposal.

The <u>delegates of the United States of America</u>, Canada and <u>Bolivia</u> said that they would prefer the text to remain unchanged. The reference to the Fixed-Satellite Service followed logically from paragraph a), which specified that the purpose of the Regional Conference was to carry out detailed planning for the Broadcasting-Satellite and Fixed-Satellite Services in Region 2.

After some discussion, it was <u>decided</u> to <u>approve</u> paragraph e) in its existing form and to add a new paragraph f) reading as follows :

"that in drafting the detailed plan referred to above, account should also be taken of the Terrestrial Services with which the band is shared."

"invites the Administrative Council" paragraph

Approved.

Annex A, as amended, was approved.

Annex B

<u>Approved</u>, subject to the insertion of the words "a Regional" before "Administrative Radio Conference" in the title.

Annex C

Title, introductory paragraph and "considering" paragraph a)

Approved.

"considering" paragraph b)

Document No. 376-E Page 14

The <u>delegate of France</u> proposed that the words "Fixed-Satellite Service" be replaced by "other services".

Approved as amended.

"considering" paragraph c) and "invites the IFRB" section

Approved.

Annex C, as amended was approved.

Annex D

The <u>delegate of Venezuela</u> suggested that Annex D be placed in square brackets pending the circulation of a similar but more comprehensive Resolution which Committee 6 had adopted and which might well supersede the text in Annex D.

It was so decided.

Annex D was approved on that understanding.

3. <u>Technical data used for drawing up the Plan(s) and to be used</u> in implementing the Plan(s) (paragraphs 3.7 and 3.8.1) (Document No. DT/48)

The <u>Chairman</u> said that the Chairman of the Editorial Committee had drawn his attention to two omissions from Document No. 224 (B.3). He had accordingly drafted texts for paragraph 3.7 on channel spacing - the exact frequencies to be supplied later by Working Group 5A - and paragraph 3.8.1 on orbital spacing.

Document No. DT/48 was approved.

The meeting rose at 2235 hours.

The Secretary :

The Chairman

A. PETTI

D. KANE

BROADCASTING SATELLITE CONFERENCE

Document No. 377-E 25 March 1977 Original : French

(Geneva, 1977)

1.

COMMITTEE 5

SUMMARY RECORD

OF THE

SEVENTH MEETING OF COMMITTEE 5

(PLANNING)

Wednesday, 9 February 1977, at 1130 hrs

Chairman : Mr. A PETTI (Italy)

Sub,	jects discussed	Documents Nos.
1.	Draft resolution	DT/50
2.	Table showing correspondence between channel numbers and assigned frequencies	DT/51
3.	Draft recommendation submitted by Australia	250
4.	Draft resolution relating to experimental broadcasting-satellite systems	253

5. Communication from the delegate of France

Draft resolution (Document No. DT/50)

The <u>Chairman</u> reminded the Committee that at its last meeting it had agreed that after the present Conference the IFRB should prepare a document giving information on the geographical coordinates of the various test points and on the corresponding elevation angles. The draft resolution in Document No. DT/50 was designed to give effect to that decision.

Before inviting the Committee to consider the draft resolution paragraph by paragraph, he asked whether there were any suggestions for its title.

The <u>delegate of Canada</u> thought that the title should indicate that the resolution applied to planning in Regions 1 and 3.

It was <u>decided</u> that the draft resolution should be given its title later, due regard being paid to the delegate of Canada's comment.

First preambular paragraph

It was <u>decided</u> to replace the paragraph by the abbreviated title of the Conference just adopted at the eighth Plenary Meeting.

Preambular paragraphs a) and b)

The <u>delegate of France</u> proposed that the expression "protection ratio" should be replaced by "protection margins".

The <u>Technical Secretary</u> said that in the light of the comment made by the delegate of Canada concerning the title of the resolution, the words "for Regions 1 and 3" should be added in paragraph a) after the word "work" and at the end of paragraph b) after the word "Plan".

With those amendments, preambular paragraphs a) and b) were approved.

Preambular paragraph c)

Preambular paragraph c) was approved without change.

Introductory phrase of the operative part

The <u>Chairman</u> invited the Committee to decide whether the IFRB should publish the information requested as an annex to the Plan or as a separate document.

The <u>delegate of Japan</u> reminded the Committee that at an earlier meeting it had been agreed to publish the information in an information document. He therefore proposed that the words "an annex to the Plan" should be replaced by "a document".

The <u>Chairman</u>, replying to a question from the <u>delegate of Thailand</u>, said that the document would be published in the course of 1977.

It was <u>decided</u> to amend the introductory phrase of the operative part of the resolution to read as follows : "to prepare, with a view to its publication by the Secretary-General, during 1977, a document containing the following information :".

<u>Operative paragraph a)</u>

Operative paragraph a) was approved without change.

Operative paragraph b)

The <u>Technical Secretary</u> said that the document referred to in the paragraph was Annex 3 to Document No. 103(Rev.2).

Document No. 377-E Page 3

The <u>delegate of Spain</u> observed that some test points did not seem to lie within the polygon delimiting a country's territory. He wondered whether it was intended to publish a new document containing the definitive list of test points.

The <u>Technical Secretary</u> explained that the Annex to Document No. 103(Rev.2) had been drawn up on the basis of the latest amendments or corrections submitted by delegations. The list of test points contained in the Annex was therefore to be regarded as definitive, on the understanding that if any problems should arise they would be settled directly with the delegation concerned.

Operative paragraphs c) and d)

Operative paragraphs c) and d) were approved without change.

Operative paragraph e)

Operative paragraph e) was <u>approved</u>, the words "protection ratio" being replaced by "protection margins".

The <u>delegate of Germany (Federal Republic of</u>) thought that the draft resolution should include a note defining protection margins.

The <u>delegate of France</u> pointed out that the definition was to be found in CCIR Report 633.

It was <u>decided</u> that after the words "protection margins" in preambular paragraphs a) and b) and operative paragraph e), an asterisk should be added referring to the following footnote : "*) for the definition of protection margins, see CCIR Report 633".

The draft resolution in Document No. DT/50, as amended, was approved.

2.

Table showing correspondence between channel numbers and assigned <u>frequencies</u> (Document No. DT/51)

The <u>Chairman</u> said that the table had been drawn up in the light of the decisions taken by Committee 4 on the subject of guard bands. However, in order to avoid having figures with too many decimals, it had been decided to adopt the value 19.18 MHz for channel spacing and to reduce slightly the lower guard band fixed by Committee 4 from 14 MHz to 13.98 MHz.

The <u>Technical Secretary</u> said that only channels 1 to 24 were to be used in Region 3.

It was <u>decided</u> that the table in Document No. DT/50 should be included in the part of the Plan relating to Regions 1 and 3.

The meeting was suspended at 1215 hrs and resumed at 1415 hrs.

3. Draft recommendation submitted by Australia (Document No. 250)

The <u>delegate of Australia</u>, introducing his delegation's draft recommendation, explained that it was based on Document No. 35. The <u>delegate of India</u> supported the Australian delegation's draft recommendation, subject to an addition in preambular paragraph c).

The <u>delegate of the USSR</u> considered that the draft recommendation dealt with a purely administrative question which did not come within the Conference's competence. He thought the CCIR could hardly be invited to undertake a study which might lead it to make recommendations affecting frequency assignments. At all events, only the 1979 World Administrative Radio Conference would be competent to amend the table of frequency assignments, should it be decided that the Broadcasting-Satellite Service was to use the band 12.2-12.5 GHz.

The <u>delegates of Canada</u>, Japan, the United Kingdom and the United States of America shared the previous speaker's view.

The <u>Chairman</u> asked whether any other delegations supported the Australian draft recommendation.

Since none did, it was <u>decided</u> not to adopt the draft recommendation in Document No. 250.

The meeting was suspended at 1545 hrs and resumed at 1900 hrs.

4.

Draft resolution relating to experimental broadcasting-satellite systems (Document No. 253)

The <u>delegate of the United Kingdom</u> said that the purpose of the proposal submitted by his delegation was to enable countries which so wished to pool their resources so that they could set up an experimental broadcasting-satellite system without having to resort to lengthy and complex coordination procedures. It involved setting aside two channels which would be protected against Terrestrial Services in Regions 1 and 3 and which several administrations could use in turn for a few months in order to carry out experiments. The formula was attractive in that it would enable countries with limited resources to acquire some experience of satellite broadcasting at low cost before they embarked on the highly expensive process of setting up an operational system. All the countries in Regions 1 and 3, moreover, could benefit from the results of the experiments.

The <u>delegates of Greece and of Switzerland</u> supported the draft resolution submitted by the delegation of the United Kingdom.

The <u>delegate of the United States</u> said that the proposal could be of considerable value and deserved to be supported. It would be advisable, however, to insert a clause in the operative part of the draft resolution stipulating that experimental systems of that kind should be used in such a way that they did not interfere with the functioning of any other system set up in accordance with the sharing criteria defined in the Final Acts of the Conference.

Document No. 377-E Page 5

The <u>delegate of the USSR</u> said that in principle he had no objection to the draft resolution. However, the choice of the two channels to be set aside for experimental broadcasting would raise a practical difficulty in that all the channels mentioned in the Plan had already been allocated. It would therefore be advisable to choose the two channels from those allocated to the administrations wishing to undertake such experiments and to say so clearly in the draft resolution.

The <u>delegates of Sudan and of Egypt</u> shared the view of the previous speaker.

The <u>delegate of China</u> considered that the Committee should not adopt a draft resolution recommending the use for experimental purposes of two channels already allocated in the Plan.

The <u>delegate of the United Kingdom</u> acknowledged that the use of a channel already allocated in the Plan could pose practical problems since the various countries were doubtless going to organize their terrestrial systems on the basis of the allocations in the Plan. However, if a group of countries decided to move one of the channels allocated to it in order to carry out an experiment in another country, there was bound to be some impact on the proper functioning of the Terrestrial Services of neighbouring countries. It was precisely to obviate such difficulties that it was proposed to designate two channels which would be set aside for experimental work and which could not be used for Terrestrial Services over a given period. One way to solve the problem of the designation of the channels would be for one or two administrations to offer to make one of their channels available to administrations wishing to set up joint experimental systems.

The <u>delegate of Australia</u> pointed out that any Experimental Satellite Service had to be designed and operated in accordance with the Plan and the sharing criteria defined by the Conference.

The <u>delegates of Switzerland and of Norway</u> said that the formula put forward by the delegate of the United Kingdom was very valuable insofar as it would promote technological development, since the results of the experiments could be shared by all the countries in Regions 1 and 3. The problem of determining which channels should be set aside could probably be solved without too much difficulty.

The <u>delegate of Sudan</u> agreed that the possibility of setting up experimental systems was indeed attractive, but considered it unacceptable to set aside for that purpose one or two of the forty channels in the Plan since all those channels had already been allocated. It would have been better to submit the request at the beginning of the Conference so that it could have been taken into account in Document No. 103(Rev.2) containing the list of requests submitted by administrations.

The delegate of Saudi Arabia shared that view.

The <u>Chairman</u> pointed out that the list contained only requests submitted individually by countries. As far as the request under discussion was concerned, it was a question of setting aside two channels for use by groups of countries.

The <u>delegate of Mauritania</u> suggested that administrations should be asked whether some of them were willing to loan one of the channels allocated to them.

The <u>delegate of Australia</u> said that it was the administrations wishing to carry out experiments which should make their channels available. For instance, Australia, New Guinea and New Zealand had planned, as an initial step, to set up a single system in which the orbital positions and frequencies specified in the final plan would be used. Other countries might well follow that example.

The <u>delegate of Papua New Guinea</u> agreed with the previous speaker. In his opinion, countries wishing to carry out satellite broadcasting experiments should form a group and establish the necessary coordination procedures so that they could be taken into account in the operation of terrestrial systems.

The <u>delegate of France</u> supported the suggestion of the previous speaker. While the United Kingdom proposal was of value, the choice of the two channels to be set aside for experimental broadcasting would raise considerable difficulties insofar as it would have to be based on the use made of the channels and in particular the implications for the Terrestrial Services would have to be considered.

The <u>delegate of the USSR</u> said that the period of ten years mentioned in the draft resolution was far too long. Quite a number of broadcasting-satellite systems would already be in service in ten years' time and their operation was likely to be jeopardized by the existence of experimental systems.

The <u>delegate of Switzerland</u> thought that there were no real grounds for the doubts expressed by the previous speaker inasmuch as there would no longer be any need for experimental satellites once several operational systems were in service. Countries wishing to become familiar with the operation of such systems could then do so directly. However, the period planned for the use of channels for experimental purposes could be shortened.

The delegates of Spain and Belgium supported that suggestion.

The <u>Chairman</u> asked whether, if the Committee did not approve the draft resolution submitted by the delegation of the United Kingdom, the procedures laid down in the Final Acts of the Conference allowed for the possibility of using, on a temporary basis and without amending the Plan, some of the channels already allocated in a manner which was not entirely in conformity with the Plan. The <u>delegate of the USSR</u> suggested that the delegation of the United Kingdom should revise its draft resolution to take into account the views expressed during the discussion.

The <u>delegates of Ivory Coast</u>, <u>Mauritania and Sudan</u> supported that proposal.

The <u>delegate of the United Kingdom</u> undertook to submit a new draft resolution to the Committee.

It was so <u>decided</u>, on the understanding that the amended document should be examined in Plenary Meeting.

5. Communication by the delegate of France

Speaking as a member of the Organisation Télédiffusion-France (TDF), the <u>delegate of France</u> wished to inform delegations, particularly Working Group 5A and the Group on Planning, that the computer and staff made available to the Conference by that organization would no longer be available as from that day.

The <u>Chairman</u> expressed his thanks to the personnel of TDF for their valuable assistance.

The meeting rose at 2100 hours.

The Secretary :

The Chairman :

D. KANE

A. PETTI

BROADCASTING SATELLITE CONFERENCE

Document No. 378-E 25 March 1977 Original : English

(Geneva, 1977)

COMMITTEE 5

SUMMARY RECORD

OF THE

EIGHTH MEETING OF COMMITTEE 5

(PLANNING)

Thursday, 10 February 1977, at 1815 hrs

Chairman : Mr. A. PETTI (Italy)

Subjects discussed :

Document No.

ACHIVE

GENÈN

 Approval of the summary record of the fourth meeting of Committee 5
239
Third report of Working Group 5B
254

z. Inita report of working droup JD

3. Request by the delegation of Iceland

1. <u>Approval of the summary record of the fourth meeting of Committee 5</u> (Document No. 239)

The <u>delegate of the German Democratic Republic</u> asked that the first two lines in the last paragraph on page 7 be corrected by substituting the words "his Delegation had written to the IFRB that" for the words "his Administration had written to say that".

The <u>delegate of Switzerland</u> asked that the last sentence in the first paragraph of the summary of his statement (page 8) be amended to read : "Since all neighbouring countries had agreed to it, the spillover could be regarded as a technical spillover".

Subject to those corrections, the summary record of the fourth meeting of Committee 5 was <u>approved</u>.

2. <u>Third Report of Working Group 5B</u> (Document No. 254)

The <u>Chairman of Working Group 5B</u>, presenting its third report (Document No. 254), said that it contained the results of some planning exercises Document No. 378-E Page 2

for Region 2 and had been submitted for information in order to assist administrations in Region 2 when preparing for their future regional conference.

Some purely editorial corrections would have to be made to bring the French and Spanish texts into line with the English version.

The <u>delegate of Mexico</u> said that as the planning exercises outlined in the document were not intended to prejudge the development of the plan in Region 2, he proposed that the third sentence in the second paragraph of the introduction be deleted.

The <u>Chairman of Working Group 5B</u>, speaking as the delegate of Canada, said that although that sentence did reflect the discussions in Working Group 5B he could accept the Mexican amendment for the sake of reaching agreement quickly.

The Mexican amendment to the second paragraph of the introduction was approved.

The <u>delegate of Venezuela</u> said that as the purpose of the exercises had been to provide information that might be useful to Region 2 countries for a future regional conference, he asked that copies of the document be sent also to administrations which had not attended the present Conference with some explanatory notes indicating the source of the data contained in the annexes. It would also be useful to circulate to them IFRB's report (Document No. 27).

The <u>delegate of Cuba</u>, endorsing the comments by the previous speaker, expressed his delegation's appreciation for the help given by the IFRB and the Sub-Group (and particularly Mr. Kovačs) which had carried out the planning exercises. He hoped that Document No. 254 could be kept up to date lest the data it contained were to become obsolete by 1982 because of technological progress.

The <u>Chairman of the IFRB</u> said that he had understood from the Chairman of Working Group 5B that none of the annexes to Document No. 254 would form part of the Final Acts and accordingly the whole document, together with its annexes, could be transmitted by circular letter to administrations which had not been represented at the present Conference.

Referring to the point raised by the delegate of Cuba, he said that in accordance with No. 482 of the Radio Regulations, the IFRB could assist in bringing the document up to date as part of its task of helping in the technical planning of a regional conference.

The <u>Chairman</u> suggested that the Committee take note of Document No. 254 and that the IFRB be instructed to despatch a corrected version to administrations not present at the Conference and to keep the annexes up to date in the light of technological progress.

It was so <u>decided</u>.

The <u>Chairman of Working Group 5B</u> thanked the members of the Sub-Group and Mr. Kovacs who had worked long hours in preparing the document.

Document No. 378-E Page 3

The <u>Chairman</u> said that Committee 5 owed a debt of gratitude to the Chairman of Working Group 5B and its members who by strenuous efforts had found common ground and prepared data that would surely be most useful in preparing for the 1982 regional conference.

3. Request by the delegation of Iceland

The <u>delegate of Iceland</u> stated that he had handed in to the IFRB for inclusion in the final version of the Plan, certain amendments to the requirements of Denmark and Iceland together with a note X reading : "This assignment results from a common requirement of the Administrations of Denmark and Iceland. The service area includes the Faroe Isles and Iceland. The assignment may after consultations between the two Administrations be used by either of them."

The <u>Chairman</u> said that the request would be taken into account in preparing the final version of the Plan.

The meeting rose at 1850 hours.

The Secretary :

D. KANE

The Chairman :

A. PETTI

BROADCASTING SATELLITE CONFERENCE

Document No. 379-E 30 March 1977 Original : English

(Geneva, 1977)

COMMITTEE 5

SUMMARY RECORD

OF THE

NINTH MEETING OF COMMITTEE 5

(PLANNING)

Friday, 11 February 1977, at 0940 hrs

Chairman : Mr. A. PETTI (Italy)

Subjects discussed :

Document No.

ACHIVA

11 I T

GENEN

 Approval of the summary record of the fifth meeting of Committee 5
Report of Working Group 5A (examination of the Plan)
265

1. <u>Approval of the summary record of the fifth meeting of Committee 5</u> (Document No. 264)

The <u>delegate of Italy</u> asked for the second sentence of his intervention on page 6 to be amended as follows :

"The Italian delegation, which had not been officially informed of a request for agreement on the part of Monaco, considered that Monaco should reduce its service area in the direction of Italy to the technically necessary minimum."

The summary record was approved, with that amendment.

2. Report of Working Group 5A (examination of the Plan) (Document No. 265)

The <u>Chairman</u> invited the Committee to make a preliminary examination of Document No. 265. Approval would be deferred until the following day, when a new run-through from the computer would be available.

The <u>delegate of the United Kingdom</u>, speaking as <u>Chairman of Working</u> <u>Group 5A</u>, introduced Document No. 265 which contained a draft Plan produced by Working Group 5A-2/3 based on the requirements given in Document No. 103(Rev.2) with a number of further changes and reductions emanating from the Conciliation Group. The draft Plan was more promising, in respect both of interferences and negative margins, than either of the draft Plans produced by Sub-Working Groups 5A2 and 5A3. Moreover, the large interferences which did appear in the new draft Plan had been traced mainly to errors in the transfer of data from one computer to another, so it was to be expected that they could be substantially reduced. It had not been possible, in the draft Plan, fully to meet the wishes of every administration but all requests had been studied and only lack of time had prevented their full incorporation.

ł

He would submit a document in writing giving corrections to the draft Plan.

In reply to a question by the <u>Technical Secretary</u>, regarding the omission of AND 262 from the draft Plan, the <u>Chairman</u> explained that it had been due to a misunderstanding in the planning groups. Since the characteristics were the same, it had been thought that AND 262 duplicated AND 238 and the latter, which had been submitted first, had been chosen.

The <u>delegate of Spain</u> said that the requirement submitted by Spain, No. AND 262, referred solely to the polygon delimiting the territory of Andorra. His delegation considered that the proposal to be taken into account in the allocation of channels should be the one whose coverage area corresponded to that polygon. He had therefore submitted a note requesting the IFRB to analyze the coverage area of both requirements so as to determine which would cause least interference in the Plan, and then to choose a single group of 5 channels for Andorra.

The <u>delegate of the United Kingdom</u> assured the Spanish delegate that the entry in Document No. 265 gave the minimum value to which the latter had referred, and urged him not to introduce any extra work for the IFRB at that stage.

The <u>Chairman</u> confirmed that page 10, line 35, gave the minimum value corresponding to the territory of Andorra.

In reply to a question by the <u>delegate of Turkey</u>, as to the form in which the boresight geographical coordinates would appear in the final Plan, the <u>Technical Secretary</u> said that for the time being it was not the intention to convert them so as to show 2 decimal places although that might be considered after the Conference.

In reply to a further question by the <u>delegate of Turkey</u>, the <u>Secretary of the Committee</u> said that the maximal gain given in Column 7 had been calculated in accordance with the formula in Document No. 27, but that could be altered when the numbers were entered in the Plan so as to bring it into line with the modified formula approved in Document No. 224.

The <u>delegate of the Federal Republic of Germany</u>, commenting on page 3 of Document No. 265, suggested that "7. Polarization" should refer to "right hand circular and left hand circular polarization" instead of "1 = direct, 2 = indirect."

The <u>Technical Secretary</u> said that comment would be borne in mind during the first reading of the relevant document.

The <u>delegate of Laos</u>, referring to Channel 1, line 18, on page 6, said that beam THAO142A covered almost the whole of his country's territory and he was therefore obliged to request a reduction of the ellipse of that beam.

The <u>Chairman of Working Group 5A</u> gave a detailed explanation of the method employed by the Working Group to ensure that all ellipses were, in fact, the smallest possible and that any spillover was technical rather than intentional.

The <u>Chairman</u> asked the delegate of Laos to accept that explanation as an assurance to the effect that the spillover was indeeed technical and minimal.

The <u>delegate of the Federal Republic of Germany</u>, referring to Channel 2, line 34, on page 6 said that the real values of the figures appearing in Column 4 were 9.66 and 49.90. The same applied to all entries under serial No. D87 in Channels 6, 10, 14 and 18.

The <u>delegate of Saudi Arabia</u>, referring to Channel 2, lines 40 and 41 on page 7, announced that in order to improve the protection margins of assignments to the Yemen Arab Republic, his Administration had agreed to exchange the channels assigned to it under Serial No. 003 with those assigned under No. 0275, and vice versa. Accordingly, the entries under Serial No. 0275 in Channels 4, 8, 12, 16 and 20 should be transferred to the 0003 beam and those under Serial No. 0003 in Channels 2, 6, 10, 14 and 18 to the 0275 beam.

The <u>delegate of Egypt</u> expressed the hope that the changes announced by the previous speaker would not in any way degrade the protection margin figures given in the draft Plan.

The <u>delegate of Spein</u>, referring to Channel 4, line 35 on page 10, as well as to the relevant lines in Channels 8, 12, 16 and 20, said that his delegation had handed a note to the IFRB requesting a recalculation of the Andorra coverage.

The <u>delegate of France</u> assured the delegate of Spain that the power of the Andorra beam as shown in Document No. 265 provided coverage of the territory of Andorra only.

The <u>Technical Secretary</u> announced that the new Serial No. for AND 262 and other entries for Andorra would henceforth be 341.

The <u>delegate of Bulgaria</u> drew attention to an error in column 4 of Channel 4, line 45 on page 10 and in the corresponding entries in Channels 8, 12, 16 and 20.

The <u>Chairman of Working Group 5A</u> wondered whether changes already handed in to the IFRB and relating strictly to the correction of errors need be reported to the Committee.

The <u>Technical Secretary</u> explained the procedure for handling changes used by the IFRB. Any requests which gave rise to difficulties would be put before the Committee en bloc at a later stage. Document No. 379-E Page 4

The <u>representative of the IFRB</u> said that in accordance with a decision taken on the previous day by Working Group 5A, the IFRB had provided teams of engineers from 0800 hours to assist administrations in connection with changes of ellipses. As the problems which had arisen had proved more difficult than expected, it was felt that they should be reported to Committee 5.

Changes affecting the orientation of ellipses or calling for a reduction of the axis of the ellipse had been requested in respect of the following beams :

URS 81	NZL 287
URS 59	D 87
SUI 140	CVA 85
GRC 105	INS 28
CHN 159	INS 38
СКН 52	AUS 4, 5, 6, 7, 8 and 9
CKN 53	PNG 271
G 27	DNK 91
ZAI 328	ISL 50
BUL 20	PAK 283
SEN 222	IND 37 and 38.

Unresolved difficulties were still outstanding between Morocco and Algeria in respect of MAR 209 and ALG 251, between the Republic of Korea and the Democratic People's Republic of Korea, and the Yemen Arab Republic and the People's Democratic Republic of Yemen. A difficulty between the People's Democratic Republic of Yemen and the Sultanate of Oman would be dealt with after the Conference with the consent of both Administrations and in accordance with the procedures provided for the purpose. Belgium had requested a power increase of 1.3 dB in respect of BEL 18 because it considered that the AG for that beam had not been correctly calculated. The CHN 160 beam had been changed by agreement between the Administrations concerned. Settlement was pending of a problem still outstanding between Ethiopia and Sudan. Lastly, a considerable amount of reciprocal overspill between Ethiopia and Somalia had had to be noted without further action in view of the absence of Somalia from the Conference.

The <u>delegate of Morocco</u>, referring to the list read out by the representative of the IFRB, said that the modification requested by his Administration was merely a correction and not a modification properly speaking. With regard to the problem which subsisted between his and the Algerian delegation, he said that every effort had been made to avoid raising the matter in the Committee. However, the ellipses listed for Algeria in the draft Plan were not in accordance with the agreement which his delegation believed to have been reached during informal discussions with the Algerian delegation. Those ellipses were larger than the minimum required to cover Algerian national territory, and his delegation would be unable to accept them if they were retained in the Plan as they stood. They could, in his view, be further reduced, and his delegation would be willing to discuss the matter with the experts in the hope that a compromise solution could be found.

The <u>delegate of Algeria</u> said that Sub-Working Group 5Al had, at the specific request of his delegation, minimized the ellipses for Algeria. It should be noted that the calculations had been made by that Group and not by his Administration. The ellipse submitted that morning for beam 251 was the smallest one possible to meet his Administration's minimum requirements.
Document No. 379-E Page 5

The <u>delegate of Tunisia</u> observed that although the ellipse to which the Algerian delegate had referred had been reduced vis-a-vis its coverage of parts of Italy and Spain, it still covered a very large area of Tunisia. However, his delegation continued to hope that a satisfactory solution could be found to the problem.

The delegate of Afghanistan made the following statement :

"I would like to draw the attention of the Committee and the IFRB to the fact that four of our channels, numbered 1, 5, 9 and 13, are subjected to serious interference of (protection margin of - 2) as a result of requirements submitted by various Administrations. I have already raised this question with the Chairman of Sub-Working Group 5A, the Conciliation Group and the Administrations concerned. The results obtained by my Administration up to the present are a matter of great concern to me. I am confident that, through this Committee, some effort will be made to reduce this interference figure, having regard to the fact that the figures which appear in the plans drawn up by Sub-Working Groups 5A2 and 5A3 were perfectly satisfactory to my Administration."

The <u>Chairman of Working Group 5A</u> said that the interference figure to which the previous speaker had referred was not considered by the experts to constitute serious interference.

The <u>representative of the IFRB</u> said that a decision by the Committee was needed in respect of the modification submitted by the Democratic People's Republic of Korea, since a beam increase was involved.

The <u>delegate of the Republic of Korea</u>, referring to the proposal by the North Korean Delegation, observed that his country's territory included several islands situated at some distance from the mainland. The draft Plan in Document No. 265 contained the minimum requirements of his Administration for coverage of the service area in question. His delegation had no wish to cause intentional spillover into other countries, nor was it prepared to accept such spillover. The proposal by North Korea to increase its ellipse over and above its just and reasonable requirements was, therefore, quite unacceptable to his delegation.

The <u>delegate of the Democratic People's Republic of Korea</u> said that the beam requested by South Korea was far larger than the minimum necessary to cover its service area and in fact covered a very large share of his country's territory as well. His delegation's request for the reduction of that beam had been refused. Consequently, it was only natural that his Administration should, on a reciprocal basis, wish to provide coverage for its islands; it had chosen, for that purpose, a minimum ellipse which did not even fully cover the islands in question.

The <u>delegate of China</u> said that, under No. 405 of the Radio Regulations, beams should be reduced to the minimum dimensions necessary with a view to reducing the interference caused to other administrations. Consequently, his delegation supported the position of the Democratic People's Republic of Korea. Document No. 379-E Page 6

In reply to a request by <u>the Chairman</u>, the <u>representative of the IFRB</u> provided information on the characteristics of the requirements submitted by the Democratic People's Republic of Korea. Replying to a question by the <u>Chairman</u> <u>of Working Group 5A</u>, he said that the IFRB could not check whether the ellipses in question were the minimum necessary for the coverage originally requested, nor could it take any other action of that kind, without the prior agreement of the Administrations concerned.

The <u>Chairman</u> said that the modification submitted by the Democratic People's Republic of Korea involved an ellipse that was greater than the one in that country's original requirements and was therefore not the type of modification which the Committee had decided to accept at the present stage of its work. Consequently, the only course open to the Committee would be to take note of the request by the Democratic People's Republic of Korea and its rejection by the Republic of Korea.

The <u>delegate of the USSR</u> said that the principles adopted by Working Group 5A in relation to the structuring of ellipses had been observed by the IFRB when drawing the related maps. However, errors had been made in certain individual cases. For instance, the ellipse plotted for URS 0059 had been increased so that it now covered not only the required area in the Soviet Union but also an area of approximately the same size involving several neighbouring European countries. He wished to make it clear that the ellipse thus plotted did not correspond to the requirements submitted by his Administration. That error, together with others concerning URS 0160 and URS 0081, had been brought to the attention of the IFRB. Furthermore, he requested that the necessary action be taken to correct an error which he had just detected in respect of the orientation of the ellipse for URS 0068.

The <u>Chairman</u> said that any corrections resulting from solutions found by Administrations to the unsettled problems mentioned by the representative of the IFRB earlier in the meeting should be handed in to the Secretariat not later than 1400 hours that day.

The meeting rose at 1240 hours.

The Secretary :

D. KANE

The Chairman :

A. PETTI

INTERNATIONAL TELECOMMUNICATION UNION

BROADCASTING SATELLITE CONFERENCE

(Geneva, 1977)

Document No. 380-E 30 March 1977 Original : English

COMMITTEE 5

SUMMARY RECORD

OF THE

TENTH MEETING OF COMMITTEE 5

(PLANNING)

Friday, 11 February 1977, at 2145 hrs

Chairman : Mr. A. PETTI (Italy)

Subject discussed

Document No.

1. Preliminary examination of the Plan (continued) 265 DT/53

Preliminary examination of the Plan (Documents Nos. 265 and DT/53) 1. (continued)

Channels 8 to 20

No comments.

Channel 21

The delegate of Algeria said that entry 35 was a case of intentional spillover about which a letter had already been forwarded to the IFRB asking for the beam to be reduced to correspond to the size of the country concerned.

The Chairman of Working Group 5A said that a procedure had been set up on reduction of beams in which all delegations were treated in the same way. The minimum beam was calculated from equations using both the TDF and ITU computers and the smaller of the two selected.

The delegate of Algeria wished to place on record his delegation's formal reservation with regard to beam MRC0209A.



Document No. 380-E Page 2

Channel 22

The <u>delegate of Algeria</u> said that the above remark applied also to entry 25 (MINO223A).

The <u>delegate of Tunisia</u> said the figure in column 6 of Document DT/53 should read 135° .

Channel 23

The <u>delegate of Spain</u> indicated with satisfaction that a compromise solution had been found in relation to entry 33 (beam CVA0085A), thanks to a spirit of collaboration. The information on change of ellipse and reduction of 0.5 dB would have to be inserted manually as it did not appear in Document No. DT/53.

Channels 24 to 40

No comments.

The Chairman then invited general comments on the plan as a whole.

The <u>delegate of Afghanistan</u> drew attention to this special problem of his country which had been discussed in the Conciliation Group; the Group had called upon the views of Sub-Group 5A2/3 which had considered that an increase in power of 0.7 dB would go some way towards solving the problem without having adverse effects on the Plan. He accordingly proposed that amendment to the Committee (entry 14 in Channel 1, 45 in Channel 5, 45 in Channel 9 and 46 in Channel 13).

The <u>Chairman of Working Group 5A</u> confirmed that that increase would lower the negative protection margin for Afghanistan to a figure which would be consistent with acceptable picture quality. In his opinion, there should be no effect on the Plan.

The delegate of the Netherlands supported the request.

It was so decided.

The <u>delegate of Pakistan</u> wished to record that the Conciliation Group had not been able to settle the case of spillover from Indian beam 0038B which affected large areas of Pakistan (entry 49 in Channel 23).

1

The <u>delegate of the Yemen Arab Republic</u> asked for correction of a protection margin which had been calculated erroneously.

The <u>delegate of the Democratic People's Republic of Korea</u> referred to an official note addressed to the Chairman of the IFRB regarding a problem which had now been solved with the cooperation of the delegation of China; he asked for the beam in question to be extended in reply to the increased coverage now being asked for by the South Korean authorities.

Document No. 380-E Page 3

The <u>Chairman of the IFRB</u> said that the delegation of the Democratic People's Republic of Korea, having approached Mr. Berrada and himself, had been given the results of calculations carried out by the IFRB to ascertain the effect of an increase in beam; the delegation was informed that the IFRB had no authority to make changes and that the decision was to be made by Committee 5.

The <u>delegate of the Democratic People's Republic of Korea</u> said that the only effect of the change in Channels 14, 16, 18, 20 and 22 (to increase the major axis to 1.3^o and the minor axis to 1.1^o) would be with regard to China.

The <u>delegate of China</u> confirmed that his delegation had agreed to the change.

The <u>delegate of the Republic of Korea</u> said, with regard to the North Korean delegation's proposal to enlarge its beam size, that such enlarged expansion of beam size based on polygon points beyond its service area would cause excessive spillover which was not recognized by the Conference and was not consistent with the provisions of No. 428A of the Radio Regulations. His delegation would not accept such an undue proposal insofar as it would cause undue harmful interference to the service area of the Republic of Korea. He also wished to record that the delegation of the Republic of Korea had never requested any polygon points beyond its territory and his delegation therefore totally rejected any unfounded and illegal allegations on that subject.

The <u>delegate of the Democratic People's Republic of Korea</u> said that his delegation was satisfied with the result reached by the Conciliation Group and saw no reason to change it.

The <u>Chairman</u> noted that the increases in the axes were accepted, on condition that the calculations confirmed that the resulting increased interference was only in the direction of China and the USSR, which had agreed to accept it.

The <u>representative of the IFRB</u> asked for a decision by the Committee with regard to Iraq which was not represented at the Conference. The greater part of that country was situated in Region 3 and the remainder in Region 1. At its request, it had been included in the European broadcasting area at the 1959 Conference and also in the LF/MF Plan.

In the absence of any objections, the <u>Chairman</u> declared that on the present occasion Committee 5 considered Iraq as belonging to Region 1.

The <u>delegate of Turkey</u> referred to correction of the beam DNK 0091 which did not appear to be a reduction. In reply the <u>delegate of Denmark</u> explained that it was to be read in conjunction with entry ISL0050, as there had been a re-arrangement of five channels with the same coverage between Denmark and Iceland.

The <u>delegate of Tunisia</u> asked the IFRB to draft a Note to entry 21 in Channel 38 to the effect that the channel could be exploited in common, with the proviso that Tunisia could use it otherwise. The <u>representative</u> of the IFRB said that that would be done. Document No. 380-E Page 4

The <u>delegate of India</u> wondered if the figure of - 8 dB for entry 57 in Channel 16 was a stray error. The <u>Chairman of Working Group 5A</u> said that the Group had felt the value was excessive but could not explain it. In other such cases studies had shown that the figures were due to errors and not to the quality of the Plan itself. The <u>representative of the IFRB</u> said that certain corrections to test points had been submitted and perhaps the case in point should be rectified; the problem would be re-examined the following morning.

In reply to a query by the <u>delegate of Switzerland</u> on the effect of the change of orientation of beam TUN150, the <u>representative</u> of the IFRB said it was not possible to alter the data submitted but it was accepted that certain modifications would have to be made.

In reply to the <u>delegate of Togo</u> who wondered whether slightly increased power would not improve the situation of - 2 dB margins in all Togo's five channels, the <u>Chairman of Working Group 5A</u> said that such points needed careful study and would have to be taken up on a bilateral basis. It was the opinion of the planning experts that a - 2 dB margin could not be regarded as serious interference, as picture quality would hardly be affected. In deriving the protection margins a number of worst case factors had been taken into account but it was obvious that in reality that situation would not occur. Further to a request by the Chairman of the Committee, the Chairman of the Working Group said he would try to produce a paper showing the correspondence between the picture quality scale and negative margins from - 1 to - 4 or - 5 dB.

The <u>delegate of the Netherlands</u> agreed that the terminology "negative margin" was misleading and reference should be to interference levels and margins of + 30 or + 29. On that understanding, it was evident that 1 dB more or less was an extremely small difference.

The delegate of Morocco said that the Algerian modification in Document No. DT/53 could not satisfy his delegation. He therefore maintained the reservation formulated that morning with regard to intentional coverage of Moroccan territory and asked for it to be entered in the Remarks column.

The <u>delegate of Ethiopia</u> said that his delegation was encountering problems with the beam of Somalia, which was not represented at the Conference. If the beam in question was included in the Plan, his own delegation would wish to include a Note stating that the assignment was to be coordinated with Ethiopia and any other country concerned.

The <u>delegate of Tunisia</u> pointed out that Somalia had submitted its requirements at the preparatory meeting organized by the Arab Broadcasting Union and the IFRB had stated that the Somali ellipse was technically acceptable.

The meeting rose at 2310 hours.

.1

The Secretary :

D. KANE

INTERNATIONAL TELECOMMUNICATION UNION

BROADCASTING SATELLITE CONFERENCE

Document No. 381-E 30 March 1977 Original : English

(Geneva, 1977)

COMMITTEE 5

SUMMARY RECORD

OF THE

ELEVENTH AND LAST MEETING OF COMMITTEE 5

(PLANNING)

Saturday, 12 February 1977, at 1630 hrs

Chairman : Mr. A. PETTI (Italy)

Subjects discussed :

1.

Document No.

 Examination of the draft Plan for Regions 1 and 3 (continued)

335 + Corr. 1 and 2, DL/63

2. Completion of the work of the Committee

Examination of the draft Plan for Regions 1 and 3 (contd.) (Documents Nos. 335, Corr. 1 and 2, DL/63)

The <u>Chairman</u> explained that some of the corrections indicated the previous evening were not included in the documents but would be read out by the Secretary; Document DL/63 contained the Notes requested by the Committee and the entries in the Plan had now been listed in alphabetical order in each channel.

The <u>Technical Secretary</u> added that the symbols AFS and RHS would be replaced throughout by the symbol IFB and the symbol DMH by BEN.

The <u>Chairman</u> proposed that the Committee examine the draft Plan page by page.



The <u>delegate of India</u> made the following statement :

"The Indian delegation has studied the draft Plan contained in Document No. 335 dated 11 February 1977 and notes with disappointment and regret that of the 48 channels assigned to India, 15 have negative protection margins although its own contribution of interference to the assigned channels of adjacent countries in Region 3 is relatively small. Of these 15 channels, 6 channels have margins of - 3 dB and one of - 6 dB. This position is obviously unsatisfactory to the Indian Administration.

The Indian delegation firmly believes that had more time been available to this Conference, a better Plan giving greater satisfaction to all Administrations could have been evolved. Considering that the guidelines for planning adopted by the Conference are sound, it should be possible to achieve further improvements in the Plan. The Indian delegation therefore suggests that the Conference entrust this task to the IFRB to examine this matter further with a view to making available its suggestions for improvement to the Administrations, at least six months before the entry into force of the Final Acts of the Conference."

Channel 2

The <u>Secretary</u> said that a recalculation of test points which had been incorrectly indicated for Saudi Arabia beam ARS 0275A affected the entry in Channel 2 as well as Channel 14.

The <u>delegate of the Republic of Korea</u>, referring to entry 47, proposed increasing the e.i.r.p. in column 9b to 64.6, as the negative protection margin had been altered from - 2 to - 3 dB.

The <u>delegate of China</u> said that increase would affect interference in the direction of China and was unacceptable.

The <u>Chairman of Working Group 5A</u> explained that certain sacrifices had had to be made in an endeavour to correct an error by the Sub-Group which had "lost" one of the USSR channels. The Japanese Administration had agreed to a power reduction and China had accepted an increase in negative margin and he therefore appealed to the Republic of Korea to assist with the problem, which was the fault of the Sub-Group.

The delegate of the Republic of Korea accepted that explanation.

The <u>Chairman of Working Group 5A</u> said that for entry 51 (MLI 0327A) the e.i.r.p. would be reduced by 1 dB with a consequential power adjustment; that result, which had been reached through the conciliation procedure, applied equally to Channels 6, 10, 14 and 18.

The IFRB agreed to make the necessary corrections to the interference calculations.

Channel 3

The <u>Chairman of Working Group 5A</u>, in reply to a query by the <u>delegate</u> <u>of Singapore</u> concerning entry 59, said that the negative margin was produced by beam VTN 0325A.

Channel 4

The <u>delegate of Austria</u> said that in entry 39 the first figure in column 4 should read 12.2 instead of 12.1.

The <u>Secretary</u> said that a correction would be made in every channel with beam ARS 0003A.

Channel 13

The <u>delegate of the USSR</u>, referring to entry 58, said he would submit in writing the fact that the USSR was prepared to reduce power by 0.5 dB in four channels to alleviate the negative protection margin of Afghanistan. The delegate of Afghanistan expressed his appreciation of that action.

Channel 14

The <u>delegate of Saudi Arabia</u> said that in compliance with the request of the delegation of Iran, his delegation would reduce the beam ARS 0275 and in a spirit of cooperation between the two countries, reduce the minor axis of the said beam to 1.2° instead of 1.4°, all other parameters remaining unchanged. The delegate of Iran expressed his gratitude for the Saudi Arabian initiative.

The <u>delegate of the Republic of Korea</u> read out corrected figures for entry 46 and asked what decision had been taken at the previous meeting with regard to the request by the delegation of the Democratic People's Republic of Korea.

The <u>Chairman</u> explained that the only objection to the request had come from the Republic of Korea so the Committee had decided in favour of the inclusion provided it did not cause increased interference to the Republic of Korea. The IFRB calculations had shown that there was no such interference. If the delegation of the Republic of Korea so wished, a Note could be inserted in the Remarks column stating its views.

The <u>delegate of the Republic of Korea</u> considered that the Committee's decision to authorize such an increase was a violation of the discussions in that Committee. In order not to prolong the debate unduly, he proposed that equal beams with equal orientation be adopted, i.e., that for the beam of the Republic of Korea the major axis be 1.3° and the minor, 1.1° .

The <u>delegate of the Democratic People's Republic of Korea</u> said that with the proposed increase the South Korean beam would cause interference over 50 % of the Northern part, including the capital.

The <u>Chairman of Working Group 5A</u> said that it was impossible to determine exactly what the effect of such an increase on the Plan would be.

The Chairman asked if there were objections to the proposal.

The <u>delegates of the People's Democratic Republic of Lao</u> and <u>China</u> were opposed to the increase.

The delegate of Japan said he had no objection.

The <u>Chairman</u> asked the IFRB, if possible, to examine the consequences of such an increase so that a decision on it could be taken in Plenary meeting.

Channel 15

The <u>Secretary</u>, replying to the <u>delegate of Indonesia</u>, said that the improved results were due to a reduction in the Australian beam.

Channel 22

The <u>delegate of Tunisia</u> said that the figure 153 should be corrected to 135 in column 6 of entry 53; the same remark applied to the other four channels in which the beam was listed.

The <u>Technical Secretary</u> said that some reductions in power had been made in Channels 22, 24, 25, 26, 28, 29, 30, 32, 33 and 3⁴ in order to alleviate high levels of interference to one particular country. The delegations of Norway, Sweden, Belgium, Finland and Denmark had also agreed to reduce power to improve the situation of the German Democratic Republic on four of its channels.

Channel 23

The <u>Technical Secretary</u> said that Note 3 should be added against entry 48; also in Channels 31 and 39 and for Denmark, in Channels 27 and 35.

Notes (Document No. DL/63)

Note 1

The <u>Technical Secretary</u> explained that the Note was required in certain cases because the Committee had decided not to have a column for ΔG .

Note 2

The <u>Technical Secretary</u> asked for information on the assignments, channel and beam number against which the Note should appear.

The <u>delegate of Saudi Arabia</u> indicated that for Indonesia it was entry 44 in Channel 19. The <u>delegate of Japan</u> said that for Malaysia it was entry 52 in Channel 16. The information for Iran would be provided later.

Note 3

To be inserted, as indicated by the Technical Secretary, during consideration of the draft Plan.

<u>Note 4</u>

The <u>Technical Secretary</u> said that the Note would appear against every entry for the symbols AFS and RHS.

Note 6

The <u>Technical Secretary</u> said that certain information had already been submitted. The <u>delegate of Tunisia</u> asked for the words "upon the agreement of those countries" to be added, with regard to entry 33 in Channel 38. The <u>delegate of Algeria</u> agreed to that addition to the Note.

Note 7

The <u>Technical Secretary</u> said that the necessary information was available to the Secretariat.

The <u>delegate of Iran</u> requested a Note 8 to read "This assignment will be used with linear polarization" to appear against entry 47 in Channels 3, 7 and 15, entry 46 in Channel 11 and entry 45 in Channel 19.

The <u>delegates of the USSR</u> and <u>France</u> expressed concern about the implications of such a Note. All the planning work had been based on the use of circular polarization and the use of linear polarization would increase interference by a considerable number of dB. The <u>delegate of New Zealand</u> added that the Plan had a very tight orbital spacing of 6° and much of the benefit would be lost if a large area of one country were to use linear polarization.

The <u>delegate of Iran</u> said that no objection had been raised when he had brought up the matter in Committee 4, nor when the documents had been approved on first reading in Plenary meeting. Document No. 224(Rev.1) had a Note to the effect that the Administration of Iran expressed a reservation about circular polarization and intended to use linear polarization.

The <u>Chairman</u> pointed out that when that document had been examined he himself had expressed doubts about the advisability of including such a reservation in the Final Acts of the Conference. He proposed that the point be studied further and an answer given at a later stage.

The <u>Technical Secretary</u> asked for instructions from the Committee to meet the request by several delegations that when the Plan was printed after the Conference, the figures in column 5 (antenna aperture) should be given to 2 decimal places.

It was so agreed.

2. Completion of the work of the Committee

The <u>Chairman of Committee 5</u> thanked all participants for their collaboration and their efforts permitting the preparation of a Plan for Regions 1 and 3. He was grateful in particular to the Chairmen of Working

Document No. 381-E Page 6

Groups, to the IFRB, including the Committee Secretary and the Technical Secretary, and to the interpreters and the entire secretariat.

The meeting rose at 1850 hours.

The Secretary :

.

D. KANE

The Chairman : A. PETTI

BROADCASTING SATELLITE CONFERENCE

(Geneva, 1977)

Document No. 382-E 17 March 1977 Original : English

PLENARY MEETING

MINUTES

OF THE

TENTH PLENARY MEETING

Thursday, 10 February 1977, at 0920 hrs

Chairman : Mr. Ib LØNBERG (Denmark)

Subjects discussed :

- 1. Approval of the minutes of the fifth plenary meeting
- 2. Texts submitted by the Editorial Committee for first reading (B.3(Rev.1), Add.1 to E.5, B.10)

Document No.

251

224(Rev.l) 231(Add.l) 246 (pages 12-15)

1. Approval of the minutes of the fifth plenary meeting (Document No. 251)

The <u>delegate of Japan</u> requested the insertion of the words "Supporting the views expressed by the delegate of the United Kingdom," before the statement by the delegate of Japan in the middle of page 5 of Document No. 251.

The <u>delegate of the United Kingdom</u> said it would be more appropriate to indicate that his statement at the foot of page 10, had been made in his capacity as Chairman of Working Group 5A.

The <u>delegate of Afghanistan</u> said that while he agreed with the summary of his statement, as given on page 10, he had asked for the text of his remarks to be annexed <u>in extenso</u> to the minutes.

The Chairman said that that would be done.

Subject to those amendments, the minutes of the fifth plenary meeting (Document No. 251) were <u>approved</u>.



 Texts submitted by the Editorial Committee for first reading (B.3(Rev.1), B.5(Add.1), B.10) (Documents Nos. 224(Rev.1), 231(Add.1), 246 (pages 12-15)).

The <u>Chairman of Committee 8</u> introduced the revised third series of texts submitted by the Editorial Committee (B.3(Rev.1) - Document No. 224(Rev.1)), the addendum to the fifth series of texts on the use of energy dispersal in the Broadcasting-Satellite Service (B.5(Add.1) - Document No. 231(Add.1), and the resolutions in Document No. 246, pages 12 - 15).

Document No. 224(Rev.1)

Title

The <u>Chairman of Committee 8</u> indicated that the title of Document No. 224(Rev.1) should be corrected to read "....should be used for the application of the Plan."

On the proposal of the <u>delegate of Canada</u>, it was <u>agreed</u> that the title should read : "Technical data used in establishing the provisions and associated Plan and which should be used for their application."

Section 2

On the proposal of the <u>Chairman of Committee 8</u>, it was <u>agreed</u> to delete the explanatory note in square brackets on page 4 of Document 224(Rev.1).

The <u>delegate of the United Kingdom</u>, supported by the <u>Chairman of</u> Committee 4, said that paragraph 2.2 did not deal with propagation factors and could be more appropriately inserted in a later section of the document. He therefore proposed its deletion from Section 2.

On the proposal of the <u>representative of the IFRB</u>, it was <u>agreed</u> that paragraph 2.2, which referred to Figure 1, should be retained in its present position, and that the United Kingdom delegate should draft an additional paragraph for insertion in Section 3.13.

On the proposal of the <u>representative of the CCIR</u>, it was <u>agreed</u> that Committee 8 should change the note at the foot of page 6 in accordance with the text of paragraph 2.2

The <u>delegate of Turkey</u> pointed out that the map on page 6 was not strictly in accordance with the tables of climatic zones given in Document No. DT/29, and after a brief discussion, it was <u>agreed</u> that the map should be replaced by the tables in question.

Section 3

It was <u>agreed</u> to delete the explanatory notes from the Editorial Committee which appeared in square brackets in paragraphs 3.1 and 3.3.

The <u>delegate of the United States of America</u> said that to avoid any possibility of confusion with the formal reservations for whose submission a dead-line was set, we would like the words "a reservation" to be replaced by the word "concern" in the footnote regarding the administration of the United States of America on page 7. The Chairman noted that amendment.

The <u>delegate of the United States of America</u> remarked that the values of the protection ratios given in paragraph 3.4.3 came from Committee 5, and not from Committee 4.

The <u>delegate of Canada</u> pointed out that Figure 4 showed a protection ratio of 30 dB, whereas paragraph 3.4.3 mentioned a protection ratio of 31 dB; he wondered how the apparent contradiction could be resolved.

The <u>delegate of the United Kingdom</u> proposed that the words "for the purpose of calculating protection margins" should be added after the words "have been adopted" in paragraph 3.4.3.

The <u>delegate of the United States of America</u> supported that proposal, while recalling that the purpose of the 1 dB addition to protection ratio values recommended by Committee 4, was to allow for ageing of the transmitting tubes.

The <u>representative of the IFRB</u> said that if the term "protection margins" was introduced, it would have to be defined in the Final Acts.

It was <u>agreed</u> that for the next reading of the text Committee 8 should produce a footnote to paragraph 3.4.3 defining "protection margins" on the basis of CCIR Report 633(Rev.76).

The amendment proposed by the United Kingdom delegate to paragraph 3.4.3 was adopted.

The <u>delegate of Australia</u>, supported by the <u>delegate of India</u>, said that paragraph 3.4.2 should be deleted, and the subject of energy dispersal should be treated in a separate paragraph with the heading "Use of energy dispersal" to be inserted as paragraph 3.20 of Document No. 224(Rev.1), and the text of which would be that of paragraph 3.9.4 of Document No. 224 (page 16).

The <u>Director of the CCIR</u> proposed the deletion of paragraph 3.4.2 including Figure 4.

The <u>delegate of the United Kingdom</u> supported that proposal and further suggested that the reference to Figure 4 in paragraph 3.4.1 should be deleted and that paragraph 3.4.3 should be renumbered 3.4.2.

It was so <u>decided</u>.

The <u>delegate of the United Kingdom</u>, referring to paragraph 3.5, said that the words "taken in the example below as 150 K" in line T_a under "where" should be deleted.

It was so <u>decided</u>.

The <u>delegate of Canada</u> proposed that the last line of paragraph 3.8 should read : "525 - line system (Region 2) : values of 18 and 23 MHz".

It was so <u>decided</u>.

Document No. 382-E Page 4

The <u>Chairman of Committee 5</u> said that the square brackets round "11 MHz" in paragraph 3.9 should be deleted.

The <u>delegate of Canada</u> proposed that the last two paragraphs on page 13 of Document No. 224 should be inserted below the Table in paragraph 3.9 and appropriately numbered by the Editorial Committee.

It was so decided.

The <u>delegate of the United States of America</u> wondered whether the entry for Region 2 in the second column of the Table was necessary as there were no space services adjacent to that band edge.

The <u>delegate of the United Kingdom</u> thought that a guard band at the upper edge of the 12.2/12.5 GHz band was necessary also to protect terrestrial services in the adjacent band.

The <u>delegate of France</u> said that the words "de la largeur" in the French text of the first sub-paragraph of paragraph 3.9 should be deleted.

The <u>delegate of Canada</u> proposed the addition of the words "in the Plan" to the heading of paragraph 3.10.

It was so decided.

The <u>Chairman of Committee 5</u> said that the square brackets round "19.18" should be removed.

The <u>delegate of the United Kingdom</u> proposed that the words "two adjacent" in the first line of paragraph 3.10 should be replaced by the words "the assigned frequencies of".

It was so decided.

The <u>delegate of the Federal Republic of Germany</u>, supported by the <u>delegate of the United Kingdom</u>, proposed the deletion of the words "is considered feasible for broadcasting satellites" in the third line of paragraph 3.12.

It was so decided.

The <u>delegate of the United Kingdom</u> suggested the addition of the following sentence at the end of the first sub-paragraph of paragraph 3.13 : "Attention is also directed to Section 2.2".

It was so <u>decided</u>.

The <u>delegate of Kenya</u> proposed that the words "up to" in the third line of the second sub-paragraph of paragraph 3.13 should be replaced by the words "at least" and that the words "or even higher" should be deleted.

It was so <u>decided</u>.

The <u>Director of the CGIR</u> proposed the addition of a new paragraph 3.14.4 entitled "Direction of rotation of circularly polarized waves" and containing the two definitions appearing in CCIR Report 321.

The <u>delegate of the United Kingdom</u> supported the proposal, but thought that sub-section 3.2 might be a more appropriate place.

The <u>Chairman</u> suggested that the Editorial Committee should be requested to insert the new paragraph under sub-section 3.2.

It was so decided.

The <u>delegate of the Federal Republic of Germany</u> observed that an indication should be provided to the effect that "direct" polarization meant right-hand and "indirect", left-hand polarization.

The <u>Chairman of Committee 5</u> said that such an indication had already been transmitted to the Editorial Committee with the list of characteristics given in the Plan.

The <u>delegate of Senegal</u> requested that the French text of the penultimate sub-paragraph of paragraph 3.14.1 (beginning with the words "a et b sont respectivement ...") should be reviewed by the Editorial Committee.

The <u>delegate of Venezuela</u> requested that Figure 7 in the Spanish text should be brought into line with that appearing in the English text.

It was so <u>decided</u>.

The <u>delegate of France</u> proposed that the title of paragraph 3.15 should be brought into line with the text of the paragraph, to read : "Spacing between two channels feeding a common antenna".

The <u>delegate of Australia</u> remarked that it would be more correct to speak of "assigned frequencies" than of "channels".

The <u>delegate of the United States of America</u> suggested that paragraphs 3.7, 3.10 and 3.15, all of which dealt with channel spacing, should be grouped together near the beginning of the Section.

The <u>Chairman</u> requested the Editorial Committee to rearrange those paragraphs accordingly, taking account of the points raised by the delegates of France and Australia with regard to paragraph 3.15.

The <u>delegate of France</u> remarked that the second and third factors listed in paragraph 3.16.2 were not, in fact, additional ones.

The <u>delegate of the United Kingdom</u> suggested that the word "additional" in the first line of paragraph 3.16.2 should be deleted and that the order of paragraphs 3.16.3 and 3.16.4 should be reversed.

It was so <u>decided</u>.

Document No. 382-E Page 6

The <u>delegate of the United States of America</u> proposed that the last line of paragraph 3.16.2 should read : "the effect of yaw error increases as the beam ellipse lengthens".

It was so decided.

The <u>Chairman</u> said that the Notes from the Editorial Committee in paragraphs 3.7, 3.9, 3.10, 3.18 and 3.19 should be deleted and that a new paragraph 3.20 should be added in accordance with the proposals made earlier in the meeting by the delegate of Australia.

The 3rd series of texts (B.3(Rev.1)) was <u>approved</u> on first reading, as amended and subject to editorial changes.

The <u>Chairman</u> invited delegates to consider the addendum to the 5th series of texts (B.5(Addendum)) (Document No. 231, Addendum No. 1).

The <u>Chairman of the Editorial Committee</u> wondered whether paragraph 3.4 should not be deleted as its subject matter was dealt with elsewhere.

The <u>delegate of the United Kingdom</u>, supported by the <u>delegate of</u> <u>the USSR</u>, urged that paragraph 3.4 should be retained.

The addendum to the 5th series of texts (B.5(Addendum)) was <u>approved</u> on first reading, subject to editorial changes.

The <u>Chairman</u> invited delegates to consider Resolutions Nos. D, E and F in the 10th series of texts (B.10) (Document No. 246).

Resolution No. D

The <u>delegate of Finland</u> proposed the insertion of the words "administrative radio" between the words "competent" and "conference" in the paragraph under "resolves".

It was so decided.

Resolution No. E

The <u>delegate of the United Kingdom</u> proposed the insertion of the word "the" between the words "concerning" and "broadcasting-satellite" in considering c), the insertion of letters "i.e." between "Radio Regulations" and "Nos." in resolves 1., and the replacement of the word "occupied" in the first and second lines of the first sub-paragraph of resolves 3. by the word "necessary".

The <u>delegate of Canada</u> proposed that the word "expounded" in considering b) should be replaced by the word "contained".

It was so decided.

The <u>Chairman of Committee 6</u> said that the square brackets in resolves 2.1 should be removed.

Resolution No. F

The <u>delegate of the United Kingdom</u> proposed the deletion of the first two lines of the title and of the square brackets round the remaining three lines of the title, as well as the replacement of the words "radiocommunication services" under "resolves" by the word "stations".

It was so <u>decided</u>.

The <u>delegate of the USSR</u> wondered whether the Resolution served any useful purpose.

The <u>Chairman of Committee 6</u> said that, following proposals made earlier in the Conference, Committee 6 at its last meeting had decided that Resolution No. F should go forward in order to make it absolutely clear that the use of any orbit other than the geostationary orbit would be incompatible with the Plan.

Resolutions Nos. D, E and F in the 10th series of texts (B.10) were approved on first reading, as amended and subject to editorial changes.

The meeting rose at 1240 hours.

The Secretary-General

M. MILI

The Chairman Ib LØNBERG

BROADCASTING SATELLITE CONFERENCE

Document No. 383-E 18 March 1977 Original : English/French

(Geneva, 1977)

PLENARY MEETING

MINUTES

OF THE

ELEVENTH PLENARY MEETING

Friday, 11 February 1977, at 1335 hrs

Chairman : Mr. Ib LØNBERG (Denmark)

Subjects discussed :

Document No.

257

255, 270

243(+ Corr.1, 2 and 3)

(except page 5), DL/61

296

- 1. Tribute to the memory of the late President of the Republic of India
- 2. Texts submitted by the Editorial Committee
 225(Add.2),

 for first reading (Add.2 to B.4, B.11,
 261, 268,

 B.12, B.14, B.15)
 297, 298
- 3. Texts submitted by the Editorial Committee for second reading (R.1)
- 4. Reports of Committee 5 to the Plenary Meeting
- 5. Documents submitted by Committee 4

6. 13th series of texts submitted by the Editorial Committee for first reading (B.13)

7. Credentials

8. Statement concerning South Africa and Rhodesia



Document No. 383-E Page 2

1. Tribute to the memory of the late President of the Republic of India

The <u>Chairman</u>, speaking on behalf of the Conference and on his own behalf, expressed condolences to the delegation of India on the occasion of the sudden death of Mr. Fakhruddin Ali Ahmed, President of the Republic of India.

The Plenary Meeting <u>observed a minute's silence</u> in memory of the late President.

2.

Texts submitted by the Editorial Committee for first reading (Add.2 to B.4, B.11, B.12, B.14, B.15) (Documents Nos. 225(Add.2), 261, 268, 297, 298)

Addendum No. 2 to the 4th Series of texts (Add.2 to B.4) (Document No. 225(Add.2))

The <u>Chairman of Committee 6</u> explained that the first option in the document had been supported by certain members of Ad Hoc Joint Working Group 4/6, but that, in revising Document No. 224, the Chairmen of Committees 4 and 6 had been unable to agree on the use of energy dispersal techniques in the preparation of the Plan. He had therefore proposed the second variant, which had still not satisfied the delegations concerned; in the light of the Plenary Meeting's decisions on the revision of the 3rd Series of texts (B.3(Rev.1), Document No. 224(Rev.1)) and of the fact that energy dispersal had been used in the preparation of the Plan and would be used in its application, the Plenary Meeting might consider that neither variant was necessary.

The <u>delegates of the USSR</u> and <u>the United Kingdom</u> said that, although they preferred the second variant, they did not consider it necessary to include the provision in the Final Acts.

The <u>delegate of India</u>, supported by the <u>delegate of Iran</u>, said that he, too, preferred the second variant, but considered that the option to employ energy dispersal should appear somewhere in the Final Acts.

_It_was <u>decided</u> to retain the second variant of Note 1) to Article $\frac{4}{4}$.

The <u>delegate of Australia</u> said that the paragraph in question should include a reference to the new paragraph 3.20 added at the end of the basic technical characteristics in the revised 3rd Series of texts (Document No. 224(Rev.1), page 21), rather than to paragraph 3.4 of the 5th Series of texts (Document No. 231).

Addendum No. 2 to the 4th Series of texts (Add.2 to B.4) was <u>approved</u> on first reading, as amended and subject to editorial changes.

11th Series of texts (B.11) (Document No. 261)

The <u>Chairman of the Editorial Committee</u> explained that the text had been drafted in the form of a Recommendation because it had been considered that it did not belong among the technical data. Perhaps the square brackets might be removed from the title. The llth Series of texts was <u>approved</u> on first reading, subject to editorial changes.

12th Series of texts (B.12) (Document No. 268)

The <u>Chairman of the Editorial Committee</u> said that all the texts related to arrangements for Region 2, except for the heading of the Plan columns on page 13, which were, of course, entirely separate.

Article / 10 / - Provisions governing the Broadcasting-Satellite Service in Region 2 pending the establishment of a detailed plan

The <u>Chairman of Committee 6</u> said that the square brackets round paragraph 5 could be removed, but that the following words should be added at the end of that paragraph : "and, where appropriate, with the provisions of Article / 7 of these Final Acts."

Article / 10 /, as amended, was approved.

Annexes A and B

Approved.

Recommendation No. HH

The <u>Chairman of the Editorial Committee</u> said that the references to "b), c), d) and e) below" in recommends 1 should be changed to "2, 3, 4 and 5 below".

The <u>delegate of Algeria</u> proposed that the reference in recommends 1 should read "2, 3, 4, 5 and 6 below".

Recommendation No. HH, as amended, was approved.

Resolutions Nos. G and H

Approved.

Headings of the Plan columns

The <u>delegate of the United Kingdom</u> proposed that the title be changed to "Column headings of the Plan", that the word "aperture" in heading 5 be replaced by "beamwidth" and that heading 8 be amended to read : "<u>E.i.r.p</u>. in the direction of maximum radiation in dbW."

The <u>delegate of Sweden</u> proposed that the words ", in degrees and tenths of a degree" be added at the end of heading 5.

In reply to a comment by the <u>delegate of the United Kingdom</u> concerning heading 7, the <u>Director of the CCIR</u> said that the CCIR used the terms "clockwise" and "anti-clockwise" rather than "direct" and "indirect" and suggested that the appropriate cross-references be made. Document No. 383-E

Page 4

The column headings of the Plan, as amended, were approved.

The 12th Series of texts (B.12) was <u>approved</u> on first reading, as amended and subject to editorial changes.

14th Series of texts (B.14) (Document No. 297)

Resolution No. I

<u>Approved</u>, with removal of square brackets round the word "equivalent" in considering b) and invites the IFRB e).

Table showing correspondence between channel numbers and assigned frequencies

Approved.

The 14th Series of texts (B.14) was <u>approved</u> on first reading, as amended and subject to editorial changes.

15th Series of texts (B.15) (Document No. 298)

The 15th Series of texts (B.15) was <u>approved</u> on first reading, subject to editorial amendments.

3. <u>Texts submitted by the Editorial Committee for second reading (R.1)</u> (Document No. 257)

The 1st Series of texts (R.1) was <u>approved</u> on second reading, subject to editorial amendments.

4. Reports of Committee 5 to the Plenary Meeting (Documents Nos. 255 and 270)

The <u>Chairman of Committee 5</u> introduced the two above-mentioned documents.

It was decided to take note of Documents Nos. 255 and 270.

5. <u>Documents submitted by Committee 4</u> (Documents Nos. 243, Corr.1, 2 and 3 (except for page 5) and DL/61)

Introducing Document No. 243, the <u>Chairman of Committee 4</u> said that, when considering point 2.3, reference should be made to Document No. DT/52 which contained the values calculated by the IFRB for the power flux-densities produced in Region 2 by space broadcasting stations in Regions 1 and 3.

Referring to the first page of Document No. 243, the <u>Chairman of</u> <u>Committee 6</u> asked whether the values given in paragraph 3 of Document No. 188(Rev.2) had been approved by Committee 5.

The <u>Chairman of Committee 5</u> said that they had and that the square brackets could be removed.

On the proposal of the <u>delegate of the USSR</u>, it was <u>decided</u> to study paragraph 3 during the consideration of Document No. 296.

The <u>Chairman</u> invited the delegates to consider the Annex to Document No. 243 point by point.

The <u>delegate of the United States of America</u> said that point 2.1 corresponded to Annex 4 to Document No. 296, point 2.2 to Article 9 on page 2 of that document and to Annex 5, and point 2.3 to Annex 6.

Point 2.1

At the request of the <u>delegate of France</u>, it was <u>decided</u> to replace the words "Fixed-Satellite Service" at the end of the first paragraph by "satellite services".

The <u>delegate of Denmark</u> said that the sign < in the second equation in the first paragraph should be replaced by the sign \leq .

It was so decided.

Point 2.1, as amended, was approved.

Point 2.2

The <u>delegate of the United States of America</u> said that the second paragraph (right-hand column) was incomplete compared with the text the Committee had approved. The words "and systems in the Fixed Services in Region 3" should be replaced by "and FDM/FM systems in the Fixed Service in Region 3".

Point 2.2 was approved subject to that amendment.

Point 2.3

The <u>delegate of Canada</u> said that the text was still in square brackets in Document No. 243 because, at the time it was prepared, Committee 4 did not have the values of the power flux-density in Region 2. Those values had been calculated subsequently by the IFRB and were given in Document No. DT/52 and in the addendum to that document.

The <u>delegate of the United States of America</u> said that his delegation and the Canadian delegation had drafted a text which was intended to replace point 2.3 in Document No. 243, namely, Document No. DL/61, which was based on the existing text of point 2.3, with certain amendments deriving from a proposal submitted by the delegation of Mauritania in Committee 5. Document No. DL/61 also referred to the method of calculation explained in Document No. DT/49 and to the values given in Document No. DT/52. The delegations of the United States of America and Canada hoped that, if approved, the new text of point 2.3 would be included in the Final Acts of the Conference.

The Chairman invited participants to consider Document No. DL/61.

Document No. 383-E Page 6

The <u>delegate of the United States of America</u> said that, at the end of the first two paragraphs, the word "Annex" should be replaced by "Document". In the third paragraph, the words "some time in the future" should be replaced by "some considerable time in the future".

At the request of the <u>delegate of Algeria</u>, it was <u>decided</u> to add the word "Consequently" at the beginning of the second sentence of the first paragraph.

The <u>delegate of Italy</u> said that the Committee had unanimously taken the view that the possibility of amending the Plan must be provided for. Committee 6 had accordingly defined the procedures to be followed for amending the Plan without upsetting the position of the various services operating in the band in question. The provisions set forth in Document No. DL/61 made no allowance for possible amendments to the Plan in that they stipulated that the values of the power flux-density must remain unchanged in all cases. In his view, what should be specified was the limit value of the radiated power fluxdensity which must not be exceeded. Furthermore, the words "without the consent of administrations which may be affected" should be inserted in the sixth line of the first paragraph after the word "modulation".

That proposal was supported by the <u>delegates of the Netherlands and</u> of Mauritania.

The <u>delegate</u> of <u>Canada</u> said that the limit value above which an administration must be consulted was given in Annex 1 on page 4 of Document No. 296.

The <u>delegate of Italy</u> said that he would revert to his proposal during the consideration of Document No. 296 if it was then decided to include in that document the text in Document No. DL/61.

The first paragraph was <u>approved</u> subject to the insertion proposed by the Algerian delegate.

Referring to the second paragraph of Document No. DL/61, the <u>representative of the IFRB</u> said that there were some errors in the values given in Document No. DT/52 which would be corrected later. The second paragraph might be deleted because it dealt with a question already referred to in the first paragraph.

The <u>delegate of the United States of America</u>, supported by the <u>delegate of Canada</u>, said that the second paragraph should be retained because it contained information which would be very useful for the South American countries. Those countries needed to know the exact values of the power flux-density they would have to take into account when planning their satellite-broadcasting systems.

The <u>delegate of Algeria</u>, supported by the <u>delegate of the</u> <u>Netherlands</u>, proposed that the word "theoretical" should be inserted before the word "values" in the second line of the paragraph.

Document No. 383-E Page 7

The <u>delegate of the United States of America</u> said that the values given in Document No. DT/52 could not be regarded as theoretical because they were those which would have to be used in systems operated in accordance with the Plan. Moreover, the insertion proposed by the delegate of Algeria would mean that similar amendments would have to be made in the case of all the values of power flux-density approved so far, which would give rise to practical difficulties.

The <u>delegates of New Zealand</u> and of <u>the United Kingdom</u> agreed with the delegate of the United States of America.

The second paragraph of the document was approved without amendment.

The <u>delegate of India</u> said that the last paragraph of Document No. DL/61 might be deleted because it amounted to a general recommendation.

The <u>delegates of Italy</u>, <u>Mauritania</u> and <u>the United Kingdom</u> supported that suggestion.

It was so decided.

It was <u>decided</u> to replace point 2.3 of Document No. 243 by the text in Document No. DL/61, subject to the insertion in the first paragraph proposed by the delegate of Algeria and the deletion of the last paragraph.

Point 2.3, as amended, was approved.

Point 2.4

The <u>delegate of the United States of America</u> pointed out that the sentence appearing in Corrigendum No. 1 to Document No. 243 should be added to point 2.4.

Point 2.4 was approved without amendment.

Document No. 243, as amended, was approved.

The <u>Chairman of Committee 4</u> thanked all those who had helped his Committee to complete its work for their valuable assistance.

6. <u>13th Series of texts submitted by the Editorial Committee for first</u> reading (B.13) (Document No. 296)

The <u>Chairman of Committee 6</u> said that the Annexes in the document might be rather controversial, because Ad Hoc Joint Working Group 4/6 had not received all the necessary technical criteria approved at Committee level for proper coordination. Although many of the square brackets could be removed as a result of the approval of Document No. 243 and Corrigenda, some points still remained outstanding, especially where Annex 1 was concerned.

Article 9 - Power flux-density limits between 11.7 and 12.2 GHz to protect Terrestrial Services in Regions 1 and 3 from interference from Region 2 broadcasting-satellite space stations

Approved.

Annex 1 - Limits for determining whether a service of an administration is considered to be affected by a proposed modification to the Plan (Article / 4_/, paragraph 4.3.1)

The <u>Chairman of Ad Hoc Joint Working Group 4/6</u> said that his Group had been hampered in its work by the lack of technical data referred to by the Chairman of Committee 6 and also by the fact that it had consisted of only about a dozen delegations, which could obviously not represent the ideas of the whole Conference. Although some discussions had been held with other delegations, the Group's conclusions must be regarded as a minority report.

Paragraph 1 - Limits on the change in the wanted-to-interfering signal ratio to protect the Broadcasting-Satellite Service of other administrations' frequency assignments in conformity with the Plan

The <u>delegate of the United Kingdom</u> proposed that the title be simplified to read "Limits on the change in the wanted-to-interfering signal with respect to frequency assignments in accordance with the Plan".

After a brief discussion on the wording of the second part of the paragraph, it was <u>agreed</u> that the French text be aligned on the English, in which the word "below" used for the second time in the fifth line would be deleted.

At the suggestion of the <u>representative of the IFRB</u>, it was <u>agreed</u> that the term "equivalent co-channel interfering signal" would be defined in the final version.

Paragraph 1, as amended, was approved.

<u>Paragraph 2</u> - <u>Limits on change in power flux-density to protect the</u> <u>Broadcasting-Satellite Service in the band 11.7-12.2 GHz in</u> Region 2

The <u>Chairman of Ad Hoc Joint Working Group 4/6</u> said that two variants had been proposed by the Group because the decisions concerning the protection of the Broadcasting-Satellite Service in Region 2 had not yet been known.

The <u>delegate</u> of the United States of America proposed that, since there was as yet no plan for Region 2, the first variant should be retained.

The <u>delegate of the USSR</u> observed that the figures in the first variant had not been agreed by Committee 4 and that it was not certain whether those figures could be met under the Plan agreed upon for Regions 1 and 3. He therefore preferred the second variant. The <u>delegate of the United Kingdom</u> pointed out that the question was not one of complying with the Plan, since the frequencies in the first variant were to be used only to assess whether coordination procedures would be needed in the event of modifications. In fact, those figures were all derived from data agreed upon in Committee 4 and were patterned on Annex 4, with slight amendments to take into account certain characteristics peculiar to Region 2. He therefore agreed with the United States delegate that the first variant should be retained.

The Chairman of Ad Hoc Joint Working Group 4/6 endorsed that statement.

The first variant in paragraph 2 was approved.

<u>Paragraph 3</u> - <u>Limits on the change in power flux-density to protect the</u> <u>Terrestrial Services of other administrations</u>

The Chairman of Ad Hoc Joint Working Group 4/6 said that the value in the second sub-paragraph had been omitted because Committee 4 had taken no decision in that respect.

The <u>representative of the IFRB</u> said that the Board interpreted paragraph 3 to mean that calculations must be made on the basis of coefficients of attenuation due to rain and other atmospheric factors corresponding to the countries for which the calculations were made, not the country responsible for the broadcasting-satellite station concerned. He therefore proposed the inclusion of a footnote to the title of the whole Annex, reading "1) The limits specified in this Annex relate to the power flux-densities which could be obtained assuming free space propagation conditions."

He also observed that it was not specified for what percentage of the time the calculations in question should be made, the 99 % provided for the Broadcasting-Satellite Service or the 99.9 % stipulated by the CCIR for the Terrestrial Service. In his opinion, pfd increases should be specified to the thousandth degree.

The <u>Chairman of Ad Hoc Joint Working Group 4/6</u> endorsed the preceding speaker's first statement but said that it would be too difficult to calculate very small percentages and that it would be better to keep to the figure provided for the Broadcasting-Satellite Service.

Speaking as the <u>delegate of Italy</u>, he said that for Terrestrial Services it would clearly be more practical not to consider the protection of each station, but the general protection of the territory of a country, since the number of stations was very large and certain countries did not intend to notify all their earth stations to the IFRB. There were two ways of bringing the number of countries with which coordination must be effected down to reasonable limits : the pfd margin in the first sub-paragraph could be made higher and the figure in the second sub-paragraph lower, so as to decrease the number of consultations, although the number of countries included in the consultations would be considerable; conversely, a low figure could be kept for the pfd margin, so that consultations would have to be held on most modifications to the Plan, but the value in the second-sub-paragraph could be increased, with Document No. 383-E Page 10

the result that in most cases the consultations would involve only a few neighbouring countries. His delegation believed the second alternative to be preferable and therefore proposed to retain the figure 0.25 dB in the first sub-paragraph and to insert in the second the same figures as for Region 2, $-125 \text{ dBW/m}^2/4 \text{ kHz}$ for circular polarization and $-128 \text{ dBW/m}^2/4 \text{ kHz}$ for linear polarization.

The delegate of the United Kingdom said that the figures for Regions 1 and 3 and for Region 2 could not be correlated on a 4 kHz basis. It might be best to use the figure - 132 dBW/m²/5 MHz, appearing in Annex 5, for both polarizations.

The <u>delegate of the United States of America</u> observed that, since many of the services to be protected in Regions 1 and 3 were channelled on a 4 kHz basis, it might be more practical also to use that basis for the power flux-density limit.

The <u>delegate of the United Kingdom</u> pointed out that many countries in Regions 1 and 3 would like to use their assignments in the band allocated to the Terrestrial, Fixed and Mobile Services for television, in which case 4 kHz would not be appropriate. He proposed that the words "the values expressed in Annex 5" be inserted in the blank space in the second sub-paragraph.

The <u>delegates of India</u>, <u>New Zealand</u> and <u>Ethiopia</u> supported that proposal.

The <u>delegate of Italy</u> said that that wording would make the Plan unduly rigid and would render the modification procedure practically inapplicable. His delegation's proposal, on the other hand, would limit consultations to neighbouring countries only.

The delegate of the Netherlands supported the Italian proposal.

After a brief discussion, the <u>Chairman</u> suggested that consideration of the paragraph be deferred until a later stage of the debate.

It was so agreed.

<u>Paragraph 4</u> - <u>Limits on the change in power flux-density to protect Fixed</u> Satellite Services in the band 11.7-12.2 GHz in Region 2

The <u>Chairman of Ad Hoc Joint Working Group 4/6</u> said that two variants were proposed for that paragraph and said that the pfd figures to be inserted should probably not differ greatly from those approved for paragraph 2.

The <u>delegate of Canada</u> said he could not agree with that statement, since the Fixed-Satellite Services in Region 2 were by nature closer to the Broadcasting-Satellite Services in Regions 1 and 3 than were the Terrestrial Services. His delegation was therefore in favour of the option in subparagraph a). Moreover, to meet the wish expressed by the Italian delegate during the discussion of Documents Nos. 243 and Corrigenda and DL/61 to accommodate minor changes to the Plan, he suggested that a reference to paragraph 4.3.1.3 be included in the first line of the paragraph. The <u>delegate of Italy</u> said that there was no need for such an additional reference, since paragraph 4.3.1.3 was already mentioned in paragraph 3 of Addendum 1. He had no objection to variant a) except for the figure 0.25 dB, approval of which would completely nullify all the work of Committee 6 by making it virtually impossible to introduce even slight modifications resulting from technical advances. The figure of 2 dB had been proposed in Ad Hoc Joint Working Group 4/6.

The <u>delegate of Canada</u> observed that, since Regions 1 and 3 had proposed the figure of 0.25 dB for the protection of their Terrestrial Services, it seemed logical to insert the same figure for triggering coordination procedures in Region 2.

The delegate of the USSR supported that view.

The <u>delegate of Mauritania</u> supported the Italian delegate's remarks. The figure 2 dB should appear in paragraphs 3 and 4.

The <u>delegate of Italy</u> reiterated that adoption of a pfd limit of 0.25 dB for modification purposes in paragraph 4 was likely to lead to a situation in which all the countries of Region 2 would have to be consulted on every change in the Plan, however slight. The position with regard to paragraph 3 was quite different : the Ad Hoc Group had agreed to suggest the figure of 0.25 dB in that case because its effect would be mitigated by a reasonable figure, such as the $-125 \text{ dBW/m}^2/4$ kHz he had proposed, in the second sub-paragraph, so that only the few countries immediately concerned would have to be consulted. Retention of the figure 0.25 dB for paragraph 4 would result in an anomalous situation where only a few of the countries of Regions 1 and 3 would have to be consulted concerning modifications affecting their assignments, while coordination would have to be effected with all the Region 2 countries, thus giving them what amounted to a veto over all modifications in Regions 1 and 3.

The <u>delegate of the United Kingdom</u>, supported by the <u>delegates of</u> <u>Canada</u>, <u>the USSR</u> and <u>the United States of America</u>, proposed that subparagraph a) be retained with the figure 0.25 dB and that another sub-paragraph be added, restating the second sub-paragraph of paragraph 3 with the value - 138 dBW/m² inserted at the end.

The <u>delegate of Italy</u>, supported by the <u>delegate of Mauritania</u>, said that adoption of those values would have the very effect he had described. Moreover, the value - 138 dBW/m^2 had been proposed in Committee ⁴ but had not been accepted.

The <u>delegate of the United States of America</u> said he thought that the Italian delegate's fears were exaggerated. The value - 138 dBW/m^2 would in fact exempt many modifications from the coordination procedure; it should also be borne in mind that modifications did not automatically lead to pfd increases.

After some further discussion, it was <u>agreed</u> to defer consideration of Annex 1 until a revised version of the document was issued. The <u>Chairman of Committee 5</u> proposed that a new paragraph 5 be added at the end of Annex 1, entitled "Conditions relating to the modulating signal", referring to point 3.1 of the basic technical characteristics as approved in the revised 3rd Series of texts and reproducing the second paragraph of that point 3.1 as it appeared on page 6 of the unrevised version (B.3, Document No. 224).

It was so decided.

<u>Annex 2</u> - <u>Basic characteristics to be furnished in notices relating to space</u> stations in the Broadcasting-Satellite Service

The representative of the IFRB pointed out that Ad Hoc Joint Working Group 4/6 had requested the Board to supply explanatory notes on the characteristics and on the manner in which they were to be notified. In point 12, the characteristics "isotropic gain" and " Δ G" were very similar, and it would be advisable to delete the reference to isotropic gain to avoid confusion.

In response to a comment by the <u>delegate of France</u>, it was <u>agreed</u> that that delegate should consult with the representative of the IFRB and the Chairman of Ad Hoc Joint Working Group 4/6 with a view to improving the definition of ΔG .

The <u>delegate of Australia</u> proposed that point 11 be amended to read "Power supplied to the antenna (watts)".

The delegate of the United Kingdom, referring to point 14, proposed that "TV standards" be replaced by "TV systems" and that the words "for reference frequency" after "frequency deviation" be deleted.

Annex 2, as amended, was <u>approved</u>.

Annex 4 - Need for coordination of a fixed-satellite space station or a broadcasting-satellite space station in Region 2 with respect to the Plan (Article 7)

The <u>Chairman</u> said that the reference in square brackets could be deleted.

The <u>delegate of the United Kingdom</u> proposed the deletion of the words "on the surface of the Earth" in the fourth line.

Annex 4, as amended, was approved.

Annex 5 - Power flux-density limits to protect the Terrestrial Services in Regions 1 and 3 from interference from Region 2 broadcasting-satellite space stations in the band 11.7-12.2 GHz (Article / _/)

The <u>Chairman of Ad Hoc Joint Working Group 4/6</u> said that the square brackets round the figure - 111 could be removed, in the light of decisions taken on Document No. 243.

The <u>delegate of Egypt</u> pointed out that the word "latitude" in the second line of paragraph 1) should read "longitude".

The <u>delegate of the United Kingdom</u> proposed that the order of paragraphs 1) and 2) be reversed and that the word "and" be added at the end of the new paragraph 1).

Annex 5, as amended, was approved.

Annex 6 - Orbital position limitations

The <u>Chairman of Ad Hoc Joint Working Group 6</u> said that the square brackets round the figure 37° W could now be removed and suggested that the figure 140° E be inserted at the end of paragraph 1). Since Article 4 was mentioned in Annex 6, a cross-reference to that Annex should be inserted in the Article.

The <u>delegate of the USSR</u> proposed that Annex 6 be deleted, since the approval of paragraphs 1 and 2 of Annex 1 seemed to make that text redundant.

The <u>delegate of the United States of America</u>, supported by the <u>delegates of Canada</u> and <u>the United Kingdom</u>, said he could not agree to that proposal, since Annex 6 was an essential provision for interregional orbit sharing and avoidance of interference between regions. The text of the Annex was derived from paragraph 3 on page 1 of Document No. 243 and had been agreed upon by all delegations, including that of the USSR; indeed, that country had proposed the figure of 140° E as a solution for the problem existing at the Pacific frontier between Regions 1 and 2, where the territories of the United States and the Soviet Union were very close and where the USSR had generously agreed to use frequencies above 12.2 GHz. The subject of Annex 6 was thus quite different from that of Annex 1, and it was important to retain it in the Final Acts.

The <u>delegate of the USSR</u> said that he would not press his proposal, although he would have thought that Annexes 2 and 4 provided all the necessary technical parameters for the coordination at issue. He would, however, propose the figure 146° E instead of 140° E, as a compensation for the concession already made by his country.

The <u>delegate of the United States of America</u> observed that 140° E had been originally proposed by the USSR delegation and had been confirmed by the Chairman of Committee 5 as the most easterly position in the Plan for Region 1. The United States had accepted that figure as a compromise, although it was aware that it was not far enough west to guarantee the absence of interference to Region 2 stations; in fact, his delegation would have greatly preferred the figure 136° E.

The <u>delegate of the USSR</u> said that further studies had caused his delegation to reconsider the figure 140° E that it had proposed. If the figure 146° E was not accepted, the USSR would be obliged also to reconsider its use of frequencies between 12.2 and 12.5 GHz in the area concerned. Document No. 383-E Page 14

The <u>Chairman</u> suggested that Annex 6 should be left in abeyance pending the issue of a revised version of the outstanding sections in the 13th Series of texts.

It was so decided.

The 13th Series of texts, with the exception of paragraphs 3 and 4 of Annex 1 and Annex 6, was <u>approved</u> on first reading, as amended and subject to editorial changes.

7. Credentials

The <u>Chairman of Committee 2</u> said that the delegations of Argentina and Mongolia had handed in their credentials which had been found to be in order. Both delegations could therefore take part in the voting and sign the Final Acts.

8. Statement concerning Rhodesia and the Republic of South Africa

Speaking on behalf of the Group of African Countries, the <u>delegate of</u> <u>Senegal</u> made a statement, which is annexed below, recommending that the names of the Administrations of South Africa and Rhodesia should not appear in the Plan.

The delegates of Nigeria, Zaire and Togo supported that statement.

It was <u>decided</u> that the names of the Administrations of South Africa and Rhodesia should not appear in the Plan although the corresponding assignments would be maintained and appear under appropriate symbols.

The meeting rose at 2000 hours.

The Secretary-General :

The Chairman : Ib LØNBERG

M. MILI

Annex : 1

Document No. 383-E Page 15

ANNEX

STATEMENT BY THE GROUP OF AFRICAN COUNTRIES CONCERNING RHODESIA AND THE REPUBLIC OF SOUTH AFRICA

Science and technology, however sophisticated, have no inherent moral value. Those who are responsible for their application may intend and take it upon themselves to place them at the service of civilization, to promote the material welfare and intellectual and moral advancement of nations; or they may turn them into powerful tools of physical, intellectual and moral discrimination and oppression. The application of science and technology by men professing opposing ideologies, with a different attitude to life and a different scale of values, leads to different and even opposite results.

If a regime, in the name of its constitution, deliberately flouts and contravenes the Universal Declaration of Human Rights in its economic and social development policies, it is the responsibility of the international community to recognize the fact and to take any action needed to bring such a dangerous enterprise to an end. In particular and in all circumstances the international community has the duty to withhold the powerful instruments of technology from regimes whose constitution is grounded in ethnic discrimination and whose national and international policies tolerate repression and aggression.

Such regimes actually exist in South Africa and Rhodesia.

The natural resources comprising the radiofrequency spectrum and the geostationary orbit cannot be used productively and beneficially for all nations unless they are the subject of multilateral or international agreement and coordination.

By virtue of Resolution No. 31 of the Plenipotentiary Conference (Malaga-Torremolinos, 1973), the Government of the Republic of South Africa was excluded from the Plenipotentiary Conference and from all the other conferences and meetings of the ITU.

It is not against the peoples of Rhodesia and South Africa that the measures taken by the United Nations are directed. In fact, these peoples, whatever their race, ethnic origin or religion, are increasingly united in opposing these regimes. The United Nations have taken and will continue to take the necessary steps to safeguard the present and future interests of these peoples, as they do for the people of Namibia under their trusteeship.

The ITU, in providing for the sharing of spectrum-orbit space, must take the rights of these peoples into account.

The Group of African Countries present at this Conference, drawing the attention of the honourable delegates to the discriminatory and oppressive nature of the Rhodesian and South African regimes, recalling the decisions adopted by the United Nations on these regimes, and mindful of the equal rights of all nations, large or small, to use spectrum-orbit space, requests of the Conference that :

Annex to Document No. 383-E Page 16

- the provisions and resolutions adopted by the Conference for the benefit of Member countries should not be applied to the Government of the Republic of South Africa;
- the Republic of South Africa and the Republic of Rhodesia should not be mentioned in the Final Acts or in the Plan.

INTERNATIONAL TELECOMMUNICATION UNION

BROADCASTING SATELLITE CONFERENCE

Document No. 384-E 22 March 1977 Original : English

(Geneva, 1977)

PLENARY MEETING

ARCHIVA

U.I.T. Genèn

MINUTES

OF THE

TWELFTH PLENARY MEETING

Saturday, 12 February 1977, at 0900 hrs

Chairman : Mr. Ib LØNBERG (Denmark)

Subjects discussed :		Documents Nos.
1.	New deadline for the submission of reservations	
2.	Texts submitted by the Editorial Committee for second reading (R.2, R.3 + Corr., R.4)	262, 302 + Corr., 306
3.	Protection of Terrestrial Services in Regions 1 and 3	247
4.	Final Protocol	269
5.	Texts submitted by the Editorial Committee for first reading (B.13(Rev.1) and Add.1)	296(Rev.l) + Add.l
6.	5th Series of texts submitted by the Editorial Committee for second reading (R.5)	351

1. New deadline for the submission of counter-reservations

The <u>Secretary-General</u> announced, in reply to a request by the <u>delegate of Spain</u>, that the deadline for submitting counter-reservations was postponed until midday that day, Saturday, 12 February 1977.

2. <u>Texts submitted by the Editorial Committee for second reading</u> second series (R.2) (Document No. 262)

Article 3

The reference to " / Annex ... 7 to" should be replaced by "Article 11

of".

For reasons of economy, this document is printed in a limited number. Participants are therefore kindly asked to bring their copi to the conference since only a few additional copies can be made available.
Article 4

The <u>Chairman of the Editorial Committee</u> said that there would be a Note relating to energy dispersal (new Note 1) and that existing Note 1 would be completed to indicate that modifications to orbital positions were dealt with in an Annex.

The <u>Chairman of Committee 6</u> assured the <u>delegate of Japan</u> that "frequency assignment" in the context took into consideration all the technical characteristics which appeared in the Plan.

Article 5

The square brackets in the title to be deleted.

Article 7

Sections II and III

The <u>Chairman of Committee 6</u> pointed out that reference to footnote 1 (and the note itself) in Section II and the asterisk and footnote to Section III should both be deleted, footnote 2 in Section II being renumbered accordingly.

Article 12

Square brackets in paragraph 1 to be deleted.

The 2nd series of texts were thus approved, as amended, on second reading.

Third Series (R.3) (Document No. 302 + Corr.)

Annex 3

The <u>delegate of Canada</u> said that the word "wanted" in paragraph 2.1 should be replaced by "necessary", in accordance with the earlier decision.

The <u>delegate of Australia</u> proposed the following additional sentence to paragraph 2.3.3 :

"This criterion takes no account of possible overload in the broadcasting-satellite receiver caused by signals from Fixed, Mobile and Broadcasting Services resulting in harmful interference".

After the explanation by the <u>United Kingdom delegate</u> that the overload situation was a national concern and not likely to arise in coordination between administrations, the <u>delegate of Australia</u> agreed to withdraw the proposal.

Annex 5

The <u>delegate of the United Kingdom</u> pointed out that the words "interfered with" in paragraph 1.3 should be replaced by "wanted". Annex $/7_7$

The <u>Chairman of the Editorial Committee</u> said that in the final text the Annex would be numbered 8.

She also pointed out that associated Tables 1 and 2 (beginning on page 19) were replaced by those given in the Corrigendum and that Table 3 (page 23) was to be deleted entirely.

Table 1

The <u>representative of the CCIR</u> said that the square brackets in the Corrigendum were to be deleted, as they concerned data already used by the IFRB in its calculations for the Plan. In some cases the figures concerned countries or territories for which there was no entry in the Plan, but they were included for the sake of completeness.

The delegate of China said that the symbol MAC should be deleted from the Country symbol column.

The <u>delegate of Portugal</u> said that the figure against the entry MDR should be 4 and not 5.

The <u>delegate of Mauritania</u>, objected to the inclusion in Table 1 of the symbol AOE which did not correspond to any distinct entity as far as the Conference was concerned. The former Spanish territory had been reunified partly within the Islamic Republic of Mauritania and partly within the Kingdom of Morocco, and its requests were reflected in those countries' national coverage.

The <u>delegate of Morocco</u> supported that statement; he added that the figures 4 and 5 against the entry for Morocco should be deleted and that he would provide the ITU at a later stage with more precise information.

The <u>delegate of Algeria</u> was strongly opposed to the deletion of the symbol AOE and did not consider the matter to fall within the scope of the present Conference.

The <u>delegate of Mauritania</u> said that the Algerian delegation was in no way concerned by the point he had raised. He proposed to request the Secretary-General to delete reference to the territory in question from all documents of the Union.

The <u>delegate of the United Kingdom</u> suggested the insertion of a Note such as that appearing in the Annex to Document No. 131.

At the <u>Chairman</u>'s suggestion, it was <u>agreed</u> to defer consideration of the entire Table until a later stage of the meeting. Document No. <u>384-E</u> Page 4

(The Chairman later proposed that as Table 1 had given rise to even more problems in the meantime, it should be deleted and the map on page 6 of Document No. 224 be reinstated.

It was so decided).

Table 2

The delegate of Algeria said that entry 0250 should be deleted.

The <u>delegate of the United Kingdom</u> and the <u>Chairman of the Editorial</u> <u>Committee</u> explained that the section on Basic technical characteristics (pages 32 - 36) should precede existing pages 24 - 31. The latter also said that the Editorial Committee had had difficulty finding a text to define the exact meaning of direct and indirect polarization, as used in the Plan.

The <u>Director of the CCIR</u> said that use of "direct" and "indirect" was unusual and the more conventional terms would be more readily understood by the user. He therefore suggested rewording the first sentence of paragraph 3.2.3 to read :

"3.2.3 The terms "direct" and "indirect" used in the Plan to indicate the direction of rotation of circularly polarized waves correspond to right-hand (clockwise) and left-hand (anti-clockwise) polarization respectively according to the following definitions :"

The titles should read :

"Direct polarization (right-hand or clockwise polarization)" and "Indirect polarization (left-hand or counter-clockwise polarization)".

The above amendment was <u>approved</u>, subject to the use of <u>anti</u>-clockwise in place of counter-clockwise.

The <u>Chairman of the Editorial Committee</u> said that the use of the word "déplacement" in the French text should be aligned to the English "field".

Paragraph 3.4

The <u>Chairman of the Editorial Committee</u> explained that Note 1) had been added after the first reading.

Paragraph 3.5.1

The <u>delegate of the United Kingdom</u> pointed out that an amendment approved on first reading did not appear in the document. The first sentence should read : "The spacing between the assigned frequencies of two adjacent ..."

Paragraph 3.5.3

Similarly, the title of paragraph 3.5.3 should read : "Spacing between two channels feeding a common antenna" and the text "... the spacing between the assigned frequencies of two channels ..."

The <u>Chairman of the Editorial Committee</u> noted that the French text was correct in both cases.

Paragraph 3.7.2

The <u>delegate of Canada</u> said that the second line should read : "... in Figures 3 and 4.".

Figure 3

The <u>delegate of Thailand</u>, requested that the text for curve A' be aligned to that of curves A and B.

Paragraph 3.8

The <u>delegate of Japan</u>, supported by the <u>delegate of the Republic of Korea</u>, asked for an addition to the second line so that it read :

"625-line systems, and 525-line systems in Region 3 : 27 MHz.".

Paragraph 3.10

The Chairman of Committee 5 said that the orbital spacing of 6° had not been adhered to strictly, so the sentence should read : "The Plan for Regions 1 and 3 had been based generally on nominal ...".

Figure 5

The <u>delegate of Venezuela</u> said that the Spanish version should be aligned to the English and French, the co-polar component curve and equation corresponding not to -25 but to -30 dB.

The <u>Chairman of the Editorial Committee</u> assured him that the original had been corrected.

The <u>delegate of Canada</u> said that the reference to the CCIR document should be deleted.

The third series of texts was approved, as amended, on second reading.

4th Series of texts (R.4) (Document No. 306)

Resolution No. A

The delegate of the United Kingdom said that the words "at the" in the considering paragraph should be replaced by "by this".

Approved, as amended.

Resolutions Nos. B, C and D

Approved.

Resolution No. E

The Chairman of the Editorial Committee said that the Annexes to be referred to in considering b) were 6 and 7.

Approved, as amended.

Resolution No. F and Recommendations Nos. AA, BB, CC, DD, EE and FF

Approved.

The 4th Series of texts (R.4) was approved on second reading, as amended and subject to editorial changes.

3. Protection of Terrestrial Services in Regions 1 and 3 (Document No. 247)

The delegate of the United Kingdom said that, in view of the short time available to the Conference, his delegation would withdraw Document No. 247, being sure that it could rely on No. 117 of the Radio Regulations for the proposed provisions to be implemented.

4. Final Protocol (Document No. 269)

The Chairman explained that the list of reservations entered within the fixed time-limit was being submitted to the Plenary Meeting so that it could take note thereof. Countries would be referred to by their full names in the final document.

In reply to the delegate of the United States of America, the Secretary-General said that further counter-reservations could be submitted later, subject to the approval of the Plenary Meeting.

The delegates of Tunisia, Algeria, Pakistan and Sudan said they wanted their countries' names to be included among the countries listed in Document No. 311. In reply to the <u>delegate of Venezuela</u>, the <u>Secretary-General</u> said that the text submitted by the Republic of Panama (Document No. 248) would appear in the Final Protocol.

The <u>delegates of Mauritania</u>, <u>Senegal</u>, <u>Mali</u>, <u>Afghanistan</u> and <u>Guinea</u> said that their countries' names should be included in the list in Document No. 313.

The <u>delegate of Egypt</u> said that the Arab Republic of Egypt should be added to the list of countries in Document No. 312.

The <u>delegate of Brazil</u> said that the reference "paragraph 1 of Document No. 204" in the second line of Document No. 332 should be replaced by "Section 1 of the Annex to Document No. 204".

The <u>delegate of the United Kingdom</u> said that in Document No. 349 the words "International Telecommunication Convention" should be replaced by "Final Acts".

5. <u>Texts submitted by the Editorial Committee for first reading (B.13(Rev.1)</u> <u>and Addendum</u>) (Document No. 296(Rev.1) and Addendum No. 1)

Addendum to the revised 13th Series of texts (B.13(Rev.1) (Addendum No. 1 to Document No. 296(Rev.1))

Article / 7 - Power flux-density limits between 11.7 and 12.2 GHz to protect space services in Region 2 from interference from Regions 1 and 3 broadcasting-satellite space stations

The <u>Chairman of the Editorial Committee</u> said that the new Article should be placed after the procedural Articles of the Agreement.

The <u>delegate of Canada</u> said that the text was intended to apply to the Plan at the time of its signature and to provide long-term protection for Region 2 space stations. The authors of the Article had, however, overlooked a provision on how the calculating methods should interrelate with the new Article; he suggested that a sentence be added, reading as follows : "These limits could be exceeded, subject to the agreement of the administrations concerned".

The <u>delegate of Italy</u> supported that addition and said that the text should be checked for possible contradictions with paragraphs 2 and 4 of Annex 1.

The <u>delegate of Mauritania</u> said he could not support the Canadian amendment. The fact that the Article reflected the situation at the time of the signature of the Final Acts should be stated explicitly in the text, which, in his opinion, should be an Annex, not an Article.

After some discussion, it was <u>decided</u> to set up a small ad hoc group to study the text.

Revised 13th Series of texts (B.13(Rev.1)) (Document No. 296(Rev.1))

<u>Annex 1</u> - <u>Limits for determining whether a service of an administration is</u> considered to be affected by a proposed modification to the Plan (Article 4, paragraph 4.3.1)

The Chairman observed that paragraphs 1 and 2 had already been approved.

<u>Paragraph 3</u> - <u>Limits on the change in power flux-density to protect the terrestrial</u> services of other administrations

The <u>delegate of the United Kingdom</u>, supported by the <u>delegates of</u> <u>New Zealand</u> and <u>Japan</u>, proposed that the square brackets round the value 0.25 dB in the first sub-paragraph be removed and that the words "the limits expressed in Annex 5" be inserted in the blank space at the end of the second sub-paragraph.

The <u>delegate of Italy</u> said that he would not oppose the United Kingdom proposal at that late stage, although he was sure that the proposed figures would make coordination of modifications to the Plan very difficult.

Paragraph 3, as amended was approved.

Paragraph 4 - Limits on the change in power flux-density to protect the Fixed-Satellite Service in the band 11.7 - 12.2 GHz in Region 2

The <u>delegate of Canada</u> proposed that all the square brackets be removed from paragraph 4 and that the reference in the first line should read "paragraphs 4.3.1.3 and 4.3.1.4".

The <u>delegate of Italy</u> and the <u>Chairman of Committee 6</u> said they considered the additional reference to be inappropriate.

The <u>delegate of the United Kingdom</u> drew attention to the proposal he had made at the preceding meeting, namely, to include at the end of paragraph 4 the wording of the second sub-paragraph of paragraph 3, but with the value - 138 dBW/m² inserted at the end.

The <u>delegate of the USSR</u> supported that proposal and suggested that the Ad Hoc Group set up to consider Addendum No. 1 should also examine paragraph 4.

It was so decided.

Paragraph 5 - Conditions relating to the modulating signal

The <u>delegate of Japan</u> suggested that the text should be either deleted or extended to cover other technical parameters, since small satellites that might be launched in the near future would probably not be able to conform with the requirements of the Plan.

Document No. <u>384-E</u> Page 9

The <u>Chairman of Committee 6</u> said that, under the procedural provisions already approved, there would be no problem if the power flux-density limits were respected; if they were exceeded, the procedures for modification of the Plan would come into operation. There was therefore no need to delete or amend the text, which, in his opinion, did not seem to belong in Annex 1.

The <u>Chairman of Committee 5</u> said that the text had been transferred from the basic technical characteristics as a result of decisions taken on the revised 4th Series of texts.

Paragraph 5 was <u>approved</u>, on the understanding that it would be returned to the Annex containing the basic technical characteristics.

Annex $/\overline{6}$ / - Orbital position limitations

The <u>delegate of the United States of America</u> said that, after some discussion, his delegation had decided to accept for the purposes of the Final Acts the figure of 146° E proposed by the USSR delegation at the preceding meeting.

Annex /67, as amended, was approved.

The meeting was suspended at 1300 hours and resumed at 1515 hours.

The delegate of Canada, who had convened the Ad Hoc Group on Article / / and paragraph 4 of Annex 1, said that the Group had decided to recommend that the text in Addendum No. 1 to the revised 13th Series (B.13(Rev.1)) should appear as Article 10 of the Final Acts, with the insertion of the words "prior to any modification of the Plan" after "band 11.7 - 12.2 GHz" in the fifth line of the first paragraph and after "(longitude 35° W, latitude 8° S)" in the second line of the second paragraph and insertion of the words "on the date of its entry into force" after "in accordance with the Plan" in the seventh line of the first paragraph.

Addendum No. 1 to the revised 13th Series of texts (Addendum No. 1 to B.13(Rev.1)) was <u>approved</u> on first reading, as amended and subject to editorial changes.

The <u>delegate of Canada</u> said that the majority of the Ad Hoc Group had decided to recommend that the introductory paragraph of paragraph 4 of Annex 1 should be immediately followed by the text of sub-paragraph a), which should end with the words "Final Acts". The square brackets round the value 0.25 dB should be removed and a second paragraph, based on the second sub-paragraph of paragraph 3 of Annex 1, should be added, worded as follows : "Where an assignment in the Plan or its subsequent modification gives a power flux-density less than - 138 dBW/m² anywhere in the territory of an administration of Region 2, that administration shall be considered as not affected".

After a discussion during which the <u>delegate of Mauritania</u> asked for clarifications of the new second paragraph, the English text of paragraph 4, as amended, was <u>approved</u>, on the understanding that a suitable French version would be agreed upon.

The revised 13th Series of texts (B.13(Rev.1)) was <u>approved</u> on first reading, as amended and subject to editorial changes.

6.

5th Series of texts submitted by the Editorial Committee for second reading (R.5) (Document No. 351)

Recommendation No. GG and Corrigendum to Article 4

Approved.

Annex 2

<u>Approved</u>, on the understanding that the IFRB would provide more detailed information after the Conference.

Annexes 4 and 5 and Article 9

Approved.

Article 11

The <u>Chairman of the Editorial Committee</u> said that the reference in the first line should be to Annex 6.

The <u>delegate of the United States of America</u> said that in paragraph 5 the word "and" before "particularly" should be deleted and a comma should be added after "Radio Regulations".

Article 11, as amended, was approved.

Final Protocol, Annexes 6 and 7, Recommendation HH, Resolutions Nos. G, H and I and Column headings of the Plan

Approved.

Table showing correspondence between channel numbers and assigned frequencies

The <u>delegate of India</u> proposed that the title be changed to "Table showing channel numbers and corresponding assigned frequencies". That change would also affect heading 3 of the Column headings of the Plan.

The Table, as amended, was approved.

The 5th Series of texts (R.6) was <u>approved</u> on second reading, as amended and subject to editorial changes.

The meeting rose at 1615 hours.

The Secretary-General :

The Chairman :

M. MILI

ID LØNBERG

INTERNATIONAL TELECOMMUNICATION UNION

BROADCASTING SATELLITE CONFERENCE

Document No. 385-E 23 March 1977 Original : English

(Geneva, 1977)

PLENARY MEETING

MINUTES

OF THE

THIRTEENTH PLENARY MEETING

Saturday, 12 February 1977, at 2120 hrs and Sunday, 13 February 1977

Chairman : Mr. Ib LØNBERG (Denmark)

Subjects discussed :		Document Nos.
1.	Minutes of the Sixth and Eighth Plenary Meetings	293, 352
2.	Final Protocol	363
3.	Texts submitted by the Editorial Committee for second reading (R.6, R.7)	370, 373
4.	Approval of the draft Plan for Regions 1 and 3	335 + Corr. 1 and 2, 368

1. <u>Minutes of the Sixth and Eighth Plenary Meetings (Documents Nos. 293,</u> <u>352</u>)

Approved.

2. Final Protocol (Document No. 363)

The <u>delegate of Japan</u> said that the figures 281 and 331 should be substituted for 340 and 344 in Document No. 359.

The Plenary Meeting took note of Document No. 363.



Document No. 385-E

Page 2

3. <u>Texts submitted by the Editorial Committee for second reading (R.6,</u> R.7) (Documents Nos. 370, 373)

6th Series of texts (R.6) (Document No. 370)

The <u>Chairman of the Editorial Committee</u> pointed out that two versions of the method of calculating the power flux-density produced in the territories of Region 2 by space stations in the Broadcasting-Satellite Service in Regions 1 and 3 had inadvertently been included in the document and suggested that the Meeting should disregard the version on page 2, which had not been examined by the Editorial Committee, and should begin its consideration with Annex 1 on page 7, go through the document and then return to the Results and Annex on pages 2 to 6.

Annex 1

Paragraphs 1 and 2

Approved, with editorial changes.

Paragraph 3

The <u>Director of the CCIR</u> suggested that the words "for all angles of arrival" in the third line of the third sub-paragraph be replaced by "for any angle of arrival".

Approved, as amended.

Paragraph 4

The <u>delegate of France</u> proposed that the word "However," be added at the beginning of the second sub-paragraph, to make it clear that the two subparagraphs were not consequential upon each other.

Approved, as amended.

Annex 1, as amended, was approved.

Annex 10

The <u>delegate of Canada</u> said that the word "range" in the second line of paragraph 1) should be replaced by "band".

Approved, as amended.

Article 10

The <u>Chairman of the Editorial Committee</u> said that the reference in the third line should be to Figure 6 and that the reference in the last line should be to Annex 11.

Approved, as amended.

Document No. <u>385-E</u> Page 3

Annex 11

The <u>delegate of the United States of America</u> proposed that the units used in the method of calculation be inserted in the appropriate places.

The <u>delegate of France</u> proposed that the formula in the second indentation in paragraph 3 be changed to "A = $10 \log \frac{1}{dB(1/m^2)}$ ".

It was <u>agreed</u> that the CCIR and the IFRB would make the necessary changes.

In reply to a request for clarification of paragraph 2 by the <u>delegate</u> of <u>Mauritania</u>, the <u>delegate of Brazil</u> proposed that the words "the most easterly point of Brazil" be replaced by the coordinates "longitude 35[°] W, latitude 8[°] S" in that paragraph and in the Results paragraph on page 2.

Annex 11, as amended, was approved.

Results paragraph and Annexed Table

The <u>Vice-Chairman of the IFRB</u> said that the figures at the top of the columns on page 4 should be -37.0, -31.0, -25.0 and -19.0 instead of -13.0, -7.0, -1.0 and +5.0.

The <u>delegate of Algeria</u> pointed out that on page 3, under nominal orbit position - 25.0, his country's second beam had been omitted.

The <u>Vice-Chairman of the IFRB</u> said that the necessary correction would be made.

In reply to queries by the <u>delegates of the Vatican City State</u> and <u>Tunisia</u> concerning the omission of entries for their countries in channels 27, 31, 35, 38 and 39, the <u>delegate of Canada</u> drew attention to the passage in the Results paragraph which specified that the table related only to channels 1 to 25.

The Results paragraph and the annexed table, as amended, were approved.

The 6th Series of texts was <u>approved</u> on second reading, as amended and subject to editorial changes.

7th Series of texts (R.7) (Document No. 373)

Notes to the Plan

The <u>Chairman of Committee 5</u> said that the delegation of Iran had asked in Committee 5 that another note be inserted, concerning the use of linear polarization by Iran's broadcasting-satellite service. Other delegations had objected to such a note, pointing out that the Plan was based on circular polarization. The Iranian delegation had asked for time to reflect on the matter. The delegate of Iran made the following statement :

"The Conference is well aware of the importance attached by our delegation to the adoption of linear polarization for our future broadcastingsatellite services.

We have expressed our views on the advantages of linear polarization in the Technical Committee and have therefore requested this mode of polarization to be recorded in the new column 15 of Document No. 335.

It still remains our wish to use linear polarization. However, in the spirit of international goodwill and of the necessity of the success of this Conference, the Iranian delegation will not press the inclusion of that reference in Document No. 335."

The <u>Deputy Secretary-General</u> said that the word "allocated" in the English text of Note 5 should read "dedicated".

The 7th Series of texts (R.7) was <u>approved</u> on first and second readings, as amended and subject to editorial changes.

4. <u>Approval of the draft Plan for Regions 1 and 3</u> (Document No. 335 + Corr. 1 and 2)

The <u>Chairman of Committee 5</u>, at the request of the <u>delegate of India</u>, drew attention to that delegation's disappointment that the planning work had resulted in negative protection margins (in one case of $-6 \, \text{dB}$) for 14 of that country's 48 frequency assignments. The delegation considered that, given more time, some improvements might be made and was requesting the Plenary meeting to authorize the IFRB to carry out studies with that in view.

The Chairman of the Committee also put to the Plenary meeting the question of the enlarged ellipse of the Democratic People's Republic of Korea, which had been accepted as calculations had shown that interference would be increased only in the direction of China, which had given its agreement; however, the Republic of Korea had disputed that increase and no solution had been found in the Committee.

Satisfactory solutions had been found to several other problems that had arisen in reading the Plan, and certain reductions of power were indicated in Corrigendum No. 2. The Committee had decided to increase by 0.7 dB the power of the channels allocated to Afghanistan, but that did not appear in any document.

The delegate of India made the statement annexed hereto.

The <u>Chairman of Committee 5</u> drew attention to Document No. 368 which gave the relationship between protection margins and quality of service and stressed that even a negative margin of -3 dB in fact represented "good" picture quality.

The <u>Chairman of Working Group 5A</u> repeated that the concept of positive and negative margins had been developed for planning purposes but in reality

the reference was to interference ratios, and under no circumstances could margins of up to - 3 dB be considered harmful. The single case of - 6 dB had been studied carefully but without solution; on the other hand, in one case India had a record positive margin of + 9 dB.

In his view some slight adjustments in power levels might be made to improve the situation and the IFRB could facilitate bilateral discussions. Administrations, however, could do much for themselves, by recalculating tests points, producing higher receiving dish performance, etc.

The <u>delegate of India</u> thought it would have been useful if the abovementioned document had been circulated earlier to make the position clearer but still requested the Plenary's support for his proposal.

The <u>delegate of New Zealand</u> said he would be prepared to support the proposal, but only on condition that no other assignments were affected, or that the agreement of all Administrations concerned was sought.

The delegate of Iran supported the Indian proposal.

The <u>delegate of Indonesia</u> also endorsed the proposal, though recognizing that it was impossible to satisfy all requirements. He also would like a recalculation to check that the orbital positions requested could not be allocated to Indonesia.

The <u>delegate of the United Kingdom</u> said that the Indian request amounted to a modification procedure. The idea behind the modification procedure already agreed was that it not be undertaken unless entry into service was planned within five years. He wondered whether India was intending to use all 48 channels immediately and if not, why it could not use the most favourable ones first. He felt that to consider modifications at the present stage was contrary to the spirit and letter of the agreement reached.

The <u>delegate of India</u> agreed that none of the channels were for immediate use but thought that the IFRB might study possible refinements while the planning criteria were familiar.

The <u>delegate of Mexico</u> considered the Indian proposal reasonable but had some doubts as to the form; if improvements were to be introduced into the Plan (which formed part of the Final Acts), No. 531 of the Convention would not be respected.

The <u>Chairman</u> assumed that any proposed change, following such a study by the IFRB, would have to be made in accordance with the modification procedures adopted in the Final Acts.

The <u>delegate of the United Kingdom</u> doubted whether further refinements were possible without calling another Conference and he was concerned about giving the IFRB authority to improve a Plan which was the subject of a signed agreement.

The <u>Chairman</u> asked the Indian delegation whether it was prepared to request assistance from the IFRB in accordance with usual procedures.

The <u>delegate of India</u> said he realized that Member Administrations could consult the IFRB at any time; he had wished for approval by the Plenary so that it would carry greater weight.

The <u>Chairman</u> noted only partial support for the proposal, as it was highly unlikely that no other countries would be affected. Of course, in such a planning Conference requirements were fulfilled to a varying degree but he thought the maximum possible effort had been made. He took it that there were no objections to the IFRB being asked to study the matter and to make its findings available. The Indian Administration could then use that information at the proper time in accordance with the agreed modification procedure.

The delegate of India said he could agree with that suggestion.

It was therefore <u>decided</u> that the Plenary should request the IFRB to carry out the studies and make the findings available to all parties concerned.

In reply to questions by the <u>delegate of Singapore</u>, the <u>Chairman of</u> <u>Committee 5</u> explained that Note 1 was not marked against any country in Corrigendum No. 2 to Document No. 335 because it was intended to cover the special case of the choice of ΔG for small countries, to be agreed upon with the IFRB. It did not have to be included in the Plan at that stage.

The <u>delegate of the Republic of Korea</u> asked what conclusion the IFRB had reached concerning his country's counter-declaration to the declaration of the Democratic People's Republic of Korea to the effect that it intended to increase its beamwidth and to move the centre point of its beam 1° southward. The effect would be that the beam of the Democratic People's Republic of Korea would cover nearly all of his country's territory.

The <u>Chairman of the IFRB</u> said he had not understood that any request to the Board for the relevant calculations had emerged from the debate on the subject in Committee 5.

The <u>Chairman of Committee 5</u> pointed out that at the last meeting of the Committee the delegation of the Republic of Korea had requested calculation of the effects on the Plan of the enlargement of its beam-width to correspond to the enlargement obtained by the Democratic People's Republic of Korea. Perhaps the IFRB had not had time to carry out the necessary calculations.

The <u>Chairman of the IFRB</u> reiterated that he had not understood the Board to have been given any such instructions.

The <u>delegate of the Democratic People's Republic of Korea</u> said that according to the calculations of his delegation the coverage of his country's service area by the orbital position of the Republic of Korea would also be practically complete. The best course would be to leave matters as they stood and to try to reach optimal results at the 1979 WARC.

The <u>delegate of the Republic of Korea</u> regretted that the IFRB had not seen fit to carry out calculations which would have made it possible to solve the problem at the current Conference.

After further discussion, the <u>Chairman</u> proposed that the IFRB should be asked to study the problem after the Conference and that the Board's findings should be used in accordance with the Plan modification procedures, if necessary.

It was so decided.

The <u>Chairman</u> then invited the meeting to examine the draft Plan channel by channel and drew attention to Corrigendum No. 2.

The <u>Technical Secretary</u> explained that the above Corrigendum contained amendments adopted in Committee 5, in abridged form.

The <u>delegate of Afghanistan</u> said that for beam AFG 246 the value in column 8 should be increased to 63.4 in entry 1 of channels 1, 5, 9 and 13.

The <u>delegate of Austria</u> said that for AUT 016 the second value in column 4 should read 12.2 and not 12.1 in entry 39 of channels 4, 8 and 16, entry 38 of channel 12 and entry 34 of channel 20.

The <u>delegate of Saudi Arabia</u> said that for ARS 340 the second value in column 5 should be changed to 0.7 in entry 34 of channel 23.

The <u>delegate of Denmark</u> said that Note 3 should appear in column 9 for DNK 091 in entry 24 of channels 27 and 35.

The draft Plan for Regions 1 and 3 (Document No. 335 and Corr. 1 and 2) was <u>approved</u>, on first reading, as amended.

The draft Plan for Regions 1 and 3 was then <u>approved</u> on second reading.

The <u>delegate of Japan</u> made the following statement :

"The Japanese delegation would like to clarify its position with regard to the maps, the polygon and the checkpoints which appear in all documents or information in connection with the establishment of the Plan for the Broadcasting-Satellite Service in the 12 GHz band.

My Administration has drawn the maps and the polygon and has decided on the checkpoints in such a manner as to obtain the most efficient and reasonable system for Japan, regardless of the exact boundary of its territory, taking into account the technical and economic aspects.

Therefore, on behalf of the Japanese Government, my delegation would like to make it clear that the maps and the polygon drawn by my Administration and the checkpoints decided upon by the Administration have, as clearly stated in paragraph 5 of IFRB Circular-letter No. 367 dated 1 November 1976, no bearing on the territorial delimitation of Japan."

The <u>delegate of the United States of America</u> made the following statements :

1. "In signing the Final Acts of this Conference, the delegation of the United States of America wishes to express agreement with the statement made by

the delegation of the Federative Republic of Brazil with regard to the possible impact of the Plan adopted for Regions 1 and 3 on the development of space services in the frequency band 11.7 - 12.2 GHz in Region 2."

2. "While supporting adoption of the Final Acts of this Conference, the United States does not regard decisions taken by the Conference on technical matters within its proper scope to be determinative of the resolution of related issues currently before the United Nations Outer Space Committee, particularly its Legal Sub-Committee. The United States reserves the right to deal with the related issues in the political fora of the United Nations, as appropriate."

The <u>delegate of the Federal Republic of Germany</u> associated his delegation with the second statement of the United States delegation.

The <u>delegate of Japan</u> said that his Administration would sign the Final Acts of the Conference subject to the approval of the Japanese Government.

The meeting rose at 0145 hours on Sunday, 13 February.

The Secretary-General :

M. MILI

The Chairman : Ib LØNBERG

<u>Annex</u> : 1

ANNEX

STATEMENT BY THE DELEGATION OF INDIA

The Indian Delegation has studied the draft Plan contained in Document No. 335 dated 11 February 1977 and notes with disappointment and regret that of the 48 channels assigned to India, 15 have negative protection margins although its own contribution of interference to the assigned channels of adjacent countries in Region 3 is relatively small. Of these 15 channels, 6 channels have margins of - 3 dB and one of - 6 dB. This position is obviously unsatisfactory to the Indian Administration.

The Indian Delegation firmly believes that had more time been available to this Conference a better plan giving greater satisfaction to all Administrations could have been evolved. Considering that the guidelines for planning adopted by the Conference are sound, it should be possible to achieve further improvements in the Plan. The Indian Delegation therefore suggests that the Conference entrust this task to the IFRB to examine this matter further with a view to making available its suggestions for improvement to the Administrations, at least six months before the entry into force of the Final Acts of the Conference.

INTERNATIONAL TELECOMMUNICATION UNION

BROADCASTING SATELLITE CONFERENCE

(Geneva, 1977)

Document No. 386-E 23 March 1977 Original : English

PLENARY MEETING

GENE

SIGNATURE OF THE FINAL ACTS

Sunday, 13 February 1977, at 0930 hrs

Chairman : Mr. Ib LØNBERG (Denmark)

1. Signature of the Final Acts

2. Closing of the Conference

1. <u>Signature of the Final Acts</u>

The <u>Deputy Secretary-General</u> explained the procedure for signature of the Final Acts and then called the roll of those delegations whose credentials had been found in order.

The delegations signed the Final Acts and the Final Protocol as indicated in Annex 1.

2. <u>Closing of the Conference</u>

can't win them all."

The Chairman made the statement reproduced in Annex 2.

The Deputy Secretary-General made the statement reproduced in Annex 3.

The <u>delegate of the United States of America</u> made the following statement :

"Our Captain has piloted his craft through very troublesome skies after sounding many warnings on the perils of the trip and offering coffee, tea or milk. He has piloted his passengers to safety on what I consider to be a very successful voyage indeed. Typical of world conferences, we all arrived with high hopes and aspirations; we depart somewhat subdued but with stiffened sinews and girded loins for another encounter in 1979. We in the United States have had a perennial presidential candidate named Harold Stassen. After about his eighth successive defeat, when asked for his comments, he said "Well, you

Would the 1979 conference be blessed with the calibre of leadership that our captain has so ably demonstrated here. He speaks slowly but carries a big stick.

rage a

In times like this one searches for superlatives. In this case they are in great abundance. "Superb" is the word I found to describe a combination of leadership, patience, endurance, skill and indeed something close to divine guidance to lead us out of the jungle from whence we sought a path. His good judgment is exceeded only by his patience.

I have no doubt that all will share my wish for a long, useful and happy life to the pilot of our ship.

And so, Mr. Chairman, adieu. May God be with you - but not too soon."

The <u>delegate of Mexico</u>, in paying tribute to the Chairman's able guidance of the Conference, recalled his predecessor at the 1971 Space Conference, Mr. Gunnar Pedersen, who was well-remembered for his outstanding qualities.

He wished in particular to stress a point mentioned by the Deputy Secretary-General, namely that despite the prolonged debates the necessity of voting had been avoided. He also noted the reference to the necessity for collaboration between all the permanent organs of the Union, a vital factor for the success of such a Conference. In conclusion, he thanked the Chairman warmly and wished him every personal success.

The <u>delegate of Iran</u> said it was a privilege for him to extend sincere congratulations to the Chairman on behalf of the Asian delegations. His own delegation was particularly grateful for the Chairman's understanding of certain questions which arose and which were solved thanks to his efforts. He also wished to thank all those who had assisted the Chairman in his task and made the work of the Conference pleasant as well as profitable, and associated himself with the gratitude expressed for the tireless efforts and cooperation received from the ITU Secretariat. He had been greatly assisted in the task entrusted to him and thanked all delegations.

The <u>delegate of the Federal Republic of Germany</u> said that the present Conference was unique in that it opened new horizons in many respects. The burden of chairing such a Conference was a heavy one and he could only admire the Chairman for his skill. He also wished to thank and congratulate all who had contributed to the success of the Conference and especially the planning groups whose efforts had been so well rewarded.

The <u>delegate of India</u> said that the outcome of the Conference was of great satisfaction to his Administration, particularly as India had been one of the first countries to embark on an experimental broadcasting satellite project. They looked forward to more extensive use of such a powerful and high-credibility medium as an instrument of social and economic development. In that respect, he was expressing the views of all developing nations in the world.

The Chairman had been largely responsible for lifting the Conference out of its mood of despondency and many awkward moments had been eased by his forbearance and sense of humour. The delegate of India was therefore happy to associate himself with the expressions of gratitude and also with the thanks to all those who had contributed to the successful outcome. The <u>delegate of Nigeria</u> joined in the appreciation expressed of the Chairman's able guidance of a very difficult Conference and also of the efforts of the Secretariat and of all delegations.

The <u>Chairman of Committee 3</u> wished to pay tribute to the Chairman's handling of the affairs of the Steering Committee.

The <u>delegate of Egypt</u> congratulated the Chairman on behalf of the Arab and Islamic countries, and the CCIR, the IFRB and all those who had taken part in the work of the Conference. The successful result was largely due to the Chairman himself and he wished him a long, happy and successful life.

The <u>delegate of the United States of America</u> paid tribute to the Chairman of the Working Group which had produced the Plan, and without whose skill and leadership success would not have been possible; he considered that in recognition of his services, the outcome of the Conference should be known as the Temple Plan.

The <u>delegate of India</u> congratulated the special Vice-Chairman for conciliation activities, Mr. Fadami of Iran, who had so ably handled such questions at the LF/MF Broadcasting Conference.

The <u>delegate of Mauritania</u>, in the absence of the Chairman of the African Group, extended sincere congratulations to the Chairman for his remarkable conduct of the meetings, and his personal contribution to the progress of the Conference.

The <u>delegate of Zaire</u> added his congratulations on the Chairman's tenacity and patience, and also thanked all the organs of the ITU for their assistance.

The <u>delegate of Algeria</u> asked the Chairman to pass to his wife the congratulations and appreciation of all delegations.

The <u>Chairman</u> thanked all those who had spoken in appreciation of his services and wished all participants a safe return to their countries.

The <u>Chairman</u> then declared closed the World Broadcasting-Satellite Administrative Radio Conference, Geneva 1977.

The meeting rose at 1100 hours.

The Secretary-General :

M. MILI

The Chairman : Ib LØNBERG

Annexes : 3

ANNEXI

The following delegations signed the Final Acts and Final Protocol :

Republic of Afghanistan, Algerian Democratic and Popular Republic, Federal Republic of Germany, Kingdom of Saudi Arabia, Argentine Republic, Australia, Austria, State of Bahrain, People's Republic of Bangladesh, Belgium, People's Republic of Benin, Byelorussian Soviet Socialist Republic, Republic of Bolivia, Federative Republic of Brazil, People's Republic of Bulgaria, Republic of Burundi, Canada, Central African Empire, Chile, People's Republic of China, Republic of Cyprus, Vatican City State, Republic of Colombia, State of the Comoros, People's Republic of the Congo, Republic of Korea, Republic of Ivory Coast, Cuba, Denmark, Arab Republic of Egypt, United Arab Emirates, Ecuador, Spain, United States of America, Ethiopia, Finland, France, Gabon Republic, Ghana, Greece, Guatemala, Republic of Guinea, Republic of Haiti, Republic of Upper Volta, Hungarian People's Republic, Republic of India, Republic of Indonesia, Iran, Ireland, Iceland, Italy, Japan, Republic of Kenya, State of Kuwait, People's Democratic Republic of Lao, Lebanon, Libyan Arab Republic, Principality of Liechtenstein, Luxembourg, Democratic Republic of Madagascar, Malaysia, Republic of Mali, Republic of Malta, Kingdom of Morocco, Mauritius, Islamic Republic of Mauritania, Mexico, Monaco, Mongolian People's Republic, Federal Republic of Nigeria, Norway, New Zealand, Sultanate of Oman, Republic of Uganda, Pakistan, Republic of Panama, Papua New Guinea, Republic of Paraguay, Kingdom of the Netherlands, Republic of the Philippines, People's Republic of Poland, Portugal, German Democratic Republic, Democratic People's Republic of Korea, Ukrainian Soviet Socialist Republic, Socialist Republic of Roumania, United Kingdom of Great Britain and Northern Ireland, Republic of Senegal, Republic of Singapore, Democratic Republic of the Sudan, Sweden, Confederation of Switzerland, United Republic of Tanzania, Republic of the Chad, Czechoslovak Socialist Republic, Thailand, Togolese Republic, Tunisia, Turkey, Union of Soviet Socialist Republics, Oriental Republic of Uruguay, Republic of Venezuela, Yemen Arab Republic, People's Democratic Republic of Yemen, Socialist Federal Republic of Yugoslavia, Republic of Zaire.

ANNEX 2

STATEMENT BY MR. IB LØNBERG, CHAIRMAN OF THE BROADCASTING-SATELLITE CONFERENCE, AT THE CLOSING SESSION

Ladies and Gentlemen,

We have now arrived at the end of our five-week conference devoted to the broadcasting of the future. It was obvious to all of us that the conference would encounter difficulties because never has an ITU conference been asked to accomplish such a complicated task in such a short time. You have succeeded in accomplishing the nearly impossible, and you have therefore every reason to be proud of the result of the conference.

I think after such a conference it is important to draw out specific features. I must first express appreciation of the amount of the technical preparation made by the CCIR during recent years; some more points had to be defined during the conference but they were solved due mainly to the existence of the CCIR texts.

A second element of this conference was the great importance of the computer and data processing. As some of you will remember it was in Stockholm in 1961 that the computer was used for the first time for the analysis of a plan. This present conference could not even have been dreamt of without the assistance of the computer.

In this connection I would like to pay special tribute to the delegation of France for thier generous gesture in putting at our disposal the use of the TDF Computer. On behalf of the conference I would ask the French delegation to convey our gratitude to the President and Director of the TDF.

But still it is the human factor which was the basis for the success of this conference. I must pay tribute to the spirit of cooperation and the goodwill which were continuously present during the duration of the conference. I consider this attitude especially worthy of praise because, in addition to the normal subjects of discussion, there were the different conditions prevailing in the various continents.

Once more the structure of our long-established Union has proved to be flexible enough to permit a conference to fulfil its mandate. 112 years after its creation the ITU has just laid the foundation stone of satellite broadcasting. The conclusions of this conference will have important consequences, not only in the Union itself, but also in other international concerns interested in specific aspects of satellite broadcasting.

I should like to thank those people responsible for the success of the conference and in particular the Vice-Chairmen of the conference and the Chairmen and Vice-Chairmen of committees. They have had a tremendous task and I am grateful to them for their good advice and strong support. Especially I would mention the enormous work carried out by the Chairman of the Editorial Committee.

Annex 2 to Document No. 386-E Page 6

I must also thank the Chairmen of the Working Groups and the delegates who spent night after night preparing acceptable texts for us. And I would especially like to pay tribute to the very few doing the real planning work - they have been the real nucleus of the conference.

I also wish to thank the Secretary-General of the ITU, the Deputy Secretary-General, the Director of the CCIR, the Chairman of the IFRB and the IFRB Members for their invaluable help and guidance.

Even more than on previous occasions, I feel I have to pay tribute to the people whose work ensured the material success of the conference. I think first of the Executive Secretary and the Technical Secretary and also of the officer responsible for the delegates service, the interpretation and the documentation; the engineers and all the staff of the IFRB and of the Computer Division who worked day and night for us, and also the CCIR engineers. It goes without saying that I also thank all the administrative services : the interpreters who helped us to understand each other, the minute-writers, the Document Service and the Order of the Day staff. I also want to mention in particular all the people who worked very hard behind the scenes for this conference : the translators, the Typing Pool, the Reproduction and Distribution Services and also the messengers and the staff of this Conference Centre.

I would ask you, Mr. Deputy Secretary-General, to transmit to them my deepest thanks and appreciation for all they have done for the conference.

I wish you all a safe and speedy return to your countries.

ANNEX 3

STATEMENT BY THE DEPUTY SECRETARY-GENERAL

The signature of the Final Acts which has just been completed brings to a close another very important ITU Administrative Conference which has been of very special interest to the Broadcasting Services. It has been devoted to the drawing up of detailed plans and associated policy considerations and procedures to permit the future development of radiocommunication services not merely for the Broadcasting and Television Services, but also for the other service interests - and especially the future use of the Fixed Satellite and the Terrestrial Services in the 11.7 GHz-12.2 GHz (Regions 2 and 3) -12.5 GHz (Region 1) bands concerned.

As the Conference comes to a close, there are two ways in which we can look at the achievements.

First, we can refer to the detailed side of telecommunication policy and frequency planning; or second to the much broader international, human and social long term consequences for nations and their people at large.

Just because we have been so deeply engrossed in the telecommunication policy and frequency planning, I feel that a brief look towards the end product of the result of the Conference is appropriate.

Here at this Conference, we cannot but have only a simple view of all that the policy achievements, discussions and negotiations will mean. It is not merely the form of the investment in the equipment and the development of the detailed national plans to which there has been some allusion vis-a-vis the sharing of the spectrum concerned for the space radiocommunication service and the orderly development (and relationships) with the Terrestrial Services during the next 17 years or so.

The many millions of future users of the frequencies which you have planned here will know very little of how this Conference made possible policy agreements which you have reached, resolving also by new international telecommunication law some principles which elsewhere in the international and the United Nations family have been difficult to solve, if not impracticable. It needs no emphasis to state that without these telecommunication solutions, orderly programme transmissions and receptions could not take place.

Your results as government representatives will be of special interest to the United Nations and its Peaceful Uses Outer Space Committee. Many of you well know that the United Nations organs have been studying the benefits and implications of satellite broadcasting. They have been evolving various principles for the use of the new medium. Some principles had indeed been already incorporated in the various international telecommunication policy legislation in 1971 and 1973. Their practical application has now been augmented in a very substantial measure through the detailed planning agreements of your Conference. Here, I speak of considerations of equality of all nations and the avoidance of spill over to the maximum technical extent practicable consistent with the art of present technology.

Annex 3 to Document No. 386-E Page 8

Thus, the opportunity now exists for the orderly evolution of space technology for satellite broadcasting when required by nations in accordance with the Final Acts in which you have suitably inscribed a world agreement.

The formalities by the Members of the Union in the establishment of the agenda and the preparatory directives to the organs of the Union were arranged through the Plenipotentiary Conference and the Administrative Council as far back as 1973. However, I cannot help but recall our contacts with various national authorities and regional organizations when it became evident that much more seemed to be needed to assure adequate national and international preparation to bring to successful fruition a conference of this nature. Hence, our concern and practical arrangements for the special preparatory seminars.

I would like to express our appreciation for the world wide collaboration that prevailed throughout the various regions in those seminars, and for the frank and open discussions that took place between the many experts of the Administrations and international organizations concerned. Those preparations enabled everyone to move forward to the collective understanding and good will which has dominated the work of this Conference. The fact that all this was achieved without a formal vote speaks for itself.

At the Conference, the Union owes its gratitude to the many experts in planning and procedures who were aided by the various computer facilities, including not only our own computer, but those of the United Nations family in Geneva and of TéléDiffusion de France (TDF), which made such a significant contribution to the planning results, and in this connection we would like to pay special tribute to Mr. Sauvet-Goichon.

We thank you, Mr. Chairman, for the forebearance with us, and wish you all a pleasant return to your countries with a proud reflection of what has been made possible here for the millions of people who will have the opportunity of benefiting from the planned development of all the Broadcasting, Television and Radiocommunication Services that will result, as and when each individual country takes its national decision for investment in the facilities that could now be operated with the necessary international safeguards in the future.

"Made possible" I think, is a concept which you, as the representatives of the Members of the Union, can justly be proud of, because this is really what this Conference has done to permit the development of a new range of space and terrestrial operations of which the users and nations will reap the benefits for many years to come.

Finally, Mr. Chairman, on behalf of Mr. Mili, the Secretariat and of various colleagues in the Headquarters of the Union, I would like to thank you warmly for your remarks on the contribution that we have endeavoured to make in the services to the Members.

CONFERENCE DE RADIODIFFUSION PAR SATELLITE

Document N^O 387-F/E/S Genève, le 14 février 1977

(Genève, 1977)

21

LISTE DES PARTICIPANTS

LIST OF PARTICIPANTS

LISTA DE PARTICIPANTES

- Cette liste comprend les sections suivantes This list includes the following sections -Esta lista comprende las secciones siguientes
 - 1. Délégations Delegations Delegaciones
 - 2. Organisations internationales International Organizations Organizaciones Internacionales
 - 2.1 Nations Unies United Nations Naciones Unidas
 - 2.2 Institutions spécialisées Specialized Agencies Instituciones especializadas
 - 2.3 Organisations régionales (Art. 32 de la Convention) Regional Organizations (Art. 32 of the Convention) - Organizaciones regionales (Art. 32 del Convenio)
 - 2.4 Autres Organisations Others Organizations Otras Organizaciones
 - 3. Siège de l'Union Union Headquarters Sede de la Unión
 - 4. Secrétariat de la Conférence Secretariat of the Conference Secretaria de la Conferencia
- II. Les symboles suivants sont utilisés The following symbols are used Se utilizan los símbolos siguientes
 - C : Chef de délégation Head of delegation Jefe de delegación
 - CA : Chef adjoint de délégation Deputy Head of delegation Jefe adjunto de delegación
 - D : Délégué Delegate Delegado
 - A : Conseiller Adviser Consejero
 - 0 : Observateur Observer Observador
 - S : Membre du secrétariat de la délégation Member of delegation secretariat -Miembro de la secretaría de la delegación
 - At : Suppléant Alternate Suplente
 - + : Assistant de la délégation Assistant to the delegation Asistente de la delegación
 - * : Interprète Interpreter Intérprete

Pour des raisons d'économie, ce document n'a été tiré qu'en nombre restreint. Les participants sont donc priés de bien vouloir apporter à la conférence leurs documents avec eux, car il n'y aura que fort peu d'exemplaires supplémentaires disponibles.

1. DELEGATIONS - DELEGATIONS - DELEGACIONES

Afghanistan (Rép. d') - Afghanistan (Rep. of) - Afganistán (Rep. de)

C M. KAMRAN Khaliq Dad Head of the Technical Department Radio Afghanistan, Kabul

> Algérie (Rép. Algérienne Démocratique et Populaire) - Algeria (Algerian Democratic and Popular Republic) - Argelia (Rep. Argelina Democrática y Popular)

- C M. BOUHIRED Noureddine Conseiller technique Ministère des Postes et Télécommunications Secrétaire Général du Comité de Coordination des Télécommunications, Alger
- CA M. BAIRI Abdelkader Directeur des Equipements des Télécommunications - Ministère des Postes et Télécommunications, Alger
- CA M. ZERROUKI B. Directeur des Services techniques et de l'Equipement Radiodiffusion Télévision Algérienne, Alger
- D M. AGHA M. Chef du Département Equipement Radiodiffusion Télévision Algérienne, Alger
- D M. AHMED-BEY Bachir Ingénieur d'Etat, Chargé d'Etudes Ministère des Postes et Télécommunications Alger
- D M. ALLAN Tahar Ingénieur d'Etat, Chargé d'Etudes Ministère des Postes et Télécommunications Alger
- D M. BEDRANE Belkacem
 Sous-Directeur de la Réglementation et des Contrôles
 Ministère de l'Intérieur, Alger
- D M. BENACER Tahar
 Chef de Division
 Ministère des Postes et Télécommunications
 Alger

Algérie (Rép. Algérienne Démocratique et Populaire) - Algeria (Algerian Democratic and Popular Republic) - Argelia (Rep. Argelina Democrática y Popular) (suite)

- D M. BOUNAB R. Chef du Service F.H. Radiodiffusion Télévision Algérienne Alger
- D M. DERRAGUI M.
 Chef de Section au Département Equipement Radiodiffusion Télévision Algérienne Alger
- D M. HARBI M. Chef du Service Etudes Radiodiffusion Télévision Algérienne Alger
- D M. KADI M'Hammed Attaché Mission d'Algérie, Genève

Allemagne (Rép. féd. d') - Germany (Fed. Rep. of) - Alemania (Rep. Fed. de)

- C M. VENHAUS Heinrich Ludwig Ministerialdirigent Deutsche Bundespost Bundesministerium für das Post- und Fernmeldewesen, Bonn
- CA M. BINZ K.R. Ministerialrat Deutsche Bundespost Bundesministerium für das Post- und Fernmeldewesen, Bonn
- CA M. KRATH Herbert Ministerialrat Deutsche Bundespost Bundesministerium für das Post- und Fernmeldewesen, Bonn
- M. BLOHMER Erwin
 Assistent des Technischen Direktors
 Zweites Deutsches Fernsehen, Mainz
- D M. EDEN Hermann Fachbereichsleiter Institut für Rundfunktechnik, München
- D M. GEORGE Eberhard Postoberrat, Deutsche Bundespost Fernmeldetechnisches Zentralamt Darmstadt

<u>Allemagne (Rép. féd. d') - Germany</u> (Fed. Rep. of) - Alemania (Rep. Fed. de) (suite)

- D M. GRÖSCHEL Günther Wissenschaftlicher Mitarbeiter Deutsche Bundespost Fernmeldetechnisches Zentralamt, Darmstadt
- D M. IMELMANN Ehrhardt
 Direktor Programmbetrieb und Sendeleitung
 Deutschlandfunk, Köln
- D M. JANSEN Bernhard Oberamtsrat Deutsche Bundespost Bundesministerium für das Post- und Fernmeldewesen, Bonn
- D M. KAISER RudolfTechnischer DirektorZweites Deutsches Fernsehen, Mainz
- D Mme KRAMER Gabriele Diplom-Übersetzerin Deutsche Bundespost Bundesministerium für das Post- und Fernmeldewesen, Bonn
- D M. KRAUSE Albrecht Ministerialrat Bundesministerium des Innern, Bonn
- D Dr LINSMAYER E. (Mme)
 Botschaftsrätin
 Ständige Vertretung der Bundesrepublik
 Deutschland bei dem Büro der Vereinten
 Nationen und den anderen internationalen
 Organisationen, Genève
- D M. MÄGELE Manfred Postrat Deutsche Bundespost Fernmeldetechnisches Zentralamt, Darmstadt
- D M. MALLAU Heiko Oberpostdirektor, Deutsche Bundespost Fernmeldetechnisches Zentralamt Darmstadt
- D M. MÜLLER-RÖMER Frank Technischer Direktor, Bayerischer Rundfunk Bayerische Staatskanzlei, München

<u>Allemagne (Rép. féd. d') - Germany</u> (Fed. Rep. of) - Alemania (Rep. Fed. de) (suite)

- M. OLMS Klaus
 Postoberrat
 Deutsche Bundespost Fernmeldetechnisches
 Zentralamt, Darmstadt
- D Dr v. PLOETZ H.F. Legationsrat I. Kl. Auswärtiges Amt, Bonn
- D M. REJZEK Günther Technischer Fernmeldeamtsrat Deutsche Bundespost Fernmeldetechnisches Zentralamt Darmstadt
- D M. ROESSLER Guenter Technischer Direktor Deutsche Welle, Köln
- D M. SCHARF Albert Juristischer Direktor Bayerischer Rundfunk, München
- D M. SCHULT H.W. Oberamtsrat Deutsche Bundespost Bundesministerium für das Post- und Fernmeldewesen, Bonn
- D M. SEIDEL Günther Postdirektor Deutsche Bundespost Bundesministerium für das Post- und Fernmeldewesen, Bonn
- D M. SEIDELMANN Oskar
 Abteilungspräsident
 Deutsche Bundespost
 Fernmeldetechnisches Zentralamt
 Darmstadt
- D M. STUCKERT Klaus-Peter Regierungsdirektor Senator für Bundesangelegenheiten, Berlin (West)
- D M. SÜVERKRÜBBE Rolf Arbeitsbereichsleiter Institut für Rundfunktechnik, München

<u>Allemagne (Rép. féd. d') - Germany</u> (Fed. Rep. of) - Alemania (Rep. Fed. de) (suite)

- A Prof. BOHNKE K.F. Erich Technischer Direktor Sender Freies Berlin, Berlin (West)
- A Dr FUHR Ernst Wolfgang Justitiar Zweites Deutsches Fernsehen, Mainz
- A M. KRIEGER Horst A.C. Technischer Direktor Norddeutscher Rundfunk, Hamburg

Arabie Saoudite (Royaume de l') -Saudi Arabia (Kingdom of) - Arabia Saudita (Reino de)

- C M. OBAID Ibrahim A. Deputy Minister of PTT Ministry of PTT, Riyadh
- C M. SWAILEM Abdulmohsen Mohamed General Director of Engineering Department, Ministry of Information Riyadh
- CA M. DAGHISTANI Abdulrahm A. Director of Wireless Communication Ministry of Posts, Telephone and Telegraphs Riyadh
- D M. AL-ASSAF Hamad Assistant Supervisor for Projects Dept. Saudi Arabia Ministry of Information Riyadh
- D M. AL-MULHEM M.I. Electrical Engineer in charge of the Satellite Section Ministry of PTT, Riyadh
- D M. BABTAIN Ali Mohamed Frequency Division Chief Ministry of Information, Riyadh
- D M. SALEEM Saleh Mohammed Chief Satellite Section, Ministry of PTT Riyadh
- A M. MOHAMED Obaidulla Hidayetulla Ministry of PTT, Riyadh

Argentine (Rép.) - Argentine Republic -Argentina (Rep.)

- D M. JIMÉNEZ DÁVILA Fernando Ministro Plenipotenciario Mission permanente d'Argentine, Genève
- D M. BUDIC Domingo V.
 Secretario
 Mission permanente d'Argentine, Genève

Australie - Australia

- C M. WILKINSON Edward James First Assistant Secretary, Postal and Telecommunications Department Melbourne
- CA M. DIXON John Malcolm Sectional Engineer, Forward Planning and Development Australian Broadcasting Control Board Melbourne
- M. CARUANA V.A.A.
 Supervising Engineer
 Spectrum & International Australian
 Telecommunications Commission
 Melbourne
- D M. COHEN Alec Senior Engineer, Planning Australian Broadcasting Commission Sydney, NSW
- D M. D'ARCEY Anthony
 Principal Technical Officer
 (International Conferences)
 Postal and Telecommunications Department
 Melbourne
- M. FISHER James Hubert Thomas
 Director of Engineering Development
 Federation of Australian Commercial
 Television Stations, Melbourne
- D M. HAAGENSEN Hans Frederick International Co-ordinating Engineer (Broadcasting)
 Australian Telecommunications Commission Melbourne
- D M. KENNA Vernon Francis Consultant Federation of Australian Radio Broadcasters Sydney

Australie - Australia (suite)

D Mrs MORISON-TURNBULL J.L. First Secretary Mission permanente d'Australie, Genève

<u>Autriche - Austria</u>

- C Dr BÖNISCH Alfred Conseiller ministériel Direction générale des Postes et des Télégraphes, Wien
- CA Dipl.Ing. LETTNER Gerd Commissaire Direction générale des Postes et des Télégraphes, Wien
- CA Ing. STEINER Ernst Ingénieur, Secrétaire de la Direction générale des Postes et des Télégraphes Wien
- D Dipl.Ing. Dr BERGER Josef Abteilungsleiter Österreichischer Rundfunk, Wien
- D Dipl.Ing. BURGSTALLER Josef Hauptabteilungsleiter Österreichischer Rundfunk, Wien
- D Ing. JASCHEK Josef
 Hauptabteilungsleiter
 Österreichischer Rundfunk, Wien
- D Dr ÖLSBÖCK Karl Commissaire supérieur Direction générale des Postes et des Télégraphes, Wien
- D Dipl.Ing. WASSICZEK Norbert Direktor Österreichischer Rundfunk, Wien
- A Ing. KLEMENT Gerhard Ingénieur, Referent, Osterreichischer Rundfunk, Wien

<u>Bahreïn (Etat de) - Bahrain (State of</u>) -<u>Bahrein (Estado de</u>)

C M. AMIN Mohamed R. Ministry of Information State of Bahrain <u>Bangladesh (Rép. Populaire du)</u> -<u>Bangladesh (People's Republic of</u>) -<u>Bangladesh (Rep. Popular de</u>)

- C M. AABAD A.M.M. Chief Engineer, Bangladesh Television Dacca
- CA M. MURSHED Manzur Director Ministry of Communications, Posts, Telegraphs and Telephone Division Dacca
- D M. KHAN Shams-Ud-Dowla Installation Engineer of Bangladesh Television, Government of Bangladesh Dacca

Belgique - Belgium - Bélgica

- C M. BOUCHIER Pierre-Charles-Marie
 Directeur général aux Affaires
 Internationales
 Régie des Télégraphes et des Téléphones
 Bruxelles
- CA M. HAUSEUX Raymond Ingénieur-Directeur Radiodiffusion-Télévision Belge Bruxelles
- D M. DEVENTER Etienne Ingénieur Radiodiffusion-Télévision Belge Bruxelles
- D M. GEWILLIG Michel R.V.
 Directeur général des Services techniques
 Radiodiffusion-Télévision Belge
 Bruxelles
- D M. HANSEN Georges C.W.
 Conseiller technique
 Radiodiffusion-Télévision Belge
 Régie des Téléphones et des Télégraphes
 Bruxelles
- M. NOTERDAEME Paul
 Représentant permanent de la Belgique auprès des Nations Unies à Genève
- D M. PETRONIO Frédéric Ingénieur principal chef de service Radiodiffusion-Télévision Belge Bruxelles

Belgique - Belgium - Bélgica (suite)

- D M. de SEGESSER DE BRUNEGG Henri Représentant permanent Adjoint de la Belgique auprès des Nations Unies à Genève
- M. VAN CROMBRUCGE Jan R.M.C.
 Ingénieur civil
 Régie des Téléphones et des Télégraphes
 Bruxelles

Bénin (Rép. Populaire du) - Benin (People's Republic of) - Benin (Rep. Popular de)

- C M. BOCCO Coffi Norbert Directeur général adjoint Office de Radiodiffusion et Télévision du Bénin, Cotonou
- CA M. AMOUSSOU Comlanvi Evariste Chef Section Radio Office des Postes et Télécommunications Cotonou

<u>Biélorussie (Rép. Socialiste Soviétique</u> <u>de) - Byelorussian Soviet Socialist</u> <u>Republic - Bielorrusia (Rep. Socialista</u> <u>Soviética de</u>)

C M. ULASSIK V.Ia. Vice-Ministre des Postes et Télécommunications de la RSS de Biélorussie Minsk

Bolivie (Rép. de) - Bolivia (Rep. of) -Bolivia (Rep. de)

- C M. OSSORIO BELTRAN René G. Director General de Telecomunicaciones Dirección General de Telecomunicaciones La Paz
- D M. ARZE CHAVEZ Luis Ingeniero Departamento de Transmisiones La Paz

<u>Brésil (Rép. Fédérative du) - Brazil</u> (Federative Republic of) - Brasil (Rep. Federativa de)

C Dr ALBERNAZ João Carlos F. Assistant to the Secretary for Planning and Technology Ministry of Communications, Brasilia <u>Brésil (Rép. Fédérative du) - Brazil</u> (Federative Republic of) - Brasil (Rep. Federativa de) (suite)

- D Mme BEILER Tereza Mondino
 Assistant of the Technical Coordinator
 of Broadcasting Services Secretariat
 General Secretariat
 Ministry of Communications, Brasilia
- + M. CARLEIAL Aydano B. Research Associate for Instituto de Pesquisas Espaciais Instituto de Pesquisas Espaciais São Paulo

Bulgarie (Rép. Populaire de) - Bulgaria (People's Republic of) - Bulgaria (Rep. Popular de)

- C M. IGNATOV Ivan Ministre-Adjoint Ministère des Postes et Télécommunications Sofia
- CA M. JANEV Janko Chef du Département Radio et Télévision Ministère des Postes et Télécommunications Sofia
- D M. BALABANOV Bontcho
 Attaché de recherches
 Ministère des Postes et Télécommunications
 Sofia
- D M. PETKOV Boris
 Spécialiste
 Ministère des Postes et Télécommunications
 Sofia
- M. PETROV Ilia
 Mission permanente de la Bulgarie auprès des Nations Unies à Genève
- D M. PEYTCHEV Gueorgui
 Chargé de recherches scientifiques
 Ministère des Postes et Télécommunications
 Sofia

Burundi (Rép. du) - Burundi (Rep. of) Burundi (Rep. de)

CA M. HUMUZA Bernard Directeur technique de la Radiodiffusion Nationale du Burundi La Voix de la Révolution du Burundi Bujumbura Canada - Canada

- C M. WARREN Gabriel I. Director General International Telecommunications Branch Department of Communications, Ottawa
- CA M. BLEVIS B.C. Director General Space Technology and Applications Department of Communications, Ottawa
- CA M. BROOKS Gary C. Director WARC Activities Department of Communications, Ottawa
- D M. AMERO Ronald G.
 Head, Space Services Engineering
 Telecommunication Regulatory Service
 Department of Communications, Ottawa
- M. HEAVENOR A.E.
 Chief Frequency Assignment and
 Licensing Division
 Department of Communications, Ottawa
- D M. JONES Robert W. Chief Spectrum Policy, Broadcast Services Department of Communications, Ottawa
- D M. LESTER R.M. Director, Communications System Telesat Canada, Ottawa
- D M. MOHER M.J. Secretary Mission permanente du Canada, Genève
- D M. SIOCOS Christos A. Chief Consultant Engineer Canadian Broadcasting Administration Montreal
- D M. WEESE D.E. Manager, Systems Engineering Communication Systems Telesat Canada Ottawa
- D M. ZEITOUN Ralph Chief, Radio Television & Telecommunications Commission, Ottawa

<u>Centrafricain (Empire) - Central African</u> <u>Empire - Centroafricano (Imperio</u>)

- CA M. TOKPAN Gilbert Directeur des Télécommunications Ministère des Transports, des Postes et Télécommunications, Bangui
- D M. NIAKPAKO Gabriel Ingénieur Radio, responsable des Services Techniques de la Télévision Bangui
- M. OUOKO DELAMBAUT Adolphe
 Chef des Services Transmission
 Ministère des Transports, des Postes et
 Télécommunications, Bangui

<u>Chili - Chile</u>

С

- C M. RAMÍREZ ZEPEDA Gonzalo Jefe del Departamento de Telecomunicaciones Estado Mayor de la Defensa Nacional Santiago
- CA M. CASANUEVA ULLOA Pedro Rolando Jefe del Departamento de Telecomunicaciones Cuerpo de Carabineros, Santiago
- D M. CORDERO VALLEJOS Julio César
 Técnico, Jefe Sección Transporte de Señal
 Televisión Nacional de Chile
 Santiago
- M. LAGOS ERAZO Jaime
 Consejero
 Delegación Permanente de Chile ante los
 Organismos Internacionales, Santiago

Chine (Rép. Populaire de) - China (People's Republic of) - China (Rep. Popular de)

- M. LU Ke-chin Deputy Director of the Technical Department of the Central Broadcasting Administration Peking
- CA M. HSU Chung-ming Deputy Chief Engineer of Peking Television Peking
- D Mme CHAO Chang-ying Staff Member Mission permanente de la Chine, Genève

Chine (Rép. Populaire de) - China (People's Republic of) - China (Rep. Popular de) (suite)

- D M. FAN Kuang-ti Engineer of the Central Broadcasting Administration, Peking
- D M. HAN Ching-yu Deputy Head of the International Liaison Department of the Central Broadcasting Administration, Peking
- D M. KAO Feng-chi Engineer of Peking Television Peking
- D M. LI Wen-chieh Engineer of the Research Institute of Post and Telecommunication Sciences Peking
- D M. PAN Chen-chung Engineer of the Central Broadcasting Administration, Peking
- D M. WU Hsueh-tsai Engineer of the Ministry of Posts and Telecommunications, Peking
- D M. YANG Wei-yuan Engineer of the Research, Institute of Post and Tlecommunication Sciences Peking
- M. CHAO Chin Translator of Radio Peking Peking
- M. TSO Teng-jung Translator of Radio Peking Peking

Chypre (Rép. de) - Cyprus (Rep. of) -Chipre (Rep. de)

- C M. MICHAELIDES Roger Telecommunications Officer Ministry of Communications & Works Nicosie
- D M. ASTREOS Paul T. Chief Engineer, Cyprus Broadcasting Corporation, Nicosie

Chypre (Rép. de) - Cyprus (Rep. of) -Chipre (Rep. de) (suite)

D M. MICHAELIDES Andreas Senior Engineer Cyprus Broadcasting Corporation Nicosie

> <u>Cité du Vatican (Etat de la) - Vatican</u> <u>City State - Ciudad del Vaticano</u> (<u>Estado de la</u>)

- C M. MAFFEO Sabino Directeur Technique Radio Vatican, Cité du Vatican
- CA M. GIUDICI Pier Vincenzo Ingénieur en Chef Radio Vatican, Cité du Vatican

<u>Colombie (Rép. de) - Colombia (Rep. of) -</u> Colombia (Rep. de)

- C M. AGUILERA-BLANCO Jaime Presidente - Empresa Nacional de Telecomunicaciones Ministerio de Comunicaciones, Bogotá
- D M. CAICEDO PERDOMO José Joaquín Subsecretario de Asuntos Exteriores Ministerio de Relaciones Exteriores Bogotá
- M. CHAVES Humberto
 Jefe, División de Telefonía y Telegrafía
 Ministerio de Comunicaciones, Bogotá
- D M. DUARTE Antonio
 Empresa Nacional de Telecomunicaciones
 Ministerio de Comunicaciones, Bogotá
- M. FONSECA-TRUQUE Joaquín
 Subsecretario de Organismos y Conferencias
 Internacionales del Ministerio de
 Relaciones Exteriores
 Ministerio de Comunicaciones, Bogotá
- D M. QUIJANO-CABALLERO Joaquín
 Dr Ing., Ingeniero Consultor
 Ministerio de Comunicaciones, Bogotá

С

Comores (Etat des) - Comoros (State of the) - Comoras (Estado de las)

- C M. IBRAHIM Abdallah Directeur des Télécommunications Service de Télécommunication et de la Radio Comores, Moroni
- CA M. SALIM HAMDANE Salim Ingénieur Electronicien Service de Radio Comores de l'Etat Comorien et de l'Aviation Civile Moroni

Congo (Rép. Populaire du) - Congo (People's Republic of the) - Congo (Rep. Popular del)

- C M. BATHEAS-MOLLOMB Charles Stanislas Directeur Général de la Radiodiffusion Télévision Congolaise, Brazzaville
- CA M. POUEBA Paul Albert Chef du Service de Gestion des Fréquences Direction des Télécommunications Brazzaville
- CA M. TATY KOUKA Albert Directeur technique de la Radiodiffusion Télévision Congolaise, Brazzaville
- M. OKELI Jean-Gabriel
 Directeur
 Office des Télécommunications
 Internationales, Brazzaville

Corée (Rép. de) - Korea (Rep. of) -Corea (Rep. de)

- C M. SHIN Chung Sup Minister
 Mission permanente de la République de Corée, Genève
- M. CHOI Jin Seoung
 Assistant Manager
 Kim Jae Transmitting Station of
 Korean Broadcasting System, Seoul
- D M. KIM Nai Sung
 First Secretary
 Mission permanente de la République de Corée, Genève
- M. LEE Young Han
 Director, Frequency Division
 Office of Radio Regulator
 Ministry of Communications, Seoul

Côte d'Ivoire (Rép. de) - Ivory Coast (Rep. of the) - Costa de Marfil (Rep. de la)

- M. NOGBOU Christophe Inspecteur général, Ministère de l'Information, ler Conseiller technique du Ministre, Abidjan
- CA M. BROU Yapo Samson Directeur du Centre d'Etudes des Postes et Télécommunications Office des Postes et Télécommunications Abidjan
- M. ADJE Kadjo
 Sous-Directeur chargé de la Technique
 Radiodiffusion Ivoirienne
 Ministère de l'Information, Abidjan
- D M. DIABATÉ Sékou
 Sous-Directeur des Programmes de Transmissions
 Office des Postes et Télécommunications Abidjan
- D M. KONE Oumar Mauri Ingénieur, Conseiller technique chargé des émetteurs Ministère de l'Information, Abidjan
- D M. YAPO Jean Directeur technique Adjoint des Télécommunications internationales de la Côte d'Ivoire (INTELCI) Abidjan

Cuba

- C M. MARTINEZ ALBUERNE Carlos Director de Control Técnico Ministerio de Comunicaciones, Habana
- CA M. FERNANDEZ RODRIGUEZ Victor Jefe del Departamento Radiocomunicaciones Habana
- D M. FERNANDEZ MAC BEATH Hugo Ingeniero - Ministerio de Comunicaciones Habana

Danemark - Denmark - Dinamarca

- C M. LØNBERG Ib Deputy Head of Radio Section General Directorate of Posts and Telegraphs, Copenhagen K
- CA M. LARSEN P.V. Telegraph Inspector General Directorate of Posts and Telegraphs D Copenhagen K
- M. HANSEN Orla
 Senior Engineer General Directorate
 of Posts and Telegraphs, Copenhagen K
- D M. HEEGAARD J.A. Senior Engineer - Danmarks Radio General Directorate of Posts and Telegraphs Copenhagen K
- D M. LINDBLAD Arne H. Telegraph Inspector, General Directorate of Posts and Telegraphs, Copenhagen
- M. POULSEN Thorbjørn
 Civil Engineer, Telefonverk Føroya
 Løgtings, General Directorate of Posts
 and Telegraphs, Copenhagen K

Egypte (Rép. Arabe d') - Egypt (Arab Republic of) - Egipto (Rep. Árabe de)

- C M. ANTAR Ahmed Hosni Inspector General of Transmission Egyptian Radio and TV Federation Cairo
- CA M. BADR Mokhtar Taha Director General of Research Department Egyptian Radio and TV Federation Cairo
- D M. DINANA Taher
 Second Secretary
 Mission permanente d'Egypte, Genève
- A M. HUSSEIN Abd El-Samie Mostafa Technical Advisor, Egyptian Radio & TV Federation, Cairo

<u>Emirats Arabes Unis - United Arab Emirates -</u> <u>Emiratos Árabes Unidos</u>

C M. FANOUS Halim Jiries Director of Telecommunications Ministry of Communications Abu Dhabi

Equateur - Ecuador

- C M. AYALA L. José Subsecretario de Relaciones Exteriores (Vice-Ministro) del Ecuador Ministerio de OOPP y Comunicaciones Instituto Ecuatoriano de Telecomunicaciones Quito
 - M. LARA P. César Jefe de Planificación de la Dirección Nacional de Frecuencias del Instituto Ecuatoriano de Telecomunicaciones IETEL Ministerio de OOPP y Comunicaciones Quito
- M. MALDONADO ARMENDARIS Gonzalo E.
 Jefe de la Estación Terrena (de Com. por Satélite) del Instituto Ecuatoriano de Telecomunicaciones IETEL Ministerio de OOPP y Comunicaciones Quito
- M. VAIDEZ Rodrigo
 Embajador-Representante Permanente en Ginebra, Ministerio de Relaciones
 Exteriores del Ecuador, Quito

Espagne - Spain - España

- C M. QUINTAS CASTAÑS Valentín Ingeniero Jefe de Coordinación Técnica Dirección General de Radiodifusión y Televisión, Madrid
- CA M. CHAMORRO SANTA CRUZ Lorenzo Doctor Ingeniero, Ingeniero Jefe del Servicio de Planificación de RTVE, Radiotelevisión Española, Madrid
- D M. DE PAULA PARDAL José Ingeniero Jefe de Planificación Central Dirección General de Radiodifusión y Televisión, Madrid
- D M. GARCÍA-CEREZO Luis Primer Secretario Mission permanente d'Espagne, Genève
- D M. JIMENEZ RODRIGUEZ Manuel Jefe de Planificación red primaria Radio Televisión Española, Madrid
Espagne - Spain - España (suite)

D M. RODRÍGUEZ PASTOR Alberto Ingeniero Técnico Planificación RTVE Radio Televisión Española, Madrid

Etats-Unis d'Amérique - United States of America - Estados Unidos de América

- C The Honorable Robert E. LEE Commissioner, Federal Communications Commission, Washington
- CA M. HUFFCUTT Gordon L. Adviser, Office of International Communications Policy, Department of State, Washington
- CA M. McNAUGHTEN Neal K. Assistant Chief, Broadcast Bureau, Federal Communications Commission, Washington
- D M. ANDERSON Dexter Office of International Communications Policy, Department of State, Washington
- D M. BARNLA Jerome D. Satellite Business Systems, Washington
- D M. BREIG Charles H. Broadcast Bureau, Federal Communications Commission, Washington
- D M. BURKE J.R. Joint Council on Educational Telecommunications, Washington
- D M. EBEL A. James National Association of Broadcasters Koln-TV, Lincoln, Nebr.
- D M. FREIBAUM Jerome Office of Applications, National Aeronautics and Space Administration, Washington
- M. FRIEDLAND Sidney
 Telecommunications Attache
 US Mission in Geneva, US Department of
 State, Washington
- D M. GOULD Richard G. President, Telecommunications Systems, Washington

Etats-Unis d'Amérique - United States of America - Estados Unidos de América (suite)

- D M. HUPE Howard H. Director of Telecommunications Policy Department of Health, Education and Welfare, Washington
- M. JACOBS Edward R.
 Office of the Chief Engineer,
 Federal Communications Commission,
 Washington
- M. JAHN William H.
 Chief, Frequency Assignment Review
 Section, Office of Telecommunications
 US Department of Commerce, Washington
- M. JANSKY Donald M.
 Assistant Director, Office of
 Telecommunications Policy, Executive
 Office of the President, Washington
- D M. JERUCHIM Michel C. Valley Forge Space Center, General Electric Company, Philadelphia, Penn.
- M. KELLEHER John J.
 Vice-President, Engineering, Systematics General Corporation, Fall Church, Va
- D M. LEPKOWSKI Ronald J. Supervisor Engineer, Federal Communications Commission, Common Carrier Bureau, Washington
- D M. MILLER Edward F. NASA Lewis Research Center, Cleveland, Ohio

D

D

- M. MILLER John E. NASA Goddard Space Flight Center Greenbelt, Md
- M. REINHART Edward E. Jet Propulsion Laboratory Pasadena, Ca
- D M. SAWITZ Peter H. Senior Scientist, Operations Research Inc., Silver Spring, Md

<u>Etats-Unis d'Amérique - United States of</u> <u>America - Estados Unidos de América</u> (suite)

- D M. SHRUM Richard E. Chief, Treaty Branch, Federal Communications Commission, Washington
- D M. URBANY Francis S.
 Policy Manager, Office of Telecommunications Policy,
 Executive Office of the President,
 Washington, D.C.
- D M. WEISS Hans J. Communications Satellite Corporation Washington
- D M. WEPPLER H. Edward Engineering Director, AT & T Co Basking Ridge, N.J.
- A M. ZIMMER Thomas M. Assistant General Counsel, COMSAT General, Washington, D.C.

Ethiopie - Ethiopia - Etiopía

- C M. SEYOUM Gabre Christos Manager Radio Division, Telecommunications Service of Ethiopia, Addis Ababa
- D M. SEBHATU Tesfatsion Chief of Radio Regulatory Branch Addis Ababa

Finlande - Finland - Finlandia

- C M. TOIVOLA Keijo Ensio Director of Radio Division, Administration of Posts and Telecommunications, Helsinki
- CA M. TERÄSVUO Kalevi Ilmari Head of Section, Administration of Posts and Telecommunications, Helsinki
- D M. LOUNASTÖRMÄ Sven Chief Engineer, Yleisradio, Helsinki
- D M. SVENSSON Reijo Bertel Dipl. Engineer, Administration of Posts and Telecommunications, Helsinki

Finlande - Finland - Finlandia (suite)

M. TARKKA Ossi K. Head of Radio Relay Planning, Yleisradio, Helsinki

France - Francia

- C M. GUEURY Jean Ministre Plénipotentiaire Ministère des Affaires Etrangères Paris
- D M. ARNAUD Jean-François Ingénieur en chef, Télédiffusion de France, Paris
- D M. BISNER René-Jean-Marie
 Adjoint au chef du Bureau Centralisateur
 National pour le Contrôle des Emissions
 Radioélectriques, Secrétariat d'Etat aux
 Postes & Télécommunications, Paris
- D M. CAZEMAJOU Jean Ingénieur, CNES, Toulouse
- D M. CECILLON Jean Directeur Adjoint Radio France, Paris
- D M. DERIEUX Charles Ingénieur, CNES, Paris
- D M. DUBEC Augustin Ingénieur, Télédiffusion de France Montrouge
- D M. FILLIOL Stanislas Henri Jean
 Conseiller des Affaires Etrangères
 Ministère des Affaires Etrangères
 Paris
- D M. FORTIN Jacques Maurice Secrétaire technique, SCART, Paris
- D M. GEORGY Jean
 Ingénieur, Télédiffusion de France, Paris
- D M. GILOTAUX Pierre
 Vice-Président de la Commission Technique,
 SCART, Paris

France - Francia (suite)

- D Mme GRANIER Claude-Marie Cadre Informaticien, Télédiffusion de France, Paris
- M. HALLOUIN Claude
 Administrateur Hors Classe des Services
 du Premier Ministre, Sofirad, PARIS
- D Mlle HUET Marie Ingénieur en chef, Télédiffusion de France, Paris
- D M. LORIQUET Paul Ingénieur, Télédiffusion de France Paris
- M. MENES Jean
 Directeur Départemental Adjoint,
 Direction des Télécommunications du Réseau international, Paris
- M. MONNOT Michel
 Ingénieur des Télécommunications,
 Postes et Télécommunications, CNET
 Issy-les-Moulineaux
- D M. NOUAILLE Xavier Société Nationale de Radiodiffusion, Paris
- D M. PHAM TAT Dat Ingénieur, Télédiffusion de France Paris
- D M. POMMIER Daniel Cadre de Direction, Télédiffusion de France, Rennes
- D M. POUZET Alain Ingénieur, CNES, Toulouse
- M. POUZOLS Bernard Julien René
 Vice-Président de la Commission technique
 SCART, Paris
- D M. ROCHICCIOLI François Ingénieur, Radio France, Paris
- D M. SAUVET-GOICHON Daniel Ingénieur, Télédiffusion de France Paris

France - Francia (suite)

- D M. SICE Alain
 Conseiller d'Ambassade,
 Ministère des Affaires Etrangères
 Paris
- D M. SIMO-PONS Richard Ingénieur, CNES, Paris
- D M. TELLIER Francis Ingénieur, Radio France, Paris
- M. VOGE Jean
 Ingénieur général, Directeur délégué pour les Relations internationales à la
 Direction générale des Télécommunications
 Secrétariat d'Etat aux Postes et
 Télécommunications, Paris
- A M. GAUTIER Georges Premier Secrétaire à la Mission permanente de la France auprès des Nations Unies à Genève Ministère des Affaires Etrangères, Paris

<u>Gabonaise (Rép.) - Gabon Republic-</u> <u>Gabonesa (Rep.)</u>

- M. ONANGA Faustin Directeur technique Adjoint Radio-Télévision Gabonaise, Libreville
- M. IMOUNGA Francis
 Ingénieur, Inspecteur technique itinérant
 Radio-Télévision Gabonaise, Libreville
 - M. JAFFRES Raymond Premier Secrétaire Mission permanente du Gabon, Genève

Ghana

С

D

- CA M. BONSO-BRUCE T.N.L. Chief Engineer, Ghana Broadcasting Corp. Accra
- M. JACKSON K.A.
 Chief Communications Officer
 Ghana Frequency Registration Board
 Accra

Grèce - Greece - Grecia

- C M. METAXAS André Ambassadeur Représentant permanent de la Grèce auprès des Organisations Internationales à Genève, Ministère des Affaires Etrangères, Délégation permanente, Genève
- CA M. HAGER Constantinos Chef de Section des fréquences Direction générale des Postes et Télécommunications, Athènes
- D M^e CASSAPOGLOU Vassili
 Avocat à la Cour, Expert en matière de Droit de l'Espace et des Télécommunications internationales, Direction générale des Postes et Télécommunications, Athènes
- M. CHATZIMANOLIS Theodoros
 Engineer of Planning Department (OTE)
 General Direction of Posts and
 Telecommunications, Athènes
- D M. CHRISTOPOULOS Ioannis Engineer of Broadcasting and Television Service, General Direction of Posts and Telecommunications, Athènes
- D M. KASTANAS George
 Technical Director Broadcasting
 8 TV Service, General Direction of Posts
 and Telecommunications, Athènes
- M. MANASSIS Nicolaos
 Ingénieur de la Direction de Planification
 Direction générale des Postes et
 Télécommunications, Athènes
- M. NICOLADIS Emmanouil
 Sub-Director of Planning Department (OTE)
 General Direction of Posts and
 Telecommunications, Athènes
- D M. PAPAHAJAKIS George Head, Frequency Management Section D Communications and Electronics Directorate General Direction of Posts and Telecommunications, Athènes
- M. SIDERIS Anastase
 Premier Conseiller d'Ambassade
 Ministère des Affaires Etrangères
 Mission permanente de la Grèce, Genève

Grèce- Greece - Grecia (suite)

- D M. SIMEONIDIS George Ingénieur de la Direction de Planification Direction générale des Postes et Télécommunications, Athènes
- D M. ZANGLIS Aristidis
 Ingénieur en Chef de la Direction de Planification
 Direction générale des Postes et Télécommunications, Athènes

Guatemala

- M. ALTAN BARILLAS Mario René Asesor de Frecuencias y Asuntos Internacionales de Radio, Ministerio de Comunicaciones y OP, Guatemala
- M. CHENAL LUNA Rodolfo
 Subdirector General de Radiodifusión
 y Televisión Nacional, Ministerio de
 Comunicaciones y OP, Guatemala

<u>Guinée (Rép. de) - Guinea (Rep. of) -</u> <u>Guinea (Rep. de)</u>

- CA M. DIALLO Mamadou Saliou Secrétaire général du Comité national de Coordination des Télécommunications Ministère des Postes et Télécommunications Conakry
- D M. TOURE Sidiki Ingénieur au Ministère de l'Information et de l'Idéologie, Conakry

Haïti (Rép. d') - Haiti (Rep. of) -Haití (Rep. de)

- C M. RIMPEL Antonio Directeur général du Conseil national de Télécommunications de la République d'Haïti, CONATEL, Port-au-Prince
 - M. MICHEL Fritz A. Conseiller Exécutif, Télécommunications d'Haīti SAM Titre III, Port-au-Prince

Haute-Volta (Rép. de) - Upper Volta (Rep. of) - Alto Volta (Rep. del)

C M. SONGRÉ Pierre Claver Chef de Centre HF, Radiodiffusion Télévision Voltaïque, Ouagadougou <u>Hongroise (Rép. Pop.) - Hungarian</u> People's Republic - Húngara (Rep. Pop.)

- C M. HORN Dezsö Ministre Adjoint des Communications et des Postes, Directeur général des Postes Direction générale des PTT, Budapest
- CA Dr HORVATH Lajos Chef de Département pour les Radiocommunications, Direction générale des PTT, Budapest
- D M. HECKENAST Gábor Directeur technique, Radio Hongroise Budapest
- D M. KARÁCSONY Mihály Direction générale des PTT, Budapest
- D M. SZABÓ Miklós Chef de Département, Radio Hongroise Budapest
- D M. SZÉKELY János Conseiller supérieur Direction générale des PTT, Budapest
- D Dr VILLÁNYI Ottó Conseiller Direction générale des PTT, Budapest

Inde (Rép. de l') - India (Rep. of) -India (Rep. de)

- C M. SRIRANGAN T.V. Wireless Adviser, Government of India New Delhi
- CA M. MITRA S.N. Chief Engineer, All India Radio New Delhi
- CA M. RAYALU V.A.D. Chief Engineer, Door Darshan (Television, India), Indian Administration New Delhi
- D M. KRISHNAMURTHY Srinivasa Officer on Special Duty (TV) Door Darshan, Mandihouse New Delhi
- D M. MOHANAVELU K.S. Engineer, ISRO HQ., Bangalore

Inde (Rép. de l') - India (Rep. of) -India (Rep. de) (suite)

- D M. NARGAS B.S. Assistant Wireless Adviser Indian Administration, New Delhi
- D Dr RAO B.S. Coordinator, Industrial & Technology Programs, India Space Research Organization, Bangalore

Indonésie (Rép. d') - Indonesia (Rep. of) -Indonesia (Rep. de)

- C M. PRATOMO Thomas Aquinas Director of Frequency Management and Monitoring, Directorate General of Posts and Telecommunications, Jakarta
- D M. BRATAHALIM Sulaiman Directorate General of Posts and Telecommunications, Jakarta
- M. ISKANDAR Arfan
 Head, Division for Improvement of
 Administration Efficiency, Radio Republic
 Indonesia, Jakarta
- M. MOENTOJO Hadisuwarno
 Legal & Foreign Affairs, Chief Division in Telecommunications State Enterprise, Headquarters, Directorate General of Posts and Telecommunications, Jakarta
- M. PERANGIN ANGIN Remedi
 Senior Engineer of Transmission
 Planning Division, Perum Telekomunikasi
 (Headquarters), Bandung
- D M. PRIYATNA Abdurrasyid Legal Advisor, Directorate General of Posts and Telecommunications, Jakarta
- D M. SRI SLAMETO Chief of the Transmission Technical Division, Perum Telekomunikasi (Headquarters), Bandung

Iran - Irán

C M. NADIMI Siamak Deputy Director General of Telecommunications, Ministry of PTT, Teheran

Iran - Irán (suite)

- CA M. FADAMI Ahmad Director of Logistic & Maintenance of National Iranian Radio & Television Teheran
- D M. ARASTEH Kavouss Manager for Sound Broadcast Planning National Iranian Radio & Television Teheran
- D M. ARDEHALI S. Alireza Mohsen Director of Satellite Planning Department Telecommunication Company of Iran, TCI Teheran
- D M. HAGOPIAN Rouben Manager of Telecommunications of the National Iranian Radio Television Teheran
- D M. KHATIBI Mehdi Ingénieur, Radio-Télévision Nationale Iranienne, Teheran
- M. KIAN Mehdi
 Engineer Telecommunications Department
 National Iranian Radio & Television
 Teheran
- D Mme RADJY M.B. Iranian Delegation Member, Teheran

Irlande - Ireland - Irlanda

- C M. Ó'SÉAGHDHA S. Assistant Secretary, Department of Posts & Telegraphs, Dublin
- CA M. GRANT M. Assistant Principal Officer Department of Posts and Telegraphs, Dublin
- D M. CALLENDER T.
 Assistant Staff Engineer, Department of Posts & Telegraphs, Dublin
- D M. MOUNTJOY J.H.Executive Engineer, Department of Posts & Telegraphs, Dublin
- D M. SLOWEY E.J. Director of Engineering, Radio Telefís Éireann, Dublin

Irlande - Ireland - Irlanda (suite)

- A M. COGAN John F. Deputy Permanent Representative Mission permanente de l'Irlande, Genève
- A M. GAYNOR Sean Ambassadeur Mission permanente de l'Irlande, Genève

Islande - Iceland - Islandia

- CA M. ARNAR Gustav Head of Division General Directorate of Posts and Telecommunications, Reykjavik
- D M. FRIMANNSSON Hördur Chief Engineer, Icelandic State Broadcasting, Reykjavik

<u>Italie - Italy - Italia</u>

- C M. PETTI Angelo Dirigeant Supérieur, PTT, Rome
- D M. ALBERICO Franco Deputy Chief Service, RAI, Rome
- D M. ASLAN Piero Attaché, Mission permanente d'Italie Ministère des Affaires Etrangères, Rome
- D M. BRUNI M.C. Direttore di Sezione Ministero Poste e Telecomunicazioni, Rome
- M. BUSSINI Vittorio E.
 Dirigente Superiore Tecnico
 Azienda di Stato dei Servizi Telefonici Rome
 - M. CARMASSI Francesco Tecnico, Radiotelevisione Italiana Rome
- D M. CITO Ruggero Premier Dirigeant, Administration PT Rome
- D M. D'ANDRIA Emanuele Engineer, TELESPAZIO, Rome

п

Italie - Italy - Italia (suite)

- D M. GLORIOSO Giulio Dirigeant d'Exploitation Ministère des PTT, Rome
- D M. GRECO Antonio TELESPAZIO, Rome
- D M. LA PADULA Aldo Consigliere Tecnico Ministero PT, Rome
- D M. LARI Mario Planning Director, RAI, Rome
- D M. MASTINI Mario Chef de Service, RAI, Rome
- D M. POLACCO Gian Mario Directeur-Adjoint, RAI, Rome
- D M. RAMILLI Ubaldo Directeur de Division Administration PT Italiana, Rome
- D M. RODINO DI MIGLIONE Francesco Radiotelevisione Italiana, Rome
- D M. ROSSI Giuliano Engineer, RAI, Rome
- D M. SETTI Remo Chef-Adjoint de Service Radiotelevisione Italiana, Rome
- D M. TERZANI Carlo Directeur des Relations techniques internationales, RAI, Rome
- D M. TOMATI Lorenzo Ingénieur Chef de Service, RAI, Rome
 - Japon Japan Japon
- C++ M. MAKI Yutaka Director General of the Radio Communications Department, Radio Regulatory Bureau, Ministry of Posts & Telecommunications, Tokyo

Japon - Japan - Japon (suite)

C++ M. KADOTA Hiroshi Deputy Director General of Radio Regulatory Bureau, Ministry of Posts and Telecommunications, Tokyo

++ du 27.1.1977 au 13.2.1977

- CA M. UESHIMA Shiro Head, Frequency Division, Radio Regulatory Bureau, Ministry of Posts & Telecommunications, Tokyo
- CA M. SEO Masaki Councillor Mission permanente du Japon, Genève
- At M. ARAI Akira Special Assistant to the Director General Radio Regulatory Bureau Ministry of Posts & Telecommunications Tokyo
- At M. FUJIOKA Masayoshi Special Assistant to Radio Regulatory Bureau, Ministry of Posts & Telecommunications, Tokyo
- At M. TATENO Satoshi Deputy Head Frequency Division, Radio Regulatory Bureau Ministry of Posts & Telecommunications Tokyo
- At M. KAJITANI Yoichi First Secretary Mission permanente du Japon, Genève
- A M. HARA Hiroshi Controller Headquarters of Technical Administration & Construction NHK Ministry of Posts & Telecommunications Tokyo
- M. IZUMI Takehiro
 Assistant Coordinator Planning
 Headquarters of Technical Administration
 & Construction NHK
 Ministry of Posts & Telecommunications
 Tokyo

⁺⁺ du 10.1.1977 au 26.1.1977

Japon - Japan - Japon (suite)

- M. MOCHIZUKI Ryo
 Senior Engineer Planning
 Headquarters of Technical Administration
 & Construction NHK, Ministry of Posts
 and Telecommunications, Tokyo
- M. MATSUSHITA Misao
 Manager, Radio Frequency & Satellite
 Broadcasting Research Group
 Technical Research Laboratories, NHK
 Ministry of Posts and Telecommunications
 Tokyo
- A M. MUTOH Tonami Assistant Vice-President Engineering Television Center, Tokyo Broadcasting System Inc., Ministry of Posts and Telecommunications, Tokyo
- M. OKA Shuichito
 Technical Adviser, Electronic Industries
 Association of Japan
 Ministry of Posts & Telecommunications
 Tokyo
- ++ M. OHMI Katsuro Engineer, Radio-Frequency & Satellite Broadcasting Research Group Technical Research Laboratories, NHK Ministry of Posts & Telecommunications Tokyo
- ++ Attaché

Jordanie (Royaume Hachémite de) - Jordan (Hashemite Kingdom of) - Jordania (Reino Hachemita de)

- C M. ALKHAS Radi Chief Engineer TV, Jordan Television Amman
- CA M. KABARITI Saleh Councillor, Mission permanente de Jordanie, Genève
- D M. ABDULLA Raféa Jordanian Broadcasting Studio Engineer Jordanian Broadcasting, Amman
- D M. ASAD Rafé Abdallah Broadcasting Engineer, Broadcasting Amman Amman
- D M. ASFOURAH Osameh Head Engineering Broadcasting Jordan Broadcasting, Amman
- D M. ZU'BI Naif Video Engineer Jordan Television, Amman

<u>Kenya (Rép. du) - Kenya (Rep. of</u>) -<u>Kenya (Rep. de</u>)

- C M. AMIRA Charles Chief Engineer (Television) Ministry of Information and Broadcasting Nairobi
- M. KIMANI James Peter
 Development Engineer
 Ministry of Information and Broadcasting
 Nairobi
- M. KINYVA Peterson John
 Senior Assistant Secretary
 Ministry of Power and Communications
 Nairobi

Koweit (Etat de) - Kuwait (State of) -Kuwait (Estado de)

- C M. AL-MAZEEDI Jawad A. Chief Engineer Radio Ministry of Information, Kuwait
- CA M. DASHTY Yacob Yosif Chief Engineer, Kuwait TV Ministry of Information, Kuwait
- D M. SOUD Ibrahim A. Engineer Ministry of Communications, Kuwait

Lao (Rép. Démocratique Populaire) -Lao (People's Democratic Republic) -Lao (Rep. Democrática Popular)

- C M. BO Thao Directeur-Adjoint Direction du Plan et Services Financiers Vientiane
- CA M. LUANGPHENGSOUK Thongphét Ingénieur radioélectrique Radiodiffusion nationale, Vientiane

Liban - Lebanon - Libano

- C M. GHAZAL Maurice Directeur des Services techniques des Télécommunications Administration générale des Téléphones et Télégraphes, Beyrouth
- C M. BANNA Mahmoud Ambassador Mission permanente du Liban, Genève

Liban - Lebanon - Libano (suite)

- CA M. SINNO Ashraf Telecommunications Engineer PTT, Beyrouth
- S Mme HOMSY Ruby Secrétaire Mission permanente du Liban, Genève

<u>Libyenne (Rép. Arabe) - Libyan Arab</u> Republic - Libia (Rep. Árabe)

- CA M. ABUGOFFA Ali Khalifa Director of Long Lines Department P & T Corporation, Tripoli
- D M. ALSABEY Mohamed Saleh Frequency Coordination P & T Corporation, Tripoli
 - Liechtenstein (Principauté de) -Liechtenstein (Principality of) -Liechtenstein (Principado de)
- C M. von LEDEBUR M. Councillor Princely Government, Vaduz
 - Luxembourg Luxemburgo
- C M. REICHLING Charles Ministre plénipotentiaire Directeur des Relations Economiques Internationales Administration des Postes et Télécommunications, Luxembourg
- CA M. RETTEL Jean Ambassadeur du Luxembourg en Suisse Berne
- D Mme ANCEL-LENNERS Jacqueline
 Secrétaire de Légation
 Représentation permanente du Luxembourg
 auprès de l'ONU, Genève
- D M. DONDELINGER Charles
 Ingénieur principal
 Administration des Postes et
 Télécommunications, Luxembourg
- D M. FABER Paul
 Ingénieur en Chef
 Administration des Postes et
 Télécommunications, Luxembourg

Luxembourg - Luxemburgo (suite)

- D M. GRAAS G.
 Directeur Général
 Administration des Postes et
 Télécommunications, Luxembourg
- D M. SCHILTZ André
 Ingénieur des Studios
 Administration des PTT, Luxembourg

Madagascar (Rép. Démocratique de) -Madagascar (Democratic Rep. of) -Madagascar (Rep. Democrática de)

- C M. RAKOTOARIVELO Benjamin Chef du Service de la Recherche, de l'Equipement et de la Vulgarisation des Postes Récepteurs, Direction technique du Ministère-Conseil à la Présidence de la République Démocratique de Madagascar Tananarive
- D M. RATIARISON Maximilien
 Chef Division Transmissions, Ministère des Postes et Télécommunications
 Direction des Télécommunications
 Tananarive

Malaisie - Malaysia - Malasia

- C M. CHIN Fiang Khoon Controller of Telecommunications Department of Telecommunications Government of Malaysia, Kuala Lumpur
- C M. NG Ek Poh Director of Telecommunications Telecoms Administration Malaysia Kuala Lumpur
- D M. MANICKAM Shanmugam Controller of Telecommunications Telecommunications Department Government of Malaysia, Kuala Lumpur
- D Dr TAN Soon Hie Engineer, Department of Broadcasting Angaksapuri, Government of Malaysia Kuala Lumpur

<u>Mali (Rép. du) - Mali (Rep. of</u>) -<u>Mali (Rep. del</u>)

- C M. SIDIBE Oumar Chef de la Division technique de la Radio-Mali, Radiodiffusion Nationale du Mali, Bamako
- CA M. SISSOKO Sikon Chef de la Division des Communications par Satellite, Office des Postes et Télécommunications du Mali, Bamako

<u>Malte (Rép. de) - Malta (Rep. of</u>) -Malta (Rep. de)

- C M. SALIBA Evarist Permanent delegate of Malta to the United Nations & Specialized Agencies in Geneva
- CA M. MARMARÁ Joseph First Secretary, Mission of Malta to the United Nations & Specialized Agencies Government of Malta, Minister of Development, Valletta

<u>Maroc (Royaume du) - Morocco (Kingdom of)</u> -Marruecos (Reino de)

- C M. WAKRIM Mohamed Ingénieur en Chef, Chef de la Division des Transmissions, Ministère des PTT Rabat
- CA M. AKALAY Mourad Chef du Service des Radiocommunications Ministère des PTT, Rabat
- CA M. DRISSI QEYTONI Bennacer Directeur technique, Radiodiffusion Télévision Marocaine, Rabat
- CA M. TANANE M'Hamed Jamal Eddine Chef de la Division de l'Equipement à la Direction technique, Radiodiffusion Télévision Marocaine, Rabat
- D M. HAMAM Miloud Ingénieur chargé de la Gestion du Spectre des Fréquences et Relations avec les Organismes Internationaux Radiodiffusion Télévision Marocaine Rabat

<u>Maurice - Mauritius - Mauricio</u>

- C M. CHASLE Louis Raymond Ambassadeur de l'Ile Maurice Représentant permanent auprès de l'Office des Nations Unies et des Institutions spécialisées, Genève
- C M. RAMBERT J.M.H. Noël Chief Engineer, Mauritius Broadcasting Corporation, Curepipe
- D M. SAINT LAMBERT Herbert Engineer, Mauritius Broadcasting Corporation, Curepipe
- D M. TSANG MANG KIN Joseph Premier Conseiller, Ambassade de l'Ile Maurice, Bruxelles

<u>Mauritanie (Rép. Islamique de)</u> -<u>Mauritania (Islamic Rep. of</u>) -<u>Mauritania (Rep. Islámica de</u>)

- C M. LÔ Medoune Directeur technique, Office Mauritanien de Radiodiffusion, Nouakchott
- D M. MANGASSOUBA Aliou Responsable des Centres de Transmission de Nouakchott, Office des Postes et Télécommunications, Nouakchott

<u>Mexique - Mexico - México</u>

- ++ M.PADILLA-LONGORIA José A. Subdirector General de Permisos y de Asuntos Internacionales, Dirección General de Telecomunicaciones, México, DF
- ++ Presidente honorario de la Delegación
- C M. HERNÁNDEZ-G. José J. Jefe del Departamento de Asuntos Internacionales, Dirección General de Telecomunicaciones, México, DF
- CA M. VALENCIA-PÉREZ Luis Jefe del Departamento de Frecuencias Radioélectricas, Dirección General de Telecomunicaciones, México, DF

Mexique - Mexico - México (suite)

- M. ZAMUDIO-ZEA Juan Manuel
 Técnico de la Estación Terrena de
 Tulancingo
 Dirección General de Telecomunicaciones
 México
- A M. CACERES CALVILLO Miguel-Angel Segundo Secretario del Servicio Exterior Mexicano, Genève
- A M. MANTECON-GUTIERREZ José Gerente General, SATELAT, México, DF

Monaco - Mónaco

- C S.E. M. SOLAMITO César Charles Ministre Plénipotentiaire Délégué permanent auprès des Organismes Internationaux, Direction générale des Postes et Télécommunications, Monaco
- D M. AUVRAY Gustave G. Ingénieur
 Direction générale des Postes et Télécommunications, Monaco

<u>Mongolie (Rép. Pop. de) - Mongolian</u> People's Republic - Mongolia (Rep. Pop. de)

D M. BAYART Luvsandorjiin Troisième Secrétaire Mission permanente de Mongolie, Genève

Nicaragua

C M. CAJINA MEJICANO Gastón Embajador Representante Permanente Genève

> Niger (Rép. du) - Niger (Rep. of the) ~ Niger (Rep. del)

C M. IBRAHIM Idrissa Chef Division Transmission, PTT Niamey <u>Niger (Rép. du) - Niger (Rep. of the)</u> -<u>Niger (Rep. del</u>) (suite)

CA M. ZOUDI Issouf Directeur technique de l'Office de Radiodiffusion-Télévision du Niger Office de Radiodiffusion-Télévision du Niger, Niamey

> <u>Nigeria (Rép. Fédérale de) - Nigeria</u> (Federal Republic of) - Nigeria (Rep. Federal de)

- C M. INOMA Raphael Ejoh Nathan Deputy Director (International Relations) Posts and Telecommunications Department Lagos
- D M. AKINWUMI Joshua Ayodele
 Area Engineer
 Ministry of Communication, P & T Department
 Lagos
- D M. NWANKPELE Alphonsus Francis Ikem Chief Engineer Nigerian Ports Authority, Lagos
- M. UKPONG Basil M.
 Troisième Secrétaire
 Mission permanente du Nigeria, Genève

Norvège - Norway - Noruega

- C M. MORTENSEN Per Technical Director, Norwegian Telecommunications Administration, Oslo
- CA M. GRIMSTVEIT Lavrans Chief Engineer Norwegian Telecommunications Administration, Oslo
- D M. BØE Arne Head of International Relations Office Norwegian Telecommunications Administration Oslo
- D M. HEGGELUND Tom Scientist, Norwegian Telecommunications Administration, Oslo
- D M. LOEVAAS Kjell Technical Director Norwegian Broadcasting Corporation Oslo

<u>Norvège - Norway - Noruega</u> (suite)

- D M. NAERLAND G. Scientist Norwegian Telecommunications Administration Oslo D
- D M. OEVENSEN Tore Chief Engineer Norwegian Broadcasting Corporation Oslo
- D M. ØWRE Trygve Head of Division Norwegian Telecommunications Administration Oslo
- D M. STOKKE Knut N. Chief Engineer Norwegian Telecommunications Administration C Oslo
- D M. THEISEN Theis Chief Engineer Norwegian Telecommunications Administration Oslo

<u>Nouvelle-Zélande - New Zealand</u> -Nueva Zelandia

- C M. BUNDLE Robert John Principal, Telecommunications Division (Radio) - Post Office Headquarters New Zealand Post Office, Wellington
- CA M. AKED M.S. Divisional Engineer (Radio) Post Office Headquarters New Zealand Post Office, Wellington
- D M. CARTER J.P. Superintending Engineer Transmission Broadcasting Council of New Zealand Wellington
- D M. CUDBY T.R. Divisional Engineer (Microwave Transmission) Broadcasting Council of New Zealand Wellington

<u>Oman (Sultanat d') - Oman (Sultanate of)</u> - Omán (Sultanía de)

D M. AL-JABRY Rashid Haroon Broadcasting Technician Directorate General Radio & TV Ministry of Information & Culture, Muscat <u>Oman (Sultanat d') - Oman (Sultanate of)</u> -<u>Omán (Sultanía de</u>) (suite)

- M. AL-KINDY Hamad Yahya Technical Director Directorate General Radio & TV Ministry of Information & Culture, Muscat
- A Dr SOBOTKA Johann
 Adviser to the Ministry of Information & Culture - Sultanate of Oman
 Directorate General Radio & TV
 Ministry of Information & Culture, Muscat

<u>Ouganda (Rép. de l') - Uganda (Rep. of) -</u> <u>Uganda (Rep. de</u>)

- M. KINENE Khalid Younis Ambassador - Permanent Representative to the United Nations, New York
- CA M. ANGURA Samuel Baker Undersecretary Ministry of Transport & Communications Kampala
- M. BYEKWASO George Wilson
 Principal Assistant Secretary
 Ministry of Information & Broadcasting
 Kampala
- M. MALIK Farooq
 Supt. Engineer Sound Operations
 Senior Broadcasting Engineer
 Ministry of Information & Broadcasting
 Kampala
- A M. MUKASA Pascal International Relations Officer Extelcoms, Nairobi
- A M. SEGGUJJA Pius
 Supt. Engineer Tx
 Ministry of Information & Broadcasting
 Kampala

Pakistan - Pakistan

- C M. SALEEM Khalid Deputy Permanent Representative of Pakistan to United Nations, Genève
- CA M. YAHIA Mohammad Deputy Chief Engineer (Overseas) Pakistan Telegraph and Telephone Department Islamabad
- D M. EHSANUL HAQ Controller Engineering Operation and Maintenance, Pakistan Television Corporation Ltd., Rawalpindi
- D M. MOHAMMAD ZUBAIR Controller Engineering, Pakistan Television Corporation Ltd., Rawalpindi

Panama (Rép. de) - Panama (Rep. of) -Panamá (Rep. de)

C M. VILLAMONTE RAMOS A.P. Embajador, Representante alterno Misión Permanente de Panamá ante la ONU y Organismos Internacionales, Genève

Papua-Nouvelle Guinée - Papua New Guinea -Papua Nueva Guinea

C M. RAILTON G. Hugh Engineer, ITU Affairs and Radio International Telecommunications Division Department of Public Utilities Port Moresby

Paraguay (Rép. du) - Paraguay (Rep. of) -Paraguay (Rep. del)

C M. MONTANARO CANZANO Sabino Ernesto Jefe del Departamento Técnico Administración de Telecomunicaciones Asunción

Pays-Bas (Royaume des) - Netherlands (Kingdom of the) - Países Bajos (Reino de los)

C M. NEUBAUER F.R. Director of Radio Affairs, Netherlands PTT Administration, The Hague Pays-Bas (Royaume des) - Netherlands (Kingdom of the) - Países Bajos (Reino de los) (suite)

- CA M. DE ZWART Hendrik K. Chief Frequency Management Section Netherlands PTT Administration The Hague
- D M. DALHUISEN Aart Directeur général, Radio Nederland Hilversum
- D M. DE VRIJER F.W. Scientific Advisor, Philips Research Laboratories, Eindhoven
- M. DITO J.C.
 Deputy Head of the Radio Television
 Broadcasting Department, Directorate
 of Radio Affairs, Netherlands PTT
 The Hague
- M. DOEVEN Jan
 Engineer, Broadcast Frequencies and
 Coverage, Sound and Television Broadcasting
 Department, Directorate for Radio Affairs
 The Hague
- D M. EDENS J.W. Chief Engineer, N.V. Philips'Gloeilampen fabrieken, Eindhoven
- D M. FRANKE C. Official Frequency Management Section Netherlands PTT Administration The Hague
- D Prof. Dr GELUK Jan J. Director Technical Developments Radio Nederland, Hilversum
- D M. MEERBURG Arend Y.
 Second Secretary
 Mission permanente des Pays-Bas, Genève
- M. SCHUTTENHELM Emile A.
 Conseiller pour les Affaires Internationales
 Nederlandse Omroep Stichting (NOS)
 Hilversum
- D M. de TROYE Jan Louis Membre du Conseil d'administration de la Nederlandse Omroep Stichting (NOS) Hilversum

<u>Pays-Bas (Royaume des) - Netherlands</u> (Kingdom of the) - Países Bajos (Reino <u>de los</u>) (suite)

- D M. VAN DEN BERG Willem F. Cadre de la Division des Affaires générales, Nederlandse Omroep Stichting Hilversum
- M. VAN WILLIGEN Peter W.
 Chef de la Division des Affaires générales, Nederlandse Omroep Stichting Hilversum
- D M. VIJZELAAR Pieter Senior Engineer, NOS, Netherlands Broadcasting Foundation, Hilversum
- D M. WAGENAAR William Joh. Alb. Vice-Président, NOS, Netherlands Broadcasting Foundation, Hilversum
- D M. WILBERS W.A.F. Directeur de la Direction Radio-Télévision et presse, Ministère de la Culture Rijswijk
- D M. WILLEMS P.E. Inspector of PTT PTT Administration, The Hague

Philippines (Rép. des) - Philippines (Rep. of the) - Filipinas (Rep. de)

- C M. CARLOS Zosimo Deputy Director Telecommunication Control Bureau, Board of Telecommunications, Quezon City
- D M. BISUNA Ramon Communications System Engineer Philcomsat, Makati, Rizal
- D M. ESPEJO C.V. Attaché, Mission permanente des Philippines Genève
- D M. LIM Vivencio T. Director Operation Engineering Makati, Rizal

Pologne (Rép. Pop. de) - Poland (People's Rep. of) - Polonia (Rep. Popular de)

- C M. KOZZOWSKI Konrad First Deputy Minister, Ministry of Posts and Telecommunications, Warszawa
- CA Mme SMOLENSKA Halina Director of Radio Department, Ministry of Posts and Telecommunications, Warszawa
- D M. FIECKO Tadeusz
 Conseiller à la Mission permanente de la Pologne à Genève
- D M. JAKUBIK Jerzy Ingénieur, Ministère des PTT, Warszawa
- D M. KUPCZYK Zbyszko-Henryk Engineer, Ministry of Posts and Telecommunications, Warszawa
- D M. STROJWAS Tadeusz Senior Expert, Ministry of Foreign Affairs, Warszawa
- D M. WESO⊅OWSKI Czes≯aw Engineer, Ministry of Posts and Telecommunications, Warszawa
- M. ZYGIEREWICZ Janusz
 Head of Laboratory
 Ministry of Posts and Telecommunications
 Warszawa

Portugal

- C M. DE CARVALHO Adriano Ambassadeur, Représentant permanent du Portugal auprès de l'Office des Nations Unies, Genève
- CA M. FRANCO Domingos António Pires Ingénieur en Chef PTT Portugais, Lisboa
- D M. ALBUQUERQUE Celso João de Ingénieur en Chef Radiodifusão Portuguesa, Lisboa
- D Mme BANDEIRA Maria Teresa Rodrigues Ingénieur en Chef Radiotelevisão Portuguesa, Lisboa

- Portugal (suite)
- D M. OLIVEIRA Vito Manuel Batista Ribeiro de Directeur technique Radiodifusão Portuguesa, Lisboa

<u>Qatar (Etat du) - Qatar (State of</u>) -<u>Qatar (Estado de</u>)

- D M. BRIGGS Ronald
 Director of Engineering
 Ministry of Information, Doha
- D M. ABOU-TALEB Mohammad
 Earth Station Manager
 Telecommunication Department
 Ministry of Communication and Transport
 Doha

<u>République Démocratique Allemande -</u> <u>German Democratic Republic - República</u> Democrática Alemana

- C M. CZERWINSKI Bruno Deputy Minister, Ministry of Posts and Telecommunications, Berlin
- CA M. KAHLE Wolf Chief of Sector Ministry of Posts and Telecommunications Berlin
- D M. BÖTTCHER Horst
 Scientific Adviser
 Ministry of Posts and Telecommunications
 Berlin
- D M. BROBERG Fritz Scientific Adviser, Ministry of Posts and Telecommunications, Berlin
- D Mme CALOV Hannelore Scientific Adviser, Ministry of Posts and Telecommunications, Berlin
- D M. GÖTZE Herbert Chief of Sector Ministry of Posts and Telecommunications Berlin
- D M. HENSE Uwe Scientific Adviser Ministry of Posts and Telecommunications Berlin

République Démocratique Allemande -German Democratic Republic - República Democrática Alemana (suite)

- D Dr LIEBSCH Winfried
 Scientific Adviser
 Ministry of Posts and Telecommunications
 Berlin
- D Dr MICHEEL Hans-Jürgen Counsellor of Embassy Permanent Representation of the GDR to the UN-Headquarters and to other International Organizations, Genève
- Mme RADEMACHER Sigrid
 Interprète
 Ministerium für Post- und Fernmeldewesen
 Berlin
- S Mme DEMMLER Petra Interprète, Ministerium für Post- und Fernmeldewesen, Berlin

République Populaire Démocratique de Corée - Democratic People's Republic of Korea - República Popular Democrática de Corea

- C M. RYE HYON Kim Director of the Department of the International Relations Ministry of Posts and Telecommunications Pyongyang
- D M. HONG UN SON Translator of the International Department Ministry of Posts and Telecommunications Pyongyang
- D M. ZEN MYEN GUN Senior Instructor Ministry of Communications, Pyongyang
- A M. MOUN GYENG Kim Third Secretary of the Permanent Mission of the Democratic People's Republic of Korea, Genève
- A M. PAK II Bou Second Secretary of the Permanent Mission of the Democratic People's Republic of Korea, Genève

<u>République Socialiste Soviétique</u> <u>d'Ukraine - Ukrainian Soviet Socialist</u> <u>Republic - República Socialista Soviética</u> <u>de Ucrania</u>

C M. SAVANTCHUK V.A. Vice-Ministre des Postes et Télécommunications, Kiev

> Roumanie (Rép. Socialiste de) - Roumania (Socialist Republic of) - Rumania (Rep. Socialista de)

- C M. AIRINEI Gheorghe
 Vice-Ministre, Ministère des Transports et Télécommunications, Bucuresti
- CA M. CEAUSESCU Constantin Directeur général Adjoint Ministère des Transports et Télécommunications, Bucuresti
- CA M. SPATARU Alexandru President of the Roumanian Commission for Space Activity Ministry of Transport and Telecommunications, Bucuresti
- D M. CHIRICÁ Andrei Ingénieur Chef Ministère des Transports et Télécommunications, Bucuresti
- D M. COSTEANU Teodor Ingénieur principal, Ministère des Transports et des Télécommunications Bucuresti
- D M. ENCIU Gheorghe Ingénieur, Ministère des Postes et Télécommunications, Bucuresti
- A M. COSTESCU Aurel Premier Secrétaire, Mission permanente de Roumanie à Genève
- A M. IVASCU Constantin Deuxième Secrétaire, Mission permanente de Roumanie à Genève

Royaume-Uni de Grande-Bretagne et d'Irlande du Nord - United Kingdom of Great Britain and Northern Ireland -Reino Unido de Gran Bretaña e Irlanda del Norte

- C M. BAPTISTE Donald Eric Assistant Secretary, Radio Regulatory Department, Home Office, London
- CA M. CARTER A.O. Assistant Secretary, Broadcasting Department, Home Office, London
- D M. ASHTON R.H.J. First Secretary United Kingdom Mission, Genève
- D M. BOLINGBROKE P.J. Executive Engineer British Post Office, London
- D M. DUNELL Wilfred Maurice Consultant
 Post Office External Telecommunications Executive, London
- D Miss EAGLES Stella Eleanor Higher Executive Officer Radio Regulatory Department Home Office, London
- D M. GILBERT Michael A.E.
 Head of Group
 Network Planning Department
 British Post Office, London
- D M. GODDARD Michael Head of Microwave and Maritime Branch Directorate of Radio Technology Home Office, London
- M. GRAHAM G.A.
 Chief Assistant to Director of Engineering British Broadcasting Corporation London
- D M. HAMMOND Charles W.F. Senior Scientific Officer Cabinet Office, London
- D M. HOWARTH Frederick H.
 Principal, Radio Regulatory Department Home Office, London

Royaume-Uni de Grande-Bretagne et d'Irlande du Nord - United Kingdom of Great Britain and Northern Ireland -Reino Unido de Gran Bretaña e Irlanda del Norte (suite)

- M. MACKIE James R.
 Head of Section TD10.3
 Telecommunications Development Department
 British Post Office, London
- M. PHILLIPS Geoffrey J.
 Head of Radio Frequency Group
 Research Department
 British Broadcasting Corporation
 Tadworth, Surrey
- M. TEMPLE Stephen Robin
 Head of Space Services Branch
 Directorate of Radio Technology
 Home Office, London
- M. WITHAM Alfred Lawrence
 Chief Engineer (Network)
 Independent Broadcasting Authority
 Winchester

<u>Sénégal (Rép. du) - Senegal (Rep. of) -</u> <u>Senegal (Rep. del)</u>

- C M. MBODJI DIONE Alioune Directeur des Télécommunications Office des Postes et Télécommunications du Sénégal, Dakar
- D M. MBAYE Malick Directeur technique Adjoint, Office de Radiodiffusion Télévision du Sénégal Dakar

<u>Singapour (Rép. de) - Singapore (Rep. of</u>) - C Singapur (Rep. de)

- C M. RAJASINGAM R.G. Manager, Regulation Policy Telecommunication Authority of Singapore Singapore
- CA M. TAN Sebastian C.H. Senior Executive Engineer Broadcasting Department Government of Singapore, Singapore

Somalie (Rép. Démocratique) - Somali Democratic Republic - Somalí (Rep. Democratica)

- C M. OSMAN Abdillahi Said Ambassador, Somali Permanent Mission to the United Nations, Genève
- D M. MOHAMED DIRIÉ Ismail Deuxième Secrétaire auprès de la Mission de Somalie, Genève

<u>Soudan (Rép. Démocratique du) - Sudan</u> (Democratic Rep. of the) - Sudan (Rep. Democrática del)

- C M. GAMAL Abdel Wahab Assistant Director General, Department of Telecommunications, Khartoum
- CA M. HAGAHMED Abdulla Sirageldin Assistant Director of Telecommunication Department for Installation Department of Telecommunication, Khartoum
- D M. AHMED Idris Yousif Senior Engineer, Department of Telecommunications, Khartoum
- D Mme ELNAHAS Nafissa Abdalla Electronic Engineer, Sudan B.C. Service Omdurman

Suède - Sweden - Suecia

- C M. ÅSDAL Carl-Gösta Technical Director Swedish Telecommunications Administration Stockholm
 - M. MALMGREN C. Gunnar Chief Engineer, Swedish Telecommunications Administration, Stockholm
- D M. DANIELSSON Sune Head of Section, Ministry of Foreign Affairs, Stockholm
- D M. GARTELIUS Ulf Head of Research Laboratory Sveroges Radio, Stockholm

D

Suède - Sweden - Suecia (suite)

- D M. JAENSSON Rolf B.I. Staff Engineer, Swedish Telecommunications Administration, Stockholm
- D M. PETTERSSON Percy H. Staff Engineer, Swedish Telecommunications Administration, Stockholm
- D M. ROHDIN Arne Director of Engineering Sveriges Radio, Stockholm
- D M. STENBERG Berndt V. Executive Officer, Swedish Telecommunications Administration, Stockholm
- A M. MARTIN-LÖF Johan Senior Technical Adviser Ministry of Industry, Stockholm

<u>Suisse (Confédération) - Switzerland</u> (Confederation of) - Suiza (Confederación)

- C M. STEFFEN Charles Chef de la Division de la Radio et de la Télévision à la Direction générale des PTT, Berne
- CA M. SCHWARZ Ernst Chef de la Section Emetteurs à la Division de la Radio et de la Télévision de la Direction générale des PTT, Berne
- D Dr BERNATH Konrad W. Chef de la Section technique de Radiodiffusion à la Division des Recherches et du Développement de la Direction générale des PTT, Berne
- D M. EBERT Walter Ingénieur diplômé, Chef de la Section Planification et Réception à la Division de la Radio et de la Télévision de la Direction générale des PTT, Berne
- D M. GALLI Pier-Luigi
 Ingénieur ETS à la Division de la Radio et de la Télévision de la Direction générale des PTT, Berne

<u>Suisse (Confédération) - Switzerland</u> (Confederation of) - Suiza (Confederación) (suite)

- Mme HOFER Irène Collaboratrice diplomatique à la Section des Affaires scientifiques internationales de la Direction des Organisations internationales, Entreprise des PTT, Berne
- M. HUMM Bernard
 Ingénieur diplômé à la Division de la
 Radio et de la Télévision de la Direction
 générale des PTT, Berne
- D M. JUNOD Benoit Collaborateur diplomatique Département politique fédéral Berne
- D M. WEY Emil Chef de la Section technique des Liaisons dirigées à la Division des Recherches et du Développement de la Direction générale des PTT, Berne
- M. ZÜST Hansruedi
 Ingénieur diplômé, Directeur technique à
 la Société suisse de Radiodiffusion et
 Télévision, Berne
- A M. PITTET Ernest Directeur technique, Radio-Télévision SSR, Lausanne

<u>Tanzanie (Rép. Unie de) - Tanzania</u> <u>(United Republic of) - Tanzania (Rep.</u> <u>Unida de)</u>

CA M. ODUNGA Stephen Executive Radio Engineer Radio Tanzania, Dar-es-Salaam

> Tchad (Rép. du) - Chad (Rep. of the) -Chad (Rep. del)

- C M. HAMID Kante Ingénieur de Radiocommunication Directeur technique à la Radiodiffusion nationale tchadienne, N'Djamena
- D M. DJASSIBE Tingabaye Directeur des Télécommunications Office national des Postes et Télécommunications, N'Djamena

п

C

<u>Tchécoslovaque (Rép. Socialiste)</u> -<u>Czechoslovak Socialist Republic</u> -<u>Checoslovaca (Rep. Socialista)</u>

- C Ing. JÍRA Jirí
 Ministre-Adjoint des Postes et Télécommunications de la RS tchécoslovaque
 Ministère des Postes et Télécommunications
 Prague
- CA Ing. MARŠÍČEK Jaroslav Chef du Département de Radiocommunication Ministère des Postes et Télécommunications Prague
- D Ing. KRÁLÍK František
 Chef de la Section des Fréquences
 Ministère des Postes et Télécommunications
 Prague
- D M. STRNAD Zdenek Conseiller, Division internationale Ministère des PTT de la RS tchécoslovaque Prague
- A M. JERKOV Jerko Directeur-Adjoint du Centre technique OIRT, Prague
- A M. KODIŠ Václav Ministry of Foreign Affairs Prague
- M. RYVOLA Richard
 Institut des Recherches des PTT
 Ministère des Postes et Télécommunications
 Prague
- M. ZELKO Jaroslav
 Engineer for Scientific and Technology
 Federal Ministry for Scientific and
 Technology, Prague

Thaïlande - Thailand - Tailandia

- C M. P.SAKORN Suchart Acting Chief Office of Frequency Management Post & Telegraph Department, Bangkok
- Mlle CHAPANON Udom
 Chief of Administrative Section
 International Services Division
 Post & Telegraph Department, Bangkok

Thailande - Thailand - Tailandia (suite)

M. PORNSUTEE Kraisorn Chief of Technical Section Office of Frequency Management Post & Telegraph Department, Bangkok

Togolaise (Rép.) - Togolese Republic -Togolesa (Rep.)

- M. AMOUZOUGAH Assiongbor Conseiller technique Ministère de l'Information des Postes et Télécommunications, Lomé
- CA M. MENSAH-ASSIAKOLEY Komessan Chef de la Division Haute Fréquence Radiodiffusion, Lomé
- D M. AHOUANDJINOU Codjo Ingénieur électronicien PTT Togo, Lomé
- D M. AYIKOE Kossivi Inspecteur des Télécommunications PTT, Lomé

Tunisie - Tunisia - Tunez

- C M. BEN HAMIDA Slaheddine Directeur général, Radiodiffusion Télévision Tunisienne, Tunis
- CA M. HADIJI Salah Ingénieur en Chef Directeur des Transmissions Radiodiffusion Télévision Tunisienne Tunis
- M. BEN YOUSSEF Taieb
 Ingénieur divisionnaire, Sous-Directeur
 des Faisceaux Hertziens, Radiodiffusion
 Télévision Tunisienne, Tunis
- D M. GHARBI Hassine Ingénieur, Chef du Service des Etudes des Projets et de la Gestion de Fréquence Radiodiffusion Télévision Tunisienne Tunis
- D M. RIAHI Moncef
 Secrétaire d'Ambassade
 Mission permanente de Tunisie, Genève

Turquie - Turkey - Turquia

- C M. YAVUZALP Ercüment Ambassadeur, Représentant permanent de la Turquie auprès de l'Office européen des Nations Unies et des autres Organisations internationales en Suisse, Genève
- D M. AKYÜZALP Nurgün Directeur-Adjoint du Département des Télégraphes et Téléphones, Direction générale des PTT, Ankara
- D M. ERDEN Doğan
 Conseiller technique du Directeur général
 Radiodiffusion Télévision de Turquie
 Ankara
- D M. ERGINER Kemal Engineer, General Directorate of PTT Ankara
- D M. ERTEM Yalçin
 Directeur-Adjoint du Département de la Planification technique
 Radiodiffusion Télévision de Turquie Ankara
- D M. ESEN Hasmet
 Directeur des Recherches techniques
 Radiodiffusion Télévision de Turquie
 Ankara
- D M. KÖKER Sakir Ingénieur diplômé Radiodiffusion Télévision de Turquie Ankara
- D M. UTKAN Hasim Premier Secrétaire de la Délégation permanente de Turquie auprès de l'Office européen des Nations Unies, Genève

<u>Union des Républiques Socialistes</u> <u>Soviétiques - Union of Soviet Socialist</u> <u>Republics - Unión de Repúblicas Socialistas</u> <u>Soviéticas</u>

C M. BADALOV A.L. Vice-Ministre des Postes et Télécommunications de l'URSS, Ministère des Postes et Télécommunications de l'URSS, Moscou

Union des 1	Républiques Socialistes
Soviétiques	s - Union of Soviet Socialist
Republics -	- Union de Repúblicas
Socialistas	Soviéticas (suite)

- CA M. BORODITCH S.V. Professor, Deputy Director, Radio Research Institute, URSS Administration, Moscou
- CA M. MOTINE E.A. Directeur du Département des Relations extérieures, Ministère des Postes et Télécommunications de l'URSS, Moscou
- D M. BOGDANOV P.A. Ingénieur en Chef, Ministère des Postes et Télécommunications de l'URSS, Moscou
- D M. EPANECHNIKOV K.V. Vice-Directeur du Département Ministère de l'Industrie des Télécommunications, Moscou
- D M. FROLAGINE Aleksandr Assistant to the Minister of Posts and Telecommunications of the USSR, Moscou
- D M. KALACHNIKOV N.I. Professeur, Institut des Télécommunications Moscou
- M. KARABAK I.M.
 Ingénieur principal, Ministère des Postes et Télécommunications de l'URSS, Moscou
- M. KOURAKOV P.S.
 Vice-Directeur du Département
 Ministère des Postes et Télécommunications de l'URSS, Moscou
- M. KRAPOTINE O.S.
 Ingénieur, Ministère des Postes et Télécommunications de l'URSS, Moscou
 - M. KRIVOCHEEV M.I. Professeur, Institut scientifique Moscou
- M. MAIORSKII B.G.
 Ministère des Affaires Etrangères de l'URSS, Moscou

С

D

С

D

D

D

D

<u>Union des Républiques Socialistes</u> <u>Soviétiques - Union of Soviet Socialist</u> <u>Republics - Unión de Repúblicas Socialistas</u> <u>Soviéticas</u> (suite)

- M. ROUSANOV I.V.
 Vice-Directeur du Département
 Comité de Télévision et de Radio de l'URSS, Moscou
- D M. TIMOFEEV V.V. Ingénieur principal, Ministère des Postes et Télécommunications de l'URSS, Moscou
- A M. ISHCHENKO Alexander Senior Engineer of Engineering Department Ministry of Posts and Telecommunications Moscou
- A M. KOZYRITSKI Gleb First Secretary of the Mission of USSR Genève
- ++ Mme GOLLENDER V.N. Dactylographe Ministère des Postes et Télécommunications Moscou

++ Dactylographe

S Mme VAKHROMITCHEVA L.N. Ingénieur, Ministère des Télécommunications Moscou

<u>Uruguay (Rép. Orientale de l') - Uruguay</u> (Oriental Rep. of) - Uruguay (Rep. Oriental del)

- C M. REAL Juan José Embajador, Ministerio Relaciones Exteriores Genève
- CA M. BOSCH Pablo Ministro del Servicio Exterior Ministerio Relaciones Exteriores, Genève

<u>Venezuela (Rép. de) - Venezuela (Rep. of</u>) -<u>Venezuela (Rep. de</u>)

- C M. MARTÍNEZ Carlos Julio Ministerio de Comunicaciones, Caracas
- D M. COOK Leopoldo Ingeniero Electricista, Ministerio de Comunicaciones, Caracas

Yémen (Rép. Arabe du) - Yemen Arab Republic - Yemen (Rep. Árabe del)

- M. TARCICI Adnan Ambassadeur, Délégué permanent auprès de l'Office des Nations Unies et des Organisations internationales à Genève
- CA M. MOGBEL Hussain Radio Engineer, Radio and TV Corporation Sanaa

Yémen (Rép. Démocratique Populaire du) -Yemen (People's Democratic Republic of) -Yemen (Rep. Democrática Popular del)

- M. AZZANI Mohamed Ali Director of Broadcasting Transmission Aden
- M. YAQUOOB Mohamed Director of Microwave Link Project Aden

Yougoslavie (Rép. Socialiste Fédérative de) - Yugoslavia (Socialist Federal Rep. of) - Yugoslavia (Rep. Socialista Federativa de)

- M. DULOVIĆ Ljubomir Directeur de la Direction fédérale des Radiocommunications yougoslaves Beograd
- Dr BOSANAC Toma Professeur à l'Université de Zagreb Zagreb

M. DAVID Janoš Ingénieur principal du Département de Développement, RTV Novi Sad, Novi Sad

- M. GALIC Roman Conseiller technique du Directeur général à la RTV Zagreb, Zagreb
- M. GEORGIEV Branko Ingénieur principal du Département de Développement à la RTV Skopje, Skopje
- D M. HRANISAVLJEVIC Lazar Conseiller de la Radiodiffusion à la Direction fédérale des Radiocommunications yougoslaves, Beograd

Yougoslavie (Rép. Socialiste Fédérative de) - Yugoslavia (Socialist Federal Rep. of) - Yugoslavia (Rep. Socialista Federativa de) (suite)

- D M. LUMEZI Kole Chef technique d'Emission et des Radiocommunications à la RTV Priština Priština
- D M. MLADENOVIC Vladimir Chef de Département pour la Planification de Fréquences à la Direction générale des Radiocommunications yougoslaves, Beograd
- D M. MORI Peter Directeur technique à la RTV Ljubljana Ljubljana
- D M. RADOMAN Vladimir Inspecteur des PTT et Radiocommunications Secrétariat d'Economie de Montenegro Titograd
- D M. RAJIĆ Milija Conseiller supérieur, Direction fédérale des Radiocommunications yougoslaves Beograd
- D Dr STOJANOVIC Ilija Professeur à l'Université de Beograd RTV Beograd, Beograd
- D M. ŽAGAR Vlatko Director of Development Department General Direction of Radiocommunications Sarajevo
- A M. ANGELI RADOVANI Kosta Conseiller, Direction fédérale des Radiocommunications, Beograd
- A M. BRAJAN Berislav Conseiller de la Radiotelevizija Zagreb Direction fédérale des Radiocommunications Beograd
- A M. JANKOVIC Milenko Directeur technique de RTV Beograd Direction fédérale des Radiocommunications Beograd

Zaire (Rép. du) - Zaire (Rep. of) -Zaire (Rep. del)

- C Citoyen MWAMBA KASONGO Ingénieur, Sous-Directeur des Transmissions à la Direction des Etudes et de la Planification, Office national des Postes et Télécommunications Kinshasa
- CA Citoyen YEMBI NSAMPALA Sous-Directeur des Fréquences à la Direction de l'Exploitation des Télécommunications, Office national des Postes et Télécommunications, Kinshasa
- D Citoyen MBONGO-IYEME Conseiller de Cabinet Ministère des PTT Kinshasa
- D Citoyen N'SIALA-MAVAMBU Mbuetete Directeur de l'Exploitation des Télécommunications, Office national des Postes et Télécommunications, Kinshasa

Zambie (Rép. de) - Zambia (Rep. of) -Zambia (Rep. de)

- M. KUMWENDA Chonga Nelson
 Broadcasting Engineer
 Zambia Broadcasting Services, Lusaka
- M. MHENDE Ronald Ronnie
 Assistant Senior Maint. Engineer
 Television Studios and Transmission
 Zambia Broadcasting Services, Lusaka

2. ORGANISATIONS INTERNATIONALES - INTERNATIONAL ORGANIZATIONS - ORGANIZACIONES INTERNACIONALES

- 2.1 <u>Nations Unies United Nations -</u> Naciones Unidas
- C M. PEREK Lubos Chief, Outer Space Affairs Division UN - New York

M. DMITRICHEV Timour Legal Officer, Legal Department UN - New York

M. ZOUPANOS Théodore S. External Relations and Inter-Agency Affairs Officer UN - Genève

Programme des Nations Unies pour l'Environnement - United Nations Environment Programme - Programa de la Naciones Unidas para el medio ambiente

M. BIRYUKOV Gennady Programme Officer United Nations Environment Programme

Bureau du Coordonnateur des Nations Unies pour les secours en cas de catastrophe -Office of the United Nations Disaster Relief Co-ordinator (UNDRO) - Oficina del Coordinador de las Naciones Unidas para el Socorro en Caso de Desastre

M. KORJENKO Viatcheslav Co-Ordinator Officer, UNDRO, Genève

2.2 <u>Institutions spécialisées - Specialized</u> <u>Agencies - Instituciones especializadas</u>

Organisation des Nations Unies pour l'éducation, la science et la culture (UNESCO) - United Nations Educational, Scientific and Cultural Organization (UNESCO) - Organización de las Naciones Unidas para la Educación, la Ciencia y la Cultura (UNESCO)

M. SOMMERLAD E. Lloyd Chief, Division of Free Flow of Information and Communication Policies UNESCO, Paris 2.3 Organisations régionales (Art. 32 de la Convention) - Regional Organizations (Art. 32 of the Convention) -Organizaciones regionales (Art. 32 del Convenio)

Asociación de Empresas Estatales de Telecomunicaciones Andinas (ASETA) -Association des Entreprises d'Etat exploitant des Télécommunications dans le cadre de l'Accord sous-régional des Andes (ASETA) - Association of State telecommunication undertakings of the Andean Sub-regional Agreement (ASETA)

M. COLINA-MARIE Miguel Secretario General, Asociación de Empresas Estatales de Telecomunicaciones del Acuerdo Subregional Andino, Quito Ecuador

Union arabe des télécommunications -Arab Telecommunication Union - Unión Árabe de Telecomunicaciones

M. GHAZAL Maurice Directeur des Services techniques des Télécommunications, Administration générale des Téléphones et Télégraphes, Beyrouth

2.4 <u>Autres Organisations - Others Organizations</u> -<u>Otras Organizaciones</u>

<u>Agence spatiale européenne - European</u> <u>Space Agency - Agencia Espacial Europea</u>

M. ARETS Jean Chef de la Division des Affaires internationales, Paris

Dr KALTENECKER H.K. Directeur de la Division des Affaires internationales, ESA, Paris

M. ROSETTI Calin Mission Manager, Paris

Organisation internationale de radiodiffusion et télévision (OIRT) -International Radio and Television Organization - Organización Internacional de Radiodifusión y de Televisión

M. DEMENTIEV Viktor Yefimovich Director of the Technical Centre, OIRT Prague

M. JERKOV Jerko Directeur-Adjoint du Centre technique Prague Organisation internationale de télécommunications par satellite (INTELSAT) -International Telecommunications Satellite Organization - Organización Internacional de Telecomunicaciones por satélite

Dr PONTANO Benjamin A. Communications Specialist Washington

Organisation internationale de télécommunications spatiales (INTERSPOUTNIK) -International Space Telecommunication Organization - Organización Internacional de Telecomunicaciones Espaciales

M. KRUPIN Jurij Ivanovitsh General Director, International Organization of Space Communications Moscou

<u>Union asiatique de radiodiffusion</u> -<u>Asian Broadcasting Union (ABU)</u> -<u>Unión Asiática de Radiodifusión</u>

M. BALASUBRAMANYAM V. Director of Engineering Kuala Lumpur

Union de radiodiffusion des Etats Arabes -Arab States Broadcasting Union (ASBU) -Unión de Radiodifusión de los Estados Árabes

Dr YOUSIF A.M. Director, Technical Centre, ASBU Khartoum

Union des radiodiffusions et télévisions nationales d'Afrique (URTNA) - Union of National Radio and Television Organizations of Africa - Unión de las Radiodifusiones y Televisiones Nacionales de África

M. NDIAYE Amady Responsable du Centre technique URTNA, Bamako

Union européenne de radiodiffusion (UER) -European Broadcasting Union (EBU) -Unión Europea de Radiodifusión (UER)

M. BROWN Alan Senior Engineer, Bruxelles

M. GRESSMANN Rudolf Director Technical Centre, Bruxelles Union européenne de radiodiffusion (UER) -European Broadcasting Union (EBU) Unión Europea de Radiodifusión (UER) (suite)

٤.

M. de KALBERMATTEN Régis Secrétaire général, Bruxelles

M. KOPITZ Dietmar Ingénieur principal, Bruxelles

M. LEBERT Philippe F.I. Informaticien, Bruxelles

M. MERTENS Henri Ingénieur en Chef, Bruxelles

Union internationale des radio-amateurs -International Amateur Radio Union (IARU) -Unión Internacional de Aficionados de Radio

M. EATON Noël B. Président, Newington, Conn.

M. JOHNSON Bruce Alan Assistant Secretary, Newington, Conn.

M. RÖTTGER Jürgen Assistant International Amateur Radio Union Londres

M. STEVENS Roy F. Secretary International Amateur Radio Union Londres

3. SIEGE DE L'UNION - HEADQUARTERS OF THE UNION - SEDE DE LA UNIÓN

Secrétariat général

M. M. Mili, Secrétaire général

Assistante : Mlle M. Knight

M. R.E. Butler, Vice-Secrétaire général

Assistante : Mlle P. Taillefer

M. R. Cook, Département des conférences et services communs

M. M. Bardoux, Département du personnel

M. R. Prélaz, Département des finances

M. I. Uygur, Département de l'ordinateur

IFRB

M. F.G. Perrin, Président

Assistante : Mme J. Fox

M. C.W. Sowton, Vice-Président

Assistante : Mlle M. Iglesias

M. A. Berrada, membre <u>Assistante</u> : Mme D. Phené

M. S. Fujiki, membre Assistante : Mme A. Tovey

M. A. Gromov, membre

Assistante : Mme M. Zinovieff

M. W. García-Rios, ingénieur

M. C. Glinz, ingénieur

CCIR

M. R.C. Kirby, Directeur

Assistante : Mme S. Dherin

M. N.V. Gadadhar, conseiller supérieur

M. Y.Y. Mao, conseiller supérieur

M. R.F. Froom, conseiller

M. A.W. Boyle, ingénieur

Assistante : Mme K. Wagner

CCITT

M. L. Burtz, Directeur

M. I. Savitsky, conseiller supérieur

M. F. Bigi, ingénieur

- 4/1 -

Secrétaire de la Conférence: M. M. Mili, Secrétaire généralSecrétaire exécutif: M. A. Winter-JensenAssistant: M. P.A. Traub (17-21.I.1977)Secrétaire technique: M. A.A. MattheyConseiller juridique, chargédes relations extérieures: M. D.N. Tonou Zekri

4.2 <u>Séances plénières, commissions</u>

Séance plénière : M. H. Pouliquen

Commission	1	:	М.	Η.	Pouliquen
**	2	:	М.	A.	Winter-Jensen
11	3	:	М.	J.	Schuwey
19	8	:	Μ.	R.	Macheret

Assistantes

-	Mme Ch.	Boccard
-	Mlle R.	Einhorn
-	Mlle S.	Frankel
-	Mlle F.	Maurice
-	Mlle D.	Service
-	Mme J. 1	Teyssier

4.3

4.1

Division technique

n

**

M. A.A. Matthey, Secrétaire technique

Commission 4 : M. J. Rutkowsky Assisté de M. M. Ahmad

- 5 : M. D. Kane
- 6 : M. R. Pluss

Assistants

- M. J.P. Berthet
- M. J.J. Bozonnet
- M. I. Dolezel
- M. F. Ekman
- M. T. Horie
- M. L.S. Huang
- M. M. Khabiri
- M. J.J. Koehli
- M. P. Korovenkov
- M. G. Kovacs
- M. R. Macheret
- M. D. Nasution
- M. G. Renn
- Mme M. Sage
- M. M. Sant
- M. L. Sonesson
- M. S. Tsukada
- M. J. Wyss

£,

14

.+

- Mlle M.L. Arocena
- M. T. Bahi
- Mlle M. Bieri
- Mme R. Reinhard
- Mme M.C. Revenga
- M. G. Serlooten
- M. A. Sigrist

4.4

Division "Réarrangement du Règlement des radiocommunications"

Commission 7 : M. A. Zaccagnini

Assistants

- M. J. Pelegri - Mile S. Peter
- Mile G. Soto Robles
- M. I. St.Q. Severin
- M. P.A. Traub

4.5 Division "Services de la Conférence"

Service des délégués, de l'interprétation et de la documentation	:	M. U. Petignat
Relations avec la presse/Information publique	:	M. R. Fontaine
Division linguistique	:	M. J. Revoy
- traduction française	:	M. M. Brodsky
- traduction anglaise	:	M. T. Jones
- traduction espagnole	:	Mlle M.A. Delgado
Service des interprètes	;	Mme M. Johner
Service des procès-verbalistes	:	Mlle J. Barley
Salles	:	Mme M. Grand
Contrôle des documents	:	Mme L. Jeanmonod
Dactylographie	:	M. J. Escudero
Reprographie	:	MM. R. Probst P. Constantin A. Schaffner
Renseignements	:	Mme M.M. de Rejod
Economat	:	M. J. Barreau
Distribution des documents	:	M. W. Gantert
Messagers	:	M. C. Glappey
Huissiers	:	M. G. Brunet

INTERNATIONAL TELECOMMUNICATION UNION

BROADCASTING SATELLITE CONFERENCE

Document No. 388-E 4 March 1977

\ _

(Geneva, 1977)

FINAL LIST OF DOCUMENTS

A. Basic Documents of the Conference

	Doc. No.		Doc. No.
Conference Chairmen and Vice-Chairmen	46(Rev.)	<u>Committee 3</u> (Budget Control Committee)	
<u>Terms of reference of</u> <u>Committees</u> <u>List of participants</u>	47 387	<u>Summary Records</u> lst Meeting 2nd " 3rd "	74 193 263
<u>Plenary Meeting</u>		<u>Report</u> Report	217
Ist Meeting 2nd " 3rd " 4th " 5th " 5th " 6th " 7th " 8th " 9th " 10th " 11th " 12th " 13th " Signing Ceremony Committee 1 (Steering Committee)	81 95 167 199 251+Corr.1 293 314 352 353 382 383 382 383 384 385 386	Committee 4 (Technical) Summary Records 1st Meeting 2nd " 3rd " 4th " 5th " 6th " 7th " 8th " 9th " 10th " 12th " 13th "	82 94 145 147 182 202 211 267 288 289 290 374 375
<u>Committee 2</u> (Credentials Committee)		<u>Reports</u> 1st Report	108(Rev.)
<u>Summary Records</u> 1st Meeting 2nd "	72 271+Corr.	2nd " 3rd " 4th "	122 159 177+ Corr.1,2
<u>Report</u> Report	241	ουπ "	243+ Corr.1,2,3
	- - 		

ARCHII U.I.T. GENEN

	Doc. No.		Doc. No.
Committee 5 (Planning) Summary Records lst Meeting 2nd " 3rd " 4th " 5th " 6th "	136 137 183 239 264 376	<u>Committee 6</u> (Procedures)(cont.) <u>Report</u> Report <u>Committee 7</u> (Rearrangement of the RR) Summony Records	101
7th " 8th " 9th " 10th " 11th " <u>Reports</u>	377 378 379 380 381	Ist Meeting 2nd " 3rd " <u>Report</u> Report	118 155+Corr. 184 RR7
lst Report 2nd " <u>Committee 6</u> (Procedures) <u>Summary Records</u>	255 270 ·	<u>Committee 8</u> (Editorial) <u>Summary Record</u> lst Meeting	146
lst Meeting 2nd " 3rd " 4th " 5th " 6th " 7th " 8th " 9th "	83 114 115 119 203+Add. 244+Corr. 292 321 354		

B. The attached Annex 1 contains the complete List of Documents in numerical order.

C. The List of Documents of the "RR" series is shown in Annex 2

Document No. 388-E Page 3

PL = Plenary Meetings C = Committee WG = Working Group

ANNEX 1

LIST OF DOCUMENTS

No.	Origin	Title	Destination
l + Corr.	SG	Agenda of the Conference	PL
2 +Corr. +Add.1,2	SG	Report by the Joint Working Party of the CCIR Study Groups on the Conference for the Planning of the Broadcasting-Satellite Service in the 12 GHz Band	C.4,C.5 and C.6
3	SG	Credentials of delegations	C.2
λţ	SG	Possible re-arrangement of the Radio Regulations and the additional Radio Regulations	C.7
5	Chile	Proposals on the agenda	C.4,C.5 and C.6
6 +Corr.1,2_	SG	Final Report by CCIR Working Party PLEN./2	C.4 and C.5
7 +Add.1,2,3	United States of America	Proposals for the work of the Conference	C.4,C.5 and C.6
8	France	General technical principles proposed to the World Administrative Radio Conference for the Planning of the Broad- casting-Satellite Service in Frequency Bands 11.7 - 12.2 GHz (in Regions 2 and 3) and 11.7 - 12.5 GHz (in Region 1)	C.4 and C.5
9	Canada	Proposals for the work of the Conference	C.4,C.5 and C.6
10	Switzerland	The need for the Conference to establish a frequency assign- ment/orbital position plan	C.4,C.5 and C.6
11 +Corr.	Italv	Regulatory procedures	C.6
12	New Zealand	Proposals for the work of the Conference	C.4,C.5 and C.6

· · · · · · · · · · · · · · · · · · ·			
No.	Origin	Title	Destination
13 +Corr.	New Zealand	World-wide satellite broadcasting planning considerations	C.4 and C.5
14	USSR	Proposals for the work of the Conference	C.4,C.5 and C.6
15 +Corr.1,2	Fed.Rep.of Germany	Procedure for the determination of the limiting power flux-density at the boundary of a country operating a Broadcasting-Satellite Service and for predicting the power flux-density produced by a terrestrial station in the band 11.7 - 12.2 GHz (12.5 GHz in Region 1)	C.4 and C.5
16 +Add.1,2,3	SG	List of foreseeable requirements for the Broadcasting-Satellite Service in the 12 GHz band	C.5
- 17	United Kingdom	Use of the geostationary orbit by broadcasting satellites in the band 11.7 - 12.2 GHz (12.5 GHz in Region 1) and by Region 2 fixed satellites in the band 11.7 - 12.2 GHz	C.4 and C.5
18	Japan	Proposals for the work of the Conference	C.4,C.5 and C.6
. 19 + Corr	SG	Convening of the Conference	$_{\rm PL}$
20	SG	Contributions of unexempted Recognized Private Operating Agencies and International Organizations	C.3
21	SG	Budget for the Conference	C.3
22	France	Use of energy dispersal in the Broadcasting-Satellite Service	с.4
23	Liberia	Proposals for the work of the Conference	C.4 and C.5
24	Mexico	Proposals for the Conference	C.4,C.5 and C.6
25 + Corr.1,2,3	SG	Invitations to the Conference	PL
26 +Add.	SG	Notifications to International Organizations	PL
27	SG	Report of the IFRB	C.4,C.5 and C.6

No.	Origin	Title	Destination
28(Rev.) + Corr.1,2,3, 4,5,6,7	SG	Situation of countries with respect to the International Telecommunication Convention	PL
29(Rev.)	Kenya	Proposals for the work of the Conference	C.4,C.5 and C.6
30(Rev.)	Nigeria	Proposals for the work of the Conference	C.4,C.5 and C.6
31	France	Contribution to the study of reception characteristics	C.4 and C.5
32	France	Contribution to the study of general reception configurations	C.4 and C.5
33	Brazil	Proposals for the work of the Conference	C.5 and C.6
34	Australia	Principles of radio-frequency (RF) channel selection in the 12 GHz band	C.4 and C.5
35	Australia	Possible sharing of the band 11.7 - 12.5 GHz between the Broadcasting-Satellite and Terrestrial Services	c.4
36	Australia	Earth-space links for the Broad- casting-Satellite Service - draft Recommendation	C.4
37	Australia	Recommended changes to the Radio Regulations	c.6
38	Australia	Studies of interference from a 12 GHz Broadcasting-Satellite Service into a line-of-sight radio-relay system	C.4
39	Australia	Areas of concern to users of the Fixed-Satellite Service; out-of- band emissions below 11.7 GHz	C.4 and C.5
40	Denmark	Proposals for the work of the Conference	C.4
4 <u>1</u>	Norway	Spurious emission of a broad- casting-satellite repeater	C.4

٠

.

д

t,

Annex 1 to Document No. 388-E Page 6

No.	Origin	Title	Destination
42	Venezuela	Proposals for the work of the Conference	C.4 and C.5
43	Yugoslavia	Proposals for the work of the Conference	C.5
կկ	Italy	Basis of determining whether a service of an administration is considered to be affected by a proposed modification to the plan	c.6
45	New Zealand	Additional world-wide satellite broadcasting planning considerations	C.5
46(Rev.)	PL	Conference Chairmen and Vice- Chairmen	
47	$_{\rm PL}$	Terms of reference of committees	-
48	SG	Allocation of documents (Nos. 1 to 39)	$_{\rm PL}$
49	Mexico	Frequency allocations - 10 kHz to 275 GHz	c.6
50 + Corr.	USSR	The maximum power flux-density produced by a space station in the Broadcasting-Satellite Service at the edge of the service area	с.4
51	USSR	The figure of merit of receiving equipment for individual reception at about 12 GHz	C.4
52	India	Proposal for the work of the Conference	C.5
53	India	Preferred orbits for tropical countries	C.4
54	Egypt	Comments on Doc. No. 2 (CCIR)	C.5

-

No.	Origin	Title	Destination
55	SG	Memorandum concerning the terms of	PL
		reference of the World Administrative Radio Conference for the planning of the Broad- casting-Satellite Service in the 12 GHz frequency band	
56	Iran - Senegal	Election of an additional Vice- Chairman to the Conference	$_{\rm PL}$
57	SG	Secretariat of the Conference	-
58	France	Contribution to the choice of diameter of the reference receiving antenna for purposes of planning	C.4 and C.5
59	C.5	Confirmation of requirements	C.5
60	Cuba	Proposals for the work of the Conference	C.5
61	Sudan	Attenuations due to rain as obtained from meteorological data using CCIR reports 233, 563 and 564	с.4
62	Sudan	Some technical considerations in connection with multibeams covering the same country	c.4
63	Chile	Comments on Document No. 5, point 2.1	C.4
64	Chile	Comments on Document No. 5, point 2.2	C.5
65	Chile	Comments on Document No. 5, point 2.3	ć.6
66	WG 4A	Maximum dimension of an antenna beam in the Broadcasting- Satellite Service	с.4
67 ·	WG 4A	Antenna reference patterns	c.4
68	WG 4A	Protection ratio between two frequency-modulation television systems	c. 4
J			

.

Annex 1 to Document No. 388-E Page 8

No.	Origin	Title	Destination
69	WG 4A	Type of modulation, bandwidth and channel spacing	C.4
70(Rev.)	C.6	Organization of the Working Groups	C.6
71	Fed. Rep. of Germany	Pointing accuracy of a broadcasting-satellite transmitting antenna	C.4
72	C.2	Summary record of the first meeting of Committee 2 (Credentials)	C.2
73	Japan	Computer programme for planning the Broadcasting-Satellite Service in the 12 GHz band	C.5
74	C.3	Summary record of the first meeting of Committee 3 (Budget control)	C.3
75	WG ЦВ-Ц	Minimum size of a receiving antenna for the Broadcasting- Satellite Service	WG 4B
76	Spain	Foreseeable requirements for the Broadcasting-Satellite Service in the 12 GHz band for the Principality of Andorra	PL
77	WG 4A	Polarization of emissions	C.4
78	WG 4A	Tuning range of television receivers	C.4
79	WG 4A	Station keeping of satellites	с.4
80	WG 4A	Angle of elevation of receiving antennae	C.4
81 + Corr.	PL	Minutes of the first plenary meeting	PL
82	с.4	Summary record of the first meeting of Committee 4	C.4

No.	Origin	Title	Destination
83	c.6	Summary record of the first meeting of Committee 6	C.6
84	C.5	Structure of Committee 5	C.5
85	WG 4A-2	Radio propagation factors	C.4
86(Rev.)	WG 4A-1	Pointing accuracy of satellite antenna	C.5
87	Sudan	The effects of dust storms on microwave propagation	WG 4A-2
88	WG 6C	Report of Working Group 6C to Committee 6	C.6
89	WG 4B	Protection ratios and other primary specifications of maximum allowable interference between services	C.4
90	WG 4B	Information regarding the use of energy dispersal in planning the Broadcasting-Satellite Service	с.4
91	WG 4B	Factors to be considered in the choice of polarization for planning the Broadcasting- Satellite Service	c.4
92	WG 4B	Power flux-density at edge of service area	с.4
93	WG 4B	Figure of merit (G/T) of a broadcasting-satellite receiving station	C.4
94	C.4	Summary record of the second meeting of Committee 4	с.4
95	PL	Minutes of the second plenary meeting	PL
96(Rev.2)	WG 4B-3	Out-of-band spurious emissions from broadcasting-satellites - planning considerations	C.4
No.	Origin	Title	Destination
--------------------------------------	---------	---	--------------
97	WG 4A	Difference between the e.i.r.p. directed towards the edge of the coverage area and that on the axis of the beam	c.4
98	USA	Safeguarding future access to the geostationary orbit	C.5
99(Rev.)	₩G 4B-2	Sharing criteria between space services within Region 2	c.4
100	SG	List of documents (1 to 100)	-
101	c.6	First report of Committee 6 relating to the form of the Final Acts of the Conference	PL
102	WG 4B	The use of energy dispersal in planning for the Broadcasting-Satellite Service	c.4
103(Rev.2) +Annex 3 + Corr.1,2	C.5	List of requirements	C.5
1.04	WG 4A3	Definitions	WG 4A
105(Re v.)	WG 4A	Radio propagation factors	C.4
106	WG 4B	Characteristics of broadcasting-satellite receiving systems for community reception	с.4
107	WG 4B3	Out-of-band spurious emissions, draft recommendation	c.4
108(Rev.) + Corr.1,2	C.4	Technical parameters required for planning	C.4,C.5 & PL
109	WG 4B3	Up-link considerations for the Broadcasting-Satellite Service	C.4
110	WG 5B	Planning principles	C.5
111(Rev.3) + Corr.	WG 4B	Protection requirements for sharing between services in the 12 GHz band	C.4
112	WG 4B	Choice of polarization to be used in planning for the Broadcasting-Satellite Service	C.4
113(Rev.)	WG 4B	Out-of-band spurious emissions from broadcasting satellites, planning consideratio	C.4 ns
114 + Corr.	c.6	Summary record of the second meeting of Committee 6	c.6 ·

....

.

٠

No.	Origin	Title	Destination
115	c.6	Summary record of the third meeting of Committee 6	C.6
116	Egypt	Proposals concerning the work of the Conference	PL
117(Rev.)	WG 4A	Cross-section of antenna beams	с.4
118	C.7	Summary record of the first meeting of Committee 7	C.7
119	C.6	Summary record of the fourth meeting of Committee 6	c.6
120	Spain	Foreseeable requirements for the Broadcasting-Satellite Service in the 12 GHz band for the Principality of Andorra	PL
121 + Corr.	Colombia, Congo, Ecuador, Gabon, Kenya, Uganda, Zaire	Equatorial countries	C.5 and C.6
122	C.4	Technical parameters for planning (Report No. 2)	C.5 and PL
123 + Add.	C.5	Instructions for the Sub-Working Group of Group 5A on orbit position assignments to be used in the establishment of the Plan	WG 5
124	C.5	Intentional extended service area	WG 5
125	Sudan	Maximum antenna beamwidth corresponding to the feasible satellite output power	C.4 and C.5
126 + Add.	SG	Situation concerning expenditure for the Broadcasting-Satellite Conference at 21 January 1977	C.3
127	Yugoslavia	Basic consideration on the planning principles	C.5 and PL
128	C.4	Draft CCIR Recommendation on out-of-band spurious emissions	PL
129(Rev.)	WG 4A	Limitation of output power in the satellite transmitter	с.ч

No.	Origin	Title	Destination
130(Rev.)	WG 4A	Effects of propagation on cross-polariza- tion	C. 4
131	SG	Note by the Secretary-General (Symbols designating countries of a geographical area)	
132	WG 4A3	Definitions	WG 4A
133(Rev.)	c.8	B.1	$_{ m PL}$
134	WG 5A3	Note from Sub-Working Group 5A3 on the beams to be used for the first planning studies under its terms of reference	WG 5A
135	WG 5A3	Note from Sub-Working Group 5A3 on the provisional proposals for the orbit positions	WG 5A
136 + Corr.	C.5	Summary record of the first meeting of Committee 5	C.5
137	C.5	Summary record of the second meeting of Committee 5	C.5
138	WG 5A3	Problems of the parts of Regions 1 and 3 with the highest density of requirements in the orbital arc	WG 5A
139	WG 5A2	Progress report, Sub-Working Group 5A2	WG 5A
140	WG 5A2 and 5A3	Report of joint meeting of Sub-Working Groups 5A2 and 5A3 on planning parameters	WG 5A
141	WG 6B2	Procedures for coordination, notification and registration for stations of terrestrial services in the bands 11.7- 12.2 GHz (in Regions 2 and 3) and 11.7- 12.5 GHz (in Region 1) in the case broadcasting-satellite stations are involved	WG бв
142	France	Note from the French Delegation to the Broadcasting-Satellite Conference	PL
143	United States of America	Contribution to the question of intraser- vice sharing in the Broadcasting-Satellite Service	с.4
144	WG 4B3	Draft Recommendation to the CCIR on up- link considerations	WG 4B
145	c.4	Summary record of the third meeting of Committee 4	C.4

No.	Origin	Title	Destination
146	C.8	Summary record of the first meeting of Committee 8	C.8
147	C.4	Summary record of the fourth meeting of Committee 4	C.4
148	WG 4B3	Up-link carrier-to-interference ratio - Planning consideration	WG 4B
149	C.5	Note by the Chairman of Committee 5 List of test points	C.5
150	WG 5A2	Report of Sub-Working Group 5A2	WG 5A
151	WG C.2	Credentials	C.2
152	WG 6D	Terms of reference of Working Group 6D	c.6
153	WG 482 and 484	Reference antenna diameter for a fixed- satellite earth station for calculating interference from satellites in the Broadcasting Service	WG 4B
154	Chairman	Note by the Chairman of the Conference (Replacement of Vice-Chairman)	-
155 + Corr.	C.7	Summary Record of the second meeting of Committee 7	C.7
156	C.4	Note from the Chairman of Committee 4 to Committee 6	C.6
157(Rev.2)	WG 4A	Relation between the antenna beamwidth and the feasible output power from a satellite	c.4
158	Venezuela	Proposals for planning the Broadcasting- Satellite Service in Region 2	C.5
159	c.4	Third report of Committee 5 (Technical parameters for planning)	C.5, C.6 and PL
160	Cuba	Proposals for the work of Committee 5	C.5
161(Rev.)	WG 4B3	Spurious emissions - Power flux-density levels, planning considerations	WG 4B
162	WG 4B3	Information relating to up-links for the Broadcasting-Satellite Service	WG 4B
163	WG 4B3	Draft Recommendation to Administrations and to the CCIR on up-links for the Broadcasting-Satellite Service	WG 4B

•

No.	Origin	Title	Destination
164(Rev.)	WG 4B3	Draft recommendation to the CCIR on the implementation and operation of up-links for the Broadcasting-Satellite Service	WG 4B
165	Colombia, Ecuador, Kenya, Uganda, Zaire	Planning of the orbit	C.5
166	WG 4B	Interregional sharing	WG 4B
167	PL	Minutes of the third Plenary Meeting	PL
168(Rev.)	WG 4B	Interregional sharing criteria to protect the Fixed-Satellite Service against interference from the Broadcasting- Satellite Service organized in a priori plan	с.4
169(Rev.) + Corr.	WG 4B	Criteria for sharing between the Broadcasting-Satellite Service and Terrestrial Services	C.4
170	WG 5A3	Report of Sub-Group 5A3	WG 5A
171	Comoros	Transfer of right to vote	-
172 + Add. 1 & 2	C.5	Annex 3 (revised) to the List of requirements	C.5
173	Australia	The interdependence of receiver design, channel grouping and sharing criteria	C.4 and C.5
174(Rev.)	WG 6A1	Report of Sub-Working Group 6Al	WG 6A
175	c.8	B.2	PL
176		(This number has not yet been allocated)	
177+Corr.1,2	C.4	Fourth Report of Committee 4 (Technical parameters for planning)	C.5, C.6 and PL
178	C.5	Note by the Chairman of Committee 5	Working Groups C.5
179	Iceland, Sweden	Proposal for further work of Sub-Group 5A2	WG 5A
180	WG 4B2	Interference within and between space services using the geostationary satellite orbit in the frequency band 11.7 - 12.2 GHz	WG 4B
l	1		

No.	Origin	Title	Destination
181	Australia	Sovereignty of the geostationary orbit	C.4, C.5 and C.6
182	C.4	Summary record of the fifth meeting of Committee 4	C.4
183	C.5	Summary record of the third meeting of Committee 5	C.5
184	C.7	Summary record of the third meeting of Committee 7	C.7
185 + Corr.	WG 6B	First Report of Working Group 6B	с.б
186	Argentina, Cuba, Mexico, Paraguay, Venezuela	Changes to be made in Document DL/22(Rev.1) C.5
187	WG 6A	Report of Working Group 6A	c.6
188(Rev.2)	Drafting Group 4B	Interregional sharing criteria	C.4
189 + Add.	C.5	Minimum requirements	C.5
· 190	с.4	Note from the Chairman of Committee 4 to the Chairman of Committee 6	C.6
191	с.4	Note from the Chairman of Committee 4	C.5
192	с.4	Note from the Chairman of Committee 4	c.6
193	C.3	Summary record of the second meeting of Committee 3	C.3
194	c.6	Note to the Chairman of Committee 4 from the Chairman of Committee 6	C.4
195	WG 5A2	Proposals for improving the planning work for final Plan	WG 5 A
196	WG 6A2	Report of Sub-Working Group 6A2	WG 6A
197(Rev.)	с.4	Note from the Chairman of Committee 4	C.4
198	France	Note by the French Delegation (relative to the Principality of Andorra)	-
199	PL	Minutes of the fourth Plenary Meeting	$_{\rm PL}$
200	-	List of documents (101 to 200)	-

No.	Origin	Title	Destination
201	Mauritius	Statement	PL
202	C.4	Summary record of the sixth meeting of Committee 4	C.4
203 + Add.	C.6	Summary record of the fifth meeting of Committee 6	C.6
204(Rev.)	WG 5B	First report of Working Group 5B	C.5
205	Spain	Note by the Spanish Delegation to the Broadcasting-Satellite Conference	-
206	Morocco, Tunisia, Mauritania, Egypt, Sudan, Syria, Lebanon, Jordan, Yemen (A.R.), Yemen (P.D.R. of), Kuwait, Qatar, United Arab Emirates, Bahrain, Oman, Pakistan, Malaysia, Bangladesh, Indonesia, Uganda, Senegal, Chad, Niger	Islamic programme coverage	PL
207	Chairman	Requirements to be taken into consider- ation for planning in Regions 1 and 3	PL.
208	C.5	Intentional extended service area	WG 5A & C.5
209	Saudi Arabia	(Multiple coverage beams)	
210	C.6	First series of texts adopted by Committee 6	c.8
211	C.4	Summary record of the seventh meeting of Committee 4	C.4
212	WG 5A5	Permitted characteristics of assign- ments in the Broadcasting-Satellite Service in the band 11.7-12.5 GHz	WG 5A
213	WG 2 ·	2nd report (Credentials)	C.2
214	C.4	Note from the Chairman of Committee 4	C.6
215	United Kingdom	Provisions to regulate the services in the 12 GHz band	C.6

¢

.

No.	Origin	Title	Destination
216 + Add.	4B Editorial Group	Information concerning the power flux-density limit necessary to protect the Fixed-Satellite Service in Region 2	WG 4B
217	C.3	Final report of the budget control committee to the Plenary Meeting	PL
218	United States	Suggested amendment to Document No. 204	WG 5B
219	United Arab Emirates	Proxy	-
220	WG 6A	Second report of Working Group 6A	C.6
221	WG 6B	Second report of Working Group 6B	C.6
222	Viet Nam	(Archip ela gos of the Paracels and Spratleys)	-
223 + Corr.	Colombia	(Concepts of the geostationary orbit)	C.5 & C.6
224(Rev.)	C.8	B.3(Rev.)	PL
225 + Add. 1, 2	c.8	B.4	PL
226	WG 5B	Second report by Working Group 5B	C.5
227 + Add.	WG 6D	Resolution relating to the updating of the master international frequency register for Regions 1 and 3 on the date of entry into force of the Final Acts	C. 6
228	Turkey	Islamic programme coverage	PL
229	Ecuador	Geostationary orbit	$_{\rm PL}$
230	United States	Power flux-density limits to initiate coordination	C.4 & C.6
231 + Add.	C.8	B.5	PL
232	c.8	в.б	PL
233	C.5	Antenna beam characteristics	C.5
234	C.8	B.7	PL
235	C.8	в.8	PL
236	Comoros	Final protocol	PL
237	Indonesia	Final protocol	PL
238	Iran	Islamic programme coverage	PL
239	C.5	Summary record of the fourth meeting of Committee 5	C.5

ŋ

)

3

No.	Origin	Title	Destination
240	France	Final protocol	PL
241	C.2	Report of Committee 2	PL
242	c.8	B.9	PL
243 + Corr. 1,2,3	C.4	Fifth report by Committee 4	C.5, C.6 and PL
244 + Corr.	c.6	Summary record of the sixth meeting of Committee 6	c.6
245	France	Final protocol	PL
246	C.8	B.10	PL
247	United Kingdom	Protection of terrestrial services in Regions 1 and 3	PL
248	Panama	Statement concerning frequency registrations for the Broadcasting- Satellite Service in its territory, including the Panamanian territory known as the Canal Zone	PL
249	Indonesia	Reflections on state sovereignty in space	-
250 _.	Australia	Draft Recommendation to the CCIR relating to the use of the 12.2-12.5 GHz band	C.5
251 + Corr.	PL	Minutes of the fifth Plenary Meeting	PL
252	Ivory Coast	Final Protocol	PL
253	United Kingdom	Draft resolution relating to the use of experimental broadcasting satellites by administrations in Regions 1 and 3	C.5
254(Rev.)	WG 5B	Third report of Working Group 5B to Committee 5	C.5
255 + Corr.	C. 5	First report by Committee 5	PL
256 + Corr.	WG 5A2/3	Report of Sub-Working Group 5A2/3	C.5
257	C.8	R.l	PL
258	Zaire	Final protocol	PL
259	Iran	Final protocol	PL
260	Afghanistan	Final protocol	PL
261	C.8	B.11	PL
262	C.8	R.2	$_{\rm PL}$
263	C.3	Summary record of the third and last meeting of Committee 3	C.3

·

4

۲

ł

		1	···· ··· ··· ···
No.	Origin	Title	Destination
264	C.5	Summary record of the fifth meeting of Committee 5	C.5
265	WG 5A-2/3	Draft plan for Regions 2 and 3	WG 5A & C.5
266(Re v.)	Byelorussia, Bulgaria, Hungarian (P.R.), Mongolia, Poland, GDR, Ukraine, Czechoslovakia, USSR	Final protocol	PL
267	C.4	Summary record of the eighth meeting of Committee 4	C.4
268	C.8	B.12	$_{\rm PL}$
269	SG	Recapitulatory list of final protocols	$_{\rm PL}$
270	C.5	Second report of Committee 5	$_{\rm PL}$
271 + Corr.	C.2	Summary record of the second meeting of Committee 2	C.2
272	United States, France, United Kingdom	Summary record of the second meeting of Committee 2 - Document No. 271	PL
273	Germany (F.R.of)	Summary record of the second meeting of Committee 2 - Document No. 271	$_{\rm PL}$
274	Zambia	Final protocol	$_{\rm PL}$
275	Bangladesh	Final protocol	PL
276	Mauritania	Final protocol	$_{\rm PL}$
277	Algeria	Islamic programme coverage	PL
278	India	Final protocol	PL
279	Mauritius	Final protocol	PL
280	Liechtenstein	Final protocol	PL
281	Congo	Final protocol	PL
282	Upper Volta	Final protocol	PL
283	Australia	Final protocol	PL
284	Guatemala	Final protocol	PL
285	Ghana	Final protocol	$_{\rm PL}$
286	Nigeria	Final protocol	PL
287	WG 5A5	Report of Sub-Working Group 5A5	C.5

2

\$

)

3

-

No.	Origin	Title	Destination
288	C.4	Summary record of the ninth meeting of Committee 4	C.4
289	c.4	Summary record of the tenth meeting of Committee 4	C.4
290	C.4	Summary record of the eleventh meeting of Committee 4	c.4
291	Philippines	Final protocol	$_{\rm PL}$
292	С.б	Summary record of the seventh meeting of Committee 6	C.6
293	PL	Minutes of the sixth plenary meeting	$_{\rm PL}$
294	Oman	Final protocol	$_{\rm PL}$
295(Rev.2)	Germany (F.R.of), Austria, Belgium, Canada, Denmark, United States, Finland, France, Ireland, Italy, Japan, Luxembourg, Monaco, Norway, New Zealand, Netherlands, Portugal, United Kingdom, Sweden	The geostationary orbit	PL
296(Rev.) + Add.	c.8	B.13(Rev.)	PL
297	C.8	B.14	$_{\rm PL}$
298	C.8	B.15	$_{\rm PL}$
299	Mali	Final protocol	PL
300	c.6	Note from the Chairman of Committee 6	$_{\rm PL}$
301	Kenya	Final protocol	PL
302 + Corr.	C.8	R.3	PL
303	Bangladesh	Final protocol	$_{\rm PL}$
304	Senegal	Final protocol	$_{\rm PL}$
305	Benin	Final protocol	PL
306	C.8	R.4	PL
307	Togo	Final protocol	PL
308	Guinea	Final protocol	PL

(

ŧ.

.

No.	Origin	Title	Destination
309	Australia, New Zealand, Papua New Guinea	Final protocol .	PL
310	Bolivia	Final protocol	PL.
311	Saudi Arabia, Bahrain, Egypt, United Arab Emirates, Kuwait, Morocco, Mauritania, Oman, Yemen (A.R. Yemen (P.D.R. of)	Final protocol	PL
312	Algeria, Saudi Arabia, Kuwait, Morocco, Sudan, Tunisia, Yemen (A.R.)	Final protocol	PL
313	Algeria, Saudi Arabia, Bahrain, Bangladesh, Egypt, United Arab Emirates, Kuwait, Malaysia, Morocco, Oman, Pakistan, Sudan, Tunisia, Yemen (A.R.), Yemen (P.D.R. of)	Final protocol	PL
314	PL	Minutes of the seventh plenary meeting	PL
315	Tanzania	Final protocol	PL
316	Brazil	Final protocol	PL
317	Sudan	Final protocol	PL
318	Yemen (P.D.R.of)	Final protocol	PL
319	Central African Empire	Final protocol	PL
320	Bahrain, United Arab Emirates	Final protocol	PL
321	C.6	Summary record of the eighth meeting of Committee 6	C.6

•

٦

3

No.	Origin	Title	Destination
322	China	nina Final protocol	
323	Lao P.D.R. Final protocol		PL
324	Saudi Arabia	Final protocol	PL
325	Pakistan	Final protocol	PL
326	Byelorussia, Bulgaria, Hungarian P.R., Poland, German D.R., Ukraine, Czechoslovakia, USSR	ussia, Final protocol ia, ian P.R., , German Ukraine, oslovakia,	
327	Thailand	Final protocol	$_{\rm PL}$
328	Burundi Final protocol		PL
329	Chad	Final protocol	PL
330	Ethiopia	Final protocol	PL
331	Colombia, Congo, Ecuador, Gabon, Kenya, Uganda, Zaire	Final protocol	PL
332 `	Brazil	Final protocol	PL
333	Algeria	Final protocol	PL
334	Venezuela	Final protocol	PL
335+Corr.	C.5	Draft Plan for Regions 1 and 3	C.5
336	Uruguay	Final protocol	PL
337	Bulgaria	Final protocol	PL
338	Panama	Final protocol	PL
339	Argentina	Final protocol	· PL
340	Korea (Rep. of)	Final protocol	$_{\rm PL}$
341	Pakistan	Final protocol	PL
342	India	Final protocol	PL
343	Brazil	Final protocol	PL
344	Korea D.P.R.	Final protocol	PL
345	Korea D.P.R.	Final protocol	PL
346	Tunisia	Final protocol	PL
347	Morocco	Final protocol	PL
348	Turkey	Final protocol	PL

•

No.	Origin	Title	Destination
349	United Kingdom	Final protocol	PL
350	Mexico	Final protocol	$_{\rm PL}$
351	C.8	R.5	PL
352	PL	Minutes of the eighth plenary meeting	PL
353	PL	Minutes of the ninth plenary meeting	PL
354	C.6	Summary record of the ninth and last meeting of Committee 6	C.6
355	United States	Document No. 248	$_{\rm PL}$
356	Morocco, Mauritania	Final protocol	PL
357	Japan	Final protocol	PL
358	Japan	Final protocol	$_{\rm PL}$
359	Japan	Final protocol	$_{\rm PL}$
360	Germany (F.R.of) Austria, Belgium, Canada, Denmark, United States, Finland, France, Ireland, Italy, Luxembourg, Monaco, Norway, Netherlands, United Kingdom, Sweden	Final protocol	PL
361	India	Final protocol	$_{\rm PL}$
362	Korea (Rep.of)	Final protocol	PL
363	SG	Second recapitulatory list of final protocols	PL
364	Spain	Final protocol	$_{\rm PL}$
365	USSR	Documents Nos. 272 and 273	PL
366	Algeria	Final protocol	$_{\rm PL}$
367	Algeria	Final protocol	$_{\rm PL}$
368	WG 5A	Notes concerning negative margins and protection ratios for the Broadcasting- Satellite Service within the Regions 1 and 3 Plan	C.5
369	German D.R.	Documents Nos. 272 and 273	PL

ž

3

ì

ŧ

_

No.	Origin	Title	Destination
370	C.8	R.6	PL
371	United States, France, United Kingdom	Document No. 365	PL
372	Germany (F.R. of)	Documents Nos. 365 and 369	PL
373	C.8	R.7	PL
374	C.4	Summary record of the twelfth meeting of Committee 4	с.4
375	с.4	Summary record of the thirteenth and last meeting of Committee 4	C.4
376	C.5	Summary record of the sixth meeting of Committee 5	C.5
377	C.5	Summary record of the seventh meeting of Committee 5	C.5
378	C.5	Summary record of the eighth meeting of Committee 5	C.5
379	C.5	Summary record of the ninth meeting of Committee 5	C.5
380	° C.5	Summary record of the tenth meeting of Committee 5	C.5
381	C.5	Summary record of the eleventh and last meeting of Committee 5	C.5
382	PL	Minutes of the tenth plenary meeting	PL
383	PL	Minutes of the eleventh plenary meeting	$_{\rm PL}$
384	$_{\rm PL}$	Minutes of the twelfth plenary meeting	PL
385	PL	Minutes of the thirteenth plenary meeting	PL
386	PL	Minutes of the closing meeting	-
387	SG	List of participants	-
388	SG	List of documents	-

ś

ŧ

Ś

Document No. 388-E Page 25

ANNEX 2

2

1

k

ţ

No.	Origin	Title	Destination
RR1	SG	Possible re-arrangement of the Radio Regulations and the Additional Radio Regulations	C.7
RR2	Canada	Possible re-arrangement of the Radio Regulations	C.7
RR3	SG	Report by the IFRB on the possible re-arrangement of the Radio Regulations and the Additional Radio Regulations	C.7
RR4	Mexico	Re-arrangement of the Radio Regulations	C.7
RR5	WG 7A	Draft - First report of Working Group 7A to Committee 7	C.7
RR6 + Corr	WG 7A	Report of Working Group 7Al to Committee 7	C.7
RR7	C.7	Report of Committee 7 to the Plenary Meeting	PL